# Period integrals of Kudla-Millson lifts and *L*-functions associated to vector valued modular forms



Ingmar Metzler

Cycling towards injectivity



## Period integrals of Kudla-Millson lifts and *L*-functions associated to vector valued modular forms

Vom Fachbereich Mathematik der Technischen Universität Darmstadt zur Erlangung des Grades eines Doktors der Naturwissenschaften (Dr. rer. nat.) genehmigte Dissertation

> von Ingmar Metzler, M.Sc. B.Sc. aus Darmstadt

Referent: Prof. Dr. Jan Hendrik Bruinier Korreferent: Prof. Dr. Nils Scheithauer

Darmstadt 2024

Ingmar Metzler: Period integrals of Kudla–Millson lifts and *L*-functions associated to vector valued modular forms Darmstadt, Technische Universität Darmstadt Tag der Einreichung: 30.08.2024 Tag der mündlichen Prüfung: 15.10.2024

Universitäts- und Landesbibliothek Darmstadt Auflage 1 Jahr der Veröffentlichung der Dissertation auf TUPrints: 2024 URN: urn:nbn:de:tuda-tuprints-287116

Die Veröffentlichung steht unter folgender Creative Commons Lizenz: Namensnennung – Weitergabe unter gleichen Bedingungen 4.0 International CC BY-SA 4.0 – https://creativecommons.org/licenses/by-sa/4.0/

This work is licensed under a Creative Commons License: Attribution-ShareAlike 4.0 International CC BY-SA 4.0 - https://creativecommons.org/licenses/by-sa/4.0/

Dieses Werk wurde mit Hilfe von  $\mathsf{KOMA}$ -Script und  $\mathsf{LAT}_{FX}$  gesetzt.

For my parents and my beloved wife.

#### Abstract

The present thesis establishes a new converse theorem for Borcherds products. Moreover, the injectivity of the Kudla–Millson theta lift is demonstrated in the O(n, 2) case in greater generality than is currently available in the literature. Both results are derived under the assumption of a single hyperbolic split of the base lattice. Additionally, symmetric square type *L*-series associated to elliptic vector valued modular forms are examined and special values of these series are linked to cycle integrals of Kudla–Millson liftings.

For vector valued elliptic modular forms, asymptotic bounds for their Fourier coefficients are tailored to the specific setting and the structure of their index set with respect to Hecke actions is analysed. Subsequently, symmetric square type L-series are associated to these forms which are realised via a Rankin–Selberg integral and meromorphic continuation is concluded. Further, formulae for the operation of Hecke algebras are extended and proven to imply product expansions of the aforementioned L-series.

In the orthogonal setting, the construction of special divisors is reviewed, before integrals of Kudla–Millson liftings over these divisors are explicitly related to special values of the aforementioned *L*-series by means of the Siegel–Weil formula. The relations that emerge in this context are then exploited in order to derive the injectivity of the Kudla–Millson lift and, by means of a duality statement, a converse theorem for Borcherds products.

#### Zusammenfassung

In der vorliegenden Arbeit wird ein neuer Umkehrsatz für Borcherdsprodukte bewiesen. Außerdem wird die Injektivität des Kudla-Millsons lifts im O(n, 2) Fall in größerer Allgemeinheit nachgewiesen als die aktuelle Literaturlage belegt. Beide Resultate werden unter der Annahme eines einzigen hyperbolischen Splits des zu Grunde liegenden Gitters gezeigt. Zusätzlich werden *L*-Reihen vom "symmetric square" Typ, welche zu vektorwertigen Modulformen assoziiert werden, untersucht und spezielle Werte derselben mit Zykelintegralen von Kudla-Millson lifts in Verbindung gebracht.

Für vektorwertige Modulformen werden asymptotische Schranken ihrer Fourierkoeffizienten an den Rahmen der Arbeit angepasst und die Struktur ihrer Indizes im Hinblick auf die Wirkung von Hecke Algebren untersucht. Nachfolgend werden diesen Formen die oben benannten *L*-Reihen zugeordnet und als Rankin–Selberg Integrale realisiert, woraus eine meromorphe Fortsetzung gefolgert wird. Außerdem gelingt es, Formeln für die Wirkung von Hecke Algebren zu erweitern und daraus Produktentwicklungen der zuvor konstruierten *L*-Reihen zu berechnen.

Im ortogonalen Fall wird die Konstruktion spezieller Divisoren dargelegt, um anschließend Integrale von Kudla–Millson Lifts darüber unter zu Hilfenahme der Siegel–Weil Formel in Termen spezieller *L*-Werte auszudrücken. Die sich hieraus ergebenden Relationen werden ausgenutzt, um die Injektivität des Kudla–Millson Lifts zu zeigen und, mit Hilfe eines Dualitätsresultats, einen Umkehrsatz für Borcherdsprodukte zu folgern.

#### Danksagung

Viele Menschen aus meinem Umfeld haben durch ihre fortwährende Unterstützung zu dieser Dissertation beigetragen. Ich möchte Euch allen gerne dafür danken!

Ganz besonders danken möchte ich meinem Doktorvater Jan Hendrik Bruinier. Er hat dieses spannende Thema für meine Arbeit vorgeschlagen und mich durch seinen Weitblick und seinen Erfahrungsschatz durch die Herausforderungen der vorliegenden Arbeit gelenkt. Ohne seine kontinuierliche Unterstützung wäre diese Arbeit nicht möglich gewesen.

Außerdem möchte ich gerne Nils Scheithauer danken. Während meiner Promotion hatte er ein offenes Ohr für die Herausforderungen meiner Arbeit und hat durch seine Vorschläge häufig neue Perspektiven eröffnet, um diese anzugehen. Ich möchte ihm zudem dafür danken, dass er einen Promotionsausschuss in den Semesterferien angeboten hat, um mir neben dem Zusammenschreiben noch Zeit mit meine Tochter einzuräumen.

Besonderer Dank geht auch an Markus Schwagenscheidt. Die mathematischen Diskussionen mit ihm waren besonders motivierend um Ergebnisse im Rahmen meiner Arbeit zu verbessern und ich danke ihm dafür, dass er mir mit seinem Erfahrungsschatz und gezielten inhaltlichen Vorschlägen dabei geholfen hat, die zahlreichen Puzzleteile meiner Arbeit zu sortieren.

Ich möchte gerne meiner Familie danken, welche mich durch meine Promotion begleitet und durch ihre kontinuierliche Unterstützung im Alltag zum Gelingen meiner Arbeit beigetragen hat. Ganz besonders möchte ich dabei meinen Eltern Volkmar und Susanne Metzler und meiner Frau Leonie danken. Meiner Mutter Susanne gebührt besonderer Dank dafür, dass sie die zeitintensive Aufgabe wahrgenommen hat, diese lange Arbeit korrekturzulesen. Meiner Frau möchte ich zudem neben all ihrer Unterstüzung explizit dafür danken, dass sie die zahllosen Stunden, welche ich am Wochenende für diese Arbeit aufbrachte, akzeptierte, sowie für ihre Bemühungen meine Kenntnisse auf dem Gebiet der Ornithologie zu erweitern.

Ein Dank geht auch an Paul Kiefer und Riccardo Zuffetti für regelmäßige hilfreiche mathematische Diskussionen und inhaltliche Hinweise, auch über den Kontext dieser Arbeit hinaus, sowie Korrekturen zur vorliegenden Arbeit. Oliver Stein möchte ich gerne für konstruktive Hinweise zu Hecke Operatoren danken.

Danken möchte ich außerdem Patrick Holzer und Felix Pennig für Korrekturen zu Teilen dieser Arbeit.

Der Autor wurde durch den LOEWE-Schwerpunkt USAG und das DFG-Projekt GAUS im Rahmen dieser Arbeit unterstützt.

## Contents

0	Introduction			
I.	Pr	Preliminaries		
1	Fun	Idamentals		
	1.1	Quadratic spaces and lattices	13	
	1.2	$p\text{-adic}$ numbers and the adele ring $\hdots \hdots \hd$	22	
	1.3	Dual groups and Fourier analysis	25	
	1.4	Hilbert symbol	30	
	1.5	Quadratic field extensions	32	
2	Clas	assical modular forms		
	2.1	Introduction	41	
		2.1.1 A geometric approach	43	
		2.1.2 Constructions and generalisations	55	
	2.2	Elliptic modular forms	64	
П	Automorphic forms			
3	Sym	Symplectic theory		
	3.1	Symplectic modular forms	75	
	3.2	Weil representation	78	
		3.2.1 Classical Weil representation and relation	84	
	3.3	Vector valued modular forms	90	
		3.3.1 Discrete Weil representation	91	
		3.3.2 Some constructions	93	
		3.3.3 Indices of Fourier coefficients	113	
		3.3.4 Bounds on Fourier coefficients	121	

4	Ortł	nogonal	l theory		129			
	4.1	Group properties						
		4.1.1	Orthogonal groups		129			
		4.1.2	Spin groups		132			
	4.2	Modula	ar varieties	•	141			
		4.2.1	Models	•	141			
		4.2.2	Modular groups	•	146			
		4.2.3	Boundary components		150			
		4.2.4	Modular forms		151			
		4.2.5	Special divisors		157			
5	Automorphic forms on SL <sub>2</sub> 16							
	5.1	Prelim	inaries		161			
	5.2	Theta o	distributions		164			
		5.2.1	The classic setting		165			
	5.3	Eisenst	tein series	•	166			
		5.3.1	Principal series representations		167			
		5.3.2	Eisenstein series		171			
	5.4	Compa	aring theta and Eisenstein series		175			
		5.4.1	Siegel–Weil formula		175			
		5.4.2	Kudla–Millson Schwartz form	•	177			
		5.4.3	A geometric integral	•	179			
					102			
	Int	ertwin	ing worlds		183			
6	<i>L</i> -series							
	6.1	Introdu	uction	•	185			
	6.2	Classic	al Hecke theory	•	189			
		6.2.1	Dividing the level		204			
	6.3	Vector	valued Hecke theory		210			

	6.2.1	Dividing the level	204
6.3	Vector	valued Hecke theory	210
	6.3.1	Dividing the level	216
6.4	<i>L</i> -func	tions of modular forms	232
	6.4.1	$L\text{-}series$ of vector valued modular forms $\hdots \hdots \$	232
	6.4.2	Product expansions	238
	6.4.3	Rankin–Selberg L-series	255

7	Automorphic lifts				
	7.1	Borche	erds Lift	267	
	7.2	2 Kudla–Millson lift			
		7.2.1	The Kudla–Millson Schwartz form	270	
		7.2.2	The Kudla–Millson lift	272	
		7.2.3	Linking lifts	275	
	7.3	Injectiv	vity of a theta lift	277	
		7.3.1	Cycle integrals of Kudla-Millson lifts	278	
		7.3.2	Injectivity for a hyperbolic split	283	
IV	IV Appendix				
Α	A Supplementing proofs				

	A.1	Lattices	293					
	A.2	Miscellaneous	298					
	A.3	Relations of Hecke operators	300					
	A.4	Gauss sums	304					
	A.5	Whittaker functions and asymptotics	308					
В	Further background material							
	B.1	Some categorical notions	313					
	B.2	Representation theory	316					
	B.3	Symplectic and metaplectic groups	319					
	B.4	Figures	331					
Index								
Gl	Glossary							
Bi	Bibliography 3							

## 0 Introduction

#### A converse theorem for Borcherds products

In 1998, Richard Borcherds was awarded the fields medal in recognition of his pioneering contributions to the field of automorphic forms and mathematical physics [OR]. He had introduced vertex operator algebras as well as a multiplicative lift of automorphic forms and employed these tools to prove the Conway–Norton moonshine conjecture. The constructed lift is nowadays referred to as *Borcherds lift* and maps vector valued elliptic modular forms to orthogonal automorphic forms with infinite product expansions. These are also referred to as *Borcherds products* which provide a diverse array of examples of orthogonal modular forms, a fact that was instrumental in the jury's decision to bestow the fields medal upon Borcherds. In this thesis we prove a converse theorem for Borcherds products, demonstrating that all orthogonal modular forms that may a priori be targeted by the Borcherds lift are, in fact, in its range up to a nonzero constant factor.

To be more explicit, let (L, q) be an even lattice of signature  $(m^+, 2)$  and  $\mathbb{D}$  be the associated Grassmannian which carries a natural complex structure. The natural arithmetic subgroup to consider is the *discriminant kernel*  $\Gamma(L) \leq O(L)$  which is a finite index subgroup stabilising the discriminant group of L. On the elliptic side, let  $\mathcal{L} = L'/L$  and inspect holomorphic modular functions  $f : \mathbb{H} \to \mathbb{C}[\mathcal{L}]$  of weight  $k \in \mathbb{Z}/2$ , transforming with the dual Weil representation  $\overline{\rho}_L$  which may possess poles at the cusp. This space is referred to as the space of *weakly holomorphic modular forms* and denoted by  $\mathcal{M}^!_{L^-,k}$ . Let  $f \in \mathcal{M}^!_{L^-,k}$  and recall that such a function has a Fourier expansion

$$f(\tau) = \sum_{\mu \in L'/L} \sum_{m \in \mathbb{Z} - \mathbf{q}(\mu)} a(\mu, m) e^{2\pi i \tau m} \mathbf{e}_{\mu}$$
(0.1)

where  $\mathfrak{e}_{\mu}$  denotes the standard basis vector of  $\mathbb{C}[\mathcal{L}]$  being 1 in the  $\mu$  component and 0 elsewhere. Assume f has integral principal part, meaning that  $a(\mu, m) \in \mathbb{Z}$  for all m < 0. Then the Borcherds lift associates to f a meromorphic modular form  $\Psi(z, f)$  on  $\mathbb{D}$  for  $\Gamma(L)$  with unitary multiplier system of finite order such that

- a) the weight of  $\Psi(\cdot, f)$  is given by a(0,0)/2,
- b) the divisor of  $\Psi(\cdot, f)$  is equal to

$$\frac{1}{2} \sum_{\mu \in L'/L} \sum_{n>0} a(\mu, -n) Z(\mu, n), \qquad (0.2)$$

where

$$Z(\mu, n) = \sum_{\substack{\lambda \in L + \mu \\ \mathbf{q}(\lambda) = n}} \{ z \in \mathbb{D} \mid z \perp \lambda \}$$

is the *Heegner divisor* of discriminant  $(\mu, n)$  (cf. Definition 4.70),

c) the target form  $\Psi(\cdot, f)$  has explicit infinite product expansions.

As previously stated, this lift gives rise to a variety of new orthogonal modular forms and also to the monster denominator formula, which is utilised to prove the moonshine conjecture. In [Bor98, Pr. 16.10] Borcherds himself poses the question whether his lift may be reversed. That is to say, given a meromorphic modular form F for  $\Gamma(L)$  whose zeros and poles are supported on special divisors  $Z(\mu, n)$ , is there a form  $f \in \mathcal{M}^!_{L^-,k}$  such that  $\Psi(z, f)$  equals F up to a constant factor? In this thesis we provide the following answer (cf. Theorem 7.19).

**Theorem 0.1.** Assume that L splits a hyperbolic plane and  $m^+ > 3$ . Then every meromorphic modular form F with respect to  $\Gamma(L)$  whose divisor is a linear combination of special divisors  $Z(\mu, n)$  is (up to a nonzero constant factor) the Borcherds lift  $\Psi(z, f)$ of a weakly holomorphic modular form  $f \in \mathcal{M}^{!}_{L^{-}, 1-m^{+}/2}$ .

The case of  $m^+ = 3$  is also addressed in Section 7.3. More specifically, the same result holds for Witt rank 1 of the lattice L and a weaker version for Witt rank 2. The statement above improves upon the previously strongest result [Bru14, Thm. 1.2], in that only one hyperbolic split is required and no transition to a sublattice is needed. In order to prove Theorem 0.1, a method developed by Bruinier [Bru02] is utilised to characterise the above converse statement in terms of the injectivity of another theta lift, the *Kudla–Millson* lift.

#### The Kudla–Millson lift

In the 1980s Kudla and Millson [KM86] introduced special Schwartz forms  $\varphi_{\rm KM}$  on the symmetric spaces attached to the classical groups  $O(m^+, m^-)$ ,  $U(m^+, m^-)$ , and  $\text{Sp}(m^+, m^-)$ , taking values in closed differential forms. Their principal objective was to investigate cohomology classes of special cycles by means of a theta correspondence, generalising the celebrated work of Hierzebruch and Zagier [HZ76] on Hilbert modular surfaces. More precisely, in the orthogonal case, Kudla and Millson symmetrised the Schwartz forms  $\varphi_{\text{KM}}$  over a base lattice L and let the Weil representation of the symplectic group act on  $\varphi_{\text{KM}}$  in order to obtain a kernel (cf. Subsection 7.2.1 for details)

$$\Theta_{\mathrm{KM}}(\tau, z), \quad \tau \in \mathbb{H}, \quad z \in \mathbb{D}$$

in a symplectic and an orthogonal variable, which transforms automorphically in both variables. These may then be employed as an integral kernel to shift automorphic objects from the symplectic to the orthogonal setting or vice versa. Such an association is referred to as a *theta lift*. The authors continued their work in [KM87] in great generality and we recommend [KM90] and [BF04] as an introductory source. The applications of the Kudla–Millson theta correspondence range from the study of the cohomology of orthogonal and unitary Shimura varieties [KM86] and Arakelov theory of Shimura varieties [Kud04a] over specific counting problems [EGS23], constructing mock modular forms and higher dimensional error functions [FK17] to proving a converse theorem for Borcherds products [Bru02] [Bru14].

In this thesis, we consider the Kudla–Millson lift on the space of vector valued elliptic modular forms associated to the Weil representation  $\rho_L$  of an even lattice L of signature  $(m^+, 2)$ . To be more explicit, let  $k = \operatorname{rank}(L)/2$  and  $S_{L,k}$  denote the space of cusp forms for the Weil representation  $\rho_L$ . Then

$$\Lambda_{\mathrm{KM}}: \mathcal{S}_{L,k} \to \mathcal{H}^{(1,1)}(Y_L), \quad f \mapsto \int_{\mathrm{Mp}_2(\mathbb{Z}) \setminus \mathbb{H}} \langle f, \Theta_{\mathrm{KM}} \rangle \operatorname{Im}^k \, \mathrm{d}\mu$$

defines a linear map to the square integrable harmonic differential forms of Hodge type (1,1) on the variety  $Y_L = \Gamma(L) \setminus \mathbb{D}$  and is referred to as the *Kudla–Millson lift*. Here  $\mu$  denotes the hyperbolic measure on the upper half-plane  $\mathbb{H}$  on which Mp<sub>2</sub>( $\mathbb{Z}$ ) acts via Möbius transforms. The question of its injectivity already arose in [KM90] and may be used to compute the rational Picard number of the underlying Shimura variety [Ber+16], as well as to derive properties of cones generated by special cycles [BM19] [Zuf22]. However, the main application we have in mind is the converse Theorem 0.1 for Borcherds products. There have been multiple results on the injectivity of the Kudla–Millson lift over the past two decades presented in [Bru02, Thm. 5.12 p. 139], [BF10, Cor. 4.11 p. 37], [Bru14, Thm. 5.3 p. 331], [Zuf24, Thm. 6.1 p. 22], and [Ste23, Cor. 7.8 p. 25]. These results are based on three fundamentally different methods which are sketched in Section 7.2. While there has been a recent success in unifying the results above by Zuffetti and the author [MZ23, Thm. 6.2 p. 24] by generalising Zuffetti's method, this has not shed any new light on the converse theorem for Borcherds products. The most potent result<sup>1</sup> for this application is Theorem 7.10 by Bruinier which provides injectivity under the assumption that the lattice L splits a hyperbolic plane as well as a scaled hyperbolic plane.

In the context of this thesis, we present a new approach to proving the injectivity of the Kudla–Millson lift by means of computing cycle integrals on the orthogonal variety. This approach eliminates the need for a scaled hyperbolic split and leads to the following result (cf. Theorem 7.16).

**Theorem 0.2.** Let (L, q) be an even lattice of signature  $(m^+, 2)$  with  $m^+ > 3$ . Assume that L splits a hyperbolic plane. Then the Kudla–Millson lift  $\Lambda_{\text{KM}} : S_{L,1+m^+/2} \to \mathcal{H}^{(1,1)}(Y_L)$  is injective.

The case of  $m^+ = 3$  is also discussed in Subsection 7.3.2 and proven for the lattice L having Witt rank 1. It should be stressed that in the case of no hyperbolic split, Bruinier constructed a subspace of the kernel and demonstrated that it is in general non-trivial even in the simple instance of lattices of prime level [Bru14, Sec. 6.1].

The general proof strategy is to compute special cycle integrals of Kudla–Millson liftings  $\Lambda_{\text{KM}}(f)$  of cusp forms  $f \in S_{L,k}$ . For suitably chosen cycles, these may be expressed in terms of special values of certain *L*-series L(f,s) associated to the form f. If  $\Lambda_{\text{KM}}(f)$  vanishes, its cycle integral must vanish as well, yielding roots of the aforementioned *L*-series. As a consequence, a relation is established between the Fourier coefficients of f appearing in the construction of L(f,s). This ultimately permits the deduction of their vanishing, which renders the initial form to vanish as well. Since the Kudla–Millson lift is linear, injectivity is inferred. Nonetheless, the appearing cycle integrals are interesting in their own right and we proceed with describing their computation below.

#### Cycle integrals

We describe how certain cycle integrals of Kudla–Millson liftings may be expressed in terms of special values of L-series. In order to construct these special cycles, note that for

<sup>&</sup>lt;sup>1</sup>This result is, in fact, identical to [MZ23, Thm. 5.1 (i) p. 19] in this particular signature.

 $\ell \in L$  of positive norm, there is a subvariety of the Grassmannian of the same type

$$\mathbb{D}_{\ell} = \{ x \in \mathbb{D} \mid x \subseteq \ell^{\perp} \} \subset \mathbb{D}.$$

When considering the stabiliser  $\Gamma_{\ell}$  of  $\ell$  in  $\Gamma(L)$ ,

$$\mathcal{Z}(\ell) := \Gamma_{\ell} \backslash \mathbb{D}_{\ell} \to \Gamma(L) \backslash \mathbb{D} = Y_L$$

defines a (in general relative) cycle of  $Y_L$ . Completing  $\Lambda_{\text{KM}}(f)$  with an adequate power of a Kähler form  $\Omega$  allows for integration over  $\mathcal{Z}(\ell)$ :

$$\int_{\mathcal{Z}(\ell)} \Lambda_{\mathrm{KM}}(f) \wedge \Omega^{m^+ - 2}.$$

For the sake of simplicity, assume throughout the introduction that  $L = \mathbb{Z}\ell \oplus K$  splits for some lattice K of signature  $(m^+ - 1, 2)$ . Then the theta series  $\Theta_{\text{KM}} = \Theta_{\mathbb{Z}\ell} \otimes \Theta_{\text{KM},K}$ splits as a tensor product on  $\mathbb{D}_{\ell}$  which, neglecting convergence issues, allows for isolating the integral over the divisor  $\mathcal{Z}(\ell)$  as follows.

$$\int_{\mathcal{Z}(\ell)} \Lambda_{\mathrm{KM}}(f) \wedge \Omega^{m^{+}-2} = \int_{Z(\ell)} \int_{\mathrm{Mp}_{2}(\mathbb{Z}) \setminus \mathbb{H}} \langle f, \Theta_{\mathrm{KM}} \rangle \operatorname{Im}^{k} \, \mathrm{d}\mu \wedge \Omega^{m^{+}-2}$$
$$= \int_{\mathrm{Mp}_{2}(\mathbb{Z}) \setminus \mathbb{H}} \sum_{\substack{\lambda \in (\mathbb{Z}\ell)'/(\mathbb{Z}\ell) \\ \delta \in K'/K}} f_{\lambda \oplus \delta} \cdot \overline{\theta_{\mathbb{Z}\ell,\lambda}} \int_{\mathcal{Z}(\ell)} \overline{\theta_{\mathrm{KM},K,\delta}} \wedge \Omega^{m^{+}-2} \operatorname{Im}^{k} \, \mathrm{d}\mu.$$
(0.3)

Here, subscripts in  $\lambda$  and  $\delta$  indicate components of the corresponding vector valued modular forms. The inner integral  $\int_{\mathcal{Z}(\ell)} \overline{\theta_{\mathrm{KM},K,\delta}} \wedge \Omega^{m^+-2}$  may be expressed in terms of an adelic standard theta integral which is suitable for an application of the Siegel–Weil formula<sup>2</sup>. Effectively, it is replaced by an Eisenstein series. Retranslating this to the classical setting reveals the expression as a scalar product

$$\int_{\mathcal{Z}(\ell)} \Lambda_{\mathrm{KM}}(f) \wedge \Omega^{m^{+}-2} \stackrel{\cdot}{=} \int_{\mathrm{Mp}_{2}(\mathbb{Z}) \setminus \mathbb{H}} \left\langle f, \Theta_{\mathbb{Z}\ell} \otimes E_{K,k-1/2} \right\rangle \mathrm{Im}^{k} \, \mathrm{d}\mu. \tag{0.4}$$

Here,  $E_{K,k-1/2}$  denotes an Eisenstein series of weight k - 1/2 associated to the lattice K. Note that  $E_{K,k-1/2}$  is a Poincaré series, technically offering the possibility of unfolding. Selecting a primitive element  $\ell_0 \in (\mathbb{Z}\ell)'$  and performing a subsequent calculation yields

<sup>&</sup>lt;sup> $^{2}$ </sup>Compare Subsection 5.4.1 for details about its application.

the following special L-value (cf. Theorem 7.14).

**Theorem 0.3.** Let (L,q) be an even lattice of signature  $(m^+, 2)$ , set  $k = 1 + m^+/2$ , let  $f \in S_{L,k}$  and select some primitive  $\ell_0 \in L'$  of positive norm. If  $m^+ > 3$ , then

$$\int_{\mathcal{Z}(\ell_0)} \Lambda_{\mathrm{KM}}(f) \wedge \Omega^{m^+ - 2} = C \cdot \sum_{n \in \mathbb{N}} \frac{a(n\overline{\ell_0}, n^2 \operatorname{q}(\ell_0))}{n^{m^+}},$$

where  $a(\lambda, n)$  denote the Fourier coefficients of the initial form  $f \in S_{L,k}$  (cf. (0.1)) and C is an explicit constant.

Before analysing the *L*-series presented above, it is imperative to elucidate the simplifications made previously and delineate an approach to the general case. Firstly, for general  $\ell \in L$  such that the lattice *L* does not split the lattice  $\mathbb{Z}\ell$ , a splitting sublattice  $M = \mathbb{Z}\ell \oplus K \leq L$  is considered. Subsequently, by applying lifting operators between vector valued modular forms to the respective discriminant forms, we may replace *f* by another form  $\uparrow f$  to the discriminant form of the sublattice *M*, thus resulting in the theta function splitting once more as a tensor product. In order to relate the inner integral in (0.3) over the special cycle  $\mathbb{Z}(\ell)$  to a standard theta integral and apply the Siegel–Weil formula, it is necessary to pass to a covering of the cycle and rewrite it in terms of the GSpin setting (compare Subsection 7.3.1 for details). Another obstacle is that the integral resulting from unfolding (0.4) is not convergent which is circumvented by passing to an analytic continuation and verifying that the function agrees with the resulting *L*-series on a right half-plane, thereby extending the identity to the edge of convergence. Ultimately, the coefficients of the initial form *f* must be recovered from the expression involving  $\uparrow f$ .

#### Symmetric square L-series

The *L*-series appearing in Theorem 0.3 are of independent interest and are analogous to symmetric square *L*-series in the scalar valued case. The latter have already been investigated by Shimura playing a key role in establishing the *Shimura lift* [Shi73] and led to further investigations in [Shi75]. When extending the construction of these *L*-series in Shimura's work to the vector valued setting, a straightforward approach is to associate a certain *L*-series to every component function of  $f \in \mathcal{M}_{L,k}$ , meaning for fixed  $\mu \in L'/L$  in (0.1). However, upon closer examination it becomes evident that their natural analogues in the setting of vector valued modular forms intertwine the components. For the sake of simplicity, we present only a special instance of the series discussed in the main body in Subsection 6.4.1. Namely, for a holomorphic vector valued modular form  $f \in \mathcal{M}_{L,k}$  with Fourier expansion as in (0.1) and an anisotropic vector  $\ell \in L'$ , we define

$$L_{\ell}(f,s) = \sum_{n \in \mathbb{N}} \frac{a(n\overline{\ell}, n^2 \operatorname{q}(\ell))}{n^s}$$
(0.5)

for  $s \in \mathbb{C}$ . In Subsection 3.3.4 we gather bounds of Fourier coefficients from existing literature and adjust these for our intended purpose. This is done in order to determine a right half-plane on which  $L_{\ell}(f, s)$  converges absolutely with respect to the parameter s, thereby defining a holomorphic function. This is always the case for  $\operatorname{Re}(s) > k + 1/2$ and we refer to Corollary 6.82 for sharper bounds. In Subsection 6.4.3 we proceed by realising the *L*-series in (0.5) as a Rankin–Selberg convolution and prove the convergence and analyticity of its unfolded counterpart in a certain region (cf. Proposition 6.97). Furthermore, we conclude that it admits meromorphic continuation to the whole complex plane and note that a symmetry property is inherited from an Eisenstein series. In fact, the main body of the thesis treats much more general *L*-series than the instance presented in (0.5).

A key feature of the *L*-series investigated by Shimura is that these admit infinite product expansions provided they are induced by Hecke eigenforms. This property may be replicated in the vector valued setting. In [BS08], Bruinier and Stein develop a theory of Hecke operators for vector valued modular forms, presenting a construction that is suitable for concrete computations. Additionally, the action of a variety of Hecke operators on Fourier coefficients of vector valued modular forms is also already presented in [BS08, Prop. 4.3 p. 258]. To be more precise, for a prime p not dividing the level of L and an eigenform  $f \in \mathcal{M}_{L,k}$  for the Hecke operator  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$ , there exists a constant C such that

$$\sigma_p a(\lambda, n) = p^{2k-2} a(\lambda/p, n/p^2) + C \cdot a(\lambda, n) + a(p\lambda, p^2n).$$

$$(0.6)$$

Here,  $a(\lambda, n)$  denote the Fourier coefficients of f as in (0.1). The recursion relation (0.6) is suitable for eliminating the prime p from the index n in (0.5). However, the case of primes p dividing the level of the lattice L is not contained in the source. Therefore, we derive a recursion formula analogous to (0.6) for primes dividing the level of the lattice L in Proposition 6.71 based on [Ste15]. However, in the general case, the recursion formula has ambiguities due to the phenomenon of torsion in the discriminant form. For maximal lattices, though, these ambiguities are resolved, resulting in a perfect recursion

for certain forms. In this case, we derive an infinite product expansion for *L*-series as in (0.5). Simplifying, compared to the main body of the thesis, results in the following statement (cf. Corollary 6.92).

**Theorem 0.4.** Assume L to be maximal with  $2 \nmid \text{lev}(L)$  and select a simultaneous eigenform  $f \in S_{L,k}$  of all  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$  that is invariant under O(L'/L). Write

$$f(\tau) = \sum_{\mu \in \mathcal{L}} \sum_{m \in \mathbb{Z} + \mathbf{q}(\mu)} a(\mu, m) e^{2\pi i \tau m} \mathbf{e}_{\mu}$$

for the Fourier expansion of f. For an element  $\ell \in L'$  such that  $lev(L) \cdot q(\ell)$  is square free writing  $(\lambda, t) := (\bar{\ell}, q(\ell)) \in \mathcal{L} \times \mathbb{Q}^{\times}$  yields the following product expansion

$$\begin{split} &\sum_{n \in \mathbb{N}} \frac{a(n\lambda, n^2 t)}{n^s} \\ &= a(\lambda, t) \cdot \prod_{p \nmid \text{lev}(L)} \frac{1 + \delta_{p \nmid t} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} \cdot p^{k-1-s}}{1 - \left(\frac{\sigma_p}{p^{k-1}} - (1 - p^{-1}) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right) \cdot p^{k-1-s} + p^{2(k-1-s)}}{\frac{1 + K_{L,p} \left\{ \delta_{\lambda_p \neq 0}(1 - p^{-1}) + \delta_{\lambda_p = 0} \cdot \delta_{p \nmid t}, \quad 2 \mid R_p \\ -\delta_{\lambda_p = 0} \cdot p^{-1/2} \left(\frac{-t}{p}\right) \epsilon_p, \qquad 2 \nmid R_p \right.}{1 - \left(\frac{\sigma_p}{p^{k-1}} - \delta_{2 \mid R_p}(1 - p^{-1}) K_{L,p}\right) p^{k-1-s} + p^{2(k-1-s)}}}. \end{split}$$

Here,  $G_{\mathcal{L}}$  and  $K_{L,p}$  are certain Gauss sums,  $\lambda_p$  denotes the projection of  $\lambda \in \mathcal{L}$  to its p-component in  $\mathcal{L}_p$ ,  $C(\lambda_p)$  is a specified integer vanishing if  $\lambda_p = 0$ ,  $\chi_{\mathcal{L}}$  is a quartic character and  $\epsilon_p$  equals 1 or i. Further, the quantity  $R_p$  is given in Definition 1.38.

In this context, we prove absolute convergence of the infinite product expansions as well as non-vanishing of the rational factors in  $p^{-s}$  within a specified range.

#### Injectivity of the Kudla–Millson lift

Turning to the question of injectivity, we recall that assuming  $f \in S_{L,k}$  lying in the kernel of the Kudla–Millson lift renders the special *L*-value in Theorem 0.3 to vanish. More explicitly, for any primitive  $\ell_0 \in L'$  of positive norm, the *L*-series in (0.5) admits the root  $L_{\ell_0}(f, m^+) = 0$ . In conclusion, we have established relations stating that certain weighted sums of Fourier coefficients of any form f lying in the kernel of  $\Lambda_{\rm KM}$  must equal zero. This will be exploited to prove that the form f had to vanish in the first place, thereby implying the triviality of the kernel of  $\Lambda_{\rm KM}$ . In the case of the lattice L splitting a hyperbolic plane, this is possible with an elegant trick. The key idea is to represent  $L_{\ell_0}(f, s)$  as a recombination of subseries of the same type through an inclusion-exclusion argument. The assumption regarding the lattice Lthen guarantees that a substantial amount of these subseries vanish at  $m^+$ , effectively thinning out the original series  $L_{\ell_0}(f, m^+)$ . By exploiting absolute convergence of the L-series it will be concluded that the first coefficient of the series in (0.5) had to vanish, which is exactly the Fourier coefficient  $a(\overline{\ell_0}, q(\ell_0))$  of the form f. However, the set of Fourier coefficients representable in this fashion is, again due to the hyperbolic split, exhaustive, implying the vanishing of the initial form f. The details of this reasoning are presented in Subsection 7.3.2 and result in Theorem 0.2 above. Finally, an application of [Bru14, Thm. 4.2 p. 330] yields Theorem 0.1.

## Part I

## **Preliminaries**

In this part, the basics of the theory are provided before turning the attention to the theory of automorphic forms which lie at the heart of this thesis, as well as, the subject of modern number theory as a whole.

### **1** Fundamentals

For the sake of efficiency, the following content is not provided in chronological order. However, the majority of fundamental concepts that are not typically included in a regular Master's programme, although essential for comprehending this thesis, are addressed in the subsequent sections. Part of the following content has already been discussed in my Master's thesis and is restated here for the sake of convenience.

#### 1.1 Quadratic spaces and lattices

The following content is primarily found in [Ehl03, Sec. 1.1], [Cas78, Sec. 7], [Bos08], [Kne02], [Lam73], [Lam05], and [Joh98]. Let R be a unitary, commutative, associative ring and M be a finitely generated R module and Fr(R) the fraction field if R is integral.

**Definition 1.1.** A quadratic form is a map  $q: M \to R$  such that

(i) 
$$q(rx) = r^2 q(x)$$
 for all  $r \in R$  and  $x \in M$ ,

(ii)  $M \times M \ni (x, y) \mapsto \mathbf{b}(x, y) \coloneqq \mathbf{q}(x + y) - \mathbf{q}(x) - \mathbf{q}(y)$  is R bilinear.

Further, if R is integral, we allow  $q: M \to K$  for a field K/Fr(R).

- a) The form b is called *associated bilinear form* (to q).
- b) The pair (M, q) is called *quadratic module* over R. In case R = F is a field, it is called *quadratic space* and in this case, M is denoted by V.
- c) For a basis  $(m_i)$  of a free module M, the matrix  $S = (b(m_i, m_j))_{i,j}$  is called a *Gram* matrix and  $det(M) \coloneqq det(q) \coloneqq det(S) \in R/(R^{\times})^2$  the determinant of M or b or q.
- d) If  $det(q) \in R^{\times}$  (or  $\in K^{\times}$ ), the form q or module (M, q) is called *regular*.

Example 1.2. Let F be a field of characteristic different from 2 and  $V = F^n$ , then

$$V \ni x \mapsto x^T \operatorname{diag}(a_1, \dots, a_n) x \in F$$
 (1.1)

defines a quadratic form, where  $a_i \in F$ .

There is a notion of morphisms and a natural process of constructing new quadratic modules from known modules.

**Definition 1.3.** Let  $(M_1, q_1)$ ,  $(M_2, q_2)$  be quadratic modules over R mapping to the same ring.

- a) A morphism  $\sigma : (M_1, q_1) \to (M_2, q_2)$  of quadratic modules is an R module homomorphism  $\sigma : M_1 \to M_2$  translating quadratic forms  $q_1 = q_2 \circ \sigma$ . An injective/bijective morphism is called *isometry/isomorphism*.
- b) Define  $(M_1, q_1) \oplus (M_2, q_2)$  as the quadratic module (M, q) with  $M \coloneqq M_1 \oplus M_2$  as R modules and  $q: (m_1, m_2) \mapsto q_1(m_1) + q_2(m_2)$ .
- c) For  $r \in R$ , we refer to  $(M_1, r \cdot q_1)$  as the scaled lattice  $M_1(r)$ .

Example 1.4. For  $y \in (M, q)$  with  $q(y) \in R^{\times}$  the reflection at y is defined as the map  $\tau_y : m \mapsto m - b(m, y) q(y)^{-1}y$  and determines an isometry. It is involutive, maps y to -y, and any  $m \in M$  with b(m, y) = 0 to itself.

If in Example 1.2 an element  $a_j$  was 0 for some j, the form would be considered on a smaller space where  $a_i \in F^{\times}$  for all indices i. In fact, any quadratic space is isometrically isomorphic to a form as in (1.1). As a consequence, regularity may be assumed in general. Further, multiplying  $a_i$  with a nonzero square does not alter the isomorphism type of the space (V, q), so that  $a_i$  are regarded as elements of  $\mathbb{F}^{\times}/(\mathbb{F}^{\times})^2$ . Over rings like  $\mathbb{Z}$  the situation is more delicate (cf. Remark 1.13).

Example 1.5. In case of  $F = \mathbb{R}$  and  $V = \mathbb{R}^m$ , every regular form q is isomorphic to

$$x \mapsto x^T \operatorname{diag}(\underbrace{1, \dots, 1}_{r \text{ times}}, -1, \dots, -1)x,$$

where the tuple  $(r, m - r) =: (m^+, m^-)$  is called *signature* or *type* of (V, q) and is a complete invariant. We shall denote (V, q) by  $\mathbb{R}^{(m^+, m^-)}$ .

**Definition 1.6.** Let (M, q) be a quadratic module.

- a) For a subset  $N \subseteq M$  we refer to  $N^{\perp} := \{x \in M \mid b(x, y) = 0 \text{ for all } y \in N\}$  as the *orthogonal complement* of N.
- b) Further, (M, q) is called *non-degenerate*, if  $M^{\perp} = \{0\}$ .
- c) An element  $0 \neq x \in M$  is called *isotropic*, if q(x) = 0. We refer to (M, q) as being *isotropic* if it possesses an isotropic element and *anisotropic*, otherwise.

d) If (V, q) is a quadratic space, the dimension of a maximal totally isotropic subspace is called *Witt index* or *Witt rank* and is an invariant of (V, q).

Example 1.7. The standard isotropic regular module has Gram matrix  $\begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}$ . It is denoted by H and called *hyperbolic plane*. If  $e_1, e_2$  denotes the standard basis, we have  $e_i^{\perp} = \langle e_i \rangle$ , i.e.  $e_i$  is isotropic, and  $(e_1 + e_2)^{\perp} = \langle e_1 - e_2 \rangle$ , rendering  $H \simeq \begin{pmatrix} 2 & 0 \\ 0 & -2 \end{pmatrix}$  provided  $2 \in \mathbb{R}^{\times}$ . Note that the Witt index of H is 1.

**Remark 1.8.** Any regular quadratic space splits into an orthogonal sum of an anisotropic space and hyperbolic planes, a fact known as *Witt's decomposition theorem* (cf. [Lam05, I.4 Thm. 4.1 p. 12]) rendering the Witt index r well defined as the number of hyperbolic planes. Also compare Theorem 4.6.

**Definition 1.9.** Let (V, q) be a quadratic space over F with basis  $(v_i)_{i=1}^m$  and R be a unital subring of F.

- a) The *R* module  $L = \sum_{i=1}^{m} Rv_i$  together with q is called *quadratic R lattice*,  $(v_i)_{i=1}^{m}$  is called *lattice basis* of *L* and *m* the *rank* of *L*. Equivalently, an *R* lattice *L* may be defined as an *R* submodule of (V, q) such that  $L \otimes_R F \simeq V$ .
- b) Further, define the discriminant disc $(L) := (-1)^{\frac{m^2-m}{2}} \det(L)$  of L.

Note that the determinant of a quadratic lattice is independent of the chosen basis if viewed as an element of  $F^{\times}/(F^{\times})^2 \cup \{0\}$ .

Example 1.10. Prominent examples for lattices in this thesis are  $\mathbb{Z}$  lattices in quadratic spaces over  $\mathbb{Q}$  or  $\mathbb{Z}_p$  lattices in quadratic spaces over  $\mathbb{Q}_p$ . In fact, every  $\mathbb{Z}$  lattice in a quadratic  $\mathbb{Q}$  vector space V lifts to a  $\mathbb{Z}_p$  lattice  $L_p$  in a  $\mathbb{Q}_p$  vector space  $V_p$ , by extending the bilinear form b onto the tensor product  $V_p = \mathbb{Q}_p \otimes_{\mathbb{Q}} V$  and setting  $L_p = \mathbb{Z}_p \otimes_{\mathbb{Z}} L$ . A fundamental concrete example for  $2 is the lattice <math>\mathbb{Z}_p^n$  with form (here,  $r \in \mathbb{Z}_p^{\times}$ )

$$q_p^{n,r}: x \mapsto x^T \operatorname{diag}(1, \dots, 1, r)x.$$
(1.2)

**Proposition 1.11.** For  $2 every non-degenerate quadratic <math>\mathbb{Z}_p$  lattice (L,q) is isomorphic to an orthogonal sum of scaled forms as in (1.2). More precisely, let  $\dim(V) = m$ , then there are  $m_i \in \mathbb{N}$  for  $1 \leq i \leq n$  such that  $\sum_{i=1}^n m_i = m$ ,  $r_i \in \mathbb{Z}_p^{\times}$  and a strictly increasing sequence of integers  $\nu_i$ , such that

$$(L,\mathbf{q}) \simeq \bigoplus_{i=1}^{m} \left( \mathbb{Z}_p^{n_i}, p^{\nu_i} \cdot \mathbf{q}_p^{n_i,r_i} \right).$$

Further, two of the above orthogonal sums are isomorphic to each other precisely when the associated ensembles of invariants  $m_i, \nu_i, \epsilon_i \coloneqq \left(\frac{r_i}{p}\right)$  are identical.

**Corollary 1.12.** Assume  $p \neq 2$ . Every  $\mathbb{Z}_p$  lattice has an orthogonal basis and there is a cancellation property for  $\mathbb{Z}_p$  lattices.

**Remark 1.13.** In fact, the case p = 2 is more complicated as the representation is not unique! However, these forms are required in Chapter 6 so that we refer to [Cas78, 8.4 p. 117] as well as [Joh98, 7.5 p. 881] and at least note that the fundamental building blocks of regular 2-adic quadratic lattices are, up to scaling with powers of 2, given by the Gram matrices

(r) with 
$$r \in \{1, 3, 5, 7\}$$
,  $H = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}$ ,  $A_2 = \begin{pmatrix} 2 & 1 \\ 1 & 2 \end{pmatrix}$ . (1.3)

Further, there is a cancellation property with respect to hyperbolic planes, i.e. for  $L_1, L_2$ quadratic  $\mathbb{Z}_2$  lattices with  $L_1 \oplus H \simeq L_2 \oplus H$ , we have  $L_1 \simeq L_2$  (also cf. [OMe73, 93:14 p. 256]). This remains true if H is replaced by a scaled hyperbolic plane.

**Definition 1.14.** The constituents  $(\mathbb{Z}_p^{n_i}, p^{\nu_i} \cdot q_p^{n_i,r})$  in Proposition 1.11 are referred to as *Jordan components* and are also denoted  $(p^{\nu_i})^{\epsilon_i n_i}$ , where  $\epsilon_i = (\frac{r_i}{p})$ .

In case p = 2, the characterisation is more complicated, incorporating the *oddity* and two different *types* of forms (cf. [Joh98, 7.4 p. 380]).

After having learnt about characterisations of  $\mathbb{Z}_p$  lattices and realising that a  $\mathbb{Z}$  lattice L lifts to local lattices  $L_p \coloneqq L \otimes \mathbb{Z}_p$ , it is natural to utilise local data in order to describe the global situation. Before continuing, note that with the correct identifications

$$L = \bigcap_{p < \infty} L_p.$$

**Definition 1.15.** On the set of non-degenerate  $\mathbb{Z}$  lattices we introduce an equivalence relation by associating a lattice L with the collection of the isometry classes of  $[L \otimes \mathbb{R}, (L_p)_{p < \infty}]$ . These collections are called *genera* (singular: *genus*).

In fact, the local isomorphisms do not characterise  $\mathbb{Z}$  lattices up to  $\mathbb{Z}$  isomorphy; the relation is strictly coarser. Nonetheless, if the Gram matrix is indefinite and the rank is greater than 2, the significant case for this thesis, these equivalent classes are close. In fact, they coincide for  $|\det(L)| < 128$  and divergences obey strict conditions (cf. [Joh98,

15.7 p. 378 and 15.9]). Also, two  $\mathbb{Z}$  lattices  $L_1, L_2$  have the same genus, if, and only if,  $L_1 \oplus H \simeq L_2 \oplus H$ .

*Example* 1.16. The following  $\mathbb{Z}$  lattices have the same genus (note that  $41 \in (\mathbb{Z}_2^{\times})^2$  as well as  $2 \in (\mathbb{Z}_{41}^{\times})^2$ ) and are not isomorphic over  $\mathbb{Z}$  (cf. [Cas78, p. 129]):

$$\begin{pmatrix} 2 & 0 \\ 0 & 164 \end{pmatrix} \quad \text{and} \quad \begin{pmatrix} 4 & 0 \\ 0 & 82 \end{pmatrix}.$$

The following definition plays a major role for one of the main theorems of the present thesis (cf. Theorem 7.16).

**Definition 1.17.** A lattice (L, q) is said to *split* another lattice M, if there is a third lattice K, such that  $L \simeq K \oplus M$  is an orthogonal decomposition. In case M is a hyperbolic plane, we also speak of a *hyperbolic split*.

Before introducing dual lattices and discriminant forms the primary consideration of lattices is concluded by sketching the Witt ring associated to a field F. To this end, another construction principle for lattices besides direct sums is required.

**Definition 1.18.** For two quadratic modules  $(M_i, q_i)$  over the same ring their tensor product  $(M_1 \otimes M_2, q_1 \otimes q_2)$  may be defined canonically by forming the tensor product of the associated bilinear forms, so that  $q_1 \otimes q_2(m_1 \otimes m_2) = q_1(m_1) q_2(m_2)$ .

The set M(F) of all isometry classes of finite dimensional quadratic spaces over a field F represents a commutative semiring with the orthogonal sum and the tensor product as concatenations. By Witt's cancellation Theorem, the additive monoid is a cancellation monoid. Consequently, there is a natural abelian group  $\operatorname{Groth}(M(F))$  containing M(F) as a submonoid, called *Grothendieck group* of M(F). It consists of pairs  $(x, y) \in M(F)^2$  modulo the equivalence  $(x, y) \equiv (x', y') \iff x \oplus y' = x' \oplus y$ , so that, for instance, (0, y) is the additive inverse of  $(y, 0) \in M(F)$ .<sup>1</sup> The tensor product extends naturally onto this structure, equipping  $\operatorname{Groth}(M(F))$  with the structure of a commutative ring.

**Remark 1.19.** For a field F, the quotient of the ring  $(\operatorname{Groth}(M(F)), \otimes)$  and the ideal generated by hyperbolic planes is called *Witt ring* of F and denoted W(F). The elements of the Witt ring are in bijection to the isometry classes of anisotropic quadratic spaces [Lam05, II.1 Prop. 1.4 p. 29].

<sup>&</sup>lt;sup>1</sup>Recall the construction of rational numbers from the ring of integers.

**Remark 1.20.** Note that the discriminant is well defined on the Witt ring (cf. Definition 1.9 b)). Further, the signature  $(m^+, m^-)$  of a quadratic  $\mathbb{R}$  space descends to an invariant on  $W(\mathbb{R})$  when it is modified to  $\operatorname{sig}(V) = m^+ - m^-$ . For a  $\mathbb{Z}$  lattice (L, q) we will also write  $\operatorname{sig}(L) \coloneqq \operatorname{sig}(L \otimes_{\mathbb{Z}} \mathbb{R})$ .

#### Dual lattices and discriminant forms

In the following, we assume the lattice L to be non-degenerate.

**Definition 1.21.** For a non-degenerate R lattice (L, q), define its *dual lattice* 

$$L' \coloneqq \{x \in L \otimes Q(R) \mid b(x, y) \in R \text{ for all } y \in L\}.$$

The lattice L is called

- (i) integral, if  $L \subseteq L'$  or equivalently  $b(L, L) \subseteq R$  or  $S_{i,j} \in R$  for all indices,
- (ii) even, if  $q(L) \subseteq R$ ,
- (iii) unimodular, if L = L'.

Note that every lattice which is unimodular or even is automatically integral. Further, in case of  $2 we find <math>2 \in \mathbb{Z}_p^{\times}$ , so that the notions of being even and integral coincide for  $\mathbb{Z}_p$  lattices.

- *Example* 1.22. a) The  $\mathbb{Z}_p$  lattice  $p^{\nu}q_p^{n,r}$  from Example 1.10 is even; also compare Table 1.1 for L'.
  - b) The hyperbolic plane H from Example 1.7 is unimodular over all rings R.
  - c) For R with  $2, 3 \in R^{\times}$  consider  $(R^2, q)$  with Gram matrix  $A_2$  from Remark 1.13, meaning  $q(x) = x_1^2 + x_1x_2 + x_2^2$ . Clearly, the vectors  $\ell_1 = \frac{1}{3} \cdot \begin{pmatrix} 1 \\ 1 \end{pmatrix}, \ell_2 = \begin{pmatrix} 1 \\ 0 \end{pmatrix}$  are part of the dual lattice and since  $|\langle \ell_1, \ell_2 \rangle / L| = 3 = \det(A_2)$  which equals |L'/L| by Remark 1.30, these represent a basis of L'.

**Remark 1.23.** a) For a non-degenerate R lattice L and  $c \in R^{\times}$  we find  $(cL)' = c^{-1}L'$ . b) If L is a non-degenerate R lattice with Gram matrix S, then  $L' = S^{-1}L$ .

Before turning to discriminant forms, we require some technical notions for reference.

**Definition 1.24.** Let (L, q) be an R lattice.

a) For  $0 \neq l_1, l_2 \in L \otimes_R \operatorname{Fr}(R)$  and  $z \in R \setminus R^{\times}$  such that  $z \cdot l_2 = l_1$ , we say that  $l_2$ divides  $l_1$  and write  $l_2 \mid l_1$ . If, in addition,  $l_2 \in L$ , we say z divides  $l_1$ , written  $z \mid l_1$ and set  $l_1/z \coloneqq l_2$  which might not be unambiguous!
- b) The element  $l_1 \neq 0$  is said to be *primitive* in L, if  $l/z \notin L$  for all  $z \in R \setminus R^{\times}$ .
- c) An even L is called *maximal* if there is no even R lattice in  $L \otimes_R Q(R)$  properly containing L.

**Remark 1.25.** Let R be a principal ideal domain.

- a) Any R lattice is contained in a maximal lattice.
- b) If L is maximal and  $L \otimes_R Q(R)$  splits a hyperbolic plane, so does  $L = M \oplus H$  where M is automatically a maximal R lattice.
- c) In case  $R = \mathbb{Z}_p$   $(p < \infty)$  and  $\operatorname{rk}(L) \ge 5$  the space  $L \otimes \mathbb{Q}_p$  splits a hyperbolic plane. By the theorem of Hasse–Minkowski every indefinite quadratic  $\mathbb{Q}$  space of dimension at least 5 splits off a hyperbolic plane. This classical result is attributed to Meyer.

The theorem of Meyer implies the following result about the Witt rank over  $\mathbb{Q}$ .

**Remark 1.26.** Let (L, q) be an even  $\mathbb{Z}$ -lattice of signature  $(m^+, m^-)$ . If  $|\operatorname{sig}(L)| \geq 3$ , then the Witt rank r of  $L \otimes \mathbb{Q}$  is maximal, i.e.  $r = \min\{m^+, m^-\}$ .

Finally, discriminant forms may be derived from lattices.

**Remark 1.27.** If an R lattice L is integral, the bilinear form induces a form

$$\overline{\mathbf{b}}: L'/L \times L'/L \to F/R.$$

If a lattice L is even, the quadratic form induces a form

$$\overline{\mathbf{q}}: L'/L \to F/R.$$

The latter is called *discriminant form*.

We state some apparent observations before introducing more parameters.

- *Example* 1.28. a) Since the hyperbolic plane H is unimodular, we find  $H'/H = \{0\}$ and  $\overline{q} = 0$ .
  - b) Let (L, q) be an even  $\mathbb{Z}_p$  lattice and  $d \in \mathbb{Z}_p$ . Then  $L(d) = L \leq L' \leq L(d)'$ , so that there is a natural map  $L'/L \hookrightarrow L(d)'/L(d)$ .

**Definition 1.29.** Let (L, q) be an even  $\mathbb{Z}$  lattice.

a) The smallest number  $N \in \mathbb{N}$ , such that  $N q(\ell) \in \mathbb{Z}$  for all arguments  $\ell \in L'$  is called the *level* of L.

	L	L'	D	N
p>2	$(\mathbb{Z}_p^n, p^\nu \operatorname{q}_p^{n,r})$	$\left(p^{-\nu}\cdot\mathbb{Z}_p^n,p^{\nu}\operatorname{q}_p^{n,r}\right)$	$((\mathbb{Z}/p^{\nu}\mathbb{Z})^n, p^{-\nu}\overline{\mathbf{q}}_p^{n,r})$	$p^{\nu}$
p = 2	$(\mathbb{Z}_p^n, p^{\nu} \operatorname{q}_p^{n,r})$	$\left(p^{-(\nu+1)} \cdot \mathbb{Z}_p^n, p^{\nu} \operatorname{q}_p^{n,r}\right)$	$\left( (\mathbb{Z}/p^{\nu+1}\mathbb{Z})^n, p^{-(\nu+2)} \mathbf{q}_p^{n,r} \right)$	$p^{\nu+2}$
	$(\mathbb{Z}_p^2, p^{\nu}H)$	$\left(\mathbb{Z}_p^2/p^{\nu},p^{\nu}H\right)$	$\left( (\mathbb{Z}/p^{\nu}\mathbb{Z})^2, p^{-\nu}H \right)$	$p^{\nu}$
	$(\mathbb{Z}_p^2, p^{\nu}A_2)$	$(p^{-\nu}\mathbb{Z}_p^2, p^{\nu}A_2)$	$\left((\mathbb{Z}/p^{\nu}\mathbb{Z})^2, p^{-\nu}A_2\right)$	$p^{\nu}$

- Table 1.1: Examples of even lattices, their discriminant forms, and levels. Here, p is a prime,  $n, \nu \in \mathbb{N}_0$ , the form  $q_p^{n,r}$  is given in Example 1.10. In addition, note that H and  $A_2$  (cf. Remark 1.13) denote the Gram matrix, meaning a factor of  $\frac{1}{2}$  has to be taken into account when moving on to the associated quadratic form.
  - b) The *level* of  $\ell \in L$  is the number  $N_{\ell} \in \mathbb{N}$ , such that  $b(\ell, L) = N_{\ell}\mathbb{Z}$ .
  - c) A prime number p is referred to as being *bad* and as *good* otherwise. We also speak of these primes as *places*.

**Remark 1.30.** (cf. addendum) Let L be an even non-degenerate  $\mathbb{Z}$  lattice of rank m.

- a) We find  $det(L)L' \leq L$  as well as  $NL' \leq L$ .
- b) We have  $|L'/L| = |\det(L)| = \operatorname{vol}(L)^2$ , in particular, L'/L is finite.
- c) Further, the following division relations are true:  $lev(L) \mid 2 det(L) \mid 2 lev(L)^m$  and  $lev(L) \mid det(L)^2$ . In case of odd level N or even rank m, we find  $lev(L) \mid det(L)$ .
- d) The level lev(L) and the determinant det(L) have the same prime divisors.

Further, if the rank of the lattice L is odd, then its level is divisible by 4. Equipped with these statements, discriminant groups may be defined and analysed locally.

**Definition 1.31.** In case of an even  $\mathbb{Z}$  lattice L, the group L'/L is called *discriminant group*. Its *level* is the level of L.

**Remark 1.32.** In fact, any finite abelian group  $\mathcal{L}$  with a non-degenerate quadratic form to  $\mathbb{Q}/\mathbb{Z}$  arises as a quotient L'/L of a  $\mathbb{Z}$  lattice L (cf. [Wal63, 4 p. 294]). Further, for any even  $\mathbb{Z}$  lattice (L, q) we have *Milgram's formula* [MH73, App. 4]

$$\sum_{\lambda \in L'/L} \exp(2\pi i \cdot \overline{\mathbf{q}}(\lambda)) = \sqrt{|L'/L|} \cdot \exp(2\pi i \cdot \operatorname{sig}(L)/8).$$
(1.4)

As a consequence, the signature  $sig(\mathcal{L}) \coloneqq sig(L) \mod 8$  is well defined.

*Example* 1.33. For a  $\mathbb{Z}$  lattice, the group  $\mathcal{L} := L'/L$  factors orthogonally into maximal p subgroups

$$L'/L \simeq \bigoplus_{p < \infty} \mathcal{L}_p.$$

By the classification of finitely generated abelian groups  $L'/L \simeq \prod_{p < \infty} \mathcal{L}_p$  is immediate. Note that if  $x \in \mathcal{L}_p$ ,  $y \in \mathcal{L}_{p'}$ , then  $\deg(x) \overline{b}(x, y) \equiv 0$ , so that the denominator of  $\overline{b}(x, y)$ must be a divisor of  $\deg(x)$ . But the same holds for y, implying  $\overline{b}(x, y) \equiv 0$ . Further, we have  $\mathcal{L}_p = L'_p/L_p$  with  $L_p = L \otimes_{\mathbb{Z}} \mathbb{Z}_p$ , so that considerations of discriminant forms are reduced to local objects, signifying the relevance of Definition 1.15 and Proposition 1.11. In particular, Table 1.1 is essentially exhaustive.

**Remark 1.34.** An even lattice L is maximal if, and only if, L'/L is anisotropic. In particular, unimodular lattices are maximal. Further, by Example 1.33, the  $\mathbb{Z}$  lattice L is maximal if, and only if,  $L_p$  is maximal for all primes p.

**Definition 1.35.** Let  $\mathcal{L}$  be a discriminant group and  $n \in \mathbb{N}$ . Then  ${}^{n}\mathcal{L}$  denotes the elements which are *n*-th powers in  $\mathcal{L}$  and  ${}_{n}\mathcal{L}$  denotes the *n*-torsion of  $\mathcal{L}$ .

**Remark 1.36.** For a discriminant group  $\mathcal{L}$  the subgroup  ${}^{n}\mathcal{L}$  is the orthogonal complement of  ${}_{n}\mathcal{L}$  and the following sequence is exact

$$0 \longrightarrow {}_{n}\mathcal{L} \stackrel{\iota}{\longrightarrow} \mathcal{L} \stackrel{\cdot n}{\longrightarrow} {}^{n}\mathcal{L} \longrightarrow 0.$$

Example 1.37. Consider  $p^{\nu}q_p^{n,r}$  (cf. Example 1.10) and  $k \in \mathbb{N}$  with  $\nu_p \coloneqq \nu_p(k)$ . Then

$${}_{k}\mathcal{L} = \left\{ z \in \mathbb{Z}_{p} \mid p^{\nu - \nu_{p} + 2\delta_{p=2}} \mid z \right\}^{n}$$

The following technical definition is required in Section 6.4.2 and in particular in Proposition 6.71, where (1.6) is needed.

**Definition 1.38.** Let (L, q) be an integral  $\mathbb{Z}$  lattice. By [Ste15, Lemma 4.3] for any  $n \in \mathbb{N}$ and odd prime p, there is an orthogonal decomposition of  $L/p^n L$  into  $\mathbb{Z}/p^n \mathbb{Z}$  submodules

$$\left(\bigoplus_{i} L_{i}\right) \oplus \left(\bigoplus_{j} M_{j}\right) \oplus N,$$

where  $L_i = (\mathbb{Z}/p^n\mathbb{Z})v_i$  is one dimensional with  $b(v_i, v_i) \in (\mathbb{Z}/p^n\mathbb{Z})^{\times}$ ,  $M_j = (\mathbb{Z}/p^n\mathbb{Z})v_j$ with  $b(v_j, v_j) \in p^k(\mathbb{Z}/p^n\mathbb{Z})^{\times}$  where  $1 \leq k \leq n-1$ , and  $b(N, N) \subseteq p^n\mathbb{Z}$ . We may assume the  $M_j$  being sorted with respect to increasing valuation k of  $q(v_j)$ . Define

$$n_k = \#\{1 \le i \le m : \nu_p(\mathbf{b}(v_i, v_i)) = k\} \text{ and } R_p^n \coloneqq \sum_{k=0}^{n-1} (n-k)n_k.$$
 (1.5)

The most prominent case in this thesis is the one of n = 1, in which case we write  $R_p := R_p^1$ and note that

$$R_p = n_0 = \#\{1 \le i \le m : \nu_p(\mathbf{b}(v_i, v_i)) = 0\}$$
(1.6)

equals the number of basis elements of L/pL such that the *p*-valuation of their norm is zero.

# 1.2 *p*-adic numbers and the adele ring

Number theorists are in general interested in the rational numbers  $\mathbb{Q}$  or even primarily the discrete case of  $\mathbb{Z}$ . Sustainably, analytical approaches have been instrumental in developing sophisticated tools to simplify or even completely solve such discrete algebraic questions. The following section presents tools from Topology and functional Analysis (more precisely integration theory) that have the potential to facilitate solutions of purely algebraic questions, while also offering an intrinsic mathematical appeal.

**Definition 1.39.** For a field F, a mapping  $|\cdot| : F \to [0,\infty]$  is called *absolute value function* (abs), if

- a)  $|\cdot|$  is definite, that is  $|a| = 0 \implies a = 0$ .
- b)  $|\cdot|$  satisfies the triangle inequality, i.e. for all  $a, b \in F$ :  $|a + b| \le |a| + |b|$ .
- c)  $|\cdot|$  is multiplicative, meaning for all  $a, b \in F$  we have  $|ab| = |a| \cdot |b|$ .

Further, such a function is said to be *non-Archimedean* if the following stricter version of b) is true:

$$|a+b| \le \max\{|a|, |b|\}.$$

The inequality above is referred to as the strong triangle inequality.

Example 1.40. The most prominent examples are the standard abs on  $\mathbb{Q}$  as well as the so called *p*-adic absolute value functions for a prime number *p*. The latter measure the appearance of *p* in the prime factor decomposition of a rational number. More precisely, let

 $p \in \mathbb{N}$  be a prime number and  $0 \neq q \in \mathbb{Q}$ . Furthermore, let  $a, b, k \in \mathbb{Z}$  with gcd(a, b) = 1, such that gcd(a, b, p) = 1 and  $q = p^k \frac{a}{b}$ . Then setting  $|q|_p \coloneqq p^{-k}$  gives rise to an *abs* on  $\mathbb{Q}$  which is called *p*-adic absolute value function.

As expected, positive powers of *p*-adic *abs* are again *abs*, the same holds true for powers to  $0 < \alpha < 1$  of the standard absolute value function on  $\mathbb{Q}$ . Clearly, these classes of *abs* generate the same topology on  $\mathbb{Q}$ , therefore two *abs* related this way are called *equivalent*. Another absolute value function is the trivial one, mapping every element to 1.

**Proposition 1.41** (Ostrowski, [BS, 1.4.2 Thm. 3 p. 37]). The list of absolute value functions on  $\mathbb{Q}$  presented in Example, i.e. the standard abs, the class of p-adic abs, and the trivial abs 1.40 is complete up to equivalence.

The Theorem of Ostrowski reveals that the definition of absolute value functions is eminently restrictive and that it is interconnected with the arithmetic structure of the field  $\mathbb{Q}$ . In fact, one can prove that any locally compact field with non-discrete topology is isomorphic to  $\mathbb{R}$ ,  $\mathbb{C}$ , or a finite extension of  $\mathbb{Q}_p$  or  $\mathbb{F}_p(t)$ . The latter two classes are exactly the so called *local fields*, which might be characterised by being complete with respect to a discrete valuation (absolute value), and possess a finite residue class field.

**Definition 1.42.** The completion of  $\mathbb{Q}$  with respect to a *p*-adic *abs* is called *field of p-adic numbers* and denoted  $\mathbb{Q}_p$ . It is referred to as the *local field* at that *place p*, in contrast to the *global* field  $\mathbb{Q}$ . The function  $|\cdot|_p$  extends uniquely to  $\mathbb{Q}_p$  and the fact that it is non-Archimedean results in the unit ball (which is open as well as compact)

$$\mathbb{Z}_p \coloneqq \{x \in \mathbb{Q}_p \mid |x|_p \le 1\} = \overline{\mathbb{Z}}^{|\cdot|_p}$$

being a ring. It is called ring of *p*-adic integers, is a principal ideal domain and its unit group consists exactly of all elements with absolute value 1. As a consequence,  $\mathbb{Q}_p$  is a locally compact, totally disconnected group. According to convention, define  $\mathbb{Q}_{\infty} := \mathbb{R}$ and  $\mathbb{Z}_{\infty} := \mathbb{Z}$  and write ' $p < \infty$ 'for the set of all primes.

**Remark 1.43.** Note that since  $\mathbb{Z}_p$  is compact for  $p < \infty$ , as well as open, the standard notation for the closure of balls is inadequate. For a fixed p, the field  $\mathbb{Q}_p$  may be identified with the set of all Laurent series  $\sum_{k=k_0}^{\infty} c_k p^k$  for integer coefficients  $0 \le c_k < p$  and an integer  $k_0$ . Then  $\mathbb{Z}_p$  consists of the subring of power series.

It becomes apparent that the *p*-adic numbers encode arithmetic information about the rationals in an analytic fashion. Hence, a desirable tool would be an algebraic and topological structure containing all *p*-adic numbers  $\mathbb{Q}_p$ . In order to capitalise on functional analytical tools, a locally compact ring *R* seems to be preferable. However, in this case *R* would be a locally compact  $\mathbb{Q}$  vector space and by a well known result from André Weil it had to be finite dimensional, conflicting with the requested properties. In order to obtain a structure which is at least additively locally compact, the inductive limit of the topological groups

$$\prod_{p \in S} \mathbb{Q}_p \times \prod_{p \notin S} \mathbb{Z}_p$$

with product topology is taken, where S is a finite set of primes. The resulting structure  $\mathbb{A}_f$  is called *ring of finite adeles* and is a locally compact additive group with a topologically incompatible multiplication. Further, it contains the ring  $\hat{\mathbb{Z}} := \prod_{p < \infty} \mathbb{Z}_p \subseteq \mathbb{A}_f = \hat{\mathbb{Z}} \otimes_{\mathbb{Z}} \mathbb{Q}$ .

Definition 1.44. The topological product

$$\mathbb{A} \coloneqq \mathbb{A}_f \times \mathbb{R}$$

is called *Adele ring*. It is, again, a locally compact abelian group and its Haar measure  $\mu$  is obtained by taking the product measure of the Haar measures of  $\mathbb{Q}_p$  (the same holds for  $\mathbb{A}_f$  where we assume the normalisation  $\mu_f(\hat{\mathbb{Z}}) = 1$ ). The group of multiplicatively invertible elements  $\mathbb{A}^{\times}$  of  $\mathbb{A}$  is called *Idele group* and becomes a locally compact group with the inductive limit topology with respect to  $\mathbb{Z}_p^{\times}$  as open compact subgroups in the components. Its Haar measure  $\mu^{\times}$  is again the product of the Haar measures of  $\mathbb{Q}_p^{\times}$ .<sup>2</sup>

Example 1.45. For  $s \in \mathbb{C}$  with  $\operatorname{Re}(s) > 1$  we have that [Dei10, 5.3 Prop. 5.3.4 p. 128]

$$\zeta(s) = \int_{\hat{\mathbb{Z}} \cap \mathbb{A}^{\times}} \prod_{p < \infty} |x_p|_p^s \ \mathrm{d}\mu_f^{\times}$$

The integral expression above contains, at a glance, an infinite product. In fact, the product is effectively finite for every argument and this type of function is usually considered when integrating adelically. The above identity was generalised by John Tate in his thesis [Tat16, [1]]. More precisely, for a Schwartz–Bruhat function f on  $\mathbb{A}^{\times}$  (cf. Definition 1.53), the Zeta integral

$$\zeta(f,s) \coloneqq \int_{\mathbb{A}^{\times}} f(x) \cdot \prod_{p \le \infty} |x_p|_p^s \, \mathrm{d}\mu^{\times}$$

<sup>&</sup>lt;sup>2</sup>These are given by their respective scaled additive measure  $\mu$  such that  $\mu^{\times}(t) = \frac{p}{p-1} \frac{\mu(t)}{|t|_p}$ , where  $\frac{\infty}{\infty} = 1$ .

converges locally uniformly for  $\operatorname{Re}(s) > 1$ , has a meromorphic extension to  $\mathbb{C}$  with known poles and fulfils the functional equation

$$\zeta(f,s) = \zeta(\hat{f}, 1-s).$$

The construction presented above may be generalised by intertwining the integrand with finite characters. Further references are the elaborate [Kud04b], the comfortably accessible [Dei10, 6 p. 139–156], the concise [Bum98, 3.1 pp. 254–277], as well as Tate's original thesis [Tat16, [1] pp. 1–44].

The function

$$|\cdot|: \mathbb{A}^{\times} \to \mathbb{R}_{>0}, \qquad (x_p)_p \mapsto \prod_{p \le \infty} |x_p|_p$$

used above as an integral kernel defines a homomorphism whose kernel we denote by  $\mathbb{A}_1$ . It plays a significant role in classifying finite characters on  $\mathbb{A}^{\times}/\mathbb{Q}^{\times}$ .

Another important feature of adelic formulations is the so called *Hasse principle* or *local-global principle*, named after Helmut Hasse. The core concept is that global properties of  $\mathbb{Q}$  correspond to an ensemble of properties at all places  $p \leq \infty$ . A concrete example for this is the Theorem of Hasse-Minkowski (cf. [Has23] or [Has24]) or the fact that  $\mathbb{Q}^{\times} < \mathbb{A}_1$ .

### **1.3** Dual groups and Fourier analysis

A convenient setting for this theory is that of *locally compact abelian* groups (lca). Details are treated in depth in [Rud90, 1–2 pp. 1–58], which may serve as a comprehensive source for acquiring the theory. The examples of particular interest for this thesis are discussed in [Dei10, 5.4 pp. 129-138 and 6.3 pp. 146–152] and [Bum98, 3.1 pp. 254–262], featuring p-adic numbers and the adele ring.

**Definition 1.46.** A topological group  $(G, \cdot)$  is called *locally compact* if it is Hausdorff and every element of G possesses a compact neighbourhood.

A positive measure  $\mu$  on the borel  $\sigma$ -Algebra of G is called a *Haar measure*, if it satisfies the following conditions:

- a)  $\mu$  is locally finite, meaning for any compact set  $K \subseteq G$  we have  $\mu(K) < \infty$ .
- b)  $\mu$  is regular.
- c)  $\mu$  is left invariant, hence for any subset  $H \subseteq G$  and  $g \in G$  the identity  $\mu(gH) = \mu(H)$  holds.

By a theorem of Haar [Els11, Thm. 3.12 p. 362], any locally compact group possesses a non-trivial Haar measure  $\mu$  which is unique up to a constant factor. As a consequence, for any  $g \in G$  the measure  $\mu_g(A) \coloneqq \mu(Ag)$  must coincide with  $\delta_G(g) \cdot \mu$  for an appropriately chosen factor  $\delta_G(g) \in \mathbb{R}_{>0}^{\times}$ . Moreover,  $\delta_G$  defines a continuous homomorphism which is called *modular quasi character* of G (cf. [DE14, 1.4 Thm. 1.4.1 p. 14]).

**Definition 1.47.** A character of a lca G is a continuous homomorphism  $\chi : G \to \mathbb{T}$ . The group of characters  $G^*$  with the compact open topology is again a lca and called (*Pontryagin*) dual group .<sup>3</sup>

*Example* 1.48. The prototype of a character is the exponential function. In case of  $(\mathbb{R}, +)$ , define

$$x \mapsto \exp(2\pi i \cdot a \cdot x) =: e_{\infty}(a, x),$$

for  $a \in \mathbb{R}$ . In case of  $(\mathbb{Q}_p, +)$  for any  $p < \infty$  and  $a \in \mathbb{Q}_p$  define

$$x \mapsto \exp(-2\pi i \cdot a \cdot x) =: e_p(a, x).$$

Note that by continuity, the mapping has to be constant on some neighbourhood  $p^k \mathbb{Z}_p$  of 0. Hence, it is well defined by multiplicativity. In fact, this list of characters for  $\mathbb{Q}_p$  is faithful and exhaustive. In particular,  $\mathbb{Q}_p$  is isomorphic to its dual group via  $a \mapsto (x \mapsto e_p(a, x))$ .

The following theorem, initially proven by Pontryagin in a more restrictive setting and later proven in general by Egbert van Kampen and André Weil, justifies the name *dual* group.

**Theorem 1.49** (Pontryagin). The bidual group of G is canonically isomorphic to G, in formulae  $(G^*)^* \simeq G$ .

As a consequence, a lea G is compact if, and only if,  $G^*$  is discrete.

With the notion of a dual group, Fourier transformations may be defined. Recall that we write  $L^1(G,\mu)$  for the space of integrable  $\mathbb{C}$ -valued functions on  $(G,\mu)$ .

**Definition 1.50.** Let  $f \in L^1(G, \mu)$ , then the mapping

$$G^* \ni \chi \mapsto \int_G f(g) \cdot \chi^{-1}(g) \, \mathrm{d}\mu(g)$$

is called *Fourier transform* of f and denoted by  $\hat{f}$  or  $\mathcal{F}(f)$ .

<sup>&</sup>lt;sup>3</sup>So characters are unitary, one dimensional representations of the group G, whereas for the notion of quasi characters unitaricity is not demanded.

Before illustrating this definition by specifying the main application within this thesis, a well established Lemma is stated.

**Remark 1.51.** Let  $H \leq G$  be a compact subgroup and  $\psi \in \text{Hom}(G, \mathbb{C}^{\times}) \cap L^{\infty}_{\text{loc}}(G)$ . Then

$$\mathcal{F}(\mathbb{1}_H \psi)(\chi) = \begin{cases} \mu(H), & \text{if } \psi|_H = \chi|_H, \\ 0, & \text{else.} \end{cases}$$

*Proof*: For arbitrary  $h \in H$  the following calculation immediately yields the result

$$\mathcal{F}(\mathbb{1}_H\psi)(\chi) = \psi(h)\chi^{-1}(h) \int_H \psi(g)\chi^{-1}(g) \,\mathrm{d}\mu(g) = \left(\psi\chi^{-1}\right)(h) \cdot \mathcal{F}(\mathbb{1}_H\psi)(\chi).$$

In case of  $G = \mathbb{Q}_p$ , the transform may be identified with a function on  $\mathbb{Q}_p$  (cf. Example 1.48), resulting in the classical Fourier transformation in case of  $p = \infty$  if the measure  $\mu$  is scaled properly. In case of a non-degenerate quadratic space  $(V_p, q)$  over  $\mathbb{Q}_p$ , the bilinear form b provides an isomorphism to the linear dual of  $V_p$ , which induces an isomorphism of Pontryagin duals via

$$V_p \ni a \mapsto \left( x \mapsto \psi_p(\mathbf{b}(a, x)) =: \psi_p^{\mathbf{b}}(a, x) \right) \in V_p^*, \tag{1.7}$$

for a non-trivial character  $\psi_p$  of  $\mathbb{Q}_p$ . For convenience, we fix the choice  $\psi_p(x) \coloneqq e_p(1, x)$ (cf. Example 1.48). These isomorphisms give rise to a Fourier transformation on  $L^1(V_p)$ 

$$\mathcal{F} \coloneqq \mathcal{F}_{\psi_p^{\mathrm{b}}} : L^1(V_p) \ni f \mapsto \left( a \mapsto \int_{V_p} f(x) \cdot \psi_p^{\mathrm{b}}(a, x)^{-1} \, \mathrm{d}\mu(x) \right).$$
(1.8)

In fact, there is, analogously to the Schwartz space in the real setting, a subspace  $S(V_p) < L^1(V_p)$  on which  $\mathcal{F}_{\psi_p^{\rm b}}$  defines an automorphism. Moreover, the space  $S(V_p)$  is characterised by this property<sup>4</sup> (cf. [Osb75, Thm. 1 p. 42]).

Example 1.52. Let  $f \in L^1(V_p, \mu_p)$  and b be a non-degenerate bilinear form on  $V_p$ ,  $\psi_p$  be a non-trivial character of  $\mathbb{Q}_p$  and  $c \in V_p$ . Then we find the following calculation rules.

a)  $\mathcal{F}[f(\cdot + c)](a) = \psi_p^{\mathbf{b}}(c, a) \cdot \mathcal{F}[f](a)$ 

<sup>&</sup>lt;sup>4</sup>The classic approach from Bruhat is via an inductive limit involving Lie subquotient groups and hence requires extensive background knowledge in differential geometry, while the characterisation of Osborne is elementarily accessible. In addition, a key feature of the space and its maximality with respect to this feature is apparent.

- b)  $\mathcal{F}[\psi(c, \cdot)f](a) = \mathcal{F}[f](a-c)$
- c) Let  $p < \infty$  and  $(L_p, q) \subset (V_p, q)$  be an even lattice. Then

$$\mathcal{F}[\mathbb{1}_{L_p}] = \mu_p(L_p) \cdot \mathbb{1}_{L'_p} \tag{1.9}$$

by Remark 1.51 with  $\psi = 1$  and  $H = L_p$ .

d) Let  $p < \infty$ ,  $(L_p, q) \subset (V_p, q)$  be an even lattice, and  $\lambda \in \mathbb{Q}_p^{\times}$ . Then

$$\mathcal{F}[f(\lambda \cdot)](a) = \frac{1}{|\lambda|_p^m} \mathcal{F}[f](a/\lambda).$$

This follows from the fact that  $A \mapsto \mu_p(\lambda A)$  defines a Haar measure as well, meaning it must agree with  $\mu_p$  modulo a non-trivial factor. Evaluation on  $L_p$  in case  $p < \infty$ yields the desired result.

**Definition 1.53.** Let  $p < \infty$ . For a  $\mathbb{Q}_p$  vector space  $V_p$ , a *Schwartz–Bruhat* function is a locally constant function with compact support. These functions constitute a vector space denoted by  $\mathcal{S}(V_p)$ .

Clearly, for  $l \in V_p$  and  $\lambda \in \mathbb{Q}_p$ , the function  $\mathbb{1}_{l+\lambda L_p}$  is a Schwartz–Bruhat function. On the other hand, by decomposing the support of  $f \in \mathcal{S}(V_p)$ , we may represent it as a linear combination of such functions.

**Remark 1.54.** Let  $p < \infty$ ,  $(L_p, q) \subset (V_p, q, \mu_p)$  be an even lattice,  $l \in V_p$ , and  $\lambda \in \mathbb{Q}_p^{\times}$ . Then the observation  $\mathbb{1}_{l+\lambda L_p}(x) = \mathbb{1}_{\lambda L_p}(x-l)$ , as well as  $\mathbb{1}_{\lambda L_p}(x) = \mathbb{1}_{L_p}(x/\lambda)$ , imply in conjunction with Example 1.52 that

$$\mathcal{F}[\mathcal{F}[\mathbb{1}_{l+\lambda L_p}]](a) = \mathcal{F}[\psi_p^{\mathrm{b}}(-l, \cdot)\mathcal{F}[\mathbb{1}_{\lambda L_p}]](a)$$

$$= \mathcal{F}[|\lambda|_p^m \mathcal{F}[\mathbb{1}_{L_p}](\lambda \cdot)](a+l)$$

$$= \mathcal{F}[\mathcal{F}[\mathbb{1}_{L_p}]]\left(\frac{a+l}{\lambda}\right)$$

$$= \mu_p(L_p)^2 |L'_p/L_p| \cdot \mathbb{1}_{L_p}\left(\frac{a+l}{\lambda}\right)$$

$$= \mu_p(L_p)^2 |L'_p/L_p| \cdot \mathbb{1}_{-l+\lambda L_p}(a)$$

$$= \mu_p(L_p)^2 |L'_p/L_p| \cdot \mathbb{1}_{l+\lambda L_p}(-a)$$

In particular, the Fourier transform defines an automorphism of  $\mathcal{S}(V_p)$ . It is apparent,

that the usual choice of Haar measure, in the following denoted by  $\mu_p^{\rm b}$ , has normalisation

$$\mu_p^{\mathrm{b}}(L_p) \coloneqq \sqrt{|L'_p/L_p|}^{-1} = \sqrt{|\det(S)|_p}$$
(1.10)

as this results in

$$\mathcal{F}_{\psi_n^b}^2[f](a) = f(-a) \tag{1.11}$$

for  $f \in \mathcal{S}(V_p)$ , rendering the Fourier transformation idempotent. As a consequence, *Parseval's formula* [Els11, 3.14 p. 199] holds as well as the Theorem of Plancherel, extending the result to  $L^2$ . With this normalisation the measure  $\mu_p^{\rm b}$  is called *self dual* and the product measure on  $V_{\mathbb{A}}$  assigns the value 1 to the compact space  $V_{\mathbb{A}}/V_{\mathbb{Q}}$  (cf. [Bum98, Prop. 3.1.3 p. 261]).

Schwartz-Bruhat functions may also be defined for the adele ring. Let V be a  $\mathbb{Q}$  vector space of dimension m and  $V_p = V \otimes_{\mathbb{Q}} \mathbb{Q}_p$  such as  $V_{\mathbb{A}} = V \otimes_{\mathbb{Q}} \mathbb{A}$ . Then an element  $f \in \mathcal{S}(V_{\mathbb{A}})$ might be directly defined as a linear combination of products of the form

$$V_{\mathbb{A}} \ni a \mapsto \prod_{p \le \infty} f_p(a_p),$$

where  $f_p \in \mathcal{S}(V_p)$  and  $f_p = \mathbb{1}_{\mathbb{Z}_p^m}$  for almost all  $p < \infty$ . Note that this space agrees with the inductive limit of finite tensor products of the local Schwartz–Bruhat spaces  $\mathcal{S}(V_p)^5$ . Further, the space  $\mathcal{S}(V_A)$  is generated by products as above when restricting possible choices of  $f_p$  at finite places to characteristic functions of elements of local discriminant groups associated to even quadratic lattices in V (cf. [KY10, p. 2286]).

For the final part of this section we return to characters, discussing the adelic case.

**Remark 1.55.** The instances of Example 1.48 may be combined to an adelic character. In fact, the mapping

$$\mathbb{A} \ni a \mapsto \left\lfloor x \mapsto \prod_{p \le \infty} e_p(a_p, x_p) \right\rfloor \in \mathbb{A}^*$$

defines an isomorphism  $\mathbb{A} \simeq \mathbb{A}^*$ . Write  $e(a, x) \coloneqq \prod_{p \leq \infty} e_p(a, x)$  and note that these characters are trivial on  $\mathbb{Q}$  if, and only if,  $a \in \mathbb{Q}$ , so that  $(\mathbb{A}/\mathbb{Q})^* \simeq \mathbb{Q}$  (equipped with the discrete topology). Moreover,  $\psi$  will denote the *standard character*  $\psi(x) = e(1, x)$ , with

<sup>&</sup>lt;sup>5</sup>For inducing the inclusions vectors of each  $\mathcal{S}(V_p)$  have to be fixed which are taken to be  $\chi_{\mathbb{Z}_p}$  for  $p < \infty$ . For further information consider [Bum98, p. 300-301].

components  $\psi_p$ , if not specified further.

Characterising the characters of the multiplicative group  $\mathbb{A}^{\times}$ , however, is more subtle and will not be discussed extensively.

**Remark 1.56.** Characters of  $\mathbb{A}^{\times}/\mathbb{Q}^{\times}$  are called *Hecke characters* and decompose into a finite character and a quasi character of the form  $a \mapsto |\cdot|^s$  for some  $s \in \mathbb{C}$ . Finite Hecke characters are in bijection to primitive Dirichlet characters (cf. [Bum98, Prop. 3.1.2 p. 259], [Dei10, Lem. 6.3.2 p. 147]), a link whose application becomes relevant in Section 5.3 and which is treated in detail in [Opi18, 1.10 p. 20]. If  $\chi$  is a Dirichlet character, then  $\chi_p$  will denote the associated local character on  $\mathbb{Q}_p^{\times}$ .

Further, quadratic Dirichlet characters are representable by the Hilbert symbol, which is the subject of the following section.

**Remark 1.57.** Remark 1.55 may be upgrade to characters of  $V_{\mathbb{A}}$  by sending

$$V_{\mathbb{A}} \ni a \mapsto \left[ x \mapsto \prod_{p \le \infty} e_p^{\mathbf{b}}(a_p, x_p) \right] \in V_{\mathbb{A}}^*$$

As a consequence, Fourier analysis as above may be carried out in the adelic setting, where computations are essentially reduced to treating finitely many places. For convenience, write  $e_f^{\rm b}(a,x) \coloneqq \prod_{p < \infty} e_p^{\rm b}(a_p, x_p)$  as well as  $e^{\rm b}(a, x) \coloneqq \prod_{p < \infty} e_p^{\rm b}(a_p, x_p)$ 

#### 1.4 Hilbert symbol

The Hilbert symbol is a number theoretic bilinear form and induces characters which are essential for understanding the Weil representation which itself is of critical significance for the space of cusp forms appearing in the main theorem of this thesis.

The curious reader may find a compact and, concerning the framework of this thesis, comprehensive investigation of Hilbert symbols in [Ser12a, I.III p. 19–26] by Serre. Based on this reference, the following section provides the bare minimum of information required to grasp the conceptional use throughout this thesis.

**Definition 1.58.** Let F be a field. The Hilbert symbol  $\mathcal{H}$  of  $a, b \in F^{\times}$  is defined as

$$\mathcal{H}(a,b) = \begin{cases} 1, & \text{if } z^2 - ax^2 - by^2 = 0 \text{ has a non-trivial solution } (x,y,z) \in F^3, \\ -1, & \text{otherwise.} \end{cases}$$

This association is naturally viewed as a mapping  $F^{\times}/(F^{\times})^2 \times F^{\times}/(F^{\times})^2 \to \{\pm 1\}.$ 

**Remark 1.59.** Let  $a, a', b \in F^{\times}$  and  $F_b = F(\sqrt{b})$ . Then  $\mathcal{H}(a, b) = 1$  holds if, and only if, a belongs to the group  $N_{F_b^{\times}}$  of norms of  $F_b^{\times}$ .

- i)  $\mathcal{H}(a, -a) = \mathcal{H}(a, 1-a) = 1$ ,
- ii)  $\mathcal{H}(a,b) = 1 \implies \mathcal{H}(aa',b) = \mathcal{H}(a',b),$
- iii)  $\mathcal{H}(a,b) = \mathcal{H}(a,-ab) = \mathcal{H}(a,(1-a)b)$ , if  $1-a \neq 0$  in the last expression.

The Hilbert symbol may be evaluated in terms of the Legendre symbol. In order to to perform this reduction, a characterisation of  $F^{\times}/(F^{\times})^2$  is advantageous, which is provided in [Ser12a] in case of  $F = \mathbb{Q}_p$ . As a result, the following formulae are obtained.

**Proposition 1.60.** If  $F = \mathbb{Q}_p$  and  $a, b \in \mathbb{Q}_p$  are written as  $p^{\alpha}u, p^{\beta}v$  with  $|u|_p = |v|_p = 1$ , the Hilbert symbol of a, b admits the following representation:

$$\begin{aligned} \mathcal{H}(a,b) &= (-1)^{\alpha\beta\varepsilon(p)} \cdot \left(\frac{u}{p}\right)^{\beta} \left(\frac{v}{p}\right)^{\alpha} & \text{if } p \neq 2, \\ \mathcal{H}(a,b) &= (-1)^{\varepsilon(u)\varepsilon(v) + \alpha\omega(v) + \beta\omega(u)} & \text{if } p = 2, \end{aligned}$$

where  $\left(\frac{u}{p}\right) \coloneqq \left(\frac{\overline{u}}{p}\right) \equiv \overline{u}^{\frac{p-1}{2}} \mod p$  is defined by the common Legendre symbol of  $\overline{u}$  in  $\mathbb{F}_p$ and  $\overline{u}$  denotes the image of u under the canonical mapping  $\mathbb{Z}_p^{\times} \to (\mathbb{Z}/p\mathbb{Z})^{\times}$ . Furthermore, the notation  $\varepsilon(u) \coloneqq \frac{u-1}{2} \mod 2$  as well as  $\omega(u) \coloneqq \frac{u^2-1}{8} \mod 2$  has been used.

Once the above formulae are established, they may be utilised to derive the following theorem.

**Theorem 1.61.** For any  $p \leq \infty$ , the Hilbert symbol  $\mathcal{H}_p$  defines a non-degenerated bilinear mapping on the  $\mathbb{F}_2$  vector space  $\mathbb{Q}_p^{\times}/(\mathbb{Q}_p^{\times})^2$ .

The following examples of finite Hecke characters are central for the Weil representation of  $SL_2(\mathbb{A})$  and it is recommended to compare these with [Dei10, Lemma 6.3.2 p. 147].

Example 1.62. Let  $b \in \mathbb{Q}^{\times}$ , then for every  $p \leq \infty$ ,  $\chi_p^b : \mathbb{Q}_p^{\times} \to \{-1, 1\}$ ,  $a \mapsto \mathcal{H}_p(a, b)$  defines a character. Furthermore,  $\chi_p^b$  is trivial on  $\mathbb{Z}_p^{\times}$  for almost all p and the product over all  $\chi_p^b(a_p)$  equals 1 for  $(a_p)_{p \leq \infty} \in \mathbb{Q}^{\times} \subset \mathbb{A}^{\times}$ , giving rise to a character

$$\chi^b : \mathbb{A}^{\times} / \mathbb{Q}^{\times} \to \{-1, 1\}, \qquad (a_p)_{p \le \infty} \mapsto \prod_{p \le \infty} \chi^b_p(a_p).$$

For a quadratic  $\mathbb{Q}$  vector space  $(V, \mathbf{q})$  of dimension m, the associated character is denoted by  $\chi_V$  and equals  $\chi^b$  with  $b = (-1)^{\frac{m(m-1)}{2}} \det(V)$ . Note that the mapping  $(V, \mathbf{q}) \mapsto \chi_V$ is well defined on the underlying additive group of the Witt ring  $W(\mathbb{Q})$  (cf. Section 1.1). Similarly, define  $\chi_L$  for a quadratic  $\mathbb{Z}$ -lattice L and  $\chi_L = \mathcal{H}_p(\cdot, \operatorname{disc}(L))$  for a local lattice.

In fact, all quadratic Dirichlet characters admit a representation via the Hilbert symbol as above. Each quadratic character, given by a fundamental discriminant  $D \equiv 0, 1 \mod 4$ via the Kronecker symbol  $m \mapsto \left(\frac{D}{m}\right)$ , corresponds to the Hilbert symbol  $\mathcal{H}_{\mathbb{A}}(\cdot, D)$ .<sup>6</sup> Here,

$$\mathbb{A}^2 \ni (a,b) \mapsto \mathcal{H}_{\mathbb{A}}(a,b) \coloneqq \prod_{p \le \infty} \mathcal{H}_p(a_p,b_p)$$

denotes the product of the local Hilbert symbols. Further, these characters are related to quadratic field extensions (cf. [Zag13]), which are all of the form  $\mathbb{Q}(\sqrt{D})/\mathbb{Q}$  and will be discussed in the subsequent section.

## 1.5 Quadratic field extensions

In the following, elementary statements about quadratic field extensions of  $\mathbb{Q}$  and  $\mathbb{Q}_p$  are presented. The reader will find adequate references in [Neu06], [Hel90] as well as [Zag13] and a brief overview in [Fre90, Appendix A].

In the following, let R be a principal ideal domain and Q(R) its field of fractions. In later sections the case  $R = \mathbb{Z}_p$  for  $p \leq \infty$  will be primarily considered which is the reason for restricting the class of admissable rings R. If the field in question is  $\mathbb{Q}_p$  for  $p < \infty$ , we associate the choice  $R = \mathbb{Z}_p$ , in case of  $\mathbb{Q}$ , the ring  $R = \mathbb{Z}$  is considered.

**Definition 1.63.** Let F/Q(R) be a field extension, then the roots of monic polynomials from R[X] lying in F form a ring which is denoted  $\mathcal{O}_F$  and referred to as *ring of integers* of F. Its elements are referred to as being *integral*.

*Example* 1.64. As a principal ideal domain, R is a unique factorisation domain and we have  $\mathcal{O}_{Q(R)} = R$ , so that there are no integral elements strictly above R in Q(R). In particular,  $\mathcal{O}_{\mathbb{Q}} = \mathbb{Z}$  and  $\mathcal{O}_{\mathbb{Q}_p} = \mathbb{Z}_p$  for  $p < \infty$ .

**Lemma 1.65.** If F/Q(R) is algebraic, then every  $\mathcal{O}_F$  module M in F is a free R module of rank [F : Q(R)] In particular, M admits an R basis.

<sup>&</sup>lt;sup>6</sup>This relation is established in detail in [Opi18, 1.10 p. 20].

*Example* 1.66. Let  $0, 1 \neq d \in \mathbb{Z}$  be square free and define

$$b = \begin{cases} \sqrt{d}, & \text{if } d \equiv 2, 3 \mod 4, \\ (1 + \sqrt{d})/2, & \text{if } d \equiv 1 \mod 4. \end{cases}$$

Then,  $\mathcal{O}_F = \mathbb{Z} + b\mathbb{Z}$  for  $F = \mathbb{Q}(\sqrt{d})$ .

- **Definition 1.67.** a) Let  $\operatorname{tr}_{F/Q(R)}(x)$  and  $\operatorname{N}_{F/Q(R)}(x)$  denote the *trace* and *norm* of an element  $x \in F$  for a finite extension, given by the trace and determinant, respectively, of the associated linear Mapping  $M_x : F \to F$ ,  $y \mapsto xy$ .
  - b) The so called *discriminant* of (an R basis  $(\alpha_i)$  of)  $\mathcal{O}_F$  for separable F/Q(R) is given by  $D = \det(\operatorname{tr}_{F/Q(R)}(\alpha_i\alpha_j)_{i,j}) \in R/(R^{\times})^2$ . The discriminant  $D_{F/Q(R)}$  of F/Q(R) is the ideal generated by all discriminants of bases.<sup>7</sup>

*Example* 1.68. For  $\mathbb{Q}(\sqrt{d})$  as above, a computation yields

$$D = \begin{cases} 4d, & \text{if } d \equiv 2, 3 \mod 4, \\ d, & \text{if } d \equiv 1 \mod 4, \end{cases}$$
(1.12)

so that  $F = \mathbb{Q}(\sqrt{d}) = \mathbb{Q}(\sqrt{D}) =: F_D$  and  $\mathcal{O}_F = \mathbb{Z} + \frac{D+\sqrt{D}}{2}\mathbb{Z}$  for all choices of d. Discriminants of the above form correspond to determinants of binary even quadratic lattices over  $\mathbb{Q}$ .

Further, the norm  $q = N_{F_D/\mathbb{Q}}$  restricted to  $\mathcal{O}_{F_D}$  induces the structure of a quadratic lattice with Gram matrix (with respect to the basis  $(1, \frac{D+\sqrt{D}}{2})$ )

$$\begin{pmatrix} 2 & D \\ D & \frac{D^2 - D}{2} \end{pmatrix},$$

so that  $(F \otimes_{\mathbb{Q}} \mathbb{R}, q)$  is of signature

$$(1, 1),$$
 if  $D > 0,$   
 $(2, 0),$  if  $D < 0.$ 

Further, note that  $\operatorname{disc}(\mathcal{O}_F, \mathbf{q}) = (-1)(-D) = D$ . In particular, if  $D = \pm p$  is prime, we find that  $L'/L \simeq (\mathbb{Z}/p\mathbb{Z}, rx^2/p)$  for some  $p \nmid r$ .

<sup>&</sup>lt;sup>7</sup>In fact, this is a principle ideal, so that one basis suffices. As a consequence, the ideal D and generating element D will not be distinguished.

**Definition 1.69.** A finitely generated  $\mathcal{O}_F$  module  $\mathfrak{a}$  is called *fractional ideal*. The collection of such ideals forms an abelian group<sup>8</sup>  $\mathscr{J}_{\mathcal{O}_F} = J_F$  with inversion defined by  $\mathfrak{a}^{-1} = \{x \in F \mid x\mathfrak{a} \subseteq \mathcal{O}_F\}$  and identity element  $\mathcal{O}_F$ . The quotient of  $J_F$  and the principal fractional ideals  $(a) = a\mathcal{O}_F$  for all  $a \in F$  is called *ideal class group* and denoted  $Cl_F$ , so that the following sequence is exact:

$$1 \longrightarrow \mathcal{O}_F^* \longrightarrow F^* \longrightarrow J_F \longrightarrow Cl_F \longrightarrow 1.$$

For  $R = \mathbb{Z}$  and an algebraic number field  $F/\mathbb{Q}$ , the number  $|Cl_F|$  is finite and called *class* number of  $F/\mathbb{Q}$ .

The function  $N_{F_D/\mathbb{Q}}$  defines a quadratic form on  $F_D$ , so that every fractional ideal  $\mathfrak{a}$  of  $F_D$  is naturally equipped with the structure of a quadratic lattice. For an ideal  $0 \neq \mathfrak{a} \triangleleft \mathcal{O}_F$ , the mapping  $\mathfrak{a} \to [\mathcal{O}_F : \mathfrak{a}]$  extends to a homomorphism  $\mathfrak{N} : J_F \to \mathbb{R}_+^{\times}$  which is called *absolute norm*.

Example 1.70. In fact,  $q = N_{F_D/\mathbb{Q}}/\mathfrak{N}(\mathfrak{a}) : \mathfrak{a} \to \mathbb{Z}$  defines an integral (even) quadratic form on the fractional ideal  $\mathfrak{a}$ . This is true since  $N_{F_D/\mathbb{Q}}(a) = \mathfrak{N}(a\mathcal{O}_F)$  by definition, meaning that  $\mathfrak{N}(\mathfrak{a}) \mid N_{F_D/\mathbb{Q}}(a)$  for all  $a \in \mathfrak{a}$ . As a consequence, each fractional ideal is naturally an even  $\mathbb{Z}$  lattice.

As mentioned above, fractional ideals admit a unique factorisation into prime ideals of  $\mathcal{O}_F$  with exponents in  $\mathbb{Z}$ . However, a prime ideal  $\mathfrak{p} \triangleleft R$  might not stay prime when passing to  $\mathcal{O}_F$  (meaning to  $\mathfrak{p} \equiv \mathfrak{p} \mathcal{O}_F$ ), so that there are prime ideals  $\mathfrak{q}_i \triangleleft \mathcal{O}_F$ , such that

$$\mathfrak{p} = \prod_{i=1}^{r} \mathfrak{q}_i^{e_i} \tag{1.13}$$

for adequate  $e_i \in \mathbb{N}_0$  in a unique fashion. The exponents  $e_i$  are called *ramification index* and the degree of the field extension

$$f_i \coloneqq [\mathcal{O}_F / \mathfrak{q}_i : R / \mathfrak{p}]$$

is called *inertia degree* of  $q_i$  over p.

<sup>&</sup>lt;sup>8</sup>Note that every element of this group admits a unique factorisation into prime ideals with exponents in  $\mathbb{Z}$ .

**Proposition 1.71.** If F/Q(R) is separable, then the following fundamental identity holds

$$[F:Q(R)] = \sum_{i=1}^{r} e_i f_i$$

There are the following notions which are advantageous for investigating ramification.

**Definition 1.72.** Let  $\mathfrak{p}$  factor as in (1.13).

- a) The prime ideal  $\mathfrak{p}$  is said to *split completely* or *totally split* in F if r = [F : Q(R)]. On the other hand,  $\mathfrak{p}$  is said to be *nonsplit* or *indecomposed* if r = 1.
- b) Further, the prime ideal  $\mathfrak{q}_i$  in the decomposition of  $\mathfrak{p}$  is said to be unramified over  $\mathcal{O}_F$  if  $e_i = 1$  and  $(\mathcal{O}_F/\mathfrak{q}_i)/(R/\mathfrak{p})$  is separable, otherwise it is called ramified. If, additionally,  $f_i = 1$  for all i, then  $\mathfrak{p}$  is called *totally ramified*.
- c) Further, the ideal  $\mathfrak{p}$  is called *unramified*, if all  $\mathfrak{q}_i$  are unramified, and the extension F/Q(R) is called *unramified*, if all of its prime ideals are unramified in F.

In fact, ramification barely occurs. For a given extension F/Q(R), only finitely many prime ideals are ramified – these are exactly the prime divisors of the so called *different* (ideal). The *different ideal* of F/Q(R) is given by the dual lattice of  $\mathcal{O}_F$  with respect to the bilinear form  $(x, y) \mapsto \operatorname{tr}_{F/Q(R)}(x, y) \coloneqq \operatorname{tr}_{F/Q(R)}(xy)$  and is denoted by  $\mathfrak{d}_F$ .

**Remark 1.73.** Let  $\mathfrak{a}$  be a fractional ideal of F/Q(R) as in Example 1.70. Then the dual ideal can be computed by means of the different ideal via

$$\mathfrak{a}' = \mathfrak{d}_F^{-1}\mathfrak{a}$$

Example 1.74. In case of  $F_D/\mathbb{Q}$ , we find  $\mathfrak{d}_{F_D} = (\sqrt{D})$ , so that  $\mathfrak{a}' = (\sqrt{D})^{-1}\mathfrak{a}$ . Further, for a prime ideal (identified with the number)  $p \in \mathbb{Z}$ , the occurrence of a case described in Definition 1.72 may be identified by means of the associated character:

$$p \text{ splits in } F_D \iff \left(\frac{D}{p}\right) = 1,$$

$$p \text{ remains prime in } F_D \iff \left(\frac{D}{p}\right) = -1$$

$$p \text{ ramifies in } F_D \iff \left(\frac{D}{p}\right) = 0.$$

In order to investigate the global situation, local considerations prove to be useful. This is why the following definition for non-Archimedean fields is established. **Definition 1.75.** A finite field extension of local fields F/K (with K = Q(R)) is called *unramified*, if

$$[F:K] = [\mathcal{O}_F/\mathfrak{m}_F: R/\mathfrak{m}_R],$$

where  $\mathfrak{m}$  refers to the maximal ideal of the respective ring of integers.

Note that by Proposition 1.71, this coincides with Definition 1.72.

# Valuations and global to local relations

This section is interconnected with Section 1.5, as the rings considered in elementary algebraic number theory are exactly these for which valuation rings can be naturally and consistently be associated. In Subsection 1.5 a prime factorisation in  $\mathscr{J}_R$  for special types of rings R was introduced. Rings in which such a factorisation is possible are called *Dedekind domains* and these form a frame predestined to study discrete valuation rings (comparable to  $\mathbb{Z}_p$ ). The reader further interested in the material will find the content provided in this thesis as well as additional relations in [Neu06] and [CF10, Chap. 1 p. 1].

**Definition 1.76.** A discrete valuation ring D is a principal ideal domain that is local, i.e. possesses exactly one maximal ideal  $\mathfrak{m}_D$ .

Given such a ring D, the maximal ideal  $\mathfrak{m}_D$  is generated by an element  $\pi$ , which is uniquely determined up to a unit and called the *uniformiser* of D. Hence, every element  $d \in D$  may be written as  $d = \varepsilon \cdot \pi^n$  with a unit  $\varepsilon \in D$  and  $n \in \mathbb{N}_0$ . The number n is denoted by  $\nu_{\mathfrak{p}}(d)$ , where we write  $\mathfrak{p} = \mathfrak{m}_D$ , and is called the *valuation* of d. The mapping extends uniquely to the quotient field of D and behaves in such a fashion, that  $d \mapsto \rho^{\nu_{\mathfrak{p}}(d)}$ for  $0 < \rho < 1$  defines a non-Archimedean absolute value function  $|\cdot|_{\mathfrak{p}}$  on Q(D).<sup>9</sup>

**Definition 1.77.** Let R be an integral domain, but not a field. Then, R is called *Dedekind* domain if every proper ideal factors into primes. Equivalently, it may be characterised by its localisations at every prime ideal  $\mathbf{p}$  being discrete valuation rings.

*Example* 1.78. Let  $R = \mathbb{Z}$ , then every prime ideal  $\mathfrak{p}$  is represented by a prime number p. The localisation at  $\mathfrak{p}$  identifies with

$$\mathbb{Z}_{\mathfrak{p}} = \left\{ \frac{x}{y} \mid x \in \mathbb{Z}, y \notin \mathfrak{p} \right\},\$$

<sup>&</sup>lt;sup>9</sup>In fact, every non-trivial absolute value function  $|\cdot|$  on Q(D) with  $|D| \leq 1$  must be of this form.

meaning any number in  $\mathbb{Q}$ , whose nominator is not divided by p. The associated absolute value function is given by  $z \mapsto |p|^{-\nu_p(z)}$  in accordance with Example 1.40. We see that  $\mathbb{Z}_p$ is the unit ball with respect to this abs and that in fact  $\mathbb{Q}_p$  is the completion of  $Q(\mathbb{Z}_p)$ along the abs. The same topological relation holds for  $\mathbb{Z}_p$  and  $\mathbb{Z}_p$ .

Therefrom, the prime ideals  $\mathfrak{p}$  of a Dedekind domain R give rise to complete non-Archimedean fields  $\overline{Q(R)}^{|\cdot|_{\mathfrak{p}}} =: Q(R)_{\mathfrak{p}}$ , similar to  $\mathbb{Q}_p$  over  $\mathbb{Q} = Q(\mathbb{Z})$  with ring of integers  $\overline{R_{\mathfrak{p}}} = \{x \in Q(R)_{\mathfrak{p}} \mid |x|_{\mathfrak{p}} \leq 1\}$  and maximal ideal  $\mathfrak{m}_{\overline{R_{\mathfrak{p}}}} = \{x \in Q(R)_{\mathfrak{p}} \mid |x|_{\mathfrak{p}} < 1\}$ . These fields might be considered in order to characterise the field Q(R) or compute invariants of Q(R). Usually, Q(R) is chosen to be a global field, in our setting a finite extension of  $\mathbb{Q}$  and the prime ideals  $\mathfrak{p}$  of  $\mathcal{O}_{Q(R)}$  are called *places* of Q(R) or  $\mathcal{O}_{Q(R)}$ , while the fields  $Q(R)_{\mathfrak{p}}$  are referred to as *local fields*<sup>10</sup> associated to Q(R).

In many cases the investigation of global properties might be reduced to the computation of local objects. A few supplementing examples will be presented in the following. For instance,

$$\mathscr{J}_R = \bigoplus_{\mathfrak{p} \text{ prime}} \mathscr{J}_{R_\mathfrak{p}} = \bigoplus_{\mathfrak{p} \text{ prime}} \mathscr{J}_{\overline{R_\mathfrak{p}}}, \qquad (1.14)$$

so that global fractional ideals are completely described in terms of local ideals.<sup>11</sup> This should not be surprising, as  $\mathscr{J}_{\overline{R_p}}$  is generated by one element, exactly the prime ideal  $\mathfrak{p}$ and the left hand side of (1.14) equals the free group in exactly these prime ideals. In fact, the local objects are almost always trivial, so that only finitely many computations have to be carried out. For instance, consider two R modules M, N in a Q(R) vector space V, which span V over Q(R). Then for almost all  $\mathfrak{p}$ , we find  $R_{\mathfrak{p}}M = R_{\mathfrak{p}}N$ . Further,  $M = \bigcap_{\mathfrak{p} \text{ prime }} R_{\mathfrak{p}}M$ .

*Example* 1.79. Let L be a  $\mathbb{Z}$  lattice in  $\mathbb{Q}^n$ , then  $\mathbb{Z}_p L = \mathbb{Z}_p^n$  for almost all p and

$$\bigcap_{p < \infty} \mathbb{Z}_p L = L$$

Next, two invariants of classical algebraic number theory are related to the local case, as they are useful for our proceedings. For two free<sup>12</sup> R modules M, N in a finite dimensional

<sup>&</sup>lt;sup>10</sup>This notion might be defined rigorously and abstractly.

 $<sup>^{11}\</sup>mathrm{Note}$  that this implies that the class number is locally finite as well.

<sup>&</sup>lt;sup>12</sup>In fact, requiring the modules to be free is not necessary, as by equation (1.14) the index can be defined locally and gives rise to one and only one ideal in  $\mathscr{J}_R$ .

Q(R) vector space V, there is a linear isomorphism  $\sigma: M \to N$  that extends to V. Define

$$[M:N]_R \coloneqq R\det(\sigma)$$

as a principal ideal, the *module index* of N in M. It agrees with the group index, if  $R = \mathbb{Z}$  and  $M \supseteq N$ .

**Definition 1.80.** The *discriminant* of an R module spanning a finite dimensional nondegenerate quadratic Q(R) vector space V is defined to be

$$D(M) \coloneqq D(M/R) \coloneqq [M':M]_R,$$

where M' is the dual module with respect to the bilinear form associated to V (cf. Section 1.1).

Note that  $D(R_{\mathfrak{p}}M/R_{\mathfrak{p}}) = D(M/R)R_{\mathfrak{p}}$  as well as  $D(\overline{R}M/\overline{R}) = D(M/R)\overline{R}$ , and that in case of M being free,

$$D(M) = (\det(S))$$

where S is the Gram matrix to the trace on M. This, in fact, accords with Definition 1.67. Similarly to the above, the following relation may be established for duals:

$$(M\overline{R})' = M'\overline{R}.$$

Last, the discriminant and different ideal behave well when splitting considerations to local field extensions.

**Remark 1.81.** For a prime ideal  $\mathfrak{p}$  in R and prime ideals  $\mathfrak{B} \mid \mathfrak{p}$  in  $\mathcal{O}_F$ , we find

$$\mathfrak{d}(\mathcal{O}_F/R)\overline{\mathcal{O}_{F,\mathfrak{B}}} = \mathfrak{d}(\overline{\mathcal{O}_{F,\mathfrak{B}}}/R_\mathfrak{p}) \qquad \text{and} \qquad D(\mathcal{O}_F/R)\overline{R_\mathfrak{p}} = \prod_{\mathfrak{B}|\mathfrak{p}} D(\overline{\mathcal{O}_{F,\mathfrak{B}}}/\overline{R_\mathfrak{p}})$$

**Remark 1.82.** The different ideal and discriminant give rise to a class of easily accessible lattices and are important constructions. Note that a lattice L in<sup>13</sup>  $F = \mathbb{Q}_p(\sqrt{D})/\mathbb{Q}_p$ is distinct from its dual  $\mathfrak{d}_F L$  (cf. Remark 1.73) if, and only if,  $\mathfrak{d}_F$  is not trivial, which happens if, and only if, there is a prime ideal  $\mathfrak{B}$  in  $\mathcal{O}_F$  that divides  $\mathfrak{d}_F$ . This case occurs only if  $p \mid D$ , as  $\mathfrak{d}_F = (\sqrt{D})$  (cf. Example 1.74) and (p) is the only prime ideal in  $\mathcal{O}_{\mathbb{Q}_p}$ .

<sup>&</sup>lt;sup>13</sup>We explicitly state that D is assumed to be the discriminant of the extension, as the statement fails in case of p = 2, otherwise.

We conclude: the lattice  $(\mathfrak{a}, N/N(\mathfrak{a}))$  in  $\mathbb{Q}_p(\sqrt{D})/\mathbb{Q}_p$  is not unimodular if, and only if,  $\mathbb{Q}_p(\sqrt{D})/\mathbb{Q}_p$  is ramified which is equivalent to  $p \mid D$ .

# 2 Classical modular forms

Modular forms are at the origin of the present thesis. Some notions and theorems of the classical theory are necessary for the later investigations in Sections 3.3 and 7.3. These are collected in a compact fashion in Section 2.2 of this chapter.

Apart from that, the author decided to devote a whole section to the motivation of classical modular forms, in order to build up an intuition for readers who are not familiar with a conceptual approach to modular forms. The path followed can be found in [DS05], [GBL10, III.59 p. 250], and [FB06, V p. 255], whereas introductory material offering different perspectives is, for instance, found in e.g. [Dei10], [KK98], [Iwa95], [Bum98], [Hec26], and [Bor97]. The curious reader will find elaborate material in [BC79], which is beyond the scope of this thesis, as well as a brief inspiring introduction to the general theory in [Bor65].

### 2.1 Introduction

Familiarity with at least classical modular forms<sup>1</sup> was an advantage in reading this thesis. Due to the sheer complexity<sup>2</sup> of the theory and the broad mathematical requirements to grasp its content, its introduction is often not realised conceptually. Nevertheless, a rough idea as well as fundamental concepts are discussed in this overview, which is based on a talk given by Solomon Friedberg at ICERM in 2013 [Fri13].

Based on mathematical experience, functions on a topological group G which are invariant under the action of a discrete (compact) subgroup  $\Gamma$  (or K) are particularly interesting. In a concrete setting, additional conditions may be imposed on these functions and the action of  $\Gamma$  frequently involves characters of G.

<sup>&</sup>lt;sup>1</sup>These associated with  $\mathrm{SL}_2(\mathbb{Z})$  on  $\mathrm{SL}_2(\mathbb{R})/\mathrm{SO}_2(\mathbb{R}) \simeq \mathbb{H}$  which are also referred to as *elliptic modular* forms.

<sup>&</sup>lt;sup>2</sup>Cf. Robert Langlands on automorphic forms: 'It is a deeper subject than I appreciated and, I begin to suspect, deeper than anyone yet appreciates. To see it whole is certainly a daunting, for the moment even impossible, task.' according to [Fri13]

Example 2.1. Let  $G = \mathbb{R}$  and  $\Gamma = \mathbb{Z} < \mathbb{R}$ , then a  $\mathbb{Z}$  invariant function f is called *periodic* and gives rise to a Fourier expansion under reasonable circumstances (cf. [Con94, I.5 Thm. 5.11 p. 22]). In this sense,  $\mathbb{Z}$  acts on  $\mathbb{R}$  as a group if automorphism and f is an *automorphic function* (invariant under the action) with respect to  $\Gamma$  on G. In particular fmay be regarded as a function on the quotient  $\mathbb{R}/\mathbb{Z} \sim \mathbb{S}^1$  and by considering functions on  $\mathbb{S}^1$  all automorphic functions on  $\mathbb{R}$  with respect to  $\mathbb{Z}$  are induced.

- **Remark 2.2.** a) In case of  $G = \operatorname{SL}_2(\mathbb{R})$  and a congruence subgroup  $\Gamma$  the theory of classical modular forms is obtained. Recall that by the Iwasawa decomposition  $\mathbb{H} \simeq \operatorname{SL}_2(\mathbb{R})/\operatorname{SO}_2(\mathbb{R})$ , where the latter group is compact. A function f on G to  $\mathbb{C}$  which is right invariant under  $K = \operatorname{SO}_2(\mathbb{R})$  and invariant under  $\Gamma$  via the operation of the *Petersson operator* is called *classical modular form* if it fulfils certain analytic properties.<sup>3</sup> Usually, holomorphy or meromorphy is assumed. By imposing weaker conditions, for instance, real smoothness and satisfying certain differential equations, involving the hyperbolic Laplace operator, Maas forms are obtained (cf. [Bum98, 1.9 pp. 103-118]). This opens up the field to functional analysis, particularly operator theory, and representation theory. The latter was incorporated into the theory of automorphic forms by Gelfand and Fomin 1952 (cf. [GF52]<sup>4</sup>, while a reference for the former application is found in [Iwa95].
  - b) In the adelic setting, similar functions may be considered. For instance,  $G = \mathbb{A}$  may be chosen along with functions f which are invariant under the operation of  $\Gamma = \mathbb{Q}$ (Note that  $\mathbb{Q} < \mathbb{A}$  is discrete and the quotient is compact). Further, the group  $(\mathbb{A}/\mathbb{Q})^*$ may be identified with  $\mathbb{Q}$  (in fact, with discrete topology - cf Remark 1.55), so that fadmits a Fourier series analogously to the case  $\mathbb{R}/\mathbb{Z}$ . If G is set to be  $\mathrm{GL}_1(\mathbb{A}) = \mathbb{A}^{\times}$ and  $\Gamma = \mathrm{GL}_1(\mathbb{Q}) = \mathbb{Q}^{\times}$ , automorphic forms are multiplicative functions on G, which are invariant under  $\Gamma$  and called *Hecke* characters (cf. Section 1.3).
  - c) Similarly, adelisations of rational groups may be considered, for instance,  $SL_2(\mathbb{A})$  (cf. [Bor91]). In this setting certain functions which transform with a character under a compact subgroup K from the right and are invariant under  $SL_2(\mathbb{Q})$  from the left may be related to classical modular forms (cf [Bum98, 3.6 p. 341] and the concrete example in Section 5.2.1), providing a broader context for classical modular forms.

 $<sup>^{3}</sup>$ In fact, this classical setting allows for a colourful geometric interpretation, discussed over the course of the subsequent sections.

<sup>&</sup>lt;sup>4</sup>An English translation of that source is available by the American Mathematical Society.

#### 2.1.1 A geometric approach

The viewpoints on automorphic forms are manifold. However, the classical case admits a certain concrete geometric interpretation which shall be discussed over the course of the subsequent sections. In short, classical automorphic forms (a generalised notion of modular forms) are global meromorphic differentials of the compactified coarse moduli space of complex elliptic curves. The following sections are devoted to fill these words with meaning for the reader unfamiliar with the aforementioned concepts.

#### Elliptic curves

At the origin of the geometrical approach lie *elliptic curves*, objects which are of a particular interest for number theorists. In fact, the ancient Greeks had already investigated equations such as

$$aX + bY = c$$
  $a, b, c \in \mathbb{N}$ 

and asked for its solutions for X and Y. In a modern context, the coefficients would be chosen from  $\mathbb{Z}$  and it is well known, that the equation always has infinitely many solutions over  $\mathbb{Q}$  and admits an integer solution, if, and only if,  $gcd(a, b) \mid c$ . All of these solutions are computable via the Euclidean Algorithm.

An immediate problem arising from the above is to determine the solution of equations of the form

$$aX^n + bY^n = c$$

for natural numbers n which is exactly *Fermat's last Theorem* for a, b, c = 1. Questions of the type presented above are the subject of *Diophantine Geometry*. In fact, the case of a quadratic polynomial is solved by the Theorem of *Hasse–Minkowsky*, which relates solvability over  $\mathbb{Q}$  to solvability over all p-adic numbers  $\mathbb{Q}_p$ . In this case there are infinitely many rational solutions. If, however, the degree of an irreducible polynomial is greater than 4, the number of rational solutions is finite, which was proven by Faltings<sup>5</sup> (1983) and conjectured by Mordell (1922).

At the verge of these cases lies the class of elliptic curves which correspond to cubic polynomials. They can have infinitely many rational solutions or finitely many, manifesting a bridge between both cases. This feature alone suffices to render them interesting for

 $<sup>{}^{5}</sup>$ In fact, he received the fields medal for his proof of the conjecture in 1986.

number theorists.

In the following, we assume the characteristic of the field F to be different from 2 and 3.<sup>6</sup>

**Definition 2.3.** An *elliptic curve* over a field F, written E/F is given by a cubic equation

$$Y^2 = X^3 + aX + b (2.1)$$

where  $a, b \in F$  and the associated discriminant  $\Delta := -2^4(4a^3 + 27b^2)$  does not vanish. Equation (2.1) is called affine Weierstraß equation and the parameter  $j := -1728 \frac{(4a)^3}{\Delta}$  is called *j*-invariant.

More geometrically, an elliptic curve is defined to be a projective, non-singular curve of genus 1 with a distinguished point. This results in the curve E having a cubic Weierstraß equation, which may look far more complicated than the above but may be transformed into that form via rational coordinate transformations, provided char $(F) \neq 2, 3$ . The condition on the discriminant translates to the curve being non-singular and the *j*-invariant parametrises elliptic curves over a field. The distinguished point may be chosen to be  $\infty$  in the projective space. Compare Figure 2.1.1 for illustrations.

Nonetheless, there is more to elliptic curves than points in a plane – an algebraic structure! In fact, the Theorem of Riemann–Roch induces a natural addition on an elliptic curve, implying that every such curve can be understood as an abelian variety. A theorem of Mordell–Weil states that for any number field F the elliptic curve E/F as a group bears the structure of a finitely generated abelian group. The rank of it's free component is called *rank* of the elliptic curve and determines whether there are infinitely many solutions or not. In fact, the rank of an elliptic curve is the subject of the famous conjecture of Birch and Swinnerton-Dyer. Their conjecture relates the Laurent coefficients at a critical point of an L-series associated to the elliptic curve to different parameters of the curve. In particular, the pole order at the critical point is conjectured to equal the rank of the curve.

The aim of the following sections is to construct the coarse moduli space of complex elliptic curves with the aim of realising classical automorphic forms as meromorphic differentials on that space.

 $<sup>^{6}</sup>$ This assumption guarantees that the Weierstraß equation can then be chosen to be much simpler.



Figure 3.1: Three elliptic curves



Figure 3.2: Two singular cubic curves.

Figure 2.1.1: Illustration of elliptic curves over  $\mathbb{R}$  (cf. [Sil13, III.1 p. 43]). Indeed, cubic curves are smooth if, and only if,  $\Delta \neq 0$ . Also note that the last equation transforms to  $y^2 = x^3 - 3x + 2$  under  $x \mapsto (x - 1)/3, y \mapsto y/3$ , so it can be transformed to match Definition 2.3.

#### The coarse moduli space of complex elliptic curves

In order to construct the desired moduli space, a concrete realisation of complex elliptic curves is advantageous. Hence, these are related to tori in the following by the well known Weierstraß  $\wp$  function.

Let  $\Lambda < \mathbb{C}$  be a lattice of rank 2. Similar to the examples above,  $\Lambda$ -automorphic functions  $f : \mathbb{C} \to \overline{\mathbb{C}}$  may be considered, meaning functions that fulfil

$$\forall \lambda \in \Lambda, z \in \mathbb{C} : f(z + \lambda) = f(z).$$

A naive attempt in order to construct such a function would be the summation over simple poles at the lattice points, which fails due to divergence. The next natural candidate is a sum over poles of order 2 which may be realised as a convergent series with the following correction terms:

$$\wp_{\Lambda}: \mathbb{C} \to \overline{\mathbb{C}}, \qquad z \mapsto \frac{1}{z^2} + \sum_{0 \neq \lambda \in \Lambda} \left( \frac{1}{(z-\lambda)^2} - \frac{1}{\lambda^2} \right).$$

This function is called Weierstraß p-function and represents a meromorphic  $\Lambda$ -automorphic function on  $\mathbb{C}$ .<sup>7</sup> In fact,  $\wp_{\Lambda}$  may be used to classify elliptic functions. First, note that every even elliptic function can be written as a rational function in  $\wp_{\Lambda}$  (cf. [FB06, Thm. 3.2 p. 275]). But the differential of  $\wp_{\Lambda}$  is odd, so that the field of elliptic functions must be a quadratic extension of  $\mathbb{C}(\wp_{\Lambda})$ . A direct computation confirms that the associated minimal polynomial is given by

$$(\wp_{\Lambda}')^2 = 4 \cdot \wp_{\Lambda}^3 - g_2 \cdot \wp_{\Lambda} - g_3, \qquad (2.2)$$

where

$$g_2 = 60 \sum_{\lambda \in \Lambda \setminus \{0\}} \lambda^{-4}, \qquad g_3 = 140 \sum_{\lambda \in \Lambda \setminus \{0\}} \lambda^{-6}.$$

These coefficients are called *Eisenstein series* and represent modular forms<sup>8</sup>. Nonetheless, equation (2.2) and (2.1) are of the same form, so that it appears to be not too far off to seek relating the associated solution sets. To be precise, the map to projective space

$$\mathbb{C}/\Lambda \ni z \mapsto [\wp_{\Lambda}(z), \wp'_{\Lambda}(z), 1] \in \mathbb{P}^{2}(\mathbb{C})$$

<sup>&</sup>lt;sup>7</sup>Such a function, meaning an  $\Lambda$ -automorphic function on  $\mathbb{C}$ , is called *elliptic function*. The name is derived from the fact that these functions were discovered to occur as inverses to so called elliptic integrals – integrals which parametrise the arc length of an ellipse.

<sup>&</sup>lt;sup>8</sup>However, this is not evident at the moment.

is a natural candidate, which maps 0 to infinity. In fact the map is bijective, in the sense that it relates elements of the Torus  $\mathbb{C}/\Lambda$  to elements in  $\mathbb{P}^2(\mathbb{C})$  which fulfil equation (2.2) for independent variables  $(\wp_\Lambda, \wp'_\Lambda)$ . The associated elliptic curve is denoted  $E_\Lambda/\mathbb{C}$ . Exporting the group structure from  $\mathbb{C}/\Lambda$  yields exactly the addition induced by the Theorem of Riemann–Roch, so that the above mapping is, in fact, a biholomorphic homomorphism  $\mathbb{C}/\Lambda \simeq E_\Lambda/\mathbb{C}$ . Two of these tori  $\mathbb{C}/\Lambda$  and  $\mathbb{C}/\Lambda'$  can be shown to be biholomorphically equivalent<sup>9</sup> if, and only if,  $\Lambda' = \alpha \Lambda$  for some  $\alpha \in \mathbb{C}^{\times}$ , meaning they arise from each other by homogeneous stretching and rotation.

After complex elliptic curves have been discussed briefly, the construction of the associated moduli space will be carried out. The approach below comprises the topological realisation followed by an outline of equipping it with a complex structure as well as its compactification.

In order to parametrise a space of elliptic curves, considering bases of lattices appears to be an advantageous starting point. A basis of a lattice  $\Lambda$  is given by a pair of complex noncollinear numbers  $(z_1, z_2)$ . Since scaling does not change the class of the quotient  $\mathbb{C}/\Lambda$  we are interested in, considering  $\tau := \frac{z_1}{z_2}$  suffices. Further, note that changing the orientation of the basis does not give rise to another lattice, so that  $\tau \in \mathbb{H} = \{z \in \mathbb{C} \mid \text{Im}(z) > 0\}$ may be assumed, effectively fixing an orientation. In summary, the space  $\mathcal{B}$  of unordered bases of lattices modulo scaling by  $\mathbb{C}^{\times}$  may be viewed as the upper half plane  $\mathbb{H}$ :

$$\mathcal{B}/\mathbb{C}^{\times} \simeq \mathbb{H}.$$

However, this is not a one to one description of elliptic curves, since there is in general more than one basis for a lattice. In fact, a suitable  $\mathbb{Z}$  linear combination of basis elements  $(z_1, z_2)$  represents a basis of the same lattice again. Such a base change can be identified with a matrix M with coefficients in  $\mathbb{Z}$  and the most primitive examples of these are translation and rotation:

$$T(z_1, z_2) = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix} (z_1, z_2) = (z_1 + z_2, z_2), \qquad S(z_1, z_2) = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} (z_1, z_2) = (-z_2, z_1).$$

Abusing the information of reversibility of this base change

$$M \in \operatorname{GL}_2(\mathbb{Z}) = \operatorname{SL}_2(\mathbb{Z}) \cup \operatorname{SL}_2(\mathbb{Z}) \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}$$

<sup>&</sup>lt;sup>9</sup>In fact, such a biholomorphism is a group isomophism if, and only if, it preserves 0, as the biholomorphisms are identified to be of the form  $x \mapsto \alpha x + \beta$  by lifting them to the covering spaces  $\mathbb{C}$ , which only allow for affine linear transformations as biholomorphisms.

is derived. Since orientation had already been fixed, only base changes from  $SL_2(\mathbb{Z}) =: \Gamma(1)$ need to be considered which form a complete set of representatives. Note that the action of  $SL_2(\mathbb{Z})$  and  $\mathbb{C}^{\times}$  commute on  $\mathcal{B}$ , so that the *Modular curve* 

$$\operatorname{SL}_2(\mathbb{Z}) \setminus \mathcal{B} / \mathbb{C}^{\times} \simeq \operatorname{SL}_2(\mathbb{Z}) \setminus \mathbb{H} =: Y(\Gamma(1))$$

is well defined. In order to conclude that  $\gamma \in \Gamma(1)$  operates via *Möbius transformations*<sup>10</sup> on  $\mathbb{H}$  note that

$$\gamma\left(\begin{smallmatrix} \tau\\1 \end{smallmatrix}\right) = \begin{pmatrix} a\tau+b\\c\tau+d \end{pmatrix} \simeq \begin{pmatrix} \frac{a\tau+b}{c\tau+d}\\1 \end{pmatrix} \in \mathbb{H}.$$

In order to picture the geometry which will be induced on this quotient set, a fundamental domain which is a connected full set of representatives in  $\mathbb{H}$  will be considered.<sup>11</sup> Such a domain for  $Y(\Gamma(1))$  is given by

$$\mathcal{F}_{\Gamma(1)} = \left\{ \tau \in \mathbb{H} \mid |\operatorname{Re}(\tau)| \le \frac{1}{2}, |\tau| \ge 1 \right\}$$
(2.3)

and is illustrated in Figure 2.1.2.

In order to glue the fundamental domain properly to a geometric representation of  $Y(\Gamma(1))$ , ambiguities have to be investigated. While the interior of  $\mathcal{F}_{\Gamma(1)}$  does not contain multiple representatives of one class, its boundary is full of redundance which is illustrated in Figure 2.1.3. Utilising that  $\mathcal{F}_{\Gamma(1)}$  is a fundamental domain for  $\Gamma(1)$ , the identity

$$\operatorname{SL}_2(\mathbb{Z}) = \langle S, T \rangle$$

may be derived rendering the examples of transformations presented essentially exhaustive.

The next step would be the introduction of a complex structure for the topological space  $Y(\Gamma(1))$  as well as its compactification.

In order to endow  $Y(\Gamma(1))$  with the desired structure, charts have to be constructed. Around most points of the quotient, this is trivial, since the natural projection from  $\mathbb{H}$  to  $Y(\Gamma(1))$  is one to one in a sufficiently small neighbourhood. The only exceptions to this are the image of *i* and  $\rho$ . This phenomenon occurs due to the fact that these points in  $\mathbb{H}$  have a non-trivial stabiliser in  $\mathrm{SL}_2(\mathbb{Z})$ , rendering the projection not injective on any neighbourhood. To address this problem, the *Cayley* transform is applied to transfer the setting to the open unit disc, where it is solved via unwrapping (details are treated in

<sup>&</sup>lt;sup>10</sup>In fact, this computation and the above motivation clarify the sudden appearance of Möbius transformations in the vast majority of textbooks on this subject.

<sup>&</sup>lt;sup>11</sup>In fact, ambiguities on null sets are neglected.



Figure 2.1.2: Illustration of  $\mathcal{F}_{\Gamma(1)}$  (blue). Its border is marked in orange in addition to the upper half of the unit circle (grey) for the purpose of orientation. The corners match  $\rho := \exp(2\pi i/6)$  and  $\rho^2$ .

[DS05, 2.2 p. 50]).

Next, the curve  $Y(\Gamma(1))$  will be compactified, offering the application of advanced tools from algebraic geometry to study the resulting object. With this intention, the following chart, compatible with the above, is considered. Note that

$$\mathbb{H} \ni \tau \mapsto e^{2\pi i\tau} \in \dot{B}_1(0)$$

induces a biholomorphic map from  $\mathbb{H}/\langle T \rangle$  (a vertical strip) to the open unit disc without zero  $\dot{B}_1(0)$ . Restricting this map to all elements  $\tau \in \mathbb{H}$  with  $\mathrm{Im}(\tau) > c$  for a fixed number c > 1 then provides a biholomorphic map from a vertical strip to a smaller open disc without its centre.<sup>12</sup> The situation is illustrated in Figure 2.1.4. The consequential step is

 $<sup>^{12}</sup>$ As a consequence, every periodic (in real direction) function that is holomorphic on an upper half plane



Figure 2.1.3: Illustration of ambiguities in  $\mathcal{F}_{\Gamma(1)}$  together with the transformations they arise from. The orange lines correspond to the intersection of  $\mathcal{F}_{\Gamma(1)}$  and its transform.

to formally add an element ' $\infty$ ' to  $Y(\Gamma(1))$  and declare the geometry around it by the coordinate map  $\psi$ . This process yields a compactification of the curve  $Y(\Gamma(1))$  which is denoted by  $X(\Gamma(1))$  and referred to as the *Baily-Borel compactification*.

has a Fourier series. This is seen by pulling the function to the punctured unit disc, where it has a Laurent series around 0. Resubstituting the variable for  $e^{2\pi i\tau}$  yields the desired Fourier series on an upper halfplane, which is truncated to the left. From this point of view, being holomorphic in  $\infty$  is meaningful and translates to all Fourier coefficients being zero for negative indices.



Figure 2.1.4: Illustration of the construction of a chart at ' $\infty$ '. The blue arrow indicates the geometrical correspondence, while the orange lines are mapped to each other.

#### Classical modular forms as differentials

Previously, the space  $X(\Gamma(1))$  has been constructed as a compactification of an important geometric description of elliptic curves. Hence, it appears natural to investigate functions on this space. Note that due to compactness, all holomorphic functions on  $X(\Gamma(1))$ are trivial, leading to meromorphic functions as the next potential class to examine. However, *Klein*'s *j*-invariant can be proven to be a biholomorphism of  $X(\Gamma(1))$  and the Riemann sphere, mapping  $\infty$  to  $\infty$ , so that *j* parametrises elliptic curves over  $\mathbb{C}$ . As a consequence, meromorphic functions on  $X(\Gamma(1))$  accord with rational functions in *j*, resulting in an immense space with a very simple structure. Fortunately, there is a natural class of function like objects attached to  $X(\Gamma(1))$ , which yields finite dimensional spaces of utmost significance in number theory. Instead of recognising functions only, meromorphic differentials might be considered.

The sheaf of meromorphic differentials of degree n on  $\mathbb{C}$  is given by associating to each open set  $V \subseteq \mathbb{C}$  objects

$$f(q)(\mathrm{d}q)^n$$
,

where f is a meromorphic function on V. These are identified as sections through tensor powers of the cotangent space. The above definition translates instantly to manifolds and we shall be interested in global sections, leading to a compatibility criterion for local descriptions with respect to coordinate transformations. Let g be such a transformation, then we have

$$g^*(f(q)(\mathrm{d}q)^n) = f(g(q))g'(q)^n(\mathrm{d}q)^n,$$

where  $g^*$  denotes the pullback along g. However, it is not evident how to construct differentials on  $X(\Gamma(1))$  unless a different perspective is chosen. In fact, a differential might be pulled back along the natural projection  $\pi : \mathbb{H} \to X(\Gamma(1))$ . The resulting differential  $f(\tau)(d\tau)^n$  must then be invariant under pullbacks from  $\Gamma(1)$ , so that

$$f(\tau)(d\tau)^n = \gamma^* (f(\tau)(d\tau)^n)$$
  
=  $f(\gamma\tau)(\gamma(\tau)')^n (d\tau)^n$   
=  $f(\gamma\tau) \left(\frac{a(c\tau+d)-c(a\tau+b)}{(c\tau+d)^2}\right)^n (d\tau)^n$   
=  $f(\gamma\tau) \left(\frac{ad-bc}{(c\tau+d)^2}\right)^n (d\tau)^n$   
=  $\frac{f(\gamma\tau)}{(c\tau+d)^{2n}} (d\tau)^n$ 

has to be true for all  $\gamma \in \Gamma(1)$ . We introduce  $j(\gamma, \tau) := (c\tau + d)$ , the so called *factor of automorphy* and observe that j defines a cocycle:

$$j(\gamma \cdot \gamma', \tau) = j(\gamma, \gamma'\tau)j(\gamma', \tau).$$
(2.4)

Further, having the transformation property of  $f(d\tau)^n$  in mind, we define the *Petersson* operator of weight  $k \in \mathbb{Z}$  given by

$$(f|_k\gamma)(\tau) = j(\gamma,\tau)^{-k} f(\gamma\tau).$$
(2.5)

This defines a right operation of  $\Gamma(1)$  on the space of meromorphic (holomorphic) functions on  $\mathbb{H}$ . In addition to being invariant under pullbacks from  $\Gamma(1)$ , the differentials  $f(d\tau)^n$ have to be meromorphic in  $\infty$ , which translates to their Fourier expansion having only finitely many coefficients of negative index.

Combining the information above results in a bijective correspondence of meromorphic differentials of degree k/2 (for  $k \in 2\mathbb{N}_0$ ) on  $X(\Gamma(1))$  and the following objects.

**Definition 2.4.** An *automorphic form* of weight  $k \in \mathbb{Z}$  on  $\mathbb{H}$  for  $\Gamma(1)$  is a meromorphic function f on  $\mathbb{H}$ , that fulfils the following properties.

- i) f is invariant with respect to the Petersson operator  $|_k$  for all  $\gamma \in \Gamma(1)$ .
- ii) f is meromorphic in  $\infty$ .

This alternative description via functions on  $\mathbb{H}$  is a convenient tool for constructing such differentials on  $X(\Gamma(1))$ . The resulting spaces will be infinite dimensional, but the alternative definition suggests the following modification. Instead of demanding f to be meromorphic, this property might be replaced by being holomorphic<sup>13</sup>, resulting in the definition of classical modular forms which span finite dimensional spaces  $\mathcal{M}_k(\Gamma(1))$  for each k:<sup>14</sup>

$$\dim(\mathcal{M}_k)_{k\in\mathbb{N}} = (0, 0, 0, 1, 0, 1, 0, 1, 0, 1, 0, 2, \ldots).$$

Further, instead of considering  $\Gamma(1)$  as a group of automorphisms on  $\mathbb{H}$ , other discrete groups may be studied. A fruitful setting is given by *Fuchsian* groups  $\Gamma$  (cf. [Iwa95,

<sup>&</sup>lt;sup>13</sup>This condition translates to orders of vanishing in the cusps and fix points of  $\Gamma(1)$  for differentials, so it is not immeditely intuitive from a geometrical viewpoint.

<sup>&</sup>lt;sup>14</sup>In fact, by i), every holomorphic modular form is 0 for even  $k \in \mathbb{N}$ . Upper bounds for the other cases up to 12 are found by means of the residue theorem of complex Analysis and are met by Eisenstein series and the Ramanujan  $\tau$ -function  $\Delta$ .

Chap. 2 p. 39]), which act discontinuously, rendering the quotient  $\Gamma \setminus \mathbb{H}$  a  $T_2$  space, so that it may be endowed with the structure of a manifold. Popular choices for groups are special subgroups of  $\Gamma(1)$ , introduced in the following.

Let  $\Gamma(N)$  denote the kernel of the natural projection

$$\operatorname{SL}_2(\mathbb{Z}) \to \operatorname{SL}_2(\mathbb{Z}/N\mathbb{Z})$$

The index of  $\Gamma(N)$  in  $\Gamma(1)$  is finite, so that a set of representatives of  $\Gamma(N) \setminus \mathbb{H}$  is constructed by gluing finitely many translates of  $\mathcal{F}_{\Gamma(1)}$ . A group  $\Gamma$  lying in between  $\Gamma(N) \leq \Gamma \leq \Gamma(1)$  is called a *congruence subgroups*. These play a major role in the classical theory of modular forms. The reason why restricting the theory to these groups is that the associated spaces of modular forms offer a rich operator theory, called *Hecke* theory (cf. Section 6.2). This theory proves valuable when modular forms are associated to *L*-series or when elementary geometry<sup>15</sup> is considered (cf. [Bum98, 1.4 p. 41]). We shall have a glance at common examples of congruence subgroups: the group

$$\Gamma_0(N) \coloneqq \left\{ \gamma \in \Gamma(1) \middle| \gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix}, c \equiv 0 \mod N \right\}$$

for some  $N \in \mathbb{N}$  occurs naturally, when considering elliptic curves with torsion data and we clearly find  $\Gamma(N) \leq \Gamma_0(N)$ . Further,  $\Gamma_1(N)$  is defined to be the kernel of

$$\Gamma_0(N) \ni \gamma \mapsto d \in (\mathbb{Z}/N\mathbb{Z})^{\times}$$

and  $\Gamma(N)$  equals the kernel of the homomorphism

$$\Gamma_1(N) \ni \gamma \mapsto b \in \mathbb{Z}/N\mathbb{Z}.$$

Both projections are surjective, so that the number of cosets may be computed elementarily. In summary,

$$\Gamma(N) \trianglelefteq \Gamma_1(N) \trianglelefteq \Gamma_0(N) \le \operatorname{SL}_2(\mathbb{Z}).$$

At this point, the setting of a group  $\Gamma$  acting discontinuously on the upper half plane  $\mathbb{H}$  might appear to be too specific unless the universal cover property of  $\mathbb{H}$  is taken into account which clarifies the broadness of this choice. In fact, any compact Riemann

<sup>&</sup>lt;sup>15</sup>There is a scalar product on the space of cusp forms, the Petersson product, which is compatible with Hecke operators.
surface M of genus strictly greater than 1 has the upper half plane  $\mathbb{H}$  as its universal cover and its fundamental group  $\pi_1(M)$  may be realised as a group  $\Gamma$  that acts discontinuously on  $\mathbb{H}$  (cf. [Bum98, 3.7 p. 349]).

### 2.1.2 Constructions and generalisations

Before constructing classes of examples of modular forms and presenting more general notions than in the proceedings above, there are a few elementary facts to be collected about the action of  $\operatorname{GL}_2^+(\mathbb{R})$  on  $\mathbb{H}$ . Recall that the group  $\operatorname{GL}_2(\mathbb{R})$  acts on  $\mathbb{C} \setminus \mathbb{R}$  via Möbius transformations. Namely,  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{GL}_2(\mathbb{R})$  acts on  $\tau \in \mathbb{C} \setminus \mathbb{R}$  via

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \tau = \frac{a\tau + b}{c\tau + d}.$$

This family of biholomorphic mappings of  $\mathbb{C} \setminus \mathbb{R}$  is exhaustive. We also recall the factor of automorphy  $j(\gamma, \tau) = c\tau + d$  and note that by direct computation

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \tau = \frac{a\tau + b}{c\tau + d}$$

$$= \frac{[a(u+iv)+b][c(u-iv)+d]}{(cu+d)^2 + (cv)^2}$$

$$= \frac{ac(u^2+v^2)+bd+ad(u+iv)+bc(u-iv)}{(cu+d)^2 + (cv)^2}$$

$$= \frac{ac(u^2+v^2)+bd+(ad+bc)u+iv(ad-bc)}{(cu+d)^2 + (cv)^2}$$

$$= \frac{ac(\tau)^2+bd+(ad+bc)u}{|j(\gamma,\tau)|^2} + i\det(\gamma)\frac{v}{|j(\gamma,\tau)|^2}$$

As a consequence, we find

$$\operatorname{Im}(\gamma \tau) = \det(\gamma) \frac{\operatorname{Im}(\tau)}{|j(\gamma, \tau)|^2}.$$

### **Eisenstein series**

The investigation above portrays an elegant and aesthetic abstract approach to modular forms, but the existence of such functions has not been verified yet. This is where Eisenstein series come into play, as the first examples and most elementarily accessible class of modular forms.

Since a modular form has to be invariant with respect to the Petersson operator, sym-

metrising seems to be an obvious approach. For a function  $f : \mathbb{H} \to \mathbb{C}$ , the sum

$$\sum_{\gamma \in \Gamma(1)} f|_k \gamma$$

is a natural candidate, which fulfils, at least formally, the invariance with respect to the operator  $|_k$ . The next step is to choose the function f, such that the sum converges and is holomorphic. This, however, is hopeless for a holomorphic function f and it appears to be reasonable to restrict the summation. For simplicity<sup>16</sup>, f shall be assumed to be 1-periodic, resulting in the group  $\Gamma(1)_{\infty} = \langle \pm 1, T \rangle$  not changing the summand, so that

$$\sum_{\gamma \in \Gamma(1)_{\infty} \setminus \Gamma(1)} f|_k \gamma$$

would be an appropriate choice for an invariant function. If f is, for instance, chosen to be constant, the sum converges locally uniformly for k > 2 and defines a holomorphic function. In this case, the resulting function

$$E_k := \sum_{\gamma \in \Gamma(1)_{\infty} \setminus \Gamma(1)} 1|_k \gamma \tag{2.6}$$

is called *Eisenstein series* of weight k and the following computation yields that it is even holomorphic at  $\infty$ , with value 1:

$$\begin{split} 2\zeta(k)E_k &= 2\zeta(k) \cdot \sum_{\substack{\gamma \in \Gamma_{\infty}(1) \setminus \Gamma(1)}} j(\gamma, \tau)^{-k} \\ &= \zeta(k) \cdot \sum_{\substack{c,d \in \mathbb{Z} \\ (c,d)=1}} \frac{1}{(c\tau + d)^k} \\ &= \sum_{\substack{c,d \in \mathbb{Z} \\ (c,d) \neq 0}} \frac{1}{(c\tau + d)^k} \\ &= \sum_{\substack{\lambda \in \Lambda_{\tau} \setminus \{0\}}} \frac{1}{\lambda^k}. \end{split}$$
(\*)

The first equality is valid, since

$$\Gamma(1)_{\infty} \setminus \Gamma(1) \to \left\{ (c,d) \in \mathbb{Z}^2 \mid (c,d) = 1 \right\}, \qquad (\overline{c},\overline{d}) \mapsto (c,d)$$

<sup>&</sup>lt;sup>16</sup>A more general construction is known by the term *Poincaré series* (cf. Subsection 3.3.2).

defines a bijection and the representation via  $(\star)$  is the one commonly used in introductory texts on the topic. This sum also immediately yields that  $E_k$  is holomorphic in  $\infty$ , by letting  $\text{Im}(\tau) \to \infty$ . An adept computation involving the partial fraction expansion of the cotangent function<sup>17</sup> yields the Fourier expansion of the Eisenstein series (cf. [DS05, p. 5]):

$$E_k = 1 - \frac{2k}{B_k} \sum_{n>0} \sigma_{k-1}(n) \cdot e^{2\pi i n\tau},$$
(2.7)

where  $B_k$  is the k-th Bernoulli–number and

$$\sigma_s(n) = \sum_{d|n} d^s \tag{2.8}$$

denotes the so called *divisor function* of degree  $s \in \mathbb{C}$ . The latter is a classical, multiplicative number theoretic function, meaning that the coefficients of  $E_k$  bear number theoretic information.

The Fourier expansion (2.7) together with the fact that the spaces of modular forms are finite dimensional may be utilised to derive unimagined identities between different sequences of Fourier coefficients. For instance,  $E_4^2 = E_8$  has to hold, since the space  $\mathcal{M}_8$ is 1 dimensional. As a result,

$$\sigma_7(n) = 120 \cdot \sum_{k=0}^n \sigma_3(n-k)\sigma_3(k)$$

is derived immediately.

Eisenstein series are not only the first examples of and most easily accessible modular forms, but to some extent *all* classical modular forms. To verify this claim note that the discriminant, technically defined above in Definition 2.3, can be written as

$$\Delta = \frac{E_4^3 - E_6^2}{1728}$$

and does not vanish on  $\mathbb{H}$ , by definition. It vanishes in  $\infty$  and, as a consequence, defines a linear bijection

$$f \mapsto \Delta \cdot f, \qquad \mathcal{M}_k \to \mathcal{S}_{k+12} < \mathcal{M}_{k+12},$$

<sup>&</sup>lt;sup>17</sup>In fact, this perspective offers another point for constructing modular forms from rational functions in the variable q – cf. [Fra20].

where  $S_k$  denotes the subspace of functions in  $\mathcal{M}_k$  which vanish in  $\infty$  and is called the space of *cusp forms*. Clearly,  $\mathcal{M}_k = \mathbb{C}E_k \oplus S_k$ , so that all modular forms may be written as homogenous polynomials in  $E_4$  and  $E_6$  due to the known dimension of  $\mathcal{M}_k$  up to the choice k = 12.

This closes the brief overview about a geometrical approach to classical modular forms as well as the particular case of Eisenstein series.

### Non-holomorphic Eisenstein series

Examining Equation (2.6) results in realising that the series is convergent, if and only if, k > 2. Nonetheless, a similar function exists for k = 2 and even lower values of k, but these are not holomorphic. In order to generate such functions, a technique, named after Hecke, is commonly used.<sup>18</sup> The general approach discussed in Section 5.3 resembles this technique which is reason enough to devote a subsection to its introduction. The reader may find additional information in [DS05, Sec. 4.10 p. 147] or [Iwa95, Sec. 3.2 p. 61].

**Definition 2.5.** For  $s \in \mathbb{C}, k \in \mathbb{Z}$  and  $\tau \in \mathbb{H}$ , let

$$E_k(\tau, s) := \sum_{\gamma \in \Gamma(1)_{\infty} \setminus \Gamma(1)} \operatorname{Im}(\tau)^s |_k \gamma$$

be the augmented<sup>19</sup> Eisenstein series of weight k.

In case of  $\operatorname{Re}(s) > 1 - \frac{k}{2}$ , this series converges absolutely in a uniform fashion on compact subsets, so that it is analytic on the half plane  $\{s \in \mathbb{C} \mid \operatorname{Re}(s) > 1 - \frac{k}{2}\}$ . Note that  $\lim_{s\to 0} E_k(\tau, s) = E_k(\tau)$ , in case of k > 2, so that this family of functions reproduces the classical Eisenstein series presented above. In case of  $k \leq 2$ , the series is not convergent for  $s \to 0$ , so that the naive definition of Eisenstein series given in Section 2.1.2 fails and a more advanced approach is required. This is exactly what the above series  $E_k(\tau, s)$ are capable of, due to their remarkable analytic properties being reflected in the next proposition.

**Proposition 2.6.** With  $E_k(\tau, s)$  associate the complete Eisenstein series

 $G_k^*(\tau, s) := \pi^{-s} \Gamma(k/2 + s) \zeta(2s) E_k(s - k/2).$ 

<sup>&</sup>lt;sup>18</sup>Compare, for instance, 'Darstellung von Klassenzahlen als Perioden von Integralen 3. Gattung aus dem Gebiet der elliptischen Modulfunktionen' found in [Hec83, p. 412].

<sup>&</sup>lt;sup>19</sup>The series is augmented by the parameter s.

This function has analytic continuation to the whole plane of complex numbers in s, except for the case when k = 0. In that case, the function has simple poles at s = 0, 1. In any case it is invariant under the transformation  $s \mapsto 1 - s$ .

This result suggests using  $E_k(\tau, 0)$  as a generalisation of classical Eisenstein series, which will then be non-holomorphic in case of k = 2. In fact, the Fourier expansion of that function looks like expected, but with the addition of a non-holomorphic correction term:

$$E_2(\tau) = -\frac{3}{\pi} \cdot \operatorname{Im}(\tau)^{-1} + 1 - 24 \sum_{n>0} \sigma_1(n) q^n.$$

More generally, the Fourier expansion of the automorphic function  $G_0^*(\tau, s) = \zeta^*(2s)E_0(\tau, s)$ is given by (cf. [Bum98, proof of Thm. 1.6.1 p. 66])

$$G_0^*(\tau, s) = \zeta^*(2s)y^s + \zeta^*(2(1-s))y^{1-s}$$

$$+ \sum_{\substack{r=-\infty,\\r\neq 0}}^{\infty} 2|r|^{s-1/2} \sigma_{1-2s}(|r|) \sqrt{y} K_{s-1/2}(2\pi |r|y) e^{2\pi i r x},$$
(2.9)

where

$$K_s(y) = \frac{1}{2} \int_0^\infty e^{-y(t+1/t)/2} t^s \frac{dt}{t}$$

is the *K* Bessel function and  $\sigma_s$  is the sum divisor function given in (2.8). The known applications of  $G_0^*$  are manifold (compare for instance the Rankin–Selberg methon [Bum98, 1.6 p. 65]). Note that Proposition 2.6 is derived from the Fourier expansion (2.9) for k = 0.

Define the Maas differential  $operator^{20}$ 

$$R_k := 2i\,\partial_z + \frac{k}{y}.$$

A simple calculation confirms that it satisfies the relations

$$(R_k f)|_{k+2}\gamma = R_k(f|_k\gamma)$$
 and  $R_k y^s = (s+k)y^{s-1}$ ,

<sup>&</sup>lt;sup>20</sup>In fact, the original operator looks slightly different, as it had been designed for Maas wave forms. Minor modifications were necessary to make it work as desired in the present setting.

for smooth functions f and  $\gamma \in SL_2(\mathbb{R})$ . As a consequence, we find

$$R_k G_k^*(s) = \left(s + k - \frac{k}{2}\right) \cdot \Gamma\left(s + \frac{k}{2}\right) \pi^{-s} \zeta(2s) \cdot E_{k+2}\left(s - \frac{k}{2} - 1\right)$$
$$= \Gamma\left(s + \frac{k+2}{2}\right) \pi^{-s} \zeta(2s) \cdot E_{k+2}\left(s - \frac{k+2}{2}\right)$$
$$= G_{k+2}^*(s)$$

and, in particular, Proposition 2.6 for k > 0. Further, the Fourier expansion of  $G_k^*$  follows from that of  $G_0^*$  by iterative application of differential operators, ultimately delivering that of  $E_k$ .

In fact, Proposition 2.6 might also be proven by means of theta functions (cf. [DS05, 4.10 p. 147 as.]), which are to be introduced in the next section before briefly relating them to Eisenstein series via a theorem of Siegel.

### Theta series

Theta series are special modular forms related to quadratic lattices. Hence, let (L, q) be a quadratic  $\mathbb{Z}$  lattice of rank  $m \in \mathbb{N}$  and associate the theta function

$$\theta_L : \mathbb{H} \to \mathbb{C}, \qquad \tau \mapsto \sum_{x \in L} e^{2\pi i q(x)\tau}$$

with it.<sup>21</sup> The sum above converges locally uniformly for positive definite q, defines a holomorphic function in  $\tau$  and is 1-periodic, if L is even. The transformation with respect to the inversion S is computed via Poisson summation which yields

$$\theta_{L'}(S\tau) = \left(\frac{\tau}{i}\right)^{m/2} \det(\mathbf{q}) \ \theta_L(\tau).$$

Here the root is defined by using the main branch of the logarithm. Strictly speaking, this is not the transformation property of a classical modular form for  $\Gamma(1)$ . However, it resembles the desired property and for the special choices  $\det(q) = 1$  (equivalently L = L'), as well as,  $m \in 8\mathbb{N}$  it is a classical modular form. If, on the other hand, L is at least even,  $\theta_L$  still represents a modular form for a subgroup of  $\Gamma(1)$ . In fact, [Ebe12, 3.1 Thm. 3.2 p. 87] states the following.

**Theorem 2.7.** For an even lattice (L,q) of rank  $m \in 2\mathbb{N}$  and level  $N \in \mathbb{N}$ , the function

<sup>&</sup>lt;sup>21</sup>In case the quadratic lattice is encoded in the Gram matrix S, we shall denote the theta series by  $\theta_S$ .

 $\theta_L$  is a modular form of weight  $\frac{m}{2}$  for  $\Gamma_0(N)$  with character<sup>22</sup>  $\left(\frac{\operatorname{disc}(L)}{\cdot}\right)$ , meaning that for all  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma_0(N)$ , we have

$$\theta_L(\gamma \tau) = \left(\frac{\operatorname{disc}(L)}{d}\right) j(\gamma, \tau)^{m/2} \theta_L(\tau)$$

and that the form is holomorphic at the cusps.

*Example* 2.8. An as famous as immediate application of theta series is that to the problem of representation numbers of quadratic forms. Let (L, q) be an even positive  $\mathbb{Z}$  lattice of rank m, then

$$\theta_L(\tau) = \sum_{k \ge 0} r_L(k) e^{2\pi i k \tau},$$

where  $r_L(k) = \# \{x \in L \mid q(x) = k\}$  is the representation number of n by (L, q). In fact, in case of  $8 \mid m$  and det(q) = 1,  $\theta_L$  is a classical modular form of weight m/2 as mentioned above, allowing for exact formulae for  $r_L$  in terms of coefficients of Eisenstein series in case  $m \leq 16$ . For higher ranks  $m \geq 24$  estimates with bounded error are obtained at least:

$$r_L(k) = \begin{cases} -\frac{m}{B_{d/2}} \sigma_{m/2-1}(k), & \text{if } m = 8, 16, \\ -\frac{m}{B_{d/2}} \sigma_{m/2-1}(k) + \mathcal{O}(k^{m/4}), & \text{else.} \end{cases}$$

The bound is derived from a general, elementary bound for coefficients of modular forms, named after Hecke (cf. Prop. 2.24), whereas the exact formulae for the low dimensional cases are due to the space of modular forms being one dimensional in this instance.

Another application of theta functions would be the proof of the functional equation of the Riemann Zeta function  $\zeta$ , as the completed version

$$\zeta^*(s) = \zeta(s)\Gamma(s/2)\pi^{-s/2}$$

is essentially the Mellin transform of  $(\theta_{\mathbb{Z},x^2/2} - 1)$ . As a consequence, the transformation law for that theta function may be abused, to represent  $\zeta^*$  as an integral that is symmetric in s and 1 - s, yielding<sup>23</sup> the following symmetry property

$$\zeta^*(s) = \zeta^*(1-s).$$

<sup>&</sup>lt;sup>22</sup>Note that the character is trivial, when the respective group is restricted to  $\Gamma_1(N)$ .

 $<sup>^{23}\</sup>mathrm{A}$  detailed computation is found in [Dei10, Thm. 6.1.2 p. 140]

### Theorem of Siegel

The (arithmetic) Theorem of Siegel established a formula for weighted representation numbers of special quadratic lattices (cf. [Sie51a], [Sie51b]). As a consequence, an analytic result, namely realising Eisenstein series as a weighted sum of theta series, is obtained.<sup>24</sup> It is this version which we shall refer to as Siegel's Theorem over the course of this thesis. The reader may consult [Fre13, V.1.3 p. 261] or [KK98] for information exceeding the scope of the following brief overview.

As seen in the preceeding section, theta series correspond to positive definite lattices. In fact, representation numbers of the latter may be used for properly combining theta series in order to construct the full range of classical modular forms. In order to substantiate this claim, let  $S \in M_n(\mathbb{Z})$  be symmetric positive definite and  $G \in \mathbb{Z}^{n \times m}$ , then  $G^T S G$  is again symmetric positive definite. In fact, the number of matrices G, such that  $G^T S G$ coincide with a fixed choice of matrix is finite. More specifically, for a symmetric matrix  $T \in M_m(\mathbb{Z})$ ,

$$#(S,T) \coloneqq # \left\{ G \in M_{n,m}(\mathbb{Z}) \mid G^T S G = T \right\}$$

is finite. It is called *representation number* of T by S. Further, on the set of symmetric positive definite matrices  $\text{Pos}_n(\mathbb{Z})$ , the operation

$$S.G = G^T S G$$

with  $G \in \operatorname{GL}_n(\mathbb{Z})$  defines a right action, which preserves evenness as well as unimodularity. This information suffices to state the analytic version of Siegel's Theorem.

**Theorem 2.9** (Siegel, Witt). Let  $S_1, \ldots, S_h$  be representatives of the classes of positive definite even unimodular quadratic  $\mathbb{Z}$ -lattices of rank m and  $k = \frac{m}{2}$ . Then

$$\sum_{j=1}^{h} \frac{\#(S_j, S_j)^{-1}}{\sum_{k=1}^{h} \#(S_k, S_k)^{-1}} \cdot \theta_{S_j} = E_k.$$

Altogether, the above theorem reveals that Eisenstein series may be obtained by a suitable combination of theta series, presenting them as an alternative foundation of the theory of classical modular forms.

The statement of Theorem 2.9 is briefly revisited in Section 5.4.1 where the *Siegel–Weil formula*, an intricate generalisation, is presented. In a phrase, it relates an integral

<sup>&</sup>lt;sup>24</sup>The stated version was developed by Witt and can be proven with elementary Hecke theory.

(resembling the weighted sum) of a theta distribution<sup>25</sup> to a special value (in the parameter s - cf. Section 2.1.2) of an Eisenstein series.

 $<sup>^{25}</sup>$ These are adelic distributions corresponding to classical theta functions in special cases (cf. Subsection 5.2.1).

# 2.2 Elliptic modular forms

This section is designed to provide a brief recollection of the classical theory and will serve as a point of reference in subsequent sections.

**Definition 2.10.** a) Define the *upper/lower half plane* as

$$\mathbb{H} \coloneqq \{\tau \in \mathbb{C} \mid \operatorname{Im}(\tau) > 0\}, \qquad \mathbb{H}^- \coloneqq \{\tau \in \mathbb{C} \mid \operatorname{Im}(\tau) < 0\}.$$

b) Let  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in GL_2(\mathbb{R})$ . Then the associated *Möbius transform* is given by

$$\mathbb{H} \cup \mathbb{H}^{-} \ni \tau \mapsto \gamma \tau \coloneqq \frac{a\tau + b}{c\tau + d} \in \mathbb{H} \cup \mathbb{H}^{-}.$$

c) For  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{GL}_2(\mathbb{R})$  and  $\tau \in \mathbb{C} \setminus \mathbb{R}$  define the factor of automorphy as

$$j(\gamma, \tau) \coloneqq (c\tau + d). \tag{2.10}$$

The following relations shed some light on these definitions.

**Remark 2.11.** a) For  $\gamma, \gamma' \in GL_2(\mathbb{R})$  and  $\tau \in \mathbb{C} \setminus \mathbb{R}$  we find the *cocycle relation* 

$$j(\gamma\gamma',\tau) = j(\gamma,\gamma'\tau) \cdot j(\gamma',\tau).$$
(2.11)

b) For  $\tau \in \mathbb{C} \setminus \mathbb{R}$  and  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathrm{GL}_2(\mathbb{R})$ , we write  $\tau = u + iv$  with  $u, v \in \mathbb{R}$  and find

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \tau = \frac{a\tau + b}{c\tau + d}$$
$$= \frac{ac|\tau|^2 + bd + (ad + bc)u}{|j(\gamma, \tau)|^2} + i\det(\gamma)\frac{v}{|j(\gamma, \tau)|^2}.$$

In particular, we have

$$\operatorname{Im}(\gamma\tau) = \frac{\operatorname{Im}(\tau)}{|j(\gamma,\tau)|^2}.$$
(2.12)

As a consequence, the subgroup  $\operatorname{GL}_2(\mathbb{R})^+ < \operatorname{GL}_2(\mathbb{R})$  of matrices with positive determinant acts on  $\mathbb{H}$  and  $\operatorname{GL}_2(\mathbb{R})/\operatorname{GL}_2^+(\mathbb{R}) \simeq \langle \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} \rangle$  has order 2. The centre  $\mathbb{R}\mathcal{I}_2 < \operatorname{GL}_2(\mathbb{R})^+$  acts trivially and its quotient acts transitively on  $\mathbb{H}$  (cf. (3.22)).

We consider the following arithmetic subgroup of  $\operatorname{GL}_2^+(\mathbb{R})$ .

**Definition 2.12.** The group  $\Gamma(1) := \operatorname{SL}_2(\mathbb{Z}) \leq \operatorname{GL}_2^+(\mathbb{R})$  is called the *modular group*. Its special elements

$$T \coloneqq \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, \qquad S \coloneqq \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$$

are referred to as *translation* and *inversion*.

**Definition 2.13.** Let  $k \in \mathbb{Z}$  and  $f : \mathbb{H} \to \mathbb{C}$  be a holomorphic function. Then f is called *modular form* of weight k if it satisfies the following two conditions:

- a) For all  $\gamma \in \Gamma(1)$  it transforms with  $f(\gamma \tau) = j(\gamma, \tau)^k f(\tau)$ .
- b) The function f is holomorphic at  $\infty$ .

The space of such functions is denoted  $\mathcal{M}_k(\Gamma(1))$ .

In order to give the last condition meaning, note that f is invariant under the translation matrix T, meaning it is periodic. As a consequence, it possesses a Fourier expansion

$$f(\tau) = \sum_{n \in \mathbb{Z}} a(n) \exp(2\pi i \tau n)$$
(2.13)

for all  $\tau \in \mathbb{H}$ . Here,  $a(n) \in \mathbb{C}$  are complex numbers<sup>26</sup> and the series converges (and hence absolutely – cf. Lemma A.19) on the upper half plane. In the context of Fourier series, write

$$q \coloneqq e(\tau) \coloneqq \exp(2\pi i\tau). \tag{2.14}$$

With this notation, the expansion of f becomes

$$f(\tau) = \sum_{n \in \mathbb{Z}} a(n)e(n\tau) = \sum_{n \in \mathbb{Z}} a(n)q^n.$$
(2.15)

**Definition 2.14.** With the notation as above, we say that f is holomorphic/meromorphic/ vanishes at  $\infty$ , if the power series in (2.15) has that property in the variable q in 0.

We define an operator leaving the space of modular forms invariant.

**Definition 2.15.** a) For a function  $f : \mathbb{H} \to \mathbb{C}, k \in \mathbb{Z}$ , and  $\gamma \in \mathrm{GL}_2^+(\mathbb{R})$  define the *Petersson (slash) operator* of weight k as

$$|_{k}: \mathbb{C}^{\mathbb{H}} \to \mathbb{C}^{\mathbb{H}}, \quad f|_{k}\gamma(\tau) \coloneqq \det(\gamma)^{k/2} j(\gamma, \tau)^{-k} f(\gamma\tau).$$
(2.16)

<sup>&</sup>lt;sup>26</sup>For a note on the Im( $\tau$ ) independence of a(n), compare A.2.

b) On  $\mathbb{H}$  write  $\tau = u + iv$  and define the hyperbolic measure

$$\mu \coloneqq \frac{\mathrm{d}u \,\mathrm{d}v}{v^2}.\tag{2.17}$$

c) For  $\delta > 0$ , the set

$$\mathcal{S}(\delta) \coloneqq \{\tau \in \mathbb{H} \mid \mathrm{Im}(\tau) > \delta, |\mathrm{Re}(\tau)| < 1/\delta\}$$
(2.18)

is called the *Siegel domain* associated to  $\delta$ .

The Petersson slash operator defines a right action of  $\operatorname{GL}_2^+(\mathbb{R})$  on  $\mathbb{C}^{\mathbb{H}}$  preserving continuity, real smoothness and analyticity. Letting a matrix  $\gamma \in \operatorname{GL}_2^+(\mathbb{Q})$  act preserves having a Fourier expansion and it preserves properties at  $\infty$  of elements of  $\mathcal{M}_k(\Gamma(1))$ . Further, the measure  $\mu$  is invariant under pullback via Möbius transformations from  $\operatorname{GL}_2^+(\mathbb{R})$ . Finally, Siegel domains have finite measure and may be used to prove that  $\Gamma(1)$ acts properly discontinuously on  $\mathbb{H}$ , implying that the quotient  $Y(\Gamma(1)) \coloneqq \Gamma(1) \setminus \mathbb{H}$  carries a  $T_2$  topology (cf. [Miy06, 1.5 p. 17]). A visualisation of a set of representatives in  $\mathbb{H}$  of  $Y(\Gamma(1))$  is presented in Figure 2.1.2.

The following series represents a canonical choice of a modular form.

**Definition 2.16.** Let  $\Gamma_{\infty} \coloneqq \langle T \rangle$  and consider for  $k \geq 3$  the *Eisenstein series* 

$$E_k(\tau) \coloneqq \frac{1}{2} \cdot \sum_{\gamma \in \Gamma_\infty \setminus \Gamma(1)} 1|_k \gamma(\tau).$$
(2.19)

The series  $E_k$  converges normally on  $\mathbb{H}$  and defines a holomorphic function. Further, it vanishes for  $2 \nmid k$  and in fact, by the invariance condition, any  $f \in \mathcal{M}_k(\Gamma(1))$  has to vanish due to the transformation under  $-\mathcal{I}_2$  for uneven k. A comparison with the partial fraction expansion of the cotangent (cf. [DS05, p. 5]) delivers the following Fourier expansion:

$$E_k(\tau) = 1 - \frac{2k}{B_k} \sum_{n>0} \sigma_{k-1}(n) q^n, \qquad (2.20)$$

where  $B_k$  denotes the k-th Bernoulli number and

$$\sigma_s(n) = \sum_{d|n} d^s \tag{2.21}$$

denotes the *divisor function* of degree  $s \in \mathbb{C}$  (with  $\operatorname{Re}(s) > 0$ ).

Another example of automorphic forms are theta functions which are revisited in the more general setting of vector valued modular forms in Definition 3.29.

**Definition 2.17.** For (L, q) a positive definite even  $\mathbb{Z}$  lattice of rank m, define

$$\theta_L : \mathbb{H} \to \mathbb{C}, \qquad \tau \mapsto \sum_{x \in L} e(\mathbf{q}(x)\tau) = \sum_{n \in \mathbb{N}} r_L(n)q^n.$$
(2.22)

For unimodular lattices, these are modular forms for  $\Gamma(1)$  of weight m/2. However, instead of merely considering the arithmetic subgroup  $\Gamma(1)$  as a group of transformations, a wider family of groups due to Hecke may be considered.

**Definition 2.18.** Let  $N \in \mathbb{N}$ . Then the kernel of the natural (surjective) projection

$$\operatorname{SL}_2(\mathbb{Z}) \to \operatorname{SL}_2(\mathbb{Z}/N\mathbb{Z})$$

is denoted by  $\Gamma(N)$  and called the *principal congruence subgroup* of *level* N. Further, a subgroup  $\Gamma \leq \Gamma(1)$  is called *congruence subgroup*, if there is some  $\Gamma(N) \leq \Gamma$ .

Remark 2.19. The following two classes of examples represent congruence subgroups:

$$\Gamma_0(N) \coloneqq \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{SL}_2(\mathbb{Z}) \middle| c \equiv 0 \mod N \right\},$$
(2.23)

$$\Gamma_1(N) \coloneqq \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma_0(N) \middle| a \equiv d \equiv 1 \mod N \right\}.$$
(2.24)

We find for N > 1 that

 $\Gamma(N) \triangleleft \Gamma_1(N) \triangleleft \Gamma_0(N) < \Gamma(1),$ 

where the natural kernel morphisms are given by

$$\Gamma_0(N) \ni \begin{pmatrix} a & b \\ c & d \end{pmatrix} \mapsto d \in (\mathbb{Z}/N\mathbb{Z})^{\times}, \qquad \Gamma_1(N) \ni \begin{pmatrix} a & b \\ c & d \end{pmatrix} \mapsto b \in (\mathbb{Z}/N\mathbb{Z}).$$

In particular, any congruence subgroup has finite index in  $\Gamma(1)$ .

The action of  $\operatorname{GL}_2^+(\mathbb{R})$  on  $\mathbb{H}$  may be extended to the closure  $\overline{\mathbb{H}} = \mathbb{H} \cup \mathbb{P}^1(\mathbb{R})$  in a natural way. The modular group  $\Gamma(1)$  then acts transitively on  $\mathbb{P}^1(\mathbb{Q}) = \mathbb{Q} \cup \{\infty\}$ .

**Definition 2.20.** For a congruence subgroup  $\Gamma$ , the equivalence classes  $\Gamma \setminus \mathbb{P}^1(\mathbb{Q})$  are referred to as *cusps*. For a meromorphic function  $f : \mathbb{H} \to \mathbb{C}$  with Fourier expansion at

Notation	Behaviour on $\mathbb H$	Behaviour at cusps	Name
$\mathcal{S}_k(\Gamma,\chi)$	holomorphic	vanishings	cusp forms
$\mathcal{M}_k(\Gamma,\chi)$	holomorphic	holomorphic	(holomorphic) modular forms
$\mathcal{M}_k^!(\Gamma,\chi)$	holomorphic	meromorphic	weakly holomorphic modular forms
$\mathcal{A}_k(\Gamma,\chi)$	meromorphic	meromorphic	meromorphic modular forms
$\mathcal{R}_k(\Gamma,\chi)$	real analytic	no condition	real analytic modular forms
$\mathcal{C}_k^m(\Gamma,\chi)$	$\mathcal{C}^m$	no condition	$\mathcal{C}^m$ modular functions

Table 2.1: Table of different types of modular forms. All of the forms f are assumed to transform with respect to  $\Gamma$  like in Definition 2.21, meaning for all  $\gamma \in \Gamma$  they satisfy  $f(\gamma \tau) = \chi(\gamma) j(\gamma, \tau)^k f(\tau)$ .

 $\infty$  and a representative c of a cusp, we select  $\gamma \in \Gamma(1)$  with  $\gamma \infty = c$ . Then the Fourier expansion of  $f|_k \gamma$  at  $\infty$  is referred to as the *Fourier expansion* of f at c (for  $|_k$ ). It is not unique, but its properties like vanishings or being holomorphic are well defined.

The modular curve associated to some congruence subgroup  $\Gamma$  is denoted by  $Y(\Gamma) \coloneqq \Gamma \setminus \mathbb{H}$ . It is a Riemannian manifold. By the theory of *Baily–Borel*, it may be compactified by adding its cusps to it, yielding the compactified modular curve  $X(\Gamma) \coloneqq \Gamma \setminus \mathbb{H}^*$ , where  $\mathbb{H}^* = \mathbb{H} \cup \mathbb{P}^1(\mathbb{Q})$ . As a consequence,  $Y(\Gamma)$  is a quasi projective variety.

**Definition 2.21.** Let  $\Gamma \leq \Gamma(1)$  be a congruence subgroup,  $k \in \mathbb{Z}$ , and  $\chi : \Gamma \to \mathbb{T}$  be a character. Then a holomorphic function  $f : \mathbb{H} \to \mathbb{C}$  is called a *modular form* for  $\Gamma$  of weight k with character  $\chi$ , if it satisfies the following two properties.

- a) For all  $\gamma \in \Gamma$  it transforms with  $f(\gamma \tau) = \chi(\gamma) j(\gamma, \tau)^k f(\tau)$ .
- b) The function f is holomorphic at  $\infty$ .

The space of such functions is denoted by  $\mathcal{M}_k(\Gamma, \chi)$ .

The conditions on such functions f may be relaxed or tightened. Table 2.1 presents an overview of typical variants of Definition 2.21. The spaces  $\mathcal{M}_k(\Gamma, \chi)$  are finite dimensional and if  $\Gamma(N) \leq \Gamma$ , then  $T^N \in \Gamma$ , so that f is invariant under a possibly higher power of  $T^N$  and hence possesses a Fourier expansion. Examples of such more general modular forms may be constructed again as Eisenstein series or, more generally, as Poincaré series (cf. [Miy06, 2.6 p. 61]). Further, theta functions represent modular forms with character.

**Theorem 2.22** ([Ebel2, 3.1 Thm. 3.2 p. 87]). For an even lattice (L, q) of rank  $m \in 2\mathbb{N}$ and level  $N \in \mathbb{N}$ , the function  $\theta_L$  is a modular form of weight  $\frac{m}{2}$  for  $\Gamma_0(N)$  with character<sup>27</sup>

<sup>&</sup>lt;sup>27</sup>Note that the character vanishes, when the respective group is restricted to  $\Gamma_1(N)$ .

 $\left(\frac{\operatorname{disc}(L)}{\cdot}\right)$ , meaning that for all  $\gamma = \left(\begin{smallmatrix} a & b \\ c & d \end{smallmatrix}\right) \in \Gamma_0(N)$ , we have

$$\theta_L(\gamma \tau) = \left(\frac{\operatorname{disc}(L)}{a}\right) j(\gamma, \tau)^{m/2} \theta_L(\tau)$$

and that the form is holomorphic at the cusps.

Apart from these examples, modular forms with characters appear naturally, when considering these without character as a byproduct of representation theory of finite groups (cf. Remark 6.8).

**Remark 2.23.** There is a natural pairing serving as a scalar product on the space of cusp forms. For  $f \in \mathcal{M}_k^!(\Gamma)$  and  $g \in \mathcal{M}_k^!(\Gamma')$  congruence modular forms, so that  $\Gamma(N) \leq \Gamma \cap \Gamma'$ for some  $N \in \mathbb{N}$  and  $f \cdot g \in \mathcal{S}_{2k}(\Gamma(N))$ , set

$$\langle f,g \rangle \coloneqq \frac{1}{[\Gamma(1):\Gamma(N)]} \cdot \int_{\Gamma(N) \setminus \mathbb{H}} f \cdot \overline{g} \cdot \mathrm{Im}^k \, \mathrm{d}\mu.$$
 (2.25)

Then, the expression is absolutely convergent (cf. Example 3.52) and independent of the choice of N. It is referred to as the *Petersson product* of f and g.

In Section 6.4, *L*-functions, another significant number theoretic construct, will be associated to modular forms. For their convergence, asymptotic bounds on Fourier coefficients of modular forms are required. The following is a naive bound that already suffices for a vast array of cases.

**Proposition 2.24** ([KK98, p. 200]). For  $f \in S_k(\Gamma_0(N), \chi)$  and  $\gamma \in \Gamma(1)$ . Denote the Fourier expansion of  $f|_k \gamma$  by  $\sum_{n_0 \leq n \in \mathbb{Z}/N} a(n)e(n\tau)$ , then we have that

$$a(n) = \mathcal{O}(n^{k/2}), \qquad n \to \infty.$$

While the advantage of the proposition above is that its proof is relatively accessible, there are more effective bounds which become relevant in boundary cases.

**Theorem 2.25** ([Ran39b, Thm. 2, p. 358]). Let  $f \in S_k(\Gamma(N))$  with Fourier expansion  $\sum_{n_0 \leq n \in \mathbb{Z}/N} a(n)e(n\tau)$ . Then

$$a_n = \mathcal{O}(n^{k/2-1/5}), \qquad n \to \infty.$$

The strongest possible bound is the following (cf. [Miy06, Thm. 4.5.17 p. 150]).

**Theorem 2.26** (Ramanujan-Petersson-Deligne). For  $f \in S_k(\Gamma_0(N), \chi)$  with Fourier expansion  $\sum_{n_0 \leq n \in \mathbb{Z}/N} a(n)e(n\tau)$ , we have for gcd(n, N) = 1 and any  $\varepsilon > 0$  that

$$a(n) = \mathcal{O}(n^{k/2-1/2+\varepsilon}), \qquad n \to \infty.$$

Note that for primes dividing the level, Atkin–Lehner theory may be utilised to deduce similar bounds (cf. [Kna92, Thm. 9.27 p. 289]) and also 3.81. Additional bounds are presented in Section 3.3.4.

Part II

**Automorphic forms** 

# **3** Symplectic theory

In Section 2.2, an introduction to the theory of classical modular forms associated with the group  $SL_2$  has been presented, followed by a couple of generalisations. A further avenue for exploration would be groups of higher rank, such as  $SL_3$ . However, it turns out that the proper generalisation in this context is the symplectic setting. This will be briefly discussed in the following section, after which the focus will return to the essential subcase for this thesis, namely the elliptic setting, which will include the Weil representation and vector valued modular forms. The chapter concludes with a collection of asymptotic bounds for Fourier coefficients of vector valued modular forms which will be used to derive convergence results of L-series.

## 3.1 Symplectic modular forms

We provide a brief overview of the symplectic notions of modular forms – called *Siegel* modular forms. For a slightly more elaborate but concise description of the following, we refer to [Fre91, Chap. 1], while the German source [Fre83] provides a more comprehensive picture.

As in case of the orthogonal group, symplectic groups may be defined as the invariance group of a bilinear form. For that purpose, let  $g \in \mathbb{N}$  and  $\mathcal{I}_g$  denote the identity matrix of rank g. Then

$$J = \begin{pmatrix} 0 & \mathcal{I}_g \\ -\mathcal{I}_g & 0 \end{pmatrix} \in M_{2g}$$

denotes the standard alternating matrix.

**Definition 3.1.** Let R be a commutative ring with unit. Then the symplectic group of genus  $g \in \mathbb{N}$  over R is defined to be

$$\operatorname{Sp}_{g}(R) \coloneqq \left\{ M \in M_{2g}(R) \mid M^{T}JM = J \right\}.$$

$$(3.1)$$

Example 3.2. In case g = 1, the conditions reduce to the case of the special linear group

$$\operatorname{Sp}_1(R) = \operatorname{SL}_2(R)$$

As in case of SL<sub>2</sub>, there is also a variety of symplectic matrices that may be presented more explicitly in the case of a general ring R. Let  $S \in M_g(R)$  be symmetric, i.e.  $S^T = S$ and  $U \in GL_q(R)$ . Then the following are symplectic matrices.

$$N(S) \coloneqq \begin{pmatrix} \mathcal{I}_g & S \\ 0 & \mathcal{I}_g \end{pmatrix},\tag{3.2}$$

$$M(U) \coloneqq \begin{pmatrix} U^T & 0\\ 0 & U^{-1} \end{pmatrix}, \tag{3.3}$$

$$J = \begin{pmatrix} 0 & \mathcal{I}_n \\ -\mathcal{I}_n & 0 \end{pmatrix}.$$
 (3.4)

In case  $R = \mathbb{Z}$  or R = F a field, the group  $\operatorname{Sp}_g(R)$  is generated by matrices as in (3.2) and (3.4). Further, the unitary group U(g) of rank g embeds into  $\operatorname{Sp}_g(\mathbb{R})$  via

$$U(g) \hookrightarrow \operatorname{Sp}_{g}(\mathbb{R}), \qquad U \mapsto \begin{pmatrix} \operatorname{Re}(U) & \operatorname{Im}(U) \\ -\operatorname{Im}(U) & \operatorname{Re}(U) \end{pmatrix}.$$
 (3.5)

This is a maximal compact subgroup and it agrees with  $SO_2(\mathbb{R})$  in the instance of  $Sp_1(\mathbb{R}) = SL_2(\mathbb{R})$ .

As in case of the special linear group of rank 2, there is a theory of congruence subgroups and of multiplier systems which are described in [Fre91]. In order to define modular forms in the symplectic setting, an analogue of the upper half plane has to be constructed.

**Definition 3.3.** Let  $g \in \mathbb{N}$ . Then the *Siegel upper half space* of *degree* g is defined to be

$$\mathbb{H}_q \coloneqq \{Z = X + iY \in M_{2q}(\mathbb{C}) \mid Z^T = Z, Y > 0\}.$$

Here, X, Y denote the real, respectively imaginary part, of Z and the condition Y > 0means that Y is a positive definite matrix, i.e. as a Gram Matrix of a real quadratic vector space.

The Siegel upper half space is convex and as such connected and carries a natural complex structure. Analogous to the  $SL_2$  case, there is an action of  $Sp_q(\mathbb{R})$  on the upper

half space  $\mathbb{H}_g$ . Namely, let

$$Z \in \mathbb{H}_g, \quad M = \begin{pmatrix} A & B \\ C & D \end{pmatrix} \in \mathrm{Sp}_g(\mathbb{R}).$$

Then CZ + D is invertible and the matrix

$$MZ \coloneqq (AZ + B)(CZ + D)^{-1} \tag{3.6}$$

lies in  $\mathbb{H}_g$ , again. This association defines an action of  $\mathrm{Sp}_g(\mathbb{R})$  on  $\mathbb{H}_g$  via biholomorphic maps and all biholomorphic maps on the upper half space are of this form. For instance, the action of the matrices in (3.2), (3.3), and (3.4) looks like the following.

$$N(S)Z = Z + S, \qquad M(U)Z = U^T Z U, \qquad JZ = -Z^{-1}$$

Note that to the map

$$\mathbb{H}_q \ni Z \mapsto \det(CZ + D),$$

there is a holomorphic square root. We fix one of these roots and denote it by

$$Z \mapsto \det(CZ + D)^{1/2}.$$

**Definition 3.4.** Let  $r \in \mathbb{Z}$ , V be a finite dimensional  $\mathbb{C}$  vector space, and

$$f: \mathbb{H}_q \to V$$

be holomorphic. Then f is called holomorphic Siegel modular form of weight r if it satisfied the following transformation property for all  $M = \begin{pmatrix} A & B \\ C & D \end{pmatrix} \in \operatorname{Sp}_g(\mathbb{Z})$ 

$$f(MZ) = \det(CZ + D)^r f(Z)$$
(3.7)

and for any  $Z_0 \in \mathbb{H}_q$ , the expression f(Z) is uniformly bounded for  $\operatorname{Im}(Z) > \operatorname{Im}(Z_0)$ .

We note that such a function is invariant under the operation of matrices N(S) with  $S \in M_g(\mathbb{Z})$  symmetric which form a  $\mathbb{Z}$  lattice with bilinearform induced by the trace. Hence, it possesses a Fourier expansion of the form

$$f(Z) = \sum_{S \text{ rational, } S = S^T} a(S) \ e\left(\operatorname{tr}(SZ)\right).$$
(3.8)

The condition on a modular form f being universally bounded on domains with  $\text{Im}(Z) \ge$  $\text{Im}(Z_0)$  for any fixed  $Z_0 \in \mathbb{H}_g$  translates to the Fourier coefficients a(S) vanishing unless  $S \ge 0$  is semi positive definite.

In case of genus greater than 1, there are harsher restrictions on the Fourier coefficients given by the so called Koecher principle.

**Theorem 3.5** (Koecher principle). Let  $f : \mathbb{H}_g \to \mathbb{C}$  be a holomorphic function with the properties that

- a) f(Z+S) = f(Z) for all  $S \in M_q(\mathbb{Z})$  with  $S^T = S$ ,
- b)  $f(U^T Z U)$  for all  $U \in \mathrm{SL}_q(\mathbb{Z})$ .

Then the condition g > 1 implies that f has a Fourier expansion of the form

$$f(Z) = \sum_{\substack{S=S^T,\\S\geq 0 \text{ even}}} a(S) \ e\left(\operatorname{tr}(SZ)\right).$$

In particular, for any  $Z_0 \in \mathbb{H}_g$  the expression f(Z) is uniformly bounded for  $\operatorname{Im}(Z) \geq \operatorname{Im}(Z_0)$ .

There is also a rich theory of Siegel modular forms reproducing a variety of results from the classical setting of genus g = 1 and we refer to [Fre83] for details before turning towards the essential case for this thesis being  $\text{Sp}_1 = \text{SL}_2$  and its Weil representation.

## 3.2 Weil representation

Vector valued modular forms for the Weil representation of  $SL_2$  play a significant role within the scope of this thesis (cf. Section 3.3). However, when forms of half integral weight are taken into account, it becomes necessary to consider a twofold cover, the *metaplectic* group Mp<sub>2</sub>, of SL<sub>2</sub>. Such a cover exists and an abstract construction is sketched in the addendum (cf. Section B.3). However, these details are not required for the purpose of the thesis and we will be presenting a briefer more ad hoc approach in the following.

As mentioned in the chapter above, the symplectic group over a field is generated by the elements N(S) and J (cf. (3.2) and (3.4)). In case of genus g = 1 and a field F we follow [Lan98, XI.2 pp. 209-214] and write  $a \in F^{\times}$  and  $b \in F$ . Then we find the following relations

$$J^2 = M(-1), (3.9)$$

$$M(a)N(b)M(a^{-1}) = N(ba^2), (3.10)$$

$$M(a) = JN(a^{-1})JN(a)JN(a^{-1}).$$
(3.11)

We note that

$$F \ni b \mapsto N(b) \in \mathrm{SL}_2(F), \qquad F^{\times} \ni a \mapsto M(a) \in \mathrm{SL}_2(F)$$

are embeddings. These matrices may be used to represent any element in  $SL_2(F)$ .

**Lemma 3.6.** Every element of  $SL_2(F)$  has a unique representation of the form

$$N(b)M(a)$$
 or  $N(b_1)M(a)JN(b_2)$ 

where  $a \in F^{\times}$  and  $b, b_1, b_2 \in F$ .

In fact, we have the following stronger assertion.

**Proposition 3.7.** Let F be a field. The group  $SL_2(F)$  is, up to isomorphy, the free group generated by an injective additive homomorphism N from F and an element J such that M defined by (3.11) is an injective multiplicative homomorphism from  $F^{\times}$  modulo the relations (3.9) and (3.10).

This reveals that in order to construct a representation of  $SL_2(F)$ , solely the action of N(b) and J have to be established and the relations stated in Proposition 3.7 have to be verified. Also from a practical point of view, relation (3.11) alongside Lemma 3.6 suggest computing the action of M(a) in order to manipulate emerging expressions conveniently. In the following, we will describe the respective action of matrices.

The setting of this paragraph is mainly a brief overview of necessary notions from [KRY06, 8.5 pp. 320-342] – the source also covers the case of higher genus. We will restrict to the case of a prime p > 2 and refer to the source for the case of p = 2 which is similar in nature. Let  $\psi_p$  denote the standard additive character of  $\mathbb{Q}_p$  meaning the one from Example 1.48 with the choice a = 1 and  $V_p$  a vector space of dimension m.<sup>1</sup> Recall that to a Schwartz–Bruhat form  $\phi \in \mathcal{S}(V_p)$ , the Fourier transform described in (1.8) will be denoted  $\hat{\phi}$ . Then, for  $a \in \mathbb{Q}_p^{\times}$  and  $b \in \mathbb{Q}_p$  a projective representation of  $SL_2(\mathbb{Q}_p)$  on  $\mathcal{S}(V_p)$ ,

<sup>&</sup>lt;sup>1</sup>Selecting different characters only requires minor modifications.

is given by the following operator

$${}^{\mathrm{pr}}r_p\left[\begin{pmatrix}1&b\\0&1\end{pmatrix}\right]:\phi\mapsto\psi(b\,\mathbf{q}(x))\cdot\phi(x),\tag{3.12}$$

$${}^{\mathrm{pr}}r_p\left[\begin{pmatrix}0&1\\-1&0\end{pmatrix}\right]:\phi\mapsto\hat{\phi}(x),\tag{3.13}$$

$${}^{\mathrm{pr}}r_p\left[\begin{pmatrix}a&0\\0&a^{-1}\end{pmatrix}\right]:\phi\mapsto|a|_p^{m/2}\phi(ax).$$
(3.14)

In fact, it satisfies the relation  ${}^{\mathrm{pr}}r_p(g_1){}^{\mathrm{pr}}r_p(g_2) = c_L(g_1,g_2){}^{\mathrm{pr}}r_p(g_1g_2)$  for the Leray cocycle  $c_L$ . That cocycle is given by

$$c_L(g_1, g_2) = \gamma(\psi \circ q(g_1, g_2)),$$

where  $\gamma$  is the Weil index of a character of second degree and the associated quadratic form is given by the Leray invariant of the isotropic subspaces  $(Y, g_1, Y, Yg_2^{-1})$  (cf. [LV80, 1.6 p. 47 as.] for an extensive discussion or [Rao93, Introduction and Thm. 4] for a brief reference). The cocycle amounts to a parametrisation of the *metaplectic extension*  $\overline{\mathrm{SL}}_2(\mathbb{Q}_p)$ of  $\mathrm{SL}_2(\mathbb{Q}_p)$  satisfying

$$1 \to \mathbb{T} \to \overline{\mathrm{SL}}_2(\mathbb{Q}_p) \to \mathrm{SL}_2(\mathbb{Q}_p) \to 1$$
 (3.15)

via  $\operatorname{SL}_2(\mathbb{Q}_p) \times \mathbb{T}$ . Namely,

$$(g, z) \mapsto [g, z]_L$$
, where  $[g_1, z_1]_L \cdot [g_2, z_2]_L = [g_1g_2, c_L(g_1, g_2)z_1z_2].$ 

For later global use this cocycle has to be modified, as it is not trivial on compact open subgroups.<sup>2</sup> The Weil representation in Leray coordinates is given by

$$\omega_{V_p}([g,z]_L)\phi(x) = \chi_V(x(g)) \left( z\gamma\left(\psi_{p,\frac{1}{2}}\right) \right)^{\#} \gamma\left(\psi_{p,\frac{1}{2}} \circ V\right)^{-1} \operatorname{pr} r_p(g) \phi(x)$$
(3.16)

where

$$\# = \begin{cases} 1 & \text{if } m \text{ is odd,} \\ 0 & \text{if } m \text{ is even,} \end{cases} \qquad \qquad \psi_{p,a}(x) \coloneqq e_p(a,x),$$

<sup>&</sup>lt;sup>2</sup>This issue is fixed explicitly in [KRY06, (8.5.11)], resulting in a complex cocycle that is trivial on compact open subgroups as well as the standard parabolic subgroup.

and

$$\chi_V(x(g)) = \mathcal{H}_p\left(x(g), \operatorname{disc}(q)\right), \quad \text{where} \quad x\left(\begin{pmatrix} a & b \\ c & d \end{pmatrix}\right) = \begin{cases} c, & \text{if } c \neq 0, \\ d, & \text{else.} \end{cases}$$
(3.17)

Recall that  $\mathcal{H}_p$  denotes the Hilbert symbol on  $\mathbb{Q}_p$  described in Section 1.4. Note also that  $\chi_V$ , as well as,  $\gamma$  are well defined on isometry classes of quadratic spaces (V, q) and the latter even defines a character of the Witt Ring (cf. Definition 1.19) mapping to the 8-th roots of unity.

Example 3.8. In case of  $\dim(V_p) = m$  is even, the representation factors through  $SL_2$ .<sup>3</sup> In that case, the concrete operations are given by

$$\begin{split} & \omega_{V_p} \left[ \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} \right] : \phi \mapsto \psi(b \operatorname{q}(x)) \cdot \phi(x), \\ & \omega_{V_p} \left[ \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \right] : \phi \mapsto \gamma(V_p) \cdot \hat{\phi}(x), \\ & \omega_{V_p} \left[ \begin{pmatrix} a & 0 \\ 0 & a^{-1} \end{pmatrix} \right] : \phi \mapsto \chi_p(a) |a|_p^{m/2} \phi(ax), \end{split}$$

where  $\gamma(V_p)$  is the *local splitting index* defined in [Kud94, Thm. 3.1].

This closes the brief overview of the local case and we turn towards sketching an adelisation of the construction.

## Adelisation

The concepts above, namely the constructed group and their representations may be transferred to the adelic setting. For instance, define

$$\operatorname{SL}_2(\mathbb{A}) \coloneqq \prod_{p \leq \infty}' \operatorname{SL}_2(\mathbb{Q}_p)$$

 $<sup>^{3}</sup>$ In fact, that statement is an if, and only if, meaning the associated representation is ordinary on SL<sub>2</sub> in that case exactly.

to be the restricted product with respect to  $SL_2(\mathbb{Z}_p)$ .<sup>4</sup> Similarly to the local case, there is an extension

$$1 \longrightarrow \mathbb{T} \longrightarrow \overline{\operatorname{SL}}_2(\mathbb{A}) \longrightarrow \operatorname{SL}_2(\mathbb{A}) \longrightarrow 1.$$
(3.18)

The Weil representation is given as a direct product. In order to prove that this definition is meaningful, the reader verifies that  $SL_2(\mathbb{Z}_p)$  operates trivially on almost all components of a Schwartz Bruhat function from  $\mathcal{S}(V_{\mathbb{A}})$  (cf. [Yos84, Lemma 2.1 p. 190]). So that

$$\omega_{\mathbb{A}}([g_{\mathbb{A}}, z]_L) = \bigotimes_{p \le \infty} \omega_{V_p}([g_p, z_p]_L)$$
(3.19)

might be realised as a *restricted tensor product* (cf. [Bum98, pp 300-303]).

Note that this definition is given in Leray coordinates, as were the above representations, even though the group itself is more conveniently parametrised by the cocycle from [KRY06, (8.5.11)], so that it is trivial on a suitable compact open subgroup. There is a product formula for the Weil indices

$$1 = \prod_{p \le \infty} \gamma(\psi \circ q_p) = 1,$$

which is also valid for the modified local Weil indices  $\gamma(V)$ , used in [KY10] to describe Fourier coefficients of adelic Eisenstein series in terms of Whittaker functions (also cf. [Rao93]).

We continue by stating some properties of  $SL_2$  and its metaplectic extension based on [KY10, 1 p. 2278]. We restate the following notation

$$M(F_p) \coloneqq \left\{ \begin{pmatrix} a & 0\\ 0 & a^{-1} \end{pmatrix} =: M(a) \mid a \in F_p^{\times} \right\},$$
(3.20)

$$N(F_p) := \left\{ \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix} =: N(b) \mid b \in F_p \right\}.$$
(3.21)

Recall that  $SL_2(\mathbb{R})$  admits the so called *Iwasawa* decomposition. Namely, that  $SO_2(\mathbb{R}) < \mathbb{R}$ 

<sup>&</sup>lt;sup>4</sup>Since the product does not depend on changes to finitely many components, we select an arbitrary open subgroup at the Archemedian place.

 $SL_2(\mathbb{R})$  is a maximal compact subgroup, such that  $SL_2(\mathbb{R})/SO_2(\mathbb{R}) \simeq \mathbb{H}$ . Here,

$$k_{\vartheta} = \begin{pmatrix} \cos(\vartheta) & \sin(\vartheta) \\ -\sin(\vartheta) & \cos(\vartheta) \end{pmatrix} \in K_{\infty} \coloneqq \mathrm{SO}_{2}(\mathbb{R})$$

constitutes a parametrisation of  $K_{\infty}$  via  $\vartheta \in \mathbb{T}^{5}$  Consequently, we find for  $\tau = u + iv \in \mathbb{H}$ and  $g_{\tau} := N(u)M(\sqrt{v}) \in P_{\infty}$  that

$$g_{\tau}k_{\vartheta} = [N(u)M(\sqrt{v})]k_{\vartheta}$$

$$= \left[ \begin{pmatrix} 1 & u \\ 0 & 1 \end{pmatrix} \cdot \begin{pmatrix} \sqrt{v} & 0 \\ 0 & 1/\sqrt{v} \end{pmatrix} \right] \cdot \begin{pmatrix} \cos(\vartheta) & \sin(\vartheta) \\ -\sin(\vartheta) & \cos(\vartheta) \end{pmatrix}$$

$$\mapsto g_{\tau}k_{\vartheta}i = \tau = u + iv \in \mathbb{H}$$
(3.22)

defines a bijection  $\operatorname{SL}_2(\mathbb{R})/\operatorname{SO}_2(\mathbb{R}) \simeq \mathbb{H}$ . As a consequence, the Iwasawa decomposition

$$\mathrm{SL}_2(\mathbb{R}) = P_\infty K_\infty \tag{3.23}$$

follows, where  $P_{\infty} := P(\mathbb{R}) = N(\mathbb{R})M(\mathbb{R}) < SL_2(\mathbb{R})$  denotes the standard Borel group of upper triangular matrices and the representation is unique.<sup>6</sup> This decomposition will be presented in more detail in the prerequisites of Chapter 5.

Similar to the above, define  $K_p := \operatorname{SL}_2(\mathbb{Z}_p)$  as well as  $K_f := \operatorname{SL}_2(\hat{\mathbb{Z}}) = \prod_{p < \infty} K_p$ . Then, for the maximal compact subgroup  $K_{\mathbb{A}} := K_{\infty}K_f < \operatorname{SL}_2(\mathbb{A})$ , an analogous decomposition  $\operatorname{SL}_2(\mathbb{A}) = P_{\mathbb{A}}K_{\mathbb{A}}$  is acquired. Such a decomposition is used to compute the action of  $\operatorname{SL}_2$ , globally or locally, on the principal series representation, discussed in Section 5.3.1.

Recall that  $SL_2(\mathbb{A})$  is described in terms of  $\overline{SL}_2(\mathbb{A})$  and  $\mathbb{T}$  by means of a cocycle in accordance with (3.18) that is given by local Leray cocycles. In the following, we assume these to be modified as in [KRY06, (8.5.11)] in order to yield a cocycle that is trivial on  $K_p$  which is advantageous for global use. To the group  $P(\mathbb{A})$  there is a splitting homomorphism  $P(\mathbb{A}) \to \overline{SL}_2(\mathbb{A})$ , so that when  $P'_{\mathbb{A}}$  denotes the inverse image, we have

$$P(\mathbb{A}) \times \mathbb{T} \simeq P'_{\mathbb{A}}.$$

Similarly, the group  $K_{\infty}$  come with a corresponding group in  $K'_{\infty}$  in  $\overline{\mathrm{SL}}_2(\mathbb{A})$ . However, for  $K_f$ , passing to a subgroup is necessary, due to the place p = 2. If that component is

<sup>&</sup>lt;sup>5</sup>Note that other authors oftentimes choose  $\vartheta \mapsto -\vartheta$ , even though this results in a negative direction of rotation. However, as a consequence, the standard character does not have to be conjugated in the transformation formula of  $\Theta$  functions (cf. (5.12)).

<sup>&</sup>lt;sup>6</sup>The group P is a parabolic subgroup which manifests itself in the notation.

replaced with  $K_0(4)_2 = \overline{\Gamma_0(4)}$ , which is the completion of  $\Gamma_0(4)$  in  $K_2 = \mathrm{SL}_2(\mathbb{Z}_2)$ , then

$$K_0(4) = K_0(4)_2 \times \prod_{2$$

is an open compact subgroup of  $SL_2(\mathbb{A}_f)$  and there is a splitting morphism  $K_0(4) \to \overline{SL}_2(\mathbb{A})$ .

Further, the group  $SL_2(\mathbb{A})$  might be exhausted by simple components – a feature known as *strong approximation*, after Kneser [Kne65]. Namely, for any compact open subgroup  $K_{\mathbb{A}_f}$ , multiplication by  $G_{\mathbb{R}}$  and  $SL_2(\mathbb{Q})$  suffices to reconstruct the whole group  $SL_2(\mathbb{A})$ :

$$\operatorname{SL}_2(\mathbb{A}) = \operatorname{SL}_2(\mathbb{Q}) \left( \operatorname{SL}_2(\mathbb{R}) \times K_f \right),$$

where  $SL_2(\mathbb{Q})$  is to be understood as lying diagonally in  $SL_2(\mathbb{A})$ . If  $K_0$  is chosen to be an open compact subgroup in  $K_0(4)$  and identified with its image in  $\overline{SL}_2(\mathbb{A})$  under the splitting morphism, the same result holds for the metaplectic group:

$$\overline{\mathrm{SL}}_2(\mathbb{A}) = \overline{\mathrm{SL}}_2(\mathbb{Q}) \,\overline{\mathrm{SL}}_2(\mathbb{R}) K_0. \tag{3.24}$$

### 3.2.1 Classical Weil representation and relation

In the extension  $\overline{\mathrm{SL}}_2(\mathbb{Q}_p)$  there is a twofold cover  $\mathrm{Mp}_2(\mathbb{Q}_p)$  included [Kud94] satisfying the analogue of (3.15), namely

$$1 \to \{\pm 1\} \to \operatorname{Mp}_2(\mathbb{Q}_p) \to \operatorname{SL}_2(\mathbb{Q}_p) \to 1.$$
(3.25)

Two fold covers of  $SL_2(\mathbb{Q}_p)$  are unique up to isomorphism and are relevant for considering half integral weight modular forms. In addition to generalising the weight to the half integral case, vector valued modular forms, like in Section 3.3, may be considered. For these, the most prominent choice of representation is the Weil representation we have encountered above.

In fact, there is also a classical notion of Weil representations on  $Mp_2(\mathbb{Z})$ , which may be taken to be the subgroup of  $Mp_2(\mathbb{R})$  projecting to  $SL_2(\mathbb{Z})$ , dating back to Schoenberg in [Sch39]. We shall understand such a representation adelically before verifying that it, in fact, coincides with the classical definition found in the literature (cf. [Str13] which builds upon [Sch09]). The classical notion is related to discriminant forms (cf. Remark 1.27) which are without exception representable by quotients of even  $\mathbb{Z}$  lattices. Let (L, q) be such an even non-degenerate lattice. From that object we obtain in a natural way a quadratic space  $(V_p, q)$  by taking a tensor product with  $\mathbb{Q}_p$ . The sections above explain how a representation of  $Mp_2(\mathbb{A})$  on  $\mathcal{S}_{\mathbb{A}}(V_{\mathbb{A}})$  is induced by this – the associated Weil representation  $\omega_{\mathbb{A}}$ . We follow [Kud03] and [BY09] for sketching the extraction of a discrete representation on a finite dimensional space from  $\omega_{\mathbb{A}}$ . Let  $K' < Mp_2(\mathbb{A})$  be the preimage of  $SL_2(\hat{\mathbb{Z}}) < SL_2(\mathbb{A}_f)$  under the natural projection to  $SL_2(\mathbb{A})$ . Recall that there is a canonical splitting morphism of  $SL_2(\mathbb{Q})$  into  $Mp_2(\mathbb{A})$ . Then

$$Mp_2(\mathbb{A}) = SL_2(\mathbb{Q}) Mp_2(\mathbb{R}) K'.$$
(3.26)

Note that  $\operatorname{SL}_2(\mathbb{Z}) \simeq \operatorname{Mp}_2(\mathbb{Q}) \cap \operatorname{Mp}_2(\mathbb{R})K'$  and that for any  $\gamma \in \operatorname{Mp}_2(\mathbb{Z})$  there are unique elements  $\gamma_0 \in \operatorname{SL}_2(\mathbb{Z})$  and  $\gamma' \in K'$  such that

$$\gamma_0 = \gamma \gamma'.$$

The association  $\operatorname{Mp}_2(\mathbb{Z}) \ni \gamma \mapsto \gamma' \in K'$  defines a homomorphism and the idea is to construct a finite dimensional subspace of  $\mathcal{S}(V_{\mathbb{A}_f})$  that arises naturally from L and is invariant under the action of  $\operatorname{Mp}_2(\mathbb{Z})$  via  $\omega_f$ . Then we had a finite dimensional representation of  $\operatorname{Mp}_2(\mathbb{Z})$ .

Note that for  $\hat{L} = L \otimes \hat{\mathbb{Z}}$  we have that  $\mathbb{1}_{\mu+\hat{L}} \in \mathcal{S}(V_{\mathbb{A}_f})$  with  $\mu \in L'/L$ . Recall Example 1.33 to verify that this notation is meaningful and the denoted functions generate a space  $\mathcal{S}_{L'/L} < \mathcal{S}(V_{\mathbb{A}_f})$  of complex dimension |L'/L|. Next, we verify that this space is invariant under  $\omega_{\mathbb{A}_f}|_{\mathrm{Mp}_2(\mathbb{Z})}$ . For that purpose, it is sufficient to consider the representation locally. Further, (3.16) reveals that the action of the representation  $\omega_p$  only differs by an element of  $\mathbb{T}$  from the action of  $\mathrm{pr}_{r_p}$ . But the latter defined a projective representation of  $\mathrm{SL}_2(\mathbb{Q}_p)$ and hence it suffices to consider the action of elements of  $\mathrm{SL}_2(\mathbb{Z})$  via  $\mathrm{pr}_p$ . The group  $\mathrm{SL}_2(\mathbb{Z})$  is, however, generated by the two matrices T representing translation by 1 and Srepresenting an inversion at i (cf. Definition 2.12).

By (3.12), we find for  $l \in L'_p$  that

$${}^{\mathrm{pr}}r_p(T)\mathbb{1}_{l+L_p}(l+x) = e_p(q(l+x))\mathbb{1}_{l+L_p}(l+x) = e_p(q(l))\mathbb{1}_{l+L_p}(l+x) \stackrel{\cdot}{=} \mathbb{1}_{l+L_p}(l+x)$$

For the action of S we remark that by (3.13) it suffices to consider only the operation of the Fourier transformation. By assuming a self dual measure, Remark 1.54 then yields in

the notation of Remark 1.57 that

$$\mathcal{F}[\mathbb{1}_{l+\hat{L}}(x)](y) = e_f^{\mathbf{b}}(-l,y) \cdot \mathbb{1}_{\hat{L}'}(y)$$

which is again in  $S_{L'/L}$ , as  $e_f^{\rm b}(l, \cdot)$  descends to a character on L'/L. Hence, we obtain a representation

$$\rho_L : \operatorname{Mp}_2(\mathbb{Z}) \ni \gamma \mapsto \omega^*_{\mathbb{A}_f}(\gamma') : \mathcal{S}_{L'/L} \to \mathcal{S}_{L'/L}, \qquad (3.27)$$

where  $\omega^*$  denotes the dual representation. Recall that the normalisation of the local measure

$$\mu_p^{\rm b}(L_p) = \sqrt{|L'_p/L_p|}^{-1} = \sqrt{|\det(S)|_p}$$

implies  $\hat{\mu}^{b}(\hat{L}) = \sqrt{|L'/L|}^{-1} = \sqrt{|S|_{\infty}}^{-1}$  for the measure of the finite adele module, where S denotes the Gram matrix of L. Of course, there is a scalar product on the space  $S_{L'/L}$  that will be normalised to  $\langle \mathbb{1}_{\hat{L}}, \mathbb{1}_{\hat{L}} \rangle = 1$  for reasons that will become clear later on. Note that  $S_{L'/L}$  in fact carries the structure of a commutative  $\mathbb{C}$  algebra, by considering convolution as a product. We have

$$[\mathbb{1}_{l+L_p} * \mathbb{1}_{l'+L_p}](y) = \int_{\mathbb{Q}_p} \mathbb{1}_{l+L_p}(x) \mathbb{1}_{l'+L_p}(y-x) \, \mathrm{d}\mu_p(x)$$
$$= \int_{L_p} \mathbb{1}_L(y-(l+l')-x) \, \mathrm{d}\mu_p(x)$$
$$= \mathbb{1}_{l+l'+L_p}(y)$$

meaning that the convolution product coincides with the one induced by the group structure of the discriminant form L'/L. Hence, we have proven the following lemma.

**Lemma 3.9.** For an even non-degenerate lattice L the space  $S_{L'/L}$  equipped with the convolution product is isometrically isomorphic to the group algebra  $\mathbb{C}[L'/L]$ 

Further, we conclude the existence of a natural representation on the group algebra  $\mathbb{C}[L'/L]$ , based on the discussion succeeding (3.27).

**Corollary 3.10.** For an even  $\mathbb{Z}$  lattice there is a natural representation of  $Mp_2(\mathbb{Z})$  on the group algebra  $\mathbb{C}[L'/L]$ , which we denote  $\rho_L$  and call (integral) Weil representation.

We would like to describe this representation more explicitly. For that purpose, the group  $Mp_2(\mathbb{Z})$  has to be examined more closely. We begin by studying the twofold

metaplectic cover of  $\operatorname{GL}_2^+(\mathbb{R})$  which we will denote by  $\overline{\operatorname{GL}_2^+}(\mathbb{R})$ . In order to provide a convenient model for this group, we require the *factor of automorphy j* that has been given in Definition 2.10. Equipped with that notion,  $\overline{\operatorname{GL}_2^+}(\mathbb{R})$  may be realised as the group of pairs  $(\gamma, \phi)$ , where  $\gamma \in \operatorname{GL}_2^+(\mathbb{R})$  and  $\phi$  denotes a holomorphic square root of  $j(\gamma, \cdot)$  on  $\mathbb{H}$ . Multiplication is given by

$$(\gamma, \phi) \cdot (\gamma', \phi') = (\gamma \gamma', \phi(\gamma' \cdot) \phi'). \tag{3.28}$$

Note that  $\overline{\operatorname{GL}_2^+}(\mathbb{R})$  inherits an action on  $\mathbb{H}$  from  $\operatorname{GL}_2(\mathbb{R})^+$  via pullback along the covering map. Namely, for  $(\gamma, \phi) \in \overline{\operatorname{GL}_2^+}$  with  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$  and  $\tau \in \mathbb{H}$ , set

$$(\gamma, \phi)\tau \coloneqq \gamma\tau = \frac{a\tau + b}{c\tau + d}.$$
(3.29)

Also, the preimage of  $SO_2$  in  $\overline{\operatorname{GL}_2^+}(\mathbb{R})$  will be denoted  $\overline{SO}_2$  and defines a double cover. Recall that the preimage of  $SL_2(\mathbb{Z})$  in  $\overline{\operatorname{GL}_2^+}(\mathbb{R})$  is denoted by  $Mp_2(\mathbb{Z})$ . It is then generated by the two elements

$$\overline{T} := \left( \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, 1 \right), \tag{3.30}$$

$$\overline{S} \coloneqq \left( \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}, \sqrt{\tau} \right) \tag{3.31}$$

which project to the two generators T and S of  $SL_2(\mathbb{Z})$  (cf. Definition 2.12). The element

$$\overline{Z} := \overline{S}^2 = (\overline{ST})^3 = \left( \begin{pmatrix} -1 & 0\\ 0 & -1 \end{pmatrix}, i \right)$$
(3.32)

generates the centre of  $Mp_2(\mathbb{Z})$ .

**Remark 3.11.** To ease the notation, an element  $\gamma \in \text{SL}_2(\mathbb{Z})$  will be associated to  $(\gamma, \phi) \in \text{Mp}_2(\mathbb{Z})$  for the standard choice of square root  $\phi$  of the factor of automorphy  $j(\gamma, \tau)$ . We will also write  $\tilde{\gamma} = (\gamma, \phi)$ . Note that this association does not define a section of groups as it is not a homomorphism, but only in terms of covering maps!

Similarly, we embed  $\Gamma_{\infty}$  into  $Mp_2(\mathbb{Z})$  by sending  $\gamma \mapsto (\gamma, 1)$ . This, however, is a section in the category of groups.

Comparing the transformation property of theta functions in the adelic setting with

the transformation rule of theta functions in the classical setting (cf. [Bor98, Thm. 4.1 p. 505]) yields the identification of  $\rho_L$  as the classical Weil representation [BY09, p. 639]. The concrete formulae are given as follows.

**Remark 3.12.** Let (L, q) be an even non-degenerate lattice and  $\mathcal{L} = (L'/L, \overline{q})$  be the associated discriminant form. It suffices to describe the action of  $\rho_L$  on the generators  $\overline{T}$  and  $\overline{S}$ . These operations are given by

$$\rho_L(\overline{T})\mathbf{e}_{\lambda} = e(\overline{\mathbf{q}}(\lambda)) \cdot \mathbf{e}_{\lambda}, \tag{3.33}$$

$$\rho_L(\overline{S})\mathfrak{e}_{\lambda} = \frac{e(-\operatorname{sig}(\mathcal{L})/8)}{\sqrt{|\mathcal{L}|}} \cdot \sum_{\mu \in \mathcal{L}} e(-\beta(\lambda,\mu))\mathfrak{e}_{\mu}.$$
(3.34)

From the remark above it is apparent that the representation  $\rho_L$  only depends on the discriminant form  $\mathcal{L}$ .

**Definition 3.13.** Let  $(\mathcal{L}, \overline{q})$  be a discriminant form. Then  $\rho_{\mathcal{L}} \coloneqq \rho_L$  is a unitary representation of  $Mp_2(\mathbb{Z})$  on  $\mathbb{C}[\mathcal{L}]$  that is fully determined by (3.33) and (3.34).

We require some more insights into this representation for later investigations and will, in the following, recollect a number of useful facts based on [BS08].

**Remark 3.14.** a) Note that the generator of the centre  $\overline{Z}$  operates for  $\lambda \in \mathcal{L}$  as

$$\rho_{\mathcal{L}}(\overline{Z})\mathfrak{e}_{\lambda} = e(-\operatorname{sig}(\mathcal{L})/4) \cdot \mathfrak{e}_{-\lambda}.$$
(3.35)

As a consequence, the Weil representation  $\rho_{\mathcal{L}}$  factors through  $Mp_2(\mathbb{Z})/\langle \overline{Z}^2 \rangle \simeq$  $SL_2(\mathbb{Z})$ , if, and only if,  $sig(\mathcal{L}) \equiv 0 \mod 2$ .

b) Another useful action which is computable is the one of  $\overline{U} = \overline{ST}^{-1}\overline{S}^{-1}$ :

$$\rho_{\mathcal{L}}(\overline{U}^m)\mathfrak{e}_{\lambda} = \frac{1}{|D|} \sum_{\mu,\nu\in D} e\left(-mq(\mu) + \beta(\mu,\lambda-\nu)\right)\mathfrak{e}_{\nu}$$
(3.36)

found in [BS08, 2 Lem. 2.3 p. 254].

c) Further, following [BS08, 2 Lem. 2.1 p. 253] define for integers  $a, d \in \mathbb{Z}$  fulfilling gcd(a, N) = 1 = gcd(d, N) such that  $a \cdot d \equiv 1 \mod N$  the element

$$R_d \coloneqq \overline{ST}^d \overline{S}^{-1} \overline{T}^a \overline{ST}^d$$

It has the form  $R_d = (\gamma, \phi)$  with  $\gamma \equiv \begin{pmatrix} a & 0 \\ 0 & d \end{pmatrix} \mod N$ . Then

$$\rho_{\mathcal{L}}(R_d)\mathbf{e}_{\lambda} = \frac{G_{\mathcal{L}}(d)}{G_{\mathcal{L}}(1)} \cdot \mathbf{e}_{d\lambda},\tag{3.37}$$

where  $G_{\mathcal{L}}(d) = \sum_{\nu \in \mathcal{L}} e(d \,\overline{\mathbf{q}}(\nu))$  are Gauss sums (cf. Definition A.16).

**Remark 3.15.** Let  $(L, q) = (L_1, q_1) \oplus (L_2, q_2)$  be a direct composition into even lattices. Recall that  $\mathbb{C}[\mathcal{L}] = \mathbb{C}[\mathcal{L}_1] \otimes \mathbb{C}[\mathcal{L}_2]$ . Then  $\rho_L = \rho_{L_1} \otimes \rho_{L_2}$ .

**Remark 3.16.** In case of even rank it had been noted above that the representation  $\rho_{\mathcal{L}}$  factors through  $SL_2(\mathbb{Z})$ . Further, writing  $N := \text{lev}(\mathcal{L})$  we find that  $\rho_{\mathcal{L}}$  is trivial on  $\Gamma(N)$  for even rank, so that it factors through the finite group

$$\operatorname{SL}_2(\mathbb{Z}/N\mathbb{Z}) \simeq \Gamma(1)/\Gamma(N).$$

In case of odd rank, the oddity formula [Joh98, Chap. 15 p. 383 (30)] implies  $4 \mid N$ , in particular,  $\mathcal{L}$  contains 2-adic Jordan components. In this case, there is a well known section (see also [Shi73, p. 447])

$$s: \Gamma_0(4) \to \overline{\Gamma}_0(4), \qquad \gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \mapsto \left( \begin{pmatrix} a & b \\ c & d \end{pmatrix}, \epsilon_d^{-1} \begin{pmatrix} c \\ \overline{d} \end{pmatrix} \sqrt{j(\gamma, \tau)} \right)$$
(3.38)

with  $\epsilon_d = 1$  or *i*, depending on whether  $d \equiv 1$  or 3 mod 4. Here,  $\overline{\Gamma_0}(4)$  denotes the preimage of  $\Gamma_0(4)$  under the projection  $\overline{\operatorname{SL}}_2(\mathbb{Z}) \to \operatorname{SL}_2(\mathbb{Z})$ .<sup>7</sup> The argument at the end of [Bor00, Thm. 5.4 p. 330] yields that  $\rho_{\mathcal{L}}$  is trivial on  $s(\Gamma(N))$  and factors through the central extension of  $\operatorname{SL}_2(\mathbb{Z}/N\mathbb{Z})$  by  $\{\pm 1\}$  given by

$$\operatorname{Mp}_2(\mathbb{Z})/s(\Gamma(N)).$$

Finally, we require the following piece of notation for the computation of Fourier expansions of non-holomorphic Eisenstein series.

**Definition 3.17.** For  $\lambda, \mu \in \mathcal{L}$  and  $(\gamma, \phi) \in Mp_2(\mathbb{Z})$ , we introduce the following notation for the  $\lambda, \mu$ -coefficient:

$$\rho_{\lambda,\mu}(\gamma,\phi) \coloneqq \langle \rho_L(\gamma,\phi) \mathbf{e}_{\lambda}, \mathbf{e}_{\mu} \rangle. \tag{3.39}$$

<sup>&</sup>lt;sup>7</sup>Note that only the section on  $\Gamma(N)$  maps to Mp<sub>2</sub> and  $\Gamma_0(N)$  is mapped to a central extension by  $\mathbb{C}^{\times}$ .

## 3.3 Vector valued modular forms

The theory of scalar valued modular forms is rich and attempting to generalise it in different directions is apparent. Recall that the compellingly simple construction of Jacobi's theta function

$$\sum_{n\in\mathbb{Z}}q^n$$

yields a function that is not modular in the standard sense, but almost. In addition, classical functions like the Dedekind eta function are almost modular in the sense that they transform with a multiplier, which is correcting signs. Also recall that Theorem 2.22 yields a whole class of theta functions that transform almost like a modular form - up to a character. Characters are unital one dimensional representations and cocycles appear when projecting representations from coverings to the underlying groups. So it appears natural to generalise the concept of modular forms to include a (possibly projective) representation. However, the step from allowing a character to allowing a multidimensional representation may not be immediately intelligible and we will recall an example. Consider the generalised theta function (compare Definition 2.17) for an even positive definite lattice (L, q)

$$\theta_{\lambda+L} : \mathbb{H} \to \mathbb{C}, \quad \sum_{x \in \lambda+L} q^{\mathbf{q}(x)},$$

with  $\lambda \in L'/L$  and  $q = \exp(2\pi i \tau)$ . Note that then

$$\theta_{\lambda+L}|_{n/2}S$$

is in fact a linear combination of theta functions of this very form  $\theta_{\mu+L}$  for  $\mu \in L'/L$ . (cf. [Ebe12, 3.1 p. 81 (T2)]). To rephrase this phenomenon: there is a linear action intertwining the different theta functions  $\theta_{\lambda+L}$  associated to an even lattice L. In fact, their action is compatible in the above sense.

**Theorem 3.18** ([Bor98, Thm. 4.1 p. 505]). For an even lattice (L,q), the theta series  $(\theta_{\lambda+L})_{\lambda\in L'/L}$  is a vector valued modular form for Mp<sub>2</sub>( $\mathbb{Z}$ ) that transforms under the Weil representation associated to the discriminant form  $(L'/L, \overline{q})$ .

This means that the linear intertwining of the individual components may be described by a complex representation, yielding a new perspective on the topic at hand. Indeed, the theory of vector valued modular forms is suited to unify all of the above phenomena in a single framework and we shall, in the following, review the elementary parts of its
theory. The first published record of a call for the development of the theory of vector valued modular forms the author is aware of is from Selberg [Sel65]. The reader may find a source for vector valued modular forms in a quite general setting in [Gan14]. The subcase that is required for this thesis is the one of vector valued modular forms for the Weil representation  $\rho_L$  of Mp<sub>2</sub>(Z) and it will be discussed in the following. However, note that not all naturally occurring vector valued modular forms are associated to these types of representations – compare for instance [Zhu96].

# 3.3.1 Discrete Weil representation

For the following, compare [Str13] and [BS08].

One of the most significant examples of representations for considering vector valued modular forms is the discrete Weil representation introduced in Subsection 3.2.1.

Recall that for a discriminant form  $(\mathcal{L}, \overline{\mathbf{q}})$  there is a unitary representation  $\rho_{\mathcal{L}}$ : Mp<sub>2</sub>( $\mathbb{Z}$ )  $\rightarrow$  Aut( $\mathbb{C}[\mathcal{L}]$ ) introduced in (3.27) and described in Remark 3.12. As before, set  $e(x) = \exp(2\pi i x)$  and recall that the representation is determined by the actions of the two generators  $\overline{T} = (\begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, 1)$ , and  $\overline{S} = \begin{pmatrix} \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}, \sqrt{\tau} \end{pmatrix}$  of Mp<sub>2</sub>( $\mathbb{Z}$ ) given by

$$\begin{split} \rho_{\mathcal{L}}(\overline{T}) \mathfrak{e}_{\lambda} &= e(\overline{\mathbf{q}}(\lambda)) \cdot \mathfrak{e}_{\lambda}, \\ \rho_{\mathcal{L}}(\overline{S}) \mathfrak{e}_{\lambda} &= \frac{e(-\operatorname{sig}(\mathcal{L})/8)}{\sqrt{|\mathcal{L}|}} \cdot \sum_{\mu \in \mathcal{L}} e(-\beta(\lambda,\mu)) \mathfrak{e}_{\mu}. \end{split}$$

After this recap, we may introduce vector valued modular forms.

**Definition 3.19.** A function  $f : \mathbb{H} \to \mathbb{C}[L'/L]$  is called *vector valued modular function* of weight  $k \in \frac{1}{2}\mathbb{Z}$  for  $\Gamma \leq \mathrm{Mp}_2(\mathbb{Z})$  if for all  $\gamma = (M, \phi) \in \Gamma$ 

$$f(\gamma \tau) = \phi(\tau)^{2k} \cdot \left[\rho_L(\gamma)f\right](\tau).$$

It carries the predicate holomorphic, weakly holomorphic, meromorphic, or real analytic, if it satisfies the usual conditions on  $\Gamma \setminus \mathbb{H}$  respectively. The associated spaces are denoted by  $\mathcal{M}_{L,k}, \mathcal{M}_{L,k}^!, \mathcal{A}_{L,k}, \mathcal{R}_{L,k}$  in accordance with Table 2.1.

By Remark 3.16, we recognise that every component function of such a vector valued modular form is a scalar valued modular form for  $\Gamma(N)$  for some  $N \in \mathbb{N}$ . However, following [Bor98, Exa. 2.2 p. 500] we realise that any scalar valued modular form may be induced to a vector valued modular form to Mp<sub>2</sub>( $\mathbb{Z}$ ) for some Weil representation. Hence, we we will stick with the case of vector valued modular forms of level 1; meaning for the full group  $Mp_2(\mathbb{Z})$ .

**Remark 3.20.** In case of  $2k \neq \operatorname{sig}(\mathcal{L}) \mod 2$ , we find  $\mathcal{M}_{L,k} = \{0\} = \mathcal{R}_{L,k}$ .

*Proof*: In fact, applying  $|_{\mathcal{L}}\overline{Z}$  to a function f twice yields

$$e(\operatorname{sig}(\mathcal{L})/2 - k) \cdot f.$$

Now if f was a modular form for  $Mp_2(\mathbb{Z})$ , i.e. invariant under that action, we concluded that either f = 0 or  $e(k - \operatorname{sig}(\mathcal{L})/2) = 1$  which is equivalent to  $2k \equiv \operatorname{sig}(\mathcal{L}) \mod 2$ .  $\Box$ 

**Remark 3.21.** If  $f \in \mathcal{M}_{L,k}^!$  we find that

$$f|_{L,k}\overline{T} = \sum_{\lambda \in L'/L} e(-\overline{\mathrm{q}}(\lambda)) f_{\lambda}(\tau+1) \mathfrak{e}_{\lambda}.$$

By comparing components we infer that  $e(-\overline{q}(\lambda)(\tau+1))f_{\lambda}(\tau+1) = e(-\overline{q}(\lambda)(\tau))f_{\lambda}(\tau)$ , i.e.  $e(-\overline{q}(\lambda)(\cdot))f_{\lambda}(\cdot)$  is 1 periodic and holomorphic in a neighbourhood of infinity, meaning it possesses a Fourier expansion. As a consequence, we find

$$f(\tau) = \sum_{\lambda \in L'/L} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) e(n\tau) \mathfrak{e}_{\lambda}$$
(3.40)

for appropriate numbers  $a(\lambda, n) \in \mathbb{C}$ . In particular, the set of admissible indices (cf. Definition 3.67)

$$\mathscr{I} = \{(\lambda, n) \in \mathcal{L} \times \mathbb{Q} \mid n \equiv \overline{\mathbf{q}}(\lambda)\}$$

interpreted as a subset of the dual space  $(\mathcal{M}_{k,L}^!)'$  is separating.

**Remark 3.22.** Let  $f \in \mathcal{M}_{L,k}^!$  with Fourier expansion as in (3.40). Applying  $|_{\mathcal{L}}\overline{Z}$  to f yields

$$f_{\lambda} = e[(\operatorname{sig}(\mathcal{L}) - 2k)/4] \cdot f_{-\lambda}$$
(3.41)

for any  $\lambda \in \mathcal{L}$ . In particular, assuming  $\lambda = -\lambda$  which may only happen if  $\lambda$  is trivial in all p components of  $\mathcal{L}$  for p > 2, we conclude that  $f_{\lambda}$  vanishes, unless  $\operatorname{sig}(\mathcal{L}) \equiv 2k \mod 4$ . On the level of Fourier coefficients (3.41) means that these fulfil the following symmetry property

$$a(-\lambda, n) = e[(\operatorname{sig}(\mathcal{L}) - 2k)/4] \cdot a(\lambda, n).$$
(3.42)

In fact, applying  $|_{\mathcal{L}}\overline{Z}$  (cf. Example 3.14 a)) yields

$$\sum_{\lambda \in L'/L} \sum_{n \in \overline{q}(\lambda) + \mathbb{Z}} a(\lambda, n) e(n\tau) \mathfrak{e}_{\lambda} = e(\operatorname{sig}(\mathcal{L})/4 - k/2) \sum_{\lambda \in L'/L} \sum_{n \in \overline{q}(\lambda) + \mathbb{Z}} a(\lambda, n) e(n\tau) \mathfrak{e}_{-\lambda}.$$

While properties of vector valued modular forms have already been discussed, we have yet to verify their existence. Hence, we will have a look at the following trivial example, before turning towards more interesting construction principles.

Example 3.23. Let L be unimodular, i.e.  $\mathcal{L} = L'/L = \{0\}$ . Then  $\mathbb{C}[\mathcal{L}] \simeq \mathbb{C}$  and necessarily  $\operatorname{rk}(L) = m \equiv 0 \mod 8$ , so that the discrete Weil representation  $\rho_L$  acts trivially by Remark 3.12. This implies that the notion of a vector valued modular form for the Weil representation coincides with the classical notion of a scalar valued modular form in that case. Also note that by Remark 3.20 the weight k must be even in that context.

## 3.3.2 Some constructions

There are various construction principles for vector valued modular forms. We will present a selection of these principles and begin by making direct observations. We will then introduce concrete classes of modular forms and subsequently move on to discuss more advanced construction principles before touching on some structure results.

### Combining known forms

Clearly, we may abstractly induce forms from subrepresentations, but this has already been cast aside as a special technical case when explaining that we stick to forms for the full group  $Mp_2(\mathbb{Z})$ . Instead, we recall ideas from the classical setting where we have encountered two accessible construction principles: products and symmetrisation. The latter has been used to construct Eisenstein series, which will also be revisited below, while the further offered a trivial construction of new forms and will be stated first.

**Remark 3.24.** Consider finitely many discriminant forms  $(\mathcal{L}_i, \overline{\mathbf{q}}_i)$  for  $1 \leq i \leq m$  and for  $k_i \in \mathbb{Z}/2$  modular forms  $f_i \in \mathcal{M}_{\mathcal{L}_i, k_i}$  with Fourier expansion

$$f_i(\tau) = \sum_{\lambda_i \in \mathcal{L}_i} \sum_{n_i \in \overline{\mathbf{q}}_i(\lambda_i) + \mathbb{Z}} c_i(\lambda_i, n_i) e(n_i \tau) \mathfrak{e}_{\lambda_i}.$$

For  $\mathcal{L} = \bigoplus_i \mathcal{L}_i$  we use the notation  $\underline{\lambda} = (\lambda_i)_i \in \bigoplus_i \mathcal{L}_i = \mathcal{L}$  as well as  $\underline{n} = (n_i)_i$  to define

the function

$$\bigotimes_{i} f_{i} : \mathbb{H} \ni \tau \mapsto \sum_{\underline{\lambda} \in \mathcal{L}} \sum_{\substack{\underline{n} \in \mathbb{Q}^{m}, \\ \forall i: n_{i} \in \overline{\mathbf{q}}_{i}(\lambda_{i}) + \mathbb{Z}}} c(\underline{\lambda}, \underline{n}) e(\underline{n}\tau) \cdot \mathbf{e}_{\underline{\lambda}} \in \mathbb{C}[\mathcal{L}]$$

where  $c(\underline{\lambda}, \underline{n}) = \prod_i c_i(\lambda_i, n_i), e(\underline{n}\tau) = e(\sum_i n_i\tau)$ , and  $\mathfrak{e}_{\underline{\lambda}} = \bigotimes_i \mathfrak{e}_{\lambda_i}$ . Then for  $k = \sum_i k_i$  we have that

$$f_1 \otimes \ldots \otimes f_m \coloneqq \bigotimes_i f_i \in \mathcal{M}_{\mathcal{L},k}.$$

The above remark offers trivial means to construct new forms from already known ones, but so far we have not encountered a single true vector valued modular form. Hence, selecting an even lattice L of level lev $(L) \mid N$  and a modular form for  $\Gamma(N)$ , we may inject this form into a component of  $\mathbb{C}[L'/L]$ . This yields a vector valued modular form for  $\Gamma(N)$ which may be symmetrised to a vector valued modular form for the full group  $\mathrm{SL}_2(\mathbb{Z})$ . Such a construction has been given by Scheithauer in [Sch15, Thm. 3.1 p. 7] along others and we will present a special case of the more delicate variant for  $\Gamma_0(N)$ .

Note that the Weil representation acts in fact via a character  $\chi_{\mathcal{L}}$  under  $\Gamma_0(\text{lev}(L))$  on the zero component of  $\mathbb{C}[L'/L]$ . This character is given by (cf. [Sch15, p. 7] for details)

$$\chi_{\mathcal{L}}\left(\begin{pmatrix}a&b\\c&d\end{pmatrix}\right) = \left(\frac{a}{|\mathcal{L}|}\right)e[(a-1)\operatorname{oddity}(\mathcal{L})/8].$$

**Proposition 3.25.** Let L be an even lattice,  $N \in \mathbb{N}$  with  $lev(L) \mid N$  and  $\chi_{\mathcal{L}}$  be the character of  $SL_2(\mathbb{Z})$  presented above. Given a modular form  $f \in \mathcal{M}_k(\Gamma_0(N), \chi_{\mathcal{L}})$ , we find

$$F(\tau) \coloneqq \sum_{\gamma \in \Gamma_0(N) \setminus \Gamma(1)} f|_k \gamma \rho_L(\gamma)^{-1} \mathfrak{e}_0 \in \mathcal{M}_{L,k}.$$

Remarkably, a significant portion of vector valued modular forms are of this type. Namely, if lev(L) is assumed to be square free and we choose  $F \in \mathcal{M}_{L,k}$  that is invariant under the action of O(L'/L), then F may be represented as such a lift of a  $\Gamma_0(\text{lev}(L))$ scalar valued modular form (cf. [Sch15, Cor. 5.5 p. 24]).

#### Siegel theta series

We will briefly review Siegel theta series, based on [Bor98]. In this subsection let (L, q) be an even lattice of signature  $(m^+, m^-)$  and rank  $m = m^+ + m^-$ . Further, let  $\mathcal{P} : \mathbb{R}^m \to \mathbb{C}$  be a polynomial. We identify  $\mathbb{R}^m$  with the underlying space of the quadratic

space  $\mathbb{R}^{(m^+,m^-)}$  (cf. Example 1.5). Then we say that  $\mathcal{P}$  has degree  $(\kappa^+,\kappa^-)$ , if  $\mathcal{P}$  has degree  $\kappa^+$  in the first  $m^+$  variables and degree  $\kappa^-$  in the last  $m^-$  variables.

Denote by  $\sigma : L \otimes_{\mathbb{Z}} \mathbb{R} \to \mathbb{R}^{(m^+,m^-)}$  an isometry and set  $z \coloneqq \sigma^{-1}(\mathbb{R}^{0,m^-})$ . Then  $z^{\perp} = \sigma^{-1}(\mathbb{R}^{m^+,0})$  and  $V = z^{\perp} \oplus z$  with  $z \in \mathbb{D}$ . For any element  $v \in V_{\mathbb{R}} = L \otimes \mathbb{R}$  introduce the decomposition  $v = v_{z^{\perp}} + v_z \in z^{\perp} \oplus z$ . With this notation, we may introduce Siegel theta functions.

**Definition 3.26.** Let  $\lambda \in L'/L$  and  $\mathcal{P}$  be homogeneous of degree  $(\kappa^+, \kappa^-), \tau = u + iv \in \mathbb{H}$ and  $\sigma, z$  be as above as well as  $\alpha, \beta \in V$ . Define

$$\theta_{L,\lambda}(\tau,\alpha,\beta,\sigma,\mathcal{P}) \coloneqq v^{\frac{m^{-}}{2}+\kappa^{-}} \sum_{\mu \in L+\lambda} \exp\left(-\frac{\Delta}{8\pi y}\right) (\mathcal{P})(\sigma(\mu+\beta)) \times e\left(\tau \operatorname{q}[(\mu+\beta)_{z^{\perp}}] + \overline{\tau} \operatorname{q}[(\mu+\beta)_{z}] - \operatorname{b}(\mu+\beta/2,\alpha)\right),$$
(3.43)

where  $\Delta$  is the Laplacian.

**Remark 3.27.** Note that the prefactor  $v^{\frac{m^-}{2}+\kappa^-}$  has been introduced in order to completely shift the transformation to the holomorphic variable as in [MZ23].

**Remark 3.28.** In the particularly relevant case of  $\alpha = 0 = \beta$  these elements are ommitted from the notation and the function may be written in more plane terms. To this end, note that  $q(v) = q(v_{z^{\perp}}) + q(v_z)$  so that we write  $q_z(v) \coloneqq q(v_z)$  and call  $q_z^+$  given by  $q_z^+(v) \coloneqq q(v) - 2q_z(v) = q(v_{z^{\perp}}) - q(v_z)$  the standard majorant of q with respect to  $z \in \mathbb{D}$ . With this notation, we find for  $\tau = u + iv \in \mathbb{H}$  that

$$\theta_{L,\lambda}(\tau,\sigma,\mathcal{P}) = v^{\frac{m^{-}}{2} + \kappa^{-}} \sum_{\mu \in L + \lambda} \exp\left(-\frac{\Delta}{8\pi y}\right) (\mathcal{P})(\sigma(\mu)) e\left(u \operatorname{q}(\mu) + iv \operatorname{q}_{z}^{+}(\mu)\right).$$
(3.44)

We have defined such a Siegel theta function for every  $\lambda \in \mathcal{L} = L'/L$  and consider these to carry local information. Piecing them together creates a complete theta function.

**Definition 3.29.** Let  $\mathcal{P}$  be homogeneous of degree  $(\kappa^+, \kappa^-)$ ,  $\tau = u + iv \in \mathbb{H}$  and  $\sigma$ , z be as above and  $\alpha, \beta \in V$ . Define the *Siegel* theta function of L to be

$$\Theta_L(\tau, \alpha, \beta, \sigma, \mathcal{P}) \coloneqq \sum_{\lambda \in \mathcal{L}} \theta_{L,\lambda}(\tau, \alpha, \beta, \sigma, \mathcal{P}) \mathfrak{e}_{\lambda}.$$
(3.45)

In case  $\alpha = \beta = 0$ , these are omitted from the notation. The same holds in case  $\mathcal{P} = 1$ .

Borcherds has proven the following transformation property for Siegel theta functions.

**Theorem 3.30** ([Bor98, Thm. 4.1 p. 505]). For  $\mathcal{P}$  a homogeneous polynomial of degree  $(\kappa^+, \kappa^-)$  and every  $\gamma = \left( \begin{pmatrix} a & b \\ c & d \end{pmatrix}, \phi \right) \in \operatorname{Mp}_2(\mathbb{Z})$ , we find

$$\Theta_L(\gamma\tau, a\alpha + b\beta, c\alpha + d\beta, \sigma, \mathcal{P}) = \phi^{m^+ + 2\kappa^+ - (m^- + 2\kappa^-)} \rho_L(\gamma) \Theta_L(\tau, \alpha, \beta, \sigma, \mathcal{P}).$$
(3.46)

As usual, the proof of the transformation property is reduced to the cases of the two generators  $\overline{T}$  and  $\overline{S}$  of Mp<sub>2</sub>( $\mathbb{Z}$ ). The first identity is a triviality, while the second is achieved with the typical tool of Poisson summation.

In Section 5.2, theta functions are introduced in adelic fashion, depending on a choice of Schwartz function which is why we introduce the following semiclassical perspective.

#### Theta functions for general Schwartz forms

The Siegel theta functions introduced above already represent a fairly general framework for theta functions which suffices for most applications. However, the kernel symmetrised in that setting has a very particular form and while technically being adequate for the applications within this thesis, formulae appearing in the later parts of this thesis are much more amenable in a different notation, arising from a more conceptual approach to theta functions. In order to approach another construction perspective recall the classical setting for theta functions (cf. Definition 2.17) which were constructed by building a family of Schwartz functions  $\varphi(\tau) \in S(V_{\mathbb{R}})$  from the form q and symmetrising  $\varphi(\tau)$  over the lattice L. These series were clearly periodic and constructed in such a fashion that Poisson summation yielded a transformation property for the action of  $S \in SL_2(\mathbb{Z})$ . However, in case of an indefinite quadratic form, the choice  $V_{\mathbb{R}} \ni x \mapsto \exp(2\pi i q(x)\tau)$  does not define a Schwartz function, for it is not rapidly decreasing. Hence, we will have to

select a different variant of that prototype in order to guarantee convergence. In addition, it is unclear how to construct a whole family  $\varphi(\tau) \in \mathcal{S}(V_{\mathbb{R}})$  of Schwartz functions. In order to shed light on a possible solution we have to take another perspective on the classical situation.

We note that in the classical setting for a positive definite quadratic space (V, q) of even signature and

$$\varphi: V_{\mathbb{R}} \to \mathbb{C}, \qquad v \mapsto \exp(-2\pi \operatorname{q}(v))$$

we find for  $\tau = u + iv \in \mathbb{H}$ ,  $g_{\tau} = N(u)M(\sqrt{v})$  as in (3.22) and  $\omega_{\infty}$  the Weil representation of Example 3.8 that

$$\omega_{\infty}(g_{\tau})\varphi(x) = e^{2\pi i \operatorname{q}(v)u} v^{m/4} \varphi(\sqrt{v}x) = v^{m/4} \exp(2\pi i \operatorname{q}(x)\tau).$$
(3.47)

With that new perspective, we may define theta functions associated to Schwartz forms. First, recall that there is a section  $P_{\infty} \to \operatorname{Mp}_2(\mathbb{R})$  of the standard Borel subgroup, so that  $g_{\tau}$  may directly be interpreted as an element in  $\operatorname{Mp}_2(\mathbb{R})$ .

**Definition 3.31.** Let  $\varphi \in \mathcal{S}(V_{\mathbb{R}})$  and  $\lambda \in \mathcal{L} = L'/L$ . Then

$$\theta_{L,\lambda}(\tau;\varphi) = v^{-m/4} \cdot \sum_{x \in L+\lambda} [\omega_{\infty}(g_{\tau})\varphi](x)$$

is called *theta function* associated to  $\varphi$  and the coset  $\lambda + L$ .

We note that we may let  $O(V_{\mathbb{R}})$  act on  $\varphi$  via the left regular representation, i.e.

$$h: \varphi \mapsto \varphi(h^{-1} \cdot) \in \mathcal{S}(V_{\mathbb{R}}).$$

In that sense,  $\theta_{L,\lambda}$  may be understood as a function on  $\mathbb{H} \times O(V_{\mathbb{R}})$ . Combining the above theta functions for different cosets of the dual lattice yields the following theta function.

**Definition 3.32.** Let  $\varphi \in \mathcal{S}(V_{\mathbb{R}})$ . Then

$$\Theta_L(\tau,h;\varphi) = v^{-m/4} \cdot \sum_{x \in L'} [\omega_{\infty}(g_{\tau})\varphi](h^{-1}x) \mathfrak{e}_{x+L}$$

is called *theta function* associated to  $\varphi$ . In case the Schwartz form  $\varphi$  is invariant under the action of a maximal compact  $K \subseteq O(V_{\mathbb{R}})$ , we find that the dependence on h factors through  $\mathbb{D}$  and we write  $\Theta_L(\tau, z; \varphi)$  for  $z \in \mathbb{D}$ . Also, for fixed z, we omit this variable.

A prominent example is the classical case based on (3.47) given that  $(V_{\mathbb{R}}, \mathbf{q})$  is positive definite. However, in case of an indefinite quadratic form, the naively chosen function  $V_{\mathbb{R}} \ni x \mapsto \exp(-2\pi \mathbf{q}(x))$  does not define a Schwartz function, for it is not rapidly decreasing. Recall that we had for an element  $z \in \mathbb{D}$  defined the standard majorant

$$\mathbf{q}_z^+(v) = \mathbf{q}(v_{z^\perp}) - \mathbf{q}(v_z)$$

which is positive definite. This enables us to associated theta functions to indefinite quadratic spaces by introducing a parameter  $z \in \mathbb{D}$ .

*Example* 3.33. With  $q_z^+ : V_{\mathbb{R}} \to \mathbb{R}$  as above,  $\varphi(x) = e(i q_z^+(x))$ , and  $\tau = u + iv \in \mathbb{H}$  we find

$$\Theta_L(\tau, z; \varphi) = \sum_{x \in L'} e\left( u \operatorname{q}(x) + iv \operatorname{q}_z^+(x) \right) \mathfrak{e}_{x+L}$$

We will revisit this construction later in Section 5.2, where it will be given adellically. Before finishing the current subsection we note that theta functions do factor, provided the associated Schwartz forms do.

**Remark 3.34.** Let the lattice  $L = L_1 \oplus L_2$  split and assume  $\varphi_{\infty} \in \mathcal{S}(V_{\infty})$  splits as a tensor product  $\varphi_{\infty} = \varphi'_{\infty} \otimes \varphi''_{\infty} \in \mathcal{S}(V_{21\infty}) \otimes \mathcal{S}(V_{2,\infty})$ . Then, if the theta function  $\Theta(\tau; \varphi_{\infty})$  is absolutely convergent, it also decomposes as a tensor product

$$\Theta_L(\tau;\varphi_\infty) = \Theta_{L_1}(\tau;\varphi'_\infty) \otimes \Theta_{L_2}(\tau;\varphi''_\infty).$$
(3.48)

Proof: A straight forward calculation yields the desired result

$$\begin{split} \Theta_{L}(\tau;\varphi_{\infty}) \\ &= v^{-\frac{m}{4}} \cdot \sum_{\substack{(\lambda_{1},\lambda_{2}) \in L_{1}^{\prime}/L_{1} \oplus L_{2}^{\prime}/L_{2}}} \sum_{l_{1} \in L_{1}+\lambda_{1}, l_{2} \in L_{\ell}+\lambda_{2}} \left(\omega_{\infty}(g_{\tau})\varphi_{\infty}\left((l_{1},l_{2})\right)\right) \mathbf{e}_{\lambda_{1}+\lambda_{2}} \\ &= \sum_{\substack{\lambda_{1} \in L_{1}^{\prime}/L_{1}}} v^{-\frac{1}{4}} \sum_{l_{1} \in L_{1}+\lambda_{1}} \left(\omega_{\infty}(g_{\tau})\varphi_{\infty}^{\prime}\right)(l_{1}) \cdot v^{-\frac{m-1}{4}} \sum_{l_{2} \in L_{2}+\lambda_{2}} \left(\omega_{\infty}(g_{\tau})\varphi_{\infty}^{\prime\prime}\right)(l_{2}) \mathbf{e}_{\lambda_{1}+\lambda_{2}} \\ &= \sum_{\substack{\lambda_{1} \in L_{1}^{\prime}/L_{1}\\ \lambda_{2} \in L_{2}^{\prime}/L_{2}}} \theta_{L_{1},\lambda_{1}}(\tau;\varphi_{\infty}^{\prime}) \cdot \theta_{L_{2},\lambda_{2}}(\tau;\varphi_{\infty}^{\prime\prime}) \mathbf{e}_{\lambda_{1}+\lambda_{2}} \\ &= \Theta_{L_{1}}(\tau;\varphi_{\infty}^{\prime}) \otimes \Theta_{L_{2}}(\tau;\varphi_{\infty}^{\prime\prime}). \end{split}$$

### Lifts between discriminant forms

Assuming we have an even lattice L and a sublattice  $M \leq L$ , it is natural to ask, whether modular forms for one of these lattices may be used to construct modular forms for the other. In fact, there is a rather natural construction for relating functions in the spaces  $\mathbb{C}[M'/M]^{\mathbb{H}}$  and  $\mathbb{C}[L'/L]^{\mathbb{H}}$ . However, its following discussion is technical and the reader only interested in references may consult (3.49) and (3.51) as well as Proposition 3.36.



Figure 3.3.1: Visualisation of natural projections and injections between quotients related to  $M \leq L$ . Diagonal arrows correspond to trivial mappings. The significant rectangle for our construction is framed in blue. Passing to the generated group algebras, the diagram remains the same.

To this end, let

$$M \le L \le L' \le M'$$

be a chain of lattices of finite index, so that  $L/M \leq L'/M \leq M'/M$  – compare Figure 3.3.1.

Recall the construction of the coordinate mappings  $M'/M \ni \mu \mapsto (f \mapsto f_{\mu})$ . An element  $f \in \mathcal{A}_{M'/M,k}$  is determined by its values under all of these mappings. Hence, it suffices to construct from such an f images for the respecting mappings  $L'/L \ni \nu \mapsto g_{\nu}$ . We may reduce to the case  $f(\tau) \in \mathbb{C}^{M'/M}$  by fixing  $\tau$ .

Clearly, an element in  $\mathbb{C}^{M'/L}$  has a natural reduction to an element in  $\mathbb{C}^{L'/L}$ , by pulling back along the natural injection  $\iota : L'/L \to M'/L$ . Hence, constructing from  $f(\tau) \in \mathbb{C}^{M'/M}$ a natural element in  $\mathbb{C}^{M'/L}$  suffices. If there was a natural section s to the projection  $M'/M \to M'/L$ , applying the pullback of it to  $f(\tau)$  yielded the desired result. However, the projection  $\pi : M'/M \to M'/L$  might not be injective, in fact its kernel is L/M, so that it represents a |L/M| fold cover of M'/L on which the kernel L/M acts on the fibres. Hence, for an arbitrary section  $s_0$  to  $\pi$  we obtain for every  $\nu \in L/M$  another section  $s_{\nu} : \mu \mapsto s_0(\mu) + \nu$ . Their images cover M'/M disjointly so that the mapping

$$s^*: \sum_{\nu \in \ker(\pi)} (s_{\nu})^*: \mathbb{C}^{M'/M} \to \mathbb{C}^{M'/L}$$

is well defined and, in particular, independent of any choice of  $\nu$  and  $s_0$ . As a consequence, there is the natural map

$$\iota^* \circ s^* : \mathbb{C}^{M'/M} \to \mathbb{C}^{M'/L} \to \mathbb{C}^{L'/L}.$$

In conclusion, there is an operator

$$\downarrow_L^M : \mathbb{C}[M'/M]^{\mathbb{H}} \to \mathbb{C}[L'/L]^{\mathbb{H}}, \quad g \mapsto \downarrow_L^M g, \tag{3.49}$$

with

$$(\downarrow_L^M g)_{\overline{\mu}} = \sum_{\nu \in L/M} g_{\mu+\nu} \tag{3.50}$$

where  $\mu \coloneqq s_0 \circ \iota(\overline{\mu})$  is an arbitrary but fixed lift of some  $\overline{\mu} \in L'/L$  to M'/M.

**Remark 3.35.** The operator  $\downarrow_L^M$  translates the theta function associated with M to a theta function of L. In the notation of Definition 3.32 this means

$$\downarrow_{L}^{M} (\Theta_{M}(\tau, z; \varphi)) = \downarrow_{L}^{M} \left( \sum_{\lambda \in M'/M} \sum_{x \in M+\lambda} (\omega_{\infty}(g_{\tau})\varphi)(x, z) \mathbf{e}_{\lambda} \right)$$
$$= \sum_{\lambda \in L'/L} \sum_{x \in L+\lambda} (\omega_{\infty}(g_{\tau})\varphi)(x, z) \mathbf{e}_{\lambda}$$
$$= \Theta_{L}(\tau, z; \varphi).$$

The operator  $\downarrow_L^M$  clearly preserves smoothness and polynomial or exponential growth conditions so that it preserves the property of being a modular (cusp) form, if only it translated the transformation property. This will become apparent later, so that we continue to ask for its adjoint operator with respect to the Petersson scalar product.

Let  $f \in \mathcal{A}_{L,k}, g \in \mathcal{A}_{M,k}$  such that their Petersson scalar product is defined and compute

$$\langle f, \downarrow_L^M g \rangle_L = \sum_{\overline{\mu} \in L'/L} f_{\overline{\mu}} \cdot \overline{\downarrow_L^M g_{\overline{\mu}}}$$

$$= \sum_{\overline{\mu} \in L'/L} f_{\overline{\mu}} \cdot \overline{\sum_{\alpha \in L/M} g_{\mu+\alpha}}$$

$$= \sum_{\overline{\mu} \in L'/L} f_{\overline{\mu}} \cdot \overline{\sum_{\alpha \in L'/M} g_{\alpha}}$$

$$= \sum_{\mu \in L'/M} f_{\overline{\mu}} \cdot \overline{g_{\mu}}$$
$$= \sum_{\mu \in M'/M} \left(\uparrow_L^M f\right)_{\mu} \cdot \overline{g_{\mu}}$$
$$= \langle\uparrow_L^M f, g\rangle_M$$

where  $\uparrow_L^M f$  denotes the naive embedding from  $f \in \mathbb{C}^{L'/L}$  to  $\mathbb{C}^{M'/M}$  given by  $f_{\overline{\mu}}$ , if  $\mu \in L'/M$  and 0, otherwise. Explicitly,

$$\left(\uparrow_{L}^{M}f\right)_{\mu} = \begin{cases} f_{\overline{\mu}}, & \text{if } \mu \in L'/M, \\ 0, & \text{otherwise,} \end{cases}$$
(3.51)

which defines the lift

$$\uparrow_L^M: \mathcal{A}_{L,k} \to \mathcal{A}_{M,k}, \quad g \mapsto g_M. \tag{3.52}$$

Note that these lifts are not inverse to each other, as  $\downarrow_L^M \circ \uparrow_L^M = |L/M| \cdot \text{id}$  and  $\uparrow_L^M \circ \downarrow_L^M$  equals  $|L/M| \cdot \text{id}$  only on the in general proper subspace given by the image of  $\uparrow_L^M$ , whose elements are supported on L'/M. In fact, the latter gives a complete description of the image of  $\uparrow_L^M$ , in particular, any element  $f \in \mathcal{M}_{M,k}$  supported on L'/M satisfies  $f_{\mu} = f_{\mu+\mu'}$  for  $\mu' \in L/M$  (cf. [Bru14, Prop. 3.3]).

In fact, there is an underlying more conceptual construction of such operators which is also the origin of the naming convention in this thesis [Sch15, Sec. 4] and [Bru14, Sec. 3]. In fact, for a discriminant group D and an isotropic subgroup H, the group  $D' := H^{\perp}/H$ defines a discriminant group of the same signature. Then there are operators

$$\begin{split} \uparrow^D_{D'} \colon M_{k,D'} \to M_{k,D}, \qquad g \mapsto \sum_{\mu \in H^\perp} g_{\mu+H} \mathfrak{e}_{\mu}, \\ \downarrow^D_{D'} \colon M_{k,D} \to M_{k,D'}, \qquad f \mapsto \sum_{\mu \in H^\perp} f_\mu \mathfrak{e}_{\mu+H}. \end{split}$$

The relation to the operators constructed in this thesis is explained in [Sch15, p. 16] and is the following: The subgroup  $H := L/M \leq M'/M =: D$  is isotropic and we find  $H^{\perp} = L'/M$ , yielding the natural isomorphism  $D' = H^{\perp}/H \simeq L'/L$ . The formulae are then identical to these above and the respective results of Scheithauer and Bruinier yield that the operators do in fact transfer the transformation properties with respect to the respective Weil representations. That is, the operators map modular forms to modular forms.

Note that the more abstract description is in fact not more general, but equivalent, for if  $H \leq D$ , its preimage  $\pi^{-1}(H) \subseteq M'$  under the projection  $\pi : M' \to D$  is an even sublattice. In fact, if  $H \coloneqq L/M$ , we obtain  $\pi^{-1}(H) = L$ .

In conclusion we find the following proposition.

**Proposition 3.36.** For  $M \leq L$  even non-degenerate lattices and  $k \in \mathbb{Z}/2$ , there are linear operators

$$\downarrow_L^M: \mathcal{A}_{M,k} \to \mathcal{A}_{L,k}, \qquad \uparrow_L^M: \mathcal{A}_{L,k} \to \mathcal{A}_{M,k}$$

fulfilling the following properties.

- a) The operators map modular (cusp) forms to modular (cusp) forms.
- b) The lift  $\downarrow_L^M$  is compatible with theta functions, i.e.  $\downarrow_L^M (\Theta_M) = \Theta_L$ .
- c) The operators are adjoint to each other on the space of cusp forms:

$$\langle \cdot, \downarrow_L^M (\cdot) \rangle_L = \langle \uparrow_L^M (\cdot), \cdot \rangle_M.$$

This identity also holds, if one of the cusp forms is replaced by a modular form.

d) The image  $B := \operatorname{im}(\uparrow_L^M)$  is the subspace consisting of functions that are supported on L'/M. We find

$$\downarrow_L^M \circ \uparrow_L^M = |L/M| \cdot \mathrm{id},$$
$$\uparrow_L^M \circ \downarrow_L^M |_B = |L/M| \cdot \mathrm{id}_B.$$

Note that the latter identity is an abuse of notation, since  $\uparrow_L^M$  had to formally be endowed with a restriction of its target to B.

#### Eisenstein series

The following exposition is based upon [Kie21, 2.5 pp. 30-36]. The statements are essential for our investigations, more specifically for deriving meromorphic continuation of the symmetric square type *L*-functions arising from the Rankin–Selberg type integrals in Subsection 6.4.3. In the following, let (L, q) be a non-degenerate lattice and  $k \in \mathbb{Z}/2$  be a half integral number. **Definition 3.37.** Let  $\lambda \in \mathcal{L}$  be isotropic and  $k \in \mathbb{Z}/2$ . Define the *Eisenstein series* 

$$E_{L,\lambda,k}(\tau,s) \coloneqq \frac{1}{2} \sum_{\gamma \in \overline{\Gamma_{\infty}} \setminus \operatorname{Mp}_{2}(\mathbb{Z})} \left(\operatorname{Im}(\tau)^{s} \mathfrak{e}_{\lambda}\right)|_{L,k} \gamma.$$
(3.53)

By construction this function is  $Mp_2(\mathbb{Z})$  modular of weight k.

**Lemma 3.38.** The Eisenstein series  $E_{L,\lambda,k}$  converges normally on  $\mathbb{H}$  for  $\operatorname{Re}(s) > 1 - \frac{k}{2}$ , is real analytic in  $\tau$  and an eigenfunction of the hyperbolic Laplace operator of weight k with Eigenvalue s(s + k - 1).

**Remark 3.39.** Recall that the Weil representation  $\rho_L$  acts trivially on  $\Gamma(\text{lev}(L))$ , so that the components of  $E_{L,\lambda,k}(\tau, s)$  are scalar-valued Eisenstein series for  $\Gamma(\text{lev}(L))$ . These, however, have meromorphic continuation in s.

Bruinier and Kühn compute the Fourier expansion of these Eisenstein series. Note that they do so with respect to the dual Weil representation, which is why the following is a slight reformulation of their statement, taking that difference into account.

**Proposition 3.40** ([BK03, Prop. 3.1 p. 1695]). For sig( $\mathcal{L}$ )  $\equiv 2k \mod 2$ , the Eisenstein series  $E_{L,\lambda,k}$  has the Fourier expansion

$$E_{L,\lambda,k}(\tau,s) = \sum_{\mu \in L'/L} \sum_{n \in \mathbb{Z} + \overline{\mathbf{q}}(\mu)} c_{\lambda}(\mu, n, s, v) \mathbf{e}(nu) \mathbf{e}_{\gamma}$$
(3.54)

where the coefficients  $c_{\lambda}(\mu, n, s, v)$  are given by

$$\begin{cases} (\delta_{\lambda,\mu} + i^{\operatorname{sig}(\mathcal{L}) + 2k} \delta_{-\lambda,\mu}) v^s + 2\pi v^{1-k-s} \frac{\Gamma(k+2s-1)}{\Gamma(k+s)\Gamma(s)} \cdot \sum_{x \in \mathbb{Z} \setminus \{0\}} |2c|^{1-k-2s} H_c(\beta, 0, \mu, 0), & \text{if } n = 0, \\ \frac{2^k \pi^{s+k-1}}{\Gamma(s+k)} \mathcal{W}_s(4\pi nv) \cdot \sum_{c \in \mathbb{Z} \setminus \{0\}} |c|^{1-k-2s} H_c(\lambda, 0, \mu, n), & \text{if } n > 0, \\ \frac{2^k \pi^{s+k-1}}{\Gamma(s)} \mathcal{W}_s(4\pi nv) \cdot \sum_{c \in \mathbb{Z} \setminus \{0\}} |c|^{1-k-2s} H_c(\lambda, 0, \mu, n), & \text{if } n < 0, \end{cases}$$

Here,  $H_c$  denotes the generalised Kloostermann sum

$$H_{c}(\lambda, m, \mu, n) = \frac{e^{-\pi i \operatorname{sgn}(c)k/2}}{|c|} \sum_{\substack{0 \neq d \mod c \\ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \Gamma(1)_{\infty} \setminus \Gamma(1)/\Gamma(1)_{\infty}}} \overline{\rho}_{\mu,\lambda} \begin{pmatrix} a & b \\ c & d \end{pmatrix} \operatorname{e} \left(\frac{am + nd}{c}\right), \quad (3.55)$$

the matrix coefficients  $\rho_{\mu,\lambda}$  are as in Definition 3.17, and the Whittaker functions  $W_s$  are given in Definition A.20.

For the notation  $\tilde{\gamma} \in Mp_2(\mathbb{Z})$  for an element  $\gamma \in SL_2(\mathbb{Z})$ , compare Remark 3.11.

*Proof*: In [BK03] only the case  $sig(\mathcal{L}) + 2k \equiv 0 \mod 4$  is considered. In order to be certain that, apart from the obvious change to the constant coefficient, no additional term has to be altered, we compute the Fourier coefficients by consulting [Bru02, 1.2.3 p. 23]. Begin with

$$c_{\lambda}(\mu, n, s, v) = \frac{1}{2} \int_{0}^{1} \Big\langle \sum_{(\gamma, \phi) \in \widehat{\Gamma(1)_{\infty}} \setminus \operatorname{Mp}_{2}(\mathbb{Z})} \operatorname{Im}(\tau)^{s} \mathfrak{e}_{\lambda}|_{k} \gamma, \mathfrak{e}_{\mu}(n\overline{\tau}) \Big\rangle.$$

Note that there are 4 non-trivial representatives  $(\gamma, \phi)$  with  $\gamma_{2,1} = 0$ , namely  $\overline{Z}^0, \overline{Z}^1, \overline{Z}^2, \overline{Z}^3$ . The remaining matrices are represented by the double coset  $\Gamma(1)_{\infty} \setminus \Gamma(1) / \Gamma(1)_{\infty}$ . Recall that by Remark 3.20 we may reduce to the case  $\operatorname{sig}(\mathcal{L}) \equiv 2k \mod 2$ , so that  $\overline{Z}$  acts trivially and we find

$$\begin{split} c_{\lambda}(\mu,n,s,v) &= \int_{0}^{1} \langle \mathfrak{e}_{\lambda}, \mathfrak{e}_{\mu}(n\overline{\tau}) \rangle \, \mathrm{d}u \cdot v^{s} + \int_{0}^{1} \langle \mathfrak{e}_{\lambda}|_{L,k} \overline{Z}, \mathfrak{e}_{\mu}(n\overline{\tau}) \rangle \, \mathrm{d}u \cdot v^{s} \\ &+ \sum_{\substack{\gamma \in \Gamma(1)_{\infty} \backslash \Gamma(1) \\ \gamma_{2,1} \neq 0}} \int_{0}^{1} \frac{\mathrm{Im}(\tau)^{s}}{|j(\gamma,\tau)|^{2s}} j(\gamma,\tau)^{-k} \big\langle \rho(\tilde{\gamma})^{-1} \mathfrak{e}_{\lambda}, \mathfrak{e}_{\mu}(n\overline{\tau}) \big\rangle \, \mathrm{d}u \end{split}$$

Define  $\epsilon_k \coloneqq i^{\operatorname{sig}(\mathcal{L})+2k}$  and recall that by Remark 3.22 we have

$$\mathfrak{e}_{\lambda}|_{L,k}\overline{Z} = \epsilon_k \mathfrak{e}_{-\lambda}.$$

Further, we find that

$$\left\langle \rho(\tilde{\gamma})^{-1} \mathfrak{e}_{\lambda}(m\gamma\tau), \mathfrak{e}_{\mu}(n\overline{\tau}) \right\rangle = \overline{\rho_{\mu,\lambda}(\tilde{\gamma})} \mathrm{e}(m\gamma\tau) \mathrm{e}(-n\tau).$$

Note that by unitaricity, the expression  $\overline{\rho_{\mu,\lambda}(\tilde{\gamma})}$  does not depend on altering  $\tilde{\gamma}$  from the right by an element of  $\overline{\Gamma(1)}_{\infty}$ . With these computations, we obtain

$$c_{\lambda}(\mu, n, s, v) = \delta_{0,n} \left( \delta_{\lambda,\mu} + \epsilon_k \delta_{-\lambda,\mu} \right) v^s + \sum_{\gamma \in \Gamma(1)_{\infty} \setminus \Gamma(1), \gamma_{2,1} \neq 0} \int_0^1 \frac{\operatorname{Im}(\tau)^s}{|j(\gamma, \tau)|^{2s}} j(\gamma, \tau)^{-k} \overline{\rho_{\mu,\lambda}(\tilde{\gamma})} e(-n\tau) \, \mathrm{d}u = \delta_{0,n} \left( \delta_{\lambda,\mu} + \epsilon_k \delta_{-\lambda,\mu} \right) v^s$$

$$+\sum_{\substack{\gamma\in\Gamma(1)_{\infty}\setminus\Gamma(1)/\Gamma(1)_{\infty}\\\gamma_{2,1}\neq 0}}\overline{\rho_{\mu,\lambda}(\tilde{\gamma})}v^{s}\exp(2\pi v)\int_{-\infty}^{\infty}\frac{\mathrm{e}(-nu)}{|j(\gamma,\tau)|^{2s}}j(\gamma,\tau)^{-k}\,\mathrm{d}u$$

Now writing  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$ , we have

$$\sqrt{j(\gamma,\tau)} = \operatorname{sgn}(c)\sqrt{c}\sqrt{\tau+d/c},$$

so that we find for the appearing integral

$$v^{s} \exp(2\pi v) \int_{-\infty}^{\infty} \frac{\mathrm{e}(-nx)}{|j(\gamma,\tau)|^{2s}} j(\gamma,\tau)^{-k} \,\mathrm{d}u$$
$$= v^{s} \exp(2\pi v) |c|^{-k-2s} \operatorname{sgn}(c)^{k} \int_{-\infty}^{\infty} \frac{\mathrm{e}(-nu)}{(\tau+d/c)^{k+s} (\overline{\tau}+d/c)^{s}} \,\mathrm{d}u$$
$$= v^{s} \exp(2\pi v) |c|^{-k-2s} \operatorname{sgn}(c)^{k} e\left(\frac{nd}{c}\right) \int_{-\infty}^{\infty} \frac{\mathrm{e}(-nu)}{\tau^{k+s}\overline{\tau}^{s}} \,\mathrm{d}u$$
$$= v^{s} \exp(2\pi v) |c|^{-k-2s} \operatorname{sgn}(c)^{k} e\left(\frac{nd}{c}\right) \int_{-\infty}^{\infty} \frac{\mathrm{e}(-nu)}{\tau^{k+s}\overline{\tau}^{s}} \,\mathrm{d}u.$$

The integral

$$\int_{-\infty}^{\infty} \frac{\mathbf{e}(-nu)}{\tau^{k+s}\overline{\tau}^s} \, \mathrm{d}u = \int_{-\infty}^{\infty} \frac{\mathbf{e}(-nu)}{(v-iu)^{k+s}(v+iu)^s} \, \mathrm{d}u$$

is then rewritten by Bruinier and Kühn in terms of Whittaker functions as

$$2^{k}\pi^{s+k}i^{-k}|n|^{s+k-1}y^{-s}\mathcal{W}_{s}(4\pi ny)\begin{cases} \Gamma(k+s)^{-1}, & n>0,\\ \Gamma(s)^{-1}, & n<0, \end{cases}$$

and

$$2^{2-k-2s}\pi i^{-k}\frac{\Gamma(k+2s-1)}{\Gamma(k+s)\Gamma(s)}y^{1-k-2s}, \quad \text{in case of } n=0.$$

Finally, we find for the sum

$$\sum_{\substack{\left(\begin{array}{c}a\\c\end{array}\right)\\c\neq 0}}|c|^{-k-2s}\operatorname{sgn}(c)^{k}\overline{\rho}_{\mu,\lambda}(\tilde{\gamma})e\left(\frac{nd}{c}\right)$$
$$=\sum_{c\neq 0}|c|^{-k-2s}\operatorname{sgn}(c)^{k}\sum_{\substack{0\neq d\mod c\\\left(\begin{array}{c}a\\c\end{array}\right)\\\in\Gamma(1)_{\infty}\setminus\Gamma(1)/\Gamma(1)_{\infty}}}\overline{\rho}_{\mu,\lambda}(\tilde{\gamma})e\left(\frac{nd}{c}\right)$$
$$=\sum_{c\neq 0}|c|^{1-k-2s}H_{c}(\lambda,0,\mu,n).$$

**Remark 3.41.** The Fourier expansion of a slight generalisation, with non-trivial periodic term, of the above is computed in [Völ18, Prop 3.6.7 p. 58]. In that case, however, the series is not an eigenfunction of the hyperbolic Laplacian anymore, but suffers from a shift in the parameter s.

We also introduce the following notation for simplifying statements further below.

$$c_{\lambda}(\mu, n, s, v) = \begin{cases} (\delta_{\lambda, \mu} + i^{\operatorname{sig}(\mathcal{L}) + 2k} \delta_{-\lambda, \mu}) v^s + c_{\lambda}(\mu, 0, s) v^{1-k-s}, & n = 0, \\ c_{\lambda}(\mu, n, s) \mathcal{W}_s(4\pi n v), & n \neq 0. \end{cases}$$
(3.56)

**Proposition 3.42.** Let k > 2, then  $E_{L,\lambda,k}(\tau,0)$  defines a vector valued holomorphic modular form with respect to  $\rho_L$  and  $Mp_2(\mathbb{Z})$ , in short  $E_{L,\lambda,k}(\tau,0) \in \mathcal{M}_{L,k}$ . Its Fourier expansion is of the form

$$E_{L,\lambda,k}(\tau) = \mathfrak{e}_{\lambda} + (-1)^k \mathfrak{e}_{\lambda} + \sum_{\gamma \in L'/L} \sum_{0 < n \in \mathbb{Z} + \overline{\mathbf{q}}(\gamma)} c_{\lambda}(\gamma, n) \mathfrak{e}_{\lambda}(n\tau),$$

where we set  $c_{\lambda}(\gamma, n) \coloneqq c_{\lambda}(\gamma, n, 0)$ .

According to [Hej83, p. 372] we have the following transformation property.

**Proposition 3.43.** For an isotropic element  $\lambda \in \mathcal{L}$ , we find the following functional equation

$$E_{L,\lambda,k}(\tau,s) = \frac{1}{2} \sum_{\mu \in \text{Iso}(L'/L)} c_{\lambda}(\mu,0,s) E_{L,\mu,k}(\tau,1-k-s).$$
(3.57)

From that transformation, a symmetry property for a special value of the Eisenstein series is derived which may naively play a role in investigating the values of special L-functions in Subsection 6.4.3 – also compare Remark 6.99 on that matter.

**Corollary 3.44.** Imposing the additional condition  $\lambda = -\lambda$  on an isotropic element  $\lambda \in \mathcal{L}$ and assuming k > 2 for the purpose of convergence, we obtain for the choice  $s = s_0 = 0$ that

$$E_{L,\lambda,k}(\tau,s_0) = \frac{2-\delta_{\lambda,0}}{2} \left(1+i^{\operatorname{sig}(\mathcal{L})+2k}\right) \cdot E_{L,\lambda,k}(\tau,1-k-s_0).$$
(3.58)

From the Fourier expansion of  $E_{L,\lambda,k}$  and the properties of the Whittaker function  $\mathcal{W}_s$ (cf. [Erd81, 6.9 p. 264], [Olv+10, Sec. 13.14], and [DAR84, Sec. 13 p. 189]) we derive an asymptotic bound for the Eisenstein series. **Corollary 3.45.** We find for  $\tau = u + iv \in \mathbb{H}$  and  $\sigma = \max\{\operatorname{Re}(s), \operatorname{Re}(1-k-s)\}$  that

$$E_{L,\lambda,k}(\tau,s) = \mathcal{O}(v^{\sigma}), \qquad v \to \infty.$$
 (3.59)

In fact, the constant required to bound may be chosen locally uniformly in s.

Proof: Indeed, the constant term of the expansion gives rise to the growth ~  $v^{\sigma}$ , while the other terms may be locally uniformly bounded in s. Note that the coefficients  $\rho_{\mu,\lambda}$ are universally bounded by 1, since  $\rho$  is unitary, effectively bounding the Kloostermann sum (3.55) in the Fourier expansion of  $E_{L,\lambda,k}$ . This gives rise to a universal bound of the infinite series in the coefficients in (3.54) on right half planes by comparing it to the Riemann  $\zeta$  function. The exponential terms may be uniformly bounded in s on vertical strips, as may the  $\Gamma$  factors in the Numerator. Further, the Beta function is entire and as such continuous, yielding a locally uniformly bound in s. Finally, Lemma A.21 yields the desired bound on the remaining Whittaker functions to justify an absolute bound on the series of all other coefficients by bounding it by a geometric series.

**Remark 3.46.** This implies the same bound for the scalar valued Eisenstein series by selecting the lattice L to be unimodular.

### Parabolic Poincaré series

The construction of Eisenstein series above may be significantly generalised. Instead of sticking with  $\mathfrak{e}_{\lambda}$  for some  $\lambda \in \mathcal{L}$  with  $\overline{q}(\lambda) = 0$  isotropic for a kernel, we may select a general element  $\lambda \in \mathcal{L}$  and adjust for the additional prefactor that came with the transformation under  $\overline{T}$  by cancelling it with a proper choice of the following form

$$\mathfrak{e}_{\lambda}(m\tau) \coloneqq e(m\tau) \cdot \mathfrak{e}_{\lambda}.$$

Now for  $m \in \mathbb{Z} + \overline{q}(\lambda)$ , we find that  $\mathfrak{e}_{\lambda}(m\tau)|_{L,k}\overline{T} = \mathfrak{e}_{\lambda}(m\tau)$  for any weight k. The generalisation goes further. In fact, instead of symmetrising  $\operatorname{Im}(\tau)^s$  we may consider much more general functions. To this end, let  $\psi : \mathbb{R}_{>0} \to \mathbb{C}$  be piecewise infinitely differentiable and for reasons of convergence, assume there is an exponent  $\alpha \in \mathbb{R}_{\geq 0}$  such that  $\psi(v) = \mathcal{O}(v^{\alpha})$  for  $v \to 0$ . Then we find

$$\psi(v)\mathbf{e}_{\lambda}(m\tau)|_{L,k}\overline{T} = \psi(v)\mathbf{e}_{\lambda}(m\tau),$$

so that the following expression is well defined.

**Definition 3.47.** Let  $\lambda \in L'/L$  and  $m \in \overline{q}(\lambda) + \mathbb{Z}$ . Further, let  $\psi \in \mathcal{L}^1_{loc}(\mathbb{R}_{>0})$ . Then

$$P_{\lambda,m,\psi,k}(\tau) \coloneqq \frac{1}{2} \sum_{(\gamma,\phi)\in\overline{\Gamma_{\infty}}\setminus\operatorname{Mp}_{2}(\mathbb{Z})} \psi(v)\mathfrak{e}_{\lambda}(m\tau)|_{L,k}(\gamma,\phi)$$
(3.60)

is called the *(parabolic) Poincaré series* of weight k with parameters  $\lambda$ , m and  $\psi$ .

Clearly,  $\psi$  may be replaced by a family  $\psi_s$  of functions with the analogous property depending on a complex parameter s and we write  $P_{\lambda,m,\psi,k}(\tau,s)$  for the Poincaré series in that case. This way, we may reproduce the Eisenstein series from the section above. The Fourier series of such parabolic Poincaré series for special choices of  $\psi$  may be computed. Compare [Bru02, Thm. 1.4 p. 19] for the choice  $\psi = 1$  and for different choices of  $\psi$  which play a role in the field of Maass forms consult [Völ18, p. 54].

**Remark 3.48.** Assume there is  $\alpha \in \mathbb{R}$  such that  $\psi(v) = \mathcal{O}(v^{\alpha})$  for  $v \to 0$  and that the weight fulfils  $k \geq 2(1 - \alpha)$ . Then  $P_{\lambda,m,\psi,k}(\tau)$  may be proven to be locally uniformly convergent, by comparing it to the classical Eisenstein series  $E_k(\tau, \alpha)$ . Hence, if  $\psi$  is (locally) continuous or holomorphic, so is  $P_{\lambda,m,\psi,k}(\tau)$ .

**Remark 3.49.** Note that  $\psi \mathfrak{e}_{\lambda}(m\tau)|_{L,k}\overline{Z} = \psi \mathfrak{e}_{-\lambda}(m\tau)$ , so that the kernel, and as such also the Poincaré series, are invariant under the action of  $\overline{Z}^2$ .

### Petersson scalar product

As in the scalar valued case, the space of cusp forms is equipped with a scalar product. This subsection serves two purposes: firstly, it introduces this product, and secondly, it realises pairings of cusp forms against more general automorphic forms which are required for constructing a class of L-series as Rankin–Selberg type integrals in Subsection 6.4.3.

**Definition 3.50.** Let  $f \in S_{L,k}$  and  $g \in \mathcal{M}_{L,k}$ . Define the *Petersson product* of f and g to equal

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, g \rangle \cdot \operatorname{Im}^{k} \, \mathrm{d}\mu, \qquad (3.61)$$

where  $d\mu$  is the standard invariant hyperbolic measure on  $\mathbb{H}$  as in Definition 2.15 b) and  $\mathcal{F}_{\Gamma(1)}$  denotes a fundamental domain for  $\Gamma(1)$  (cf. (2.3)).

We will need to verify that this expression is convergent in order to obtain a well defined notion. In fact, we require a more general convergence statement for a later application in the context of the Rankin–Selberg method and will state it here. Note first, that an integral of the form

$$\int_{\mathcal{F}_{\Gamma(1)}} f \, \mathrm{d}\mu$$

is absolutely convergent, as long as  $f \in L^{\infty}(\mathcal{F}_{\Gamma(1)})$ . In fact, it is easy to verify that the volume of  $\mathcal{F}_{\Gamma(1)}$  is finite, by computing the volume of an encompassing Siegel domain.

**Definition 3.51.** A function  $f : \mathbb{R} \to \mathbb{C}$  is called *rapidly decreasing* towards  $\infty$ , if for all  $n \in \mathbb{N}$  we have

$$f(x) = \mathcal{O}(x^{-n}), \qquad x \to \infty.$$
 (3.62)

A family of functions  $f_i$  is called rapidly decreasing, if all  $f_i$  are. It is called *uniformly* rapidly decreasing, if the constants in (3.62) may be chosen independently of the index *i*.

- *Example* 3.52. a) Let  $f : \mathbb{R} \to \mathbb{C}$  be rapidly decreasing towards  $\infty$  and  $p : \mathbb{R} \to \mathbb{C}$  be of at most polynomial growth. Then the product  $f \cdot p$  is rapidly decreasing.
  - b) Let  $f: \mathbb{H} \to \mathbb{C}$  have a convergent Fourier expansion of the form

$$f(\tau) = \sum_{0 < n \in \mathbb{Z}/N} a(n) e(n\tau)$$

for some natural number  $N \in \mathbb{N}$ . Then for  $\tau = u + iv \in \mathbb{H}$ , the family f(u + iv) is uniformly rapidly decreasing for  $v \to \infty$ .

- c) Let f, g be as in Definition 3.50. Then  $\langle f, g \rangle$  is uniformly (in  $\operatorname{Re}(\tau)$ ) rapidly decreasing towards  $\infty$ .
- d) Let L be a non-degenerate quadratic lattice that splits  $L = L_1 \oplus L_2$ . Given  $f \in S_{L,k}$ ,  $g \in \mathcal{M}_{L_1,k_1}$  and an Eisenstein series  $E_{L_2,\lambda,k_2}$  as in Definition 3.37 the product

$$\langle f, g \otimes E_{L_2,\lambda,k_2}(\,\cdot\,,s) \rangle$$
 (3.63)

is uniformly rapidly decreasing towards  $\infty$  for all fixed s for which  $E_{L_2,\lambda,k_2}$  does not have a pole. Further, the prefactor for the bound can be chosen locally uniformly in the parameter s (cf. Corollary 3.45).

In fact, part b) is verified by the following computation. Note first, that the Fourier series must then be absolutely convergent on  $\mathbb{H}$  by a standard argument for power series (cf. Lemma A.19). In fact, we find that for any  $\delta > 0$  the series is absolutely bounded on

the upper half plane  $\{\tau \mid \text{Im}(\tau) > \delta\}$  by

$$\sum_{0 < n \in \mathbb{Z}/N} |a(n)| \exp(-2\pi n\delta).$$
(3.64)

Consequently, we infer

$$\begin{split} |f(\tau)| &\leq \sum_{0 < n \in \mathbb{Z}/N} |a(n)| \exp(-2\pi nv) \\ &= \exp(-\pi v/N) \sum_{0 < n \in \mathbb{Z}/N} |a(n)| \exp(-\pi nv) \exp(-\pi (n-1/N)v) \\ &\leq \exp(-\pi v/N) \sum_{0 < n \in \mathbb{Z}/N} |a(n)| \exp(-\pi n\delta). \end{split}$$

Recall that the series on the right hand side equals a finite value, since the Fourier expansion of f is absolutely convergent at  $i\delta/2$ . Consequently, we find  $f(u+iv) = \mathcal{O}(\exp(-\pi v/N))$ for  $v \to \infty$  uniformly in u.

In order to verify that (3.63) is rapidly decreasing we restrict to the case Re(s) > 1 - k/2 first and note that

$$\begin{split} &|\langle f,g\otimes \sum_{\gamma\in\overline{\Gamma_{\infty}}\setminus\mathrm{Mp}_{2}(\mathbb{Z})}\left(\mathrm{Im}(\tau)^{s}\mathfrak{e}_{\lambda}\right)|_{L,k}\gamma\rangle|\\ &\leq \left|\sum_{\gamma\in\overline{\Gamma_{\infty}}\setminus\mathrm{Mp}_{2}(\mathbb{Z})}\mathrm{Im}(\tau)^{s}|_{k}\gamma\right|\cdot|\langle f,g\otimes\rho_{\mathcal{L}_{2}}(\gamma)^{-1}\mathfrak{e}_{\lambda}\rangle|. \end{split}$$

Observe that

$$|\langle f,g\otimes\rho_{\mathcal{L}_2}(\gamma)^{-1}\mathfrak{e}_{\lambda}\rangle|\leq \sum_{\mu\in\mathcal{L}_2}|\langle\mathfrak{e}_{\lambda},\rho_{\mathcal{L}_2}(\gamma)\mathfrak{e}_{\mu}\rangle|\cdot|\langle f,g\otimes\mathfrak{e}_{\mu}\rangle|.$$

Since  $\rho_{\mathcal{L}_2}$  is unitary, we find that the prefactor in the sum is universally bounded by 1. Further,  $\langle f, g \otimes \mathfrak{e}_{\mu} \rangle$  possesses a Fourier expansion as in part b), so that it remains to recall that

$$\sum_{\in \overline{\Gamma_{\infty}} \setminus \operatorname{Mp}_{2}(\mathbb{Z})} \operatorname{Im}(\tau)^{s}|_{k} \gamma = E_{k}(\tau, s)$$

 $\gamma$ 

is by Corollary 3.45 locally in s uniformly bounded in u by a single polynomial for  $v \to \infty$ . The concrete polynomial is given by  $v^{\max\{\operatorname{Re}(s), 1-k-\operatorname{Re}(s)\}}$ .

Deriving the statement for more general parameters  $s \in \mathbb{C}$  is more technical. The idea is

to use the Fourier expansion of  $E_{L,\lambda,k}(\tau, s)$  presented in Theorem 3.40 and note that it has meromorphic continuation (this is derived from the scalar valued case for the coordinate functions). Then, the *L*-series appearing in the Fourier expansion has meromorphic continuation and is independent of v, so that it may be locally uniformly bounded as long as it has no pole. Further, in order to bound the Whittaker function  $\mathcal{W}_s$ , we refer to Definition A.20 to recall its definition and to  $[Olv+10, (13.19.3)]^8$  in order to verify that the asymptotic behaviour  $\mathcal{W}_s(v) \sim e^{-1/2v}$  declared in [Olv+10, (13.14.21)] is true for a locally uniform bound in s. Consequently, only the constant term of the Fourier expansion plays a role which may again be asymptotically bounded by  $v^{\max\{\operatorname{Re}(s), 1-k-\operatorname{Re}(s)\}}$ . Note that part c) is a subcase of part b) and part a) is immediate.

By part c) and a) of the above Example, we conclude the following Corollary implying that the Petersson product in Definition 3.50 is well defined.

**Corollary 3.53.** Assume f, g to be chosen as in Definition 3.50 and let  $p : \mathbb{H} \to \mathbb{C}$  be a function of at most polynomial growth in  $\operatorname{Im}(\tau) \to \infty$ . Then the integral

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, g \rangle \cdot p \, \mathrm{d}\mu \tag{3.65}$$

converges absolutely.

Further, we obtain by part d) of Example 3.52 that the following Rankin–Selberg type integrals yield holomorphic functions.

**Corollary 3.54.** Let L be a non-degenerate quadratic lattice that splits  $L = L_1 \oplus L_2$ . Given  $f \in S_{\mathcal{L},k}$ ,  $g \in \mathcal{M}_{\mathcal{L}_1,k_1}$  and an Eisenstein series  $E_{L_2,\lambda,k_2}$  as in Definition 3.37, the integral

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, g \otimes E_{L_2,\lambda,k_2}(\,\cdot\,,s) \rangle \cdot v^k \,\mathrm{d}\mu \tag{3.66}$$

converges absolutely for  $\operatorname{Re}(s) > 1 - k/2$ . In fact, if  $E_{L_2,\lambda,k}$  has no pole at some  $s \in \mathbb{C}$  the convergence is locally uniformly, so that the integral in (3.66) defines a holomorphic function in s.

#### The case of higher genus

Many of the ideas that have been presented above for the  $SL_2$  case may be reproduced in case of genus g > 1. There is, for instance, also a metaplectic extension which at the

<sup>&</sup>lt;sup>8</sup>The reader may also consult [Olv+10, (5.2.4)] for the definition of the appearing *Pochhammer* symbol.

infinite place may also be realised by pairs

$$\left(\begin{pmatrix}A & B\\ C & D\end{pmatrix}, \phi(Z)\right),$$

where  $\phi$  is a holomorphic square root of (CZ + D). Its integral points  $\operatorname{Mp}_{2g}(\mathbb{Z})$  act via the discrete Weil representation on  $\mathbb{C}[\mathcal{L}^g]$ , where  $\mathcal{L} = L'/L$  is the discriminant form as before. For  $\underline{\lambda}, \underline{\mu} \in \mathcal{L}^g$  we understand  $\operatorname{b}(\underline{\mu}, \underline{\lambda}) = \sum_{j=1}^g \operatorname{b}(\mu_j, \lambda_j)$ . Then the action of  $\rho_L$  is given by the following formulae.

a) For  $U \in \operatorname{GL}_q(\mathbb{Z})$  we find

$$\rho_L\left(M(U),\sqrt{\det(U)}\right)\mathfrak{e}_{\underline{\lambda}} = \sqrt{\det(U)}^{m^- - m^+}\mathfrak{e}_{U^{-1}\underline{\lambda}}.$$

b) For  $S \in M_g(\mathbb{Z})$  symmetric we have

$$\rho_L\left(N(S),1\right)\mathfrak{e}_{\underline{\lambda}} = e[\operatorname{tr}(\overline{\mathbf{q}}(\lambda)S)]\mathfrak{e}_{\underline{\lambda}}.$$

c) For J the standard alternating matrix we find

$$\rho_L\left(-J,\sqrt{\det(Z)}\right)\mathfrak{e}_{\underline{\lambda}} = \frac{e[-g\frac{(m^+-m^-)}{8}]}{|L'/L|^{g/2}}\sum_{\underline{\mu}\in\mathcal{L}^g}e[-\operatorname{b}(\underline{\mu},\underline{\lambda})]\mathfrak{e}_{\underline{\mu}}.$$

We note that a holomorphic function

$$f:\mathbb{H}_g\to\mathbb{C}[\mathcal{L}^g],f(Z)=\sum_{\underline{\lambda}\in\mathcal{L}^g}f_{\underline{\lambda}}\mathfrak{e}_{\underline{\lambda}}$$

transforming like a modular form for the Weil representation  $\rho_L$  fulfils

$$f|_{L,k}(N(S),1) = \sum_{\underline{\lambda} \in \mathcal{L}^g} e[-\operatorname{tr}(\overline{\mathbf{q}}(\underline{\lambda})S)] f_{\underline{\lambda}} \mathfrak{e}_{\underline{\lambda}},$$

meaning  $e[-\operatorname{tr}(\overline{q}(\underline{\lambda})(Z+S))]f_{\underline{\lambda}}(Z) = e[-\operatorname{tr}(\overline{q}(\underline{\lambda})Z)]f_{\underline{\lambda}}(Z)$ , i.e. it is periodic and hence possesses a Fourier expansion just as in the case of a trivial representation (cf. (3.8)).

For additional details, the reader may consult [KRY06] and [Zha09].

## 3.3.3 Indices of Fourier coefficients

In this subsection we lay the foundation for investigations of Fourier coefficients in the context of examining the injectivity of the Kudla–Millson lift in Section 7.3. In case the base lattice in this section does not split a hyperbolic plane, partial orders will play a key role whose relevance arises from the action of Hecke algebras on Fourier coefficients of the vector valued modular forms involved. In fact, these partial orders are induced by the action of monoids. Consequently, we begin our investigation with introducing monoidal preorders before defining crucial notions for the investigation of indices of Fourier coefficients of automorphic forms.

#### Monoid preorders

**Definition 3.55.** Let  $(M, \cdot)$  be a monoid and S be a set.

- a) We say that  $(M, \cdot)$  acts on S, written  $M \odot S$ , if there is a monoid-morphism  $\sigma : (M, \cdot) \to \operatorname{Aut}(S)$ . For  $m \in M$  and  $s \in S$  write  $m \cdot s := \sigma(m)(s)$ .
- b) For an action  $M \bigcirc S$  define a preorder on S via

$$s_1 \preceq s_2 \iff \exists m \in M : s_2 = m \cdot s_1.$$

The following examples are meant to build some intuition and also lay the foundation for the key instance we have in mind.

- *Example* 3.56. a) For the monoid  $(\mathbb{N}_0, +)$  and the set  $\mathbb{N}_0$ , where the action is given by addition, we find that  $\leq$  equals the usual order on  $\mathbb{N}_0$ .
  - b) Let  $(\mathbb{N}_0, +)$  be the monoid and  $S = \mathbb{Q}^{\times}$ . Pick a prime p and a natural number  $k \in \mathbb{N}$ and let  $\mathbb{N}_0$  act via  $q \mapsto n \cdot q \coloneqq p^{nk} \cdot q$ . Then  $q_1 \preceq q_2$  encodes that  $q_2$  is a multiple of  $q_1$  by a power of  $p^k$ .
  - c) Let  $(\mathbb{N}, \cdot)$  be the monoid and  $S = \mathbb{Q}^{\times}$ . Then  $\mathbb{N} \oplus \mathbb{Q}^{\times}$  via multiplication and the relation  $\leq$  detects natural multiples.
  - d) As a refinement of the above, let  $P \subseteq \mathbb{P}$  be a set of primes and for every  $p \in P$  select  $k_p \in \mathbb{Z}$ . Construct the monoid  $\bigoplus_{p \in P} (\mathbb{N}_0, +)$  and  $S = \mathbb{Q}^{\times}$  with action

$$q\mapsto (n_p)_p\cdot q\coloneqq \left(\prod_{p\in P}p^{n_pk_p}\right)\cdot q$$

where multiplication on the right denotes multiplication in  $\mathbb{Q}^{\times}$ . Note that the choice  $P = \{p\}$  is b), the choice  $P = \mathbb{P}$  and  $k_p = 1$  for all  $p \in \mathbb{P}$  equals part c). Unique factorisation domains may be treated similarly.

The notion of a preorder is too weak for the application in later stages of this thesis. However, the desired property of antisymmetry may be characterised in terms of the monoid action.

**Lemma 3.57.** For the preorder  $\leq$  induced by  $(M, \cdot) \ominus S$  the following are equivalent:

- a) The preorder  $\leq$  is a partial order (i.e. antisymmetric).
- b) For all  $m_1, m_2 \in M$  and all  $s \in S$  we find that  $m_2m_1 \cdot s = s$  already implies  $m_1 \cdot s = s$ .

*Proof*: Assume there are  $m_1, m_2 \in M$  and some  $s \in S$  such that  $m_2m_1 \cdot s = s$ , in other words  $s \leq m_1 \cdot s \leq m_2m_1 \cdot s = s$ . If we assume antisymmetry, we obtain  $m_1 \cdot s = s$ , settling the first implication.

On the other hand, assuming b) and given  $s_1 \leq s_2$  and  $s_2 \leq s_1$ , we spell that out as there are  $m_1, m_2 \in M$  such that  $s_2 = m_1 s_1$  and  $s_1 = m_2 s_2 = m_2 m_1 s_1$ . By assumption we find  $s_2 = m_1 s_1 = s_1$ .

*Example* 3.58. All subexamples of Example 3.56 define partial orders since there is no unit other than the neutral element and no other element acts with a fixed point.

This prompts us to define the following notion.

**Definition 3.59.** A monoid action  $M \odot S$  is called *faithful* if only the neutral element  $e \in M$  acts as the identity on S. It is further called *free* if  $e \in M$  is the only element fixing a point in S.

**Remark 3.60.** A monoid action  $M \odot S$  that is free induces a partial order. Further, every free action is faithful.

Clearly, we would like to apply the notion of monoidal preorders to discriminant forms and consider the following example as a prototype.

Example 3.61. Let (L, q) be an even lattice of level  $N \in \mathbb{N}$ . With the notation of Example 3.56 part d), we select Q to be a subset of the prime divisors of N, choose  $S = \bigoplus_{q \in Q} \mathcal{L}_q = \bigoplus_{q \in Q} \mathcal{L}'_q / \mathcal{L}_q$ , and declare a similar action

$$\lambda \mapsto (n_p)_p \cdot \lambda \coloneqq \left(\prod_{p \in P} p^{n_p k_p}\right) \cdot \lambda.$$

We remark that the associated preorder is almost never antisymmetric. In fact, let  $2 \neq q \in Q$  and assume there is  $p \in P$  with  $p \neq q$ . Then, multiplication by p acts as an automorphism on  $\mathcal{L}_q$  and there must be some number  $r_p$  such that multiplication by  $p^{r_p}$  acts as the identity on  $\mathcal{L}_q$ . In consequence, select some non-trivial  $\lambda \in \mathcal{L}_q \leq S$  and verify that

$$(p^{r_p-1} \cdot p^1) \cdot \lambda = \lambda \qquad \land \qquad p^1 \cdot \lambda \neq \lambda. \tag{3.67}$$

However, (3.67) contradicts part b) of Lemma 3.57. In case Q holds only one element and  $P \subseteq Q$ , the preorder will be antisymmetric.

When considering indices of Fourier coefficients of vector valued automorphic forms, however, not only the associated discriminant form  $\mathcal{L}$  plays a role but pairs taken from  $\mathcal{L} \times \mathbb{Q}$ . Hence, it is natural to ask how to combine actions onto products of sets.

**Definition 3.62.** Let  $(M_i, \cdot)$  be monoids and  $(S_i)$  be sets with an  $M_i$  monoid action  $\sigma_i$ .

a) Then we call

$$\sigma: \prod_{i} (M_{i}, \cdot) \to \operatorname{Aut}\left(\prod_{i} S_{i}\right), \qquad (m_{i}) \cdot (s_{i}) \coloneqq (\sigma_{i}(m_{i})s_{i}) \in \prod_{i} S_{i}$$

the product action of the  $\sigma_i$ .

b) In case there is a monoid  $(M, \cdot)$  that embeds into each  $(M_i, \cdot)$ , the induced action via a diagonal embedding into the product

$$\sigma: M \to \operatorname{Aut}\left(\prod_{i} S_{i}\right), \qquad m \cdot (s_{i}) \coloneqq (\sigma_{i}(m)s_{i}) \in \prod_{i} S_{i}$$

is called the *diagonal action* of  $(M, \cdot)$  via the family  $(\sigma_i)$ .

**Remark 3.63.** If in the above definition for a diagonal action one of the  $\sigma_i$  is faithful, so is the product of the family  $(\sigma_i)$ . Then, however, it induces a partial order.

The following is the prime example for the application in this thesis.

Example 3.64. Let  $M = (\mathbb{N}, \cdot)$  and (L, q) be an even lattice. Then  $\mathbb{N}$  acts naturally on  $\mathcal{L} = L'/L$  via multiplication and on  $\mathbb{Q}^{\times}$  via multiplication with squares. The resulting diagonal action of the product is given by letting  $n \in \mathbb{N}$  act by

$$n \cdot (\lambda, q) = (n\lambda, n^2 q) \in \mathcal{L} \times \mathbb{Q}^{\times}.$$
(3.68)

Refine the above by selecting some  $N \in \mathbb{N}$  and replacing M by the submonoid of  $(\mathbb{N}, \cdot)$  generated by prime numbers that are (co)prime to N. The action defines a partial order.

**Remark 3.65.** Example 3.64 may be realised in the following fashion. Select  $P \subseteq \mathbb{P}$  to be the collection of primes coprime to N, consider Example 3.56 d) with  $k_p = 2$  and Example 3.61 with  $k_p = 1$  and Q representing all prime divisors of lev(L). Then taking the diagonal action of  $\bigoplus_{p \in P}(\mathbb{N}_0, +)$  via the actions described above yields the desired result. The advantage of decomposing along primes becomes apparent in Subsection 6.4.2.

The following definition subsumes partial orders generated by the cases of Example 3.64 that are crucial for investigating the action of Hecke operators on Fourier coefficients of vector valued modular forms for the Weil representation.

**Definition 3.66.** Let  $N \in \mathbb{N}$  and define  $\preceq_N (\preceq^N)$  to be the partial order on  $\mathcal{L} \times \mathbb{Q}^{\times}$ induced by the monoid action from Example 3.64 with the monoid  $M \subseteq (\mathbb{N}, \cdot)$  being generated by primes dividing (being coprime to) N.

Explicitly, this means that for  $s_0, s_f \in S = \mathcal{L} \times \mathbb{Q}^{\times}$  and some prime  $p \in \mathbb{P}$  we define

$$s_{0} \leq_{p} s_{f} : \iff \text{ there is some } k \in \mathbb{N} \text{ such that } p^{k} \cdot s_{0} = s_{f},$$
  

$$\text{there is a finite set of primes } \{p_{i} \in \mathbb{P} : 0 \leq i \leq k, p_{i} \mid N\}$$
  

$$s_{0} \leq_{N} s_{f} : \iff \text{ and corresponding elements } s_{i} \in S \text{ such that}$$
  

$$s_{0} \leq_{p_{0}} s_{1} \leq_{p_{1}} \cdots \leq_{p_{k}} s_{k} = s_{f},$$
  

$$\text{there is a finite set of primes } \{p_{i} \in \mathbb{P} : 0 \leq i \leq k, p_{i} \nmid N\}$$
  

$$s_{0} \leq^{N} s_{f} : \iff \text{ and corresponding elements } s_{i} \in S \text{ such that}$$
  

$$s_{0} \leq_{p_{0}} s_{1} \leq_{p_{1}} \cdots \leq_{p_{k}} s_{k} = s_{f}.$$

The corresponding strict partial orders are denoted  $\prec_N (\prec^N)$ .

### **Indices of Fourier coefficients**

First, we will have a look at indices of Fourier coefficients of vector valued modular forms associated to an even quadratic lattice (L, q) of signature  $(m^+, m^-)$ . Recall that the possible indices lie in the set  $\mathcal{L} \times \mathbb{Q}$ , where  $\mathcal{L} = L'/L$  denotes the discriminant form.

**Definition 3.67.** For an even quadratic lattice (L, q) of signature  $(m^+, m^-)$ , define the set of *admissible* indices with respect to (L, q) as

$$\mathscr{I} := \{ (\lambda, n) \in \mathcal{L} \times \mathbb{Q} \mid \overline{q}(\lambda) \equiv n \}.$$

Further, define the set of *representable* indices with respect to (L, q) as

$$\mathscr{R} \coloneqq \left\{ (\lambda, n) \in \mathcal{L} \times \mathbb{Q} \mid \exists \ell \in L' : (\lambda, n) = (\overline{\ell}, q(\ell)) \right\}$$

Here,  $\overline{\ell}$  denotes the natural projection of  $\ell$  to  $\mathcal{L}$ . Define the subsets

$$\mathscr{I}^{\times} \coloneqq \mathscr{I} \cap \mathcal{L} \times \mathbb{Q}^{\times} \subseteq \mathscr{I}, \qquad \mathscr{R}^{\times} \coloneqq \mathscr{R} \cap \mathscr{I}^{\times} \subseteq \mathscr{R},$$

with non-vanishing second component. The subsets with positive/negative second component are denoted  $\mathscr{I}^{\pm}, \mathscr{R}^{\pm}$ . Further, we require the *primitively representable* pairs

$$\mathscr{R}_1 \coloneqq \left\{ (\lambda, n) \in \mathcal{L} \times \mathbb{Q} \mid \exists \ell \in L' \text{ primitive } : (\lambda, n) = (\overline{\ell}, q(\ell)) \right\}.$$

With these notions, we may formulate the following key lemma for prooving the injectivity of the Kudla–Millson lift under the assumption of a hyperbolic split.

**Lemma 3.68.** In case  $L \simeq K \oplus H$  splits a hyperbolic plane we find  $\mathscr{R}_1 = \mathscr{R} = \mathscr{I}$ , i.e. any possible index  $(\lambda, n)$  may be written as  $(\overline{\ell_0}, q(\ell_0))$  for some primitive element  $\ell_0 \in L'$ .

Proof: Let  $(\lambda, n) \in \mathscr{I}$ , choose a representative  $\ell \in L'$  with  $\overline{\ell} = \lambda$ , meaning  $q(\ell) \equiv n \mod \mathbb{Z}$ , and note that the class  $\overline{\ell} \in \mathcal{L}$  is invariant under translating  $\ell$  by elements in H. Since  $L = K \oplus H$  splits a hyperbolic plane  $H = \langle e_1, e_2 \rangle_{\mathbb{Z}}$ , where  $e_1, e_2$  denote the standard generators with  $q(e_1) = q(e_2) = 1$  and  $b(e_1, e_2) = 1$ , we find

$$q(\ell) = q(\ell|_K) + q(\ell|_H)$$

and for all  $m_1, m_2 \in \mathbb{Z}$  that

$$q(m_1e_1 + m_2e_2) = m_1m_2$$

so that the primitive vector

$$\ell_{\lambda} \coloneqq \ell|_{K} \oplus [(n - q(\ell|_{K}))e_{1} + e_{2}]$$

fulfils the desired property  $(\overline{\ell_{\lambda}}, \mathbf{q}(\ell_{\lambda})) = (\lambda, n)$ . In particular,  $(\lambda, n) \in \mathscr{R}_1 \subseteq \mathscr{R} \subseteq \mathscr{I}$ .  $\Box$ 

**Remark 3.69.** The action of Example 3.64 on  $\mathcal{L} \times \mathbb{Q}^{\times}$  and the corresponding partial orders from Definition 3.66 restrict to  $\mathscr{I}^{\times}$  and  $\mathscr{R}^{\times}$ . They may, in the same spirit be

extended to  $\mathcal{L} \times \mathbb{Q}$  and then restricted to  $\mathscr{I}$  and  $\mathscr{R}$ , if antisymmetry is given up (also cf. Example 3.61).

The monoid action of Example 3.64 on  $\mathcal{L} \times \mathbb{Q}$  may be partly reversed for good primes. **Definition 3.70.** For a pair  $(\lambda, n) \in \mathcal{L} \times \mathbb{Q}$  and  $k \in \mathbb{N}$  with gcd(k, lev(L)) = 1 define

$$(\lambda, n)/k \coloneqq (\lambda/k, n/k^2).$$

Note that in the notation above and for natural numbers  $k_1, k_2 \in \mathbb{N}$  satisfying  $gcd(k_1, lev(L)) = gcd(k_2, lev(L)) = 1$  we find that

$$[(\lambda, n)/k_1]/k_2 = (\lambda, n)/(k_1k_2) = [(\lambda, n)/k_2]/k_1.$$

Next assume we have an element  $(\lambda, n) \in \mathscr{I}^{\times}$ . Then, necessarily  $n \in \mathbb{Z}/\operatorname{lev}(L)$  so that for  $k \in \mathbb{N}$  with  $\operatorname{gcd}(k, \operatorname{lev}(L)) = 1$ , we conclude  $(\lambda, n)/k \notin \mathscr{I}^{\times}$ , unless  $n/k^2 \in \mathbb{Z}/\operatorname{lev}(L)$ . So there must be elements in  $\mathscr{I}$  which are minimal with respect to division.

**Definition 3.71.** Let  $N \in \mathbb{N}$ .

a) The pair  $(\lambda, n) \in \mathscr{I}^{\times}$  (respectively contained in  $\mathscr{R}^{\times}$ ) is called *N*-primitive, if there is no  $(\mu, m) \in \mathscr{I}^{\times}$  (respectively contained in  $\mathscr{R}^{\times}$ ) with

$$(\mu, m) \prec^N (\lambda, n).$$

b) A number  $n \in \mathbb{Z}/N$  is called *N*-square free (also *N*- $\mathbb{Z}$ ), if for any prime  $p \nmid N$  we find  $n/p^2 \notin \mathbb{Z}/N$ .

In case of good primes, N-primitive indices may be classified by their second component as follows.

**Lemma 3.72.** For  $N \in \mathbb{N}$  with  $lev(L) \mid N$  and  $p \in \mathbb{P}$  the following are true.

- a) A pair  $(\lambda, n) \in \mathscr{R}^{\times}$  is N-primitive if, and only if, n is N-square free.
- b) If  $(\lambda, n) \in \mathscr{I}^{\times}$  with  $p \nmid N, p^2 \mid n$ , and  $(\lambda, n)$  only differs (additively in the second component) from an element in  $\mathscr{R}$  by  $(0, p^2 \cdot z) \in \mathcal{L} \times p^2 \mathbb{Z}$ , we find that it cannot be *N*-primitive.

*Proof*: Clearly, if n is N-square free, there is nothing to show.

For the other direction write  $(\lambda, n) = (\lambda, q(\ell_{\lambda}))$  with  $\ell_{\lambda} \in L'$  projecting to  $\lambda$ . Assume

there was some prime  $p \nmid N$  such that  $p^2 \mid \text{num}(q(\ell_{\lambda}))$  (which is equivalent to  $p^2 \mid q(\ell_{\lambda})$ , since  $p \nmid \text{lev}(L)$  and  $\text{den}(q(\ell_{\lambda})) \mid \text{lev}(L)$ ).

We have  $\mu \coloneqq \lambda/p \iff p\mu = \lambda$ , so that there must be some  $\ell_{\mu} \in L'$  projecting to  $\mu$ . This means  $\ell_{\lambda} = p\ell_{\mu} + l$  for some  $l \in L$ . Hence,

$$q(\ell_{\lambda}) = p^2 q(\ell_{\mu}) + p b(\ell_{\mu}, l) + q(l)$$
(\*)

but since  $p^2 | q(\ell_{\lambda})$ , so does  $p^2 | q(\ell_{\lambda}) - p^2 q(\ell_{\mu}) \in \mathbb{Z}$ , meaning  $p b(\ell_{\mu}, l) + q(l) = p^2 \cdot r$  for some  $r \in \mathbb{Z}$ . Hence, the numerator on the right side of (\*) is divisible by  $p^2$  so that

$$q(\ell_{\lambda})/p^2 = q(\ell_{\mu}) + r.$$

This, however, means nothing else than

$$q(\ell_{\lambda})/p^2 \equiv \overline{q}(\mu) = \overline{q}(\lambda/p)$$

and hence,  $(\lambda/p, q(\ell_{\lambda})/p^2)$  was admissible, so that  $(\lambda, n)$  could not be primitive. Now, in case  $n = q(\ell_{\lambda}) + p^2 \cdot z$  with  $z \in \mathbb{Z}$ , we find  $p^2 | q(\ell_{\lambda})$  and the same argument as above works.

**Lemma 3.73.** Assume there are  $\lambda, \sigma, \mu \in \mathcal{L}$  and  $m, n \in \mathbb{Z}$  with gcd(n, m) = 1 as well as  $n\lambda = \sigma = m\mu$ . Then there is  $\delta \in \mathcal{L}$  such that  $m\delta = \lambda$  and  $n\delta = \mu$ . In other words



Proof: For the notation compare Definition 1.35. We begin with the left diagram and note that  $\sigma \in {}^{n}\mathcal{L} \cap {}^{m}\mathcal{L}$ . However, multiplication by m acts as an automorphism on  $\mathcal{L}_{p}$  for all primes p with  $p \nmid m$ , so that  $\mu \in {}^{n}\mathcal{L}$ . In other words, there is  $\delta_{\mu} \in \mathcal{L}$  with  $n\delta_{\mu} = \mu$ . Now by assumption  $nm\delta_{\mu} = mn\delta_{\mu} = m\mu = \sigma = n\lambda$ , meaning  $\lambda - m\delta_{\mu} \in {}_{n}\mathcal{L}$ . However, by symmetry of the diagram, we also find  $\lambda \in {}^{m}\mathcal{L}$ , so that  $\lambda - m\delta_{\mu} \in {}_{n}\mathcal{L} \cap {}^{m}\mathcal{L}$ . This means, there is an m-th root of  $\lambda - m\delta_{\mu}$  which we will call  $\delta'$  and which lies in  ${}_{n}\mathcal{L}$ , since multiplication by m acts as an automorphism on this submodule. In total, we find for  $\delta \coloneqq \delta_{\mu} + \delta'$  that

$$n\delta = n\delta_{\mu} + n\delta' = n\delta_{\mu} = \mu$$
, and  
 $m\delta = m\delta_{\mu} + m\delta' = m\delta + \lambda - m\delta = \lambda$ .

We will have to explain the purpose of these investigations which originate from an approach to proving the injectivity of the Kudla–Millson theta lift.

For a lattice of signature  $(m^+, 2)$  with level N and a modular form  $f \in \mathcal{M}_{L,k}$  we find that  $\mathscr{I}$  is the set of possible indices of Fourier coefficients of f. If f is further a cusp form, these lie in  $\mathscr{I}^+$  which decomposes into maximal cones with respect to the partial order  $\prec^N$ . For an index  $i \in \mathscr{I}$  we speak of the value of i on f when referring to the Fourier coefficient with index i.

By Proposition 6.54 we find that for a prime  $p \nmid N$  the Hecke operator  $\mathcal{T}(p^2)$  relates the values of indices on f which are linked by  $\prec_p$ . If f is an eigenform, the value of some index in  $\mathscr{I}$  may actually be reduced to the value of a minimal index with respect to  $\prec_p$ , in many cases. In fact, the same is true for the partial order  $\prec^N$ .

The role of the representable indices  $\mathscr{R}$  is the following. Recall that a major goal of this thesis is to prove the injectivity of the Kudla–Millson lift. For that purpose we consider cycle integrals of liftings. In Subsection 7.3.1 we learn that coefficients with indices in  $\mathscr{R}$  are exactly these that appear in descriptions of the aforementioned cycle integrals (cf. Theorem 7.14). The same is true for different descriptions of the Fourier expansion of Kudla–Millson liftings appearing in the literature (cf. [Bru14], [Zuf24], and [MZ23]). Hence, the set  $\mathscr{R}$  contains the indices that play a role for the Kudla–Millson lift. The purpose of the primitively representable indices  $\mathscr{R}_1$  is then the following. The values of these indices represent the first terms of certain L-series associated to f which are introduced in Subsection 6.4.1. These series are shown to admit roots if the form f is annihilated by the Kudla-Millson lift. Now if these indices happen to be minimal with respect to  $\prec^N$ , their value on f can be shown to vanish if f is an eigenform for a certain Hecke algebra. Then, however, all indices related to it by  $\prec^N$  have to vanish on f. In many cases, this suffices to prove that f has to vanish, if its Kudla–Millson lift does. By employing a Hecke equivariance result for theta lifts [Yos85], the assumption of an eigenform may be eliminated, implying the injectivity of the Kudla–Millson theta lift.

However, for now, results obtained by this procedure are less general than Theorem 7.16.

## 3.3.4 Bounds on Fourier coefficients

Recall that a vector valued modular form  $f \in \mathcal{M}_{L,k}$  has a Fourier expansion of the form

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot q^n \mathfrak{e}_{\lambda}.$$

Typically these coefficients encode arithmetic information, in the case of theta series representation numbers of lattices, for instance. This alone justifies the desire for asymptotic bounds of these coefficients, as  $n \to \infty$ . In our case, in Section 6.4, we are associating *L*-series to vector valued modular forms and exploit that the asymptotic behaviour of the coefficients  $a(\lambda, n)$  determines the range of convergence of these series.

In the following, we will collect information on this asymptotic behaviour. In fact, there is an abundance of bounds for scalar valued modular forms available and one could hope to lift these results to the vector valued case. Clearly, the Weil representation intertwines the different components of f, so directly applying bounds for scalar valued modular forms for Mp<sub>2</sub>( $\mathbb{Z}$ ) cannot work and we will have to do further investigations. Along these lines, we attempt to understand the behaviour of the discrete Weil representation  $\rho_L$  better with respect to subgroups of Mp<sub>2</sub>( $\mathbb{Z}$ ) and state the following result.

**Proposition 3.74** ([Bor00, Thm. 5.4]). Suppose that  $\mathcal{L}$  is a discriminant form of level dividing N. If b and c are divisible by N, then  $\gamma = \left( \begin{pmatrix} a & b \\ c & d \end{pmatrix}, \sqrt{c\tau + d} \right) \in \operatorname{Mp}_2(\mathbb{Z})$  acts on the Weil representation by

$$\rho_L(g)\mathfrak{e}_{\lambda} = \chi_{\mathcal{L}}(g)\mathfrak{e}_{d\lambda}$$

where  $\chi_{\mathcal{L}}$  denotes the character of  $\overline{\Gamma}_0(N)$  defined in [Bor00, Thm. 5.4].

However, in case of  $4 \nmid N$ , the character  $\chi_{\mathcal{L}}$  is trivial on the preimage of  $\Gamma(N)$  in  $\operatorname{Mp}_2(\mathbb{Z})$ . In the other case, meaning  $4 \mid N$ , the section  $s(\Gamma(N))$  in  $\operatorname{Mp}_2(\mathbb{Z})$  is contained in the kernel of the character (cf. [Bor00, Lem. 5.3 p. 329]). Hence, the component functions  $f_{\lambda}$  of a vector valued modular form are actually contained in  $\mathcal{M}_k(\Gamma(\operatorname{lev}(L)))$ .

As a consequence, for the purpose of bounding the coefficients of the vector valued modular form f, it suffices to bound the asymptotic behaviour of scalar valued modular forms for  $\Gamma(N)$ . There is the following standard bound available in the literature (cf. [Shi71, Lem. 3.62 p. 90], [Shi73, p. 447]).

**Lemma 3.75.** Let  $k \in \mathbb{Z}/2$  and  $N \in \mathbb{N}$  with  $4 \mid N$  if k is not integral. Suppose that

 $f \in \mathcal{S}_k(\Gamma(N))$  has Fourier expansion

$$f(\tau) = \sum_{n \in \mathbb{Z}} a(n) q^{n/N}$$

Then

$$a(n) \in \mathcal{O}(n^{k/2})$$
 for  $n \to \infty$ 

Also compare [JLL18] deriving that if the component functions of a vector valued modular form satisfy this bound, then it is a cusp form. Combining this bound with the bounds on coefficients of Eisenstein series yields the following result.

**Corollary 3.76.** Let  $f \in \mathcal{M}_k(\Gamma(N))$  with Fourier expansion as above. Then

$$a(n) \in \mathcal{O}(n^{k-1}).$$

In many cases there are sharper bounds on the asymptotic behaviour known, for instance, the bound  $\mathcal{O}(n^{k/2-1/5})$  by Rankin in the even case (cf. Theorem 2.25). However, even sharper bounds are to be found in the literature for forms to larger congruence subgroups. Consequently, the idea is to reduce the cases relevant to us to the setting of these bigger congruence subgroups.

In fact, further analysis of the situation yields that the component functions  $f_{\lambda}$  for  $\lambda \in \mathcal{L}$  may indeed be described by scalar valued modular forms for  $\Gamma_1(\text{lev}(L))$  with character. Recall that  $\Gamma(\text{lev}(L)) \triangleleft \Gamma_1(\text{lev}(L))$  with abelian quotient, which remains true under the section s so that by Remark 6.8, we may reduce to the case of a modular form for  $\Gamma_1(\text{lev}(L))$  with character.

In a subsequent step, this information may be utilised to reduce to the even simpler case of  $\Gamma_0$  modular forms. However, First note that a scalar valued modular form f that has been transferred by the Petersson operator still possesses a Fourier expansion. Hence, it is meaningful to speak of a Fourier expansion of forms that have been altered by such a procedure. The proof idea of the following lemma is found in [DS05, 1.2.11 p. 24].

**Lemma 3.77.** Let  $\Gamma$  be a congruence subgroup,  $k \in \mathbb{Z}/2$ ,  $f \in \mathcal{M}_k(\Gamma)$ , and select  $\gamma \in \mathrm{GL}_2^+(\mathbb{Q})$ . Then  $f|_k\gamma$  has a Fourier expansion and its constant term vanishes if f is a cusp form.

Proof: Assuming  $\gamma \in \mathrm{Mp}_2(\mathbb{Z}), f|_k \gamma$  must have a Fourier expansion. In fact, recall that there is  $N \in \mathbb{N}$  such that  $\Gamma(N) \leq \Gamma$ . However,  $s(\Gamma(N)) \leq \mathrm{Mp}_2(\mathbb{Z})$  is normal. Hence,  $f|_k \gamma$  is N periodic and holomorphic at  $\infty$ , so that it possesses a Fourier expansion.

In the general case,  $\gamma$  may be written as  $\gamma = \alpha \gamma m a'$  where  $\alpha \in Mp_2(\mathbb{Z})$  and  $\gamma' = r \cdot \left( \begin{pmatrix} a & b \\ 0 & d \end{pmatrix}, \sqrt{d} \right)$  with  $r \in \mathbb{Q}^+$  and gcd(a, b, d) = 1. Then also  $f|_k \gamma$  must have a Fourier expansion.

With this tool at hand, we may reduce to the case of a modular form for  $\Gamma_1(N)$  without character which, in turn, reduce to modular forms for  $\Gamma_0(N)$  with character factoring through  $\Gamma_1(N)$ .

**Lemma 3.78.** Let  $N \in \mathbb{N}$ ,  $k \in \mathbb{Z}/2$ ,  $\psi : \Gamma_1(N) \to \mathbb{T}$  be a character that is trivial on  $\Gamma(N)$ , and

$$f = \sum_{n \in \mathbb{N}/N} a(n)q^n \in \mathcal{S}_k(\Gamma_1(N), \psi).$$

Assume that there is some  $\alpha \in \mathbb{R}$  such that for all characters  $\chi : \Gamma_0(N) \to \mathbb{T}$  that are trivial on  $\Gamma_1(N)$  and any choice  $g = \sum_{n \in \mathbb{N}} b(n)q^n \in \mathcal{S}_k(\Gamma_0(N^2), \chi)$  we have  $b(n) \in \mathcal{O}(n^{\alpha})$ . Then also

$$a(n) \in \mathcal{O}(n^{\alpha}).$$

**Remark 3.79.** The statement above, meaning the transferal of bounds, remains true if the space of cusp forms S is replaced by holomorphic modular forms  $\mathcal{M}$ , weakly holomorphic modular forms  $\mathcal{M}^!$  or meromorphic modular forms  $\mathcal{A}$ .

Proof: Assume f to be as above and  $k \notin \mathbb{Z}$  implying  $4 \mid N$  (the case of  $k \in \mathbb{Z}$  is similar but easier to prove). We will prove that  $f(N \cdot)$  is a modular form without character for  $\Gamma_1(N^2)$ . Recall that there is a section  $s : \Gamma_0(N) \to \overline{\operatorname{SL}}_2(\mathbb{Z})$  given by (3.38) which is implicitly used to define modular forms of half integral weight and may be utilised to pull back some computations to  $\Gamma_0(N)$ . Let  $\gamma = \left( \begin{pmatrix} a & b \\ c & d \end{pmatrix}, \phi(\tau) \right) \in \overline{\Gamma}_1(N^2)$  and compute

$$\begin{pmatrix} \begin{pmatrix} N & 0 \\ 0 & 1 \end{pmatrix}, 1 \end{pmatrix} \cdot \begin{pmatrix} \begin{pmatrix} a & b \\ c & d \end{pmatrix}, \phi(\tau) \end{pmatrix} \cdot \begin{pmatrix} \begin{pmatrix} 1/N & 0 \\ 0 & 1 \end{pmatrix}, 1 \end{pmatrix}$$
$$= \begin{pmatrix} \begin{pmatrix} N & 0 \\ 0 & 1 \end{pmatrix}, 1 \end{pmatrix} \cdot \begin{pmatrix} \begin{pmatrix} a/N & b \\ c/N & d \end{pmatrix}, \phi(\tau/N) \end{pmatrix}$$
$$= \begin{pmatrix} \begin{pmatrix} a & Nb \\ c/N & d \end{pmatrix}, \phi(\tau/N) \end{pmatrix} \in \overline{\Gamma}_1(N).$$

Now, without loss of generality, we may pass to the case of N being a square and  $3 \mid \nu_2(N)$ . Under this assumption, we infer from (3.38) that if  $\gamma \in s(\Gamma_1(N^2))$  then also the

conjugate above is contained in  $s(\Gamma_1(N))$ . In fact, it even lies in  $s(\Gamma(N))$ . Next, define  $M_N \coloneqq \left( \begin{pmatrix} N & 0 \\ 0 & 1 \end{pmatrix}, 1 \right) \in \overline{\operatorname{GL}_2^+}(\mathbb{Q})$  and conclude that for any  $\gamma \in s(\Gamma_1(N^2))$  we find that

$$f(M_N \gamma \tau) = f(M_N \gamma M_N^{-1} M_N \tau)$$
  
=  $\psi(M_N \gamma M_N^{-1}) f(M_N \tau) \phi(\tau)^k$ 

Recall that by assumption  $\psi$  was trivial on  $s(\Gamma(N))$ . As a result, we have that

$$f(N\tau) = \sum_{n \in \mathbb{N}/N} a(n)q^{Nn} \in \mathcal{S}_k(\Gamma_1(N^2)).$$

Recall that the space  $S_k(\Gamma_1(N^2))$  may be represented as a direct sum of spaces of modular forms for  $\Gamma_0(N^2)$  with character (cf. Remark 6.8 and Example 6.9) so that the coefficients fulfil  $a(n) \in \mathcal{O}(n^{\alpha})$  by assumption.

As a consequence, it suffices to cite asymptotic bounds of Fourier coefficients for modular forms for  $\Gamma_0(N)$  with nebentypus. There are some sharper bounds available in this setting. For integral weight modular forms, there is the Deligne bound, stated in Thm. 2.26, providing the best possible bound. However, it had been formulated only for natural numbers *n* being coprime to the level *N* of the modular form. Due to the structure of the Hecke algebra (cf. Proposition 6.41) it will then suffice to only prove a similar bound for coefficients at primes that divide the level. In fact, by Atkin–Lehner theory, it suffices to do so for newforms. The following is due to Ogg and Li and part of [Li74, Thm. 3 p. 295].

**Proposition 3.80.** Let  $k \in \mathbb{Z}_{>0}$  and  $f \in \mathcal{S}_k^{\text{new}}(\Gamma_0(N), \chi)$  with Fourier expansion

$$f(\tau) = \sum_{n=1}^{\infty} a(n) e(n\tau)$$

be a normalised form that is an eigenform for the Hecke algebra  $\mathscr{H}_N$ . Then for  $p \mid N$  we find the following

- i) If  $\chi$  is not induced by a character modulo N/p, then  $|a(p)| = p^{k/2-1/2}$ .
- ii) If  $\chi$  is a character modulo N/p and  $p^2 \nmid N$ , then  $a(p)^2 = \chi(p)p^{k-1}$ .
- iii) If  $\chi$  is a character modulo N/p and  $p^2 \mid N$ , then a(p) = 0.

As a Corollary we obtain the following bound on Fourier coefficients.

**Theorem 3.81.** Let  $k \in \mathbb{Z}_{>0}$  and  $f \in S_k(\Gamma_0(N), \chi)$  with Fourier expansion

$$f(\tau) = \sum_{n=1}^{\infty} a(n)e(n\tau).$$

Then

$$a(n) \in \mathcal{O}_{\varepsilon}(n^{k/2-1/2+\varepsilon}).$$

However, for half integral weight forms, the situation is more delicate. Nevertheless, there are the following two bounds available in the literature. The first has been proven by Blomer and Harcos and is derived from [BH08, Cor. 2 p. 55].

**Theorem 3.82.** Let  $k \in (\mathbb{Z}/2) \setminus \mathbb{Z}$  with  $k \geq 5/2$ ,  $N \in \mathbb{N}$ , and  $\chi$  be a character modulo 4N. Let

$$f(\tau) = \sum_{n=1}^{\infty} a(n)e(n\tau)$$

be in  $\mathcal{S}_k(\Gamma_0(4N), \chi)$ . Then for  $t \in \mathbb{N}$  and  $n \nmid (2N)^{\infty}$  we find

$$a(tn) \in \mathcal{O}_{\varepsilon}(n^{k/2-5/16+\varepsilon}).$$

Note that the above theorem is also true in case of k = 3/2, if the choice of cusp form is reduced to the orthogonal complement of theta functions.

The second bound goes back to Iwanicc [Iwa87] and Duke [Duk88] in a special case and has been generalised by Waibel [Wai18, Thm. 1 p. 186].

**Theorem 3.83.** Let  $k \in (\mathbb{Z}/2) \setminus \mathbb{Z}$  with  $k \geq 5/2$ ,  $4 \mid N \in \mathbb{N}$ , and  $\chi$  be a character modulo 4N. Let

$$f(\tau) = \sum_{n=1}^{\infty} a(n) e(n\tau)$$

be in  $S_k(\Gamma_0(4N), \chi)$ . Then it holds for indices  $n = tv^2w^2$  with t squarefree,  $v \mid N^{\infty}$ , and gcd(w, N) = 1 that

$$a(n) \in \mathcal{O}_{\varepsilon}(n^{k/2-1/2+\varepsilon}v^{1/2}).$$

Note that the above implies  $a(n) \in \mathcal{O}_{\varepsilon}(n^{k/2-1/4+\varepsilon})$ , which is the Weil bound.

In conclusion we have the following bounds on Fourier coefficients of vector valued modular forms.

**Corollary 3.84.** Let  $2 \leq k \in \mathbb{Z}/2$  and  $f \in \mathcal{S}_{L,k}(Mp_2(\mathbb{Z}))$  with Fourier expansion

$$f( au) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n au) \otimes \mathfrak{e}_{\lambda}.$$

Assume  $n \in \mathbb{Q}$  is in a progression with an upper bound on the number of bad primes appearing in the prime number decomposition of n. Then for all  $\lambda \in \mathcal{L}$ 

$$a(\lambda, n) \in \begin{cases} \mathcal{O}_{\varepsilon}(n^{k/2 - 1/2 + \varepsilon}), & 2 \mid \operatorname{rank}(L), \\ \mathcal{O}_{\varepsilon}(n^{k/2 - 5/16 + \varepsilon}), & 2 \nmid \operatorname{rank}(L). \end{cases}$$

For general n and  $t \in \mathbb{Q}$ , the following bound is true

$$a(\lambda, tn^2) \in \begin{cases} \mathcal{O}_{\varepsilon}(n^{k-1+\varepsilon}), & 2 \mid \operatorname{rank}(L), \\ \mathcal{O}_{\varepsilon}(n^{k-1/2+\varepsilon}), & 2 \nmid \operatorname{rank}(L). \end{cases}$$

In later applications, certain progressions of indices of Fourier coefficients of the vector valued form f are considered. Hence, we conclude with statements about how the bounds above transfer to setting we have in mind.

**Corollary 3.85.** Let  $f \in \mathcal{M}_{k,L}(Mp_2(\mathbb{Z}))$  be a modular form with Fourier expansion

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \mathbb{Z} + \lambda^2/2} a(\lambda, n) \cdot e(n\tau) \otimes \mathfrak{e}_{\lambda}.$$

Assume that there is  $\sigma \in \mathbb{R}$  such that for all  $\lambda \in \mathcal{L}$  we find  $a(\lambda, n) \in \mathcal{O}(n^{k/2-\sigma})$ . Select  $d \in \mathbb{N}$  and fix a pair  $(\lambda, n)$ . Then

$$a(m\lambda, m^d \cdot n) \in \mathcal{O}(m^{d(k/2-\sigma)}).$$
(3.69)

*Proof*: We find

$$|a(m\lambda, m^d n)| \le \sum_{\mu \in \mathcal{L}} |a(\mu, m^d n)| \in \mathcal{O}(m^{d(k/2-\sigma)}).$$

**Remark 3.86.** Clearly, the above corollary is also valid for progressions  $\mathfrak{s} : \mathcal{L} \times \mathbb{Q} \to \mathcal{L} \times \mathbb{Q}$ such that the projection to the second component of  $\mathfrak{s}$  is in  $\mathcal{O}(m^d)$  for some d > 0 which is independent of  $\lambda \in \mathcal{L}$ . Formalising this statement, however, interfered with readability.
The section will be concluded with a collection of bounds that are required for proving convergence of L-series in Section 6.4.1.

**Corollary 3.87.** Let  $2 \leq k \in \mathbb{Z}/2$  and  $f \in \mathcal{S}_{L,k}(Mp_2(\mathbb{Z}))$  with Fourier expansion

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n\tau) \mathfrak{e}_{\lambda}.$$

Select  $t \in \mathbb{Q}$ , then the following bound is true for all  $\lambda \in \mathcal{L}$ 

$$a(\lambda, tn^2) \in \begin{cases} \mathcal{O}_{\varepsilon}(n^{k-1+\varepsilon}), & 2 \mid \operatorname{rank}(L), \\ \mathcal{O}_{\varepsilon}(n^{k-1/2+\varepsilon}), & 2 \nmid \operatorname{rank}(L). \end{cases}$$

If in addition  $n \in \mathbb{Q}$  is assumed to be coprime to lev(L) we obtain

$$a(\lambda, tn^2) \in \begin{cases} \mathcal{O}_{\varepsilon}(n^{k-1+\varepsilon}), & 2 \mid \operatorname{rank}(L), \\ \\ \mathcal{O}_{\varepsilon}(n^{k-5/8+\varepsilon}), & 2 \nmid \operatorname{rank}(L). \end{cases}$$

In order to recollect the different bounds relevant to our setting, we introduce a parameter  $\sigma \in \mathbb{R}$  associated to some  $f \in \mathcal{M}_{L,k}(\mathrm{Mp}_2(\mathbb{Z}))$  describing the improvement of the asymptotic growth of its Fourier coefficients compared to the Hecke bound, i.e.  $\sigma \in \mathbb{R}$ is chosen such that in the case of a progression which is linear in n we have

$$a(\lambda, n) \in \mathcal{O}_{\varepsilon}(n^{k/2-\sigma+\varepsilon}).$$
 (3.70)

In case  $f \notin S_{L,k}$  and sufficiently large weight, we select the obvious bound  $\sigma = 1 - k/2$ . In case of  $f \in S_{L,k}$ , Table 3.1 provides an overview of possible choices, also for a variant of progression in the index. In other cases, for theta functions of low weight, for instance, the parameter  $\sigma$  has to be determined separately.

σ	$\begin{vmatrix} n \in \mathbb{N} \\ a(\lambda, n) \end{vmatrix}$	$\begin{array}{l} n \in \mathbb{N} \\ a(\lambda, tn^2) \end{array}$	$\begin{array}{c}n \nmid \operatorname{lev}(L)^{\infty}\\ a(\lambda, tn^2)\end{array}$
$\begin{array}{c c} 2 \mid \operatorname{rk}(L) \\ 2 \nmid \operatorname{rk}(L) \end{array}$	$\begin{vmatrix} 1/2 \\ 0 \end{vmatrix}$	$\frac{1/2}{1/4}$	$1/2 \\ 5/16$

Table 3.1: To a cusp form  $f \in S_{L,k}$  of weight  $k \geq 2$  we associate the parameter  $\sigma \in \mathbb{R}$  above, determined by the asymptotic growth of its Fourier coefficients  $a(\lambda, n)$  relative to the Hecke bound, i.e.  $a(\lambda, n) \in \mathcal{O}_{\varepsilon}(n^{k/2-\sigma+\varepsilon})$ . Its direct application is determining the range of convergence of *L*-series associated to *f* (cf. Subsection 6.4.1). The number  $t \in \mathbb{Q}$  appearing above allows for a shift in the index in order to consider the variant  $a(\lambda, tn^2) \in \mathcal{O}_{\varepsilon}(n^{k-2\sigma+\varepsilon})$ .

# 4 Orthogonal theory

The preceding chapter focussed on introducing the symplectic setting for modular forms. This chapter presents a similar theory in the orthogonal setting. First, properties of orthogonal groups and the construction of the spin group are reviewed. Subsequently, models of modular varieties in this context, the standard arithmetic subgroups and modular forms are introduced. Finally, a brief review of special divisors which play a significant role for cycle integrals in Section 7.3 is provided.

## 4.1 Group properties

In this section we summarise the theory of orthogonal and Spin groups in the context of our investigations.

#### 4.1.1 Orthogonal groups

**Definition 4.1.** Let (M, q) be a free non-degenerate quadratic R module. Then the associated *orthogonal group* is defined as

$$O(M) := \{ \gamma \in Aut(M) \mid b(\gamma x, \gamma y) = b(x, y) \text{ for all } x, y \in M \}.$$

Here, b denotes the associated bilinear form. Further, the subgroup of elements with determinant 1 is denoted SO(M). Further, if M = V is an  $\mathbb{R}$  vector space of signature  $(m^+, m^-)$ , we write  $O(m^+, m^-) \coloneqq O(V)$ .

One of the most significant examples of orthogonal transformations are reflections.

Example 4.2. Let (V, q) be a quadratic space over K. Then every  $u \in V$  with non-trivial norm q(u) induces a *reflection* at the hyperplane  $u^{\perp}$  via  $\tau_u : v \mapsto v - b(v, u) q(u)^{-1}u$ . In fact, we have  $\tau_u \in O(V)$ . The name reflection derives from the properties  $\tau_u(u) = -u$ and  $\tau_u(v) = v$  for  $v \in u^{\perp}$ . In particular,  $\tau_u^2 = id$  and we find  $det(\tau_u) = -1$ , so that  $\tau_u \notin SO(V)$ . We will present more examples, which are discussed in more detail in [Fis10, 5.5 p. 303]. Example 4.3. Let V be a real vector space of dimension 2 with standard scalar product as bilinear form, so that  $x \mapsto q(x) = \frac{1}{2} \langle x, x \rangle$  defines the quadratic form.

a) The map from the unit circle

$$\mathbb{T} \to \mathrm{SO}(V), \quad \vartheta \mapsto k_\vartheta \coloneqq \begin{pmatrix} \cos(\vartheta) & -\sin(\vartheta) \\ \sin(\vartheta) & \cos(\vartheta) \end{pmatrix}$$
(4.1)

defines an injective group homomorphism from the complex unit circle. The associated map may be visualised as a rotation by the angle  $\vartheta$ .

b) We find that for any  $\vartheta \in \mathbb{T}$  the map represented by the matrix

$$r_{\vartheta} \coloneqq \begin{pmatrix} \cos(\vartheta) & \sin(\vartheta) \\ \sin(\vartheta) & -\cos(\vartheta) \end{pmatrix}$$

is also orthogonal. In fact, it is elementary to verify that any element in O(2,0) is represented by either  $k_{\vartheta}$  or  $r_{\vartheta}$  for a suitable choice of  $\vartheta \in \mathbb{T}$ , yielding a homeomorphism  $O(2,0) \simeq \mathbb{T} \sqcup \mathbb{T}$ , if O(2,0) is equipped with the relative topology.

**Remark 4.4.** In case of M = V a K vector space, the group O(M) may be defined as the stabiliser of q of the natural right action  $GL(V) \ni g \mapsto q \circ g$  on quadratic forms on M, as long as the characteristic of K is different from 2.

**Remark 4.5.** If (M, q) is a free non-degenerate quadratic R module and  $R \to S$  is a map of rings, we have a natural induced map  $O(M) \to O(M \otimes S)$ . In particular, for a  $\mathbb{Z}$  lattice  $O(L) \hookrightarrow O(L \otimes S)$  for  $S \in \{\mathbb{Z}_p, \mathbb{Q}_p, \mathbb{R}\}$ .

The following two theorems are essential for our investigations and central to the orthogonal theory. The extension result is due to Witt, while the generation result is due to Cartan–Dieudonné.

**Theorem 4.6** ([Cas78, Thm. 4.1]). Let (V, q) be a quadratic space over a field of characteristic different from 2 with regular subspaces  $V_1, V_2$ . Suppose there is an isomorphism  $\sigma : (V_1, q |_{V_1}) \to (V_2, q |_{V_2})$ , then  $\sigma$  extends to an automorphism of (V, q).

**Theorem 4.7** ([Gro02, Thm. 6.6]). Let (V, q) be a quadratic K vector space with char $(K) \neq 2$  of dimension m. Then every element in O(V) is the product of at most m reflections.

**Remark 4.8** ([Fis10, 5.5.5, 5.5.6, p. 307]). In fact, the examples presented above provide a complete list of orthogonal elements for definite quadratic  $\mathbb{R}$  spaces. Let (V, q) be such a space of dimension m, and  $\gamma \in O(V)$ , then there are  $r^+, r^-, r \in \mathbb{N}_0$  with  $r^+ + r^{-1} + 2r = m$ , and  $\vartheta_i \in (0, \pi) \cup (\pi, 2\pi)$  such that  $\gamma$  is represented by a matrix

$$\mathcal{I}_{r^+} \oplus (-1) \cdot \mathcal{I}_{r^-} \oplus \bigoplus_{i=1}^r k_{\vartheta_i}$$

with respect to a suitable orthogonal basis of V. Further, in case of a complex quadratic space, all Gram matrices are diagonal.

Recall that  $\operatorname{GL}_n(\mathbb{R}) \simeq \operatorname{Aut}(V)$  carries the trace topology of  $\mathbb{R}^{n \times n}$ .

**Lemma 4.9.** Let (V,q) be a regular real quadratic space. Then O(V) is closed. Further, it is compact, if, and only if, V is definite.

*Proof*: Recall that S denotes the Gram matrix of q. Consider the map  $M_n(\mathbb{R}) \ni g \mapsto g^T S g$ , wich is continuous as it is polynomial in the coordinates of  $M_n(\mathbb{R})$ . Then O(V) is given by the preimage of S and as such closed.

Recall further that  $\sqrt{q(\cdot)}$  defines a norm on V if, and only if, q is definite. This norm induces an operator norm on the operator algebra  $\mathscr{B}(V) \simeq M_n(\mathbb{R})$  whose unit ball contains O(V). Since  $M_n(\mathbb{R})$  is finite dimensional, all of its norms are equivalent and O(V) is compact as it is closed and totally bounded.

Assume  $(V, \mathbf{q}) \simeq H$  is a hyperbolic plane. Further, we find for any  $a \in \mathbb{R}^{\times}$  that

$$\operatorname{diag}(a, a^{-1})^T H \operatorname{diag}(a, a^{-1}) = H,$$

meaning  $s_a := \operatorname{diag}(a, a^{-1}) \in \mathcal{O}(H)$ . However, in the standard operator topology we have  $\|s_a\| = \max\{|a|, |a^{-1}|\}$ , meaning  $\mathcal{O}(H)$  is unbounded. Now if (V, q) was not a hyperbolic plane but indefinite, we may assume a hyperbolic split, rendering  $\mathcal{I}_{m-2} \oplus s_a \in \mathcal{O}(V)$  with  $\|\mathcal{I}_{m-2} \oplus s_a\| = \|s_a\|$ , settling the general case.

Before turning to constructing the general spin group, which will be an extension of the special orthogonal group SO, we will state some topological properties of SO and O in the following remark which is not trivial to proove.

**Remark 4.10.** Let (V, q) be a real quadratic space of signature  $(m^+, m^-)$ .

- a) If (V, q) is definite, O(V) is compact and has two connected components, one of which is given by SO(V). Then SO(V) is the identity component and is even pathconnected, as may be proven by means of Remark 4.8.
- b) If (V, q) is indefinite, O(V) is non-compact and has four connected components. The components correspond to whether the elements preserve the orientation on the maximal (positive and negative) definite subspaces. SO(V) then has two connected components, corresponding to elements (not) changing the orientation on both maximal positive and negative definite subspaces. Further,

$$O(m^+) \times O(m^-) < O(m^+, m^-),$$
  
 $SO(m^+, m^-) \cap (O(m^+) \times O(m^-)) < SO(m^+, m^-),$   
 $SO(m^+) \times SO(m^-) < SO^+(m^+, m^-)$ 

are maximal compact subgroups, where  $\mathrm{SO}^+$  denotes the identity component. We find that

$$O(m^+, m^-) / O(m^+) \times O(m^-) \simeq SO^+(m^+, m^-) / SO(m^+) \times SO(m^-).$$

#### 4.1.2 Spin groups

In the following section the Clifford algebra is constructed in order to derive the general Spin group from it. This group functions as a cover of the special orthogonal group and is a valid choice for defining Shimura varieties in the orthogonal setting, since these embed naturally into Siegel varieties and are hence defined over  $\mathbb{Q}$ , making the transition to a number field superfluous. For basic facts about the tensor product and tensor algebra we refer to [GW20] and [Lan02, XVI-7 p. 632]. The primary source for the introduction of the Clifford algebra and the Spin group is [Bru+08, 2.2 p. 129] which may be supplemented with [Kit99].

Recall the definition of tensor products.

**Definition 4.11.** Let R be a ring and M, N be R modules. A *tensor product* of M, N is an R module  $M \otimes_R N$  together with an R bilinear map  $\tau : M \times N \to M \otimes_R N$ , denoted  $(m, n) \mapsto m \otimes n$ , which is universal for the following diagram.



This means for every R module and R bilinear map  $\alpha : M \times N \to P$  there exists exactly one linear map  $\beta : M \otimes_R N \to P$ , such that  $\alpha = \beta \circ \tau$ .

**Remark 4.12.** i) The pair  $(M \otimes_R N, \tau)$  is unique up to unique isomorphism.

ii) If  $(m_i), (n_j)$  are generating sets of M, N, respectively, then the *elementary tensors*  $(m_i \otimes n_j)$  form a generating set of  $M \otimes_R N$ , in particular, the tensor product is a finitely generated R module, if M, N are.

Example 4.13. Given a ring homomorphism  $f : R \to S$ , the R module M induces an S module via tensoring. Namely,  $M \otimes_R S$ , where S is regarded as an R module via f carries the structure of an S module, since S itself is an S module. More precisely, the multiplication in S is associative, so that for  $r \in R$ ,  $s_1, s_2 \in S$  we find that the corresponding actions commute  $f(r) \cdot (s_1 \cdot s_2) = (f(r) \cdot s_1) \cdot s_2$ , so that S inherits an S right action and is rendered an (R, S) bimodule. Then, however,  $M \otimes_R S$  carries the structure of an S (right) module. This is called extension of scalars.

A common realisation is what we have encountered in Section 1.1 in the context of quadratic modules, where  $R \hookrightarrow S$  injects and the name extension of scalars becomes quite graphical. If, for instance, R = K is a field, and F/K its algebraic closure and  $(b_i)$  denotes a basis of the vector space V, then  $V(F) \coloneqq V \otimes_K F$  is a vector space over F with basis  $(b_i)$  again. This process is then called *complexification* of the K vector space V.

The tensor product as an operation on modules has remarkable properties, constituting the structure of the Witt ring.

- **Remark 4.14.** a) The modules  $M \otimes_R N$  and  $N \otimes_R M$  are isomorphic via  $m \otimes n \mapsto n \otimes m$ , i.e. the tensor product is a commutative operation.
  - b) The modules  $(M_1 \otimes_R M_2) \otimes_R M_3$  and  $M_1 \otimes_R (M_2 \otimes_R M_3)$  are isomorphic via  $(m_1 \otimes m_2) \otimes m_3 \mapsto m_1 \otimes (m_2 \otimes m_3)$ , i.e. the tensor product is an associative operation. This is why we write  $m_1 \otimes m_2 \otimes m_3$  for  $(m_1 \otimes m_2) \otimes m_3$  and  $\bigotimes_{i \in I} M_i$  for the tensor product of a finite family  $(M_i)_{i \in I}$  of R modules.
  - c) The map  $R \otimes_R M \to M$ ,  $r \otimes m \mapsto rm$  defines an R linear isomorphism, i.e. the module R functions as a unit element with respect to the tensor product.

d) For a family of R modules  $M_i$  and an R modules N, the map  $(m_i)_i \otimes n \mapsto (m_i \otimes n)_i$ defines an isomorphism  $(\bigoplus_i M_i) \otimes_R N \simeq \bigoplus_i (M_i \otimes N)$ , i.e. the tensor product commutes with direct sums or in short *distributivity*.

For an R module M, we seek an R algebra, that canonically includes R, as well as M. The tensor product appears to possess the correct properties to be beneficial in that quest. Let M be an R module and  $l \in \mathbb{N}_0$ . We write  $M^{\otimes 0}$  for R and  $M^{\otimes l+1} \coloneqq M^{\otimes l} \otimes_R M$ . Then from the associativity of the tensor product we obtain for any  $l_1, l_2 \in \mathbb{N}_0$  a bilinear associative map

$$M^{\otimes l_1} \otimes M^{\otimes l_2} \to M^{\otimes (l_1+l_2)}, \quad (m_1, m_2) \mapsto m_1 \otimes m_2$$

This declares a ring structure on the direct sum  $\bigoplus_{l=0}^{\infty} M^{\otimes l}(M)$  rendering it an algebra. **Definition 4.15.** Let M be an R module. The *tensor algebra*  $\mathcal{T}(M)$  of M is defined as

$$\mathcal{T}(M) \coloneqq \bigoplus_{l=0}^{\infty} M^{\otimes l}.$$
(4.2)

The product of two elements  $m, n \in \mathcal{T}(M)$  is denoted  $m \otimes n$ . For  $l \in \mathbb{N}_0$ , we refer to  $M^{\otimes l}$ in  $\mathcal{T}(M)$  as the tensors of *degree* l and denote the summand by  $\mathcal{T}^l(M)$ , rendering  $\mathcal{T}(M)$ a graded associative algebra with canonical injection  $\iota : M \to \mathcal{T}(M)$ .

**Remark 4.16.** Note that for  $l \in \mathbb{N}_0$  the association  $M \mapsto \mathcal{T}^l(M)$  is functorial. In fact, if  $(M_i), (N_i)$  are finite families of R modules and  $f_i : M_i \to N_i$  are R linear maps. Then we may concatenate  $\prod_i f_i$  with the tensor map  $\tau : \prod_i N_i \to \bigotimes_i N'_i$  to obtain a multilinear map  $\prod_i M_i \to \bigotimes_i N_i$  that factors uniquely through  $\bigotimes_i M_i$  and gives by the universal property of  $\bigotimes_i M_i$  rise to a linear map

$$\bigotimes_{i} f_{i} : \bigotimes_{i} M_{i} \to \bigotimes_{i} N_{i}.$$

$$(4.3)$$

This shows that  $\mathcal{T}^l: M \mapsto M^{\otimes l}, f \mapsto f^{\otimes l} \coloneqq \bigotimes_{i=1}^l f$  is functorial in the category of R modules.

**Remark 4.17.** For an R module map  $f : M \to N$  the family of associated maps  $(\mathcal{T}^{l}(f))$ induces a map  $\mathcal{T}(M) \to \mathcal{T}(N)$  which we denote by  $\mathcal{T}(f)$ . In fact, for a finite family of elements  $m_i \in M$  we find

$$\mathcal{T}(f)(\otimes_i m_i) = \otimes_i f(m_i),$$

so that we obtain an algebra morphism and hence end up with a functorial association  $\mathcal{T}: M \mapsto \mathcal{T}(M), \quad f \mapsto \mathcal{T}(f)$  of R modules to graded R algebras.

**Theorem 4.18.** The tensor algebra satifies the following universal property: For any linear map  $f: M \to A$  for an associative R algebra A, there is a unique linear algebra morphism  $\rho: \mathcal{T}(M) \to A$  such that the following diagramm commutes.

$$\begin{array}{c} \mathcal{T}(M) \xrightarrow{\exists ! \rho} A \\ & \uparrow & & \uparrow \\ M & & & & \\ M & & & & \\ \end{array}$$

Example 4.19. The universal property of  $\mathcal{T}(M)$  implies that the mapping defined by  $J: M \to M \subset \mathcal{T}(M), \quad m \mapsto -m$  lifts to a map  $\mathcal{T}(M) \to \mathcal{T}(M)$ . It is called the canonical automorphism of  $\mathcal{T}(M)$ .

There are several algebras derived from the tensor algebra. Recall that morphisms from  $\mathcal{T}(M)$  to commutative algebras factor through the quotient by the commutator ideal  $\langle [m,n] \rangle$ .

**Definition 4.20.** Let M be an R module.

- a) Let  $\mathcal{C} := \langle [\mathcal{T}(M), \mathcal{T}(M)] \rangle$  denote the commutator ideal of  $\mathcal{T}(M)$ . Then the symmetric algebra of M is defined to be the quotient  $\operatorname{Sym}(M) := \mathcal{T}(M)/\mathcal{C}$ . The product is denoted by ,... The image of  $\mathcal{T}^{l}(M)$  in  $\operatorname{Sym}(M)$  is denoted by  $\operatorname{Sym}^{l}(M)$  and called the *l*-th symmetric power of M.
- b) Let  $\mathcal{I} = \langle \{m \otimes n + n \otimes m \mid m, n \in \mathcal{T}(M)\} \rangle$  denote the anticommutator ideal of  $\mathcal{T}(M)$ . Then the *exterior algebra* of M is defined to be the quotient  $\bigwedge(M) \coloneqq \mathcal{T}(M)/\mathcal{I}$ . The product is denoted by  $\wedge$ . The image of  $\mathcal{T}^{l}(M)$  in  $\bigwedge(M)$  is denoted by  $\bigwedge^{l}(M)$  and called the *l*-th *exterior power* of M.

The following universal property is directly inherited from the one of the tensor algebra.

**Remark 4.21.** By construction,  $M^l \to \mathcal{T}^l(M) \to \operatorname{Sym}^r(M)$  is universal for multilinear symmetric maps.

Example 4.22. Let V be an F vector space of dimension n. Then  $\bigwedge^{n}(V)$  has dimension one and we find by functoriality of  $\bigwedge^{l}$  that for any endomorphism  $f: V \to V$ , we have an associated linear map  $\bigwedge^{n}(f): \bigwedge^{n}(V) \to \bigwedge^{n}(V) \simeq F$ . Clearly,  $\operatorname{Hom}_{VS}(F,F) \simeq F$ canonically, so that  $\bigwedge^{n}(f)$  may be interpreted as an element in F. It is called the *determinant* of f. **Remark 4.23.** Let  $n \in \mathbb{N}_0$  and  $\mathfrak{S}_n$  denote the *n*-th symmetric group, i.e. the automorphism group of the set of *n* elements. Then there is a canonical injection  $\iota_n : \mathfrak{S}_n \hookrightarrow \mathfrak{S}_{n+1}$  by letting an element  $\sigma \in \mathfrak{S}$  act on the first *n* elements of  $\{1, \ldots, n+1\}$  and fixing the last element. This gives rise to a filtered inductive system and the colimit along this system is called the *general finite symmetric group* and denoted  $\mathfrak{S}_f$ .

It is clear that  $\mathfrak{S}_n$  acts on  $\mathcal{T}^n(M)$  via  $\sigma(\otimes_{1 \leq i \leq n} m_i) \coloneqq \otimes_i m_{\sigma(i)}$ . This action extends to a linear action of  $\mathfrak{S}_f$  on  $\mathcal{T}(M)$  and gives rise to the ideal

$$\mathcal{C} \coloneqq \langle \{m - \sigma(m) \mid m \in \mathcal{T}(M), \sigma \in \mathfrak{S}_f \} \rangle.$$

The quotient  $\mathcal{T}(M)/\mathcal{C}$  is isomorphic to the symmetric algebra  $\operatorname{Sym}(M)$ .

**Definition 4.24.** Assume (M, q) is a quadratic R module mapping to R. Consider the ideal  $\mathcal{I} := \langle \{m \otimes m - q(m) \mid m \in M\} \rangle \trianglelefteq \mathcal{T}(M)$ . Then the quotient  $\mathcal{C}_M := \mathcal{T}(M)/\mathcal{I}$  is called the *Clifford algebra*.

**Remark 4.25.** a) Write  $m \cdot n \coloneqq m \otimes n \in \mathcal{C}_M$ . Then

$$m \cdot m = q(m), \qquad m \cdot n + n \cdot m = b(m, n).$$
 (4.4)

In particular, we have  $m \cdot n = -n \cdot m$ , if, and only if, m and n are orthogonal to each other.

- b) Both R as a ring and M as a module are canonically injected into  $\mathcal{C}_M$ .
- c) If M is free and of finite rank with basis  $b_1, \ldots, b_n$  then the family

$$b_{i_1} \cdots b_{i_r} \qquad (1 \le r \le n, 1 \le i_1, < \ldots < i_r \le n)$$

forms a basis of  $\mathcal{C}_M$ , rendering it a free R module of rank  $2^n$ .

*Example* 4.26. If char(R)  $\neq 2$  and  $q \equiv 0$ , we find  $C_M \simeq \bigwedge(M)$ .

**Remark 4.27.** The Clifford algebra satisfies the following universal property: For any linear map  $f: M \to A$  for an associative unital R algebra A, such that  $f(m)^2 = q(m) \cdot 1 \in A$  for all  $m \in M$ , there is a unique linear algebra morphism  $\rho: \mathcal{C}_M \to A$  such that the following diagramm commutes.



Note that, as a consequence, any isometry  $M_1 \to M_2$  lifts to an R algebra morphism  $\mathcal{C}_{M_1} \to \mathcal{C}_{M_2}$  so that the Clifford construction yields a functor from quadratic R modules with isometries as morphisms to the category of unital associative R algebras.

From now on, we will consider the case of free quadratic R modules (M, q). There are two canonical automorphisms on  $C_M$  which will become important for constructing and working with the Spin group (cf. Subsection 5.4.3).

**Definition 4.28.** Let (M,q) be a free quadratic R module.

- a) Multiplication by -1 induces an isometry on M. By the universal property of  $\mathcal{C}_M$  it lifts to a map  $J : \mathcal{C}_M \to \mathcal{C}_M$  called the *canonical automorphism*.
- b) Define the *canonical involution* by

$$\cdot^T : \mathcal{C}_M \to \mathcal{C}_M, \qquad m_1 \cdot \dots \cdot m_k \mapsto m_k \cdot \dots \cdot m_1.$$
 (4.5)

c) Define the *Clifford norm* on  $\mathcal{C}_M$  by

$$N: \mathcal{C}_M \to \mathcal{C}_M, \qquad m \mapsto m^T \cdot m.$$
 (4.6)

Example 4.29. a) For  $m \in M \leq \mathcal{C}_M$  we find  $m^T = m$  and hence N(m) = q(m).

We are ultimately interested in finite dimensional quadratic vector spaces over fields of characteristic 0, so assume in the following that  $2 \in \mathbb{R}^{\times}$  and that M has the finite orthogonal basis  $(b_i)_{i=1}^n$ .

**Definition 4.30.** Write  $\mathcal{C}_M^0$  for the subalgebra generated by products of an even number of basis elements  $b_i$  and call it the *even* Clifford algebra. On the contrary, the submodule  $\mathcal{C}_M^1$  is defined as the submodule generated by products of an odd number of basis elements.

- **Remark 4.31.** a) The decomposition  $\mathcal{C}_M = \mathcal{C}_M^0 \oplus \mathcal{C}_M^1$  as a modules defines a  $\mathbb{Z}/2\mathbb{Z}$  grading on  $\mathcal{C}_M$ . In case  $-1 \neq 1 \in \mathbb{R}$ , note that  $\mathcal{C}_M^0 = (\mathcal{C}_M)^J$  equals the fix space of the canonical automorphism J.
  - b) Let  $\delta \coloneqq b_1 \cdots b_n \in \mathcal{C}_M$ . Then

$$\delta b_i = (-1)^{n-1} b_i \delta, \qquad \delta^2 = (-1)^{(n-1)n/2} 2^{-n} \det(\mathbf{q}) \in R/(R^{\times})^2.$$
 (4.7)

Further, this special element determines the centre of the Clifford algebra

$$Z(\mathcal{C}_M) = \begin{cases} R & \text{if } n \text{ is even,} \\ R + R\delta & \text{if } n \text{ is odd,} \end{cases}$$

$$Z(\mathcal{C}_M^0) = \begin{cases} R + R\delta & \text{if } n \text{ is even,} \\ R & \text{if } n \text{ is odd.} \end{cases}$$

A sizeable range of examples is found in [Bru+08, p. 132] that is further extended by [Kit99, 1.5 p. 24]

Now that the Clifford algebra has been introduced we turn towards the main application of this construction within the scope of our quest: the construction of the Spin group. Note that for every invertible element  $x \in \mathcal{C}_M^{\times}$  we obtain a linear map

$$v_x: M \to M, \quad m \mapsto xmJ(x)^{-1}.$$

**Definition 4.32.** Define the *Clifford group* of M as

$$\mathfrak{G}_M \coloneqq \{ x \in \mathcal{C}_M^{\times} \mid v_x(M) = M \}$$

The induced representation  $v : \mathfrak{G}_M \to \operatorname{Aut}_R(M)$  is referred to as the vector representation of  $\mathfrak{G}_M$ .

Example 4.33. Note that  $x \in \mathfrak{G}_M \cap V$  gives rise to the reflection  $\tau_x$  at the hyperplane  $x^{\perp}$  that had already appeared in Example 4.2. In fact, with w = rx + sy for  $y \in x^{\perp}$ ,  $r, s \in R$ , we find

$$v_x(w) = xwJ(x)^{-1} = -rxxx^{-1} - sxyx^{-1} = -rx + syxx^{-1} = \tau_x(w).$$

Lemma 4.34. Let  $-1 \neq 1 \in R$ 

- a) We find  $\ker(v) = R^{\times}$ .
- b) The mapping  $N : \mathfrak{G}_M \to R^{\times}$  defines a homomorphism.
- c) Moreover,  $v : \mathfrak{G}_M \to \operatorname{Aut}(M)$  only targets the subgroup  $O(M) \subseteq \operatorname{Aut}(M)$ .

*Proof*: It suffices to verify that  $\ker(v) \subseteq R^{\times}$ . Let  $x \in \ker(v)$ , then  $x = x_0 + x_1 \in \mathcal{C}^0_M \oplus \mathcal{C}^1_M$ and we have  $xmJ(x)^{-1}$  for all  $m \in M$ . This means

$$x_0 m = m x_0,$$
$$x_1 m = -m x_1$$

However, these *m* generate  $C_M$ , so that we find  $x_0 \in Z(C_M) \cap C_M^0 = R$ . Also, the second condition can be tested with basis elements  $m = b_i$  of *M*. Then we find  $b_i x_1 = x_1 b_i$  and

hence by (4.7)  $x_1 = 0$ .

b) For  $m \in M$  and  $x \in \mathfrak{G}_M$ , we have  $v_x(m) = xmJ(x)^{-1} \in M$  and hence the identity  $v_x(m) = -J(v_x(m))^T$ . This implies

$$xmJ(x)^{-1} = v_x(m) = -J(v_x(m))^T = (x^T)^{-1}mJ(x^T)$$

which in turn provides

$$m = (x^T x)mJ(x^T x)^{-1} = N(x)mJ(N(x))^{-1} = v_{N(x)}(m)$$

However, the kernel of v had been identified to equal  $R^{\times}$  and hence  $N(x) \in R^{\times}$ . c) Recall that for  $x \in \mathfrak{G}_M, m \in M$  we have  $v_x(m) \in M$ , so that

$$q(v_x(m)) = N(v_x(m)) = J(x^{-1})^T m^T x^T x m J(x^{-1}) = q(m),$$

meaning  $v_x \in \mathcal{O}(M)$ .

With these preparations we may define a cover of the special orthogonal group that is regularly used to define orthogonal Shimura varieties, the general *Spin* group. The advantage of using it in order to define varieties in place of the orthogonal group is that the corresponding varieties embed into Siegel varieties and are hence defined over  $\mathbb{Q}$ , meaning it is not necessary to pass to an extension of  $\mathbb{Q}$ .

**Definition 4.35.** The general Spin group and Spin group of M are defined as

$$\operatorname{GSpin}_M \coloneqq \mathfrak{G}_M \cap \mathcal{C}_M^0, \tag{4.8}$$

$$\operatorname{Spin}_M \coloneqq \operatorname{GSpin}_M \cap \ker(\mathbf{N}).$$
 (4.9)

**Remark 4.36.** In case R = K is a field of characteristic unequal to 2, we write V = M and note that the orthogonal group is generated by reflections (cf. Theorem 4.7), so that the special orthogonal group is the subgroup of elements given by an even number of reflections. Consequently, we obtain the following exact sequences

$$1 \longrightarrow K^{\times} \longrightarrow \mathfrak{G}_{V} \longrightarrow \mathcal{O}(V) \longrightarrow 1,$$
$$1 \longrightarrow K^{\times} \longrightarrow \operatorname{GSpin}_{V} \longrightarrow \mathcal{SO}(V) \longrightarrow 1.$$

**Remark 4.37.** The above exact sequence implies  $\mathfrak{G}_V/K^{\times} \simeq \mathcal{O}(V)$ , so by Lemma 4.34, the Clifford norm induces a homomorphism  $O(V) \to K^{\times}/(K^{\times})^2$  which we denote by  $\mathcal{N}$ and call the Spinor norm.

*Example* 4.38. We find for an anisotropic vector  $x \in V$  and the corresponding reflection  $\tau_x$  that  $\mathcal{N}(\tau_x) \equiv \mathcal{N}(x) \equiv \mathcal{q}(x) \mod (K^{\times})^2$ .

Remark 4.39. We obtain an exact sequence

$$1 \longrightarrow \{\pm 1\} \longrightarrow \operatorname{Spin}_V \xrightarrow{v} \operatorname{SO}(V) \xrightarrow{\mathcal{N}} K^{\times}/(K^{\times})^2.$$

Note that if L is indefinite, then  $\operatorname{Spin}_{V_{\mathbb{R}}}$  is connected (cf. [PRR93, Prop. 7.6 p. 407]).

We end this subsection by describing  $\text{Spin}_{L_p}$  in more concrete terms, following [Att15, p. 20]. Let L be an R lattice of rank m with  $R \in \{\mathbb{Z}, \mathbb{Z}_p\}$  and basis  $(b_i)_i$ . Then the Clifford algebra  $\mathcal{C}_L$  has the basis

$$\{b_1^{e_1} \cdot \ldots \cdot b_m^{e_m}, | e_i \in \{0, 1\}\}$$

as an R module. The involution J identifies with the restriction from  $\mathcal{C}_V$  and the  $\mathbb{Z}/2\mathbb{Z}$ grading on it descends to  $C_L$ . Hence, it is possible to define the spin groups as in Definition 4.35, meaning in explicit terms

$$\operatorname{GSpin}_{L} \coloneqq \{ x \in \mathcal{C}_{L}^{0} \mid x \cdot J(x) \in \mathbb{R}^{\times}, xLJ(x)^{-1} = L \},$$

$$(4.10)$$

$$\operatorname{Spin}_{L} \coloneqq \{ x \in \mathcal{C}_{L}^{0} \mid x \cdot J(x) = 1, x L J(x)^{-1} = L \}.$$

$$(4.11)$$

#### Eichler transforms

We have already seen some examples of orthogonal transformations constructed via the vector representation of the Clifford group. However, there is another significant class that plays a key role for the existence of Fourier expansions of orthogonal modular forms – Eichler transforms.

**Definition 4.40.** For  $x \in L$  and  $y \in V_{\mathbb{R}}$  with q(x) = b(x, y) = 0 the vector representation of  $1 + xy \in \text{Spin}(V_{\mathbb{R}})$  is called the *Eichler transform* of x and y and is denoted by E(x, y). **Remark 4.41.** In explicit terms, we find for  $v \in V_{\mathbb{R}}$  the following formula

$$E(x,y)(v) = v - b(v,x)y + b(v,y)x - q(y)b(v,x)x.$$
(4.12)

In fact, a computation reveals that

$$v_{1+xy}(w) = (1+xy)wJ(1+xy)^{-1}$$

$$= (1 + xy)w(1 - xy)$$
$$= w - wxy + xyw - xywxy.$$

Note that wxy = (b(w, x) - xw)y = b(w, x)y - x(b(w, y) - yw), notice that [x, y] = 0and xx = q(x) = 0 to learn that

$$wxy = (b(w, x) - xw)y = b(w, x)y - x(b(w, y) + yw),$$
$$xyw = (b(x, y) - yx)w,$$
$$xywxy = x b(w, x) q(y).$$

This yields the desired result.

**Remark 4.42.** If  $x, y \in L$ , then E(x, y) restricts to an isometry of L. Additionally, E(x, y) also preserves  $v \in L'/L$  in this case.

Further, Eichler transforms have the following properties.

Lemma 4.43. Let x, y be as above.

- a) We find E(x, y)(x) = x.
- b) We find E(x, y)(y) = y + 2q(y)x.
- c) For  $y_1, y_2 \in x^{\perp}$  we find  $E(x, y_1) \circ E(x, y_2) = E(x, y_1 + y_2)$ .

## 4.2 Modular varieties

#### 4.2.1 Models

A fundamental reference for the following is [Bru02, p. 78], while [Bru+08, 2.4 p. 136] provides a brief and accessible overview.

This section will explore various ways in which orthogonal spaces that serve as domains for automorphic forms can be realised. As previously stated, (V, q) denotes a regular rational quadratic space of signature  $(m^+, m^-)$ . Let  $G = O(m^+, m^-)$  denote the orthogonal group of the real quadratic vector space  $V_{\mathbb{R}} = V \otimes \mathbb{R}$ . A canonical candidate for a domain to study modular forms on is the symmetric domain G/K, where K is a maximal compact subgroup. The resulting symmetric space is hermitian, if, and only if, either  $m^+$  or  $m^-$  is equal to 2. As a consequence, we shall restrict to assuming  $m^- = 2$  from Subsection 4.2.2 onwards.

#### The Grassmannian

We begin with a model providing a conveniently simple descritpion, the Grassmannian. It is a special case of a Flag manifold.

**Definition 4.44.** Let  $V_{\mathbb{R}}$  be a real quadratic space of signature  $(m^+, 2)$ . Then its Grassmannian is defined to be

$$\mathbb{D}(V_{\mathbb{R}}) \coloneqq \{ z \subseteq V_{\mathbb{R}} \mid \dim(z) = 2 \text{ and } q \mid_{z} < 0 \}.$$

$$(4.13)$$

We would like to realise  $\mathbb{D}$  as a quotient G/K for some maximal compact  $K \leq G$ . Note that by Witt's extension theorem (Theorem 4.6), the group  $O(V_{\mathbb{R}})$  acts transitively on  $\mathbb{D}$  and that any real orthogonal group associated to a definite space is compact (cf. Lemma 4.9). So fix  $z_0 \in \mathbb{D}$ , then  $K := O(z_0) \times O(z_0^{\perp}) \hookrightarrow O(V_{\mathbb{R}}) = G$  is included in the stabiliser of  $z_0$ . However, an element of  $O(V_{\mathbb{R}})$  leaving  $z_0$  invariant, must also leave  $z_0^{\perp}$  invariant, so that K is, in fact, identified to equal the stabiliser of  $z_0$ . Additionally, the group K is maximally compact among subgroups of G. Further,  $G/K \simeq \mathbb{D}$ , so that the Grassmannian might be equipped with a symmetric structure, inhertited from the quotient. Nevertheless, the clear advantage of the model is the apparent description, while the models below provide different additional insights.

#### The projective Model

Another model is induced through projective space and called the *projective model*. It offers the advantage of an immediate complex structure. Consider  $V_{\mathbb{C}} = \mathbb{C} \otimes_{\mathbb{Q}} V$ , which induces a projective variety

$$P(V_{\mathbb{C}}) = (V_{\mathbb{C}} \setminus \{0\}) / \mathbb{C}^{\times}.$$

Equipping this space with the final topology of the natural projection results in a compact space. Further, it naturally carries the structure of a complex analytic manifold. In fact, it is the standard example of such a space. We consider the closed subvariety

$$\mathscr{N} \coloneqq \{ [Z] \in P(V_{\mathbb{C}}) \mid \mathbf{b}(Z, Z) = 0 \},\$$

which, as a consequence, bears an analytic structure. It is called *zero quadric*. Note that elements  $Z \in V_{\mathbb{C}}$  may be decomposed into X + iY for  $X, Y \in V_{\mathbb{R}}$ , where the elements X, Y are referred to as the *real* and *imaginary part* of the vector z. As a consequence, we find

$$b(X + iY, X \pm iY) = b(X, X) \mp b(Y, Y) + i(b(X, Y) \pm b(X, Y))$$

yielding  $b(Z,\overline{Z}) = b(X,X) + b(Y,Y)$  and that b(Z,Z) = 0 is equivalent to b(X,X) = b(Y,Y) and b(X,Y) = 0. We consider the open subset of the zero quadratic

$$\mathcal{K} \coloneqq \{ [Z] \in P(V_{\mathbb{C}}) \mid b(Z, Z) = 0, b(Z, \overline{Z}) < 0 \}$$

which inherits the complex structure of  $\mathscr{N}$ . Note that the orthogonal group  $O(V_{\mathbb{R}})$  acts, by Witt's extension theorem (cf. Theorem 4.6), transitively on  $\mathcal{K}$ .

In the following, we reduce to the case  $m^- = 2$ . Then the space  $\mathcal{K}$  may be explicitly related to the Grassmannian.

**Proposition 4.45.** The assignment  $\mathcal{K} \ni [Z] \mapsto \mathbb{R}X + \mathbb{R}Y \in \mathbb{D}$  defines a continuous and open two to one map.

Proof: In fact, the real part X and imaginary part Y of a representative Z of  $[Z] \in \mathcal{K}$ fulfil q(X) = q(Y) < 0 and b(X, Y) = 0, as we have computed above. As a consequence, they form an orthogonal normed basis of an element of D. Note that the pair (X, -Y)associated to  $\overline{Z}$  gives rise to the same element in D, so that the mapping is at least two to one, where the ambiguity of the cover is given by the orientation of the basis that is considered. If there was another preimage, it had to arise from (X,Y) via an orthogonal transformation. If the transformation was special, it reduced to multiplying Z with an element of T, acting as the identity on  $\mathcal{N}$ , if it was not special, it reduced to  $(X,Y) \mapsto (X,-Y)$ .

Checking the topological conditions is left to the reader and plays no major role in the following exposition.  $\hfill \Box$ 

In fact, the space  $\mathcal{K}$  decomposes into two connected components which agree with the sheets of the twofold cover above. We fix one of these sheets and denote it by  $\mathcal{K}^+$ , the other by  $\mathcal{K}^-$ . Then the elements of  $O(V_{\mathbb{R}})$  preserving  $\mathcal{K}^+$  form a subgroup  $O^+(V_{\mathbb{R}}) < O(V_{\mathbb{R}})$ . This may be characterised as follows: recall that the Spinor norm  $\mathcal{N} : O(V_{\mathbb{R}}) \to \{\pm 1\}$  defines a homomorphism (cf. Remark 4.37) so that its product with the determinant map  $\mathcal{N} \cdot \det$  represents a homormophism, as well. The kernel of this map equals  $O^+(V_{\mathbb{R}})$ , which is open and closed in  $O(V_{\mathbb{R}})$ . Its complement  $O(V_{\mathbb{R}}) \setminus O^+(V_{\mathbb{R}})$  consists of these orthogonal transformations that interchange  $\mathcal{K}^+$  and  $\mathcal{K}^-$ .

The projective model provides a natural complex structure that may be exported to

the Grassmannian  $\mathbb{D}$  via Proposition 4.45. In the next section, we identify  $\mathcal{K}$  with a tube domain.

#### The tube domain model

The last model that is to be discussed is the tube domain model, reminiscent of the upper half plane. In the following, we restrict to the hermitian case of  $m^+ = 2$ . We choose a nonzero isotropic vector  $e_1 \in V$  and some  $e_2 \in V$  with  $b(e_1, e_2) = 1.^1$  Next, we define the quadratic space  $W = e_1^{\perp} \cap e_2^{\perp}$  which is of type (n - 1, 1) (meaning it is Lorentzian), so that

$$V = W \oplus \mathbb{Q}e_1 \oplus \mathbb{Q}e_2. \tag{4.14}$$

We write  $Z = (z, a, b) \in V_{\mathbb{C}}$  for an element in the associated complex space given by  $z + ae_1 + be_2$  with  $z \in W_{\mathbb{C}}$ .

Assume, we wanted to construct an element  $[Z] \in \mathcal{K}$  from some  $z \in W_{\mathbb{C}}$ . We recall that the conditions on such a representative Z for inducing an element in the zero quadric  $\mathscr{N}$ imply that for the decomposition Z = X + iY with  $X, Y \in V_{\mathbb{R}}$  we must have that X, Yhave equal negative weight and span a 2 dimensional subspace of  $V_{\mathbb{R}}$ . Hence, one of these must have an  $e_2$ -component, meaning  $b(Z, e_1) \neq 0$ . This prompts us to set (z, a, 1) as a candidate for Z, with a complex number  $a \in \mathbb{C}$  which has yet to be determined. If we now require  $[z + ae_1 + e_2] \in \mathscr{N}$ , we obtain by definition

$$0 = b((z, a, 1), (z, a, 1))$$
  
= 2 q(z) + 2a + 2 q(e<sub>2</sub>)

yielding  $a = -q(z) - q(e_2)$ . We realise that  $e_2$  could without loss of generality have been assumed to fulfil  $q(e_2) = 0$ , by replacing it with  $e_2 - q(e_2)e_1$ , since  $q(e_2 - q(e_2)e_1) = 0$ and  $b(e_1, e_2 - q(e_2)e_1) = b(e_1, e_2)$ . If we write z = x + iy with  $x, y \in W_{\mathbb{R}}$ , again, we find for the real and imaginary part

$$(z, -q(z) - q(e_2), 1) = (x, q(y) - q(x) - q(e_2), 1) + i(y, b(x, y), 0).$$

<sup>&</sup>lt;sup>1</sup>Should the rational space V be not isotropic, we may perform the construction in the real setting.

Now the second condition to lie in  $\mathcal{K}$  reads

$$0 > b((z, -q(z) - q(e_2), 1), (\overline{z}, -\overline{q(z)} - q(e_2), 1))$$
  
= 2(q(x) + q(y)) + 2(- Re(q(z)) - q(e\_2)) + 2q(e\_2) (4.15)  
= 4q(y).

Meaning lying in  $\mathcal{K}$  is equivalen to q(y) < 0. As a consequence, we restrict to the subspace

$$\mathcal{H}_{m^+}^{\pm} \coloneqq \{ z = x + iy \in W_{\mathbb{C}} \mid \mathbf{q}(y) < 0 \}, \tag{4.16}$$

which is an analogy of the union of the upper and lower halfplane. We see in particular, that the constructed association is injective, as  $(z, -q(z) - q(e_2), 1)$  is determined by z. In order to recall that given  $[Z] \in \mathcal{K}$ , we had seen that Z must have a nonzero  $e_2$ component, so that we may without loss of generality pick a representative Z = (z, a, 1)for some  $a \in \mathbb{C}$ . Above we had seen that this already implies  $a = -q(z) - q(e_2)$ , which results in a unique element  $z \in \mathcal{H}^{\pm}$ . In conclusion, we have proven the bijectivity result of the following proposition.

Proposition 4.46. The map

$$\mathcal{H}_{m^+}^{\pm} \ni z \mapsto [(z, -q(z) - q(e_2), 1)] \in \mathcal{K}$$

defines a biholomorphic association.

**Definition 4.47.** Define  $\mathcal{H}_{m^+}^+$ , to be the preimage of  $\mathcal{K}^+$  under the above isomorphism as the generalised upper half plane or tube domain model.

To emphasise the analogy to the upper half plane model in the elliptic case, we set

$$\mathcal{C} \coloneqq \{ y \in W_{\mathbb{R}} \mid \mathbf{q}(y) < 0 \}$$

and note that it possesses two connected components, each of which forms a cone. One component is given by

$$\mathcal{C}^+ \coloneqq \mathcal{H}_{m^+}^+ \cap i\mathcal{C},$$

so that

$$\mathcal{H}_{m^+}^+ = W_{\mathbb{R}} + i\mathcal{C}^+. \tag{4.17}$$

In the following the superscript + of the tube domain model will frequently be omitted.

The action of  $O^+(V_{\mathbb{R}})$  on  $\mathcal{K}^+$  may be pulled back to an action on the tube domain  $\mathcal{H}_{m^+}$ .

#### 4.2.2 Modular groups

Similarly to the classic case of elliptic modular forms, the base space will be constructed as a quotient of G/K by an arithmetic subgroup of  $O(V_{\mathbb{R}})$  and compactified. In the following, we consider lattices which comprise a central class of objects in this thesis as a foundation for the construction. Let (L, q) be an even  $\mathbb{Z}$  lattice of signature  $(m^+, m^-)$ . Recall that  $V = \mathbb{Q} \otimes_{\mathbb{Z}} L$  is the associated rational quadratic space.

Note that O(L) also acts on L'. In fact, let  $\ell \in V$ ,  $\gamma \in O(L)$  and  $(b_i)$  be a basis of L. Then  $(\gamma b_i)$  also constitutes a basis of L. Hence, we may test  $\gamma \ell$  against  $(\gamma b_i)$  for integrality. Writing  $\ell$  in the basis  $(b_i)$  yields that  $\gamma \ell \in L'$  if, and only if,  $\ell \in L'$ . As a consequence, we obtain a natural projection

$$O(L) \to \operatorname{Aut}(L'/L).$$
 (4.18)

Of course, the same is true for any subgroup of O(L), in particular the following definition is meaningful.

**Definition 4.48.** Let  $H \leq O(L)$  be a finite index subgroup. Then the kernel of the natural map obtained from restricting (4.18)

$$H \hookrightarrow \mathcal{O}(L) \to \operatorname{Aut}(L'/L)$$
 (4.19)

is called the *discriminant kernel* subgroup of H. It is denoted  $H_L$ . In case of the choice  $H = O^+(L)$  or  $SO^+(L)$ , also write  $\Gamma(L) = H_L$  or  $\Gamma_L = H_L$ .

Typically, the group H is chosen to be  $O^+(L) := O(L) \cap O^+(V)$ , the subgroup which preserves orientation or  $SO^+(L) = SO(L) \cap O^+(V)$ . These cases will be assumed if not specified further.

Note that the group  $\Gamma(L)$  has finite index in O(L). Next, recall that for any  $d \in \mathbb{Z} \setminus \{0\}$ the lattice L(d) was defined as  $(L, d \cdot q)$ , so that (L(d))' = (1/d)L'. The orthogonal group O(L(d)) agrees with O(L) if embedded canonically into  $GL(V_{\mathbb{R}})$ .<sup>2</sup> Further, we find the embedding  $L'/L \hookrightarrow L(d)'/L(d) = L(d)'/L$ . Clearly, any element  $\sigma \in Aut(L(d)'/L)$ induced from O(L) restricts to an element  $\sigma' \in Aut(L(d_2)'/L)$  for  $d_2 \mid d$ . As a special case, any element in the kernel of  $O(L) \to Aut(L(d)'/L)$  also lies in the kernel of  $O(L) \to$ 

<sup>&</sup>lt;sup>2</sup>To do so, realise L(d) as  $(\sqrt{d} \cdot L, q)$ .

 $\operatorname{Aut}(L(d_2)'/L)$ , meaning there is the following inclusion

$$\Gamma(L(d)) \le \Gamma(L(d_2)) \le \Gamma(L) \le \mathcal{O}^+(L) \le \mathcal{O}(L)$$
(4.20)

of subgroups of finite index.

Example 4.49. Consider the case of a hyperbolic plane H presented in Example 1.7. In the following, for a nonzero integer d, the discriminant kernel  $\Gamma(H(d))$  is computed. Assume we had an orthogonal transformation  $\gamma \in O(H)$ , described in coordinates as  $\gamma = \begin{pmatrix} b_1 & b_2 \\ b_3 & b_4 \end{pmatrix}$ . Then we find

$$\gamma^{T} H \gamma = \begin{pmatrix} 2 b_{1} b_{3} & b_{2} b_{3} + b_{1} b_{4} \\ b_{2} b_{3} + b_{1} b_{4} & 2 b_{2} b_{4} \end{pmatrix} = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}$$

From this equation we infer that in every column of the matrix representation of  $\gamma$ , an entry must vanish. Further, a whole row cannot vanish, as the side diagonal entries were contradicted (and bijectivity of  $\gamma$ ). So either  $b_1 = b_4 = 0$  or  $b_2 = b_3 = 0$ . Then the side diagonals read  $b_1b_4 = 1$  or  $b_2b_3 = 1$ . Since the entries will have to be integers, this may be reformulated as  $b_1 = b_4 = \pm 1$  and  $b_2 = b_3 = \pm 1$ . So the only orthogonal matrices are

$$\begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}, \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix}, \begin{pmatrix} -1 & 0 \\ 0 & -1 \end{pmatrix}, \begin{pmatrix} 0 & -1 \\ -1 & 0 \end{pmatrix}.$$

This group is clearly generated by the matrices  $-\mathcal{I}, H$ .

Recall that H was unimodular, so that  $L(d)' = \mathbb{Z}/d \cdot e_1 + \mathbb{Z}/d \cdot e_2$  and  $L(d)'/L(d) = L(d)'/L \simeq \mathbb{Z}/d\mathbb{Z} \times \mathbb{Z}/d\mathbb{Z}$  where the latter carries the quadratic form  $(x, y) \mapsto xy/d \in \mathbb{Q}/\mathbb{Z}$ . Hence, for any  $d \in \mathbb{Z}$ , the automorphisms induced by O(L(d)) = O(L) in Aut(L(d)'/L) are generated by the images of

$$-\mathcal{I}:(x,y)\mapsto (-x,-y)\in \mathbb{Z}/d\mathbb{Z}\times\mathbb{Z}/d\mathbb{Z},$$
$$H:(x,y)\mapsto (y,x)\in \mathbb{Z}/d\mathbb{Z}\times\mathbb{Z}/d\mathbb{Z}.$$

This results in the following discriminant kernel:

$$\Gamma(H(d)) = \begin{cases} \{\pm \mathcal{I}, \pm H\}, & d = \pm 1, \\ \{\pm \mathcal{I}\}, & d = \pm 2, \\ \{\mathcal{I}\}, & |d| > 2. \end{cases}$$
(4.21)

Remark 4.50. The discriminant kernel is, other than the orthogonal group, functorial

in the lattice. Explicitly, if there are two lattices  $M \leq L$ , then  $\Gamma(M) \leq \Gamma(L)$ . For the identification of elements in O(L) and O(M), these are embedded into  $O(L \otimes_{\mathbb{Z}} \mathbb{Q})$ .

Proof: Denote  $V = L \otimes_{\mathbb{Z}} \mathbb{Q} = M \otimes_{\mathbb{Z}} \mathbb{Q}$  and let  $\gamma \in \Gamma(M) \subset O(V)$ . Then  $\gamma$  acts trivially on M'/M. In particular, it acts trivially on the subgroup  $L/M \leq M'/M$ , meaning it must act on L, so that  $\gamma \in O(L)$ . However, the trivial action on M'/M is preserved under pushforward via  $M'/M \to M'/L$ . Conclusively, the element  $\gamma$  lies in O(L) and acts trivially on M'/L and hence on the subgroup  $L'/L \leq M'/L$ . Intersection with  $O^+(V_{\mathbb{R}})$ does not interfere with any of the presented arguments, implying  $\Gamma(M) \leq \Gamma(L)$ .  $\Box$ 

Note that  $\Gamma(M)$  in the above Remark has finite index in  $\Gamma(L)$  which follows from the fact that it must contain  $\Gamma(L(d))$  for some  $d \in \mathbb{N}$ , again by Remark 4.50. In analogue to the elliptic case, this prompts us to define the notion of a congruence subgroup.

**Definition 4.51.** Let *L* be as above and  $d \in \mathbb{N}$  be a natural number. Then  $\Gamma(L(d))$  is called its *d*-th *principal congruence* subgroup. Further, a subgroup  $\Gamma \leq O^+(L)$  is called a *congruence subgroup* with respect to *L*, if there is some  $d \in \mathbb{N}$  such that  $\Gamma(L(d)) \leq \Gamma$ .

Note that all of these groups have finite index in O(L) and that the notion of congruence subgroups depends solely on the associated rational quadratic space  $V = L \otimes_{\mathbb{Z}} \mathbb{Q}$ .

**Remark 4.52.** For the hermitian case of  $m^- = 2$ , there is a theorem stating that the opposite is also true, provided  $m^+ \ge 3$ . We will see below that this theorem does not hold in case of  $m^+ = 1$ .

In fact, there are some accidental isomorphisms that realise symplectic cases via orthogonal groups. Most notably the elliptic case may be realised via the O(1, 2) setting. *Example* 4.53. Assume V has signature (1, 2), then the associated real space has quadratic form  $x_1x_2 - x_0^2$  and we may realise it as

$$V_{\mathbb{R}} = \left\{ \begin{pmatrix} x_1 & x_0 \\ x_0 & x_2 \end{pmatrix} \middle| x_i \in \mathbb{R} \right\},\$$

so that  $q = \det$ . Then for  $g \in GL_2(\mathbb{R})$  the transformation  $V \ni v \mapsto gvg^T$  is orthogonal as long as  $\det(g) = \pm 1$  is presumed. The corresponding natural association defines an isogeny with kernel  $\pm \mathcal{I}$  which descends to an isogeny

$$\operatorname{SL}_2(\mathbb{R}) \to \operatorname{O}^+(V_{\mathbb{R}}).$$

The image has index 2 and adding the reflection  $v \mapsto -v$  to the image generates the whole group. The association

$$\mathbb{P}^{1}(\mathbb{C}) \ni \tau \mapsto \begin{cases} \begin{pmatrix} 1 & 0 \\ 0 & 0 \end{pmatrix}, & \text{if } \tau = \infty, \\ \begin{pmatrix} \tau^{2} & \tau \\ \tau & 1 \end{pmatrix}, & \text{else,} \end{cases}$$

defines a biholomorphic map onto the zero quadric  $\mathscr{N} \subset \mathbb{P}(V_{\mathbb{C}})$ . The induced action of  $\mathrm{PSL}_2(\mathbb{R})$  is given by Möbius transformations. In fact, let  $\gamma \in \mathrm{SL}_2(\mathbb{R})$ , then

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} \cdot \begin{pmatrix} \tau^2 & \tau \\ \tau & 1 \end{pmatrix} \cdot \begin{pmatrix} a & c \\ b & d \end{pmatrix} = \begin{pmatrix} (b+a\tau)^2 & (b+a\tau)(d+c\tau) \\ (b+a\tau)(d+c\tau) & (d+c\tau)^2 \end{pmatrix}$$
$$\simeq \begin{pmatrix} (\gamma\tau)^2 & \gamma\tau \\ \gamma\tau & 1 \end{pmatrix}.$$

The space  $\mathcal{K}$  in that model identifies with the upper and lower half plane, where  $\mathcal{K}^+$  represents exactly the upper (or lower) half plane. In fact, the condition to lie in  $\mathcal{K}$ , namely  $b(\tau, \overline{\tau}) < 0$ , reads

$$0 > (\tau, \tau^2, 1) \begin{pmatrix} -2 & \\ & 1 \\ & 1 \end{pmatrix} \begin{pmatrix} \overline{\tau} \\ \overline{\tau^2} \\ 1 \end{pmatrix} = -2|\tau|^2 + \tau^2 + \overline{\tau}^2 = -4y^2 \iff y \neq 0.$$

The discriminant kernel of L may also be realised over  $\operatorname{GSpin}_L$ . In fact, we have the following theorem.

**Theorem 4.54** ([Att15, Cor 2.5.3]). Let (L, q) be a non-degenerate *R*-lattice. Then the image of GSpin(*L*) in SO(*L*) under the natural map equals  $\Gamma(L)$ , which is the discriminant kernel of SO<sup>+</sup>(*L*), where SO<sup>+</sup>(*L*) denotes the intersection of SO(*L*) with SO<sup>+</sup>( $L \otimes \mathbb{R}$ ) if  $R = \mathbb{Z}$  or SO<sup>+</sup>( $L \otimes \mathbb{Q}_p$ ) if  $R = \mathbb{Z}_p$ .

We will often consider the following variety in analogue to the classical case.

**Definition 4.55.** Let  $\Gamma \leq \Gamma(L)$  be a subgroup of finite index. Then denote  $\Gamma \setminus \mathbb{D}$  by  $Y_{\Gamma}$ . In case of  $\Gamma = \Gamma(L)$ , we also write  $Y_L = Y_{\Gamma(L)}$ .

#### 4.2.3 Boundary components

Additional background material on the subject is found in the base source [BF01] for this subsection and [Kie21].

Throughout this section we shall assume  $m^- = 2$ , so that L is a lattice of signature  $(m^+, 2)$ . Boundary components of G/K are represented by isotropic spaces and the projective model is relatively accessible for the purpose of describing boundary components. Hence, we consider the boundary of  $\mathcal{K}^+$  in the zero quadric  $\mathcal{N}$ .

In fact, the boundary points are described as follows. If a point in  $[Z] \in \mathcal{N}$  should be a boundary point of  $\mathcal{K}$ , its representative Z = X + iY has to fulfil q(X) = q(Y) = 0 and b(X, Y) = 0. This might happen, if the vectors X and Y are collinear or not, giving rise to a one dimensional or respectively two dimensional isotropic subspace. In fact, X and Y are collinear if, and only if, [X + iY] may be represented by a real vector (namely either X or Y). Otherwise, there is no real representation. These points are then collected into components.

Definition 4.56. We have the following classification of boundary points.

- a) If the space spanned by X, Y is one dimensional, we call [Z] a special boundary point.
- b) If the space spanned by X, Y is two dimensional, we call [Z] a generic boundary point.
- c) A set consisting of one special boundary point is called zero dimensional *boundary* component.
- d) For an isotropic plane  $I \subset V_{\mathbb{R}}$ , the collection of generic boundary points which may be represented by elements in  $I \otimes_{\mathbb{R}} \mathbb{C}$  is called one dimensional *boundary component*.

The reason for bundling these points is due to the following structural result about one dimensional boundary components.

**Remark 4.57.** There is a bijective correspondence between boundary components and isotropic subspaces of  $V_{\mathbb{R}}$ . Moreover, one dimensional boundary components are isomorphic to upper half planes  $\mathbb{H}$ .

In fact, the image of  $I \otimes_{\mathbb{R}} \mathbb{C}$  in  $\mathscr{N}$  for a two dimensional isotropic space  $I \subset V(\mathbb{R})$  may be identified with  $\mathbb{H} \cup \mathbb{R} \cup \{\infty\}$  as in the classical case. **Definition 4.58.** A boundary component is called *rational*, if the corresponding isotropic space I is defined over  $\mathbb{Q}$ , meaning for  $[Z] \in \mathcal{N}$  that there is a representative Z = X + iY, such that  $X, Y \in V_{\mathbb{Q}} \subset V_{\mathbb{R}}$ .

*Example* 4.59. Assume  $V_{\mathbb{Q}}$  is anisotropic and  $\Gamma < O^+(V)$  is commensurable with O(L). Then the rational boundary of  $Y_{\Gamma}$  is empty and  $X_{\Gamma} = Y_{\Gamma}$  is compact.

#### 4.2.4 Modular forms

The following content is essentially contained in [Bru02, 3.3 p. 84]. Another reference providing a brief overview is [Kie22].

Similar to the elliptic case, we will describe modular forms, which may be interpreted as global sections of a hermitian line bundle over the modular variety  $Y(\Gamma)$ , as functions on the cover G/K of that space with a certain transformation property. In order to state this transformation property analogously to the classical case, we are required to introduce the factor of automorphy in the orthogonal setting.

Recall that there is a natural action of  $O^+(V_{\mathbb{R}})$  on the projective model  $\mathcal{K}$  of G/K, which is transferred by the biholomorphic association discussed above to the tube domain model  $\mathcal{H}_{m^+}$ . We use the notation of the preceding subsections and select for  $z \in \mathcal{H}_{m^+}$  a specific representative of its corresponding element  $[Z] \in \mathcal{K}^+$  following Proposition 4.46. In fact, selecting  $V_{\mathbb{C}} \ni \tilde{z} = z - (q(z) + q(e_2))e_1 + e_2$  gives rise to an embedding of  $\mathcal{H}_{m^+}$ into the preimage of  $\mathcal{K}^+$  in  $V_{\mathbb{C}}$ . The latter is exactly the cone over the chosen embedding and will be denoted  $\tilde{\mathcal{K}}^+$ .

Starting from the perspective of the cone, it is relatively straightforward how to define modular functions in analogy to the classical case:

**Definition 4.60.** A function  $F : \tilde{\mathcal{K}}^+ \to \mathbb{C}$  is called *modular* of weight  $r \in \mathbb{Z}$  with respect to a congruence subgroup  $\Gamma \leq \Gamma(L)$  if it satisfies the following two properties.

- a) For all  $t \in \mathbb{C}^{\times}$  the function is homogeneous of degree -r, meaning  $F(tZ) = t^{-r}F(Z)$ .
- b) For all  $\gamma \in \Gamma(L)$  the function is invariant under pullback:  $F(\gamma Z) = F(Z)$ .

From that definition, it is immediate how to generalise it to modular functions with a character. In fact, let  $\chi : \Gamma \to \mathbb{T}$  ba a character, then b) may be generalised to read  $F(\gamma Z) = \chi(\gamma)F(Z)$ . In order to translate that action to the function on the embedded tube domain  $\mathcal{H}_{m^+}$ , we need to introduce the factor of automorphy for which we will have to consider the action of  $O^+(V_{\mathbb{R}})$  that is transferred from the canonical action on  $\mathcal{K}$ . We find for  $\gamma \in O^+(V_{\mathbb{R}})$  and  $z \in \mathcal{H}_{m^+}$  with corresponding lift  $\tilde{z} \in \tilde{\mathcal{K}}^+$  (cf. Proposition 4.46) that<sup>3</sup>

$$\gamma(\tilde{z}) = z' - ae_1 + b(e_2 - q(e_2)e_1)$$

for some  $a, b \in \mathbb{R}$  and another  $z' \in \mathcal{H}_{m^+}$ . By definition, we obtain via normalising that the preimage of  $[\gamma(\tilde{z})]$  in  $\mathcal{H}_{m^+}$  equals

$$\gamma(z) = b^{-1} z'$$
 where  $b = b(\gamma(\tilde{z}), e_1).$ 

If we repeat the procedure by letting another  $\gamma_2 \in O^+(V_{\mathbb{R}})$  act on z', we find that the new prefactor b' (in place of b) fulfils a cocycle relation. This prompts us to define.

**Definition 4.61.** Let  $\gamma \in O^+(V_{\mathbb{R}})$  and  $z \in \mathcal{H}_{m^+}$  corresponding to  $\tilde{z}$  as above. Then the *factor of automorphy* is defined to be

$$j(\gamma, z) \coloneqq \mathbf{b}(\gamma(\tilde{z}), e_1).$$

Lemma 4.62. The factor of automorphy fulfils

$$j(\gamma, z)(\widetilde{\gamma z}) = \gamma(\tilde{z}) \tag{4.22}$$

as well as the cocycle relation

$$j(\gamma_1 \gamma_2, z) = j(\gamma_1, \gamma_2 z) j(\gamma_2, z).$$
 (4.23)

In addition, the imaginary part of an element  $z \in \mathcal{H}_{m^+}$  transforms with the factor of automorphy in analogy to the elliptic case presented in (2.12).

**Remark 4.63.** For  $\gamma \in O^+(V_{\mathbb{R}})$  and  $z \in \mathcal{H}_{m^+}$ , we have the following transformation property

$$q(\operatorname{Im}(\gamma z)) = \frac{q(\operatorname{Im}(z))}{|j(\gamma, z)|^2}$$
(4.24)

*Proof*: By (4.15) reading  $4q(y) = b(\tilde{z}, \tilde{\overline{z}})$  and (4.22) we obtain

$$q(\text{Im}(\gamma z)) = \frac{1}{4} \left( \frac{\gamma(\tilde{z})}{\mathbf{b}(\gamma \tilde{z}, e_1)}, \frac{\gamma(\tilde{z})}{\mathbf{b}(\gamma \tilde{z}, e_1)} \right)$$
$$= \frac{1}{4 |\mathbf{b}(\gamma \tilde{z}, e_1)|^2} \left( \gamma \tilde{z}, \gamma \tilde{z} \right)$$

<sup>&</sup>lt;sup>3</sup>Recall that we may replace  $e_2$  by  $e_2 - q(e_2)e_1$  to guarantee  $q(e_2) = 0$ .

$$= \frac{q(\operatorname{Im}(z))}{|\mathbf{b}(\gamma \tilde{z}, e_1)|^2}.$$

In the light of Lemma 4.62, we may translate Definition 4.60 to the tube domain model. To be explicit, let

$$F_{\mathcal{H}}(z): \mathcal{H}_{m^+} \to \mathbb{C}, \qquad z \mapsto F(\tilde{z}) = F(z - (q(z) + q(e_2))e_1 + e_2).$$

Then  $F_{\mathcal{H}}$  satisfies the following transformation property for  $\gamma \in \Gamma$  (cf. (4.22))

$$F_{\mathcal{H}}(\gamma z) = F(\widetilde{\gamma z})$$

$$= F(j(\gamma, z)^{-1}\gamma \widetilde{z})$$

$$= j(\gamma, z)^{r}\chi(\gamma)F(\widetilde{z})$$

$$= j(\gamma, z)^{r}\chi(\gamma)F_{\mathcal{H}}(z)$$
(4.25)

and we see that there is a bijective correspondence between such functions.

Technically, (4.25) already yields the notion of a modular function on  $\mathcal{H}_{m^+}$ . However, as we will consider the Borcherds lift in Section 7.1, we require a more general notion of modular forms than these; namely to characters and have to introduce multiplier systems. To this end, let  $\ln(j(\gamma, z))$  denote a fixed logarithm of  $j(\gamma, z)$ . For a rational number  $r \in \mathbb{Q}$ , set

$$j(\gamma, z)^r \coloneqq \exp\left(r \cdot \ln(j(\gamma, z))\right)$$
.

Then there is a map  $w_r : \mathcal{O}^+(V_{\mathbb{R}}) \times \mathcal{O}^+(V_{\mathbb{R}}) \to \mathbb{T}$  restricting to the roots of unity of order bounded by den(r) such that

$$j(\gamma_1\gamma_2, Z)^r = w_r(\gamma_1, \gamma_2) \cdot j(\gamma_1, \gamma_2 z)^r j(\gamma_2, z)^r$$

which only depends on  $r \mod \mathbb{Z}$ . With this preparation, we may replace the notion of a character with a more general notion, satisfying the same transformation property to cancel the above factor when transforming.

**Definition 4.64.** Let  $\Gamma \leq O^+(V_{\mathbb{R}})$  and  $r \in \mathbb{Q}$ . A multiplier system of weight r for  $\Gamma$  is a map

$$\chi: \Gamma \to \mathbb{T}, \quad \text{s.t.} \quad \chi(\gamma_1 \gamma_2) = w_r(\gamma_1, \gamma_2)^{-1} \cdot \chi(\gamma_1) \chi(\gamma_2)$$

With the notion of a multiplier system, modular forms on the tube domain  $\mathcal{H}_{m^+}$  may

be defined. For simplicity we assume  $m^+ \ge 3$  first, in order to simplify to the following definition.

**Definition 4.65.** Assume  $m^+ \geq 3$ . Let  $\Gamma \leq \Gamma(L)$  be a subgroup of finite index and  $\chi$  be a multiplier system for  $\Gamma$  of weight  $r \in \mathbb{Q}$ . A holomorphic function F on  $\mathcal{H}_{m^+}$  is called a *holomorphic modular function* of weight r and multiplier system  $\chi$  with respect to  $\Gamma$ , if it fulfils

$$F(\gamma z) = \chi(\gamma)j(\gamma, z)^r F(z)$$
(4.26)

for all  $\gamma \in \Gamma$ . Write  $\mathcal{M}_r(\Gamma, \chi)$  for the space of such functions.

Recall that in the classical case, there has been a growth condition towards the boundary. In case of a holomorphic modular function this condition is superfluous by the Köcher principle (cf. [Bru02, Prop. 4.15 p. 109]) in case of  $m^+ \ge 3$ . If  $m^+ = 2$ , however, the situation is more convoluted and we would like to briefly touch on how to proceed. Assuming  $m^+ = 2$  and that there is no isotropic element in V, the quotient  $\Gamma \setminus \mathbb{D}$  is compact and we may stick with Definition 4.60 for integral weight  $r \in \mathbb{Z}$  and impose holomorphicity or meromorphicity.<sup>4</sup> In case there is an isotropic element, we are required to speak about vanishing at the cusps. For this instance, we sketch how to derive a Fourier expansion of such functions.<sup>5</sup>

Recall for  $x, y \in L$  the Eichler transform E(x, y) of Definition 4.40. Then by Remark 4.42, we find that the Eichler transform  $E(x, y) \in \operatorname{Aut}(L)$  is contained in  $\Gamma(L)$ . In fact, for  $d \in \mathbb{N}$  and  $y \in d \cdot L$ , we even find  $E(x, y) \in \Gamma(L(d))$ . These transformations may be used to associate a Fourier expansion to a modular function. For that purpose, recall the construction of the tube domain model, in particular of the subspace  $W_{\mathbb{R}} < V_{\mathbb{R}}$  whose rational points are described in (4.14). Instead of constructing W directly as a vector space, we may, as long as  $V_{\mathbb{Q}}$  is isotropic, select a primitive isotropic vector  $e_1 \in L$  and choose  $e_2 \in L'$  such that  $b(e_1, e_2) = 1$ . Then the lattice  $M \coloneqq L \cap e_1^{\perp} \cap e_2^{\perp}$  is a Lorentzian, i.e. it has signature  $(m^+ - 1, 1)$ . If we set  $W_{\mathbb{R}} \coloneqq K \otimes \mathbb{R}$ , we obtain the same construction as for the tube domain model. The difference is that this time we have a lattice  $M \subset L$ as a base of the construction.

Select  $z \in \mathcal{H}_{m^+}$  with associated element  $\tilde{z} = z + ae_1 + e_2 \in V_{\mathbb{C}}$ .

<sup>&</sup>lt;sup>4</sup>In case of non-integral weight  $r \in \mathbb{Q} \setminus \mathbb{Z}$  one has to consider covers of  $\mathcal{K}^+$ , based on the denominator or r.

<sup>&</sup>lt;sup>5</sup>This is also valid in case of  $m^+ \ge 3$ , as there will always be isotropic vectors by Meyer's theorem.

Next, notice that for  $l \in M$  we find that  $E(e_1, l)$  acts as follows (cf. (4.12))

$$E(e_1, l)(\tilde{z}) = \tilde{z} - b(\tilde{z}, e_1)l + b(\tilde{z}, l)e_1 - q(l)b(\tilde{z}, e_1)e_1$$
  
=  $\tilde{z} - l + b(z, l)e_1 - q(l)e_1$   
=  $(z - l) + (a + b(z, l) - q(l))e_1 + e_2.$ 

Hence, the induced action of  $E(e_1, l)$  on  $\mathcal{H}_{m^+}$  is given by  $z \mapsto z - l$  and we obtain for the corresponding factor of automorphy that  $j(E(e_1, l), z) = b(E(e_1, l)\tilde{z}, e_1) = 1$ . As a consequence, every holomorphic function  $F : \mathcal{H}_{m^+} \to \mathbb{C}$  transforming with respect to  $\Gamma(L(d))$  as in (4.26) for some  $d \in \mathbb{N}$  possesses a Fourier expansion

$$F(z) = \sum_{\lambda \in \rho + K'/d} c(\lambda) e(\mathbf{b}(\lambda, z))$$
(4.27)

for some  $\rho \in M \otimes \mathbb{Q} = W_{\mathbb{Q}}$  which is unique modulo K'/d.

With this preparation, one may define modular forms also in case  $m^+ \geq 3$ .

**Definition 4.66.** Let V be isotropic. Let  $\Gamma \leq \Gamma(L)$  be a subgroup of finite index and  $\chi$  be a multiplier system for  $\Gamma$  of weight  $r \in \mathbb{Q}$ . A holomorphic function F on  $\mathcal{H}_{m^+}$  is called a *holomorphic modular function* of weight r and multiplier system  $\chi$  with respect to  $\Gamma$ , if it fulfils

$$F(\gamma z) = \chi(\gamma)j(\gamma, z)^r F(z)$$
(4.28)

for all  $\gamma \in \Gamma$  and for the associated Fourier expansion (4.27) the coefficient  $c(\lambda)$  vanishes, unless their index  $\lambda$  lies in  $\mathcal{C}^+$ . We write  $\mathcal{M}_r(\Gamma, \chi)$  for the space of such functions.

As already alluded to, the second condition stated above about coefficients vanishing unless being contained in the cone manifesting the imaginary part of the tube domain model is automatic in case of  $m^+ \geq 3$  by the Köcher Principle [Bru02, Prop 4.15 p. 109]. With the notion of holomorphic modular forms, the meromorphic version may be defined in a straightforward fashion.

**Definition 4.67.** Let V be isotropic. An orthogonal meromorphic modular form with respect to the group  $\Gamma \leq \Gamma(L)$  of finite index of weight  $r \in \mathbb{Q}$  and multiplier system  $\chi$  is a function F on  $\mathcal{H}_{m^+}$  with transformation property

$$F(\gamma z) = \chi(\gamma)j(\gamma, z)^r F(z)$$

for all  $\gamma \in \Gamma$  that may be written as a quotient  $F = F_1/F_2$  with  $F_1 \in \mathcal{M}_s(\Gamma, \chi_1)$  and  $F_2 \in \mathcal{M}_t(\Gamma, \chi_2)$  for suitable parameters  $s, t \in \mathbb{Q}$  and multiplier systems  $\chi_1, \chi_2$ .

Analogously to the classical case presented in Chapter 2, there is a natural class of modular forms that may be constructed to prove the existence of such forms (cf. Subsection 2.1.2). To this end, we define, in analogy to the classical case, the following operator.

**Definition 4.68.** Let f be a function on  $\mathcal{H}_{m^+}$  and  $k \in \mathbb{Z}$ . Then for  $\gamma \in \Gamma(L)$ 

$$f|_k \gamma(z) \coloneqq j(\gamma, z)^{-k} f(\gamma z)$$

defines a function on  $\mathcal{H}_{m^+}$  and the induced right operation of  $\Gamma(L)$  preserves holomorphicity and meromorphicity, respectively. The operator  $|_k$  is referred to as *Petersson operator* of weight k or slash operator.

As in the elliptic setting, the Petersson operator leaves the space of modular forms invariant.

An example of orthogonal modular forms has yet to be constructed! Recall Eisenstein series in the classical setting introduced in Definition 2.16. These are constructed to be the simplest possible instance of a  $\Gamma(1)$  invariant function, by symmetrising the constant function with respect to the Petersson operator. The same procedure will be mimicked in the orthogonal setting.

**Definition 4.69.** Let  $k \in \mathbb{Z}$ ,  $z \in L$  be a cusp, i.e. a primitive isotropic vector, and  $\Gamma \leq \Gamma(L)$  be a subgroup of finite index. Then

$$E_{k,z}(Z) \coloneqq \sum_{\sigma \in \Gamma} 1|_k \sigma$$

as a function on  $\mathcal{K}^+$  is called *Eisenstein series*.

The series converges for k > l + 2 by comparison to an Epstein zeta function. As in the elliptic setting non-holomorphic Eisenstein series may be defined. We will not require these in the following, but the curious reader may consult [Kie22, 7 p. 2867]. Further, non trivial examples of modular forms may be constructed via the Borcherds lift, relating weakly holomorphic vector valued elliptic modular forms to meromorphic orthogonal modular forms (cf. Section 7.1).

#### 4.2.5 Special divisors

In algebraic geometry divisors are a common tool for investigating the structure of varieties. A feature of the orthogonal setting is that the associated Shimura varieties have an abundance of divisors being induced by sub-varieties of the same type in all codimensions. These give rise to so called *special cycles* which will play a key role in proving the injectivity of the Kudla–Millson lift in Section 7.3. In the following, we describe the construction of these codimension 1 special cycles.

Recall that we assume the rational quadratic space (V, q) to have signature  $(m^+, 2)$  and select some  $v \in V$  with q(v) > 0. Then the subspace  $V_v \coloneqq v^{\perp}$  has signature  $(m^+ - 1, 2)$ and the complement in the projective model  $\mathcal{K}^+$  denoted by

$$\mathcal{Z}_{v} \coloneqq \left\{ [z] \in \mathcal{K}^{+} \mid (z, v) = 0 \right\}$$

$$(4.29)$$

is an analytic divisor on  $\mathcal{K}^+$ , which is exactly the hermitian symmetric domain corresponding to  $(V_v, \mathbf{q})$ . This means

$$\mathcal{Z}_v \simeq \mathbb{D}_v \coloneqq \{ U \subset V_v(\mathbb{R}) \mid \dim(U) = 2 \text{ and } q \mid_U < 0 \}.$$

$$(4.30)$$

Here,  $\mathbb{D}_v$  can be realised as the Grassmannian attached to the stabiliser  $O(V_{\mathbb{R}})_v$  of v in  $O(V_{\mathbb{R}})$ . It has become apparent that the divisor  $\mathcal{Z}_v$  coincides with a subsymmetric domain of the same type and these divisors are referred to as *special divisors*. Its corresponding description in the tube domain model looks as follows. Write  $v = v_W + ae_2 + be_1$  as above (cf. (4.14)) so that

$$\mathcal{Z}_v \simeq \left\{ z \in \mathcal{H}^+ \mid a \operatorname{q}(z) - \operatorname{b}(z, v_W) - a \operatorname{q}(e_2) - b = 0 \right\} \subset \mathcal{H}^+.$$

This description gives rise to the term *rational quadratic divisor*. Further, the following combination of these divisors appears naturally in many applications (cf. Theorem 7.3).

**Definition 4.70.** Select  $\lambda \in L'/L$  as well as a rational number n > 0. Then

$$Z(\lambda, n) \coloneqq \sum_{\substack{v \in \lambda + L \\ q(v) = n}} \mathcal{Z}_v \tag{4.31}$$

defines an analytic divisor on  $\mathcal{K}^+$  which is called *Heegner divisor* of discriminant  $(\lambda, n)$ .

Recall that arithmetic subgroups  $\Gamma \leq O(L)$  have been considered which give rise to

modular varieties  $Y_{\Gamma} = \Gamma \setminus \mathbb{D}$ . If the arithmetic group  $\Gamma$  lies in the discriminant kernel, i.e. acts trivially on the discriminant group L'/L, then the Heegner divisor descends to an algebraic divisor on  $Y_{\Gamma}$ , which is also denoted by  $Z(\lambda, n)$ . This may be verified by Chow's lemma.

Alternatively, the Heegner divisors might be symmetrised over L'/L to yield an algebraic divisor on  $Y_{\Gamma}$ , namely

$$Z(m) \coloneqq \frac{1}{2} \sum_{\lambda \in L'/L} Z(\lambda, m).$$
(4.32)

We also require the following more primitive divisor which is directly induced by (4.29). Let  $\Gamma_v := \Gamma \cap O(V_{\mathbb{R}})_v$  be the intersection with the stabiliser of v. Then we find that

$$\mathcal{Z}(v) \coloneqq \Gamma_v \backslash \mathbb{D}_v \to Y_\Gamma \tag{4.33}$$

defines a (in general relative) cycle. These cycles play a fundamental role in the proof of the injectivity of the Kudla–Millson lift in Section 7.3.1.

The existence of such a family of algebraic divisors is distinctive for orthogonal and unitary groups.

# 5 Automorphic forms on $SL_2$

In the preceding sections, the concept of modular forms has been introduced in classical terms, both in the orthogonal and elliptic setting. In particular, Eisenstein series and theta functions have been discussed, both playing a central role in the theory of automorphic forms. Furthermore, there is a rather representation theoretic perspective on modular forms involving the adelic points of the respective group. An introduction to this perspective is presented in [Dei10], while a more thorough discussion can be found in [Bum98] and [Gel+90]. We restrict ourselves to describing the analogous versions of classical theta functions and Eisenstein series on  $SL_2(\mathbb{A})$  in an ad hoc fashion and encourage the reader to consult the literature for additional background information.

The following sections introduce theta distributions 5.2 and Eisenstein series 5.3 associated to Schwartz–Bruhat functions are introduced before relating both to each other by means of the Siegel–Weil formula (5.4.1).

### 5.1 Preliminaries

Before introducing examples of automorphic forms, we will have to state some facts about the group  $SL_2(\mathbb{R})$  based on [Dei10, Chap. 3 p. 81]. Recall the following three embeddings into  $SL_2(\mathbb{R})$ :

$$m: (\mathbb{R}^{\times}, \cdot) \ni a \mapsto m(a) \coloneqq \begin{pmatrix} a & 0\\ 0 & a^{-1} \end{pmatrix},$$
(5.1)

$$n: (\mathbb{R}, +) \ni b \mapsto n(b) \coloneqq \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix}, \tag{5.2}$$

$$k: \mathbb{T} \ni \vartheta \mapsto k(\vartheta) \coloneqq \begin{pmatrix} \cos(\vartheta) & \sin(\vartheta) \\ -\sin(\vartheta) & \cos(\vartheta) \end{pmatrix}.$$
(5.3)

The image of m equals the subgroup of diagonal matrices in  $SL_2(\mathbb{R})$ . We will denote it by M and the subgroup  $m(\mathbb{R}_{>0})$  by  $M^0$ . Further, the image of n will be denoted by N and

the one of k, which equals  $SO_2(\mathbb{R})$ , will also be denoted by  $K_{\infty}$ .

*Example* 5.1. a) For  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{R})$  we consider the action on  $i \in \mathbb{H}$  and compute

$$\gamma i = \frac{ai+b}{ci+d} = \frac{ac+bd+i(ad-bc)}{c^2+d^2} = \frac{ac+bd}{c^2+d^2} + i\frac{1}{c^2+d^2}.$$
(5.4)

b) For  $a \in \mathbb{R}^{\times}$  and  $\tau \in \mathbb{H}$  we find

$$m(a)\tau = a^2\tau.$$

c) For  $b \in \mathbb{R}$  and  $\tau \in \mathbb{H}$  we find

$$n(b)\tau = \tau + b$$

d) Combining the last two instances, we obtain for  $u + iv = \tau \in \mathbb{H}$  that

$$n(u)m(\sqrt{v})i = n(u)iv = u + iv = \tau.$$
 (5.5)

The example above prompts the following definition.

**Definition 5.2.** Let  $u + iv = \tau \in \mathbb{H}$  and define the associated matrix  $g_{\tau} \coloneqq n(u)m(\sqrt{v}) \in$ SL<sub>2</sub>( $\mathbb{R}$ ), acting as  $g_{\tau}i = \tau$ .

With these notions, the *Iwasawa* decomposition of the group  $SL_2(\mathbb{R})$  may be constructed. Consider the stabiliser of  $i \in \mathbb{H}$  with respect to the action  $SL_2(\mathbb{R}) \oplus \mathbb{H}$  via Möbius transformations. A swift computation yields that  $Stab(i) = SO_2(\mathbb{R})$ . Recall that by Example 5.1 the action of  $SL_2(\mathbb{R})$  is transitive and we obtain the desired decomposition.

**Proposition 5.3** (Iwasawa decomposition). Let  $M^0$ , N, and  $K_{\infty} = \text{Stab}(i) = \text{SO}_2(\mathbb{R})$  be as above. Then

 $N \times M^0 \times K_{\infty} \to \mathrm{SL}_2(\mathbb{R}), \qquad (n, m, k) \mapsto nmk$ 

defines a homeomorphism.

*Proof*: In fact, we find that associating the following matrices to  $\begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{R})$  defines a topological inverse mapping.

$$m = \begin{pmatrix} \sqrt{c^2 + d^2}^{-1} & 0\\ 0 & \sqrt{c^2 + d^2} \end{pmatrix},$$
$$n = \begin{pmatrix} 1 & \frac{ac + bd}{c^2 + d^2}\\ 0 & 1 \end{pmatrix},$$
$$k = \frac{1}{\sqrt{c^2 + d^2}} \begin{pmatrix} d & -c \\ c & d \end{pmatrix}.$$

For constructing these matrices, we have employed (5.4) and (5.5).

We will also require the interplay of the factor of automorphy with the special matrices appearing in the Iwasawa decomposition.

*Example* 5.4. Let  $a \in \mathbb{R}^{\times}$ ,  $b \in \mathbb{R}$ ,  $\vartheta \in \mathbb{T}$ , and  $\tau \in \mathbb{C} \setminus \mathbb{R}$ . Then we find

$$j(m(a),\tau) = a^{-1},$$
(5.6)

$$j(n(b), \tau) = 1,$$
 (5.7)

$$j(k_{\vartheta}, i) = e^{-i\vartheta}.$$
(5.8)

Further, combining these with the cocycle relation (cf. Remark 2.11), we obtain

$$\begin{aligned} j(n(b)m(a)k_{\vartheta},i) &= j(n(b)m(a),i)j(k_{\vartheta},i) \\ &= j(n(b),m(a)i)j(m(a),i)j(k_{\vartheta},i) \\ &= a^{-1}e^{-i\vartheta}. \end{aligned}$$

For the purpose of explicitly constructing Eisenstein series in Section 5.3, we are required to express the factor of automorphy of the constitute k of an element  $\gamma \in SL_2(\mathbb{R})$  in Proposition 5.3 in terms of entries of  $\gamma$  and refer to [BY09, p. 641] on that matter.

**Remark 5.5.** Let  $\gamma = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in SL_2(\mathbb{R})$  and  $\tau \in \mathbb{H}$ . Then  $\gamma \tau = \gamma g_\tau i$ , where the Iwasawa decomposition guarantees  $\gamma g_\tau = g_{\gamma\tau} k_\vartheta = n(\beta)m(\alpha)k_\vartheta$  for suitable  $\beta, \alpha$  and  $\vartheta$ . In fact, we find

$$e^{i\vartheta} = \overline{j(k_{\vartheta},i)} = \frac{j(\gamma,\overline{\tau})}{|j(\gamma,\tau)|},\tag{5.9}$$

$$\alpha = \frac{\sqrt{v}}{|j(\gamma,\tau)|} = \sqrt{\operatorname{Im}(\gamma\tau)}.$$
(5.10)

*Proof*: Recall that the first equation of (5.9) has already been given in Example 5.4. For the second, we begin by calculating

$$\begin{pmatrix} a & b \\ c & d \end{pmatrix} g_{\tau} = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \begin{pmatrix} \sqrt{v} & u/\sqrt{v} \\ 0 & 1/\sqrt{v} \end{pmatrix} = \begin{pmatrix} \sqrt{v}a & (ua+b)/\sqrt{v} \\ \sqrt{v}c & (uc+d)/\sqrt{v} \end{pmatrix}$$
(5.11)

yielding by the Iwasawa decomposition (cf. proof of Prop. 5.3) as an associated matrix

$$k_{\vartheta} = \frac{1}{\sqrt{(cv)^2 + (uc+d)^2}} \begin{pmatrix} (uc+d) & -cv\\ cv & (uc+d) \end{pmatrix}$$

With this expression, we obtain by inserting the definition of the factor of automorphy and by writing  $\tau = u + iv$  that

$$\overline{j(k_{\vartheta},i)} = \frac{-cvi + (uc+d)}{\sqrt{(cv)^2 + (uc+d)^2}} = \frac{c(u-iv) + d}{\sqrt{[c(u+iv) + d][c(u-iv) + d]}} = \frac{j(\gamma,\overline{\tau})}{|j(\gamma,\tau)|}$$

We may also read the parameter  $\alpha$  off by employing (5.11) deriving that

$$a = \sqrt{vc^2 + (uc+d)^2/v}^{-1} = \frac{\sqrt{v}}{|j(\gamma,\tau)|} = \sqrt{\text{Im}(\gamma\tau)}$$

which finishes the proof.

After these prerequisites, we may turn towards concrete examples of automorphic forms.

# 5.2 Theta distributions

The following section introduces the first example of automorphic forms on Mp<sub>2</sub>( $\mathbb{A}$ ) in the form of theta distributions. Their construction requires less effort than the construction of Eisenstein series, which is the reason for favouring their introduction for the first section. Further information is provided in [Kud03]. Let (V, q) be a quadratic  $\mathbb{Q}$  vector space,  $V_{\mathbb{A}}$ its adelisation, and let  $\omega$  denote the Weil representation of Mp<sub>2</sub>( $\mathbb{A}$ ) on  $\mathcal{S}(V_{\mathbb{A}})$  as in (3.19).

**Definition 5.6.** Define the *theta distribution*, also known as *theta kernel*,

$$\theta(g,h;\varphi) = \sum_{x \in V(\mathbb{Q})} (\omega(g)\varphi)(h^{-1}x)$$

which is a tempered distribution in  $\varphi \in \mathcal{S}(V_{\mathbb{A}})$ , where  $g \in \mathrm{Mp}_2(\mathbb{A})$  and  $h \in \mathrm{O}(V_{\mathbb{A}})$ . Here, the latter denote the adelic points of the orthogonal group  $\mathrm{O}(V)$ .

Poisson summation yields that the theta kernel is left invariant under  $Mp_2(\mathbb{Q})$  as well as O(V). Furthermore, it is only slowly increasing on the quotient  $Mp_2(\mathbb{Q}) \setminus Mp_2(\mathbb{A})$  and  $O(V) \setminus O(V_{\mathbb{A}})$ .

We would like to compare the context of the definition above to previous notations of theta functions.

**Remark 5.7.** Comparing Definition 5.6 with Definition 3.31 we find the following identity. Select  $\lambda \in L'/L$  and  $\varphi_f = \mathbb{1}_{\lambda+L}$  as well as  $\varphi_{\infty} \in \mathcal{S}(V_{\infty})$  and  $h_{\infty} \in O(V_{\mathbb{R}})$  as well as  $h_f = \mathrm{id} \in O(V_{\mathbb{A}_f})$ . Then we find

$$\theta_{L,\lambda}(\tau,h_{\infty};\varphi_{\infty}) = \operatorname{Im}(\tau)^{-m/4} \cdot \theta(g_{\tau},h_{\infty}h_{f};\varphi_{\infty}\otimes\varphi_{f}).$$

Theta lifts and as such theta functions are central objects in the present thesis and we will present a link to Eisenstein series, the Siegel–Weil formula (cf. Section 5.4.1), which is also crucial for the main objective of this thesis of proving injectivity and surjectivity results of theta lifts. We note that in many parts of the thesis, we use classical notation, also for theta functions which is why we explicitly outline the realisation of a classical theta function by a theta distributions as above.

#### 5.2.1 The classic setting

Theta distributions may be viewed as generalisations of classical theta series (cf. Definition 2.17). In this context, let (L, q) be a classical  $\mathbb{Z}$  lattice with positive definite quadratic form and even rank m. Then  $V := \mathbb{Q} \otimes_{\mathbb{Z}} L$  is a vector space with quadratic form q and the association

$$V_{\infty} \ni x \mapsto \exp(-2\pi \operatorname{q}(x))$$

defines a Schwartz-Bruhat function on  $V_{\infty} \coloneqq V \otimes_{\mathbb{Q}} \mathbb{R}$ .

Recall that the Weil representation factors through  $SL_2$  in the even case and we may restrict to working with the  $SL_2$  setting. Further, we select  $N \in \mathbb{N}$  and introduce the abbreviation  $K_0(N)_p = \overline{\Gamma_0(N)} \leq SL_2(\mathbb{Z}_p)$  to note that

$$\Gamma_0(N) \setminus \mathrm{SL}_2(\mathbb{R}) \simeq \mathrm{SL}_2(\mathbb{Q}) \setminus \mathrm{SL}_2(\mathbb{A}) / \prod_{p < \infty} K_0(N)_p$$

via the association

$$\alpha: \Gamma_0(N)x \mapsto \operatorname{SL}_2(\mathbb{Q})(x,1) \prod_{p < \infty} K_0(N)_p$$

As a Schwartz form choose  $\varphi = e_{\infty}(i\mathbf{q}, \cdot) \cdot \prod_{p < \infty} \mathbb{1}_{L_p}$ , where  $e_{\infty}$  is the standard character at the archimedian place from Example 1.48 and let  $g_{\tau}k_{\vartheta} = g_{\infty} \in \mathrm{SL}_2(\mathbb{R})$  be the element corresponding to  $u + iv = \tau \in \mathbb{H}$  with respect to the Iwasawa decomposition as in Definition 5.2:

$$g_{\tau}k_{\vartheta} = \left[ \begin{pmatrix} 1 & u \\ 0 & 1 \end{pmatrix} \cdot \begin{pmatrix} \sqrt{v} & 0 \\ 0 & 1/\sqrt{v} \end{pmatrix} \right] \cdot \begin{pmatrix} \cos(\vartheta) & \sin(\vartheta) \\ -\sin(\vartheta) & \cos(\vartheta) \end{pmatrix} \in \mathrm{SL}_2(\mathbb{R}).$$

With this notation, we obtain

$$\theta(g_{\tau}k_{\vartheta}, 1; \varphi) = e^{-i\vartheta m/2} \cdot \sqrt{v}^{m/2} \cdot \theta_L(\tau).$$
(5.12)

Given  $\tau \in \mathbb{H}$ , define  $\psi : \tau \mapsto \theta(g_{\tau}k_{\vartheta}, 1; \varphi)j(g_{\tau}k_{\vartheta}, i)^{m/2} = \theta_L(\tau)$ . Then, for arbitrary  $\gamma \in \Gamma_0(N)$ , where N is the level of L we obtain:

$$\begin{split} \psi(\gamma.\tau) &= \theta((\gamma g_{\tau}k_{\vartheta}, 1), 1; \varphi) \cdot j(\gamma g_{\tau}k_{\vartheta}, i)^{m/2} \\ &= \theta(\gamma(g_{\tau}k_{\vartheta}, \gamma^{-1}), 1; \varphi) \cdot j(\gamma g_{\tau}k_{\vartheta}, i)^{m/2} \\ &= \theta((g_{\tau}k_{\vartheta}, 1), 1; \varphi)\chi_{V}(\gamma^{-1}) \cdot j(\gamma g_{\tau}k_{\vartheta}, i)^{m/2} \\ &= \theta_{L}(\tau) \cdot \chi_{V}(\gamma^{-1}) \cdot j(\gamma, \tau)^{m/2} \\ &= \psi(\tau) \cdot \left(\frac{\operatorname{disc}(L)}{\gamma_{2,2}}\right) \cdot j(\gamma, \tau)^{m/2}, \end{split}$$

where the isolation of the character  $\chi_V$  from Example 1.62, which is meant to act on the lower right entry of the argument, is performed locally. For that purpose the local matrices are decomposed in order to apply the formulae of Example 3.8 and we refer to [Rao93] to relate the local Weil index appearing to the values of the character  $\chi_V$ . The final step to express this character in terms of the Kronecker symbol is performed in [Opi18, Cor. 1.10.2 p. 22]. Consequently, the function  $\psi : \mathbb{H} \to \mathbb{C}$  not only equals the theta series  $\theta_L$  on  $\mathbb{H}$  but the calculation also proves that it transforms with a character under  $\Gamma_0(N)$ . Additionally, holomorphy may be verified directly by a trivial estimate of the sum, so that  $\theta_L$  is identified as a holomorphic modular form (cf. Theorem 2.22).

**Corollary 5.8.** The function  $\theta_L : \mathbb{H} \to \mathbb{C}$  is a modular form of weight  $\frac{m}{2}$  with character  $\left(\frac{\operatorname{disc}(L)}{L}\right)$  with respect to the group  $\Gamma_0(\operatorname{lev}(L))$ .

### 5.3 Eisenstein series

This section is devoted to the introduction of Eisenstein series on  $SL_2$  which play a pivotal role in applying the Siegel–Weil formula for proving the central result of this thesis (cf. Theorem 7.16). At first, principal series representations, which are indispensable

for constructing Eisenstein series, are briefly discussed, before defining Eisenstein series associated to standard sections. Subsequently, Eisenstein series are ascribed to Schwartz–Bruhat functions which lays the foundation for forging a bridge to theta kernel from the previous section by means of the Siegel–Weil formula.

This entire section relies upon [KY10], in which Kudla and Yang provide a comprehensive setting in a compact fashion that is adequate for the investigation of Eisenstein series on  $\overline{\text{SL}}_2(\mathbb{A})$ . Readers interested in further details may wish to consult the more elaborate [Bum98], which also addresses the Lie theory upon which individual results are built.

We fix a character  $\psi$  of  $\mathbb{A}/\mathbb{Q}$  which we recall are identifiable with rational numbers. For explicit computations  $\psi$  is assumed to be the standard character as in Remark 1.55 with a = 1, even though the theory works for arbitrary non-trivial characters, with proper normalisation of the Haar measures involved. In general,  $\chi$  will denote a finite character on  $\mathbb{A}^{\times}/\mathbb{Q}^{\times}$ , if it is not specified further.

#### 5.3.1 Principal series representations

The principal series representations constitute an essential component of adelic Eisenstein series. They are vital for investigating unitary representations of  $SL_2$ , as they comprise two of essentially four infinitesimal isomorphism classes of representations of  $SL_2(\mathbb{R})$  – besides the *(mock) discrete series* and the *complementary series* [Lan98, VI.6 Thm. 8 p. 123]. The reason for the appearance of these representations in the construction of Eisenstein series is described in detail in [Bum98]. The case of  $SL_2$  is discussed in [Dei10, Sec. 7.1 p. 157], a more sophisticated compact approach is given by [Gel74] or [Kud96], whereas [Bum98, p. 213, Sec. 4.5 pp. 469-489] presents the content extensively, featuring a significant representation theoretic discussion. Supplementing material is also found in [Kud94], [KR92], as well as [KRY99, pp. 353-454].

The principal series representations of  $SL_2$  are induced by characters on the Borel group of upper triangular matrices. As in the case of finite groups, induced representations can be constructed as tensor products or explicitly by right action of the comprising group. However, in the infinite case involving topology is indispensable so that working only on dense subspaces with properties which are apparently advantageous for computations is convenient. This leads to demanding analytical and symmetry features of investigated functions.

The following construction is nearly identical in the  $SL_2(\mathbb{R})$  case. Let G be a totally disconnected locally compact group with a closed subgroup H which possesses a smooth

representation  $(\pi_H, V)$ . Consider the space of smooth functions  $f: G \to V$  which satisfy

$$f(hg) = \delta_G(h)^{-1/2} \delta_H(h)^{1/2} \pi_H(h) f(g)$$

with modular quasi characters  $\delta_G$ ,  $\delta_H$  of the respective groups.<sup>1</sup> Then, the representation of G induced by  $\pi_H$  (of H) on the space of functions defined above is denoted  $\pi_G = \text{Ind}_H^G(\pi_H)$  and given by right translation of G:

$$\pi_G(g)(f)(x) = f(xg).$$

The essential case for this thesis is the choice  $G = \operatorname{SL}_2(F_p)$  with upper triangular matrices H as a maximal compact subgroup. When the representation  $\pi_H$  of H is chosen to be a multiplicative quasi character  $\chi\left(\begin{smallmatrix}a&b\\0&a^{-1}\end{smallmatrix}\right) := \chi(a)$ ,  $\operatorname{Ind}_H^G(\chi)$  consists of functions  $f: G \to \mathbb{C}$ , such that

$$f(n(b)m(a)g) = \chi(a)|a|_p \cdot f(g),$$

where  $n(b) = \begin{pmatrix} 1 & b \\ 0 & 1 \end{pmatrix}$  and  $m(a) = \begin{pmatrix} a & 0 \\ 0 & a^{-1} \end{pmatrix}$  as usual. These are called *principal series* representations in case of irreducibility and an element of them is determined by the values it attains on  $K_p$ .<sup>2</sup> The choice of particular interest is that of a Hecke quasi character, which naturally decomposes into a family of local finite characters  $\chi_p$  on  $\mathbb{Q}_p^{\times}$  and quasi characters  $|\cdot|_p^s$  (cf. Remark 1.56). Hence, the functions in the associated induced representation transform in the following fashion

$$f(n(b)m(a)g) = \chi_p(a)|a|_p^{s+1} \cdot f(g).$$
(5.13)

The collection of spaces  $\operatorname{Ind}_{H}^{G}(\chi | \cdot |_{p}^{s})$  form vector bundles over  $\mathbb{C}$  (parametrised by s) denoted by  $I_{p}(s,\chi)$  which admit holomorphic sections  $\Phi_{p}(s)$  – at least for  $\operatorname{Re}(s) > 1$ . In particular, there is a so called *spherical section*  $\Phi_{p}^{0}$ , attaining the value 1 on the neutral element of SL<sub>2</sub> and being right  $K_{p}$  invariant.

For a family of Hecke quasi characters  $\chi \cdot |\cdot|^s$  parametrised by  $s \in \mathbb{C}$ , the local representations described above (for all  $p \leq \infty$ ) may be combined via *restricted tensor products* (cf. [Bum98, p. 300 as.]). For this purpose the spherical section is fixed in each component. This results in a collection of adelic representations, denoted by  $I(s, \chi)$  and also called *principal series representations* associated to the Hecke character. A similar

<sup>&</sup>lt;sup>1</sup>Note that the factors are chosen in such a fashion that induction preserves unitaricity (cf. [Bum98, Thm. 2.6.1 p. 224]).

<sup>&</sup>lt;sup>2</sup>This is true by strong approximation (cf. [Kne65]).

construction can be done in case of  $\overline{\operatorname{SL}}_2(\mathbb{A})$  (cf. [KY10]) and the resulting principal series representation will also be denoted  $I(s,\chi)$ . In case these factor through  $\operatorname{SL}_2(\mathbb{A})$ , they are called *even*, if not, they are called *odd*. Again, there are holomorphic sections  $\Phi(s)$ , which are utilised as kernels for symmetrisation resulting in non-holomorphic Eisenstein series which are to be discussed in the following section. To be more explicit, a section  $\Phi(s)$ from  $I(s,\chi)$  is a smooth function on  $G'_{\mathbb{A}}$  such that

$$\Phi(p'g',s) = \chi(a) \cdot |a|^{s+1} \Phi(g',s) \begin{cases} 1, & \text{in the even case,} \\ z, & \text{in the odd case,} \end{cases}$$
(5.14)

where  $p' = [n(b)m(a), z] \in P'_{\mathbb{A}}$  is in the standard Borel subgroup (cf. Subsection 3.2) and the same normalisation as in [KRY06, p. 287] is chosen. A section  $\Phi(s)$  is called *standard* if its restriction to the maximal compact subgroup  $K_{\mathbb{A}} < SL_2(\mathbb{A})$  ( $K'_{\mathbb{A}} < \overline{SL}_2$ , respectively) is independent of *s* and *factorisable* if  $\Phi = \bigotimes_p \Phi_p$  is a primitive tensor. Further, every section  $\Phi(g, s) \in I(s, \chi)$  is determined by its values on  $K'_{\mathbb{A}}$  which is verified via the Iwasawa decomposition in (3.23).

The simplest example of a section at a finite place is the normalised standard section. Example 5.9. Let  $2 and. Then the section <math>\Phi_p^0 \in I_p(s, \chi)$  determined by

$$\Phi_n^0(k,s) = 1$$

for  $k \in K_p = \operatorname{SL}_2(\mathbb{Z}_p)$  is a standard section. It is called the *spherical section* (since it is right invariant under  $K_p$ ). To make sense of the above, recall that in case 2 < p, there is a section of  $K_p = \operatorname{SL}_2(\mathbb{Z}_p) \hookrightarrow \overline{\operatorname{SL}}_2(\mathbb{Q}_p)$ .

Recall that the characters of the double cover  $\overline{SO}_2(\mathbb{R})$  sitting in  $K'_{\infty}$  are of the following form (cf. [BF04, Sec. 2]).

**Remark 5.10.** For every  $l \in \mathbb{Z}/2$  there is a character of  $\overline{SO}_2(\mathbb{R})$  denoted by  $\nu_l$  with values

$$\nu_l\left(k_\vartheta, \pm \sqrt{j(k_\vartheta, \tau)}\right) \mapsto \pm \sqrt{j(k_\vartheta, i)}^{-1} = \pm e^{il\vartheta}.$$

With this family of characters, one may define concrete elements of the principal series representation at the infinite place.

Example 5.11. Assume  $\chi$  is a quadratic character. For  $l \in \mathbb{Z}/2$  imposing the conditions

$$\Phi^l_{\infty}(gk,s) = \nu_l(k)\Phi^l_{\infty}(g,s), \quad \Phi^l_{\infty}(1,s) = 1$$

on a candidate  $\Phi_{\infty}^{l}$  for a section through  $I_{\infty}(s,\chi)$  almost works. In fact, the parameter l has to satisfy a parity condition to comply with the action of  $-\mathcal{I}$  when letting it act via  $\chi$ , for otherwise the section  $\Phi_{\infty}^{l}$  is forced to vanish. The concrete condition is stated in (5.18). We refer to  $\Phi_{\infty}^{l}$  as the *normalised eigenfunction* of weight l and note that these functions span the  $K_{\infty}$  finite functions in  $I_{\infty}(s,\chi)$ .

For  $\tau = u + iv \in \mathbb{H}$  we recall the following matrices from the Iwasawa decomposition appearing in Proposition 5.3. Let  $g_{\tau} = \begin{pmatrix} 1 & u \\ 0 & 1 \end{pmatrix} \begin{pmatrix} \sqrt{v} & 0 \\ 0 & 1/\sqrt{v} \end{pmatrix}$  and  $k_{\vartheta} = \begin{pmatrix} \cos(\vartheta) & \sin(\vartheta) \\ -\sin(\vartheta) & \cos(\vartheta) \end{pmatrix}$  for some radiant  $\vartheta$ . Inserting these, we find

$$\Phi_{\infty}^{l}(g_{\tau}k_{\vartheta},s) = \chi_{\infty}\left(\sqrt{\mathrm{Im}(\tau)}\right)\mathrm{Im}(\tau)^{(s+1)/2}\nu_{l}(k_{\vartheta})\Phi_{\infty}^{l}(1,s).$$

In particular, for  $\gamma \in Mp_2(\mathbb{Z})$  we find  $\gamma g_{\tau} = g_{\gamma\tau} k_{\vartheta}$ , resulting by Remark 5.5 in

$$\begin{split} \Phi^{l}_{\infty}(g_{\gamma\tau}k_{\vartheta},s) &= \chi_{\infty}(\sqrt{\mathrm{Im}(\gamma\tau)}) \operatorname{Im}(\gamma\tau)^{(s+1)/2} \nu_{l}(k_{\vartheta}) \Phi^{l}_{\infty}(1,s) \\ &= \operatorname{Im}(\gamma\tau)^{(s+1)/2} \left(\frac{\overline{j(\gamma,\tau)}}{|j(\gamma,\tau)|}\right)^{l} \Phi^{l}_{\infty}(1,s) \\ &= \operatorname{Im}(\tau)^{l/2} \frac{\operatorname{Im}(\gamma\tau)^{(s+1-l)/2}}{j(\gamma,\tau)^{l}}. \end{split}$$

There is an intertwining map which may be used to construct elements of the principal series representations by means of the Weil representation. In fact, when considering the action of the Weil representation in Example 3.8, for instance, and comparing it to (5.14), the following statement is inferred.

**Lemma 5.12.** Let (V,q) be a quadratic  $\mathbb{Q}$  space of dimension m with character  $\chi_V$  as in Example 1.62. Further, let  $p \leq \infty$ ,  $\omega_p$  be the associated Weil representation as in (3.16) and set  $s_0 \coloneqq \frac{m}{2} - 1$ . Then

$$\lambda_p : \mathcal{S}(V_p) \to I_p(s_0, \chi), \qquad \lambda_p(\varphi)(g') \coloneqq \omega_p(g')\varphi(0)$$
(5.15)

is a well defined map. In fact, it is almost always surjective, except for three special cases that are described in [KY10, Sec. 4 p. 2286].

Clearly, to any of the sections arising from the above intertwining operator (5.15), there is an associated canonical standard section. This method reproduces the already known examples from above but also yields new examples.

*Example* 5.13. a) Let  $V_{\infty}$  have signature  $(m^+, m^-)$ . Then for any element z of the

associated Grassmannian, there is an associated positive standard majorant  $q_z^+$  of q by decomposing orthogonally along z and flipping the sign of the negative definite part (cf. Remark 3.28). Setting

$$\varphi_{\infty}(x,z) \coloneqq e^{-2\pi \operatorname{q}_{z}^{+}(x)} \in \mathcal{S}(V_{\infty})$$
(5.16)

associates the standard section from Example 5.11 of weight  $l = \frac{m^+ - m^-}{2}$  via the intertwining operator of Lemma 5.12 in the archimedian case.

- b) Let  $L_p$  be unimodular, then the standard section associated to  $\lambda_{V_p}(\mathbb{1}_L)$  equals the spherical section  $\Phi_p^0(s)$  from Example 5.9.
- c) Next, assume  $L_p$  is simply even, let  $\mu_p \in L'_p/L_p$  and write  $\varphi_{\mu_p} \coloneqq \mathbb{1}_{\mu_p+L_p}$ . Then the action of  $\omega_p$  on  $\varphi_{\mu_p}$  is described in Section 3.2, where the Fourier transform of  $\varphi_{\mu_p}$  is computed explicitly in Remark 1.54. This gives rise to a standard section in  $I_p(\chi, s)$ .
- d) Further, we find an operator for the finite adelic part

$$\lambda_f: \mathcal{S}(V_{\mathbb{A}_f}) \to I_f(s_0, \chi) = \otimes_{p < \infty}' I_p(s_0, \chi), \qquad \lambda_f(\varphi)(g'_f) = \omega_f(g'_f)\varphi(0).$$

For  $\mu \in L'/L \simeq \hat{L}'/\hat{L}$  there is the associated element  $\varphi_{\mu} \coloneqq \mathbb{1}_{\mu+L_f} = \otimes_{p < \infty} \varphi_{\mu_p} \in \mathcal{S}(V_{\mathbb{A}_f})$  for which we find  $\lambda_f(\varphi_{\mu}) \in I_f(s_0, \chi)$ . The associated standard section will be denoted by  $\Phi_{\mu}$  and plays a major role in the construction of Eisenstein series below. Clearly, we may perform the same procedure for a Schwartz–Bruhat function  $\varphi \in \mathcal{S}(V_{\mathbb{A}})$  and associate to it a standard section  $\Phi \in I(s, \chi)$ .

With the tools described and examples constructed above, we may turn towards Eisenstein series.

#### 5.3.2 Eisenstein series

The following exposition is based on [KY10, Sec. 2] and [BY09, Sec. 2]. We begin by defining Eisenstein series in analogy to the classical case by symmetrising sections of the principal series representation.

**Definition 5.14.** For a standard section  $\Phi(g, s) \in I(s, \chi)$  define the associated *Eisenstein* series

$$\mathcal{E}(g,s;\Phi) := \sum_{\gamma \in P(\mathbb{Q}) \setminus \mathrm{SL}_2(\mathbb{Q})} \Phi(\gamma g,s).$$
(5.17)

Note that this series converges for  $\operatorname{Re}(s) > 1$ , is, by definition, left invariant under  $\operatorname{SL}_2(\mathbb{Q})$ , and has a meromorphic continuation to the whole complex plane in s due to Langlands [Lan06].<sup>3</sup>

We are interested in concrete instances of the above series and restrict, from now on, to a quadratic character  $\chi$  which may be realised by a square free integer  $d \in \mathbb{Z}$  via

$$\chi(x) = \begin{cases} \mathcal{H}_{\mathbb{A}}(x,d), & \text{even case,} \\ \\ \mathcal{H}_{\mathbb{A}}(x,2d), & \text{odd case.} \end{cases}$$

Here,  $\mathcal{H}_{\mathbb{A}}$  denotes the Hilbert symbol discussed in Section 1.4 and the even or odd case refers to the principal series representation factoring through SL<sub>2</sub> or not.<sup>4</sup> Recall that in order for the section  $\Phi_{\infty}^{l}$  of Example 5.11 to be non-trivial and hence part of  $I_{\infty}(s,\chi)$ , the following parity condition has to be satisfied

$$\begin{cases} (-1)^l = \operatorname{sig}(d), & \text{even case,} \\ l \equiv \operatorname{sig}(d)/2 \mod 2, & \text{odd case.} \end{cases}$$
(5.18)

In this context, we are able to represent classical Eisenstein series.

Example 5.15. For  $\chi$  being the trivial character and  $g_{\tau}$  as a above we find by Example 5.11 and Example 5.9 with the definition  $\Phi_f^0 = \bigotimes_p \Phi_p^0$  that

$$\mathcal{E}(g_{\tau},s;\Phi_{\infty}^{l}\otimes\Phi_{f}^{0})=\mathrm{Im}(\tau)^{l/2}\sum_{\gamma\in\Gamma_{\infty}\backslash\operatorname{SL}_{2}(\mathbb{Z})}\frac{\mathrm{Im}(\gamma\tau)^{(s+1-l)/2}}{j(\gamma,\tau)^{l}}.$$

The above example demonstrates how classical Eisenstein series may be realised. In fact, this motivates considering the Eisenstein series in (5.17) as functions on the upper half plane. Recall that any factorisable finite standard section  $\Phi_f(g,s) \in I_f(\chi,s)$  is invariant under some open subgroup  $K_0$  of  $K_0(4)$ . Hence, by strong approximation the following Eisenstein series, whose definition is motivated by the example above, determines the series  $\mathcal{E}(g', s; \Phi_{\infty}^l \otimes \Phi_f)$  completely.

**Definition 5.16.** For  $\Phi_{\infty}^{l}$  from Example 5.11 transforming with  $\nu_{l}$  under  $\overline{K_{\infty}}$  and a finite

<sup>&</sup>lt;sup>3</sup>Compare 6 Lemma 6.1 p. 91 for instance. However, that treatment is written in a completely different notation and with a different perspective than the current document (a more analytic point of view), rendering it not particularly helpful on a glance if only our setting is known.

 $<sup>^4\</sup>mathrm{This}$  will agree with the signature being even or odd in our later setting.

standard section  $\Phi_f \in I_f(\chi, s)$ , define

$$\mathcal{E}(\tau, s; \Phi_{\infty}^{l} \otimes \Phi_{f}) := \operatorname{Im}(\tau)^{-l/2} \cdot \mathcal{E}(g_{\tau}, s; \Phi_{\infty}^{l} \otimes \Phi_{f}).$$
(5.19)

The series  $\mathcal{E}(\tau, s; \Phi_{\infty}^{l} \otimes \Phi_{f})$  then defines a non-holomorphic modular form of weight land we have seen how to realise classical scalar valued Eisenstein series. However, in this thesis, vector valued modular forms play a major role and we seek means to describe these also in the adelic setting.

In the spirit of Subsection 3.2.1, we will realise non-holomorphic vector valued Eisenstein series for the discrete Weil representation (cf. Definition 3.37) in the present setting. To this end, let (V, q) be the rational quadratic space associated to an even  $\mathbb{Z}$ -lattice (L, q)and select the quadratic character  $\chi = \chi_V = \mathcal{H}_{\mathbb{A}}(\cdot, \operatorname{disc}(V))$  described in Example 1.62. Recall that in Example 5.13 d), for  $\mu \in L'/L$  the finite adelic Schwartz form  $\varphi_{\mu} := \mathbb{1}_{\mu+\hat{L}} :=$  $\otimes_{p<\infty} \mathbb{1}_{\mu+L_p}$  has been associated to a standard section  $\Phi_{\mu}$  via the intertwining operator.

**Definition 5.17.** With the notation as above, we define the following Eisenstein series

$$\mathcal{E}_{\hat{L},l}(\tau,s) \coloneqq \sum_{\mu \in L'/L} \mathcal{E}\left(\tau,s; \Phi_{\infty}^{l} \otimes \Phi_{\mu}\right) \varphi_{\mu}.$$
(5.20)

The shape of this series not only resembles that of the classical non-holomorphic vector valued Eisenstein series presented in Definition 3.37 but may be directly translated to that setting [BY09].

**Proposition 5.18.** With the notation as above we find the following identity

$$\mathcal{E}_{\hat{L},l}(\tau,s) = E_{L,0,l}\left(\tau, \frac{s+1-l}{2}\right).$$
(5.21)

For this equality, the natural identification  $\mathcal{S}_{L'/L} \simeq \mathbb{C}[L'/L]$  from Lemma 3.9 is used.

*Proof*: We follow [BY09, Sec. 2 pp. 637-642] for part of the computation. We begin our calculation by inserting the explicit term for the section  $\Phi_{\infty}^{l}$  that has been calculated in Example 5.11.

$$\mathcal{E}_{\hat{L},l}(\tau,s) = \operatorname{Im}(\tau)^{-l/2} \sum_{\mu \in \mathcal{L}} \mathcal{E}(g_{\tau},s;\Phi_{\infty}^{l} \otimes \Phi_{\mu})\varphi_{\mu}$$
$$= \frac{1}{2} \sum_{\mu \in \mathcal{L}} \sum_{\gamma \in \Gamma_{\infty} \setminus \Gamma(1)} \frac{\operatorname{Im}(\gamma\tau)^{(s+1-l)/2}}{j(\gamma,\tau)^{l}} \Phi_{\mu}(\gamma)\varphi_{\mu}.$$

Now, recall that  $\Phi_{\mu}$  was induced by  $\lambda_f(\varphi_{\mu}) \in I_f(s_0, \chi)$  via the finite intertwining operator introduced in Example 5.13. We then find via the correspondence of  $\mathcal{S}_{L'/L}$  and  $\mathbb{C}[L'/L]$ constructed before Lemma 3.9 that for  $\gamma \in \mathrm{Mp}_2(\mathbb{Z})$  the following identity is true.

$$\begin{split} \lambda_f(\varphi_\mu)(\gamma) &= (\omega_f(\gamma)\varphi_\mu) (0) \\ &= \langle \omega_f(\gamma)\varphi_\mu, \varphi_0 \rangle \\ &= \langle \varphi_\mu, \omega_f(\gamma)^{-1}\varphi_0 \rangle \\ &= \langle \overline{\omega_f(\gamma^{-1})}\varphi_0, \varphi_\mu \rangle \\ &= \langle \rho_L(\gamma^{-1})\varphi_0, \varphi_\mu \rangle \\ &= \langle \rho_L(\gamma^{-1})\mathfrak{e}_0, \mathfrak{e}_\mu \rangle \\ &= \langle \mathfrak{e}_0, \rho_L(\gamma)\mathfrak{e}_\mu \rangle. \end{split}$$

Here, we have used the correspondence (3.27) between the classical Weil representation and the dual of the adelic version on  $S_{L'/L}$ . Inserting this relation in the above expression for the Eisenstein series and using the isomorphism  $S_{L'/L} \simeq \mathbb{C}[L'/L]$  we obtain

$$\begin{split} & \mathcal{E}_{\hat{L},l}(\tau,s) \\ &\simeq \frac{1}{2} \sum_{\gamma \in \overline{\Gamma_{\infty}} \setminus \operatorname{Mp}_{2}(\mathbb{Z})} \frac{\operatorname{Im}(\gamma \tau)^{(s+1-l)/2}}{j(\gamma,\tau)^{l}} \sum_{\nu \in \mathcal{L}} \langle \rho_{L}(\gamma^{-1}) \mathfrak{e}_{0}, \mathfrak{e}_{\mu} \rangle \mathfrak{e}_{\mu} \\ &= \frac{1}{2} \sum_{\gamma \in \overline{\Gamma_{\infty}} \setminus \operatorname{Mp}_{2}(\mathbb{Z})} \left[ \operatorname{Im}(\tau)^{(s+1-l)/2} \right] \Big|_{l} \cdot \rho_{L}(\gamma^{-1}) \mathfrak{e}_{0} \\ &= \frac{1}{2} \sum_{\gamma \in \overline{\Gamma_{\infty}} \setminus \operatorname{Mp}_{2}(\mathbb{Z})} \left[ \operatorname{Im}(\tau)^{(s+1-l)/2} \mathfrak{e}_{0} \right] \Big|_{L,l} \gamma \\ &= E_{L,0,l}(\tau, \frac{s+1-l}{2}). \end{split}$$

This completes the proof.

It is possible to compute the Fourier expansion of  $\mathcal{E}(\tau, s; \Phi)$  for factorisable sections in terms of local Whittaker functions as explained in [KY10, Thm. 2.4 p. 2282]. The computations in concrete cases involve solving Gauss sums for the appearing local integrals and a sizable amount of case distinctions. This has been carried out in the special case of lattices associated to quadratic field extensions in a Master's thesis [Met19].

## 5.4 Comparing theta and Eisenstein series

Theta and Eisenstein series represent a sizable share of the theory of automorphic forms. In fact, in many cases it is known that the span of theta series encompasses the space of cusp forms (cf. [Mül24, Thm. 6.7]) which is then complemented by Eisenstein series. Further, it is worth noting the existence of a celebrated formalism for relating theta series to Eisenstein series, called the *Siegel-Weil* formula, which is the subject of the subsequent subsection. Afterwards, the Siegel-Weil formula is applied to link a certain geometric integral of the Kudla-Millson Schwartz form to Eisenstein series, as outlined by Kudla [Kud03, Sec. 4.3 pp. 328-335]. Recall that L is an even lattice of signature  $(m^+, 2)$  and the dimension of  $V = L \otimes_{\mathbb{Z}} \mathbb{Q}$  is denoted by m.

#### 5.4.1 Siegel–Weil formula

In the previous section, Eisenstein series have been associated to Schwartz–Bruhat functions. The subject of this subsection is the so called *Siegel–Weil* formula which establishes a relation between such Eisenstein series (5.17) evaluated at a special point  $s_0$  and theta integrals as described in (5.22). It is an adelic extension of the classic result Theorem 2.9 presented in Section 2.1.2 and has first been proven by André Weil [Wei65]. The curious reader may find further information in [KR94] where the result is notably extended. Recall that the classic result mentioned above reads

$$\sum_{i} \frac{\#(S_i, S_i)^{-1}}{\sum_{j} \#(S_j, S_j)^{-1}} \theta_{S_i} = E_{m/2},$$

where  $S_i$  are representatives of classes of even unimodular lattices of rank m. In other words, a weighted sum, or integral for an appropriate measure, over theta series equals an Eisenstein series of weight m/2. Weil performed the step of reinterpreting the sum as an integral and transformed the above formula into the adelic context [Wei65].

In order to state the Siegel–Weil formula, we have to introduce the aforementioned theta integral. Recall the definition of adelic theta distributions as in Definition 5.6. Namely, for a Schwartz–Bruhat function  $\varphi \in \mathcal{S}(V_{\mathbb{A}})$  we set

$$\theta(g,h;\varphi) = \sum_{x \in V_{\mathbb{Q}}} (\omega(g)\varphi)(h^{-1}x)$$

as a function in  $g \in Mp_2(\mathbb{A})$  and  $h \in O(V_{\mathbb{A}})$ . Averaging over the argument h delivers the

following standard theta integral.

**Definition 5.19.** For a Schwartz–Bruhat function  $\varphi \in \mathcal{S}(V_{\mathbb{A}})$  define the following theta integral.

$$I(g;\varphi) \coloneqq \int_{\mathcal{O}(V_{\mathbb{Q}}) \setminus \mathcal{O}(V_{\mathbb{A}})} \theta(g',h;\varphi) \,\mathrm{d}h, \qquad (5.22)$$

where dh is the standard invariant normalised measure on the quotient such that  $\operatorname{vol}(O(V_{\mathbb{Q}}) \setminus O(V_{\mathbb{A}})) = 1.$ 

By Weil's convergence criterion, the integral  $I(g; \varphi)$  above is absolutely convergent in case the rational space V is anisotropic or  $\dim(V) = m > 2 + r$ , where r denotes the Witt rank of V. In case this criterion is not matched, the integral may still be regularised as in [KR94, p. 41] and numerous results may be transferred to this setting.

With these preparations we require only one additional piece of notation to state the Siegel–Weil formula. Let  $\varphi \in \mathcal{S}(V_{\mathbb{A}})$  be a Schwartz–Bruhat form and  $\Phi \in I(s, \chi)$  its associated standard section through the principal series representation arising from the intertwining operator as described in Example 5.13 d) (also compare Lemma 5.12).

Definition 5.20. With the notation as above, write

$$\mathcal{E}(g', s; \varphi) \coloneqq \mathcal{E}(g', s; \Phi), \qquad \mathcal{E}(\tau, s; \varphi) \coloneqq \mathcal{E}(\tau, s; \Phi).$$

**Theorem 5.21** (Siegel–Weil formula, [Kud03, Thm. 4.1 p. 36]). Assume m > 2+r or that V is anisotropic. Then the theta integral  $I(g'; \varphi)$  for  $\varphi \in \mathcal{S}(V_{\mathbb{A}})$  is absolutely convergent. Further,  $\mathcal{E}(g', s; \varphi)$  is holomorphic at  $s_0 = \frac{m}{2} - 1$  and the following identity holds:

$$\mathcal{E}(g', s_0; \varphi) = \kappa \cdot I(g'; \varphi),$$

where  $\kappa = 2$ , if  $m \leq 2$ , and  $\kappa = 1$  otherwise.

In particular, the Eisenstein series associated to a lattice as it has been discussed in the previous section bears information about representation numbers of the quadratic lattice in its Fourier expansion at the critical point  $s_0 = \frac{m}{2} - 1$ . This is remarkable since the Eisenstein series is essentially build from local data, while the theta integral involves global arithmetic. Hence, theta integrals associated to different quadratic spaces may be related to each other, via the Eisenstein series, once the local data agrees. **Definition 5.22.** Let  $V_p, V'_p$  be two quadratic  $\mathbb{Q}_p$  spaces of dimension m. Suppose these spaces induce the same character  $\chi_{V_p} = \chi_{V'_p}$  (cf. Example 1.62). Two functions  $\varphi_p \in \mathcal{S}(V_p)$  and  $\varphi'_p \in \mathcal{S}(V'_p)$  are called *matching*, if they give rise to the same section via the intertwining operator  $\lambda_p(\varphi_p) = \lambda'_p(\varphi'_p)$ .

In fact, in case of m > 4 and p non-Archimedian, every  $\varphi_p \in \mathcal{S}(V_p)$  has a matching function  $\varphi'_p \in \mathcal{S}(V'_p)$ , since the local principal series  $I_p(s_0, \chi_p)$  is irreducible.

In order to compare theta integrals as suggested above, it is hence sufficient to find a non-trivial match at the archimendian place. One solution to that problem is given by the Kudla–Millson Schwartz function which is introduced in the following section.

#### 5.4.2 Kudla–Millson Schwartz form

Let  $(V, \mathbf{q})$  be a quadratic  $\mathbb{Q}$  vector space of signature  $(m^+, 2)$  and dimension m. Recall that for a complex manifold D, the space of smooth forms of Hodge type (1, 1) is denoted by  $\mathscr{A}^{(1,1)}(D)$ . Kudla and Millson have constructed a form  $\varphi_{\mathrm{KM}} \in \mathcal{S}(V_{\mathbb{R}}) \otimes \mathscr{A}^{(1,1)}(\tilde{\mathbb{D}})$ , where  $\tilde{\mathbb{D}}$  denotes the oriented Grassmannian associated to the base lattice L, with the following properties.

i) For all  $h \in O(V_{\mathbb{R}})$ 

$$h^*\varphi_{\rm KM}(h^{-1}x) = \varphi_{\rm KM}(x). \tag{5.23}$$

ii) The form  $\varphi_{\rm KM}$  has weight m/2 for  $K'_{\infty}$  for the Weil representation  $\omega_{\infty}$ , meaning

$$\omega_{\infty}(k')\varphi_{\rm KM} = \nu_{m/2}(k')\varphi_{\rm KM},\tag{5.24}$$

where the character  $\nu_l$  is presented in Remark 5.10.

iii) The form  $\varphi_{\rm KM}$  is closed, i.e.

$$\mathrm{d}\varphi_{\mathrm{KM}} = 0 \tag{5.25}$$

for the exterior differential d on  $\mathbb{D}$ .

This form *matches* the Gaussian on a positive definite space of the same dimension in the sense that it gives rise to the same section through the principal series representation. To specify the meaning of this statement, we define the following form.

**Definition 5.23.** Define the following closed  $O(V_{\infty})$  invariant (1,1) form on  $\mathbb{D}$ :

$$\Omega \coloneqq \varphi_{\rm KM}(0). \tag{5.26}$$

An explicit description in local coordinates on the tube domain model of that form is presented in [Kud03, Prop. 4.11 p. 330]. It is given by

$$\Omega = -\frac{1}{2\pi i} \left[ -\mathbf{b}(y,y)^{-2} \,\mathbf{b}(y,\,\mathrm{d}z) \wedge \mathbf{b}(y,\,\mathrm{d}\overline{z}) + \mathbf{b}(y,y)^{-1} \frac{1}{2} \,\mathbf{b}(\,\mathrm{d}z,\,\mathrm{d}\overline{z}) \right] \tag{5.27}$$

in coordinates on the tube domain model  $\mathcal{H}_{m^+}^{\pm}$  (cf. (4.16)). Its negative  $-\Omega$  is an invariant Kähler form on  $\tilde{\mathbb{D}}$  and descends to a Kähler form on the variety  $X_K$  given in (5.30).

Note that we may write

$$\varphi_{\rm KM}(x) \wedge \Omega^{m-1} = \tilde{\varphi}_{\rm KM}(x) \Omega^m$$

for a function  $\tilde{\varphi}_{\mathrm{KM}} \in \mathcal{S}(V_{\mathbb{R}}) \otimes A^{(0,0)}(\tilde{\mathbb{D}})$ . The  $\mathcal{O}(V_{\infty})$  invariance of  $\Omega$  then implies that

$$\tilde{\varphi}_{\rm KM}(hx,hz) = \tilde{\varphi}_{\rm KM}(x,z)$$

for any  $h \in O(V_{\infty})$ . Further, for  $k \coloneqq \frac{m}{2}$  we find

$$\lambda_{\infty}(\varphi_{\mathrm{KM}}(\,\cdot\,,z)) = \Phi^k(s_0)\Omega, \quad \lambda_{\infty}(\tilde{\varphi}_{\mathrm{KM}}(\,\cdot\,,z)) = \Phi^k(s_0) \tag{5.28}$$

for any  $z \in \mathbb{D}$ , where  $\lambda_{\infty}$  denotes the intertwining operator described in Lemma 5.12 and  $\Phi_{\infty}^k$  denotes the standard section in  $I_{\infty}(s, \chi_V)$  at the infinite place presented in Example 5.11. This means, the function  $\tilde{\varphi}_{\rm KM}$  will be associated by  $\lambda_{\infty}$  to the same section as the standard Gaussian  $\varphi_0$  on a space V' of signature (m, 0) if the local characters agree, i.e. they are *matching*. If further, we have two matching functions at the finite places  $\varphi_f \in \mathcal{S}(V_f)$  and  $\varphi'_f \in \mathcal{S}(V'_f)$ , the Siegel–Weil formula (Theorem 5.21) yields

$$I(g'; \tilde{\varphi}_{\mathrm{KM}} \otimes \varphi_f) = E(g', s_0; \tilde{\varphi}_{\mathrm{KM}} \otimes \varphi_f) = E(g', s_0; \varphi_0 \otimes \varphi'_f) = I(g'; \varphi_0 \otimes \varphi'_f).$$
(5.29)

Before turning towards an essential application of the Siegel–Weil formula within the scope of this thesis, we remark that a more constructive approach to the Kudla– Millson Schwartz form will be sketched in Subsection 7.2.1 while an even more explicit representation is found in [MZ23].

#### 5.4.3 A geometric integral

The following discussion evolves around [Kud03, Prop. 4.17 p. 332]. Before stating the main proposition, understanding the structure of the variety in terms of the considered reductive group is instructive.

**Lemma 5.24.** Let G be a reductive group,  $K_f \subset G(\mathbb{A}_f)$  be a compact open subgroup and  $K_{\infty} \subset G(\mathbb{R})$  be a maximal compact subgroup. Assume  $G(\mathbb{A}) = \bigsqcup_j G(\mathbb{Q})G^+(\mathbb{R})h_jK_f$  for suitable  $h_j \in G(\mathbb{A}_f)^5$  and set  $\Gamma_j \coloneqq G(\mathbb{Q}) \cap (G^+(\mathbb{R})h_jK_fh_j^{-1})$ . Then by abuse of notation we understand  $\Gamma_j \leq G^+(\mathbb{R})$  and find

$$G(\mathbb{Q})\backslash (G(\mathbb{R}) \times G(\mathbb{A}_f))/(K_{\infty} \times K_f) = \bigsqcup_j \Gamma_j \backslash (G^+(\mathbb{R})/K_{\infty}).$$

Further, the following map is bijective and equivariant

$$\bigsqcup_{j} \Gamma_{j} \backslash G^{+}(\mathbb{R}) \to G(\mathbb{Q}) \backslash G(\mathbb{A}) / K_{f}, \quad \Gamma_{j} x \mapsto G(\mathbb{Q})(x, h_{j}) K_{f}$$

Proof: By abuse of notation, we understand  $\Gamma_j \subset G(\mathbb{Q}_p)$  for all p. First, we assume  $x = \gamma_j y \in G^+(\mathbb{R})$  with  $\gamma_j \in \Gamma_j$ . Then recall that  $\Gamma_j \subset G(\mathbb{Q}) \cap h_j K_f h_j^{-1}$ , implying  $(x, h_j) \equiv (\gamma_j y, h_j) \equiv (y, \gamma_j^{-1} h_j) \equiv (y, h_j)$ , so that the map is well defined.

Next, we verify that the mapping is injective: Assume there are two  $x, y \in \bigsqcup_j \Gamma_j \setminus G^+(\mathbb{R})$ with  $(x, h_i) \equiv (y, h_j)$ . Clearly, i = j, for otherwise the decomposition of  $G(\mathbb{A})$  provided by the  $h_j$  is not disjoint. So assume  $(x, h_j) \equiv (y, h_j)$ . Then there are  $\gamma \in G(\mathbb{Q})$  and  $k \in K_f$ with  $(x, h_j) = (\gamma y, \gamma h_j k)$ . Hence,  $h_j = \gamma h_j k \iff h_j k^{-1} h_j^{-1} = \gamma \in G(\mathbb{Q})$ , meaning  $\gamma = h_j k^{-1} h_j^{-1} \in \Gamma_j$ . As a consequence, we find  $x = \gamma y$ , meaning the map is injective. For surjectivity let there be  $(x_{\infty}, x_f) \in G(\mathbb{A})$ . By assumption, we know that there

For surjectivity let there be  $(x_{\infty}, x_f) \in G(\mathbb{A})$ . By assumption, we know that there is  $\gamma \in G(\mathbb{Q})$  with  $\gamma^{-1}x_f \in h_jK_f$  for some j and  $\gamma^{-1}x_{\infty} \in G^+(\mathbb{R})$ . Then necessarily  $(x_{\infty}, x_f) = \gamma(\gamma^{-1}x_{\infty}, 1)\gamma^{-1}x_f$ , i.e.  $\Gamma_j\gamma^{-1}x_{\infty} \in \Gamma_jG^+(\mathbb{R})$  is a preimage.  $\Box$ 

In our context, the considered group will be denoted by H and represents the GSpin group associated to the lattice (L, q) of signature  $(m^+, 2)$ . Denote by  $\tilde{\mathbb{D}}$  the oriented Grassmannian of L and choose a compact open subgroup  $K_f \subset H(\mathbb{A}_f)$ . Then we consider the variety

$$X_K \coloneqq H(\mathbb{Q}) \setminus (\mathbb{D} \times H(\mathbb{A}_f)) / K_f \tag{5.30}$$

<sup>&</sup>lt;sup>5</sup>Compare [Kne65] as a reference for the method of strong approximation or [PRR93, Sec. 7.4 p. 427].

and have that

$$X_K = \bigsqcup_{j=1} \Gamma_j \backslash \mathbb{D}$$

as in the proposition above, where  $\mathbb{D}$  is a connected component of  $\tilde{\mathbb{D}}$ . We write  $Y_j := \Gamma_j \setminus \mathbb{D}$ and note that there are  $|\hat{\mathbb{Z}}^{\times} : N(K_f)|$  of these components. Here, N denotes the Clifford norm given in Definition 4.28. With this preparation we may state the following modified version of [Kud03, Prop. 4.17 p. 332].

**Proposition 5.25.** We find for a compact subgroup  $K_f \leq H(\mathbb{A}_f)$  with  $K_f \cap Z(\mathbb{A}) \simeq \hat{\mathbb{Z}}^{\times}$ under the isomorphism  $Z(\mathbb{A}) \simeq \mathbb{A}^{\times}$  that

$$(-1)^{m^{+}} \frac{1}{4} \operatorname{vol}(K_{f}) \cdot \int_{X_{K}} \theta(g'; \varphi_{\mathrm{KM}} \otimes \varphi_{f}) \wedge \Omega^{m^{+}-1} = I(g'; \tilde{\varphi}_{\mathrm{KM}} \otimes \varphi_{f}).$$

Further, the integral on the right hand side over each component  $Y_j$  of  $X_K$  is identical. Hence, for any component  $Y_j$  the following identity holds

$$(-1)^{m^{+}} \frac{1}{4} \operatorname{vol}(K_{f}) \cdot |\hat{\mathbb{Z}}^{\times} : \operatorname{N}(K_{f})| \cdot \int_{Y_{j}} \theta(g'; \varphi_{\mathrm{KM}} \otimes \varphi_{f}) \wedge \Omega^{m^{+}-1} = I(g'; \tilde{\varphi}_{\mathrm{KM}} \otimes \varphi_{f}).$$

We would like to apply this result in the more classical setting of varieties defined at the infinite place over O as in [Bor98] and recall that for H = GSpin we find

$$1 \to \mathbb{Q}^{\times} \to H(\mathbb{Q}) \to \mathrm{SO}(V) \to 1,$$

as well as

$$H(\hat{\mathbb{Z}}) \to \Gamma'(\hat{L}) \to 1,$$

where  $\Gamma'(\hat{L})$  denotes the discriminant kernel in  $\mathrm{SO}(\hat{L})$ . For  $\Gamma \leq \mathrm{SO}(L)$ , we may consider the closure  $\overline{\Gamma} < \mathrm{SO}(V_{\mathbb{A}_f})$  and further select a cover  $K_f < H(\mathbb{A}_f)$ , containing  $\hat{\mathbb{Z}}^{\times}$ . In that case, we have that the Shimura variety  $X_K$  of the pair  $(H, K_f)$  is the same as the variety associated to  $(\mathrm{SO}, \overline{\Gamma})$  (cf. [BY21, p. 1669]):

$$\mathrm{SO}(V) \setminus \mathbb{D} \times \mathrm{SO}(V_{\mathbb{A}_f}) / \overline{\Gamma} \simeq H(\mathbb{Q}) \setminus \mathbb{D} \times H(\mathbb{A}_f) / K_f.$$

In the special case of  $K_f = \operatorname{GSpin}_{\hat{L}}$ , we find that for the choice of  $h_j = e$  being equal to the neutral element that  $Y_j \simeq Y'(L) = \Gamma'(L) \setminus \mathbb{D}$ , where  $\Gamma'(L)$  is the discriminant kernel in SO. In fact, if we specialise further, to the case of L splitting a hyperbolic plane, we find that the index  $|\hat{\mathbb{Z}}^{\times} : N(K_f)|$  equals 1 so that there is only one connected component identifying with  $Y'(L) = \Gamma'(L) \setminus \mathbb{D}$ . All in all, we obtain the following corollary.

**Corollary 5.26.** Let (L, q) be an even quadratic lattice of signature  $(m^+, 2)$  and select  $K_f = \operatorname{GSpin}_{\hat{L}}$ . Further, write  $Y_L = \Gamma(L) \setminus \mathbb{D}$  for the variety associated to the discriminant kernel  $\Gamma(L)$  in O(L) and select an element  $\varphi_f \in \mathcal{S}(V_{\mathbb{A}_f})$  that is invariant under  $\Gamma(L)$ . Then

$$\int_{Y_L} \theta(g'; \varphi_{\mathrm{KM}} \otimes \varphi_f) \wedge \Omega^{m^+ - 1} = (-1)^{m^+} \frac{4c_K}{|\hat{\mathbb{Z}}^{\times} : \mathrm{N}(K_f)|} \cdot I(g', \tilde{\varphi}_{\mathrm{KM}} \otimes \varphi_f), \qquad (5.31)$$

where  $c_K = 1$  or 1/2, depending on whether  $\Gamma(L) \cap SO(V) \setminus \mathbb{D} \to \Gamma(L) \setminus \mathbb{D}$  is bijective or two to one.

Proof: If considered in the special orthogonal setting, meaning integrating over  $\Gamma'(L) \setminus \mathbb{D}$ instead of  $\Gamma(L) \setminus \mathbb{D}$  on the left hand side of (5.31), the statement follows immediately from Proposition 5.25. The only task remaining is hence the translation to the orthogonal setting. By assumption, the integrand is invariant under pullback by elements from  $\Gamma(L)$ . In fact, the form  $\Omega$  is invariant under  $O(V_{\infty})$  and  $\varphi_{\rm KM}$  fulfils (5.23). Further,  $\varphi_f$ was chosen to be invariant under  $\Gamma(L)$  as well, so that we may translate the domain of integration by  $\gamma \in \Gamma'(L) \setminus \Gamma(L)$  and combine a fundamental domain for  $\Gamma'(L) \setminus \mathbb{D}$  from translates of a domain for  $\Gamma(L) \setminus \mathbb{D}$ . We obtain that the integral in question in (5.31) equals the integral over  $\Gamma'(L) \setminus \mathbb{D}$  divided by the multiplicity of the cover  $\Gamma'(L) \setminus \mathbb{D} \to \Gamma(L) \setminus \mathbb{D}$ . Finally, we employ the determinant map and the fundamental theorem on group homomorphism or apply Lemma A.8 with G = O(V), H = SO(V) and  $K = \Gamma(L)$  to

$$[\Gamma(L): \Gamma'(L)] \le [\mathcal{O}(V): \mathcal{SO}(V)] = 2,$$

meaning that the considered cover represents at most a double-cover.

find

# Part III

# Intertwining worlds

# 6 *L*-series

The main purpose of this chapter is to introduce and investigate symmetric square type L-series associated to vector valued modular forms (cf. Definition 6.77). After a brief overview of classical L-functions in 6.1, classical Hecke theory for  $\Gamma_1(N)$  modular forms is investigated and applied to derive properties of associated L-series. Afterwards, Hecke theory for vector valued modular forms is briefly reviewed based on [BS08] and properties of Hecke algebras arising in that context are verified. In the last section, L-series are associated to vector valued modular forms. We prove their convergence and realise them as a Rankin–Selberg integral to derive meromorphic continuation. In special cases, product expansions for these L-series are proven.

# 6.1 Introduction

The concept of an L-function is loosely speaking a generalisation of the Riemann  $\zeta$ -function

$$\zeta: \{s \in \mathbb{C} \mid \operatorname{Re}(s) > 1\} \to \mathbb{C}, \qquad s \mapsto \sum_{n=1}^{\infty} \frac{1}{n^s}$$

This famous function attributed to Riemann encodes deep number theoretic insights – the distribution of prime numbers among the naturals and it is equipped with remarkable analytic properties:

- An elementary bound, involving geometric sums, yields absolute convergence of the series for s ∈ C with Re(s) > 1. In this range, it converges normally and defines a holomorphic function without roots.
- For these  $s \in \mathbb{C}$ , the series has an alternative representation as a so called *Euler* product over the finite primes:

$$\zeta(s) = \prod_{p < \infty} \left( 1 - p^{-s} \right)^{-1}.$$
(6.1)

- The Riemann zeta function  $\zeta$  has a meromorphic continuation to the whole complex plane  $\mathbb{C}$  with a single pole in s = 1 with residue 1.
- The so called *completed Riemann zeta function* Λ(s) := π<sup>-s/2</sup>Γ(s/2)ζ(s) has a point symmetry around 1/2:

$$\Lambda(s) = \Lambda(1-s). \tag{6.2}$$

There is no universally accepted definition of *L*-functions. However, inspired by the properties of the Riemann  $\zeta$ -function and later applications, the following description delimits a rough notion of what to expect of an *L*-function.

An *L*-function should be a holomorphic function *L* on a right half plane  $\{s \in \mathbb{C} \mid \text{Re}(s) > \sigma\}$  for some  $\sigma \in \mathbb{R}$  such that the following properties are fulfilled.

a) There are rational functions  $Q_p$  for any prime p, such that

$$L(s) = \prod_{p < \infty} Q_p(p^{-s})$$

on some right half plane.

b) There exist rational functions  $r_e$  and  $r_1, \ldots, r_m$ , signs  $\pm_1, \ldots, \pm_m$ , and a constant  $C \in \mathbb{C}$ , such that

$$\Lambda(s) \coloneqq C \cdot e^{r_e(s)} \cdot \prod_{l=1}^m \Gamma(r_l(s))^{\pm_l} \cdot L(s)$$

admits meromorphic continuation to  $\mathbb{C}$  with only finitely many poles all lying along the real line.

c) There is a functional equation. Explicitly, there are  $\epsilon \in \mathbb{T}$  and  $k \in \mathbb{R}$  such that

$$\Lambda(s) = \epsilon \Lambda(1 - s + k).$$

An L-series should then be a series of the form

$$L(a,s) \coloneqq \sum_{n \in \mathbb{N}} \frac{a(n)}{n^s}$$

for a sequence  $a : \mathbb{N} \to \mathbb{C}$  with the potential to represent an *L*-function in its range of convergence.

In the following, we will have a brief look at some elementary notions and the classical example of a Dirichlet *L*-series associated to a Dirichlet character.

**Definition 6.1.** A function  $a : \mathbb{N} \to \mathbb{C}$  is called *weakly multplicative*, if  $a(m \cdot n) = a(m) \cdot a(n)$  for  $m, n \in \mathbb{N}$  with gcd(m, n) = 1.

- *Example* 6.2. i) Evidently, any multiplicative function  $\chi : \mathbb{N} \to \mathbb{C}$  is weakly multiplicative.
  - ii) The sum divisor function  $\sigma_s(n) = \sum_{d|n} d^s$  for  $s \in \mathbb{C}$  with  $\operatorname{Re}(s) > 0$  is weakly multiplicative.
  - iii) if  $\nu_p$  denotes the *p*-adic evaluation of an integer, then the function  $\mathbb{N} \ni n \mapsto \lambda(n) = (-1)^{\sum_{p < \infty} \nu_p(n)}$  is weakly multiplicative.
  - iv) The Möbius function

$$n \mapsto \mu(n) := \begin{cases} 0, & \text{if } n \text{ contains a square,} \\ (-1)^{\#\{p \mid \nu_p(n) > 0\}}, & \text{else.} \end{cases}$$
(6.3)

In compact and uncommon notation, we may write  $\mu(n) = \delta_{\not \exists \Box \mid n} \cdot \lambda(n)$ .

*Example* 6.3. a) If  $a, b : \mathbb{N} \to \mathbb{C}$  are weakly multiplicative, so is their product  $a \cdot b$ , and  $a^s$  for  $s \in \mathbb{C}$  for  $\operatorname{Re}(s) > 0$ . In particular,  $\mathbb{N} \ni n \mapsto \frac{a(n)}{n^s}$  is multiplicative.

b) If  $a, b : \mathbb{N} \to \mathbb{C}$  are weakly multiplicative, so is the convolution

$$[a * b](n) = \sum_{d|n} a(d)b(n/d)$$

This product naturally arises as the coefficient sequence of  $L(a, s) \cdot L(b, s) = L(a * b, s)$ . It is commutative, admits a neutral element and an inverse to a, if, and only if,  $a(1) \neq 0$ .

**Remark 6.4.** If a is weakly multiplicative and  $\sum_{n} a(n) \neq 0$  is absolutely convergent then

$$\sum_{n=1}^{\infty} a(n) = \prod_{p} \left( \sum_{l=0}^{\infty} a\left(p^{l}\right) \right).$$

If a is in fact multiplicative,

$$\sum_{n=1}^{\infty} a(n) = \prod_{p < \infty} \frac{1}{1 - a(p)}.$$

Example 6.5. For a Dirichlet character  $\chi$  the *L*-function  $L(\chi, s) := \sum_{n=1}^{\infty} \frac{\chi(n)}{n^s}$  converges absolutely for  $s \in \mathbb{C}$  with  $\operatorname{Re}(s) > 1$  by comparing it to the Riemann zeta function  $\zeta(s)$  and admits the following *Euler product* 

$$L(\chi, s) = \prod_{p < \infty} \frac{1}{1 - \chi(p)p^{-s}}.$$

In particular, for  $\chi = \chi_0$  the trivial character on  $(\mathbb{Z}/N\mathbb{Z})^{\times}$  we have

$$L(s, \chi_0) = \prod_{p|N} (1 - p^{-s}) \cdot \zeta(s).$$

Also, the series  $L(\chi, s)$  converge conditionally in case the character  $\chi$  is non-trivial<sup>1</sup> and never vanishes at s = 1.

Should the reader be interested in obtaining a better impression of Dirichlet series he may consider the compactly presented discussion in [Zag13, pp 1-15]. Further, *L*-series may also be associated to scalar valued modular forms. In fact, if  $f \in \mathcal{M}_k(\Gamma_0(N), \chi)$  has Fourier expansion

$$f(\tau) = \sum_{n \in \mathbb{N}_0} a(n)q^n$$

we may associate to it the L-series

$$L(f,s) \coloneqq \sum_{n \in \mathbb{N}} \frac{a(n)}{n^s}.$$
(6.4)

This series converges on a right half plane due to asymptotic bounds on the Fourier coefficients a(n) (cf. Corollary 3.76) and possesses a meromorphic continuation. There is even a converse theorem for such *L*-functions due to André Weil presented in [Bum98, Thm. 1.5.1 p. 60]. Further, by developing a theory of Hecke operators, which will be partially carried out in the following section, product expansions of such *L*-series may be constructed for certain classes of modular forms (cf. Theorem 6.45).

<sup>&</sup>lt;sup>1</sup>IN case of trivial character, it has a pole at s = 1.

## 6.2 Classical Hecke theory

In this section, we explore classical Hecke theory for  $\Gamma_1(N)$  modular forms. In fact, the theory is reduced to the case  $\Gamma_0(N)$  with character. Hecke operators are endomorphisms of modular forms and allow for proving that *L*-series associated to modular forms have not only meromorphic continuation, but also admit product expansions. For the content of this section, compare [Bum98, 1.4 pp. 41-53] and [Kna92, IX.6 p. 273].

We return to elliptic modular forms for a moment. These form, for fixed weight and congruence group, a finite dimensional vector space. In order to comprehend the structure of objects considering morphisms on them is a fundamental approach. Since these are finite dimensional vector spaces the ring of endomorphisms is, of course, represented by square matrices with entries in the ground field. This, however, does not provide any meaningful insight.

Hence, the aim is to naturally construct special algebras of operators on these spaces. Recall the Petersson slash operator of weight  $k \in \mathbb{Z}$  for  $\gamma \in \mathrm{GL}_2(\mathbb{R})^+$ :

$$|_k\gamma: \mathbb{C}^{\mathbb{H}} \to \mathbb{C}^{\mathbb{H}}, \qquad f \mapsto \left( z \mapsto f|_k \gamma(\tau) \coloneqq \det(\gamma)^{k/2} j(\gamma, \tau)^{-k} f(\gamma \tau) \right).$$

This defines a right action of  $\operatorname{GL}_2(\mathbb{R})^+$  that preserves smoothness conditions.<sup>2</sup> The subspace of  $\mathcal{C}^{\operatorname{hol}}(\mathbb{H},\mathbb{C})$  on which a subgroup  $\Gamma < \operatorname{GL}_2(\mathbb{Q})^+$  that is commensurable to  $\Gamma(1)$  operates as the identity comprises the space of weakly holomorphic modular forms of weight k to  $\Gamma$ .

These operators form the basis of constructing the desired endomorphism ring. Observe that for  $g \in \operatorname{GL}_2(\mathbb{Q})^+$  and  $f \in \mathcal{M}_k(\Gamma)$  we find

$$f|_k g \in \mathcal{M}_k(g^{-1}\Gamma g).$$

The basic idea is to construct an appropriate linear combination of operators  $|_k g$  in order to cancel terms to receive a form that is invariant under  $\Gamma$ , again. The first obvious candidate of an index set would be  $g\Gamma$  in order to achieve invariance by summation. However, this does obviously not converge in general so means of forcing convergence are to be sought. A naive idea for diminishing the index set of the sum would be utilising the invariance of f to quotient out  $\Gamma$  from the left. However,  $\Gamma$  then was required to operate on the index set  $g\Gamma$  so the set, indeed, had to be  $\Gamma g\Gamma$ . This, on the other hand, appears

<sup>&</sup>lt;sup>2</sup>Note that the centraliser  $\mathbb{R} \cdot \mathcal{I}$  operates trivially if k is even and  $\mathbb{R}^+ \cdot \mathcal{I}$  if k is odd causing spaces of automorphic forms to vanish if k is odd and  $-\mathcal{I} \in \Gamma$ .

to be a promising candidate. Now, a reasonable condition to force convergence is that the quotient  $\Gamma \setminus \Gamma g \Gamma$  is finite. There is a wide range of groups for which this is possible and uniformly computable: the so called *Hecke* or *congruence* subgroups (cf. Definition 2.18).

**Remark 6.6.** Let  $\Gamma \leq \Gamma(1)$  be a congruence subgroup and  $g \in \mathrm{GL}_2(\mathbb{Q})^+$ . Then there is  $M \in \mathbb{N}$  such that  $\Gamma(M) \leq g^{-1}\Gamma g$ . In particular,  $\Gamma \setminus \Gamma g \Gamma$  is finite.

Proof: Let  $\Gamma(N) \leq \Gamma$  and choose  $N_1, N_2$  such that  $N_1g$  and  $N_2g^{-1}$  have integer entries. Define  $M = NN_1N_2$  and write  $\gamma = \mathcal{I} + M\gamma'$  for  $\gamma \in \Gamma(M)$  where  $\gamma'$  has integer entries. Clearly,  $g\gamma g^{-1} = \mathcal{I} + N(N_1g)\gamma'(N_2g^{-1}) \in \Gamma(N) \leq \Gamma$ .

Further,  $|\Gamma \setminus \Gamma g\Gamma| = [\Gamma : g^{-1}\Gamma g \cap \Gamma]$ . The intersection, however, must contain a congruence group which has finite index in  $\Gamma(1)$  and hence in  $\Gamma$ .

This statement not only yields the desired finiteness result of the symmetrised sum discussed above, directly yielding the operators that had been advertised, but also that the property of being a congruence modular form is preserved by the action of  $\operatorname{GL}_2(\mathbb{Q})^+$ . In fact, every space of congruence modular forms can be injected into a space of modular forms for  $\Gamma_1(M)$  for an adequate  $M \in \mathbb{N}$  which is beneficial for explicit computations and stated below. Note that this operation preserves the property of vanishing at cusps (cf. Lemma 3.77).

**Remark 6.7.** Let  $\Gamma \leq \Gamma(1)$  be a congruence subgroup. Then there is  $g \in \mathrm{GL}_2(\mathbb{Q})^+$  and  $M \in \mathbb{N}$  such that  $\Gamma_1(M) \leq g^{-1} \Gamma g$ .

Proof: Let  $\Gamma(N) \leq \Gamma$  and  $g = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{GL}_2(\mathbb{Q})^+$  with entries in  $\mathbb{Z}$  such that  $N \mid a^2$  and  $N \mid c^2$ . To ensure the former condition, multiply with the least common multiple of the denominators. Then for any  $\mathcal{I} + \gamma \in \Gamma_1(N \operatorname{det}(g))$  there is  $h \in \mathbb{N}$  such that

$$g(\mathcal{I} + \gamma)g^{-1} \equiv \mathcal{I} + \begin{pmatrix} a & b \\ c & d \end{pmatrix} \cdot \begin{pmatrix} 0 & h \\ 0 & 0 \end{pmatrix} \cdot \begin{pmatrix} d & -b \\ -c & a \end{pmatrix} \mod N$$
$$\equiv \mathcal{I} + h \cdot \begin{pmatrix} -ac & a^2 \\ -c^2 & ac \end{pmatrix} \mod N$$
$$\equiv \mathcal{I} \mod N.$$

Consequently,  $g\Gamma_1(N \det(g))g^{-1} \leq \Gamma(N) \leq \Gamma$ .

In particular, for any congruence modular form f of weight k there is  $g \in \mathrm{GL}_2(\mathbb{Q})^+$ and a natural number N such that  $f|_k g \in \mathcal{M}_k(\Gamma_1(N))$  ( $\iff f \in \mathcal{M}_k(g\Gamma_1(N)g^{-1})$ ). In fact, the reduction may be taken further by harnessing a decomposition of  $\Gamma_1(N)$  modular forms into  $\Gamma_0(N)$  modular forms with Dirichlet character.

**Remark 6.8** ([Miy06, Lemma 4.3.1 p. 114]). Let  $\Gamma$  be commensurable to  $\Gamma(1)$  and assume there is  $\Gamma' \leq \Gamma$  such that  $\Gamma/\Gamma'$  is finite abelian. Then for  $R_k \in \{\mathcal{A}_k, \mathcal{M}_k^!, \mathcal{M}_k, \mathcal{S}_k, \mathcal{S}_k^\perp \leq \mathcal{M}_k\}$ 

$$R_k(\Gamma') = \bigoplus_{\chi \in (\Gamma/\Gamma')^*} R_k(\Gamma, \chi).$$

Proof: The group  $\Gamma$  acts on  $R_k(\Gamma')$  via  $\gamma \mapsto (f \mapsto f|_k \gamma)$ , descending to a representation of  $\Gamma/\Gamma'$  which decomposes by Example B.13 into irreducible representations, meaning characters as the quotient is abelian.

*Example* 6.9. For  $N \in \mathbb{N}$ :  $\Gamma_1(N) \leq \Gamma_0(N)$  and  $\Gamma_0(N)/\Gamma_1(N) = (\mathbb{Z}/N\mathbb{Z})^{\times}$  by

$$\Gamma_0(N) \to (\mathbb{Z}/N\mathbb{Z})^{\times}, \qquad \begin{pmatrix} a & b \\ c & d \end{pmatrix} \mapsto \overline{a}.$$

As a consequence, we have for instance

$$\mathcal{M}_k(\Gamma_1(N)) = \bigoplus_{\substack{\chi \text{ Dirichlet}\\ \text{mod } N}} \mathcal{M}_k(\Gamma_0(N), \chi).$$
(6.5)

This sum is exploited to justify only explicitly constructing the theory of Hecke operators for  $\mathcal{M}_k(\Gamma_0(N), \chi)$ . To do so, the Petersson slash operator will be altered for  $\Gamma_0(N)$  to include characters:

$$|_{k,\chi} \begin{pmatrix} a & b \\ c & d \end{pmatrix} : f \mapsto \left( z \mapsto f|_{k,\chi} \begin{pmatrix} a & b \\ c & d \end{pmatrix} (z) = \overline{\chi(a)} \det(\gamma)^{k/2} j(\gamma, z)^{-k} f(\gamma z) \right).$$

However, in order to treat modular forms with characters, the modified Petersson slash operator has to be extended (analogously to the extension of the regular one to  $\operatorname{GL}_2(\mathbb{Q})^+$ ). Hence, an extension of  $\Gamma_0(N) < \operatorname{GL}_2(\mathbb{Q})^+$  on which the Dirichlet character  $\chi \mod N$  operates naturally has to be constructed. For a matrix  $\begin{pmatrix} a & b \\ c & d \end{pmatrix}$  evaluating a character  $\chi \begin{pmatrix} a & b \\ c & d \end{pmatrix} = \chi(a)$  clearly fails for general rationals a but may be defined with ease in case that the denominator of a is coprime to N.

To this end let  $S_N$  be the prime divisors of N and  $\mathbb{Z}_{S_N^C}$  denote the localisation along all other primes (rendering them invertible). For  $r/s \in \mathbb{Z}_{S_N^C}$  write  $\chi(r/s) \coloneqq \chi(r)/\chi(s)$ , so  $\chi$ is applicable to elements of  $\mathrm{GL}_2(\mathbb{Z}_{S_N^C})^+$ . However, it fails to be a character of this group unless restricting to the subgroup of matrices with left lower entry being in  $N\mathbb{Z}_{S_N^C}$  (an analogue of  $\Gamma_0(N)$ ).

**Definition 6.10.** Define the following subgroup of  $\operatorname{GL}_2^+(\mathbb{Q})$ 

$$G_0(N) \coloneqq \left\{ \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{GL}_2(\mathbb{Z}_{S_N^C}) \ \middle| \ c \in N\mathbb{Z}_{S_N^C} \right\} \cap \operatorname{GL}_2^+(\mathbb{Q}).$$

Recall that by Remark 6.6 there are finitely many representatives  $g_i \in \mathrm{GL}_2(\mathbb{Q})^+$  giving rise to the following decomposition into right cosets

$$T_g \coloneqq \Gamma_0(N)g\Gamma_0(N) = \bigcup_i \Gamma_0(N)g_i.$$
(6.6)

Hence, the following is an immediate definition for a natural operator on spaces of modular forms.

**Definition 6.11.** Let  $g \in G_0(N)$ ,  $k \in \mathbb{Z}$  and  $\chi$  be a Dirichlet character for  $\Gamma_0(N)$  and assume the decomposition in (6.6). Then the associated *Hecke operator* is defined to be

$$|_{k,\chi}T_g: \mathcal{C}_k^{\alpha}(\Gamma_0(N),\chi) \to \mathcal{C}_k^{\alpha}(\Gamma_0(N),\chi), \quad f \mapsto \sum_i f|_{k,\chi}g_i.$$
(6.7)

**Remark 6.12.** The operator  $|_{k,\chi}T_g$  above descends to an endomorphism of the space of holomorphic modular forms  $\mathcal{M}_k(\Gamma_0(N),\chi)$  and of cusp forms  $\mathcal{S}_k(\Gamma_0(N),\chi)$ .

Now that there is a relatively natural family of operators on a wide spectrum of modular forms, the structure of their algebra has to be investigated. Ultimately, this will yield product expansions of L-functions associated to cusp forms (cf. Proposition 6.42).

For  $f \in \mathcal{C}_k^{\infty}(\Gamma_0(N), \chi)$  and  $\alpha, \beta \in G_0(N)$  consider the expression

$$f|_{k,\chi}T_{\alpha}T_{\beta} = \sum_{i,j} f|_{k,\chi}\alpha_i\beta_j.$$

Since f is invariant with respect to  $|_{k,\chi}\Gamma_0(N)$ ,  $\alpha_i\beta_j$  may be pooled in families  $\Gamma_0(N)\alpha_i\beta_j$ . However, choosing other  $\alpha_i$ , which amounts to multiplying them from the left with  $\gamma_i \in \Gamma_0(N)$ , does not change the family  $\Gamma_0(N)\alpha_i\beta_j$  and multiplying the collection  $\alpha_i$  from the right by a fixed element  $\gamma \in \Gamma_0(N)$  results in permuting the cosets in (6.6) yielding a family  $\gamma_i \in \Gamma_0(N)$  such that  $\{\alpha_i\}$  is replaced by  $\{\gamma_i\alpha_i\}$ . As a consequence, the number of elements  $\alpha_i\beta_j$  occuring in  $\Gamma_0(N)\alpha_i\beta_j$  is independent of the concrete choices for  $\alpha_i, \beta_j$  and may be denoted  $n(\alpha, \beta; \sigma)$  for some  $\sigma \in \Gamma_0(N)\alpha_i\beta_j$ . As a second consequence, it is independent under multiplying  $\sigma$  with an element  $\gamma \in \Gamma_0(N)$  from the right, yielding

$$\sum_{i,j} f|_{k,\chi} \alpha_i \beta_j = \sum_{\sigma \in \Gamma_0(N) \setminus G_0(N)} n(\alpha, \beta; \sigma) f|_{k,\chi} \sigma$$
$$= \sum_{\sigma \in \Gamma_0(N) \setminus G_0(N) / \Gamma_0(N)} n(\alpha, \beta; \sigma) f|_{k,\chi} T_{\sigma}.$$

This inspires defining a multiplication on the objects  $T_{\alpha}$  by the above formula. To be precise, let  $\mathscr{H}_0(N)$  denote the free abelian group over  $\{T_{\alpha} \mid \alpha \in \Gamma_0(N) \setminus G_0(N) / \Gamma_0(N)\}$ and define a multiplication via

$$T_{\alpha}T_{\beta} \coloneqq \sum_{\sigma \in \Gamma_0(N) \setminus G_0(N) / \Gamma_0(N)} n(\alpha, \beta; \sigma) \cdot T_{\sigma}.$$
(6.8)

This is immediately recognised to be associative by explicitly writing the triple product out yielding a sum of the form

$$T_{\alpha}T_{\beta}T_{\gamma} = \sum_{\sigma \in \Gamma_0(N) \setminus G_0(N) / \Gamma_0(N)} n(\alpha, \beta, \gamma; \sigma) \cdot T_{\sigma}.$$

**Definition 6.13.** The ring  $\mathscr{H}_0(N)$  is called *abstract Hecke algebra* for  $\Gamma_0(N)$ .

**Remark 6.14.** Usually the ring  $\mathscr{H}_0(N)$  is tensorised with  $\mathbb{Q}$  (or  $\mathbb{R}$ ) over  $\mathbb{Z}$  and becomes an algebra - there will be no distinction in notation.

In order to characterise the ring  $\mathscr{H}_0(N)$ , the different symbols  $T_{\alpha}$  are classified, first. This is a multi-step process and may be inferred from the base case N = 1. In that instance, the elementary divisor theorem yields the following.

**Remark 6.15.** The set  $\Gamma(1) \setminus \operatorname{GL}_2(\mathbb{Q})^+ / \Gamma(1)$  is faithfully represented by

$$\begin{pmatrix} D_1 & 0 \\ 0 & D_2 \end{pmatrix}$$

where  $D_1, D_2 \in \mathbb{Q}_{>0}$  such that  $D_1/D_2 \in \mathbb{N}$ . In particular,  $\Gamma(1)\alpha\Gamma(1) = \Gamma(1)\alpha^T\Gamma(1)$ .

Proof: The proof is completely contained in [Bum98, 1.4.2 p. 44].

**Remark 6.16.** The set  $\Gamma(1) \begin{pmatrix} D_1 & 0 \\ 0 & D_2 \end{pmatrix} \Gamma(1)$ , with  $D_1, D_2$  as above, equals

$$\left\{D_2 \cdot M \mid M = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in M_2(\mathbb{Z}), \gcd(a, b, c, d) = 1, \det(M) = D_1/D_2\right\}.$$

Proof: First,

$$\begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \cdot \begin{pmatrix} D_1 & 0 \\ 0 & D_2 \end{pmatrix} \cdot \begin{pmatrix} \alpha' & \beta' \\ \gamma' & \delta' \end{pmatrix} = \begin{pmatrix} \alpha D_1 \alpha' + \beta D_2 \gamma' & \alpha D_1 \beta' + \beta D_2 \delta' \\ \gamma D_1 \alpha' + \delta D_2 \gamma' & \gamma D_1 \beta' + \delta D_2 \delta' \end{pmatrix}$$

Assume  $D_2 = 1$  by factoring out. Obviously, a common divisor had to divide  $D_1$ , since the determinant is  $D_1$ . However, it cannot divide  $D_1$  since then w.l.o.g.  $\beta$  must be divisible by it, preventing  $\delta$  to be. But then  $\gamma'$  and  $\delta'$  had to, leading to a contradiction.

On the other hand, by Remark 6.15 any matrix in  $\operatorname{GL}_2(\mathbb{Q})^+$  is in such a double coset and  $D_1, D_2$  are determined by the greatest common divisor of the entries and the determinant  $D_1D_2$  (paired with the condition  $D_1/D_2 \in \mathbb{N}$ ).

With these statements as a foundation we may prove the general case. Recall that we write  $\mathbb{Z}_{S_N^C}$  for the localisation of  $\mathbb{Z}$  along all primes not dividing N. Explicitly, let  $R := \{z \in \mathbb{Z} \mid \gcd(z, N) = 1\}$ , then  $\mathbb{Z}_{S_N^C} = R^{-1}\mathbb{Z}$ .

**Remark 6.17.**  $\Gamma_0(N) \setminus G_0(N) / \Gamma_0(N)$  is faithfully represented by matrices

$$\begin{pmatrix} D_1 & 0 \\ 0 & D_2 \end{pmatrix}$$

with  $D_1, D_2 \in \mathbb{Z}_{S_N^C}^{\times} \cap \mathbb{Q}^+$  such that  $D_1/D_2 \in \mathbb{N}$ . Consequently, for all  $\alpha \in G_0(N)$ , we have  $\Gamma_0(N) \alpha \Gamma_0(N) = \Gamma_0(N) \alpha^T \Gamma_0(N)$ .

*Proof*: Let  $\alpha \in G_0(N)$  and select (by virtue of Remark 6.15)  $\gamma_1, \gamma_2 \in \Gamma(1)$  such that  $\alpha = \gamma_1 \delta \gamma_2$  with  $\delta = \operatorname{diag}(D_1, D_2)$  such that  $D_2 \mid D_1$  and  $D_1/D_2 \in \mathbb{N}$ .

By Remark 6.16 we find that N cannot have a common divisor with  $D_1$  and  $D_2$ , meaning  $gcd(D_1/D_2, N) = 1$ . Further, we find for an arbitrary element  $\gamma_D = \begin{pmatrix} a & b \\ c \cdot D_1/D_2 & d \end{pmatrix} \in \Gamma_0(D_1/D_2)$  that

$$\begin{pmatrix} D_1 & 0 \\ 0 & D_2 \end{pmatrix} \cdot \begin{pmatrix} a & b \\ c\frac{D_1}{D_2} & d \end{pmatrix} = \begin{pmatrix} aD_1 & bD_1 \\ cD_1 & dD_2 \end{pmatrix} = \begin{pmatrix} a & b\frac{D_1}{D_2} \\ c & d \end{pmatrix} \cdot \begin{pmatrix} D_1 & 0 \\ 0 & D_2 \end{pmatrix}.$$

Hence,  $\gamma_2 = \begin{pmatrix} a_2 & b_2 \\ c_2 & d_2 \end{pmatrix}$  may be altered by an element  $\gamma_D$  like above from the left, if  $\gamma_1$  is altered, respectively. The left lower entry of the resulting matrix  $\gamma_D\gamma_2$  is  $c\frac{D_1}{D_2}a_2 + dc_2$  and we endeavour to select c, d such that this number is congruent to zero modulo N. Let  $g \coloneqq \gcd\left(\frac{D_1}{D_2}a_2, c_2\right)$ . The Euclidean algorithm yields a solution (cN, dN) such that  $g\left(c\frac{D_1}{D_2g}a_2 + dc_2/g\right) \cdot N = gN$ . By passing to the solution  $(c', d') \coloneqq (cN \pm c_2, dN \mp \frac{D_1}{D_2}a_2)$ 

we obtain a tuple for which gcd(c', d') = 1, remaining valid for the pair  $\left(\frac{D_1}{D_2}c', d'\right)$ .<sup>3</sup> As a consequence, the Lemma of Bézout implies the existence of the desired matrix  $\gamma_D \in \Gamma_0(D_1/D_2)$ , such that  $\gamma_D \gamma_2 \in \Gamma_0(N)$ . Hence, w.l.o.g.  $\gamma_2 \in \Gamma_0(N)$  implying  $\gamma_1 = \alpha \gamma_2^{-1} \delta^{-1} \in G_0(N) \cap \Gamma(1) \leq \Gamma_0(N)$ .

This characterisation of the symbols appearing in the Hecke algebra is, in fact, more useful than it might initially appear to be. Indeed, it may be used to prove that the algebra  $\mathscr{H}_0(N)$  is commutative. First, however, the following preparatory assertion is stated.

**Remark 6.18.** The representatives  $g_i$  in Equation (6.6) may be chosen in such a fashion that

$$\Gamma_0(N)g\Gamma_0(N) = \bigcup \Gamma_0(N)g_i = \bigcup g_i\Gamma_0(N).$$

Proof: By Remark 6.17

$$\bigcup \Gamma_0(N)g_i = \Gamma_0(N)\alpha\Gamma_0(N) = \Gamma_0(N)\alpha^t\Gamma_0(N) = \bigcup g_i^t\Gamma_0(N)$$

and the matrices  $g_i, g_i^t$  generate the same double coset. Hence,  $g_i = \gamma g_i^t \gamma'$  for  $\gamma, \gamma' \in \Gamma_0(N)$ , so  $\gamma^{-1}g_i = g_i^t \gamma' \in \Gamma_0(N)g_i \cap g_i^t \Gamma_0(N)$  are the desired representatives.  $\Box$ 

**Proposition 6.19.** The algebra  $\mathscr{H}_0(N)$  is commutative.

*Proof*: The proof is, neglecting minor modifications, carried out in [Bum98, p. 45].  $\Box$ 

This algebra is not only remarkably well behaved, but the extent of naturality of the action on the spaces of modular forms is reflected in its compatibility with the Petersson product. For that insight recall the following expression for the Petersson product from Remark 2.23 where we write  $\tau = u + iv \in \mathbb{H}$ 

$$\langle f,g\rangle = \frac{1}{[\Gamma(1):\Gamma(N)]} \int_{\Gamma(N)\backslash\mathbb{H}} f(\tau)\overline{g(\tau)} v^k \frac{\mathrm{d}x \,\mathrm{d}y}{v^2}$$

defining a scalar product on  $\mathcal{S}_k(\Gamma_0(N), \chi)$ .

**Proposition 6.20.** The algebra  $\mathscr{H}_0(N)$  acts on  $\mathcal{S}_k(\Gamma_0(N), \chi)$  as a commutative algebra of normal operators. As such, it admits an orthogonal basis of simultaneous eigenforms of  $\mathcal{S}_k(\Gamma_0(N), \chi)$ . In case  $\chi = 1$ , the operators are self adjoint.

<sup>&</sup>lt;sup>3</sup>Note that if there was a divisor, it had to be a divisor of  $D_1/D_2$ , but then it also had to divide dN. However,  $D_1/D_2$  has neither a common divisor with N nor d.

The proof is split into the following two lemmas.

**Lemma 6.21.** For  $f, g \in S_k(\Gamma_0(N), \chi)$  and  $\alpha \in G_0(N)$  we have

$$\langle f | \alpha, g \rangle = \langle f, g | \alpha^{-1} \rangle.$$
 (6.9)

*Proof*: Recall that by Remark 6.6 there is  $\Gamma' \coloneqq \Gamma(N') \leq \alpha^{-1} \Gamma(N) \alpha$ . Since the measure is Moebius invariant the transformation  $\tau \mapsto \alpha^{-1} \tau$  yields

$$\begin{split} &\langle f | \alpha, g \rangle \\ \stackrel{(*)}{=} \frac{1}{[\Gamma(1) : \Gamma']} \int_{\Gamma' \setminus \mathbb{H}} f |_{k,\chi} \alpha(\alpha^{-1}\tau) \cdot \overline{g}(\alpha^{-1}\tau) \left( \frac{\det(\alpha^{-1})}{j(\alpha^{-1},\tau)\overline{j(\alpha^{-1},\tau)}} \right)^k y^k \frac{\mathrm{d}x \, \mathrm{d}y}{y^2} \\ &= \frac{1}{[\Gamma(1) : \Gamma']} \int_{\Gamma' \setminus \mathbb{H}} \frac{\overline{\chi(a)} \det(\alpha)^{k/2}}{j(\alpha, \alpha^{-1}\tau)^k} f(\tau) \cdot \overline{j(\alpha^{-1},\tau)^{-k}g(\alpha^{-1}\tau)} \left( \frac{\det(\alpha^{-1})}{j(\alpha^{-1},\tau)} \right)^k y^k \frac{\mathrm{d}x \, \mathrm{d}y}{y^2} \\ &= \frac{1}{[\Gamma(1) : \Gamma']} \int_{\Gamma' \setminus \mathbb{H}} \frac{\overline{\chi(a)}}{\chi(d/\det(\alpha))} f(\tau) \cdot \overline{g|_{k,\chi}\alpha^{-1}}(\tau) y^k \frac{\mathrm{d}x \, \mathrm{d}y}{y^2} \\ &= \langle f, g | \alpha^{-1} \rangle. \end{split}$$

In the last step  $\alpha \in G_0(N)$  has been used, so that  $\det(\alpha) = ad \mod N\mathbb{Z}_{S_N^C}$ . However, then

$$\frac{\overline{\chi(a)}}{\chi(d/\det(\alpha))} = \chi(\det(\alpha)/ad) = \chi(ad - bc)/\chi(ad) = 1.$$

In (\*) the group  $\Gamma'$  is replaced by  $\alpha \Gamma' \alpha^{-1} \leq \Gamma(N) \leq \Gamma(1)$ . By the invariance of the measure (used in  $\downarrow$ )

$$\operatorname{vol}(\mathcal{F}_{\Gamma(1)}) \cdot [\Gamma(1) : \alpha \Gamma' \alpha^{-1}] = \operatorname{vol}(\alpha \Gamma' \alpha^{-1}) \stackrel{\downarrow}{=} \operatorname{vol}(\Gamma') = \operatorname{vol}(\mathcal{F}_{\Gamma(1)}) \cdot [\Gamma(1) : \Gamma'],$$

so that descending to a Hecke subgroup suffices implying that this inconvenience could be ignored.  $\hfill \Box$ 

**Lemma 6.22.** The adjoint of  $|_{k,\chi}T_{\alpha}$  with  $\alpha = \text{diag}(D_1, D_2)$  as in Remark 6.17 with respect to the Petersson scalar product is given by

$$(|_{k,\chi}T_{\alpha})^* = M_{\chi(D_1D_2)}|_{k,\chi}T_{\alpha}.$$

Here,  $M_z$  for  $z \in \mathbb{C}$  denotes the multiplication operator by z.

*Proof*: Clearly, Equation (6.9) states that the expression is invariant under translat-

ing  $\alpha$  from the left or right by an element in  $\Gamma_0(N)$  implying that each element  $\beta \in \Gamma_0(N) \setminus \Gamma_0(N) \alpha \Gamma_0(N)$  may be replaced by  $\alpha$  under the scalar product. By Remark 6.6 the number of these is finite, say  $n_{\alpha} \in \mathbb{N}$ . Then, however, the statement is trivial for  $\mathcal{S}_k(\Gamma(1))$ , as in that case  $\det(\alpha)\alpha^{-1} = S\alpha S^{-1}$  and scalar matrices operate trivially yielding

$$\langle f|_k T_\alpha, g \rangle = n_\alpha \langle f|_k \alpha, g \rangle = n_\alpha \langle f, g|_k \alpha \rangle = \langle f, g|_k T_\alpha \rangle.$$

In case of  $\Gamma_0(N)$  with N > 1, however, the computation is more subtle. First note that scalar matrices act by multiplication with a character and, hence, transform nearly trivially:

$$(|_{k,\chi}T_{D_2\mathcal{I}})^* = \left(M_{\overline{\chi(D_2)}}\right)^* = M_{\chi(D_2)} = |_{k,\chi}T_{D_2^{-1}\mathcal{I}}.$$

As a consequence, only  $T_{\beta}$  with  $\beta = \beta(D_1) = \begin{pmatrix} D_1 & 0 \\ 0 & 1 \end{pmatrix}$  and  $D_1 \in \mathbb{N}$  remain to be considered. Select  $m_1, m_2, \in \mathbb{Z}$  such that

$$D_1^2 m_1 - N m_2 = 1$$

and notice that

$$\begin{pmatrix} D_1m_1 & m_2 \\ N & D_1 \end{pmatrix} \cdot \begin{pmatrix} 1 & 0 \\ 0 & D_1 \end{pmatrix} \cdot \begin{pmatrix} m_1 & m_2 \\ N & D_1^2 \end{pmatrix}^{-1} = \begin{pmatrix} D_1 & 0 \\ 0 & 1 \end{pmatrix}.$$

Employing that computation immediately yields the desired adjoint, where  $n_{\beta(D_1)} \in \mathbb{N}$  is chosen suitably:

$$\begin{split} \langle f|_{k,\chi} T_{\beta(D_1)}, g \rangle &= n_{\beta(D_1)} \langle f|_{k,\chi} \beta(D_1), g \rangle \\ &= n_{\beta(D_1)} \langle f, g|_{k,\chi} (D_1^{-1} \mathcal{I}) \beta(D_1) \rangle \\ &= \langle f, \chi(D_1) \cdot g|_{k,\chi} T_{\beta(D_1)} \rangle. \end{split}$$

Combining the computations above yields for  $\alpha = \text{diag}(D_1, D_2)$ 

$$(|_{k,\chi}T_{\alpha})^{*} = (|_{k,\chi}T_{D_{2}\mathcal{I}}T_{\beta(D_{1}/D_{2})})^{*} = M_{\chi(D_{1})}|_{k,\chi}T_{\beta(D_{1}/D_{2})} = M_{\chi(D_{1}D_{2})}|_{k,\chi}T_{\alpha},$$

meaning the operator  $|_{k,\chi}T_{\alpha}$  is in general normal. Further, it is self adjoint, if, and only if,  $\chi(D_1D_2) = 1$ . Note that the proof relied upon  $gcd(D_1, N) = 1$ .

We find the following application to the eigenvalues of Hecke operators.

Example 6.23. Note that the lemma above implies for the Hecke eigenvalue  $\sigma$  of  $T_{\alpha}$ 

$$\overline{\sigma} = \chi(D_1 D_2) \cdot \sigma. \tag{6.10}$$

This yields, in particular, for the angle  $\vartheta$  of  $\sigma$  in polar coordinates that  $2\vartheta(\sigma) \equiv -\vartheta(\chi(D_1D_2))$ , implying

$$\sigma \in \mathbb{R} \cdot \zeta_{2 \operatorname{ord}(\chi)} \subseteq \mathbb{R} \cdot \zeta_{2N}.$$
(6.11)

In particular, in case of  $\chi = 1$ , the value of  $\sigma$  must be real. If  $\chi$ , on the other hand, is quadratic, we find that  $\sigma$  must be real or purely imaginary, depending on whether  $\chi(D_1D_2)$  equals 1 or -1.

Proposition 6.20 has the significant application of yielding Euler products. In order to relate these operators directly to the coefficients of L-functions associated to modular forms, the operators are clustered in an appropriate fashion.

Recall that  $q\mathcal{I} \in (\mathbb{Z}_{S_N^C})_{>0}^{\times} \cdot \mathcal{I} < G_0(N)$  acts trivially via the Petersson slash operator as multiplication with  $\overline{\chi}(q)$ , so that any Hecke operator  $|_{k,\chi}T_{\alpha}$  is essentially represented by a matrix  $\alpha$  with integer entries. Further, let  $n \in \mathbb{N}$  with gcd(n, N) = 1 and denote for the moment<sup>4</sup> by T(n) the operator arising from summing over all  $T_{\alpha}$  where  $\alpha \in G_0(N)$  is of the form diag $(D_1, D_2)$ , such that  $D_2 \mid D_1$  (cf. Remark 6.17) and  $D_1, D_2 \in \mathbb{N}$ , with the limitation det $(\alpha) = D_1 \cdot D_2 = n$ .

By Definition 6.11

$$f|_{k,\chi}T(n) = \sum_{i} f|_{k,\chi}\beta_i \tag{6.12}$$

for suitable choices of  $\beta_i \in M_2(\mathbb{Z})$ . In fact, if M(n, N) denotes the subset of  $M_2(\mathbb{Z}) \cap G_0(N)$ consisting of matrices with determinant n we have that

$$M(n,N) = \bigcup_{i} \Gamma_0(N)\beta_i.$$
(6.13)

**Definition 6.24.** The  $\mathbb{Z}$  algebra generated by  $T_{\alpha}$  with  $\alpha \in G_0(N) \cap M_2(\mathbb{Z})$  is denoted  $\mathscr{H}_{0,\mathbb{Z}}(N)$ .

In order to carry out explicit computations, the following decomposition is useful.

**Lemma 6.25.** For  $D_1, D_2 \in \mathbb{N}$  as in Remark 6.17 there is the following disjoint decompo-

<sup>&</sup>lt;sup>4</sup>Note that this operator is renormalised in Definition 6.27!
sition

$$\Gamma_0(N) \begin{pmatrix} D_1 & 0\\ 0 & D_2 \end{pmatrix} \Gamma_0(N) = \bigcup_{\substack{a,d \in \mathbb{N}, ad = D_1 D_2\\b \mod d\\\gcd(a,b,d) = D_2}}^{\cdot} \Gamma_0(N) \begin{pmatrix} a & b\\ 0 & d \end{pmatrix}.$$
 (6.14)

In particular,

$$M(n,N) = \bigcup_{\substack{a,d \in \mathbb{N}, ad=n \\ b \mod d}}^{\cdot} \Gamma_0(N) \begin{pmatrix} a & b \\ 0 & d \end{pmatrix}.$$

*Proof*: Only (6.14) requires proof. The inclusion  $(\supseteq)$  is obvious by Remark 6.17.

Let  $g = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$  be an element of the left set. Then  $(\gamma, \delta) = (c, -a) / \gcd(a, c)$  has  $\gcd(a, c, N) = 1$ , so that  $N \mid c$  carries over to  $c/ \gcd(a, c)$ . Hence, the pair  $(\gamma, \delta)$  gives rise to a matrix  $\begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \in \Gamma_0(N)$ , such that the product  $\begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \cdot g$  has lower left entry  $(ca - ac)/\gcd(a, c) = 0$ , yielding w.l.o.g. c = 0. As a consequence, ad = n and possible multiplication with  $-\mathcal{I}$  forces a, d > 0, while  $\gcd(a, b, d) = D_2$  is deduced from Remark 6.16. Finally, multiplication with a suitable element from  $\Gamma_0(N)_{\infty} = \Gamma(1)_{\infty}$  yields  $0 \leq b < d$ .

Lastly, it remains to be verified that the decomposition is disjoint. Assume there were representatives  $\begin{pmatrix} a & b \\ 0 & d \end{pmatrix}$ ,  $\begin{pmatrix} a' & b' \\ 0 & d' \end{pmatrix}$  and  $\begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \in \Gamma_0(N)$  such that:

$$\begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \cdot \begin{pmatrix} a & b \\ 0 & d \end{pmatrix} = \begin{pmatrix} a' & b' \\ 0 & d' \end{pmatrix}.$$
 (6.15)

Solving yields

$$\begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} = \begin{pmatrix} a' & b' \\ 0 & d' \end{pmatrix} \begin{pmatrix} d & -b \\ 0 & a \end{pmatrix} \cdot \frac{1}{n}$$

from which we infer that  $\gamma = 0$  and  $\delta > 0$ . However, the matrix is an element of  $\Gamma_0(N) \leq \Gamma(1)$ , so  $\alpha = \delta = 1$ . Inserting this result in (6.15) yields  $b + d\beta = b'$  or, alternatively,  $b \equiv b' \mod d$ .

As an immediate application we find (cf. [Dei10, Prop. 2.5.4 p. 40]) the following.

**Lemma 6.26.** For  $m, n \in \mathbb{N}$  coprime to N with gcd(m, n) = 1 we have

$$T(mn) = T(m)T(n).$$

Recall that the operator T(m) is (up to now) empty in case (m, N) > 1.

*Proof*: Lemma 6.25 yields sets of representatives  $R_m, R_n$  for T(m) and T(n). Their product T(m)T(n) will, as set of representatives, admit the product set  $R_mR_n$ . That the

following mapping is well defined is clear by the definition of M(n, N) and Lemma 6.25.

$$R_m \times R_n \to \Gamma_0(N)_\infty \backslash R_{mn}, \qquad (A,B) \mapsto \Gamma_\infty \backslash AB.$$

For injectivity: assume there were (the relation  $\simeq$  in the following means that the matrices are identical module left operation of  $\Gamma_{\infty} = \langle T \rangle$ , where T denotes the translation matrix)

$$\begin{pmatrix} a_1 & b_1 \\ 0 & d_1 \end{pmatrix} \cdot \begin{pmatrix} a'_1 & b'_1 \\ 0 & d'_1 \end{pmatrix} = \begin{pmatrix} a_1a'_1 & a_1b'_1 + b_1d'_1 \\ 0 & d_1d'_1 \end{pmatrix} \\ \simeq \begin{pmatrix} a_2a'_2 & a_2b'_2 + b_2d'_2 \\ 0 & d_2d'_2 \end{pmatrix} = \begin{pmatrix} a_2 & b_2 \\ 0 & d_2 \end{pmatrix} \cdot \begin{pmatrix} a'_2 & b'_2 \\ 0 & d'_2 \end{pmatrix}.$$

Since the diagonal entries are identical and gcd(m, n) = 1 we deduce that indices do not matter for a and d. Hence, the identity of the right upper entries becomes

$$ab'_1 + b_1d' \equiv ab'_2 + b_2d' \mod dd'$$

Reduction modulo d' yields

$$ab'_1 \equiv ab'_2 \mod d'.$$

Again, by gcd(m, n) = 1 we infer gcd(a, d') = 1, yielding  $b'_1 \equiv b'_2 \mod d'$ . By assumption, however, this means  $b'_1 = b'_2$ , implying  $b_1d' \equiv b_2d' \mod dd'$  which reduces to  $b_1d' \equiv b_2d'$ mod d. Once more, gcd(d, d') = 1 yields  $b_1 = b_2$  by choice of the set of representatives. The surjectivity is implied once Lemma 6.25 is taken into consideration to deduce that  $|R_m| = \sigma_1(m)$  which is evidently weakly multiplicative.

Before employing the above results, the operator T(n) is renormalised.

**Definition 6.27.** For  $n \in \mathbb{N}$  with gcd(n, N) = 1 define  ${}^{k}T(n) \coloneqq n^{k/2-1} \cdot T(n)$ , and, more generally,  ${}^{k}T_{g} = det(g)^{k/2-1} \cdot T_{g}$ , so that

$$|_{k,\chi}{}^{k}T(n) \coloneqq n^{k/2-1} \cdot \sum_{\beta \in \Gamma_{0}(N) \setminus M(n,N)} |_{k,\chi}\beta.$$

With these modifications in place L-functions associated to modular forms (cf. (6.4)) may be investigated. First, the operation of the operators on Fourier expansions of modular forms is examined before exploiting the result to extract (incomplete) product expansions for associated L-series. Example 6.28. If  $\sum_{m=0}^{\infty} A(m)q^m = f \in \mathcal{M}_k(\Gamma_0(N), \chi)$  and  $n \in \mathbb{N}$  with gcd(n, N) = 1 the Fourier coefficients of  $f|_{k,\chi}{}^k T(n) = \sum_{m=0}^{\infty} B(m)q^m$  are given by

$$B(m) = \sum_{a | \gcd(m,n)} \overline{\chi(a)} a^{k-1} A\left(\frac{mn}{a^2}\right).$$

This is verified by the direct computation, where in the following b(d) runs through a set of representatives mod d.

$$\begin{split} f|_{k,\chi}{}^{k}T(n)(z) &= n^{k/2-1} \cdot \sum_{ad=n} \sum_{b \ (d)} \left(\frac{a}{d}\right)^{k/2} \overline{\chi(a)} f\left(\frac{az+b}{d}\right) \\ &= \sum_{ad=n} \overline{\chi(a)} \frac{a^{k}}{n} \sum_{m=0}^{\infty} A(m) e\left(\frac{amz}{d}\right) \cdot \sum_{b \ (d)} e\left(\frac{mb}{d}\right) \\ &= \sum_{m=0}^{\infty} \sum_{\substack{ad=n \\ d|m}} \overline{\chi(a)} \frac{a^{k}}{n} A(m) \cdot e\left(\frac{amz}{d}\right) \cdot d \\ &= \sum_{m=0}^{\infty} \sum_{\substack{ad=n \\ a|m}} \overline{\chi(a)} a^{k-1} A\left(\frac{md}{a}\right) \cdot e\left(mz\right) \\ &= \sum_{m=0}^{\infty} \sum_{\substack{a|gcd(m,n)}} \overline{\chi(a)} a^{k-1} A\left(\frac{mn}{a^{2}}\right) \cdot e\left(mz\right). \end{split}$$

With this example at hand, we may prove properties of eigenforms to Hecke operators.

**Definition 6.29.** Let  $N \in \mathbb{N}$  and  $\chi : \Gamma_0(N) \to \mathbb{T}$  be a Dirichlet Character modulo N. A form  $f \in \mathcal{S}_k(\Gamma_0(N), \chi)$  is referred to as a *Hecke eigenform*, if it is an eigenvector to T(n) for all  $n \in \mathbb{N}$  with gcd(n, N) = 1.

Hecke eigenforms have remarkable properties derived from the properties of the Hecke algebra.

**Proposition 6.30** ([Bum98, Prop 1.4.5 p. 48]). Assume  $f \in S_k(\Gamma_0(N), \chi)$  is a Hecke eigenform with Fourier coefficients a(m) and eigenvalues  $\lambda(n)$  for  ${}^kT(n)$ . Then

- i) In case N = 1 we have  $a(1) \neq 0$ .
- ii) If a(1) = 1, then  $\lambda(n) = a(n)$  for all n with gcd(n, N) = 1.
- iii) If a(1) = 1 the association  $n \mapsto a(n)$  for gcd(n, N) = 1 is weakly multiplicative.

*Proof*: By Example 6.28 we find for  $n \in \mathbb{N}$  with gcd(n, N) = 1 that

$$\lambda(n)a(m) = \sum_{d \mid \gcd(m,n)} \overline{\chi(d)} d^{k-1}a\left(\frac{mn}{d^2}\right).$$

Assuming gcd(m, n) = 1 we infer

$$\lambda(n)a(m) = a(mn).$$

Setting m = 1 provides  $\lambda(n)a(1) = a(n)$ . This implies ii) immediately and iii) after comparison with Lemma 6.26. In case of i) a(1) = 0 renders f = 0, contradicting the assumption of f being an eigenform.

Hecke eigenforms for which the first Fourier coefficient equals 1 are referred to as *normalised*. Their relevance is demonstrated by Proposition 6.31.

**Proposition 6.31.** If  $f \in S_k(\Gamma_0(N), \chi)$  is a normalised eigenform the associated L-function has a partial product expansion:

$$L(f,s) = \sum_{m=1}^{\infty} \frac{a(m)}{m^s} = \left(\sum_{\substack{m \in \mathbb{N} \\ p|m \Longrightarrow p|N}} \frac{a(m)}{m^s}\right) \cdot \prod_{p|N} \left(1 - a(p)p^{-s} + \overline{\chi(p)}p^{k-1-2s}\right)^{-1}.$$

Before proving this result, another beneficial application of Lemma 6.25 shedding light on the structure of  $\mathscr{H}_{0,\mathbb{Z}}(N)$  is presented.

**Lemma 6.32.** For  $r \in \mathbb{N}$  and  $k, p \nmid N$  as above we find

$$T(p^{r+1}) = T(p)T(p^r) - p \cdot T_{p \cdot \mathcal{I}} \cdot T(p^{r-1}).$$

The identity remains valid if the symbol T is replaced by  ${}^{k}T$ .

*Proof*: The proof is presented in Section A.3 based on the idea found in [Dei10, Prop 2.5.4 p 40].

*Example* 6.33. Employing Definition 6.11 and 6.27 in Lemma 6.32 yields for  $k, \chi, p$  as above and  $r \in \mathbb{N}$ 

$$|_{k,\chi}{}^{k}T(p){}^{k}T(p^{r}) = |_{k,\chi}{}^{k}T(p^{r+1}) + p^{k-1}\overline{\chi(p)} \cdot |_{k,\chi}{}^{k}T(p^{r-1}).$$
(6.16)

*Proof of Prop 6.31*: The weak multiplicativity of the coefficients for arguments coprime to N yields

$$\sum_{m \in \mathbb{N}} a(m)m^{-s} = \left(\sum_{\substack{M \in \mathbb{N} \\ p|m \Longrightarrow p|N}} \frac{a(m)}{m^s}\right) \cdot \prod_{p \nmid N} \left(\sum_{l=0}^{\infty} a(p^l)p^{-ls}\right).$$

For  $p \nmid N$  Equation 6.16 and  $\lambda(p) = a(p)$  yield for  $r \in \mathbb{N}$ 

$$a(p^{r+1}) - a(p)a(p^r) + \overline{\chi(p)}p^{k-1}a(p^{r-1}) = 0$$

translating to

$$\left(1 - a(p)X + \overline{\chi(p)}p^{k-1}X^2\right) \cdot \left(\sum_{r=0}^{\infty} a(p^r)X^r\right) = 1$$

as an identity of formal power series. Selecting  $X = p^{-s}$  yields the desired result.  $\Box$ 

**Proposition 6.34.** The algebra  $\mathscr{H}_{0,\mathbb{Z}}(N)$  has the following properties:

- a) It is commutative.
- b) It is generated (over  $\mathbb{Z}$ ) by  $T_{\mathcal{I}}$  and T(p) for primes p with  $p \nmid N$ .
- c) It acts normally on  $\mathcal{S}_k(\Gamma_0(N), \chi)$ .
- d)  $\mathscr{H}_0(N)$  and  $\mathscr{H}_{0,\mathbb{Z}}(N)$  generate the same algebra over  $\mathbb{Q}$ .

*Proof*: For a) and c) the properties are inherited from  $\mathscr{H}_0(N)$  (cf. Proposition 6.19, 6.20). Part b): Lemma 6.26 and Lemma 6.32 imply immediately that every element in  $\mathscr{H}_0(N)$  is generated by the operators mentioned in b).

Part d) is realised as follows: by Lemma 6.26 in case of r = 1, the operator  $T_{p\mathcal{I}}$  is reconstructable from T(p), but then  $T_{m\mathcal{I}}$  for  $m \in N$  with gcd(m, N) = 1 is. By the discussion before Definition 6.24 operators in  $\mathscr{H}_0(N)$  are decomposed into elements from  $\mathscr{H}_{0,\mathbb{Z}}(N)$  and  $T_{q\mathcal{I}}$  with  $q = \frac{q_1}{q_2} \in \mathbb{Z}_{S_N^{\mathbb{C}}}^{\times}$ . The latter act as multiplication with  $\chi(q_1)/\chi(q_2)$ , where  $\chi$  is a Dirichlet character mod N. However, there is  $m \in \mathbb{N}$  with gcd(m, N) = 1and  $\chi(m)\chi(q_2) = 1$ , meaning  $T_{q\mathcal{I}}$  acts as  $T_{m\mathcal{I}}$ .

Evidently, a full product expansion of L-series associated to a Hecke eigenform instead of the partial expansion of Proposition 6.31 is desirable, though cannot be expected in general, as the following example testifies.

Example 6.35. The modular form  $\Delta(\tau) + \Delta(6\tau) = f \in S_{12}(\Gamma_0(6), 1)$  is an eigenfunction of all Hecke operators  $T_g$  with  $g \in G_0(6)$ , since  $\Delta \in S_{12}$  which is one dimensional. The associated *L*-function L(f, s), however, has no Euler product, since its Fourier coefficients C(m) are not multiplicative. Indeed, C(2) and C(3) equal the Fourier coefficient  $C_{\Delta}(2), C_{\Delta}(3)$  of  $\Delta$  while C(6) equals  $C_{\Delta}(6) + C_{\Delta}(1) \neq C_{\Delta}(6) = C_{\Delta}(2) \cdot C_{\Delta}(3)$ .

However, in order to hope for a full product expansion, Hecke operators for primes dividing the level have to be constructed and investigated.

## 6.2.1 Dividing the level

We fix a level  $N \in \mathbb{N}$ . There is, in fact, a theory of Hecke operators T(p) for  $\Gamma_0(N)$  in case  $p \mid N$ . However, these are, in general, not normal and hence there is no basis of Hecke eigenforms for  $S_k(\Gamma_0(N), \chi)$  which have the strong properties stated above (cf. Example 6.35).

In order to extend the definition of Hecke operators T(n) to  $n \in \mathbb{N}$  with gcd(n, N) > 1note that the collection of matrices M(n, N) (cf. (6.13)) was empty, so it has to be redefined. In fact, instead of a congruence subgroup  $\Gamma$  acting on a suitable subgroup of  $GL_2(\mathbb{Q})^+$  the condition may be attenuated to semigroups (cf. [Shi71] or [Miy06, 4.5 p. 131]). This had implicitly been done before, when replacing  $G_0(N)$  with  $M_2(\mathbb{Z}) \cap G_0(N)$ in the investigation prior to this subsection. In the case at hand, define

$$\Delta_0(N) := \left\{ \alpha = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in M_2(\mathbb{Z}) \mid c \equiv 0 \mod N, (a, N) = 1 \right\} \cap \operatorname{GL}_2(\mathbb{Q})^{-1}$$

to extend  $G_0(N) \cap M_2(\mathbb{Z}) \subseteq \Delta_0(N)$ . Note that the conditions on a, c were chosen suitably to extend  $\chi$  to the semigroup. The semigroup  $\Delta_0(N)$  decomposes as before into the sets

$$M(n,N) := \left\{ \alpha = \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in M_2(\mathbb{Z}) \mid \det(\alpha) = n, c \equiv 0 \mod N, (a,N) = 1 \right\}.$$

In case of gcd(n, N) = 1, this agrees with the definition previously settled for, so it is, in fact, an extension. There is a geometric reason for the condition on a, emerging from symmetrising over superlattices of index n with a natural torsion condition (cf. [Kna92, IX.6 p 275]). Note that the condition on a stays intact, when such a matrix is multiplied from the left or right by an element of  $\Gamma_0(N)$ . The proof of Lemma 6.25 stays intact as

well and we find (cf. [Kna92, p. 277])

$$M(n,N) = \bigcup_{\substack{a,d \in \mathbb{N}, ad=n \\ (a,N)=1 \\ b \mod d}} \Gamma_0(N) \begin{pmatrix} a & b \\ 0 & d \end{pmatrix}.$$
 (6.17)

Now, if we define

$$f|_{k,\chi} \begin{pmatrix} a & b \\ c & d \end{pmatrix} = \overline{\chi(a)} \det(\alpha)^{k/2} j(\alpha, \tau)^{-k} f(\alpha \tau),$$

we obtain an extension of the action used above on  $\Gamma_0(N)$  that yields the same Hecke operators in case gcd(n, N) = 1. We attempt to reproduce the results of the case treated above and note that for the following [Miy06, p. 134] is helpful.

**Remark 6.36.** If  $\sum_{m=1}^{\infty} A(m)q^m = f \in \mathcal{M}_k(\Gamma_0(N), \chi)$  and  $n \in \mathbb{N}$ , the Fourier coefficients of  $f|_{k,\chi}{}^k T(n) = \sum_{m=0}^{\infty} B(m)q^m$  are given by

$$B(m) = \sum_{a | \gcd(m,n)} \overline{\chi(a)} a^{k-1} A\left(\frac{mn}{a^2}\right).$$

*Proof*: The computation is identical to that of Example 6.28:

$$\begin{split} f|_{k,\chi}{}^{k}T(n)(z) &= n^{k/2-1} \cdot \sum_{\substack{ad=n\\(a,N)=1}} \sum_{b \ (d)} \left(\frac{a}{d}\right)^{k/2} \overline{\chi(a)} f\left(\frac{az+b}{d}\right) \\ &= \sum_{\substack{ad=n\\(a,N)=1}} \overline{\chi(a)} \frac{a^{k}}{n} \sum_{m=0}^{\infty} A(m) e\left(\frac{amz}{d}\right) \cdot \sum_{b \ (d)} e\left(\frac{mb}{d}\right) \\ &= \sum_{m=0}^{\infty} \sum_{\substack{ad=n\\(a,N)=1\\d \mid m}} \overline{\chi(a)} \frac{a^{k}}{n} A(m) \cdot e\left(\frac{amz}{d}\right) \cdot d \\ &= \sum_{m=0}^{\infty} \sum_{\substack{ad=n\\(a,N)=1\\a \mid m}} \overline{\chi(a)} a^{k-1} A\left(\frac{md}{a}\right) \cdot e\left(mz\right) \\ &= \sum_{m=0}^{\infty} \sum_{\substack{al \equiv n\\(a,N)=1}} \overline{\chi(a)} a^{k-1} A\left(\frac{mn}{a^{2}}\right) \cdot e\left(mz\right). \end{split}$$

Note that  $\chi(a) = 0$  if (a, N) > 1, so the requirement may be eliminated from the sum.  $\Box$ 

**Definition 6.37.** The algebra of operators generated by  $T_{\alpha}$  for all  $\alpha \in \bigcup_{n \in \mathbb{N}} M(n, N)$  is denoted  $\overline{\mathscr{H}_{0,\mathbb{Z}}}(N)$ . It contains the algebra  $\mathscr{H}_{0,\mathbb{Z}}(N)$  from the previous section.

**Lemma 6.38.** For coprime  $m, n \in \mathbb{N}$  the Hecke operators are multiplicative:

$$T(m)T(n) = T(mn)$$

*Proof*: The first part of the proof is identical to the proof of Lemma 6.26. Only for identifying the map to fulfil surjectivity counting has to be reconsidered. Note that

$$|R_n| = \sum_{(a,N)=1,a|n} \frac{n}{a} = n \cdot \sum_{a|n/\gcd(n,N)} a^{-1} = \gcd(n,N) \cdot \sigma_1\left(\frac{n}{\gcd(n,N)}\right).$$

However,  $n \mapsto \gcd(n, N)$  is clearly weakly multiplicative, so that  $n \mapsto |R_n|$  is, yielding bijectivity from injectivity.

Together with the Lemma below the structure of T(m) where m only has prime divisors appearing in N is recognised to be trivial.

**Lemma 6.39.** If  $p \mid N$  the following reduction is valid  $T(p^r) = T(p)^r$ . It is evidently also true for  ${}^kT$  in place of T.

*Proof*: The proof is reduced to the identity  $T(p^{r+1}) = T(p)T(p^r)$ . The decomposition (6.17) yields the following sets of representatives for T(p) and  $T(p)^r$ 

$$R_p = \left\{ \begin{pmatrix} 1 & b \\ 0 & p \end{pmatrix} \middle| b \mod p \right\}, \qquad R_{p^r} = \left\{ \begin{pmatrix} 1 & b_r \\ 0 & p^r \end{pmatrix} \middle| b_r \mod p^r \right\}.$$

The product of representatives is, evidently,

$$\begin{pmatrix} 1 & b_r + p^r b \\ 0 & p^{r+1} \end{pmatrix}$$

However, this yields exactly the set of representatives  $R_{p^{r+1}}$ .

This yields a complete picture of the structure of the algebra  $\overline{\mathscr{H}_{0,\mathbb{Z}}}(N)$ .

**Corollary 6.40.** The algebra  $\overline{\mathscr{H}_{0,\mathbb{Z}}}(N)$  is commutative and generated (over  $\mathbb{Z}$ ) by the operators  $T_{q\mathcal{I}}$  for primes  $q \nmid N$  and T(p) for all primes p. It extends the action of  $\mathscr{H}_{0,\mathbb{Z}}(N)$ . Proof: This is an immediate consequence of Proposition 6.34, Lemma 6.38, and Lemma 6.39.

There is also an analogous result to Lemma 6.22 for bad primes that is more complicated and we refer to [Miy06, Thm. 4.5.4 p 136] The analogue of Proposition 6.30 is also true without the condition (n, N) = 1.

**Proposition 6.41.** Assume  $f \in S_k(\Gamma_0(N), \chi)$  is an eigenform for all  ${}^kT(n)$  with Fourier coefficients a(m) and eigenvalues  $\lambda(n)$  for  ${}^kT(n)$ . Then w.l.o.g. a(1) = 1 and

- i)  $\lambda(n) = a(n)$  for all  $n \in N$ ,
- ii) the association  $n \mapsto a(n)$  is weakly multiplicative.
- iii) For  $p \mid N$  and  $r \in \mathbb{N}$  we find necessarily  $a(p^r) = a(p)^r$ .
- iv) For  $p \nmid N$  and  $r \in \mathbb{N}$  we find  $a(p^{r+1}) = a(p)a(p^r) p^{k-1}\chi(p)a(p^{r-1})$ .

*Proof*: Except for iii) the proof ist identical to the proof of 6.30. Part iii), however, is a direct consequence of Lemma 6.39.

Further, a full product expansion of simultaneous eigenforms (cf. Prop 6.31) is obtained:

**Proposition 6.42.** If  $f \in S_k(\Gamma_0(N), \chi)$  is a simultaneous eigenform of  ${}^kT(n)$  for all  $n \in \mathbb{N}$  with Fourier coefficients a(n), a(1) = 1 may be assumed and the associated *L*-function has the product expansion:

$$L(f,s) = \sum_{m=1}^{\infty} \frac{a(m)}{m^s} = \prod_{p|N} \left(1 - a(p)p^{-s}\right)^{-1} \cdot \prod_{p \nmid N} \left(1 - a(p)p^{-s} + \overline{\chi(p)}p^{k-1-2s}\right)^{-1}.$$

Proof of Prop 6.42: From Proposition 6.31 follows, that only

$$\left(\sum_{p|m \Longrightarrow p|N} \frac{a(m)}{m^s}\right)$$

has to be factored. For  $p \mid N$  Part iii) of Proposition 6.41 translates to

$$(1 - a(p)X) \cdot \left(\sum_{r=0}^{\infty} a(p^r)X^r\right) = 1$$

as an identity of formal power series. Selecting  $X = p^{-s}$  yields the desired result.  $\Box$ 

Clearly, the question remains whether such simultaneous eigenfunctions exist. Example 6.35 already indicated a weakness in the construction: there exist Hecke eigenforms in

 $S_k(\Gamma_0(N), \chi)$  arising from cusp forms of different level or weight which may be effortlessly combined to violate multiplicative conditions. However, there is still a decomposition into eigenspaces.

**Theorem 6.43** (Hecke-Petersson). The space  $S_k(\Gamma_0(N))$  decomposes into a direct sum of eigenforms for T(p) for all primes p.

The above theorem also provides a *multiplicity one* statement for forms with nonvanishing first coefficient, as the Fourier coefficients are by Proposition 6.41 determined by the Hecke eigenvalues of T(p).

In fact, the result may even be improved. If the modular forms induced by lower weight and level are eliminated from consideration by restricting to their orthogonal complement, the theory of T(p) for primes not dividing the level N works as desired (cf. [AL70]). The remaining subspace of forms which do not arise in such a fashion is called space of *newforms*  $S_k^{\text{new}}(\Gamma_0(N), \chi)$ . It is stable under the operation of the Hecke algebra, admits an orthogonal basis of Hecke eigenforms and, if f is such an eigenform, its first Fourier coefficient is automatically different from zero. We restrict to the case of trivial character for the following statement.

**Theorem 6.44** (strong multiplicity one). The space  $S_k^{\text{new}}(\Gamma_0(N))$  admits a basis of normalised eigenforms for the operators T(p) for  $p \nmid N$ . Further, two normalised eigenforms  $f \in S_k^{\text{new}}(\Gamma_0(N))$  and  $g \in S_k(\Gamma_0(N'))$  having the same eigenvalues for all but finitely many T(p) result in N = N' and f = g. Additionally, in this case f is an eigenfunction for all T(p).

Compare [AL70, Thm 2 p 144 and Thm 4 p 150] for the origin of the theorem, [Miy71, Thm 2 p 185]<sup>5</sup> for a representation theoretic perspective.<sup>6</sup> The name multiplicity one arises from the fact that it is described in adelic terms as only admitting every smooth irreducible admissible representation once. This is true for  $GL_n$  (cf. [Jac70, Prop 11.1.1 p. 354] for the case n = 2 and [Sha74] for the generalisation to n > 2) and SL<sub>2</sub>. The part strong referes to the fact that these cuspidal automorphic representations are determined up to isomorphism locally and if they differ they must differ at infinitely many places, compare [Cas73, Thm. 2 p. 307] and [JS81].

The *L*-function associated to a newform does not only admit a product expansion but carries additional analytic properties (cf. [DS05, pp 200-205]).

<sup>&</sup>lt;sup>5</sup>note that normalised eigenforms are referred to as *primitive* in this source

<sup>&</sup>lt;sup>6</sup>For translating the representation theoretic perspective [Cas73, 3. p. 308] might be considered.

**Theorem 6.45.** Let  $f \in \mathcal{S}_k^{\text{new}}(\Gamma_0(N), \chi)$  be a normalised Hecke eigenform and L(f, s) denote its L-series. Then

- a) The series L(f, s) admits analytic continuation to all  $s \in \mathbb{C}$ .
- b) There is a product expansion

$$L(f,s) = \prod_{p < \infty} \left( 1 - a(p)p^{-s} + \overline{\chi}(p)p^{k-1-2s} \right)^{-1}$$

c) Assume f is an eigenfunction for the Fricke involution with eigenvalue  $\epsilon$ . The completed L-function  $L^*(f,s) \coloneqq \left(\frac{2\pi}{\sqrt{N}}\right)^{-s} \Gamma(s)L(f,s)$  satisfies the functional equation

$$L^*(f,s) = \epsilon \cdot i^k \cdot L^*(f,k-s).$$

That the form f is assumed in the last part to be an eigenform for the Fricke involution is no remarkable restriction. The Fricke involution is a special case of an Atkin–Lehner involution, commutes with the Hecke operators for good primes and is self adjoint [Kna92, Prop 9.19 p.281], so that these are simultaneously diagonisable. We will briefly review Atkin–Lehner involutions as these are helpful for computing product expansions of *L*-series associated to vector valued modular forms arising as a symmetrisation of scalar valued forms for  $\Gamma_0(N)$ .

## Atkin–Lehner involution

Atkin–Lehner involutions that already appeared briefly above, are of particular interest for investigating *L*-series associated to lifts of scalar valued modular forms in  $S_k(\Gamma_0(N), \chi)$ to the vector valued case. For the following, compare [AL70].

**Definition 6.46.** Let  $e \mid N$  be a divisor of N, such that gcd(e, N/e) = 1. Such a number is called a *Hall divisor* of N and this fact is denoted  $e \mid \mid N$ . For any Hall divisor  $e \mid \mid N$  select a matrix

$$\gamma_e = \begin{pmatrix} ae & b\\ cN & de \end{pmatrix} \in M_2(\mathbb{Z})$$

with determinant  $e^{7}$ . The induced operator is called an *Atkin–Lehner involution*:

$$\mathcal{M}_k(\Gamma_0(N),\chi) \to \mathcal{M}_k(\Gamma_0(N),\chi), \qquad f \mapsto f|_k \gamma_e.$$

**Remark 6.47.** a) Atkin Lehner involutions on  $\mathcal{M}_k(\Gamma_0(N), \chi)$  are well defined.

- b) For all  $e \parallel N$  the matrices  $\gamma_e$  are normalising  $\Gamma_0(N)$ .
- c) Different choices  $\gamma_e, \gamma'_e$  are equivalent mod  $\Gamma_0(N)$ .
- d) For  $e, h \mid \mid N$  we have  $\gamma_e \cdot \gamma_h = \gamma_{\operatorname{lcm}(e,h)}$ .

*Proof*: Note that for two different choices  $\gamma_e, \gamma'_e$  of matrices and  $\gamma \in \Gamma_0(N)$ 

$$\gamma_e \gamma \gamma'_e \in e^2 \cdot \Gamma_0(N).$$

Additionally,  $\gamma_e'' \coloneqq e\gamma_e'^{-1}$  is an alternative choice, meaning  $\gamma_e \gamma \gamma_e'^{-1} \in \Gamma_0(N)$  (\*). However, for an element  $\delta \in \Gamma_0(N)$ , select  $\gamma = \gamma_e^{-1} \delta \gamma_e' \in \Gamma_0(N)$  and find that  $\gamma_e \Gamma_0(N) \gamma_e^{-1} = \Gamma_0(N)$ , i.e. b).

For c), (\*) implies  $\gamma_e \gamma = \gamma' \gamma'_e$  for some  $\gamma' \in \Gamma_0(N)$ . By b), a modified  $\gamma' \in \Gamma_0(N)$  yields  $\gamma_e \gamma = \gamma'_e \gamma'$ , meaning  $\gamma_e \equiv \gamma'_e \mod \Gamma_0(N)$ .

Lastly, for a),  $f \in \mathcal{M}_k(\Gamma_0(N), \chi)$  together with b) (select  $\delta \in \Gamma_0(N)$  s.t.  $\delta \gamma_e = \gamma_e \gamma$ ) and the computation for c) imply  $f|_k \gamma_e = f|_k \delta \gamma_e = f|_k \gamma_e \gamma = f|_k \gamma'_e \gamma' = f|_k \gamma'_e$ . The preservation of vanishing conditions at cusps, however, is immediate.

## 6.3 Vector valued Hecke theory

So far we have had a look at Hecke operators acting on scalar valued modular forms in  $S_k(\Gamma_0(N), \chi)$ . However, in Section 3.3 we have introduced vector valued modular forms that are a collection of scalar valued forms interacting under transformation of the full modular group to guarantee automorphic behaviour. In this setting, *L*-series may also be defined and one might hope to reproduce some of the results in the classical case. For the purpose of deriving product expansions, specifically, an analogous theory to Hecke operators in the elliptic case has to be developed which has been carried out by [BS08].

<sup>&</sup>lt;sup>7</sup>Of course, this condition already implies that such a divisor e had to be a Hall divisor.

Recall that a vector valued modular form  $f \in \mathcal{M}_{\mathcal{L},k}(Mp_2(\mathbb{Z}))$  is associated to a discriminant form  $\mathcal{L}$  and has a Fourier expansion

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n\tau) \mathfrak{e}_{\lambda}.$$

We fix a  $\mathbb{Z}$  lattice L with  $\mathcal{L} = L'/L$ . In order to construct Hecke operators in the classical case, the action of the Petersson slash operator on a supergroup of  $\Gamma_0(N)$  was utilised. This action already exists and may be reused for acting on the component functions. The task that remains, however, is to extend the action of the Weil representation  $\rho_{\mathcal{L}}$  on  $\mathbb{C}[\mathcal{L}]$  to a supergroup of  $\mathrm{Mp}_2(\mathbb{Z})$ .

Assume  $\operatorname{sig}(\mathcal{L})$  is even so that we are in the case of  $\operatorname{SL}_2(\mathbb{Z})$  modular forms and write  $N := \operatorname{lev}(L)$ . Then the Weil representation is trivial on  $\Gamma(N)$ , meaning it factors through

$$\operatorname{SL}_2(\mathbb{Z}/N\mathbb{Z}) \simeq \operatorname{SL}_2(\mathbb{Z})/\Gamma(N)$$

Hence, a naive approach would be an extension of the Weil representation to  $\operatorname{GL}_2(\mathbb{Z}/N\mathbb{Z})$ . This, however, fails. Bruinier and Stein note that with the definitions  $S(N) \coloneqq \operatorname{SL}_2(\mathbb{Z}/N\mathbb{Z})$ ,  $G(N) \coloneqq \operatorname{GL}_2(\mathbb{Z}/N\mathbb{Z})$ , and  $U(N) \coloneqq (\mathbb{Z}/N\mathbb{Z})^{\times}$ , the following sequence is exact and splitting:

$$1 \to S(N) \to G(N) \to U(N) \to 1.$$

Hence, G(N) is a semidirect product of S(N) and U(N). Then they construct a group Q(N), such that there is a natural exact sequence

$$1 \to S(N) \to Q(N) \to U(N) \to 1$$

and, in fact,  $Q(N) \simeq S(N) \times U(N)$ . Hence, the authors choose a unitary representation of U(N) that commutes with  $\rho_{\mathcal{L}}$  on S(N) and have thus extended the representation. Their choice is consistent with the classical theory, in case the base lattice is unimodular and we will denote the extended representation by  $\rho_{\mathcal{L}}$ , again. They then construct a group Q(N) (cf. [BS08, (4.2) p. 257]) containing Mp<sub>2</sub>( $\mathbb{Z}$ ) and covering Q(N). This setting is finally suitable to reproduce the ideas of the scalar valued case.

The full picture is presented in [BS08, Sec. 3]. In the end, the authors prove that in the case we are interested in, the coset decomposition of the appearing double cosets are equivalent to the standard decomposition. We will briefly sketch the action of these operators after introducing notation and proving identities for their abstract counterparts. Note that in Lemma 6.32 and 6.39 relations of Hecke operators in the abstract Hecke algebra have been proven. We recall that the subalgebra  $\mathscr{H}_{0,\mathbb{Z}}(N)$  generated by the operators T(n) for  $n \in \mathbb{N}$  (cf. paragraph before (6.12)) was of particular interest for the theory of *L*-series and we would like to replicate part of these results in the vector valued setting. The operators considered by [BS08], however, correspond not to  $T(p^2)$ , but to so called *primitive* operators, serving as simpler building blocks of the abstract Hecke algebra. This justifies the new notation  $\mathcal{T}(p^2)$  in contrast to  $T(p^2)$  from the previous section.

**Definition 6.48.** For  $n \in \mathbb{N}$  with gcd(n, N) = 1, define the matrix

$$g(n) \coloneqq \begin{pmatrix} n & 0\\ 0 & 1 \end{pmatrix} \in \mathrm{GL}_2^+(\mathbb{R})$$

and in the notation of (6.8) set the *primitive* Hecke operator associated to n to be

$$\mathcal{T}(n) \coloneqq T_{g(n)}$$

Note that these operators may be used to decompose the previously defined operators T(n) appearing in (6.12).

**Remark 6.49.** Let  $n \in \mathbb{N}$  and p a prime not dividing lev(L), then Lemma 6.25 implies

$$T(p^n) = \sum_{l=0}^{\lfloor n/2 \rfloor} T_{p^l \cdot \begin{pmatrix} p^{n-2l} & 0 \\ 0 & 1 \end{pmatrix}} = \sum_{l=0}^{\lfloor n/2 \rfloor} T_{p^l \mathcal{I}} \cdot \mathcal{T}(p^{n-2l}).$$

Hence, for  $n \in \mathbb{N}$  with gcd(n, lev(L)) = 1 to understand the action of the family of operators T(n), it suffices to know how  $\mathcal{T}(n)$  and  $T_{n\mathcal{I}}$  act.

There is the following analog to Lemma 6.32.

**Lemma 6.50.** We find for  $r \in \mathbb{N}$  and a prime p not dividing N that

$$\mathcal{T}(p^{r+1}) = \mathcal{T}(p)\mathcal{T}(p^r) - p \cdot T_{p\mathcal{I}} \cdot \mathcal{T}(p^{r-1}) - \delta_{r=1} \cdot T_{p\mathcal{I}}.$$

*Proof*: See Section A.3 for details.

Note that this is consistent with [Miy06, Lem. 4.5.7 (1) p. 140], where  $T(p,p) = T_{p\mathcal{I}}$ and  $T(1, p^e) = \mathcal{T}(p^e)$  in our notation. Note also that the number N in Miyake may be assumed to equal 1 in the current subsection. By utilising the statement above, we may derive a relation for square indices which may prove useful later.

**Lemma 6.51.** For  $2 \leq r \in \mathbb{N}$  and a prime p not dividing N, the following relation is true:

$$\mathcal{T}(p^{r+2}) = \left[\mathcal{T}(p^2) + (1-p) \cdot T_{p\mathcal{I}}\right] \mathcal{T}(p^r) - \left(1 + \delta_{r=2}p^{-1}\right) \left(p \cdot T_{p\mathcal{I}}\right)^2 \mathcal{T}(p^{r-2}).$$

*Proof*: We apply Lemma 6.50 repeatedly to obtain the following:

$$\begin{aligned} \mathcal{T}(p^{r+2}) &= \mathcal{T}(p)\mathcal{T}(p^{r+1}) - p \cdot T_{p\mathcal{I}}\mathcal{T}(p^r) \\ &= \mathcal{T}(p) \left[\mathcal{T}(p)\mathcal{T}(p^r) - p \cdot T_{p\mathcal{I}}\mathcal{T}(p^{r-1})\right] - p \cdot T_{p\mathcal{I}}\mathcal{T}(p^r) \\ &= \left[\mathcal{T}(p^2) + (p+1) \cdot T_{p\mathcal{I}}\right]\mathcal{T}(p^r) \\ &- p \cdot T_{p\mathcal{I}} \left[\mathcal{T}(p^r) + (p+\delta_{r=2}) \cdot T_{p\mathcal{I}}\mathcal{T}(p^{r-2}) + \mathcal{T}(p^r)\right] \\ &= \left[\mathcal{T}(p^2) + (1-p) \cdot T_{p\mathcal{I}}\right]\mathcal{T}(p^r) - (1+\delta_{r=2}p^{-1}) \left(p \cdot T_{p\mathcal{I}}\right)^2 \mathcal{T}(p^{r-2}). \end{aligned}$$

In order to define an action of the abstract Hecke operators, also in the vector valued case, it is necessary to declare how the representatives in the double cosets of the operators  $\mathcal{T}(m)$  act on the group algebra  $\mathbb{C}[\mathcal{L}]$ . For  $m \in \mathbb{N}$  with gcd(m, lev(L)), the approach presented in [BS08] and sketched above yields the following action for any  $\lambda \in \mathcal{L}$ :

$$\mathbf{e}_{\lambda}|_{\mathcal{L}}g(m^2) = \mathbf{e}_{m\lambda}.\tag{6.18}$$

It also translates to the odd case, where the metaplectic cover is required and we set for  $m \in \mathbb{N}$  with gcd(m, lev(L)) = 1 analogously to the even case

$$g(m^2) \coloneqq \left( \begin{pmatrix} m^2 & 0\\ 0 & 1 \end{pmatrix}, 1 \right) \in \mathrm{Mp}_2^+(\mathbb{R}).$$
(6.19)

With this notation and the operation on  $\mathbb{C}[\mathcal{L}]$  obtained from linearising (6.18), we may define the action of the Hecke operator  $\mathcal{T}(m^2)$ .

**Definition 6.52.** Let  $m \in \mathbb{N}$  with gcd(m, lev(L)) = 1 and let  $g(m^2) \in Mp_2^+(\mathbb{R})$  be as above. Further, let

$$\operatorname{Mp}_2(\mathbb{Z}) \cdot g(m^2) \cdot \operatorname{Mp}_2(\mathbb{Z}) = \bigsqcup_i \operatorname{Mp}_2(\mathbb{Z}) \cdot h_i$$

be a right coset decomposition. Then we declare for  $f \in \mathcal{M}_{\mathcal{L},k}$  the following operation

$$f|_{L,k}\mathcal{T}(m^2) \coloneqq \det(g(m^2))^{k/2-1} \sum_i \sum_{\mu \in \mathcal{L}} (f|_k \delta_i) \otimes (\mathfrak{e}_{\mu}|_{\mathcal{L}} h_i) \,.$$

Note that unlike the scalar valued case, the factor  $\det(g(m))^{k/2-1}$  has directly been incorporated into the action instead of writing  ${}^{k}\mathcal{T}$  for a separate, rescaled operator. Since this is the only case required in the following, we decided to lighten the notation. We also note that  $|_{L,k}\mathcal{T}(m^2)$  equals exactly the primitive operator  $|_{k,A}T^*(m^2)$  in [BS08, (4.30)].

We turn towards describing how these Hecke operators manipulate the Fourier expansion of a modular form.

#### **Operation on Fourier coefficients**

First, we fix some notation for the following sections.

Notation 6.53. To lighten the notation write  $\Gamma(1)$  for  $\operatorname{Mp}_2(\mathbb{Z})$ . We fix an even  $\mathbb{Z}$  lattice (L, q) of rank m and level  $\operatorname{lev}(L)$ . Its discriminant form is denoted  $\mathcal{L} := L'/L$  and for  $n \in \mathbb{N}$  with  $\operatorname{gcd}(n, \operatorname{lev}(L)) = 1$ , we write  $|_{L,k}\mathcal{T}(n^2)$  for the Hecke operator represented by  $\overline{\Gamma(1)}g(n^2)\overline{\Gamma(1)}$  which is denoted by  $|_{k,\mathcal{L}}T(n^2)^*$  in [BS08] and represents the primitive Hecke operator associated to  $n^2$  in the existing literature. Also, we resort to eliminating the metaplectic group from the notation of the space of modular forms, writing  $\mathcal{M}_{L,k}$  or  $\mathcal{S}_{L,k}$  for the space of cusp forms, respectively.

Recall that for  $k \in \mathbb{Z}/2$  a vector valued modular form  $f \in \mathcal{M}_{L,k}$  has a Fourier expansion by Remark 3.21:

$$f(\tau) = \sum_{\lambda \in L'/L} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) e(n\tau) \mathfrak{e}_{\lambda}.$$
(6.20)

For once, understanding how the above Hecke operators manipulate the Fourier expansion of a modular form provides insights into the structure of the operators and arithmetic of the expansion. On the other hand, it has an immediate application to a family of L-series associated to modular forms. For that purpose, Bruinier and Stein have investigated the action of Hecke operators on the Fourier coefficients of such a form.

**Proposition 6.54** ([BS08, Prop. 4.3 p. 258]). Let L have even signature and p be a prime coprime to lev(L). Let  $f \in \mathcal{M}_{L,k}$  and denote the Fourier expansion as in (6.20). Then

$$f|_{L,k}\mathcal{T}(p^2) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \mathbb{Z} + \overline{\mathbf{q}}(\lambda)} b(\lambda, n) q^n \otimes \mathfrak{e}_{\lambda},$$

where

$$b(\lambda, n) = p^{2k-2}a(\lambda/p, n/p^2) + \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}p^{k-2}(p\delta_{p|n} - 1)a(\lambda, n) + a(p\lambda, p^2n)$$
(6.21)

and

$$\delta_{p|n} = \begin{cases} 1, & p \mid n, \\ 0, & p \nmid n. \end{cases}$$

Moreover, we understand that  $a(\lambda/p, n/p^2) = 0$  if  $p^2 \nmid n$ .

Here, for  $d \in \mathbb{N}$ , the Gauss sum  $G_{\mathcal{L}}(d) = \sum \lambda \in \mathcal{L}e(d\overline{q}(\lambda))$  is further described in Definition A.16 and below. Bruinier and Stein have also proven the case of odd signature.

**Theorem 6.55** ([BS08, Thm 4.10 p 263]). Let L have odd signature and p be a prime coprime to lev(L). Let  $f \in \mathcal{M}_{L,k}$  and denote the Fourier expansion as in (6.20). Then

$$f|_{L,k}\mathcal{T}(p^2) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \mathbb{Z} + \overline{\mathbf{q}}(\lambda)} b(\lambda, n) q^n \otimes \mathfrak{e}_{\lambda},$$

where

$$b(\lambda,n) = p^{2k-2}a(\lambda/p,n/p^2) + \epsilon_p^{\operatorname{sig}(\mathcal{L}) + (\frac{-1}{|\mathcal{L}|})} \left(\frac{p}{|\mathcal{L}|2^{\operatorname{sig}(\mathcal{L})}}\right) \left(\frac{-n}{p}\right) p^{k-3/2}a(\lambda,n) + a(p\lambda,p^2n)$$

$$(6.22)$$

and for an odd integer d we set

$$\epsilon_d = \begin{cases} 1, & d \equiv 1 \mod 4, \\ i, & d \equiv -1 \mod 4. \end{cases}$$

Moreover, we understand that  $a(\lambda/p, n/p^2) = 0$  if  $p^2 \nmid n$ .

Based upon this operation on the Fourier coefficients of vector valued modular forms  $f \in \mathcal{M}_{L,k}$ , product expansions of *L*-series associated to *f* may be established, provided that the form *f* is an eigenform of the respective Hecke operators. This approach is pursued in Subsection 6.4.2.

### 6.3.1 Dividing the level

Next, we will have to turn to the bad places, following Bruinier and Stein. In case  $p \mid N$ , the matrix  $g(p^2) = \begin{pmatrix} p^2 & 0 \\ 0 & 1 \end{pmatrix}$  does not belong to  $\operatorname{GL}_2(\mathbb{Z}/N\mathbb{Z})$ . However, the action of the case  $p \nmid N$  might be mimicked by defining for  $\lambda \in \mathcal{L}$  the operation

$$\mathbf{\mathfrak{e}}_{\lambda}|_{\mathcal{L}}g(p^2) \coloneqq \mathbf{\mathfrak{e}}_{p\lambda}.\tag{6.23}$$

This coincides with the action of  $g(p^2)$  in case of  $p \nmid \text{lev}(L)$  constructed in [BS08, Sec. 4] and extends to  $g(m^2)$  for  $m \in \mathbb{N}$ . Since the goal is to define associated Hecke operators, products  $\gamma g(m^2)\gamma' \in \overline{\Gamma(1)}g(m^2)\overline{\Gamma(1)}$  have to be considered, where we understand  $\overline{\Gamma(1)} = \text{Mp}_2(\mathbb{Z})$ . It is clear how to define their action, however, consistency remains to be verified. In the following we frequently drop the index  $\mathcal{L}$  from  $|_{\mathcal{L}}$  to improve readability.

**Proposition 6.56** ([BS08, Prop. 5.1]). For fixed  $\delta(m^2) = \gamma g(m^2)\gamma' \in \overline{\Gamma(1)}g(m^2)\overline{\Gamma(1)}$ different choices of  $\gamma, \gamma'$  result in the same action

$$\mathfrak{e}_{\lambda}|\gamma|g(m^2)|\gamma'|$$

*Proof*: Assume a different set of choices, such that  $\gamma g(m^2)\gamma' = \gamma_1 g(m^2)\gamma'_1$ . It is sufficient to assume  $\gamma' = \gamma_1 = 1$ , meaning  $\delta(m^2) = \gamma g(m^2) = g(m^2)\gamma'_1$  so that

$$\mathbf{e}_{\lambda}|\gamma g(m^2) = \mathbf{e}_{\lambda}|g(m^2)\gamma_1'. \tag{6.24}$$

Write  $\gamma = \left( \begin{pmatrix} a & b \\ c & d \end{pmatrix}, \pm \sqrt{c\tau + d} \right)$ , then necessarily  $\gamma'_1 = \left( \begin{pmatrix} a & b/m^2 \\ m^2c & d \end{pmatrix}, \pm \sqrt{m^2c\tau + d} \right)$ . As a consequence,  $b \equiv 0 \mod m^2$ , meaning  $\gamma \in \overline{\Gamma}^0(m^2)$  and  $\gamma' \in \overline{\Gamma}_0(m^2)$ .

However,  $\overline{\Gamma}^0(m^2)$  is generated by  $\overline{\Gamma}^0(m^2) \cap \overline{\Gamma}^0_0(N)$ ,  $T^{m^2}$  and U.<sup>8</sup> In the first case, the identity follows directly from a well known theorem by Borcherds, stating that matrices in  $\overline{\Gamma}^0_0(N)$  act by multiplication with a character in d and multiplication of the index of  $\mathfrak{e}_{\lambda}$  with d. For  $T^{m^2}$  the representation acts by multiplication, trivialising the computation. The delicate case is that of  $U = (\begin{pmatrix} 1 & 0 \\ 1 & 1 \end{pmatrix}, \sqrt{\tau + 1})$ . The operation of U is found in Lemma 2.3 of [BS08], so that the left hand side of (6.24) reads

$$\mathfrak{e}_{\lambda}|U|g(m^2) = \frac{1}{|\mathcal{L}|} \sum_{\mu,\nu \in \mathcal{L}} e\left(q(\mu) + \beta(\mu,\lambda-\nu)\right) \mathfrak{e}_{m\nu}.$$

<sup>&</sup>lt;sup>8</sup>Note that  $\gamma$  may be manipulated by multiplication with  $T^{lm^2}$  for suitable  $l \in \mathbb{Z}$  such that b is divisible by N, since (a, d) = 1 so that  $N \mid (l_1a + l_2d + b/m^2)$ . However, the same argument applies to altering c with  $U^l$  for  $l \in \mathbb{Z}$  adequately chosen.

Using the decomposition from Remark 1.36 yields

$$= \frac{1}{|\mathcal{L}|} \sum_{\substack{\mu \in \mathcal{L} \\ \nu'' \in \mathcal{L}_m}} \sum_{\substack{\nu' \in \mathcal{L}_m \\ \nu'' \in \mathcal{L}_m}} e\left(q(\mu) + \beta(\mu, \lambda - \nu'/m - \nu'')\right) \mathfrak{e}_{\nu'},$$

where  ${}_{m}\mathcal{L}$  denotes the *m*-torsion of  $\mathcal{L}$ . By a simple character argument (cf. Remark 1.51)

$$= \frac{|\mathcal{L}_m|}{|\mathcal{L}|} \sum_{\mu \in \mathcal{L}^m} \sum_{\nu \in \mathcal{L}_m} e\left(q(\mu) + \beta(\mu, \lambda - \nu/m)\right) \mathfrak{e}_{\nu}$$
$$= \frac{1}{|\mathcal{L}|} \sum_{\mu \in \mathcal{L}} \sum_{\nu \in \mathcal{L}_m} e\left(q(m\mu) + \beta(m\mu, \lambda - \nu/m)\right) \mathfrak{e}_{\nu}.$$

Considering the right hand side of (6.24) results in

$$\mathfrak{e}_{\lambda}|g(m^2)|U^{m^2} = \frac{1}{|\mathcal{L}|} \sum_{\mu,\nu\in\mathcal{L}} e\left(m^2 q(\mu) + \beta(\mu, m\lambda - \nu)\right)\mathfrak{e}_{\nu}.$$

Shifting  $\mu$  by an element  $\mu' \in \mathcal{L}_m$  does not change the sum over  $\mu$  meaning

$$= \frac{1}{|\mathcal{L}||\mathcal{L}_m|} \sum_{\mu,\nu\in\mathcal{L}} \sum_{\mu'\in\mathcal{L}_m} e\left(m^2 q(\mu) + \beta(\mu+\mu', m\lambda-\nu)\right) \mathfrak{e}_{\nu}.$$

However, the sum over  $\mu'$  is, as a sum over a character, non-vanishing if, and only if,  $\nu\in\mathcal{L}^m,$  so that

$$= \frac{1}{|\mathcal{L}|} \sum_{\mu \in \mathcal{L}} \sum_{\nu \in \mathcal{L}^m} e\left(m^2 q(\mu) + \beta(\mu, m\lambda - \nu)\right) \mathfrak{e}_{\nu}$$
$$= \frac{1}{|\mathcal{L}|} \sum_{\mu \in \mathcal{L}} \sum_{\nu \in \mathcal{L}^m} e\left(q(m\mu) + \beta(m\mu, \lambda - \nu/m)\right) \mathfrak{e}_{\nu}.$$

As a consequence, the natural extension of the action is meaningful.

**Definition 6.57.** For  $m \in \mathbb{N}$  and  $\delta(m^2) = \gamma g(m^2) \gamma' \in \overline{\Gamma} g(m^2) \overline{\Gamma}$  set

$$\mathfrak{e}_{\lambda}|\delta(m^2) \coloneqq \mathfrak{e}_{\lambda}|\gamma|g(m^2)|\gamma'.$$

Moreover this acts<sup>9</sup> weakly contravariantly in the sense of the following Proposition,

<sup>&</sup>lt;sup>9</sup>Note that it is not derived from an actual mathematical group action like in the case gcd(m, lev(L)) = 1.

being essential for the Hecke theory developed by Bruinier and Stein.

**Proposition 6.58** ([BS08, Prop. 5.4]). For coprime  $m_1, m_2 \in \mathbb{N}$  we find for elements  $\alpha \in \overline{\Gamma(1)}g(m_1^2)\overline{\Gamma(1)}$  and  $\beta \in \overline{\Gamma(1)}g(m_2^2)\overline{\Gamma(1)}$  that

$$\mathfrak{e}_{\lambda}|\alpha|\beta = \mathfrak{e}_{\lambda}|(\alpha\beta).$$

*Proof*: Write  $g_1 \coloneqq g(m_1^2), g_2 \coloneqq g(m_2^2)$  and  $\alpha = \gamma g_1 \gamma', \beta = \delta g_2 \delta'$ . By the elementary divisor Theorem we find  $\alpha\beta = \epsilon g_1 g_2 \epsilon'$ . In view of the definition of the action, the assertion reduces to the case  $\gamma = \delta' = 1$ . So far,

$$\alpha\beta = g_1\gamma'\delta g_2 = \epsilon g_1 g_2 \epsilon'. \tag{6.25}$$

First, assume  $\epsilon' = 1$ . Then  $\alpha\beta = g_1\gamma'\delta g_2 = \epsilon g_1g_2$  implies  $g_1\gamma'\delta = \epsilon g_1 \in \overline{\Gamma(1)}g_1\overline{\Gamma(1)}$ yielding with the definition of the action preceding this Proposition

$$\begin{aligned} \mathbf{\mathfrak{e}}_{\lambda} \mid g_{1}\gamma'\delta &= \mathbf{\mathfrak{e}}_{\lambda} \mid \epsilon g_{1} \\ \implies & \mathbf{\mathfrak{e}}_{\lambda} \mid g_{1}\gamma'\delta \mid g_{2} = \mathbf{\mathfrak{e}}_{\lambda} \mid \epsilon g_{1} \mid g_{2}. \end{aligned}$$

Clearly,  $|g_1|g_2 = |g_1g_2$ , so that the right hand side's action reads  $|\epsilon g_1g_2 = |\alpha\beta$ , while the left hand sides action is by definition and the slash operator defining an action of  $\overline{\Gamma(1)}$  $|g_1\gamma'|\delta g_2 = |\alpha|\beta$ .

Next, consider the case of general  $\epsilon'$ . Rearranging (6.25) yields

$$\delta g_2 \epsilon'^{-1} = \gamma'^{-1} g_1^{-1} \epsilon g_1 g_2.$$

However, since the matrix component of the left hand side has integral entries, so does the right hand. Since  $(m_1, m_2) = 1$  it may be inferred that  $\tilde{\delta} := \gamma'^{-1} g_1^{-1} \epsilon g_1 \in \overline{\Gamma(1)}$ . Introducing this notation results in

$$\delta g_2 = \tilde{\delta} g_2 \epsilon' \stackrel{(6.25)}{\iff} (g_1 \gamma') (\tilde{\delta} g_2) = \epsilon g_1 g_2.$$

However, the assertion has already been proven in case  $\epsilon' = 1$ , so that

$$\mathfrak{e}_{\lambda} \mid (g_1 \gamma') \mid (\delta g_2) = \mathfrak{e}_{\lambda} \mid (\epsilon g_1 g_2).$$

Acting from the right with  $\epsilon'$  renders this statement equivalent to the original.

218

For  $g(m^2) = \left( \begin{pmatrix} m^2 & 0 \\ 0 & 1 \end{pmatrix}, 1 \right) \in \mathrm{Mp}_2(\mathbb{Q})^+$ , pick a disjoint left coset decomposition

$$\overline{\Gamma(1)}g(m^2)\overline{\Gamma(1)} = \bigcup_i \overline{\Gamma(1)}h_i$$

**Definition 6.59.** For  $g(m^2)$  and  $h_i$  as above define the action of  $\mathcal{T}(m^2)$  on  $f \in M_{\mathcal{L},k}$  by

$$f \mapsto f|_{L,k} \mathcal{T}(m^2) \coloneqq \det(g(m^2))^{k/2-1} \cdot \sum_i \sum_{\lambda \in \mathcal{L}} f_\lambda|_k h_i \otimes \mathfrak{e}_\lambda|_{\mathcal{L}} h_i.$$
(6.26)

Before stating the main theorem of this section, a variant of Lemma 6.25 has to be established.

**Lemma 6.60.** Let  $m, n \in \mathbb{N}$  and  $R_m^*$  denote a set of representatives of  $\mathcal{T}(m) = T_{\begin{pmatrix} m & 0 \\ 0 & 1 \end{pmatrix}}$ . Then gcd(m, n) = 1 implies

$$R_m^* R_n^* = R_{mn}^*.$$

*Proof*: We begin with the case of  $\Gamma(1)$ . Equation (6.14) of Lemma 6.25 yields

$$R_m^* = \bigcup_{\substack{a,d \in \mathbb{N}, ad = m \\ b \mod d \\ \gcd(a,b,d) = 1}}^{\cdot} \Gamma(1) \begin{pmatrix} a & b \\ 0 & d \end{pmatrix}.$$
 (6.14)

The proof is heavily based upon that of Lemma 6.26, as translates of  $R^*$  form a partition of the set of representatives  $R_m$  appearing in its proof. First, we check that the following map is well defined:

$$R_m^* \times R_n^* \to \Gamma(1)_{\infty} \backslash R_{mn}^*, \qquad (A, B) \mapsto \Gamma(1)_{\infty} \backslash AB.$$

It is clear that

$$\begin{pmatrix} a & b \\ 0 & d \end{pmatrix} \cdot \begin{pmatrix} a' & b' \\ 0 & d' \end{pmatrix} = \begin{pmatrix} aa' & ab' + bd' \\ 0 & dd' \end{pmatrix}$$

is in  $R_{mn}$ , however, note that by gcd(m, n) = 1 we have

$$\gcd(aa', dd', ab' + bd') \mid \gcd(aa', dd') = \gcd(a, d) \cdot \gcd(a', d').$$

Assume  $p \mid \gcd(aa', dd', ab' + bd')$  then

$$p \mid \gcd(a,d) \implies p \mid bd' \implies p \mid b \implies p \mid \gcd(a,b,d) \perp,$$
$$p \mid \gcd(a',d') \implies p \mid ab' \implies p \mid b' \implies p \mid \gcd(a',b',d') \perp.$$

So the right hand side is, in fact, contained in  $R_{mn}^*$ . The proof of Lemma 6.26 shows that the mapping is bijective, if the non-primitive version  $R_m$  is considered. However, note that by Lemma 6.25

$$R_m = \bigcup_{\substack{a,d \in \mathbb{N}, ad = m \\ b \mod d}} \Gamma(1) \begin{pmatrix} a & b \\ 0 & d \end{pmatrix} = \bigcup_{d^2 \mid m} (d\mathcal{I}) \cdot R^*_{m/d^2},$$

so that it must, by  $d\mathcal{I} \in Z(\mathrm{GL}_2(\mathbb{R}))$ , be bijective on the level of  $R_m^*$ . In case of a square we find the identity

$$R_{m^2} = \bigcup_{\substack{a,d \in \mathbb{N}, ad = m^2 \\ b \mod d}} \Gamma(1) \begin{pmatrix} a & b \\ 0 & d \end{pmatrix} = \dot{\bigcup}_{d|m} (d\mathcal{I}) \cdot R^*_{(m/d)^2}, \tag{6.27}$$

which is used to explicitly partition  $T(m^2)$  essentially into primitive operators  $\mathcal{T}(\cdot^2)$ . Now in case of  $\overline{\Gamma(1)}$ , it suffices to consider only squares m, n. In that case, however, the computation is identical.

We also require the action of another element. Recall that we had defined the metaplectic element  $g(m^2) = \left( \begin{pmatrix} m^2 & 0 \\ 0 & 1 \end{pmatrix}, 1 \right)$  in (6.19) and define the following associated element

$$g^*(m^2) \coloneqq \left( \begin{pmatrix} 1 & 0 \\ 0 & m^2 \end{pmatrix}, m \right) = \overline{S}g(m^2)\overline{S}^{-1} \in \overline{\Gamma(1)}g(m^2)\overline{\Gamma(1)}.$$
(6.28)

**Lemma 6.61.** For  $\lambda \in \mathcal{L}$  and  $m \in \mathbb{N}$  we find

$$\mathfrak{e}_{\lambda}|_{\mathcal{L}}g^*(m^2) = \sum_{\substack{\sigma \in \mathcal{L} \\ m\sigma = \lambda}} \mathfrak{e}_{\sigma}$$
(6.29)

and conclude for the standard scalar product on  $\mathbb{C}[\mathcal{L}]$  and  $v_1, v_2 \in \mathbb{C}[\mathcal{L}]$  that

$$\langle v_1|_{\mathcal{L}}g(m^2), v_2 \rangle = \langle v_1, v_2|_{\mathcal{L}}g^*(m^2) \rangle, \qquad (6.30)$$

meaning these operators are adjoint to each other:  $(|_{\mathcal{L}}g(m^2))^* = |_{\mathcal{L}}g^*(m^2).$ 

*Proof*: By linearity it suffices to prove the identity on a basis of  $\mathbb{C}[\mathcal{L}]$ . Let  $\lambda, \mu \in \mathcal{L}$  and consider the following computation:

$$\langle \mathfrak{e}_{\lambda} | g(m^2), \mathfrak{e}_{\mu} \rangle = \langle \mathfrak{e}_{m\lambda}, \mathfrak{e}_{\mu} \rangle$$

$$= \delta_{m\lambda=\mu}$$
$$= \langle e_{\lambda}, \sum_{\substack{\sigma \in \mathcal{L} \\ m\sigma = \mu}} \mathfrak{e}_{\sigma} \rangle.$$

Note that the right entry of the scalar product may be realised as  $\mathfrak{e}_{\mu}|g^*(m^2)$ . In fact, we find by (3.34), (3.35), and  $\overline{S}^{-1} = \overline{S}^7 = Z^3 \overline{S} = \left( \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}, -i\sqrt{\tau} \right)$  that

$$\begin{aligned} \mathbf{\mathfrak{e}}_{\mu}|_{\mathcal{L}} \overline{S}g(m^2) \overline{S}^{-1} &= \frac{e(-\operatorname{sig}(\mathcal{L})7/8)}{\sqrt{\mathcal{L}}} \sum_{\nu \in \mathcal{L}} e(\mathbf{b}(\mu,\nu)) \mathbf{\mathfrak{e}}_{m\nu} | \overline{S}^{-1} \\ &= \frac{1}{|\mathcal{L}|} \cdot \sum_{\nu \in \mathcal{L}} e(\beta(\mu,\nu)) \sum_{\sigma \in \mathcal{L}} e(-\mathbf{b}(m\nu,\sigma)) \mathbf{\mathfrak{e}}_{\sigma} \\ &= \frac{1}{|\mathcal{L}|} \cdot \sum_{\nu,\sigma \in \mathcal{L}} e(\mathbf{b}(\mu-m\sigma,\nu)) \mathbf{\mathfrak{e}}_{\sigma} \\ &= \sum_{\substack{\sigma \in \mathcal{L} \\ m\sigma = \mu}} \mathbf{\mathfrak{e}}_{\sigma}, \end{aligned}$$

so that

$$g^*(m^2) = \det(g(m^2))g(m^2)^{-1} = \overline{S}g(m^2)\overline{S}^{-1}$$

acts adjointly to  $g(m^2)$ .

As a consequence, we obtain the following adjointness on the space of modular forms. **Corollary 6.62.** Let  $f, h \in S_{L,k}(\overline{\Gamma(1)}), m \in \mathbb{N}$  and  $g(m^2)$  as above. Then we find for every  $\alpha \in \overline{\Gamma(1)}g(m^2)\overline{\Gamma(1)}$  that

$$\langle f|_{L,k}\alpha,h\rangle = \langle f,h|_{L,k}\det(\alpha)\alpha^{-1}\rangle.$$
 (6.31)

The expression does not depend on the choice of  $\alpha$  in the double coset.

*Proof*: Recall that

$$\langle f|_{L,k}\alpha,h
angle = \sum_{\lambda,\mu\in\mathcal{L}} \langle f_{\lambda}|_k \alpha,h_{\mu}
angle \langle \mathfrak{e}_{\lambda}|\alpha,\mathfrak{e}_{\mu}
angle.$$

Assume first that  $\alpha \in \overline{\Gamma(1)} = \operatorname{Mp}_2(\mathbb{Z})$ . Note that the scalar valued case of Corollary 6.62 as in Lemma 6.21 is also valid for  $\Gamma(N)$  – the proof is identical but simpler, as there is no character. Recall that the component functions satisfy  $f_{\lambda}, h_{\mu} \in \mathcal{S}_k(\Gamma(\operatorname{lev}(L)))$  and combine this with the unitaricity of the Weil representation for the desired result.

In the general case, by the contravariance of the slash operator (cf. Definition 6.57), and invariance of f and h under Mp<sub>2</sub>( $\mathbb{Z}$ ), we may reduce to the choice  $\alpha = g(m^2)$ . This, however is solved by the same argument as above combined with Lemma 6.61 in the components of the group algebra.

Combining the results above leads to the following observation which is also stated in [BS08, Thm. 5.6 p. 269].

**Theorem 6.63.** The operation of  $\mathcal{T}(m^2)$  for arbitrary  $m \in \mathbb{N}$  determines a linear, cusp form preserving action on the space of modular forms. Further, it is identified to be self adjoint with respect to the Petersson scalar product and the operators satisfy

$$\mathcal{T}(m_1^2)\mathcal{T}(m_2^2) = \mathcal{T}(m_1^2 m_2^2)$$

for coprime elements  $m_1, m_2 \in \mathbb{N}$ .

We install the following piece of notation in order to state two corollaries.

**Definition 6.64.** For  $M \in \mathbb{N}$  let  $\mathcal{H}^2_M$  denote the Hecke algebra generated by all  $\mathcal{T}(p^2)$ and  $T_{p\mathcal{I}}$  for primes p with gcd(p, M) = 1.

**Corollary 6.65.** For any  $M \in \mathbb{N}$ , the algebra  $\mathcal{H}^2_M$  acts as a commutative algebra of cusp form preserving linear operators.

The Corollary follows from Theorem 6.63. In conjunction with Lemma 6.51 we conclude that for any prime p with  $p \nmid M \operatorname{lev}(L)$ , the algebra  $\mathcal{H}^2_M$  contains  $\mathcal{T}(p^{2r})$  for all  $r \in \mathbb{N}$ .

Corollary 6.65 immediately implies the existence of simultaneous eigenforms.

**Corollary 6.66.** Let  $M \in \mathbb{N}$  and assume  $W \leq S_{L,k}$  is a nonzero subspace that is invariant under the operation of  $\mathcal{H}^2_M$ . Then W has an orthonormal basis consisting of simultaneous eigenforms of  $\mathcal{H}^2_M$ .

We turn towards proving Theorem 6.63.

*Proof*: The first assertion is immediate. For the proof of the second claim, let  $f, h \in S_{L,k}$ and apply Corollary 6.62 to see that there must be a natural number  $n_m \in \mathbb{N}$  such that

$$\langle f|_{L,k} \mathcal{T}(m^2), h \rangle = \det(g(m^2))^{k/2-1} \cdot n_m \cdot \langle f|_{L,k} g(m^2), h \rangle$$

$$= \det(g(m^2))^{k/2-1} \cdot n_m \cdot \langle f, h|_{L,k} \overline{S}g(m^2) \overline{S}^{-1} \rangle$$

$$= \det(g(m^2))^{k/2-1} \cdot n_m \cdot \langle f, h|_{L,k} g(m^2) \rangle$$

$$= \langle f, h|_{L,k} T_{g(m^2)} \rangle.$$

The last assertion is a consequence of Lemma 6.60 and the fact that the contravariance of the slash operator extends to the definition for  $p \mid \text{lev}(L)$  in case of coprime numbers as in Proposition 6.58. The product  $\mathcal{T}(m_1^2)\mathcal{T}(m_2^2)$  will, as set of representatives, admit the product set  $R_{m_1^2}^*R_{m_2^2}^*$  which, by the stated lemma, equals  $R_{(m_1m_2)^2}^*$  representing the operator  $\mathcal{T}((m_1m_2)^2)$ .

Now that the existence of eigenforms is certain, a bound on the respective eigenvalues was advantageous. The following method for such bounds was discovered by Kohnen [Koh87] even though the bound itself had been known before [Wei84, Sec. 6].

**Lemma 6.67.** Let  $m \in \mathbb{N}$  and  $f \in S_{L,k}(\overline{\Gamma(1)})$  be a nonzero eigenform of  $\mathcal{T}(m^2) = T_{g(m^2)}$ with eigenvalue  $\sigma_m$ , where  $g(m^2) = \left( \begin{pmatrix} m^2 & 0 \\ 0 & 1 \end{pmatrix}, 1 \right)$ . Then

$$|\sigma_m| < m^{k-2} |\overline{\Gamma(1)} \setminus \overline{\Gamma(1)} g(m^2) \overline{\Gamma(1)}| \cdot |_m \mathcal{L}|.$$
(6.32)

In the particular case where m = p is a prime, we find

$$|\sigma_p| < p^{k-1} \cdot (p+1) \cdot |_p \mathcal{L}|. \tag{6.33}$$

Here,  ${}_{m}\mathcal{L}$  for a natural number m denotes the m-torsion of  $\mathcal{L}$ .

*Proof*: Recall the definition of  $g^*(m^2)$  from (6.28) and note that for  $\lambda \in \mathcal{L} = L'/L$  we have by Lemma 6.61 that

$$\mathfrak{e}_{\lambda}|_{\mathcal{L}}g^{*}(m^{2})|_{\mathcal{L}}g(m^{2}) = \sum_{\substack{\sigma \in \mathcal{L} \\ m\sigma = \lambda}} \mathfrak{e}_{\sigma}|_{\mathcal{L}}g(m^{2}) = \delta_{\lambda \in {}^{m}\mathcal{L}} \cdot |_{m}\mathcal{L}| \cdot \mathfrak{e}_{\lambda}.$$

First, note that Cauchy–Bunyakowsky–Schwarz implies in conjunction with Corollary 6.62 that

$$||f|_{L,k}g^*(m^2)||_2^2 \le ||f|_{L,k}g^*(m^2)g(m^2)||_2 \cdot ||f||_2.$$

Further,

$$\begin{split} \|f|_{L,k}g^{*}(m^{2})g(m^{2})\|_{2}^{2} \\ = \sum_{\lambda,\mu\in\mathcal{L}} \langle f_{\lambda}|_{k}g^{*}(m^{2})g(m^{2}), f_{\mu}|_{k}g^{*}(m^{2})g(m^{2})\rangle \langle \mathfrak{e}_{\lambda}|_{k}g^{*}(m^{2})g(m^{2}), \mathfrak{e}_{\mu}|_{k}g^{*}(m^{2})g(m^{2})\rangle \end{split}$$

$$= \sum_{\lambda,\mu\in^{m}\mathcal{L}} \langle f_{\lambda}, f_{\mu} \rangle |_{m} \mathcal{L} |^{2} \langle \mathfrak{e}_{\lambda}, \mathfrak{e}_{\mu} \rangle$$
$$\leq |_{m} \mathcal{L} |^{2} ||f||_{2}^{2}.$$

In particular, in case of gcd(m, lev(L)), we have equality in the last line. Here,  ${}_{m}\mathcal{L}$  and  ${}^{m}\mathcal{L}$  denote the *m*-torsion and *m*-multiples in  $\mathcal{L}$ , respectively. Also compare Remark 1.36. Second,

$$\sigma_{m} \cdot \langle f, f \rangle = \left\langle \mathcal{T}(m^{2})f, f \right\rangle$$
$$= m^{k-2} \sum_{h \in \overline{\Gamma(1)} \setminus \overline{\Gamma(1)}g(m^{2})\overline{\Gamma(1)}} \langle f|_{k}h, f \rangle$$
$$= m^{k-2} \cdot |\overline{\Gamma(1)} \setminus \overline{\Gamma(1)}g(m^{2})\overline{\Gamma(1)}| \cdot \left\langle f|_{k}g(m^{2}), f \right\rangle$$

where we have used, in the last line that by Corollary 6.62, the value of the inner product does not depend on the representative in the double coset. Again, by Cauchy–Bunyakowsky–Schwarz and the above calculations, we infer

$$\begin{aligned} |\sigma_m| &\leq m^{k-2} \left| \overline{\Gamma(1)} \setminus \overline{\Gamma(1)} g(m^2) \overline{\Gamma(1)} \right| \frac{\|f\|_{L,k} g(m^2)\|_2 \cdot \|f\|_2}{\|f\|_2^2} \\ &= m^{k-2} \cdot \left| \overline{\Gamma(1)} \setminus \overline{\Gamma(1)} g(m^2) \overline{\Gamma(1)} \right| |_m \mathcal{L}|. \end{aligned}$$

$$(6.34)$$

Considering the case gcd(m, lev(L)) = 1, we have equality, if, and only if,  $f|_{L,k}g(m^2)$  and f are proportional to each other. Hence, assume there is some constant  $C \in \mathbb{C}$  with

$$C \cdot \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \ e(\tau n) \otimes \mathfrak{e}_{\lambda} = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \ e(\tau n m^2) \otimes \mathfrak{e}_{m\lambda}.$$
(6.35)

Here, we see that components  $\lambda \in {}_{m}\mathcal{L}$  could not have contributed. Assume first, that gcd(m, lev(L)) = 1 in which case multiplication by m acts as an automorphism on  $\mathcal{L}$ . So there is a natural number  $l \in \mathbb{N}$ , such that  $m^{l}$  acts as the identity on  $\mathcal{L}$ . Now, unless  $\sigma_{m} = 0$ , we must have  $C \neq 0$ . In the latter case, assume  $a(\lambda, n)$  was a nonzero coefficient and possibly replace the l above by a multiple, large enough, so that  $m^{l} \nmid n$ . Applying the operator  $|_{L,k}g(m^{2})$  in succession for l iterations, we obtain (6.35) but with (C,m) replaced by  $(C^{l}, m^{l})$ . Hence, we find by comparing coefficients that  $a(\lambda, n)$  could only be different from zero if  $m^{l} \mid n$  which contradicts our assumption. Hence, we must have strict inequality in equation (6.34).

In the case of gcd(m, lev(L)) > 1, we find that if the bound was not strict, there could not be any contribution from any p part  $\mathcal{L}_p$  of  $\mathcal{L}$ , for primes  $p \mid gcd(m, lev(L))$ . Explicitly, this means that  $f_{\lambda} = 0$  for any  $\lambda$ , such that the projection to  $\mathcal{L}_p$  for p as above was different from 0. On the remaining components, the argument is the same as above.

Finally, for the choice m = p, there are p(p+1) cosets in the class  $\overline{\Gamma(1)} \setminus \overline{\Gamma(1)} g(m^2) \overline{\Gamma(1)}$ (cf. [Shi73, p. 451]), finishing the proof.

We turn our attention towards the irregular case of bad primes for the action of Hecke operators on Fourier coefficients of vector valued modular forms, in analogy to Proposition 6.54. Such a formula has been given by [Ste15, Thm. 5.4 p. 246] for bad odd primes. However, it has been pointed out by [BCJ18] that the cited Theorem is flawed. The original author has submitted a correction [Ste21], representing the previously arising but flawed coefficients in a more abstract fashion in terms of certain representation numbers of the lattice that also appear in [BK01, p. 447] as coefficients of Eisenstein series.

These representation numbers, however, are not explicit enough for the application we have in mind. In particular, they are unfit to derive product expansions of *L*-series. Hence, the coefficients had to be calculated more explicitly based on the exposition [Ste15]. To this end we have to let  $\mathcal{T}(p^2)$  explicitly act on  $f \in \mathcal{M}_{L,k}$  where the operation is given by letting right cosets of

$$\overline{\Gamma(1)}g(p^2)\overline{\Gamma(1)}$$
, where  $g(p^2) = \left( \begin{pmatrix} p^2 & 0\\ 0 & 1 \end{pmatrix}, 1 \right)$ 

operate as in Definition 6.59. Recall that by [BS08, p. 263] or [Shi73, p. 451] we find

$$\overline{\Gamma(1)}g(p^2)\overline{\Gamma(1)} = \overline{\Gamma(1)}g(p^2) \cup \bigcup_{h(p)^*} \overline{\Gamma(1)}\beta_h \cup \bigcup_{b(p^2)} \overline{\Gamma(1)}\gamma_b,$$
(6.36)

where

$$\beta_h = \left( \left( \begin{array}{cc} p & h \\ 0 & p \end{array} \right), \sqrt{p} \right), \qquad \gamma_b = \left( \left( \begin{array}{cc} 1 & b \\ 0 & p^2 \end{array} \right), p \right). \tag{6.37}$$

The operation on the scalar component functions of the vector valued modular form via the Petersson slash operator is well known and understood. As a consequence, it suffices to compute the action of the elements in (6.37) on the discriminant group. The action of

 $\gamma_b$  is presented in [Ste15, Thm. 5.1 p. 243] in greater generality than required.

**Proposition 6.68.** Let p be an odd prime,  $b \in \mathbb{Z}/p^2\mathbb{Z}$ , and  $\lambda \in \mathcal{L}$ . Then  $\gamma_b$  acts in the following fashion:

$$\mathbf{e}_{\lambda}|_{\mathcal{L}}\gamma_{b} = \sum_{\substack{\nu \in \mathcal{L} \\ p\nu = \lambda}} e(-b\,\overline{\mathbf{q}}(\nu))\mathbf{e}_{\nu}.$$
(6.38)

The action of  $\beta_h$ , however, will be extracted from [BCJ18, Prop. 5.3 p. 26] where we select the special case of l = a = 1.

**Proposition 6.69.** Let p be an odd prime and  $h \in (\mathbb{Z}/p\mathbb{Z})^{\times}$ , write  $m = \operatorname{rank}(L)$ , and select  $\lambda \in \mathcal{L}$ . Then

$$\mathbf{e}_{\lambda}|_{\mathcal{L}}\beta_{h} = p^{-m/2} \sum_{\substack{\delta \in \mathcal{L}(p) \\ p\delta = \lambda}} e(-h \,\overline{\mathbf{q}}_{p}(\delta))\mathbf{e}_{\lambda}, \tag{6.39}$$

where  $\overline{\mathbf{q}}_p$  denotes  $\overline{p \mathbf{q}} : \mathcal{L}(p) = L(p)'/L(p) \to \mathbb{Q}/\mathbb{Z}$ .

This will have to be rewritten for our purposes.

Lemma 6.70. With the notation as above, we find

$$\mathfrak{e}_{\lambda}|_{\mathcal{L}}\beta_{h} = p^{-m/2}\delta_{\lambda\in {}^{p}\mathcal{L}} \cdot e\left(-hp\operatorname{q}(\ell_{\lambda/p})\right)G_{L,p}(1,-h)\mathfrak{e}_{\lambda},$$

where  $G_{L,p}(n,h) = \sum_{v \in L/p^n L} e\left(\frac{h}{p^n} q(v)\right)$  is the Gauss sum defined in Definition A.12,  $\ell_{\lambda/p} \in L'$  is some element, such that  $p \cdot \overline{\ell_{\lambda/p}} = \lambda$ ,  ${}^{p}\mathcal{L}$  are the p-multiples in  $\mathcal{L}$  and  $\delta$  denotes Kronecker symbol.

*Proof*: Beginning with (6.39) we transform the sum by utilising the following description  $\mathcal{L}(p) = p^{-1}L'/L \simeq L'/pL$ :

$$p^{-m/2} \sum_{\substack{\delta \in \mathcal{L}(p) \\ p\delta = \lambda}} e(-h \,\overline{\mathbf{q}}_p(\delta)) \mathfrak{e}_{\lambda} = p^{-m/2} \sum_{\substack{\mu \in L'/pL \\ \mu = \lambda \mod L}} e\left(-\frac{h}{p} \,\mathbf{q}(\mu)\right) \mathfrak{e}_{\lambda},$$

where  $q(\mu)$  is do be understood as the quadratic form on L' of some preimage in L' of  $\mu \in L'/pL$ . In order for the sum above to not vanish, it is necessary that  $\lambda \in {}^{p}\mathcal{L}$  (cf. [Bar03, Lemma 5.2.1]). Note that by the isomorphism  $(L'/pL)/(L/pL) \simeq \mathcal{L}$ , the solutions

in  $\mu$  of  $\mu = \lambda \mod L$  are represented by  $\lambda + L/pL$  so that the expression above equals

$$p^{-m/2}\delta_{\lambda\in p\mathcal{L}}\cdot\sum_{v\in L/pL}e\left(-\frac{h}{p}\operatorname{q}(\ell_{\lambda}+v)\right)\mathfrak{e}_{\lambda},$$

where  $\ell_{\lambda}$  is some lift of  $\lambda$  to L'. In fact, the sum does not depend on the chosen representative, as the shift may be absorbed into v. We would like to separate the dependency on  $\lambda$  from the sum. To this end, let  $\nu \in \mathcal{L}$  be a *p*-th root of  $\lambda$ , i.e.  $p \cdot \nu = \lambda \in \mathcal{L}$ and denote a lift of  $\nu$  to L' by  $\ell_{\nu}$ . Then

$$\delta_{\lambda \in {}^{p}\mathcal{L}} \sum_{v \in L/pL} e\left(-\frac{h}{p} \operatorname{q}(\ell_{\lambda} + v)\right) = \delta_{\lambda \in {}^{p}\mathcal{L}} \cdot e\left(-\frac{h}{p} \operatorname{q}(p\ell_{\nu})\right) \cdot \sum_{v \in L/pL} e\left(-\frac{h}{p} \operatorname{q}(v)\right). \quad (*)$$

In fact, since  $\lambda \in {}^{p}\mathcal{L}$  there is  $\nu \in \mathcal{L}$  such that  $p\nu = \lambda$ . Hence, there must be  $\ell_{\nu} \in L'$  and  $\delta \in L$  such that  $p\ell_{\nu} + \delta = \ell_{\lambda}$ . Nevertheless, the term  $\delta \in L$ , may be absorbed in v and will henceforth be ignored resulting in replacing  $q(\ell_{\lambda} + v)$  by  $q(p\ell_{\nu} + v)$ . Then, however,

$$q(p\ell_{\nu}+v) = q(p\ell_{\nu}) + p\beta(\ell_{\nu},v) + q(v) \equiv q(p\ell_{\nu}) + q(v) \mod p.$$

This, in fact, means that the reduction above is valid.

Note that Stein appears to have made the same reduction, first. Then, however, he noted in [Ste21] that it was not valid because the factored expression in (\*) depended on the choice of representative. If, however, a *p*-th root  $\nu$  of  $\lambda$  is considered, there is no ambiguity as the following computations testifies: Note that two chosen  $\nu$  may differ by  $\nu' \in {}_{p}\mathcal{L}$ . As a consequence,  $\ell_{\nu}$  might be replaced with  $\ell_{\nu} + \ell_{\nu'} + \nu_{0}$  with  $\nu_{0} \in L$ . Then

$$q [p(\ell_{\nu} + \ell_{\nu'} + \nu_0)] = q [p(\ell_{\nu} + \ell_{\nu'})] + p^2 b(\ell_{\nu} + \ell_{\nu'}, \nu_0) + p^2 q(\nu_0)$$
  

$$\equiv q [p(\ell_{\nu} + \ell_{\nu'})] \mod p$$
  

$$= q(p\ell_{\nu}) + p b(\ell_{\nu}, p\ell_{\nu'}) + q(p\ell_{\nu'})$$
  

$$\equiv q(p\ell_{\nu}) \mod p.$$

In order to verify the last line, note that  $p\ell_{\nu'} \in L$ , so that the term involving the bilinear form is, in fact, in  $p\mathbb{Z}$  and we have  $q(p\ell_{\nu'}) \in \mathbb{Z}$ . On the other hand we find  $q(p\ell_{\nu'}) = 2^{-1}p \operatorname{b}(\ell_{\nu'}, p\ell_{\nu'}) \in \frac{p}{2}\mathbb{Z} \cap \mathbb{Z} = p\mathbb{Z}$ .

We follow [Ste15] in order to compute the action of Hecke operators on the Fourier expansion of vector valued modular forms more explicitly. In fact, the following is a modified version of [Ste15, Thm. 5.4 p. 246].

**Proposition 6.71.** Let p be an odd prime and  $f \in \mathcal{M}_{L,k}$  with Fourier expansion

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n\tau) \mathfrak{e}_{\lambda}.$$

Then the Fourier coefficients of  $f|_{L,k}\mathcal{T}(p^2)$  are given by

$$b(\lambda, n) = p^{2(k-1)} \delta_{\lambda \in {}^{p}\mathcal{L}} \sum_{\substack{\lambda' \in {}_{p}\mathcal{L} \\ n-p^{2} \overline{q}(\lambda/p+\lambda') \in p^{2}\mathbb{Z}}} a\left(\lambda/p + \lambda', \frac{n-p^{2} \overline{q}(\lambda/p+\lambda')}{p^{2}} + \overline{q}(\lambda/p+\lambda')\right) + p^{k-2} \delta_{\lambda \in {}^{p}\mathcal{L}} K_{L,p} g_{p} \left[1, -\chi_{p}^{R}, n - q(p \cdot \ell_{\lambda/p})\right] a(\lambda, n) + a(p\lambda, p^{2}n).$$

Here,  $K_{L,p} = p^{-m/2}G_{L,p}(1,1)$ , where  $G_{L,p}(r,1) = \sum_{\nu \in L/p^r L} e\left(\frac{1}{p^r}q(\nu)\right)$  and  $g_p[r,\chi,n] = \sum_{h \in (\mathbb{Z}/p^r \mathbb{Z})^*} \chi(h) e\left(\frac{hn}{p^r}\right)$ , while  $\ell_{\lambda/p} \in L'$  is any element such that  $\overline{p\ell_{p/\lambda}} = \lambda \in \mathcal{L}$ . Further,  $_p\mathcal{L}$  is the p-torsion of  $\mathcal{L}$ ,  $^p\mathcal{L}$  are the p-multiples and  $\delta_{\lambda \in ^p\mathcal{L}} = 1$  if  $\lambda \in ^p\mathcal{L}$  and 0, otherwise. In addition,  $_-\chi_p(h) = \left(\frac{-h}{p}\right)$  is the Legendre symbol and the exponent R has been presented in Definition 1.38 (1.6) as  $R_p$ .

*Proof*: Most of the proof is identical for higher powers of  $p^2$ , but we restrict to the operation of  $\mathcal{T}(p^2)$  in order to lighten the notation. We have

$$f|_{k,L}\mathcal{T}(p^2) = p^{k-2} \sum_{\lambda \in \mathcal{L}} f_\lambda|_k g(p^2) \ \mathfrak{e}_\lambda|_L g(p^2)$$
(6.40)

$$+ p^{k-2} \sum_{h \in (\mathbb{Z}/p\mathbb{Z})^{\times}} \sum_{\lambda \in \mathcal{L}} f_{\lambda}|_{k} \beta_{h} \mathfrak{e}_{\lambda}|_{L} \beta_{h}$$

$$(6.41)$$

$$+ p^{k-2} \sum_{b \in \mathbb{Z}/p^2 \mathbb{Z}} \sum_{\lambda \in \mathcal{L}} f_{\lambda}|_k \gamma_b \, \mathfrak{e}_{\lambda}|_L \gamma_b.$$
(6.42)

For the first part (6.40) we find

$$p^{k-2} \sum_{\lambda \in \mathcal{L}} f_{\lambda}|_{k} g(p^{2}) \mathfrak{e}_{\lambda}|_{\mathcal{L}} g(p^{2})$$

$$= p^{2(k-1)} \sum_{\lambda \in \mathcal{L}} f_{\lambda} (p^{2}\tau) \mathfrak{e}_{p\lambda}$$

$$= p^{2(k-1)} \sum_{\lambda \in \mathcal{L}} \sum_{n \in \mathbb{Z} + \overline{q}(\lambda)} a(\lambda, n) \cdot e(p^{2}n\tau) \mathfrak{e}_{p\lambda}$$

$$=p^{2(k-1)}\sum_{\lambda\in {^{p}\!\mathcal{L}}}\sum_{\lambda'\in {_{p}\!\mathcal{L}}}\sum_{n\in p^{2}(\mathbb{Z}+\overline{\mathbf{q}}(\lambda/p+\lambda'))}a(\lambda/p+\lambda',n/p^{2})\cdot e\left(n\tau\right)\ \mathfrak{e}_{\lambda}.$$

Here,  $\lambda/p$  is any fixed *p*-th root of  $\lambda$  in  $\mathcal{L}$ . Also compare Remark 1.36. The inner sum in the expression above may be rewritten as

$$\sum_{n-p^2\,\overline{\mathbf{q}}(\lambda/p+\lambda')\in p^2\mathbb{Z}}a\left(\lambda/p+\lambda',\frac{n-p^2\,\overline{\mathbf{q}}(\lambda/p+\lambda')}{p^2}+\overline{\mathbf{q}}(\lambda/p+\lambda')\right)\cdot e\left(n\tau\right)\ \mathfrak{e}_{\lambda}$$

in order to separate the integer part of the index from the rest.

Next, we evaluate the sum (6.41) involving the action of the elements  $\beta_h$ :

$$p^{k-2}\sum_{h\in(\mathbb{Z}/p\mathbb{Z})^{ imes}}\sum_{\lambda\in\mathcal{L}}f_{\lambda}|_{k}\beta_{h} \ \mathfrak{e}_{\lambda}|_{L}\beta_{h}.$$

First, we utilise Lemma 6.70 to obtain

$$p^{k-2} \sum_{h \in (\mathbb{Z}/p\mathbb{Z})^{\times}} \sum_{\lambda \in \mathcal{L}} f_{\lambda}|_{k} \beta_{h} \mathfrak{e}_{\lambda}|_{L} \beta_{h}$$
  
=  $p^{2(k-1)} \sum_{h \in (\mathbb{Z}/p\mathbb{Z})^{\times}} \sum_{\lambda \in p\mathcal{L}} p^{-k} f_{\lambda} \left(\frac{p\tau + h}{p}\right) \cdot p^{-\frac{m}{2}} e\left(-hp \operatorname{q}(\ell_{\lambda/p})\right) G_{L,p}(1, -h) \mathfrak{e}_{\lambda}.$  (\*)

Here,  $\ell_{\mu}$  for some  $\mu \in \mathcal{L}$  is understood to be an arbitrary but fixed lift of  $\mu$  to L' and  $\lambda/p$  is any fixed *p*-th root of  $\lambda$  in  $\mathcal{L}$ . In (\*), the dependency of the Gauss sum  $G_{L,p}(1, -h)$  on h may be extracted. In fact, an application of Remark A.14 yields

$$G_{L,p}(1,-h) = \left(\frac{-h}{p}\right)^{R} \cdot G_{L,p}(1,1)$$
(6.43)

with the following choice of  $R_p$ . The  $(\mathbb{Z}/p\mathbb{Z})$  module L/pL with quadratic form to  $\mathbb{Z}/p\mathbb{Z}$ decomposes into one dimensional submodules (cf. Remark A.14). Let  $R_p$  be the number of these one dimensional modules, such that their generator is non-isotropic, i.e. if v is a generator, then  $q(v) \notin p\mathbb{Z}$ .

With these computations, and by writing  $-\chi_p(h)$  for the Legendre symbol appearing in (6.43), we find the following expression for (\*):

$$p^{k-2-m/2}G_{L,p}(1,1)\sum_{h\in(\mathbb{Z}/p\mathbb{Z})^{\times}}\sum_{\lambda\in^{p}\mathcal{L}}f_{\lambda}\left(\frac{p\tau+h}{p}\right)\cdot e\left(-hp\operatorname{q}(\ell_{\lambda/p})\right)_{-}\chi_{p}(h)^{R}\mathfrak{e}_{\lambda}$$

$$= p^{k-2-m/2} G_{L,p}(1,1) \sum_{h \in (\mathbb{Z}/p\mathbb{Z})^{\times}} \sum_{\lambda \in p\mathcal{L}} \sum_{n \in \overline{q}(\lambda) + \mathbb{Z}} a(\lambda,n) e(n\tau) e\left(\frac{h}{p} [n - q(p \cdot \ell_{\lambda/p})]\right) - \chi_p(h)^R \mathfrak{e}_{\lambda}.$$

Set  $K_{L,p} := p^{-m/2} G_{L,p}(1,1)$  and compare Definition A.12 to recognise the appearing term above as the following Gauss sum:

$$g_p\left[1, -\chi_p^R, n - q(p \cdot \ell_{\lambda/p})\right] = \sum_{h \in (\mathbb{Z}/p\mathbb{Z})^{\times}} e\left(\frac{h}{p}[n - q(p \cdot \ell_{\lambda/p})]\right) \chi_p(-h)^R$$

which yields for the term in (\*) to equal

$$p^{k-2}K_{L,p}\sum_{\lambda\in p\mathcal{L}}\sum_{n\in\overline{\mathbf{q}}(\lambda)+\mathbb{Z}}a(\lambda,n)g_p\left[1,-\chi_p^R,n-\mathbf{q}(p\cdot\ell_{\lambda/p})\right]e\left(n\tau\right)\mathfrak{e}_{\lambda}$$

Last, the sum in (6.42) involving the action of  $\gamma_b$  is evaluated, using Proposition 6.68.

$$p^{k-2} \sum_{b \in \mathbb{Z}/p^2 \mathbb{Z}} \sum_{\lambda \in \mathcal{L}} f_{\lambda}|_{k} \gamma_{b} \mathfrak{e}_{\lambda}|_{L} \gamma_{b}$$

$$= p^{-2} \sum_{b \in \mathbb{Z}/p^{2} \mathbb{Z}} \sum_{\lambda \in \mathcal{L}} f_{\lambda} \left(\frac{\tau + b}{p^{2}}\right) \cdot \sum_{\substack{\nu \in \mathcal{L} \\ p\nu = \lambda}} e\left(-b \overline{q}(\nu)\right) \mathfrak{e}_{\nu}$$

$$= p^{-2} \sum_{\lambda \in \mathcal{L}} \sum_{\substack{\nu \in \mathcal{L} \\ p\nu = \lambda}} \sum_{n \in \overline{q}(\lambda) + \mathbb{Z}} a(\lambda, n) \sum_{b \in \mathbb{Z}/p^{2} \mathbb{Z}} e\left(b\frac{n - q(p\ell_{\nu})}{p^{2}}\right) e\left(\frac{n\tau}{p^{2}}\right) \mathfrak{e}_{\nu}. \quad (*')$$

Note that the inner sum is  $p^2$ , if  $n - q(p\ell_{\nu}) \in p^2\mathbb{Z}$ , otherwise the geometric formula yields

$$\sum_{b \in \mathbb{Z}/p^2 \mathbb{Z}} e\left(\frac{n - q(p\ell_{\nu})}{p^2}\right)^b = \frac{1 - e\left(n - q(p\ell_{\nu})\right)}{1 - e\left([n - q(\ell_{\nu})]/p^2\right)} = 0$$

Hence,

$$\sum_{b \in \mathbb{Z}/p^2 \mathbb{Z}} e\left(\frac{n - q(p\ell_{\nu})}{p^2}\right)^b = p^2 \delta_{n \equiv q(p\ell_{\nu}) \mod p^2 \mathbb{Z}}.$$

Further,  $p\nu = \lambda$  may only be fulfilled if  $\lambda \in {}^{p}\mathcal{L}$ . The solutions to the equation are exactly the elements  $\lambda/p + \delta$  where  $\delta \in {}_{p}\mathcal{L}$  and  $\lambda/p$  is a fixed inverse image. However, by Remark 1.36, these are all elements in  $\mathcal{L}$ . In total, we obtain the following expression for

(\*'):

$$\sum_{\lambda \in {}^{p}\mathcal{L}} \sum_{\substack{\nu \in \mathcal{L} \\ p\nu = \lambda}} \sum_{\substack{n \in \mathbb{Z} + \overline{q}(\lambda) \\ n - q(p\ell_{\nu}) \in p^{2}\mathbb{Z}}} a(\lambda, n) e\left(\frac{n\tau}{p^{2}}\right) \mathfrak{e}_{\nu}$$

$$= \sum_{\mu \in \mathcal{L}} \sum_{n - p^{2} q(\ell_{\mu}) \in p^{2}\mathbb{Z}} a(p\mu, n) e\left(\frac{n\tau}{p^{2}}\right) \mathfrak{e}_{\mu}$$

$$= \sum_{\mu \in \mathcal{L}} \sum_{m \in \mathbb{Z} + \overline{q}(\mu)} a(p\mu, p^{2}m) e(m\tau) \mathfrak{e}_{\mu}.$$

As a consequence of the above, we obtain the case of good primes (different from 2), meaning essentially Proposition 6.54.

**Corollary 6.72** (Bruinier, Stein). Let L have even signature, p be an odd prime not dividing lev(L) and  $f \in \mathcal{M}_{L,k}$  with Fourier coefficients  $a(\lambda, n)$ . Denote the Fourier coefficients of  $f|_{L,k}\mathcal{T}(p^2)$  by  $b(\lambda, n)$ . Then we find

$$b(\lambda, n) = p^{2k-2}a(\lambda/p, n/p^2) + \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}p^{k-2}(\delta_{p|n}p - 1)a(\lambda, n) + a(p\lambda, p^2n).$$
(6.44)

*Proof*: In the present case, the formulae of Proposition 6.71 read

$$b(\lambda, n) = p^{2(k-1)} a\left(\lambda/p, \frac{n}{p^2}\right)$$
  
+  $p^{k-2-m/2} G_{L,p}(1, 1) g_p \left[1, -\chi_p^R, n - q(p \cdot \lambda/p)\right] a(\lambda, n)$   
+  $a(p\lambda, p^2 n).$ 

As a consequence, only the middle term has to be evaluated. So the sums  $g_p$  and  $G_{L,p}$  are to be understood. Recall that Equation (A.5) in Remark A.18 yields the desired quotient of Gauss sums. Hence, only proving that the second Gauss sum  $g_p$  reduces to  $(\delta_{p|n}p - 1)$ is required. Note that R = m is even in our case, so that  $\chi_p^R = \chi_1$  is the trivial character. As a consequence, Remark A.15 implies

$$g_p[1, \chi_1, n - \mathbf{q}(p \cdot \ell_{\lambda/p})] = \delta_{p|n - \mathbf{q}(p \cdot \ell_{\lambda/p})}p - 1.$$

However, the denominator of the individual fractions does not contain p, since  $p \nmid \text{lev}(L)$ . Hence, investigating the numerator for divisibility by p suffices. However,  $q(p \cdot \ell_{\lambda/p})$  is clearly divisible by p, so that the question reduces to whether  $p \mid n$  or not.

# 6.4 L-functions of modular forms

*L*-functions and modular forms are closely related analytical objects in the field of number theory. In fact, there is a classical theory of the association of *L*-functions to scalar valued modular forms, as well as a converse theorem<sup>10</sup>. Furthermore, the interconnection between these classes of objects has also been exploited to deliver the famous proof of Fermat's last theorem. We refer to [DS05] and [Sil13] for more on that matter and to [Bum98] for an accessible overview of the classical theory of *L*-functions and their relation to modular forms. In the following, our focus lies on associating symmetric square type *L*-series that have appeared in [Shi73] to vector valued modular forms for the discrete Weil representation which play a key role in the present thesis.

### 6.4.1 L-series of vector valued modular forms

In this subsection, convergence statements depend on a parameter  $\sigma \in \mathbb{R}$ . Its value is given by the improvement of the asymptotic growth of Fourier coefficients compared to the Hecke bound of cusp forms and a possible, not necessarily optimal, choice is found in Table 3.1 or Corollary 6.82.

In Section 6.1 we have briefly reviewed *L*-functions associated to classical modular forms and stated their main properties in Theorem 6.45. In the following we will define similar *L*-functions associated to vector valued modular forms  $f \in \mathcal{M}_{L,k}$  for some lattice *L*. For this subsection, assume that the Fourier expansion of the modular form f is written as

$$f(\tau) = \sum_{\mu \in L'/L} \sum_{n \in \overline{\mathbf{q}}(\mu) + \mathbb{Z}} a(\mu, n) \cdot e(n\tau) \mathfrak{e}_{\mu}.$$
(6.45)

Given the structure of the operation of the Hecke algebra on Fourier coefficients, a definition resembling symmetric square L-functions comparable to these in [Shi73] is motivated. Recall that (L, q) was an even lattice of signature  $(m^+, m^-)$  and rank m. For simplicity, we will initially assume a splitting  $L = L_1 \oplus L_2$  with  $L_1 \otimes \mathbb{R}$  definite of rank  $m_1$ .

**Definition 6.73.** For  $f \in \mathcal{M}_{L,k}$ ,  $L = L_1 \oplus L_2$  an orthogonal splitting with definite  $L_1 \otimes_{\mathbb{Z}} \mathbb{R}$ , an isotropic element  $\eta \in L'_2/L_2$ , and  $r \in \mathbb{Q}^{\times}$  we define the formal series

$$L_{L_1,\eta,r}(f,s) \coloneqq \sum_{0 \neq l \in L_1'} \frac{a(\bar{l} + \eta, r \operatorname{q}(l))}{\operatorname{q}(l)^s}.$$
(6.46)

<sup>&</sup>lt;sup>10</sup>Compare [Bum98, Thm 1.5.1 p 60], for instance, for a general Theorem due to André Weil.

The origin of this definition lies in the construction of such L-series via the Rankin–Selbgerg method as in Proposition 6.97.

**Remark 6.74.** Note that in case  $\eta = 0$ , the series in (6.46) has a symmetry property inherited from the symmetry of the Fourier coefficients of the modular form f. Recall Remark 3.22, in particular (3.42), reading

$$a(-\lambda, n) = e[(\operatorname{sig}(\mathcal{L}) - 2k)/4]a(\lambda, n)$$

which is satisfied for all  $\lambda \in \mathcal{L} = L'/L$  and  $n \in \overline{q}(\lambda) + \mathbb{Z}$ . In the light of Remark 3.20, we may assume  $\operatorname{sig}(\mathcal{L}) - 2k \equiv 0, 2 \mod 4$ , in any case, so that  $e[(\operatorname{sig}(\mathcal{L}) - 2k)/4] = \pm 1$ .

It is natural to ask for the region of convergence of these *L*-series for complex arguments  $s \in \mathbb{C}$  which may be derived from asymptotic bounds on the Fourier coefficients appearing in their construction. The reader may compare Corollary 3.84 and Corollary 3.85 on that matter. We require the following technical result, before continuing.

**Lemma 6.75.** Let  $m \in \mathbb{N}$  be a natural number and  $\alpha \in \mathbb{R}$ . Then

$$\sum_{0 \neq z \in \mathbb{Z}^m} \|z\|_\infty^{-\alpha}$$

converges if  $\alpha > m$ , where  $\|\cdot\|_{\infty}$  denotes the  $\infty$ -norm.

*Proof*: For  $n \in \mathbb{N}$ , we will compute the size of the discrete sphere  $\mathbb{S}^{m-1}_{\infty}(n)$  of radius n in  $\mathbb{Z}^m$  with respect to  $\|\cdot\|_{\infty}$ . Using the abbreviation  $[n] \coloneqq \{0, \ldots, n\}$  for  $n \in \mathbb{N}$ , we find

$$S_{\infty}^{m-1}(n) = \| \cdot \|_{\infty}^{-1}(n)$$
  
= {(z\_i) \in \mathbb{Z}^m | max |z\_i| = n}  
= \| \cdot \|\_{\infty}^{-1}([n]) \ \| \cdot \|\_{\infty}^{-1}([n-1]).

Clearly,  $\# \| \cdot \|_{\infty}^{-1}([n]) = (2n+1)^m$ , so that we obtain

$$|\mathbb{S}_{\infty}^{m-1}(n)| = (2n+1)^m - (2n-1)^m.$$

Using the binomial formula yields

$$(2n \pm 1)^m = \sum_{k=0}^m \binom{m}{k} (\pm 1)^k (2n)^{m-k},$$

resulting in

$$|\mathbb{S}_{\infty}^{m-1}(n)| = (2n+1)^m - (2n-1)^m$$
$$= 2\sum_{2 \nmid k=0}^m \binom{m}{k} (2n)^{m-k}$$
$$\leq K \cdot n^{m-1}$$

for some K > 0. With that bound, we obtain

$$\sum_{0 \neq z \in \mathbb{Z}^m} \|z\|_{\infty}^{-\alpha} \le K \sum_{n \in \mathbb{N}} \frac{1}{n^{1-m+\alpha}}$$

However, the right hand series equals  $\zeta(1-m+\alpha)$  which converges exactly if  $1-m+\alpha > 1$  or, equivalently,  $\alpha > m$ .

With this tool, we may prove convergence of the investigated L-series.

**Lemma 6.76.** Assume  $L = L_1 \oplus L_2$  splits with definite  $L_1 \otimes_{\mathbb{Z}} \mathbb{R}$  of rank  $m_1$  and  $2 \leq k \in \mathbb{Z}/2$ . In case  $f \in \mathcal{M}_{L,k}$ , the series  $L_{L_1,\eta,r}(f,s)$  converges normally for  $\operatorname{Re}(s) > k - 1 + \frac{m_1}{2}$ . If, in addition,  $f \in \mathcal{S}_{L,k}$ , we find that  $L_{L_1,\eta,r}(f,s)$  converges normally for  $\operatorname{Re}(s) > \frac{k+m_1}{2} - \sigma$  for  $\sigma = 1/2$  in the even case and  $\sigma = 1/4$  in the odd case.

*Proof*: Recall that  $\sqrt{|\mathbf{q}|} : L_1 \otimes_{\mathbb{Z}} \mathbb{R} \to \mathbb{R}_{\geq 0}$  defines a norm. Hence, it is equivalent to  $\|\cdot\|_{\infty}$ , meaning, in particular, there is a constant 0 < c such that for all  $v \in L_1 \otimes_{\mathbb{Z}} \mathbb{R}$  we have

$$c \cdot \|v\|_{\infty} \le \sqrt{|\mathbf{q}(v)|}.$$

Further recall that we have for any  $\lambda \in \mathcal{L}$  a bound

$$|a(\lambda, rn^2)| \le C \cdot n^{k-2\sigma}$$

for some  $\sigma \in \mathbb{R}$  and some positive constant C > 0. For instance, we may always pick  $\sigma = 1 - k/2$  (cf. Corollary 3.76) and select  $\sigma = \frac{1}{5}$  for  $f \in S_{L,k}$  by Corollary 3.84. The best choices available in this thesis are found in Corollary 3.87. In particular, we find

$$\sum_{0 \neq l \in L_1'} \frac{|a(\bar{l} + \eta, r \mathbf{q}(l))|}{|\mathbf{q}(l)|^{\operatorname{Re}(s)}} \leq \frac{C}{c^2} \sum_{0 \neq l \in L_1'} \frac{\|l\|_{\infty}^{k-2\sigma}}{\|l\|_{\infty}^{2(\operatorname{Re}(s))}}$$
$$= \frac{C}{c^2} \sum_{0 \neq l \in \mathbb{Z}^{m_1}} \|l\|_{\infty}^{k-2(\sigma + \operatorname{Re}(s))}.$$

However, by Lemma 6.75 the latter series converges as long as

$$2(\sigma + \operatorname{Re}(s)) - k > m_1$$

holds, which is equivalent to  $\operatorname{Re}(s) > \frac{k+m_1}{2} - \sigma$ . Clearly, the bound is decreasing in  $\operatorname{Re}(s)$ , so that normal convergence on a right half plane follows immediately.

Clearly, it is dissatisfying to rely on a splitting of the lattice L in order to associate the symmetric square L-function from Definition 6.73 to a modular form. In the following, we will erase that assumption. Let  $L_1$  be a  $\mathbb{Z}$ -submodule of L such that  $L_1 \otimes \mathbb{R}$  is definite. By choosing  $L_2 := (L_1^{\perp} \cap L)$  we construct an even non-degenerate lattice  $M := L_1 \oplus L_2 \leq L$  of finite index. Then by Proposition 3.36 there is a lifting operator mapping  $f \in \mathcal{M}_{L,k}(\mathrm{Mp}_2(\mathbb{Z}))$  to  $\uparrow_L^M f \in \mathcal{M}_{M,k}$  and for  $\eta \in L'_2/L_2$  isotropic and  $r \in \mathbb{Q}^{\times}$  we may declare the L-series  $L_{L_1,\eta,k}(\uparrow_L^M f, s)$  as above. We abbreviate  $\tilde{f} := \uparrow_L^M f$  and denote its Fourier coefficients by  $\tilde{a}(\lambda, n)$ . Then

$$L_{L_{1},\eta,r}(\tilde{f},s) = \sum_{0 \neq l \in L'_{1}} \frac{\tilde{a}(\bar{l}+\eta, r \operatorname{q}(l))}{\operatorname{q}(l)^{s}}.$$

Note that if  $\eta$  is not induced by an element of L', then the coefficients  $\tilde{a}(\bar{l}+\eta, r q(l))$  vanish. Hence, we may assume  $\eta$  to be induced by an element of L'. Further,  $\tilde{a}(\bar{l}+\eta, q(l))$  vanishes unless  $l \in L'$ . But if also  $l \in L'$ , necessarily  $\tilde{a}(\bar{l}+\eta, q(l)) = a(\bar{l}+\eta, q(l))$ . Consequently, if we define  $a(\bar{l}+\eta, q(l))$  to vanish, unless  $l \in L'$ , we conclude

$$L_{L_1,\eta,r}(\tilde{f},s) = \sum_{0 \neq l \in L'_1} \frac{a(\bar{l} + \eta, r \operatorname{q}(l))}{\operatorname{q}(l)^s}.$$

This prompts us to generalise Definition 6.73.

**Definition 6.77.** For  $f \in \mathcal{M}_{L,k}$ , a splitting sublattice  $M = L_1 \oplus L_2 \leq L$  with definite  $L_1 \otimes \mathbb{R}$ , an isotropic element  $\eta \in (L' \cap L'_2)/L_2$ , and  $r \in \mathbb{Q}^{\times}$  define the formal series

$$L_{L_1,\eta,r}(f,s) \coloneqq \sum_{0 \neq l \in L'_1 \cap L'} \frac{a(l+\eta, r q(l))}{q(l)^s}.$$
(6.47)

**Remark 6.78.** In the setting of Definition 6.77 we have

$$L_{L_1,\eta,r}(f,s) = L_{L_1,\eta,r}(\uparrow_L^M f,s),$$

where the right hand side may be interpreted as in Definition 6.73. As a consequence, the definition above is a proper extension to the case without splitting lattice  $L_1$  and, in particular, Lemma 6.76 is applicable to obtain the same range of convergence.

In addition to these general series, we are also interested in the following specialisations which will appear as cycle integrals of Kudla–Millson liftings in Subsection 7.3.1.

**Definition 6.79.** Let (L, q) be an even lattice and

$$f = \sum_{\mu \in \mathcal{L}} \sum_{n \in \mathbb{Z} + \overline{\mathbf{q}}(\mu)} a(\mu, n) \cdot e(n\tau) \mathfrak{e}_{\mu} \in \mathcal{M}_{L,k}$$

be a modular form. For  $\lambda \in \mathcal{L}$  and  $r \in \mathbb{Q}^{\times}$  we define the associated symmetric square L-function

$$L_{(\lambda,r)}(f,s) \coloneqq \sum_{n \in \mathbb{N}} \frac{a(n\lambda, n^2 r)}{n^s}$$
(6.48)

and its specialisation induced by  $\ell \in L'$  satisfying  $q(\ell) \neq 0$  with  $\lambda := \overline{\ell}$  and  $r := q(\ell)$ :

$$L_{\ell}(f,s) \coloneqq L_{(\bar{\ell},\mathbf{q}(\ell))}(f,s) = \sum_{n \in \mathbb{N}} \frac{a(n\bar{\ell},n^2 \mathbf{q}(\ell))}{n^s}.$$
(6.49)

The range of convergence of these special series is derived from Lemma 6.76.

**Lemma 6.80.** For  $f \in \mathcal{M}_{L,k}$ ,  $k \geq 2$ ,  $\lambda \in \mathcal{L} = L'/L$ , and  $r \in \mathbb{Q}^{\times}$  the series  $L_{(\lambda,r)}(f,s)$  converges absolutely for  $\operatorname{Re}(s) > 2k - 1$ . If, in addition,  $f \in \mathcal{S}_{L,k}$ , we find that  $L_{(\lambda,r)}(f,s)$  converges absolutely for  $\operatorname{Re}(s) > k + 1 - 2\sigma$  for  $\sigma = 1/2$  in the even case and  $\sigma = 1/4$  in the odd case.

Proof: Begin by choosing  $\ell \in L'$  such that  $\overline{\ell} = \lambda$  and recall that we have to assume  $r = q(\ell) \neq 0$ . Note that  $(\mathbb{Q}\ell \cap L, q \mid_{\mathbb{Z}\ell})$  defines an even lattice, implying  $\ell \in (\mathbb{Q}\ell \cap L, \ell)'$ . Denote by  $\ell_0$  a primitive element of the dual  $(\mathbb{Q}\ell \cap L)'$ . Then

$$L_{\mathbb{Z}\ell,0,r} = \sum_{0 \neq l \in (\mathbb{Z}\ell)'} \frac{a(\bar{l}, r \operatorname{q}(l))}{\operatorname{q}(l)^s}$$

$$= \sum_{0 \neq n \in \mathbb{Z}} \frac{a(n\ell_0, r \operatorname{q}(\ell_0)n^2)}{\operatorname{q}(\ell_0)^s n^{2s}}.$$

In particular, if  $t \in \mathbb{Z}$  denotes the unique element with  $t\ell_0 = \ell$  we have

$$q(\ell)^{s/2} \cdot L_{\mathbb{Z}\ell,0,r/q(\ell)}(s/2) = \sum_{0 \neq n \in \mathbb{Z}} \frac{a(n\overline{\ell_0}, r(n/t)^2)}{(n/t)^s}.$$

Recall that by Lemma 6.76 this expression converges absolutely for  $\operatorname{Re}(s)/2 > \frac{k+1}{2} - \sigma$ for  $\sigma = 1 - k/2$  in general and some  $1/4 \leq \sigma < 1/2$  in case f was a cusp form (cf. Corollary 3.87). However, by considering the subseries of indices  $n \in t\mathbb{N}$  we obtain exactly  $L_{(\lambda,r)}(f,s)$ . Hence, the latter series converges absolutely for  $\operatorname{Re}(s) > k + 1 - 2\sigma$ .  $\Box$ 

For the proof of Theorem 7.16 we will require subseries of the above L-functions that are of a special shape. A naive motivation for defining these is to isolate coefficients associated to good primes for which there is a product expansion due to a well behaved Hecke theory (cf. Corollary 6.87).

**Definition 6.81.** In the setting of Definition 6.79, meaning  $f \in \mathcal{M}_{L,k}$  with Fourier coefficients  $a(\mu, n)$  and  $\lambda \in \mathcal{L} = L'/L$  as well as  $r \in \mathbb{Q}^{\times}$ , consider for some  $N \in \mathbb{N}$  the subseries

$$L^{N}_{(\lambda,r)}(f,s) \coloneqq \sum_{\substack{n \in \mathbb{N} \\ \gcd(n,N)=1}} \frac{a(n\lambda, n^{2}r)}{n^{s}}$$
(6.50)

and its specialisation induced by  $\ell \in L'$  satisfying  $q(\ell) \neq 0$  with  $\lambda := \overline{\ell}$  and  $r := q(\ell)$ :

$$L^{N}_{\ell}(f,s) \coloneqq L^{N}_{(\bar{\ell},\mathbf{q}(\ell))}(f,s) = \sum_{\substack{n \in \mathbb{N}\\ \gcd(n,N)=1}} \frac{a(n\ell, n^{2} \mathbf{q}(\ell))}{n^{s}}.$$
(6.51)

As subseries of  $L_{(\lambda,r)}(f,s)$ , these clearly converge absolutely, if the original series does, meaning Lemma 6.80 is applicable as well. Further, the concrete asymptotic bounds for Fourier coefficients of vector valued cusp forms in Section 3.3.4 that are subsumed in Corollary 3.87 provide information on how far absolute convergence may be pushed.

**Corollary 6.82.** Let  $2 \leq k \in \mathbb{Z}/2$ ,  $\ell \in L'$  be non-isotropic and  $f \in S_{L,k}$  with Fourier expansion

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n\tau) \mathfrak{e}_{\lambda}.$$

Then for  $N \in \mathbb{N}$  the series

$$L_{\ell}^{N}(f,s) = \sum_{\substack{n \in \mathbb{N} \\ \gcd(n,N)=1}} \frac{a(n\overline{\ell}, n^{2} \operatorname{q}(\ell))}{n^{s}}$$

converges absolutely for  $\operatorname{Re}(s) > k + 1 - 2\sigma$  for some  $1/4 \le \sigma \le 1/2$ . The possible choices of  $\sigma$  are given as follows.

σ	$N\in \mathbb{N}$	$\operatorname{lev}(L) \mid N^\infty$
$2 \mid \operatorname{rk}(L)$	1/2	1/2
$2 \nmid \operatorname{rk}(L)$	1/4	5/16

We turn to the application of Hecke theory to the L-functions discussed above.

### 6.4.2 Product expansions

In case  $f \in \mathcal{S}_{L,k}$  is an eigenform of a family of Hecke operators  $\mathcal{T}(m^2)$  as in Definition 6.59, the associated *L*-series considered in Definition 6.79 admits a partial product expansion. These products are induced by the action of Hecke operators on the Fourier expansion of fpresented in Proposition 6.54 and Theorem 6.55. Recall that for  $N \in \mathbb{N}$  the expression  $\mathscr{H}_N^2$ denotes the Hecke algebra generated by  $\mathcal{T}(p^2)$  for primes  $p \nmid N$ . By Corollary 6.66 these operators always admit simultaneous eigenforms unless  $\mathcal{S}_{L,k}$  is trivial.

In special cases, even full product expansions of the associated *L*-series may be derived, though requiring restrictive assumptions on the lattice *L* and the form *f* as presented in Corollary 6.92. We begin by isolating a single factor associated to a prime number based on [Shi73, Cor. 1.8 p. 451].

**Proposition 6.83.** Let p be a prime with  $p \nmid \text{lev}(L)$ ,  $N \in \mathbb{N}$  fulfil  $p \nmid N$  and assume  $f \in S_{L,k}$  is an eigenform of  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$ . Then for  $\lambda \in \mathcal{L}$  and  $0 < t \in \mathbb{Z} + \overline{q}(\lambda)$  such that  $p^2 \nmid \text{lev}(L)t$  we find that

$$\begin{split} L^{N}_{(\lambda,t)}(f,s) &= L^{Np}_{(\lambda,t)}(f,s) \\ &\cdot \begin{cases} \frac{1 + \delta_{p \nmid t} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} \cdot p^{k-1-s}}{1 - \left(\frac{\sigma_{p}}{p^{k-1}} - (1-p^{-1}) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right) \cdot p^{k-1-s} + p^{2(k-1-s)}} &, 2 \mid \mathrm{rk}(L), \\ \frac{1 - \chi_{\mathcal{L}}(p) \left(\frac{-t}{p}\right) p^{-1/2} \cdot p^{k-1-s}}{1 - \frac{\sigma_{p}}{p^{k-1}} \cdot p^{k-1-s} + p^{2(k-1-s)}} &, 2 \nmid \mathrm{rk}(L). \end{split}$$

Here,  $\chi_{\mathcal{L}}(p) = \epsilon_p^{\operatorname{sig}(\mathcal{L}) + \left(\frac{-1}{|\mathcal{L}|}\right)} \left(\frac{p}{|\mathcal{L}|^{2\operatorname{sig}(\mathcal{L})}}\right)$  and  $\epsilon_p = 1$  or i depending on wether  $p \equiv 1$  or -1 mod 4. Given  $L^N_{(\lambda,t)}(f, \cdot)$  is nonzero, the rational function in  $p^{-s}$  is well defined and does not vanish in the right half plane of absolute convergence of  $L^N_{(\lambda,t)}(f, \cdot)$  determined by Corollary 6.82.

Before proving the result, recall that for some integer d we set  $G_{\mathcal{L}}(d) = \sum_{\lambda \in \mathcal{L}} e(d \overline{q}(\lambda))$ in Definition A.16, preceding alternative descriptions of that Gauss sum.

*Proof*: Recall that the Fourier coefficients of  $f \in S_{L,k}$  are denoted  $a(\lambda, n)$  as in (6.20). For  $n \in \mathbb{N}$ , following [Shi73, p. 452], define the formal power series

$$H_n(x) \coloneqq \sum_{m=0}^{\infty} a(p^m n\lambda, (p^m n)^2 t) x^m.$$

Our goal is to prove that there is a rational function Q(x) such that

$$H_n(x) = a(n\lambda, n^2 t) \cdot Q(x). \tag{(*)}$$

Then,

$$L_{(\lambda,t)}^{N}(f,s) = \sum_{\substack{n=1\\ \gcd(n,N)}}^{\infty} a(n\lambda, n^{2}t)n^{-s}$$
$$= \sum_{\substack{n=1\\ \gcd(n,Np)}}^{\infty} H_{n}(p^{-s})n^{-s} = L_{(\lambda,t)}^{Np}(f,s) \cdot Q(p^{-s}).$$

We will compute Q(x) in the case of even and odd rank separately.

The case of even rank: In this case, Proposition 6.54 yields

$$\sigma_p \cdot a(\lambda, n) = a(p\lambda, p^2n) + \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} p^{k-2} \left( p\delta_{p|n} - 1 \right) a(\lambda, tn) + p^{2(k-1)}a(\lambda/p, n/p^2)$$

resulting for  $n, m \in \mathbb{N}$  with  $p \nmid n$  and  $p^2 \nmid \text{lev}(L)t$  in

$$\sigma_p \cdot a(n\lambda, n^2 t) = a(pn\lambda, p^2 n^2 t) + \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} p^{k-2} \left( p\delta_{p|t} - 1 \right) a(n\lambda, n^2 t), \qquad (\mathbf{I})$$
  
$$\sigma_p \cdot a(p^m n\lambda, p^{2m} n^2 t) = a(p^{m+1} n\lambda, p^{2(m+1)} n^2 t) + (p-1) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} p^{k-2} a(p^m n\lambda, p^{2m} n^2 t)$$

$$+ p^{2(k-1)}a(p^{m-1}n\lambda, p^{2(m-1)}n^{2}t).$$
(II)

Multiplying these equations by the formal variable  $x^{m+1}$  and summing them results in

$$\sigma_p x \cdot H_n(x) = H_n(x) - a(n\lambda, n^2 t) + (p-1) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} p^{k-2} x H_n(x) - \delta_{p \nmid t} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} p^{k-1} a(n\lambda, n^2 t) x + p^{2(k-1)} x^2 H_n(x).$$

A straightforward rearrangement yields

$$H_n(x) = a(n\lambda, tn^2) \cdot \left(1 + \delta_{p \nmid t} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} p^{k-1} x\right)$$
$$\cdot \left[1 - \sigma_p x + (p-1) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} p^{k-2} x + p^{2(k-1)} x^2\right]^{-1}$$

The case of odd rank: In this case, Theorem 6.55 yields

$$\sigma_p \cdot a(\lambda, n) = a(p\lambda, p^2 n) + \frac{\chi_{\mathcal{L}}(p)}{\sqrt{p}} \left(\frac{-n}{p}\right) p^{k-1} a(\lambda, n) + p^{2(k-1)} a(\lambda/p, n/p^2)$$

where we have abbreviated  $\chi_{\mathcal{L}}(p) \coloneqq \varepsilon_p^{\operatorname{sig}(A) + \binom{-1}{|A|}} \left(\frac{p}{|A|^{2^{\operatorname{sig}(A)}}}\right)$ . This results for  $p \nmid n, m \in \mathbb{N}$  and  $p^2 \nmid \operatorname{lev}(L)t$  in

$$\sigma_p \cdot a(n\lambda, n^2 t) = a(pn\lambda, p^2 n^2 t) + \frac{\chi_{\mathcal{L}}(p)}{\sqrt{p}} \left(\frac{-t}{p}\right) p^{k-1} a(n\lambda, n^2 t), \tag{I}$$

$$\sigma_p \cdot a(p^m n\lambda, p^{2m} n^2 t) = a(p^{m+1} n\lambda, p^{2(m+1)} n^2 t) + p^{2(k-1)} a(p^{m-1} n\lambda, p^{2(m-1)} n^2 t).$$
(II)

Analogously to the situation of even rank we find

$$\sigma_p x \cdot H_n(x) = H_n(x) - a(n\lambda, n^2 t) + \frac{\chi_{\mathcal{L}}(p)}{\sqrt{p}} \left(\frac{-t}{p}\right) p^{k-1} a(n\lambda, n^2 t) x$$
$$+ p^{2(k-1)} x^2 H_n(x)$$

resulting in

$$H_n(x) = a(n\lambda, n^2 t) \left(1 - \frac{\chi_{\mathcal{L}}(p)}{\sqrt{p}} \left(\frac{-t}{p}\right) p^{k-1} x\right) \cdot \left[1 - \sigma_p x + p^{2(k-1)} x^2\right]^{-1}.$$

This settles the shape of the rational factor.

Next, we will investigate the behaviour of the numerator and denominator of the rational

factor. Write

$$Q(x) = \frac{R(x,t)}{P(x)}$$

for the rational function in x. Now assume that  $L^N_{(\lambda,t)}(f,s) \neq 0$ , for otherwise there is no behaviour to consider. Then there is some  $n \in \mathbb{N}$  with gcd(n, N) = 1, such that  $a(n\lambda, n^2t) \neq 0$ . We fix this n. Recall that there exists a whole right half plane such that if  $s \in \mathbb{C}$  is contained in it, the series  $L^N_{(\lambda,t)}(f,s)$  converges absolutely. Then the proof above shows that we have for such s the identity

$$P(p^{-s}) \cdot H_n(p^{-s}) = a(n\lambda, n^2t) \cdot R(p^{-s}, t).$$

Observe that unless  $R(p^{-s}, t) = 0$ , no factor in the above equation may vanish. In fact, in case of  $p \mid t$ , we find  $R(p^{-s}, t) = 1$  regardless of s. If  $p \nmid t$ , further analysis is required. In that case, all factors in the second term of  $R(p^{-s}, t)$  that are not a power of p have absolute value 1. Hence, we find that

$$R(p^{-s}, t) \neq 0 \text{ for } \begin{cases} \operatorname{Re}(s) > k - 1, & 2 \mid \operatorname{rk}(L), \\ \operatorname{Re}(s) > k - 3/2 & 2 \nmid \operatorname{rk}(L). \end{cases}$$

These bounds are strictly better than the bounds of convergence for  $L^N_{(\lambda,t)}(f,s)$  imposed on Re(s). Hence, in the right half plane of absolute convergence of  $L^N_{(\lambda,t)}(f,s)$ , the factor  $Q(p^{-s})$  is well defined and different from zero.

The result above may be utilised to construct infinite product expansions. However, before employing this result, a remarkable notion of convergence has to be introduced.

**Definition 6.84.** Let  $(a_n)_{n \in \mathbb{N}} \in \mathbb{C}$  be a sequence of numbers. Then, the infinite product

$$\prod_{n\in\mathbb{N}}(a_n)$$

is said to be *absolutely convergent*, if the following series converges:

$$\sum_{n \in \mathbb{N}} |a_n - 1|.$$

The practical application of that notion is that an absolutely convergent product is zero, if, and only if, one of its factors equals zero. We refer the reader to [FB06, IV p. 200] for more on that matter and proceed by applying Proposition 6.83 in order to derive infinite

product expansions.

**Corollary 6.85.** Let  $N \in \mathbb{N}$ , S be a family of good primes p, and assume  $f \in S_{L,k}$  is a simultaneous eigenform of  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$  for all  $p \in S$ . Denote the Fourier expansion of f by

$$f = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) e(n\tau) \mathbf{e}_{\mu}$$

and select  $\lambda \in \mathcal{L} = L'/L$ , as well as  $t \in \overline{q}(\lambda) + \mathbb{Z}$  with  $p^2 \nmid \text{lev}(L)t$  for all  $p \in S$ . Then we find in the range of absolute convergence of  $L^N_{(\lambda,t)}(f,s)$  presented in Corollary 6.82 that

$$\begin{split} L^N_{(\lambda,t)}(f,s) &= \left(\sum_{\substack{n \in \mathbb{N} \\ \gcd(n,N)=1 \\ \forall p \in S: p \nmid n}} \frac{a(n\lambda,nt)}{n^s}\right) \\ &\cdot \prod_{\substack{p \in S \\ \gcd(p,N)=1}} \begin{cases} \frac{1 + \delta_{p \nmid t} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} \cdot p^{k-1-s}}{1 - \left(\frac{\sigma_p}{p^{k-1}} - (1-p^{-1}) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right) \cdot p^{k-1-s} + p^{2(k-1-s)}} &, 2 \mid \mathrm{rk}(L), \end{cases}$$

Further, the possibly infinite product is absolutely convergent for  $\operatorname{Re}(s) > k + 1$ .

*Proof*: The existence of the product is a consequence of Proposition 6.83 and a standard argument.

In order to prove absolute convergence of the product in the claimed range, it suffices to prove that its individual rational factors we denote by  $Q_p(p^{-s},t)$  converge to 1 for  $p \to \infty$ and all choices of  $s \in \mathbb{C}$  with  $\operatorname{Re}(s) > k + 1$  fast enough – compare Definition 6.84 on that matter. Introduce the following notation for the numerator and denominator

$$Q_p(p^{-s},t) = \frac{R_p(p^{-s},t)}{P_p(p^{-s})}.$$

Then we have

$$Q_p(p^{-s},t) - 1 = \frac{R_p(p^{-s},t) - P_p(p^{-s})}{P_p(p^{-s})}$$

and it suffices to investigate the convergence of the numerator and denominator for this expression, separately. The denominator will be shown to converge to 1. Then it suffices to prove that the numerator goes to 0 faster than linearly.<sup>11</sup> As before, we distinguish the

<sup>&</sup>lt;sup>11</sup>Hence, the series over  $Q_p(p^{-s},t) - 1$  is absolutely convergent by comparison to the Riemann  $\zeta$  function.

odd and even case, however, before specialising to one of these, recall that by Lemma 6.67 the following bound is true:

$$|\sigma_p| < p^{k-1}(p+1). \tag{6.52}$$

The Case of even signature: We begin by bounding

$$|P_p(p^{-s}) - 1| \le \left| \frac{\sigma_p}{p^{k-1}} - (1 - p^{-1}) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} \right| p^{k-1 - \operatorname{Re}(s)} + p^{2(k-1 - \operatorname{Re}(s))}$$
$$= \left( \frac{p+1}{p} + \frac{1 - p^{-1}}{p} \right) \cdot p^{k - \operatorname{Re}(s)} + p^{2(k-1 - \operatorname{Re}(s))}.$$

We see that for  $\operatorname{Re}(s) > k$ , the expression converges to 0 for  $p \to \infty$  as desired. Next, the numerator is investigated:

$$R_p(p^{-s}, t) - P_p(p^{-s})$$
  
=  $\left[\frac{\sigma_p}{p^{k-1}} + \left(\delta_{p|t} - (1-p^{-1})\right) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right] \cdot p^{k-1-s} - p^{2(k-1-s)}.$ 

We refer to Remark A.18 for a proof of the fraction of Gauss sums to be of absolute value one in order to verify that the term in brackets may be absolutely bounded by

$$\left|\frac{\sigma_p}{p^{k-1}} + \left(\delta_{p|t} - (1-p^{-1})\right)\frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right| \le (p+1) + (1-p^{-1/2}) = \left(1+2p^{-1}-p^{-2}\right) \cdot p.$$

As a consequence, the difference  $R_p(p^{-s},t) - P_p(p^{-s})$  goes to zero for  $p \to \infty$  in case of  $\operatorname{Re}(s) > k$ . It goes to 0 faster than linearly for  $\operatorname{Re}(s) > k + 1$ . This settles absolute convergence of the product in the even case.

The case of odd signature: We begin by bounding

$$|P_p(p^{-s}) - 1| \le \left| \frac{\sigma_p}{p^{k-1}} \right| \cdot p^{k-1-\operatorname{Re}(s)} + p^{2(k-1-\operatorname{Re}(s))} \\ = \frac{p+1}{p} \cdot p^{k-\operatorname{Re}(s)} + p^{2(k-1-\operatorname{Re}(s))}.$$

We see that for  $\operatorname{Re}(s) > k$ , the expression converges to 0 for  $p \to \infty$ . Next, the numerator is investigated:

$$R_p(p^{-s},t) - P_p(p^{-s})$$

$$= \left[\frac{\sigma_p}{p^{k-1}} - \chi_{\mathcal{L}}(p)\left(\frac{-t}{p}\right)p^{-1/2}\right] \cdot p^{k-1-s} - p^{2(k-1-s)}$$

The term in brackets may be bounded by

$$\left|\frac{\sigma_p}{p^{k-1}} - \chi_{\mathcal{L}}(p)\left(\frac{-t}{p}\right)p^{-1/2}\right| \le (p+1) + \sqrt{p}^{-1} = [1+p^{-1}+p^{-3/2}] \cdot p.$$

As a consequence, the difference  $Q_p(p^{-s},t) - P_p(p^{-s})$  goes to zero for  $p \to \infty$  in case  $\operatorname{Re}(s) > k$ . It goes to 0 more than linearly for  $\operatorname{Re}(s) > k + 1$ . This settles absolute convergence of the product in the odd case.

Recall that the different ranges of absolute convergence of  $L^{N}_{(\lambda,t)}(f,s)$  determined by Corollary 6.82 cover the condition  $\operatorname{Re}(s) > k + 1/2$ .

The following special case conveys the essence of the above statement but is not general enough for applications that we have in mind.

**Corollary 6.86.** Let  $2 \leq k \in \mathbb{Z}/2$  and assume  $f \in S_{L,k}$  is a simultaneous eigenform of  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$  for all primes  $p \nmid \text{lev}(L)$ . Denote the Fourier expansion of f by

$$f(\tau) = \sum_{\mu \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\mu, n) \cdot e(n\tau) \mathbf{e}_{\mu}$$

and select  $\lambda \in \mathcal{L}$  as well as  $t \in \overline{q}(\lambda) + \mathbb{Z}$  with  $p^2 \nmid \text{lev}(L)t$  for all  $p \nmid \text{lev}(L)$ . Then

$$\begin{split} L_{(\lambda,t)}(f,s) &= \left(\sum_{\substack{n \in \mathbb{N} \\ n \mid \text{lev}(L)^{\infty}}} \frac{a(n\lambda,nt)}{n^s}\right) \\ & \cdot \prod_{p \nmid \text{lev}(L)} \begin{cases} \frac{1 + \delta_{p \nmid t} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} \cdot p^{k-1-s}}{1 - \left(\frac{\sigma_p}{p^{k-1}} - (1-p^{-1}) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right) \cdot p^{k-1-s} + p^{2(k-1-s)}} & ,2 \mid \text{rk}(L), \\ \frac{1 - \chi_{\mathcal{L}}(p) \left(\frac{-t}{p}\right) p^{-1/2} \cdot p^{k-1-s}}{1 - \frac{\sigma_p}{p^{k-1}} \cdot p^{k-1-s} + p^{2(k-1-s)}} & ,2 \nmid \text{rk}(L) \end{cases}$$

is absolutely convergent as an infinite product for  $\operatorname{Re}(s) > k + 1$ .

**Corollary 6.87.** Select  $2 \le k \in \mathbb{Z}/2$  and  $f \in S_{L,k}$  with Fourier expansion

$$f(\tau) = \sum_{\mu \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\mu, n) \cdot e(n\tau) \mathfrak{e}_{\mu}$$

and let  $(\lambda, t)$  be an index of a Fourier coefficient of f such that for some  $N \in \mathbb{N}$  with

 $\operatorname{lev}(L) \mid N$  we have  $p^2 \nmid \operatorname{lev}(L)t$  for all  $p \nmid N$ . Assume f is a simultaneous eigenform for  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$  for all primes  $p \nmid N$ . Then

$$\begin{split} L_{(\lambda,t)}^{N}(f,s) &\stackrel{\text{def}}{=} \sum_{\substack{n \in \mathbb{N}, \\ \gcd(n,N)=1}} \frac{a(n\lambda, n^{2}t)}{n^{s}} \\ &= a(\lambda,t) \cdot \prod_{p \nmid N} \begin{cases} \frac{1 + \delta_{p \nmid t} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} \cdot p^{k-1-s}}{1 - \left(\frac{\sigma p}{p^{k-1}} - (1-p^{-1}) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right) \cdot p^{k-1-s} + p^{2(k-1-s)}} &, 2 \mid \mathrm{rk}(L), \\ \frac{1 - \chi_{\mathcal{L}}(p) \left(\frac{-t}{p}\right) p^{-1/2} \cdot p^{k-1-s}}{1 - \frac{\sigma p}{p^{k-1}} \cdot p^{k-1-s} + p^{2(k-1-s)}} &, 2 \nmid \mathrm{rk}(L) \end{cases}$$

converges absolutely for  $\operatorname{Re}(s) > k + 1$ . Here we have  $G_{\mathcal{L}}(d) = \sum_{\mu \in \mathcal{L}} e(d\overline{q}(\mu))$  and  $\chi_{\mathcal{L}}(p) = \epsilon_p^{\operatorname{sig}(\mathcal{L}) + \binom{-1}{|\mathcal{L}|}} \left(\frac{p}{|\mathcal{L}|^{2\operatorname{sig}(\mathcal{L})}}\right)$  with  $\epsilon_p = 1$  or i depending on wether  $p \equiv 1$  or -1 mod 4.

**Remark 6.88.** Specialising to the case of a negative definite lattice of rank 1 in Corollary 6.87 and the choice  $(\lambda, t) = (\overline{\ell_0}, q(\ell_0))$  for some suitable  $\ell_0 \in L'$ , we find by [BS08, Rem. 4.11] that  $\chi_{\mathcal{L}}(p) = \left(-\frac{q(\ell_0)}{p}\right)$ , implying

$$L^{N}_{\ell_{0}}(f,s) = a(\overline{\ell_{0}}, \mathbf{q}(\ell_{0})) \cdot \prod_{p \nmid N} \frac{1 - \delta_{p \nmid \mathbf{q}(\ell_{0})} p^{-1/2} \cdot p^{k-1-s}}{1 - \frac{\sigma_{p}}{p^{k-1}} \cdot p^{k-1-s} + p^{2(k-1-s)}}.$$

After settling the case of primes not dividing the level lev(L), we turn to the more delicate case of a prime  $p \mid lev(L)$ . Recall that in this instance the map on the discriminant form

$$\mathcal{L} \ni \lambda \to p \cdot \lambda \in \mathcal{L}$$

is not an automorphism, but has a non-trivial kernel denoted by  ${}_{p}\mathcal{L}$ . In this case, the recursion formula for Fourier coefficients of Hecke eigenforms obtained from Proposition 6.71 fans out in general, inhibiting the construction of a product expansion. However, this fan-out may be trivialised by imposing adequate constraints.

To this end, we recall that by Example 1.33 a discriminant form  $\mathcal{L}$  decomposes orthogonally into maximal p subgroups

$$\mathcal{L}\simeq igoplus_{p<\infty}\mathcal{L}_p$$

and we refer to the component of  $\lambda \in \mathcal{L}$  lying in  $\mathcal{L}_p$  as  $\lambda_p$ . With this preparation we may state appropriate conditions for a factorisation at bad places.

**Proposition 6.89.** Let  $f \in S_{L,k}$  with Fourier expansion

$$f(\tau) = \sum_{\mu \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\mu) + \mathbb{Z}} a(\mu, n) \cdot e(n\tau) \mathbf{e}_{\mu}$$

Assume  $p \neq 2$  is a prime such that f is an eigenform of  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$  and that there is no nonzero  $\lambda' \in \mathcal{L}_p$  with  $\overline{q}(\lambda') = 0$ . Select  $\lambda \in \mathcal{L}$  and  $t \in \overline{q}(\lambda) + \mathbb{Z}$  such that the p-component  $\lambda_p \in \mathcal{L}_p$  of  $\lambda$  is non-trivial or  $p^2 \nmid \text{lev}(L)t$ . If  $\lambda_p \neq 0$  assume further that fis invariant under  $O(\mathcal{L}_p)$ . Then we find for a natural number  $N \in \mathbb{N}$  with  $p \nmid N$  that

$$\begin{split} L^N_{(\lambda,t)}(f,s) &= L^{Np}_{(\lambda,t)}(f,s) \\ & \cdot \frac{1 + K_{L,p} \begin{cases} \delta_{\lambda_p \neq 0} (1 - p^{-1}) + \delta_{\lambda_p = 0} \delta_{p \nmid t}, & 2 \mid R_p \\ -\delta_{\lambda_p = 0} p^{-1/2} \left(\frac{-t}{p}\right) \epsilon_p, & 2 \nmid R_p \end{cases} \cdot p^{k-1-s} + C(\lambda_p) p^{2(k-1-s)} \\ & \frac{1 - \left(\frac{\sigma_p}{p^{k-1}} - \delta_{2|R_p} (1 - p^{-1}) K_{L,p}\right) p^{k-1-s} + p^{2(k-1-s)}}{1 - \left(\frac{\sigma_p}{p^{k-1}} - \delta_{2|R_p} (1 - p^{-1}) K_{L,p}\right) p^{k-1-s} + p^{2(k-1-s)}}. \end{split}$$

The rational factor in  $p^{k-1-s}$  extracted is well defined and nonzero for  $\operatorname{Re}(s) > \operatorname{rk}(\mathcal{L}_p)/2+k$ . Here,  $K_{L,p} = p^{-\operatorname{rk}(L)/2} \sum_{\nu \in L/pL} e\left(\frac{1}{p}q(\nu)\right)$  and  $R_p$  has been given in Definition 1.38. Further,  $\epsilon_p = 1$ , *i* depending on whether  $p = 1, 3 \mod 4$  and the constant  $C(\lambda_p) + 1$  equals the size of the orbit of  $\lambda_p \in \mathcal{L}_p$  under  $O(\mathcal{L}_p)$ . In particular,  $C(\lambda_p) = 0$ , if  $\lambda_p = 0$ .

**Remark 6.90.** Sharper bounds for the non-vanishing result are found in the proof in a variety of subcases.

*Proof*: The proof is analogous to the proof of Proposition 6.83 but more convoluted. Recall that the Fourier coefficients of  $f \in S_{L,k}$  in question are denoted by  $a(\lambda, n)$  as in Equation (6.20). For  $n \in \mathbb{N}$ , following [Shi73, p. 452], define the formal power series

$$H_n(x) \coloneqq \sum_{m=0}^{\infty} a(p^m n\lambda, (p^m n)^2 t) x^m.$$

Our goal is to prove that there is a rational function Q(x) such that

$$H_n(x) = a(n\lambda, n^2 t) \cdot Q(x). \tag{(*)}$$

Then,

$$L_{(\lambda,t)}^{N}(f,s) = \sum_{\substack{n=1\\ \gcd(n,N)}}^{\infty} a(n\lambda, n^{2}t)n^{-s}$$
$$= \sum_{\substack{n=1\\ \gcd(n,Np)}}^{\infty} H_{n}(p^{-s})n^{-s}$$
$$= L_{(\lambda,t)}^{Np}(f,s) \cdot Q(p^{-s}).$$

Recall that by Proposition 6.71, the operation of  $\mathcal{T}(p^2)$  on the Fourier coefficients of f reads

$$\sigma_p a(\lambda, n) = p^{2(k-1)} \delta_{\lambda \in ^{p}\mathcal{L}} \sum_{\substack{\lambda' \in _{p}\mathcal{L} \\ n-p^2 \,\overline{\mathbf{q}}(\lambda/p+\lambda') \in p^2 \mathbb{Z}}} a\left(\lambda/p + \lambda', \frac{n-p^2 \,\overline{\mathbf{q}}(\lambda/p+\lambda')}{p^2} + \overline{\mathbf{q}}(\lambda/p+\lambda')\right) + p^{k-2} \delta_{\lambda \in ^{p}\mathcal{L}} K_{L,p} g_p \left[1, -\chi_p {}^{R_p}, n - \mathbf{q}(p \cdot \ell_{\lambda/p})\right] a(\lambda, n) + a(p\lambda, p^2 n).$$

This results for some  $n, m \in \mathbb{N}$  with  $p \nmid n$  and  $p^2 \nmid \text{lev}(L)t$  or  $\lambda_p \neq 0$  in<sup>12</sup>

$$\sigma_p a(n\lambda, n^2 t) = p^{k-2} \delta_{\lambda \in \mathcal{PL}} K_{L,p} g_p \left[ 1, -\chi_p^{R_p}, n^2 (t - q(p \cdot \ell_{\lambda/p})) \right] a(n\lambda, n^2 t)$$

$$+ a(pn\lambda, (pn)^2 t)$$
(I)

and

$$\sigma_{p}a(p^{m}n\lambda,(p^{m}n)^{2}t) = p^{2(k-1)} \sum_{\substack{\lambda' \in_{p}\mathcal{L} \\ (p^{m}n)^{2}t - p^{2}\,\overline{\mathfrak{q}}(p^{m}n\lambda/p + \lambda') \in p^{2}\mathbb{Z}}} a\left((p^{m}n\lambda)/p + \lambda',(p^{m}n)^{2}t/p^{2}\right) + p^{k-2}K_{L,p}g_{p}\left[1,-\chi_{p}^{R_{p}},(p^{m}n)^{2}t - \mathfrak{q}(p \cdot \ell_{(p^{m}n\lambda)/p}))\right]a\left(p^{m}n\lambda,(p^{m}n)^{2}t\right) + a(p^{m+1}n\lambda,(p^{m+1}n)^{2}t).$$
(II)

The Gauss sum  $g_p$  given in Definition A.12 appearing above will be made more explicit. We begin by understanding the sum appearing in (II) meaning for  $m \in \mathbb{N}$  being different from zero. Note that for a fixed choice  $\ell_{\lambda} \in L'$  projecting to  $\lambda \in \mathcal{L}$  we have  $t = q(\ell_{\lambda}) + r$ for some integer  $r \in \mathbb{Z}$ . The expression  $g_p$  in (II) above, however, is independent of

<sup>&</sup>lt;sup>12</sup>Note that the group  $\mathcal{L}_p$  may only contain constituents of the form  $\mathbb{Z}/p\mathbb{Z}$ .

that residue r as it is multiplied by  $p^m$ . Further, recall that by Proposition 6.71, the choice of representative  $\ell_{(p^m n\lambda)/p}$  does not matter, meaning we may choose  $p^{m-1}n\ell_{\lambda}$ . As a consequence, the Gauss sum in (II) collapses to

$$g_p\left[1, -\chi_p^{R_p}, (p^m n)^2 t - q(p \cdot \ell_{(p^m n\lambda)/p}))\right] = g_p\left[1, -\chi_p^{R_p}, 0\right]$$

It should be noted that this Gauss is computed in Remark A.15. In fact, we find that

$$g_p \left[ 1, -\chi_p^{R_p}, 0 \right] = \begin{cases} 0, & \text{if } 2 \nmid R_p, \\ p - 1, & \text{if } 2 \mid R_p. \end{cases}$$

Next, the case m = 0, corresponding to (I), is considered. Here, Remark A.15 implies

$$g_p\left[1, -\chi_p^{R_p}, t - \mathbf{q}(p \cdot \ell_{\lambda/p})\right] = \begin{cases} \delta_{p \nmid t - \mathbf{q}(p\ell_{\lambda/p})} p^{1/2} \left(\frac{\mathbf{q}(p \cdot \ell_{\lambda/p}) - t}{p}\right) \epsilon_p, & \text{if } 2 \nmid R_p, \\ \delta_{p \mid t - \mathbf{q}(p\ell_{\lambda/p})} p - 1, & \text{if } 2 \mid R_p. \end{cases}$$

We deduce that the Gauss sum is indifferent to multiplication of the last argument by a square that is coprime to p. Multiplication by  $(\text{lev}(L)/p^{\nu_p(\text{lev}(L))})^2$  yields that the number  $(\text{lev}(L)/p^{\nu_p(\text{lev}(L))})^2 p^2 q(\ell_{\lambda/p})$  is an integer divisible by p.<sup>13</sup> Hence,  $\delta_{p|t-q(p\ell_{\lambda/p})} = \delta_{p|t}$  and this expression is already implicitly included in the Legendre symbol, yielding

$$g_p\left[1, -\chi_p^{R_p}, t - \mathbf{q}(p \cdot \ell_{\lambda/p})\right] = \begin{cases} p^{1/2} \left(\frac{-t}{p}\right) \epsilon_p, & \text{if } 2 \nmid R_p, \\ \delta_{p|t}p - 1, & \text{if } 2 \mid R_p. \end{cases}$$

In the next step, we investigate the index of the sum in (II). Further analysis of the discriminant group yields that  $\mathcal{L}_p = {}_p\mathcal{L}_p \simeq {}_p\mathcal{L}$  equals the *p*-torsion of the discriminant group  $\mathcal{L}$ . As a consequence, any *p* multiple  $\mu \in {}^p\mathcal{L}$  has trivial *p* component  $\mu_p$ , meaning multiplication by *p* acts as an automorphism on  ${}^p\mathcal{L}$ , yielding

$$\{p^m n\lambda/p + \lambda' \mid \lambda' \in {}_p\mathcal{L}\} = \{p^{m-1}n\lambda + \lambda' \mid \lambda' \in {}_p\mathcal{L}\}.$$

In addition, by Remark 1.36,  ${}^{p}\mathcal{L}$  is the orthogonal complement of  ${}_{p}\mathcal{L}$ , so that for  $\lambda' \in {}_{p}\mathcal{L}$ 

$$\overline{\mathbf{q}}(p^{m-1}n\lambda + \lambda') = \overline{\mathbf{q}}(p^{m-1}n\lambda) + \overline{\mathbf{q}}(\lambda')$$

<sup>&</sup>lt;sup>13</sup>Recall that we have assumed  $\mathcal{L}_p$  to be non-isotropic, implying  $p^2 \nmid \text{lev}(L)$ .

in case of m > 1 or  $\lambda_p = 0$ . Recall that  $t \in \overline{q}(\lambda) + \mathbb{Z}$  so that  $(p^m n)^2 t - p^2 q(p^{m-1}n\ell_{\lambda}) \in p^2\mathbb{Z}$ . However, we have assumed that there is no isotropic element in  $\mathcal{L}_p$ , meaning we find that the index of the sum in (II) collapses to the case  $\lambda' = 0 \in \mathcal{L}$ .

Should we have  $\lambda_p \neq 0$ , only the case m = 1 remains. We may begin arguing as above by replacing  $\lambda$  by  $\lambda - \lambda_p$  but note that there will be at least two choices of  $\lambda' \in {}_p \mathcal{L}$  this time. Namely,  $\lambda_p \neq 0$  and  $-\lambda_p$ . However, there might be even more and the number of these choices  $\lambda' \in \mathcal{L}_p$  will be denoted by  $C(\lambda_p) + 1$ . In order to proceed with the calculation the different Fourier coefficients appearing in the sum in (II) have to be related.<sup>14</sup> This is where the assumption of invariance with respect to the action of the orthogonal group comes into play. Recall that for  $\lambda_p \neq 0$ , we find  $\overline{q}(\lambda_p) \neq 0$  by assumption on the lattice. This implies the local lattice  $L_p$  to be maximal (cf. Remark 1.34) which renders  $\mathcal{L}_p$  an  $\mathbb{F}_p$ vector space. Then, mapping  $\lambda_p$  to a different choice of root with the same norm defines an isometry of subspaces, which, by Witts extension theorem 4.6, extends to an isometry of  $\mathcal{L}_p$ , i.e. an element in  $O(\mathcal{L}_p)$ . By assumption the Fourier coefficients of f are invariant with respect to this action so that we obtain  $C(\lambda_p) + 1$  contributions from this particular coefficient and identify this number as the size of the orbit of  $\lambda_p$  under  $O(\mathcal{L}_p)$ . With these manipulations, (I) and (II) become

$$\sigma_p a(n\lambda, n^2 t) = p^{k-2} \delta_{\lambda \in {}^{p}\mathcal{L}} K_{L,p} \left[ \delta_{2|R_p} (\delta_{p|t} p - 1) + \delta_{2|R_p} p^{1/2} \left( \frac{-t}{p} \right) \epsilon_p \right] a(n\lambda, n^2 t) + a(pn\lambda, (pn)^2 t)$$
(I')

and

$$\sigma_p a(p^m n\lambda, (p^m n)^2 t) = p^{2(k-1)} [\delta_{m>1} + \delta_{m=1} (C(\lambda_p) + 1)] a\left(p^{m-1} n\lambda, \frac{(p^m n)^2 t}{p^2}\right) + \delta_{2|R_p} p^{k-2} K_{L,p} \cdot (p-1) a\left(p^m n\lambda, (p^m n)^2 t\right) + a(p^{m+1} n\lambda, (p^{m+1} n)^2 t).$$
(II')

We multiply the equations above by  $x^{m+1}$  and sum them to obtain

$$\sigma_p \cdot H_n(x)x$$
  
=  $p^{2(k-1)}H_n(x)x^2 + C(\lambda_p)p^{2(k-1)}a(n\lambda, n^2t)x^2$   
+  $\delta_{2|R_p}p^{k-2}K_{L,p}\left[(p-1)H_n(x)x - a(n\lambda, n^2t)x\left\{(p-1) - \delta_{\lambda_p=0}(\delta_{p|t}p - 1)\right\}\right]$ 

<sup>&</sup>lt;sup>14</sup>In fact, in the sum in (II), there are now indices appearing for different choices of  $\lambda'$  which were not contained in the initial series  $L^N_{(\lambda,t)}(f,s)$ .

$$+ \delta_{2 \nmid R_p} p^{k-2} K_{L,p} \delta_{\lambda_p=0} p^{1/2} \left(\frac{-t}{p}\right) \epsilon_p a(n\lambda, n^2 t) x$$
$$+ H_n(x) - a(n\lambda, n^2 t).$$

This will have to be rearranged. Beforehand, note that the term in curly braces may be rewritten as

$$(p-1) - \delta_{\lambda_p=0}(\delta_{p|t}p - 1) = (p-1) - \delta_{\lambda_p=0}[(p-1 + (\delta_{p|t}p - p)] = \delta_{\lambda_p \neq 0}(p-1) - \delta_{\lambda_p=0}p(\delta_{p|t} - 1) = \delta_{\lambda_p \neq 0}(p-1) + \delta_{\lambda_p=0}\delta_{p|t}p.$$

With that transformation, we may rearrange the above equation for  $H_n(x)$  to read

$$H_{n}(x) - H_{n}(x)x\left(\sigma_{p} - \delta_{2|R_{p}}(p-1)p^{k-2}K_{L,p}\right) + H_{n}(x)x^{2}p^{2(k-1)}$$

$$= a(n\lambda, n^{2}t) \times \left[1 + p^{k-2}K_{L,p} \begin{cases} \delta_{\lambda_{p}\neq0}(p-1) + \delta_{\lambda_{p}=0}\delta_{p|t}p, & 2 \mid R_{p} \\ -\delta_{\lambda_{p}=0}p^{1/2}\left(\frac{-t}{p}\right)\epsilon_{p}, & 2 \nmid R_{p} \end{cases} \cdot x - C(\lambda_{p})p^{2(k-1)}x^{2}\right]$$

which yields the desired rational expression, once x is substituted with  $p^{-s}$ :

$$H_n(p^{-s})$$

$$= a(n\lambda, n^{2}t) \cdot \frac{1 + K_{L,p} \begin{cases} \delta_{\lambda_{p} \neq 0}(1 - p^{-1}) + \delta_{\lambda_{p} = 0}\delta_{p \nmid t}, & 2 \mid R_{p} \\ -\delta_{\lambda_{p} = 0}p^{-1/2} \left(\frac{-t}{p}\right)\epsilon_{p}, & 2 \nmid R_{p} \end{cases} \cdot p^{k-1-s} - C(\lambda_{p})p^{2(k-1-s)} \\ \frac{1 - \left(\sigma_{p} - \delta_{2 \mid R_{p}}(1 - p^{-1})p^{k-1}K_{L,p}\right)p^{-s} + p^{2(k-1-s)}}{1 - \left(\sigma_{p} - \delta_{2 \mid R_{p}}(1 - p^{-1})p^{k-1}K_{L,p}\right)p^{-s} + p^{2(k-1-s)}}.$$

However, it remains to check that the potential denominator is well defined. First, we may assume  $L^N_{(\lambda,t)}(f,s) \neq 0$  for there is nothing to show otherwise. Then we may fix n such that  $a(n\lambda, n^2t) \neq 0$  and write

$$Q(p^{-s}) = \frac{R(p^{-s}, t)}{P(p^{-s})}$$

for the numerator and denominator of the rational expression above. In the right half plane of absolute convergence of  $L^{N}_{(\lambda,t)}(f,s)$ , also  $H_{n}(p^{-s})$  converges absolutely, yielding

$$P(p^{-s})H_n(p^{-s}) = a(n\lambda, n^2t) \cdot R(p^{-s}, t)$$

in this right half plane. Note that in the above equation no factor vanishes, unless  $R(p^{-s},t)$  does, meaning it suffices to verify that  $R(p^{-s},t)$  does not vanish, to conclude that  $P(p^{-s})$  does not vanish as well. To this end, we shall specify a right half plane, such that  $|1 - R(p^{-s},t)| < 1$ . The first factor that needs to be tended to is  $K_{L,p}$ . By Remark A.18 we find

$$|K_{L,p}| = p^{-\operatorname{rk}(L)/2} |G_{L,p}(1,1)| = \sqrt{|p\mathcal{L}|}.$$

Recall that in our case the local lattice  $L_p$  was assumed to be maximal, implying  ${}_p\mathcal{L} \simeq \mathcal{L}_p$ which results in its size being equal to  $p^{\mathrm{rk}(\mathcal{L}_p)}$ . Also note that  $C(\lambda_p) = 0$  in case  $\lambda_p = 0$ . If  $\lambda_p \neq 0$ , we necessarily find  $p \mid \mathrm{lev}(L)$  and conclude that  $C(\lambda_p) \leq |\mathcal{L}_p| - 2 = p^{\mathrm{rk}(\mathcal{L}_p)} - 2$ . We will have a look at the event that  $2 \mid R_p$  first which yields

$$|1 - R(p^{-s}, t)| \le \begin{cases} \left( p^{\operatorname{rk}(\mathcal{L}_p)/2} (1 - p^{-1}) + [p^{\operatorname{rk}(\mathcal{L}_p)} - 2] p^{k-1 - \operatorname{Re}(s)} \right) p^{k-1 - \operatorname{Re}(s)}, & \lambda_p \neq 0, \\ p^{\operatorname{rk}(\mathcal{L}_p)/2 + k - 1 - \operatorname{Re}(s)}, & \lambda_p = 0. \end{cases}$$

Next, the case  $2 \nmid R_p$  is considered which gives

$$|1 - R(p^{-s}, t)| \le \begin{cases} [p^{\operatorname{rk}(\mathcal{L}_p)} - 2]p^{2(k-1-\operatorname{Re}(s))}, & \lambda_p \neq 0, \\ p^{\operatorname{rk}(\mathcal{L}_p)/2 + k - 3/2 - \operatorname{Re}(s)}, & \lambda_p = 0. \end{cases}$$

With the gathered information we may guarantee non-vanishing of the numerator for

For the case  $2 \mid R_p$  and  $\lambda_p \neq 0$ , we chose a suboptimal bound by bounding  $(1 - p^{-1}) \leq 1$ and  $(1 - p^{-\operatorname{rk}(\mathcal{L}_p)}) \leq 1/3$ . The exact choice may be computed by means of a quadratic equation. In detail, note that in this case the expression is strictly decreasing in  $\operatorname{Re}(s)$ , write  $z = p^{k-1-\operatorname{Re}(s)}$  and  $c_p = p^{\operatorname{rk}(\mathcal{L}_p)/2} \frac{1-p^{-1}}{p^{\operatorname{rk}(\mathcal{L}_p)}-2}$ . Then we seek the even point of

$$z^2 + c_p z = 1.$$

Solving the quadratic equation results in only the positive solution being acceptable,

meaning

$$p^{k-1-\operatorname{Re}(s)} = -\frac{c_p}{2} + \sqrt{\frac{c_p^2}{4}} + 1,$$

yielding

$$\operatorname{Re}(s) = k - 1 - \ln_p(2) + \ln_p\left(\sqrt{p^{\operatorname{rk}(\mathcal{L}_p)} \left(\frac{1 - p^{-1}}{p^{\operatorname{rk}(\mathcal{L}_p)} - 2}\right)^2 + 4} - p^{\operatorname{rk}(\mathcal{L}_p)/2} \frac{1 - p^{-1}}{p^{\operatorname{rk}(\mathcal{L}_p)} - 2}\right).$$

Note that in case of a good prime  $p \nmid \text{lev}(L)$ , the parameter  $R_p$  equals the rank and  $\lambda_p = 0$  vanishes trivially as the associated local lattice is unimodular, so that we may recover Proposition 6.83 in case  $p \neq 2$ . For the reduction to that case, also compare Corollary 6.72 and Remark A.18 where the respective Gauss sums are explicitly compared to each other in the even case.

As another corollary, we obtain a factorisation of L-series. Note that it suffices in light of Corollary 6.86 to reduce to the case of bad primes and compare Remark 1.34 for the appearing maximality condition.

**Corollary 6.91.** Select  $N | \text{lev}(L)^{\infty}$  and a Hall divisor  $N_0 | N$  with  $2 \nmid N_0$  such that  $L_p$  is maximal for all primes  $p | N_0$ . Further, let  $f \in S_{L,k}$  be a simultaneous eigenform of all  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$  for the primes  $p | N_0$ . Denote the Fourier expansion of f by

$$f(\tau) = \sum_{\mu \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\mu) + \mathbb{Z}} a(\mu, n) \cdot e(n\tau) \mathfrak{e}_{\mu}$$

and select an index  $(\lambda, t)$  such that for all primes  $p \mid N_0$  we have  $p^2 \nmid \text{lev}(L)t$  or  $\lambda_p = 0$ . For the primes  $p \mid N_0$  for which  $\lambda_p \neq 0$ , assume that f is invariant with respect to  $O(\mathcal{L}_p)$ . Then

$$\sum_{\substack{n \in \mathbb{N} \\ n \mid N^{\infty}}} \frac{a(n\lambda, nt)}{n^s}$$
$$= \sum_{\substack{n \in \mathbb{N} \\ n \mid (N/N_0)^{\infty}}} \frac{a(n\lambda, nt)}{n^s}$$

$$\cdot \prod_{p|N_0} \frac{1 + K_{L,p} \begin{cases} \delta_{\lambda_p \neq 0} (1 - p^{-1}) + \delta_{\lambda_p = 0} \delta_{p \nmid t}, & 2 \mid R_p \\ -\delta_{\lambda_p = 0} p^{-1/2} \left(\frac{-t}{p}\right) \epsilon_p, & 2 \nmid R_p \end{cases}}{1 - \left(\frac{\sigma_p}{p^{k-1}} - \delta_{2|R_p} (1 - p^{-1}) K_{L,p}\right) p^{k-1-s} + p^{2(k-1-s)}}$$

for  $\operatorname{Re}(s) > \max_{p|N_0} \{\operatorname{rk}(\mathcal{L}_p)\}/2 + k$  in the notation of Proposition 6.89.

Combining this factorisation with Corollary 6.86 yields the following product expansion.

**Corollary 6.92.** Assume L to be maximal and select a simultaneous eigenform  $f \in S_{L,k}$ of all  $\mathcal{T}(p^2)$  with eigenvalue  $\sigma_p$ . Write

$$f(\tau) = \sum_{\mu \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\mu) + \mathbb{Z}} a(\mu, n) \cdot e(n\tau) \mathfrak{e}_{\mu}$$

for its Fourier expansion and select an index  $(\lambda, t)$  such that for all primes p we have  $p^2 \nmid \text{lev}(L)t$  or  $\lambda_p \neq 0$ . Assume f to be invariant with respect to the action of  $O(\mathcal{L}_p)$  for these primes for which  $0 \neq \lambda_p \in \mathcal{L}_p$ . Then

$$\begin{split} & L_{(\lambda,t)}(f,s) \\ & \stackrel{\text{def}}{=} \sum_{n \in \mathbb{N}} \frac{a(n\lambda, n^2 t)}{n^s} \\ & = \begin{cases} \sum_{n \in \mathbb{N}} \frac{a(2^n\lambda, 2^n t)}{2^{ns}}, & 2 \mid \text{lev}(L), \\ a(\lambda, t), & 2 \nmid \text{lev}(L) \end{cases} \\ & \quad \cdot \prod_{p \nmid \text{lev}(L)} \begin{cases} \frac{1 + \delta_{pt} \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} \cdot p^{k-1-s}}{1 - \left(\frac{\sigma p}{p^{k-1}} - (1-p^{-1}) \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)}\right) \cdot p^{k-1-s} + p^{2(k-1-s)}} & , 2 \mid \text{rk}(L), \\ \frac{1 - \chi_{\mathcal{L}}(p) \left(\frac{-t}{p}\right) p^{-1/2} \cdot p^{k-1-s}}{1 - \frac{\sigma p}{p^{k-1}} \cdot p^{k-1-s} + p^{2(k-1-s)}} & , 2 \nmid \text{rk}(L) \end{cases} \\ & \quad \cdot \prod_{2 \neq p \mid \text{lev}(L)} \frac{1 + K_{L,p} \begin{cases} \delta_{\lambda p \neq 0} (1 - p^{-1}) + \delta_{\lambda p = 0} \delta_{p \nmid t}, & 2 \mid R_p \\ -\delta_{\lambda p = 0} p^{-1/2} \left(\frac{-t}{p}\right) \epsilon_p, & 2 \nmid R_p \end{cases} \cdot p^{k-1-s} + C(\lambda_p) \cdot p^{2(k-1-s)} \\ & \quad 1 - \left(\frac{\sigma p}{p^{k-1}} - \delta_{2|R_p} (1 - p^{-1}) K_{L,p}\right) p^{k-1-s} + p^{2(k-1-s)} \end{cases} . \end{split}$$

The product converges absolutely for  $\operatorname{Re}(s) > k + 1$  as long as the factors in the finite product are well defined. This is definitely the case for  $\operatorname{Re}(s) > \max{\operatorname{rk}(\mathcal{L}_p)}/{2 + k}$  and in this range all rational factors in  $p^{-s}$  are different from zero.

Here,  $G_{\mathcal{L}}(d) = \sum_{\lambda \in \mathcal{L}} e(d \,\overline{q}(\lambda))$  and  $K_{L,p} = p^{-\operatorname{rk}(L)/2} \sum_{v \in L/pL} e(q(v))$ , while  $\lambda_p$  denotes the projection of  $\lambda \in \mathcal{L}$  to the p-component  $\mathcal{L}_p$ . Further,  $R_p$  is given in Definition 1.38 and

$$\chi_{\mathcal{L}}(p) = \epsilon_p^{\operatorname{sig}(L) + \left(\frac{-1}{|\mathcal{L}|}\right)} \left(\frac{p}{|\mathcal{L}| 2^{\operatorname{sig}(\mathcal{L})}}\right), \quad \text{with } \epsilon_p = \begin{cases} 1, & p \equiv 1 \mod 4, \\ i, & p \equiv 3 \mod 4. \end{cases}$$

### 6.4.3 Rankin–Selberg L-series

In this section, the *L*-series associated to vector valued modular forms in Subsection 6.4.1 are constructed via an integral pairing in Rankin–Selberg fashion. We begin with a brief introduction of the central tool for the method, the unfolding trick.

#### The unfolding trick

Let G be a group,  $(X, \mu)$  be a measured space, such that  $G \odot X$  acts faithfully and invariantly with respect to the measure. Further, let  $H \leq G$  be a subgroup.

**Lemma 6.93.** Assume there is a measurable representation  $\mathcal{F}_G$  of  $G \setminus X$  and select a subgroup  $H \leq G$  with countable  $H \setminus G$ . Then

$$\mathcal{F}_H \coloneqq \bigcup_{\gamma \in H \setminus G} \gamma^{-1} \mathcal{F}_G$$

is a measurable representation of  $H \setminus X$ .

*Proof*: Let  $x \in X$ , then there is  $g \in G$  with  $gx \in \mathcal{F}_G$ . Now there is some  $\gamma \in H \setminus G$  and  $h \in H$  with  $\gamma h = g$ . Apparently,  $hx = \gamma^{-1}\gamma hx = \gamma^{-1}gx \in \gamma^{-1}\mathcal{F}_G$ .

Assume we had  $x_1, x_2 \in \mathcal{F}_H$  and  $h \in H$  with  $x_1 = hx_2$ . Then there are  $\gamma_1, \gamma_2 \in H \setminus G$  with  $x_i \in \gamma_i^{-1} \mathcal{F}_G$ , so that there are  $y_i \in \mathcal{F}_G$  such that  $\gamma_1^{-1} y_1 = x_1 = x_2 = h\gamma_2^{-1} y_2$ , meaning  $x_1 = \gamma_1 h \gamma_2^{-1} y_2 \in \mathcal{F}_G$ . Hence,  $\gamma_1 h \gamma_2^{-1} = e$ , translating to  $\gamma_1 h = \gamma_2 \in \gamma_1 H$ , implying  $\gamma_1 = \gamma_2$ , and resulting in h = e.

**Lemma 6.94.** Assume there is a measurable representation of  $G \setminus X$ , a subgroup  $H \leq G$  of at most countably infinite index and a function  $f : (X, \mu) \to \mathbb{C}$  which is invariant under H and integrable on a measurable representation of  $H \setminus X$ . Then

$$\int_{G \setminus X} \sum_{\gamma \in H \setminus G} \gamma^* f \, \mathrm{d}\mu = \int_{H \setminus X} f \, \mathrm{d}\mu.$$
(6.53)

*Proof*: We have

$$\int_{G \setminus X} \sum_{\gamma \in H \setminus G} \gamma^* f \, \mathrm{d}\mu = \sum_{\gamma \in H \setminus G} \int_{G \setminus X} \gamma^* f \, \mathrm{d}\mu = \sum_{\gamma \in H \setminus G} \int_{\gamma^{-1}(G \setminus X)} f \, \mathrm{d}\mu = \int_{H \setminus X} f \, \mathrm{d}\mu,$$

where convergence is verified from right to left.

There are some straightforward generalisations to the above setting. First, one may assume that  $G \oplus X$  acts not faithfully, but that its isotropy group of an element  $x \in X$ is normal in G and contained in H. However, even that is not necessary, but their are measure theoretic inconveniences arising when considering more general situations – compare [Neu29] for a note on that matter. Further, the last result is indifferent to null sets, meaning that conditions may be neglected on null-sets and also the notion of a fundamental domain – a set of representatives of the orbits – may be defined up to null sets.

### Constructing *L*-series

The most prominent application of the unfolding trick that has been briefly reviewed in the section above is the choice of  $X = \mathbb{H}$  the upper half plane,  $G = \Gamma(1) = \mathrm{SL}_2(\mathbb{Z})$ acting via Möbius transformations and as a subgroup  $H = \Gamma_{\infty}$ . Recall that a fundamental domain  $\mathcal{F}_{\Gamma(1)}$  is, up to a null-set, given by

$$\mathcal{F}_{\Gamma(1)} = \{ \tau \in \mathbb{H} \mid |\operatorname{Re}(\tau)| \le 1/2, |\tau| \ge 1 \}$$

(cf. Figure 2.1.2) and except for i,  $\rho$ , and  $\rho^2$ , the isotropy group is  $\{\pm \mathcal{I}\} < \Gamma(1)$ , which may be divided out. The considered measure is the hyperbolic measure on the upper half plane which is given by  $d\mu = \frac{du \, dv}{v^2}$ , where we represent  $\tau = u + iv \in \mathbb{H}$  with  $u = \operatorname{Re}(\tau), v = \operatorname{Im}(\tau)$  (cf. ).

In this context, Rankin<sup>15</sup> [Ran39b] and Selberg have considered integrals of the form

$$\int_{\mathcal{F}_{\Gamma(1)}} \varphi \cdot G_0(\tau, s) \, \mathrm{d}\mu,$$

where  $\varphi = f \cdot \overline{g} \cdot \text{Im}^k$  was essentially a product of two modular forms  $f, g \in \mathcal{M}_k(\Gamma(1))$ that was assumed to be a cusp form, so that  $\varphi$  was automorphic. Further,  $G_0$  denotes a non-holomorphic Eisenstein series of weight 0. The resulting integral would unfold and define an *L*-series with analytic continuation and functional equation [Ran39b, Thm. 3 p. 360].

In the following, we will mimic that procedure in order to construct *L*-series as in (6.47) for cusp forms  $f \in S_{L,k}$ . For that purpose, we will pair f against a non-holomorphic Eisenstein series that is twisted with a theta series. We will first prove that the unfolding trick is applicable in our context. However, we will do so in greater generality, as we have

<sup>&</sup>lt;sup>15</sup>Note that Rankin has already considered the case of the arithmetic group  $\Gamma(N)$ .

future applications in mind. As a consequence, the following technical result is meant for the reader who is interested in details. Its Corollary 6.96 suffices for the proof of Theorem 7.16 which is the main application within the scope of the present thesis.

Recall that for  $\eta \in \mathcal{L} = L'/L$  and suitable parameters  $m, \psi, k$  we have defined vector valued parabolic Poincaré series  $P_{\eta,m,\psi,k}$  (cf. Definition 3.47 and its succeeding comments).

**Proposition 6.95.** Let L be an even lattice that splits as  $L = L_1 \oplus L_2$ ,

- $f \in \mathcal{S}_{L,k}$  with Fourier coefficients  $a(\lambda, n) \in \mathcal{O}_{\varepsilon}(n^{\nu_f})$ ,
- $g \in \mathcal{M}_{L_1,k_1}$  with Fourier coefficients  $b(\lambda, n) \in \mathcal{O}_{\varepsilon}(n^{\nu_g})$ ,
- and  $P_{\eta,m,\psi,k_2} = \sum_{\gamma \in \overline{\Gamma_{\infty}} \setminus \operatorname{Mp}_2(\mathbb{Z})} \psi \mathfrak{e}_{\eta}(m \cdot)|_{L_2,k_2} \gamma \in \mathcal{A}_{L_2,k_2}$  be a parabolic Poincaré series in the notation of Definition 3.47 such that  $k = k_1 + k_2$ . Assume that  $\psi \in \mathcal{L}^{\infty}_{\operatorname{loc}}(\mathbb{R}_{>0})$  and that there are constants  $\alpha_0, \alpha_{\infty} \in \mathbb{R}$  such that  $\psi(v) \in \mathcal{O}(v^{\alpha_0})$  for  $v \to 0$  and  $\psi(v) \in \mathcal{O}(v^{\alpha_{\infty}})$  for  $v \to \infty$ .

Then

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, g \otimes P_{\eta, m, \psi, k} \rangle \cdot \operatorname{Im}^{k} \, \mathrm{d}\mu = \int_{\mathcal{F}_{\Gamma_{\infty}}} \langle f, g \otimes \psi \mathfrak{e}_{\eta}(m \cdot) \rangle \cdot \operatorname{Im}^{k} \, \mathrm{d}\mu \tag{6.54}$$

provided  $\alpha_0 > \max\{3 - k + \nu_f + \nu_g, 1 - k/2\}.$ 

Note that by assumption of f, g being modular forms we find by Theorem 2.25 and Corollary 3.85 that the parameters  $\nu_f$ ,  $\nu_g$  are explicitly selectable such that  $a(\lambda, n) = \mathcal{O}(n^{\nu_f})$  and  $b(\mu, n) = \mathcal{O}(n^{\nu_g})$ .

*Proof*: First, we assume absolute convergence of the integral on the right hand side of (6.54) and verify the result of the unfolding method. We use the abbreviations  $H := \overline{\Gamma_{\infty}}$ ,  $G = Mp_2(\mathbb{Z})$ , as well as  $h_{\eta,m} = \psi \mathfrak{e}_{\eta}(m \cdot)$  and begin by computing

$$\begin{split} \int_{\mathcal{F}_{\Gamma(1)}} \langle f, g \otimes P_{\eta, m, \psi, k} \rangle \cdot \mathrm{Im}^{k} \, \mathrm{d}\mu &= \int_{\mathcal{F}_{\Gamma(1)}} \left\langle f, g \otimes \sum_{\gamma \in H \setminus G} h_{\eta, m} |_{k_{2}} \gamma \right\rangle \cdot \mathrm{Im}^{k} \, \mathrm{d}\mu \\ &= \int_{\mathcal{F}_{\Gamma(1)}} \sum_{\gamma \in H \setminus G} \langle f, g \otimes h_{\eta, m} |_{k_{2}} \gamma \rangle \cdot \mathrm{Im}^{k} \, \mathrm{d}\mu \\ &= \int_{\mathcal{F}_{\Gamma(1)}} \sum_{\gamma \in H \setminus G} \langle f |_{k} \gamma, g |_{k_{1}} \gamma \otimes h_{\eta, m} |_{k_{2}} \gamma \rangle \cdot \mathrm{Im}^{k} \, \mathrm{d}\mu \\ &= \int_{\mathcal{F}_{\Gamma(1)}} \sum_{\gamma \in H \setminus G} \langle f |_{k} \gamma, (g \otimes h_{\eta, m}) |_{k} \gamma \rangle \cdot \mathrm{Im}^{k} \, \mathrm{d}\mu \end{split}$$

$$= \int_{\mathcal{F}_{\Gamma(1)}} \sum_{\gamma \in H \setminus G} \gamma^* \left( \langle f, g \otimes h_{\eta,m} \rangle \cdot \mathrm{Im}^k \right) \, \mathrm{d}\mu$$
$$= \int_{\mathcal{F}_{\Gamma_{\infty}}} \langle f, g \otimes h_{\eta,m} \rangle \cdot \mathrm{Im}^k \, \mathrm{d}\mu.$$
(6.55)

Subsequently, we have to prove absolute convergence of the unfolded integral. Note that

$$\langle f, g \otimes h_{\eta, m} \rangle = \left\langle \sum_{\substack{\lambda_1 \in L'_1/L_1 \\ \lambda_2 \in L'_2/L_2}} f_{\lambda_1 + \lambda_2} \mathfrak{e}_{\lambda_1 + \lambda_2}, \sum_{\substack{\lambda_1 \in L'_1/L_1 \\ \eta \in L'_1/L_1}} g_{\lambda_1} \psi \mathfrak{e}_{\lambda_1 + \eta}(m \cdot) \right\rangle$$

$$= \sum_{\substack{\lambda_1 \in L'_1/L_1 \\ \eta \in L'_1/L_1}} f_{\lambda_1 + \eta} \overline{g_{\lambda_1} \psi e(m \cdot)}$$

$$(6.56)$$

where indices of f and g denote the respective component function. Assume we had  $\psi(v) = 1$  and m = 0, then

$$\langle f, g \otimes h_{\eta, m} \rangle = \sum_{n \in \mathbb{Z}/\operatorname{lev}(L)} c(n) q^n$$

has a Fourier expansion that is convergent and hence absolutely convergent on  $\mathbb{H}$  (cf. Lemma A.19). Returning to the case of general  $h_{\eta,m}$ , we restrict to considering the behaviour  $v \to \infty$ , first and note that it suffices to prove convergence of the integral on a strip  $S \subseteq \{z \in \mathbb{C} \mid |\text{Re}(z)| \le 1/2\}$  with arbitrary positive minimum height such that  $\psi(v)$  is bounded by some positive multiple of  $v^{\alpha_{\infty}}$ . Then, we may use the same trick as in Example 3.52 d) to verify that for general  $\psi(v)$  we have the following bound on S

$$\begin{split} \left| \left\langle f, g \otimes h_{\eta, m} \right\rangle \right| &\leq |\psi(v)| \sum_{n \in \mathbb{N}/\operatorname{lev}(L)} |c(n)| e^{-2\pi v n} \\ &\leq |\psi(v)| e^{-2\pi v/(2\operatorname{lev}(L))} \sum_{n \in \mathbb{N}/\operatorname{lev}(L)} |c(n)| e^{2\pi i n(iv/2)}. \end{split}$$

The right sum however, is bounded by a constant, since the Fourier series is absolutely convergent on  $\mathbb{H}$  and the integral over S of the last expression converges absolutely.

We turn to the case  $v \to 0$  and will proceed with inserting the Fourier expansion of both modular forms f and g in (6.56) to proceed with the computation. Recall that there are Fourier coefficients, such that

$$f(\tau) = \sum_{\lambda \in L'/L} \sum_{0 < n \in \mathbb{Z} + \overline{\mathbf{q}}(\lambda)} a(\lambda, n) e(n\tau) \mathfrak{e}_{\lambda},$$

$$g(\tau) = \sum_{\mu \in L_1'/L_1} \sum_{0 \leq n \in \mathbb{Z} + \overline{\mathbf{q}}(\mu)} b(\mu, n) e(n\tau) \mathfrak{e}_{\mu}.$$

With that notation, we derive the following explicit expansion.

$$\sum_{\lambda_1 \in L_1'/L_1} f_{\lambda_1 + \eta} \overline{g_{\lambda_1}}$$
  
= 
$$\sum_{\lambda_1 \in L_1'/L_1} \sum_{0 < n \in \mathbb{Z} + \overline{q}(\lambda_1) + \overline{q}(\eta)} \sum_{0 \le n_1 \in \mathbb{Z} + \overline{q}(\lambda_1)} a(\lambda_1 + \eta, n) \overline{b(\lambda_1, n_1)} \cdot e(n\tau - n_1\overline{\tau}).$$

With this expansion, we shall prove absolute convergence of the unfolded integral. First, we note that it is sufficient to bound each of the  $|L'_1/L_1|$  summands in  $\lambda_1$  arising in (6.56) separately. Consequently, we fix  $\lambda_1$  in such a fashion that the associated term dominates the others, when multiplied with a suitable constant  $C_1 > 0$ . Next, we rewrite

$$|e(n\tau - n_1\overline{\tau})| = \exp[-2\pi v(n+n_1)].$$

In the following, we may assume  $\nu_f, \nu_g \ge 0$ . Combining these observations with (6.56) yields that the scalar product is bounded by

$$|\langle f, g \otimes h_{\eta, m} \rangle| \leq |L_1'/L_1| C_1 \sum_{\substack{0 < n \in \mathbb{Z} + \overline{\mathbf{q}}(\lambda_1) + \overline{\mathbf{q}}(\eta) \\ 0 \leq n_1 \in \mathbb{Z} + \overline{\mathbf{q}}(\lambda_1)}} n^{\nu_f} n_1^{\nu_g} \cdot \exp[-2\pi v(n+n_1)] \cdot |\psi(v)|.$$

We write N := lev(L) and note that a summation over more possible choices of  $n, n_1$  potentially increases the sum.

$$\begin{split} |\langle f,g \otimes h_{\eta,m} \rangle| &\leq |L_1'/L_1| C_1 \sum_{\substack{0 \neq n \in \mathbb{Z}/N \\ n_1 \in \mathbb{Z}/N}} |n|^{\nu_f} |n_1|^{\nu_g} \cdot \exp[-2\pi v(|n|+|n_1|)] \cdot |\psi(v)| \\ &= |L_1'/L_1| C_1 \sum_{\substack{0 \neq n \in \mathbb{Z} \\ n_1 \in \mathbb{Z}}} \frac{|n|^{\nu_f} |n_1|^{\nu_g}}{N^{(\nu_f+\nu_g)}} \cdot \exp[-2\pi v(|n|+|n_1|)/N] \cdot |\psi(v)|. \end{split}$$

We simplify the appearing series. First, we observe that we may bound the numerator

$$|n|^{\nu_f} |n_1|^{\nu_g} \le \max\{|n|, |n_1|\}^{(\nu_f + \nu_g)}.$$

Recall further that  $|\psi(v)|$  was bounded for  $v \to 0$  by  $v^{\alpha_0}$ . Hence, we may bound  $|\psi(v)| \leq \frac{C_2}{|L'_1/L_1|C_1}v^{\alpha_0}$  close to the real line for a positive constant  $C_2 > 0$  and without loss

of generality we may assume that the bound for  $\psi$  is valid on  $\mathcal{F}_{\Gamma_{\infty}}$ . Consequently, we deduce

$$\int_{\mathcal{F}_{\Gamma_{\infty}}} |\langle f, g \otimes h_{\eta, m} \rangle| \cdot \operatorname{Im}^{k} d\mu$$
  
$$\leq C_{2} \cdot \int_{0}^{\infty} \sum_{\substack{0 \neq n \in \mathbb{Z} \\ n_{1} \in \mathbb{Z}}} \frac{\max\{|n|, |n_{1}|\}^{(\nu_{f} + \nu_{g})}}{N^{\nu_{f} + \nu_{g}}} \cdot e^{-\frac{2\pi}{N}(|n| + |n_{1}|) \cdot v} v^{\alpha_{0} + k - 2} dv$$

substituting  $v \mapsto v \cdot N/[2\pi(|n| + |n_1|)]$  yields

$$= C_2 \cdot \sum_{\substack{0 \neq n \in \mathbb{Z} \\ n_1 \in \mathbb{Z}}} \frac{\max\{|n|, |n_1|\}^{\nu_f + \nu_g}}{N^{(\nu_f + \nu_g)} [2\pi(|n| + |n_1|)/N]^{\alpha_0 + k - 1}} \cdot \Gamma(\alpha_0 + k - 1)$$
  
$$= \frac{C_2}{N^{\nu_f + \nu_g}} \frac{\Gamma(\alpha_0 + k - 1)}{(2\pi/N)^{\alpha_0 + k - 1}} \cdot \sum_{\substack{0 \neq n \in \mathbb{Z} \\ n_1 \in \mathbb{Z}}} \frac{\max\{|n|, |n_1|\}^{\nu_f + \nu_g}}{(|n| + |n_1|)^{\alpha_0 + k - 1}}.$$

We are left with the task of bounding the series to the right. First, we increase it by summing over  $0 \neq (n, n_1) \in \mathbb{Z}^2$  and write  $x \coloneqq (n, n_1)$ . Then we note that  $|n| + |n_1| = ||x||_1$  and  $\max\{|n|, |n_1|\} = ||x||_{\infty}$ . However, these norms are equivalent (cf. Remark A.7), implying that the series converges, if and only if,

$$\sum_{0 \neq x \in \mathbb{Z}^2} \frac{1}{\|x\|_{\infty}^{\alpha_0 + k - 1 - (\nu_f + \nu_g)}}$$

converges. The convergence of this type of series, however, has already been established in Lemma 6.75 by computing the size of the discrete  $\|\cdot\|_{\infty}$ -sphere and comparing the result with the Riemann  $\zeta$ -function. It converges for  $\alpha_0 + k - 1 - (\nu_f + \nu_g) > 2$ , meaning

$$\alpha_0 > 3 - k + \nu_f + \nu_g.$$

A special instance of Proposition 6.95 where the Poincaré series is given by an Eisenstein series is integral for the main goal of this thesis. This yields the following observation which is relevant for the computation of cycle integrals of the Kudla–Millson liftings in Section 7.3.1.

**Corollary 6.96.** Let  $f \in S_{L,k}$ ,  $L = L_1 \oplus L_2$  split,  $g \in \mathcal{M}_{L_1,k_1}$ , and  $\eta \in L'_2/L_2$  be isotropic. Further, choose  $E_{L_2,\eta,k_2}(\tau,s)$  be the Eisenstein series from Definition 3.37 such

that  $k = k_1 + k_2$ . Then there is some  $s_0 \in \mathbb{R}$  such that

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, g \otimes E_{L_2,\eta,k_2}(\,\cdot\,,s) \rangle \cdot \operatorname{Im}^k \, \mathrm{d}\mu = \int_{\mathcal{F}_{\Gamma_{\infty}}} \langle f, g \otimes \mathfrak{e}_\eta \rangle \cdot \operatorname{Im}^{\overline{s}+k} \, \mathrm{d}\mu \tag{6.57}$$

for all  $s \in \mathbb{C}$  with  $\operatorname{Re}(s) > s_0$  and both sides are holomorphic in  $\overline{s}$ .

In order to construct the *L*-series presented in (6.47), we will have to choose *g* to equal a theta series. Assume, we had  $(L, q) = (L_1, q_1) \oplus (L_2, q_2)$  with  $(L_1, q_1)$  positive definite of rank  $m_1$  and associated theta function  $\Theta_{L_1}(\tau) \coloneqq \Theta_{L_1}(\tau; \exp(-2\pi q_1))$ . Selecting  $g = \Theta_{L_1}(\tau)$ , we obtain for  $f \in \mathcal{S}_{L,k}$  a cusp form and suitable  $s \in \mathbb{C}$  that

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, \Theta_{L_1} \otimes E_{L_2, \eta, k_2}(\cdot, s) \rangle \cdot \operatorname{Im}^k \, \mathrm{d}\mu = \int_{\mathcal{F}_{\Gamma_{\infty}}} \langle f, \Theta_{L_1} \otimes \mathfrak{e}_\eta \operatorname{Im}^s \rangle \cdot \operatorname{Im}^k \, \mathrm{d}\mu.$$

We compute the integrand before continuing:

$$\begin{split} \left\langle f, \Theta_{L_1}(\tau) \otimes \mathfrak{e}_{\eta} \right\rangle &= \left\langle \sum_{\substack{\lambda_1 \in L'_1/L_1\\\lambda_2 \in L'_2/L_2}} f_{\lambda_1 + \lambda_2} \mathfrak{e}_{\lambda_1 + \lambda_2}, \sum_{\substack{\lambda_1 \in L'_1/L_1\\\lambda_1 \in L'_1/L_1}} \theta_{L_1, \lambda_1} e_{\lambda_1 + \eta} \right\rangle \\ &= \sum_{\lambda_1 \in L'_1/L_1} f_{\lambda_1 + \eta} \overline{\theta_{L_1, \lambda_1}}. \end{split}$$

We proceed similarly to the proof of the technical Proposition 6.95. This time, however, we do not have to consider the absolute value of the integrand and a significant portion of the terms appearing will cancel, allowing for a relatively concrete computation. For that purpose, we assume  $\operatorname{Re}(s)$  to be large enough to guarantee absolute convergence. To this end, recall that f has a representation as a Fourier series as well as the definition of  $\Theta_{L_1}$ (compare for instance Definition 3.31 in conjunction with Example 3.33 in the positive definite case).

$$\begin{split} f(\tau) &= \sum_{\lambda \in L'/L} \sum_{0 < n \in \mathbb{Z} + \overline{\mathbf{q}}(\lambda)} a(\lambda, n) e(n\tau) \mathfrak{e}_{\lambda}, \\ \Theta_{L_1}(\tau) &= \sum_{\lambda_1 \in L'_1/L_1} \sum_{l \in L_1 + \lambda_1} e(\mathbf{q}(l)\tau) \mathfrak{e}_{\lambda_1}. \end{split}$$

We insert the Fourier expansion of both functions in order to obtain (recall that  $\eta \in L'_2/L_2$ 

was isotropic)

$$\sum_{\lambda_1 \in L'_1/L_1} f_{\lambda_1 + \eta} \overline{\theta_{L_1, \lambda_1}}$$

$$= \sum_{\lambda_1 \in L'_1/L_1} \sum_{0 < n \in \mathbb{Z} + \overline{q}(\lambda_1)} \sum_{l \in L_1 + \lambda_1} a(\lambda_1 + \eta, n) \cdot e(n\tau) e\left[-\overline{\tau} q(l)\right]$$

$$= \sum_{\lambda_1 \in L'_1/L_1} \sum_{0 < n \in \mathbb{Z} + \overline{q}(\lambda_1)} \sum_{l \in L_1 + \lambda_1} a(\lambda_1 + \eta, n) \cdot e[iv(n + q(l))] e\left[u(n - q(l))\right]$$

Clearly,  $n - q(l) \in \mathbb{Z}$ , so that  $\mathbb{R} \ni u \mapsto e[u(n - q(l))]$  defines a character which descends to a character on  $\mathbb{T}$ . The unit circle  $\mathbb{T}$ , however, is compact, implying by Remark 1.51 the following identity:

$$\int_0^1 e\left[u(n-\mathbf{q}(l))\right] \, \mathrm{d}u = \delta_{n,\mathbf{q}(l)},$$

where the right hand side is to be understood as a Kronecker symbol. As a consequence, we conclude

$$\int_{\mathcal{F}_{\Gamma_{\infty}}} \left\langle f, \Theta_{L_{1}} \otimes \mathfrak{e}_{\eta} \right\rangle \cdot \operatorname{Im}^{\overline{s}+k} \, \mathrm{d}\mu$$

$$= \int_{0}^{\infty} \sum_{\lambda_{1} \in L_{1}'/L_{1}} \sum_{0 \neq l \in L_{1}+\lambda_{1}} a(\lambda_{1}+\eta, \mathbf{q}(l)) \cdot e[iv2\,\mathbf{q}(l)] \cdot v^{\overline{s}+k-2} \, \mathrm{d}v$$

$$= \sum_{\lambda_{1} \in L_{1}'/L_{1}} \sum_{0 \neq l \in L_{1}+\lambda_{1}} a(\lambda_{1}+\eta, \mathbf{q}(l)) \cdot \int_{0}^{\infty} \exp[-4\pi\,\mathbf{q}(l) \cdot v] \cdot v^{\overline{s}+k-2} \, \mathrm{d}v.$$

Applying the subtitution  $v\mapsto v/4\pi\,\mathbf{q}(l)$  yields that this equals

$$= \sum_{\lambda_1 \in L'_1/L_1} \sum_{0 \neq l \in L_1 + \lambda_1} \frac{a(\lambda_1 + \eta, \mathbf{q}(l))}{(4\pi \, \mathbf{q}(l))^{\overline{s} + k - 1}} \cdot \int_0^\infty e^{-v} \cdot v^{\overline{s} + k - 2} \, \mathrm{d}v$$
$$= \frac{\Gamma(\overline{s} + k - 1)}{(4\pi)^{\overline{s} + k - 1}} \sum_{\lambda_1 \in L'_1/L_1} \sum_{0 \neq l \in L_1 + \lambda_1} \frac{a(\lambda_1 + \eta, \mathbf{q}(l))}{\mathbf{q}(l)^{\overline{s} + k - 1}},$$

where instead of summing over all elements of all cosets of  $L'_1$  by  $L_1$ , we may as well sum over the whole lattice  $L'_1$  to obtain the following, simpler description

$$=\frac{\Gamma(\overline{s}+k-1)}{(4\pi)^{\overline{s}+k-1}}\sum_{0\neq l\in L_1'}\frac{a(\overline{l}+\eta,\mathbf{q}(l))}{\mathbf{q}(l)^{\overline{s}+k-1}}.$$

Comparing the last result with Definition 6.77, we notice that the series appearing is ex-

actly  $L_{L_1,\eta,1}(f, \overline{s}+k-1)$ . Recall that there is  $\sigma \in \mathbb{R}$  such that  $a(\lambda, q(l)) \in \mathcal{O}_{\varepsilon}(q(l)^{k/2-\sigma+\varepsilon})$ - compare Table 3.1 for concrete choices of  $\sigma$ . Then a comparison with Lemma 6.76 yields that the last expression converges absolutely for  $\operatorname{Re}(s) > \frac{m_1-k}{2} + 1 - \sigma$  with  $\sigma$  as in the Lemma and defines a holomorphic function in  $\overline{s}$  in that right half plane. As a consequence, we have proven the following Proposition.

**Proposition 6.97.** Let  $f \in S_{L,k}$ ,  $k \ge 2$ ,  $L = L_1 \oplus L_2$  with positive definite  $L_1$  of rank  $m_1$ ,  $E_{L_2,\eta,k_2}(\cdot, s)$  be the Eisenstein series of Definition 3.37, and assume  $k = m_1/2 + k_2$ . Then we have the identity

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, \Theta_{L_1} \otimes E_{L_2,\eta,k_2}(\cdot,s) \rangle \cdot \operatorname{Im}^k \, \mathrm{d}\mu = \frac{\Gamma(\overline{s}+k-1)}{(4\pi)^{\overline{s}+k-1}} \cdot L_{L_1,\eta,1}(f,\overline{s}+k-1)$$
(6.58)

of holomorphic functions in  $\overline{s}$  for  $\operatorname{Re}(s) > \frac{m_1-k}{2} + 1 - \sigma$ , where  $\sigma = 1/2$  or 1/4 depending on whether  $2 | \operatorname{rk}(L)$  or not. Further, the left hand side gives rise to a meromorphic continuation of  $L_{L_1,\eta,1}$ .

There is a specialisation that plays a key role in the proof of Theorem 7.16.

**Corollary 6.98.** Let (L,q) be even such that  $L = \mathbb{Z}\ell \oplus L_2$  for some  $\ell \in L$  of positive norm and  $f \in S_{L,k}$ . Assume we have  $k > 3 - 2\sigma$  with  $\sigma$  as above. Then

$$\int_{\mathcal{F}_{\Gamma(1)}} \langle f, \Theta_{\mathbb{Z}\ell} \otimes E_{L_2,0,k-1/2}(\cdot,0) \rangle \cdot \mathrm{Im}^k \, \mathrm{d}\mu = \frac{2\Gamma(k-1)}{(4\pi \,\mathrm{q}(\ell_0))^{k-1}} \delta_{\mathrm{sig}(L)\equiv 2k} \, {}_{(4)} \cdot L_{\ell_0}(f,2k-2) \tag{6.59}$$

for a primitive  $\ell_0 \in L'$  such that  $\ell_0 \mid \ell \in L'$ . Here,  $\delta$  denotes a Kronecker symbol.

*Proof*: In Proposition 6.97, we select  $L_1 = \mathbb{Z}\ell$  and find  $m_1 = \operatorname{rk}(\mathbb{Z}\ell) = 1$ . Further, the choice s = 0 necessitates  $k > 2 + m_1 - 2\sigma = 3 - 2\sigma$  for reasons of convergence. Writing out (6.58) yields

$$\begin{split} \int_{\mathcal{F}_{\Gamma(1)}} \left\langle f, \Theta_{L_1} \otimes E_{L_2, 0, k-1/2}(\,\cdot\,, 0) \right\rangle \cdot \mathrm{Im}^k \, \mathrm{d}\mu &= \frac{\Gamma(k-1)}{(4\pi)^{k-1}} \sum_{0 \neq l \in L_1'} \frac{a(\bar{l}, \mathbf{q}(l))}{\mathbf{q}(l)^{k-1}} \\ &= \frac{\Gamma(k-1)}{(4\pi \,\mathbf{q}(\ell_0))^{k-1}} \sum_{0 \neq n \in \mathbb{Z}} \frac{a(n\overline{\ell_0}, n^2 \,\mathbf{q}(\ell_0))}{|n|^{2k-2}}. \end{split}$$

Recall that by Remark 3.20 there are no non-trivial modular forms in case  $2 \nmid \operatorname{sig}(\mathcal{L}) + 2k$ , anyway. Further, in case of  $\operatorname{sig}(\mathcal{L}) \equiv 2k + 2 \mod 4$  the series vanishes by the symmetry property of the coefficients under reflection of the index (cf. Remark 3.22). Hence,

$$\sum_{0 \neq n \in \mathbb{Z}} \frac{a(n\overline{\ell_0}, n^2 \operatorname{q}(\ell_0))}{|n|^{2k-2}} = 2 \cdot \delta_{\operatorname{sig}(L) \equiv 2k} {}_{(4)} \cdot \sum_{0 \neq n \in \mathbb{N}} \frac{a(n\overline{\ell_0}, n^2 \operatorname{q}(\ell_0))}{n^{2(k-1)}}.$$

Finally, the series on the right hand side equals  $L_{\ell_0}(f, 2(k-1))$  by Definition 6.79.

The special value  $L_{\ell_0}(f, 2(k-1))$  in (6.59) plays a key role in the context of computing cycle integrals of Kudla–Millson liftings in the subsequent chapter about theta lifts.

**Remark 6.99.** We recall that the Eisenstein series  $E_{L_2,0,k-1/2}$  appearing in (6.59) has a symmetry property (cf. Corollary 3.44). This symmetry may naively be utilised to relate the special value to another value for which properties like non-vanishing might be derivable. However, upon closer investigation we find that the right hand side of (6.59) had, under the induced reflection property, to be evaluated at s = 1 - k, where neither the *L*-series nor the  $\Gamma$ -prefactor converge.

# 7 Automorphic lifts

In the preceding sections two distinct types of varieties have been discussed, namely symplectic and orthogonal varieties as well as their associated automorphic forms. It turns out that forms of one type may be utilised to construct forms of the other type by means of so called *lifts*. The lifts considered in this thesis are of a special type, referred to as *theta lifts*. These are given by integrating against a kernel that is constructed by symmetrising a smooth object with respect to representations of a dual reductive pair. A concrete description, tailored to our setting, is presented below.

The celebrated paper [BF04] of Bruinier and Funke serves as a fundamental source for the subsequent discussion and a lattice (L, q) of signature  $(m^+, m^-)$  is assumed to be fixed.

## 7.1 Borcherds Lift

The celebrated product expansion of the discriminant function

$$\Delta = q \prod_{n=1}^{\infty} (1 - q^n)^{24}$$
(7.1)

derived from the respective product of the Dedekind eta function is well established in the field of automorphic forms. It was this expression that prompted Richard Borcherds to pursue a more conceptual approach to product expansions of modular forms.

In 1995 Borcherds published a paper [Bor95] in which he presented a rigorous method for obtaining infinite product expansions of modular forms with character for  $SL_2(\mathbb{Z})$  by lifting weakly holomorphic modular forms of weight 1/2 to the latter. The lift is multiplicative, transforming the sum of modular forms into a product of their images. Furthermore, the constant coefficient of the input form represents the weight of the resulting function [Bor95, Thm. 14.1 p. 204]. This theorem even yields information about the zeroes and poles of the target function as algebraic number theoretic data. As an example Borcherds retrieves identity (7.1) and product expansions for the Klein *j*-invariant as well as several Eisenstein series. The following selection represents some of these examples.

Example 7.1. For  $12 \cdot \theta(\tau) = 12 \sum_{n \in \mathbb{Z}} e^{2\pi i n^2 \tau} = \sum_{n \in \mathbb{Z}} c(n) q^n$  and the series  $\sum_{n \in \mathbb{Z}} a(n) q^n$  as in [Bor95, p. 204] we obtain

$$\Delta(\tau) = q^1 \prod_{n>0} (1-q^n)^{c(n^2)} \quad \text{and} \quad j(\tau) = q^{-1} \prod_{n>0} (1-q^n)^{a(n^2)}.$$

A far more intricate example is the following construction.

Example 7.2. The modified Klein's j invariant  $j(\tau) - 744 = \sum_{n \ge -1} c(n)q^n$  lifts to

$$j(\tau_1) - j(\tau_2) = q_1^{-1} \prod_{m>0, n \in \mathbb{Z}} (1 - q_1^m q_2^n)^{c(mn)}$$

known as the *monster denominator formula*. This may be utilised to endow the monster module with a multiplication, inducing a vertex operator algebra structure, which ultimately leads to solving the Moonshine conjecture.

In his renowned 1998 paper [Bor98] Borcherds significantly expanded on his initial approach, presenting a multiplicative lift [Bor98, Thm. 13.3 p. 544] that maps weakly holomorphic vector valued modular forms of weight  $1 - m^+/2$  to modular forms for certain discrete automorphism groups in  $O_{m^+,2}(\mathbb{R})$ , i.e. to the orthogonal setting. In the case of  $m^+ = 1$ , the association may be interpreted as a purely symplectic lift, due to the accidental isomorphism of  $SO_{1,2}(\mathbb{R})$  and  $Sp_1(\mathbb{R})$ . For an excellent concise source on the fundamentals which are presented in the following, see [Bru14].

It should be noted that the basis of Borcherds' construction is a theta lift utilising a Siegel theta function as in Definition 3.29 (see also Remark 3.28)

$$\Theta_L(\tau, z) = v^{\frac{m^-}{2}} \sum_{\lambda \in L'} e\left[ u \operatorname{q}(\lambda) + iv \operatorname{q}_z^+(\lambda) \right] \mathfrak{e}_{\lambda + L}$$

as an integral kernel which is paired with a weakly holomorphic vector valued modular form  $f \in \mathcal{M}_{L,k}^!$ , namely

$$\Phi(z,f) \coloneqq \int_{\mathrm{Mp}_2(\mathbb{Z})\backslash \mathbb{H}}^{\bullet} \langle f(\tau), \Theta_L(\tau, z) \rangle \frac{\mathrm{d}u \, \mathrm{d}v}{v^2}.$$
(7.2)

Here, we have written  $u + iv = \tau \in \mathbb{H}$  and  $z \in \mathbb{D}$  is an element of the Grassmannian. Further, the dot accompanying the integral sign signifies a necessary regularisation (cf. [Bor98]). This pairing is then utilised to derive the multiplicative Borcherds lift, the main features of which are collected in the following theorem.<sup>1</sup>

**Theorem 7.3** (Borcherds). Let  $f \in \mathcal{M}^{!}_{L^{-},1-m^{+}/2}$  be a weakly holomorphic modular form with Fourier coefficients  $a(\lambda, n)$  as in (3.40). Assume that  $a(\lambda, n)$  is integral for  $n \leq 0$ . Then there exists a meromorphic modular form  $\Psi(\cdot, f)$  for the discriminant kernel  $\Gamma(L)$ with unitary multiplier system of finite order, such that the following conditions are satisfied.

- a) The weight of  $\Psi(\cdot, f)$  is given by a(0,0)/2.
- b) The divisor of the target form  $\Psi(\cdot, f)$  is given by the principal part of f via

$$\frac{1}{2} \sum_{\mu \in L'/L} \sum_{n>0} a(\mu, -n) Z(\mu, n), \tag{7.3}$$

where

$$Z(\mu, n) = \sum_{\substack{v \in \mu + L \\ \mathbf{q}(v) = n}} \{ z \in \mathbb{D} \mid z \perp v \}$$

is the Heegner divisor of discriminant  $(\mu, n)$  (cf. Definition 4.70 plus the subsequent comment).

c) The target form  $\Psi(\cdot, f)$  has explicit infinite product expansions.

For further details regarding the product expansions that are constructed for different cusps, or background material on the construction itself, we refer the reader to [Bor98, Thm. 13.3 p. 544] and conclude with the following definition.

**Definition 7.4.** For a form  $f \in \mathcal{M}^{!}_{L^{-},1-m^{+}/2}$  as above, we refer to  $\Psi(\cdot, f)$  as the *Borcherds lift* of the form f or, alternatively, as the *Borcherds product* of f.

# 7.2 Kudla–Millson lift

In the 1980s Kudla and Millson [KM86] introduced special Schwartz forms  $\varphi_{\rm KM}$  on the symmetric spaces attached to the classical groups  $O(m^+, m^-)$ ,  $U(m^+, m^-)$ , and  $Sp(m^+, m^-)$ , taking values in closed differential forms. Their principal objective was to investigate cohomology classes of special cycles by means of a theta correspondence, generalising the celebrated work of Hirzebruch and Zagier [HZ76] on Hilbert modular

<sup>&</sup>lt;sup>1</sup>We denote by  $L^-$  the lattice (L, -q).

surfaces. More precisely, in the orthogonal case, Kudla and Millson symmetrised the Schwartz forms  $\varphi_{\rm KM}$  over a base lattice L and let the Weil representation of the symplectic group act on  $\varphi_{\rm KM}$  in order to obtain a kernel  $\Theta_{\rm KM}(\tau, z)$  in a symplectic and an orthogonal variable, which transforms automorphically in both variables. These may then be employed as an integral kernel to lift automorphic objects from the symplectic to the orthogonal setting or vice versa. The authors continued their work in [KM87] in great generality and we recommend [KM90] as an introductory source and [BF04] for a perspective restricting to the elliptic case on the symplectic side.

In the following subsection, the theta function  $\Theta_{\rm KM}$ , as well as the associated lift will be described in a more explicit manner. We begin by reviewing the construction of the Schwartz form  $\varphi_{\rm KM}$ .

## 7.2.1 The Kudla–Millson Schwartz form

In this section we will provide a brief overview of the construction of the Kudla–Millson Schwartz form within the context relevant to our study, while also establishing the necessary notation. It should be noted that some of its properties have already been stated in Subsection 5.4.2, as they were required in the context of the Siegel–Weil formula. This section is based on [BF04, Sec. 2, 4 p. 64], which serves as an excellent source of further background information and some of the content of the following is also found in [MZ23].

We begin by recalling the setting. Let (L, b) be an even  $\mathbb{Z}$  lattice of signature  $(m^+, m^-)$ and rank m. It should be recalled that  $V = L \otimes \mathbb{Q}$  is the associated rational quadratic space and that  $V(\mathbb{R}) = V \otimes \mathbb{R}$  denotes the enclosing real space. An orthogonal basis  $(e_i)$ of  $V(\mathbb{R})$  with  $b(e_i, e_i) = 1$  for  $1 \leq i \leq m^+$  and  $b(e_i, e_i) = -1$  for  $m^+ < i \leq m$  is selected, effectively choosing an isometry to the standard real quadratic vector space  $\mathbb{R}^{(m^+,m^-)}$ of Example 1.5. Recall that the symmetric domain associated to L has a model as the Grassmannian

$$\mathbb{D} = \{ z \subset V \mid \dim z = m^{-}, q \mid_{z} < 0 \},\$$

which will be equipped with the base point  $z_0$  spanned by  $(e_i)_{m^+ < i \leq m}$ . Further, we denote its stabiliser in SO<sup>+</sup>( $V(\mathbb{R})$ ) by K and note that it is maximal compact. Additional details can be found in Subsection 4.2.1. Recall that for any  $z \in \mathbb{D}$  we find  $V(\mathbb{R}) = z \oplus z^{\perp}$  so that each element  $v \in V(\mathbb{R})$  may be decomposed uniquely into  $v = v_z + v_{z^{\perp}} \in z \oplus z^{\perp}$ . Note
that this decomposition gives rise to the positive definite standard majorant of q, namely

$$\mathbf{q}_z(v)^+ = \mathbf{q}(v_{z^\perp}) - \mathbf{q}(v_z).$$

Let  $\mathfrak{g}$  denote the Lie algebra of SO<sup>+</sup>( $V(\mathbb{R})$ ) and  $\mathfrak{g} = \mathfrak{p} + \mathfrak{k}$  be its Cartan decomposition. Then  $\mathfrak{p} \simeq \mathfrak{g}/\mathfrak{k}$  identifies with the tangent space at  $z_0$  of  $\mathbb{D}$ . With respect to the basis above, we have

$$\mathfrak{p} \simeq \left\{ \begin{pmatrix} 0 & X \\ X^T & 0 \end{pmatrix} \middle| X \in \mathbb{R}^{m^+ \times m^-} \right\} \simeq \mathbb{R}^{m^+ \times m^-}.$$
(7.4)

We denote the  $m^-$ -forms on  $\mathbb{D}$  by  $\mathscr{A}^{m^-}(\mathbb{D})$ . Then

$$\left[\mathcal{S}(V(\mathbb{R}))\otimes\mathscr{A}^{m^{-}}(\mathbb{D})\right]^{\mathrm{SO}^{+}(V(\mathbb{R}))}\simeq\left[\mathcal{S}(V(\mathbb{R}))\otimes\bigwedge^{m^{-}}\mathfrak{p}^{*}\right]^{K},$$

where the isomorphism is given by evaluation at the base point and the form may be extended to the entire Grassmannian  $\mathbb{D}$  by the action of  $\mathrm{SO}(V(\mathbb{R}))$  via pullback. Therefore, we may represent  $\varphi_{\mathrm{KM}}$  as an element in  $\left[\mathcal{S}(V(\mathbb{R})) \otimes \bigwedge^{m^-} \mathfrak{p}^*\right]^K$ . To this end, let  $X_{\alpha,\beta} \in \mathbb{R}^{m^+ \times m^-}$  for  $1 \leq \alpha \leq m^+$  and  $1 \leq \beta \leq m^-$  denote the basis matrix with value one in the  $(\alpha, \beta)$ -entry and zero elsewhere. The corresponding dual element will be denoted by  $w_{\alpha,\beta}$  and multiplication with it by  $A_{\alpha,\beta}$ . Then, define the following differential operator where we identify the cotangent space of  $\mathbb{D}$  at  $z_0$  with  $\mathfrak{p}^*$  under (7.4):

$$\mathscr{D}^{m^+,m^-} \coloneqq \frac{1}{2^{m^-/2}} \prod_{\beta=1}^{m^-} \left[ \sum_{\alpha=1}^{m^+} \left( x_\alpha - \frac{1}{2\pi} \frac{\partial}{\partial x_\alpha} \right) \otimes A_{\alpha,\beta} \right].$$
(7.5)

This operator is applied to the standard Gaussian  $\varphi_0 \otimes 1 \in \left[\mathcal{S}(V(\mathbb{R})) \otimes \bigwedge^0 \mathfrak{p}^*\right]^K$  in order to obtain the Kudla–Millson Schwartz form

$$\varphi_{\rm KM} = \mathscr{D}^{m^+, m^-} \varphi_0 \otimes 1 \tag{7.6}$$

which is closed. Based upon this construction, a more explicit description of the form is computed in [MZ23, Sec. 2.2] which invokes products of Hermite polynomials. It is noteworthy that  $\Omega \coloneqq \varphi_{\text{KM}}(0, \cdot)$  is the Euler form of the symmetric space  $\mathbb{D}$  and in the hermitian case of  $m^- = 2$ , its negative defines a Kähler form. In the latter case, a description in coordinates on the associated tube domain model has been provided in (5.27). Of particular interest for this thesis is the hermitian case of signature  $(m^+, 2)$ , for which the form  $\varphi_{\rm KM}$  is, in fact, a (1, 1)-form.

With the Kudla–Millson Schwartz form available, the associated theta kernel may be constructed as presented in Definition 3.32.

**Definition 7.5.** Let  $\tau = u + iv \in \mathbb{H}$  and  $z \in \mathbb{D}$ . Then the Kudla–Millson theta function of the lattice L is given by

$$\Theta_{\mathrm{KM}}(\tau, z) \coloneqq \Theta_L(\tau, z; \varphi_{\mathrm{KM}}) \coloneqq v^{-m/4} \sum_{\lambda \in L'/L} \sum_{l \in \lambda + L} \left( \omega_{\infty}(g_{\tau}) \varphi_{\mathrm{KM}} \right) (l, z) \mathfrak{e}_{\lambda}.$$
(7.7)

It should be noted that this form is explicitly expressed in terms of Siegel-theta functions in [MZ23]. With the form  $\Theta_{\rm KM}$  at hand, we may construct associated lifts of automorphic forms.

#### 7.2.2 The Kudla–Millson lift

As with the Borcherds lift, the theta kernel associated to the Kudla–Millson Schwartz form may be utilised as an integral kernel to construct a theta lift.

**Definition 7.6.** Let *L* be an even lattice of signature  $(m^+, m^-)$  and rank *m*. Set k := m/2, then for a cusp form  $f \in \mathcal{S}_{L,k}(Mp_2(\mathbb{Z}))$  and  $\tau = u + iv \in \mathbb{H}$ , the association

$$f \mapsto \Lambda_{\mathrm{KM}}(f) \coloneqq \int_{\mathrm{Mp}_2(\mathbb{Z}) \setminus \mathbb{H}} v^k \langle f(\tau), \Theta_L(\tau, z; \varphi_{\mathrm{KM}}) \rangle \, \frac{\mathrm{d}u \, \mathrm{d}v}{v^2} \tag{7.8}$$

defines a map to the  $m^-$ -forms  $\mathscr{A}^{m^-}(\mathbb{D})$  and is referred to as Kudla-Millson lift.

The target form  $\Lambda_{\text{KM}}(f)$  is in general closed and inherits the invariance under the discriminant kernel subgroup  $\Gamma(L)$  of L on the orthogonal side from the theta form  $\Theta_{\text{KM}}$ . Consequently, it descends to a form on the quotient space  $Y_L = \Gamma(L) \setminus \mathbb{D}$ . In the hermitian case of  $m^- = 2$ , we obtain a map

$$\Lambda_{\rm KM}: \mathcal{S}_{L,1+m^+/2} \to \mathcal{H}^{(1,1)}(Y_L), \tag{7.9}$$

to the space of square integrable harmonic differential forms of Hodge type (1,1) on the orthogonal modular variety associated to the discriminant kernel  $\Gamma(L)$ .

The applications of the Kudla–Millson theta correspondence range from the study of the cohomology of orthogonal and unitary Shimura varieties [KM86] and Arakelov theory of Shimura varieties [Kud04a] over specific counting problems [EGS23], constructing mock modular forms and higher dimensional error functions [FK17], to proving a converse theorem for Borcherds products [Bru02] [Bru14]. It is the latter application that we have in mind which is related to the injectivity of the Kudla–Millson lift. The question of its injectivity already arose in [KM90, p. 122] and may be used to compute the rational Picard number of the underlying Shimura variety [Ber+16], as well as to derive properties of cones generated by special cycles [BM19] [Zuf22]. Since the Kudla–Millson lift has been constructed, there have been a number of advances in prooving its injectivity and we would like to present these results. The first statement found in the literature, the author is aware of, is the following.

**Theorem 7.7** ([Bru02, Thm. 5.12 p. 139]). Let (L, q) be an even lattice of signature  $(m^+, 2)$  such that it splits two hyperbolic planes  $L \simeq K \oplus H \oplus H$ . Then the Kudla–Millson lift  $\Lambda_{\text{KM}} : S_{L,1+m^+/2} \to \mathcal{H}^{(1,1)}(Y_L)$  is injective.

The proof is based upon the computation of the Fourier expansion of the target form, which reveals information about the Fourier coefficients of the initial form f and leads to the conclusion that it must vanish if its image does. The second hyperbolic split is required in order to guarantee that indices of Fourier coefficients are representable by the lattice in a certain fashion similar to Lemma 3.68.

There has been a second advance by the same author in collaboration with Jens Funke in [BF10]. In this instance, the authors develop a new strategy of proof, by utilising the doubling method to compute the  $L^2$ -norm of the Kudla-Millson lift by means of the Rallis inner product formula in order to conclude injectivity in the unimodular case for general signature. In addition, the authors consider a twist of the Kudla-Millson lift, denoted by  $\Lambda_{\text{KMF},l}$ , which was introduced by Funke and Millson [FM12] with a parameter  $l \in \mathbb{N}_0$  that allows for inputting forms of higher weight. The twisted lift  $\Lambda_{\text{KMF},l}$  maps to the space of  $\widetilde{\text{Sym}}^l(V)$  valued closed differential forms on  $Y_L$ . Here,  $\widetilde{\text{Sym}}^l(V)$  denotes the local system on  $\mathbb{D}$  associated to the *l*-the symmetric power of V (cf. Definition 4.20). Details are provided in the primary source or in a compact fashion in [BF10] and we refer to the lift  $\Lambda_{\text{KMF},l}$  as the *Kudla-Millson-Funke* lift of degree  $l \in \mathbb{N}_0$ . The latter source also contains the following injectivity statement.

**Theorem 7.8** ([BF10, Cor. 4.11 p. 37]). Assume that  $m > \max\{4, 3+r\}, m^+ > 1, m^- + l$ even, and that L is even unimodular. Then the theta lift  $\Lambda_{\text{KMF},l} : S_{L,k+l} \to \mathscr{A}^{m^-}(Y_L, \widetilde{\text{Sym}}^l(V))$  is injective.

This approach has recently been generalised by Stein to include the case of maximal

lattices by solving intricate local integrals for bad primes in the same setting.

**Theorem 7.9** ([Ste23, Cor. 7.8 p. 25]). Let  $m > \max\{6, 2l - 2, 3 + r\}$  and assume that  $m^- + l$  as well as  $k = \frac{m}{2} + l$  are even. If L'/L is assumed to be anisotropic, then  $\Lambda_{\text{KMF},l} : \mathcal{S}_{L,k+l} \to \mathscr{A}^{m^-}(Y_L, \widetilde{\text{Sym}}^l(V))$  is injective.

Nevertheless, the case with the most compelling application within this thesis remains the  $O(m^+, 2)$  setting since it relates to the Borcherds lift – a connection that is discussed in further detail below. In this case, the strongest result available in the literature is attributed to Bruinier and proven by refining his approach in [Bru02] through the development of a newform theory for vector valued modular forms in order to weaken the assumption of a second hyperbolic split.

**Theorem 7.10** ([Bru14, Thm. 5.3 p. 331]). Let (L, q) be an even lattice of signature  $(m^+, 2)$ such that it splits a hyperbolic plane and a scaled hyperbolic plane  $L \simeq K \oplus H(N) \oplus H$  for some  $N \in \mathbb{N}$ . Then the Kudla–Millson lift  $\Lambda_{\text{KM}} : S_{L,1+m^+/2} \to \mathcal{H}^{(1,1)}(Y_L)$  is injective.

It should also be noted that quite recently, Zuffetti and the author have extended this result to the case of general signature, also including the Funke–Millson twist.

**Theorem 7.11** ([MZ23, Thm. 6.2 p. 24]). Let L be an even indefinite lattice of signature  $(m^+, m^-)$ .

- i) If  $L \simeq K \oplus H(N) \oplus H$  for some even lattice K and some positive integer N, then the lift  $\Lambda_{\text{KMF},l}$  is injective.
- ii) Let  $m^- = 1$ . If  $L \simeq M \oplus H$  for some positive definite even lattice M and  $M \otimes \mathbb{Z}_p$ splits off a hyperbolic plane for every prime p, then the lift  $\Lambda_{\text{KMF},l}$  is injective.
- iii) Let  $m^+ = 1$ . Then the lift  $\Lambda_{\text{KMF},l}$  associated to L is identically zero.

It is noteworthy that this result implies all of the above. The proof of Theorem 7.11 relies on the utilisation of Borcherds' method of expressing theta functions with respect to sublattices [Bor98, Sec. 5] and realising these as Poincaré series in order to apply the unfolding method. The application of this method yields a Fourier expansion of the lift in terms of the Fourier coefficients of the initial form. In this context, the first hyperbolic split is then employed to simplify this expansion allowing to extract vanishing results for the initial coefficients, while the additional split of a scaled hyperbolic plane is required to guarantee that the lattice represents a sufficient number of indices of Fourier coefficients of the initial form. In order to rely on the weaker assumption of a *scaled* hyperbolic split, the newform theory of [Bru14, Sec. 3] is invoked.

#### 7.2.3 Linking lifts

An alternative approach to constructing the Kudla–Millson lift  $\Lambda_{\rm KM}$  based on Borcherds' additive lift  $\Phi(z, f)$ , as presented in (7.2), has been described by Bruinier and Funke and will be outlined in the following paragraphs. In their celebrated paper [BF04], based upon this construction, the authors prove a duality statement between the Borcherds and Kudla–Millson lift. In this context, they formally introduce a new space of Maass wave forms that has been foreshadowed by [Bru02] and relate it to holomorphic modular forms via a differential operator.

We will briefly present the setting and highlight the aforementioned connection between the injectivity of the Kudla–Millson lift and the converse theorem for Borcherds products, as outlined in [Bru14, Sec. 4.1].

**Definition 7.12.** Let (L, q) be an even lattice. For  $k \in \mathbb{Z}/2$  with  $k \neq 1$  denote by  $H_{L,k}$  the space of *weak Maass forms* of weight k with respect to the Weil representation  $\rho_L$ . This space consists of all real analytic functions  $f : \mathbb{H} \to \mathbb{C}[L'/L]$ , satisfying the following.

- a) f transforms as a modular form:  $f(\gamma \tau) = \phi^{2k} \rho_L(\gamma, \phi) f(\tau)$  for all  $(\gamma, \phi) \in Mp_2(\mathbb{Z})$ .
- b) f grows at most exponentially towards the cusps.
- c) f is annihilated by the hyperbolic Laplace operator of weight k given by

$$\Delta_k = -v^2 \left( \frac{\partial^2}{\partial u^2} + \frac{\partial^2}{\partial v^2} \right) + ikv \left( \frac{\partial}{\partial u} + i \frac{\partial}{\partial v} \right).$$

Such a form has a Fourier expansion on the upper half plane, however, its coefficients are not invariant under imaginary shifts. It should also be noted that the hyperbolic Laplace operator may be expressed in terms of the *Maass raising* and *lowering* operator. To be explicit, the operators

$$R_k = 2i\frac{\partial}{\partial\tau} + kv^{-1}, \qquad L_k = -2iv^2\frac{\partial}{\partial\overline{\tau}}$$

can be concatenated to construct  $-\Delta_k = L_{k+2}R_k + k = R_{k-2}L_k$ . Bruinier and Funke then craft an antilinear operator

$$\xi_k : H_{L,k} \to \mathcal{M}^!_{L^-,2-k}, \qquad f(\tau) \mapsto v^{k-2} \overline{L_k f(\tau)} = R_{-k} v^k \overline{f(\tau)}$$
(7.10)

with kernel  $\mathcal{M}_{L,k}^!$ . Recall that  $L^-$  denotes the lattice L with quadratic form rescaled by

the factor -1. The crucial point is that the operator  $\xi_k$  is in fact surjective, resulting in the exact sequence

$$0 \to \mathcal{M}_{L,k}^! \to H_{L,k} \xrightarrow{\xi_k} \mathcal{M}_{L^-,2-k}^! \to 0.$$
(7.11)

If we restrict to the preimage of  $S_{L^-,2-k}$  which will be denoted by  $H^+_{L,k}$ , this yields the following exact sequence.

$$0 \to \mathcal{M}_{L,k}^! \to H_{L,k}^+ \xrightarrow{\xi_k} \mathcal{S}_{L^-,2-k} \to 0.$$

In the following, fix k = m/2 and compare [BF04, Thm. 6.1 p. 77] to verify that the following equality is true for all  $f \in H^+_{L^-, 2-k}$ :

$$\langle \Theta_L(\tau, z; \varphi_{\mathrm{KM}}), \xi_{2-k}(f) \rangle_{L,k} - a^+(0,0)\Omega = dd^c \Phi(z,f),$$

where  $\Omega$  is the negative of a Kähler form considered in (5.23),  $a^+(0,0)$  denotes the constant term of the Fourier expansion of f and the expression  $\Phi(z, f)$  might require regularisation as in [Bru02]. Hence, for a cusp form  $g \in S_{L,k}$ , select a weak harmonic Maass form  $f \in H^+_{L^-,2-k}$  with vanishing constant Fourier coefficient such that  $\xi_{2-k}(f) = g$ , inducing a linear map

$$\Lambda(g,z) = dd^c \Phi(z,f) \tag{7.12}$$

that agrees with the Kudla–Millson lift on  $S_{L,k}$ . Based on this construction a subspace of the kernel of the linear lift  $\Lambda_{\rm KM}$  on  $S_{L,k}$  may be specified. Namely, let

$$N_{L^-,2-k} = \{ f \in H_{L^-,2-k} \mid Z(f) = 0 \in \operatorname{Div}(Y_L) \otimes \mathbb{C} \},\$$

where Z(f) denotes the divisor of f on  $Y_L$ . Then by [Bru02, Thm. 4.23 p. 114], the space  $\xi(N_{L^-,2-k})$  is contained in the kernel of  $\Lambda_{\rm KM}$  and Bruinier defines  $S_{L,k}^+$  to be the orthogonal complement of that subspace in  $S_{L,k}$ . Moreover, he outlines the characteristics of the spaces  $N_{L^-,2-k}$  and  $S_{L,k}^+$ , including their dependence on the lattice L and not merely on the discriminant form L'/L, that they are stable under  $O^+(L)$ , and that  $N_{L^-,2-k}$  vanishes, provided L splits a hyperbolic plane which implies in particular  $S_{L,k}^+ = S_{L,k}$ . Further, the constant term in the Fourier expansion of an element in  $N_{L^-,2-k}$  vanishes automatically.

The relationship (7.12) is then utilised to prove the following alternative characterisation of a converse theorem for Borcherds products. For that purpose, write  $\Lambda_{\rm KM}^+$  for the restriction of  $\Lambda$  to  $\mathcal{S}_{L,k}^+$ . **Theorem 7.13** ([Bru14, Thm 4.2 p. 330]). Suppose that  $m^+ \ge 2$  and that  $m^+$  is greater than the Witt rank of V. The following are equivalent:

- i) The map  $\Lambda_{\mathrm{KM}}^+ : \mathcal{S}_{L,k}^+ \to \mathcal{H}^{(1,1)}(Y_L)$  is injective.
- ii) Every meromorphic modular form F with respect to  $\Gamma(L)$  whose divisor is a linear combination of special divisors as in (7.3) is (up to a nonzero constant factor) the Borcherds lift  $\Psi(z, f)$  of a weakly holomorphic modular form  $f \in \mathcal{M}^!_{L^-, 2-k}$  with integral principal part.

The proof of the first implication is based on a weak converse theorem for Borcherds products of Bruinier, stating that there is a desired preimage  $f \in H_{L^-,2-k}$  in the space of weak harmonic Maass forms such that  $\Lambda_{\text{KM}}(\xi(f)) = dd^c \Phi(z, f) = 0$ . For the sake of convenience, assume that the lattice L splits a hyperbolic plane. Then, if  $\Lambda_{\text{KM}}^+$  is injective, we must have  $\xi(f) = 0$ , i.e. f is in the kernel of the operator  $\xi$ . This kernel, however, agrees with the space  $\mathcal{M}_{L^-,2-k}^!$  by (7.11), meaning f was, in fact, a weakly holomorphic modular form. In conclusion, a converse theorem for Borcherds products is derived as demanded in [Bor95, Problem 10] and [Bor98, Problem 16.10].

# 7.3 Injectivity of a theta lift

Let (L, q) be an even  $\mathbb{Z}$  lattice of signature  $(m^+, 2)$  and rank m and set k := m/2. In this section we will prove the injectivity of the Kudla–Millson lift introduced in Definition 7.6:

$$\Lambda_{\mathrm{KM}}: \mathcal{S}_{L,k} \to \mathcal{H}^{(1,1)}(Y_L).$$

First, note that the lift  $\Lambda_{\text{KM}}$  is linear, hence it suffices to prove that its kernel is trivial. As described in the introduction, the idea is for  $f \in S_{L,k}$  to integrate the form  $\Lambda_{\text{KM}}(f)$ over special cycles of  $Y_L$ . If it is assumed that f is annihilated by the Kudla–Millson lift, then these cycle integrals must vanish. The nature of the special cycles will allow us to extract information about the Fourier coefficients of the initial form f. Subsequently, we shall conclude that they vanish without exception, i.e. f = 0, proving the injectivity. In the following, abbreviate  $\mathcal{L} = L'/L$  and denote the Fourier expansion of f by

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n\tau) \mathbf{e}_{\lambda}.$$
(7.13)

### 7.3.1 Cycle integrals of Kudla–Millson lifts

The first major step is to prove that integrals of a target form  $\Lambda_{\text{KM}}(f)$  of the Kudla–Millson lift over certain divisors reduce to special *L*-values of symmetric square type *L*-functions of *f*. These divisors have been presented in Subsection 4.2.5 in (4.33) and we briefly recall their shape. Let  $\ell \in L$  have positive norm,  $\mathbb{D}$  denote the Grassmannian of *L* and  $\Gamma(L)$ denote the discriminant kernel in O(L). If  $\Gamma(L)_{\ell}$  denotes the stabiliser of  $\ell$  in  $\Gamma(L)$ , then

$$\mathcal{Z}(\ell) \coloneqq \Gamma(L)_{\ell} \setminus \mathbb{D}_{\ell} \to \Gamma(L) \setminus \mathbb{D} = Y_L$$

defines a (in general relative) cycle of  $Y_L$ , where we have set  $\mathbb{D}_{\ell} = \{x \in \mathbb{D} \mid x \subseteq \ell^{\perp}\}$ . With the application of Kudla's geometric example [Kud03, Sec. 4.3 p. 328] in mind, consider

$$\int_{\mathcal{Z}(\ell)} \int_{\mathcal{F}} \langle f(\tau), \Theta_L(\tau, z; \varphi_{\rm KM}) \rangle v^k \frac{\mathrm{d}u \, \mathrm{d}v}{v^2} \wedge \Omega^{m^+ - 2}, \tag{7.14}$$

where  $\Omega$  is the negative of a Kähler form on  $Y_L$  as in Definition 5.23 and  $\mathcal{F} = Mp_2(\mathbb{Z}) \setminus \mathbb{H}$ . In order to apply Kudla's observation [Kud03, Prop. 4.17 p. 332], several reduction steps are performed. To this end, recall that  $L_{\ell} \coloneqq L \cap \ell^{\perp}$  is a lattice of signature  $(m^+ - 1, 2)$ , such that  $M \coloneqq \mathbb{Z}\ell \oplus L_{\ell} \leq L$  defines a sublattice of finite index. If  $O(V)_{\ell}$  denotes the stabiliser of  $\ell$ , we find

$$\Gamma(L_{\ell}) \simeq \Gamma(M) \cap \mathcal{O}(V)_{\ell} \le \Gamma(L) \cap \mathcal{O}(V)_{\ell} = \Gamma(L)_{\ell}.$$

Here, we have used that the discriminant kernel is inclusion preserving (cf. Remark 4.50). Further, an application of Lemma A.8 with the choices  $G = \Gamma(L), H = \Gamma(M)$ , and  $K = \Gamma(L)_{\ell}$  necessitates  $[\Gamma(L)_{\ell} : \Gamma(L_{\ell})] \leq [\Gamma(L) : \Gamma(M)]$ . The latter index is certainly finite, and so is the multiplicity  $C(\ell)$  of the covering

$$\mathcal{Y}(\ell) \coloneqq \Gamma(L_{\ell}) \backslash \mathbb{D}_{\ell} \to \Gamma(L)_{\ell} \backslash \mathbb{D} = \mathcal{Z}(\ell).$$

Since the integrand in (7.14) is invariant under pullbacks from  $\Gamma(L)$  (cf. Definition 5.23 and (5.23)), we may rewrite the integral as an integral over  $\mathcal{Y}(\ell)$ , provided we adjust for the prefactor  $C(\ell)^{-1}$ .

We will begin the computation in the following and note that convergence will be verified backwards at a later stage. The first aim is to separate part of the integrand that belongs to the cycle  $\mathcal{Z}(\ell)$  in order to apply the Siegel–Weil formula. In Subsection 3.3.2 a lift  $\uparrow_L^M : S_{L,k} \to S_{M,k}$  has been described that behaves well with respect to the respective scalar products as well as theta functions. In fact, by Proposition 3.36 the operator  $\uparrow_L^M$  fulfils the following relation:

$$\langle f(\tau), \Theta_L(\tau, z; \varphi_{\mathrm{KM}}) \rangle = \langle \uparrow_L^M f(\tau), \Theta_M(\tau, z; \varphi_{\mathrm{KM}}) \rangle.$$

Note that  $\Theta_M(\tau, z; \varphi_{\text{KM}})$  fulfils the conditions of Remark 3.34 on  $\mathbb{D}_{\ell}$  (cf. [Fun02, Thm. 2.1 p. 294]). As a consequence, this theta function splits as a tensor product and by abbreviating  $\tilde{f} := \uparrow_L^M f$  and writing  $\varphi_{\text{KM},L_{\ell}}$  for the Kudla–Millson Schwartz form on  $L_{\ell} \otimes \mathbb{R}$ , we obtain the following identity on  $\mathbb{D}_{\ell}$ :

$$\langle f(\tau), \Theta_L(\tau, z; \varphi_{\mathrm{KM}}) \rangle = \langle \tilde{f}(\tau), \Theta_{\mathbb{Z}\ell}(\tau, z; \varphi_{0,\ell}) \otimes \Theta_{L_\ell}(\tau, z; \varphi_{\mathrm{KM}, L_\ell}) \rangle,$$

where  $\varphi_{0,\ell}(x) = \exp(-2\pi q(x))$  is the standard Gaussian on the space  $\mathbb{R}\ell$ . Now, if we swap the order of integration in (7.14) under the assumption of absolute convergence, we may integrate over only a part of the scalar product. For that purpose, write  $\lambda_1 \in (\mathbb{Z}\ell)'/\mathbb{Z}\ell$  as well as  $\lambda_2 \in (L_\ell)'/L_\ell$  and find in the notation of Definition 3.31 that

$$\int_{\mathcal{Y}(\ell)} \int_{\mathcal{F}} \langle f(\tau), \Theta_L(\tau, z; \varphi_{\mathrm{KM}}) \rangle v^k \frac{\mathrm{d}u \, \mathrm{d}v}{v^2} \wedge \Omega^{m^+ - 2}$$
$$= \int_{\mathcal{F}} \sum_{\lambda_1, \lambda_2} \tilde{f}_{\lambda_1 \oplus \lambda_2}(\tau) \cdot \overline{\theta_{\mathbb{Z}\ell, \lambda_1}(\tau; \varphi_{0,\ell})} \int_{\mathcal{Y}(\ell)} \overline{\theta_{L_\ell, \lambda_2}(\tau, z; \varphi_{\mathrm{KM}, L_\ell})} \wedge \Omega^{m^+ - 2} v^k \frac{\mathrm{d}u \, \mathrm{d}v}{v^2}.$$

Recall that  $\Theta_{\mathbb{Z}\ell}$  has no z-dependence, since the associated Grassmannian is trivial. The inner integral over  $\mathcal{Y}(\ell)$ , however, has already been explicitly related to a standard theta integral by means of Corollary 5.26 which in turn equals a special value of an Eisenstein series by the application of the Siegel–Weil formula (cf. Theorem 5.21).<sup>2</sup> In fact, we obtain under the assumption that  $m^+ - 1$  is bigger than the Witt index of  $V_\ell = L_\ell \otimes \mathbb{Q}$  that

$$\int_{Z(\ell)} \int_{\mathcal{F}} \langle f(\tau), \Theta_L(\tau, z; \varphi_{\mathrm{KM}}) \rangle v^k \frac{\mathrm{d}u \, \mathrm{d}v}{v^2} \wedge \Omega^{m^+ - 2}$$
$$= C_L(\ell) \cdot \int_{\mathcal{F}} \sum_{\lambda_1, \lambda_2} \tilde{f}_{\lambda_1 \oplus \lambda_2} \overline{\theta_{\mathbb{Z}\ell, \lambda_1}(\tau; \varphi_{0,\ell}) \mathcal{E}(\tau, s_0; \tilde{\varphi}_{\mathrm{KM}, L_\ell} \otimes \varphi_{\lambda_2})} v^k \frac{\mathrm{d}u \, \mathrm{d}v}{v^2}$$
(7.15)

for a nonzero explicit constant<sup>3</sup>  $C_L(\ell)$ . Here, the Eisenstein series is to be understood

<sup>&</sup>lt;sup>2</sup>Compare Remark 5.7 as well as Definition 5.16 for translating Theorem 5.21 from functions on the group to functions in  $\tau \in \mathbb{H}$ . A factor of  $\text{Im}(\tau)^{-k/2}$  appears in the process on both sides.

<sup>&</sup>lt;sup>3</sup>In fact, we find that  $C_L(\ell)$  is given by the product of the prefactors on the right hand side of (5.31) in

adellically and has been presented in Definition 5.20. Further, it is holomorphic at the critical point  $s_0 = k - 3/2$  while  $\varphi_{\lambda_2}$  denotes the finite Schwartz form represented by the coset of  $\lambda_2$  (cf. Example 5.13 part d)). By (5.28), the intertwining operator maps  $\tilde{\varphi}_{\text{KM},L_{\ell}}$  to the weight k - 1/2 section  $\Phi_{\infty}^{k-1/2}$  through the principal series representation. Hence, collecting terms for  $\lambda_1, \lambda_2$  and utilising Definition 5.17, the inner sum in (7.15) may again be written as a scalar product by using the identification of  $\mathbb{C}[\mathcal{L}]$  with  $\mathcal{S}_{\mathcal{L}}$  presented in Lemma 3.9:

$$\sum_{\lambda_1,\lambda_2} \tilde{f}_{\lambda_1 \oplus \lambda_2} \overline{\theta_{\mathbb{Z}\ell,\lambda_1}(\tau;\varphi_{0,\ell}) \mathcal{E}(\tau,s_0;\tilde{\varphi}_{\mathrm{KM},L_\ell}\otimes\varphi_{\lambda_2})} = \langle \tilde{f}, \Theta_{\mathbb{Z}\ell}(\tau;\varphi_{0,\ell}) \otimes \mathcal{E}_{\hat{L}_\ell,k-1/2}(\tau,s_0) \rangle.$$

Here,  $\mathcal{E}_{\hat{L}_{\ell},k-1/2}$  is an adelic version of the vector valued Eisenstein series (cf. Definition 5.17) which has been rewritten in classical terms in Proposition 5.18. In fact, (5.21) reads for  $l \in \mathbb{Z}/2$ , representing the weight,

$$\mathcal{E}_{\hat{L}_{\ell},l}(\tau, s_0) = E_{L_{\ell},0,l}\left(\tau, \frac{s_0 + 1 - l}{2}\right),\tag{7.16}$$

where  $E_{L_{\ell},0,l}$  is described in Definition 3.37. As a consequence, selecting l = k - 1/2 yields a new expression for the initial integral in classical terms:

$$\int_{\mathcal{Z}(\ell)} \int_{\mathcal{F}} \langle f(\tau), \Theta_L(\tau, z; \varphi_{\rm KM}) \rangle v^k \frac{\mathrm{d}u \, \mathrm{d}v}{v^2} \wedge \Omega^{m^+ - 2}$$
$$= C_L(\ell) \cdot \int_{\mathcal{F}} \langle \tilde{f}(\tau), \Theta_{\mathbb{Z}\ell}(\tau; \varphi_{0,\ell}) \otimes E_{L_\ell, 0, k - 1/2}(\tau, 0) \rangle v^k \frac{\mathrm{d}u \, \mathrm{d}v}{v^2}.$$
(7.17)

Recall that by Example 3.52 part d) the integral in (7.17) converges absolutely as long as the Eisenstein series  $E_{L_{\ell},0,k-1/2}$  converges. This is definitely the case if k > 5/2 is assumed (cf. Lemma 3.38) which is equivalent to  $m^+ > 3$ , justifying all steps so far. Further, the expression in (7.17) is a Petersson scalar product type integral containing a parabolic Poincaré series as in Proposition 6.95. As such, it qualifies for unfolding, provided the resulting integral converges absolutely. By Proposition 6.95, it does so for 0 > 3 - k/2, which is true in our case as long as  $m^+ > 10$ . This bound may be marginally improved, however, a bound below 7 appears to be unfeasible. Clearly, it is possible to continue the desired computation under the assumption  $m^+ > 10$ . Nevertheless, regularising the integral appears to be more attractive.

For that purpose, recall that by Definition 3.37 the series  $E_{L_{\ell},0,k-1/2}(\tau,0)$  is a special value

Corollary 5.26 with  $C(\ell)^{-1}$ .

of a series  $E_{L_{\ell},0,k-1/2}(\tau,s)$  depending on some parameter  $s \in \mathbb{C}$  which may be proven to be normally convergent if s satisfies  $\operatorname{Re}(s) > \frac{5}{4} - k/2 = \frac{5-m}{4}$  (cf. Lemma 3.38). This implies the condition  $m^+ \geq 3$  to guarantee holomorphicity to the right of s = 0. Now, instead of considering the Eisenstein series  $E_{L_{\ell},0,k-1/2}$  for s = 0, we may investigate the integral in (7.17) for general arguments  $s \in \mathbb{C}$  (also compare the analytic continuation of  $E_{L_{\ell},0,k-1/2}$  in s by Remark 3.39). Then, unfolding is possible for choices  $s \in \mathbb{C}$  with sufficiently large real part as a byproduct of Proposition 6.95. This has been spelled out in Corollary 6.96 stating explicitly that for all s with sufficiently large real part we find

$$\int_{\mathcal{F}} \left\langle \tilde{f}, \Theta_{\mathbb{Z}\ell} \otimes E_{L_{\ell}, 0, k-1/2}(\cdot, s) \right\rangle \cdot \operatorname{Im}^{k} \frac{\mathrm{d} u \, \mathrm{d} v}{v^{2}} = \int_{\mathcal{F}_{\Gamma_{\infty}}} \left\langle \tilde{f}, \Theta_{\mathbb{Z}\ell} \otimes \mathfrak{e}_{0} \right\rangle \cdot \operatorname{Im}^{\overline{s}+k} \frac{\mathrm{d} u \, \mathrm{d} v}{v^{2}}.$$

Here,  $\mathcal{F}_{\Gamma_{\infty}}$  denotes a suitable representation of  $\Gamma_{\infty} \setminus \mathbb{H}$ . We may then proceed to manipulate the integral as it is carried out directly below Corollary 6.96 to obtain a representation as an *L*-series. This will yield the special case of Proposition 6.97 that is

$$\int_{\mathcal{F}} \left\langle \tilde{f}, \Theta_{\mathbb{Z}\ell} \otimes E_{L_{\ell}, 0, k-1/2}(\cdot, s) \right\rangle \cdot \operatorname{Im}^{k} \frac{\mathrm{d}u \, \mathrm{d}v}{v^{2}} = \frac{\Gamma(\overline{s} + k - 1)}{(4\pi)^{\overline{s} + k - 1}} \cdot \sum_{0 \neq l \in (\mathbb{Z}\ell)'} \frac{\tilde{a}(\overline{l}, \mathbf{q}(l))}{\mathbf{q}(l)^{\overline{s} + k - 1}}.$$
 (7.18)

Here,  $\tilde{a}(\lambda, n)$  denote the Fourier coefficients of  $\tilde{f}$  as in (7.13). The series on the right hand side identifies with  $L_{\mathbb{Z}\ell,0,1}(\tilde{f}, \bar{s} + k - 1)$  given in Definition 6.73. For that naive definition, the lattice  $\mathbb{Z}\ell$  was assumed to split from the initial lattice. However, the succeeding more thorough Definition 6.77 also allowed for considering splitting sublattices and Remark 6.78 concluded that

$$\sum_{0 \neq l \in (\mathbb{Z}\ell)'} \frac{\tilde{a}(\bar{l}, \mathbf{q}(l))}{\mathbf{q}(l)^{\bar{s}+k-1}} = L_{\mathbb{Z}\ell,0,1}(\tilde{f}, \bar{s}+k-1)$$
$$= L_{\mathbb{Z}\ell,0,1}(f, \bar{s}+k-1)$$
$$= \sum_{0 \neq l \in (\mathbb{Z}\ell)'} \frac{a(\bar{l}, \mathbf{q}(l))}{\mathbf{q}(l)^{\bar{s}+k-1}}.$$

To ease the notation before continuing, we may select a primitive element  $\ell_0 \in (\mathbb{Z}\ell)' \cap L'$ and write

$$L_{\mathbb{Z}\ell,0,1}(f,\overline{s}+k-1) = \sum_{0 \neq l \in (\mathbb{Z}\ell)'} \frac{a(\overline{l},q(l))}{q(l)^{\overline{s}+k-1}} = \sum_{0 \neq n \in \mathbb{Z}} \frac{a(n\overline{\ell_0},n^2 q(\ell_0))}{q(n\ell_0)^{\overline{s}+k-1}}.$$

Notice that the Fourier coefficients of cusp forms have a symmetry property enforced by

the action of the generator  $\overline{Z} \in Mp_2(\mathbb{Z})$  of the centre (cf. Remark 6.74), yielding that the series vanishes unless  $2k \equiv sig(L) \mod 4$ . In our case  $2k = m^+ + 2 \equiv m^+ - 2 \mod 4$ . The right side is, in fact, the signature, so that the negative and positive indices of the series contribute the same terms, yielding

$$\begin{split} L_{\mathbb{Z}\ell,0,1}(f,\overline{s}+k-1) &= \sum_{0 \neq n \in \mathbb{Z}} \frac{a(n\overline{\ell_0}, n^2 \operatorname{q}(\ell_0))}{\operatorname{q}(n\ell_0)^{\overline{s}+k-1}} \\ &= \frac{2}{\operatorname{q}(\ell_0)^{\overline{s}+k-1}} \cdot \sum_{n \in \mathbb{N}} \frac{a(n\overline{\ell_0}, n^2 \operatorname{q}(\ell_0))}{n^{2(\overline{s}+k-1)}} \\ &= \frac{2}{\operatorname{q}(\ell_0)^{\overline{s}+k-1}} \cdot L_{\ell_0}(f, 2(\overline{s}+k-1)), \end{split}$$

where we use the notation of Definition 6.79 in the last line. By utilising the asymptotic growth of the Fourier coefficients  $a(\lambda, n)$ , absolute convergence of  $L_{\ell_0}(f, 2(\overline{s} + k - 1))$  may be proven in a right half plane. This is carried out in Lemma 6.80, proving that absolute convergence is guaranteed for  $2 \cdot \text{Re}(s) > 3 - k - 2\sigma$  with some positive  $\sigma$  depending on the asymptotic behaviour of the progression of Fourier coefficients. Concrete choices of  $\sigma$ have been derived in Corollary 3.87 based on bounds by [Ran39b], [BH08], and [Wai18] and collected in Table 3.1. The currently relevant cases are also contained in Lemma 6.80 and read

$$\sigma = \begin{cases} 1/2, & 2 \mid \operatorname{sig}(L), \\ 1/4, & 2 \nmid \operatorname{sig}(L). \end{cases}$$

If we were to insert the critical point s = 0, the conditions read

$$m^{+} > 4(1 - \sigma) = \begin{cases} 2, & 2 \mid \operatorname{sig}(L), \\ 3, & 2 \nmid \operatorname{sig}(L). \end{cases}$$
(7.19)

Assuming (7.19), we terminate the manipulation of the right hand side of (7.18) and apply our findings to (7.17) in order to infer the following Theorem.

**Theorem 7.14.** Let (L, q) be an even lattice of signature  $(m^+, 2)$ , set  $k := 1 + \frac{m^+}{2}$ , let  $f \in S_{L,k}$  and select some  $\ell \in L$  of positive norm. If  $m^+ > 3$ , then

$$\int_{\mathcal{Z}(\ell)} \Lambda_{\rm KM}(f) \wedge \Omega^{m^+ - 2} = 2C_L(\ell) \frac{\Gamma(k - 1)}{(4\pi \,\mathrm{q}(\ell_0))^{k - 1}} \cdot L_{\ell_0}(f, m^+), \tag{7.20}$$

where  $\Lambda_{\rm KM}$  is the Kudla-Millson lift,  $\Omega$  the negative of a Kähler form presented in

Definition 5.23,  $\mathcal{Z}(\ell)$  is the special divisor in (4.33),  $C_L(\ell)$  is a nonzero explicit constant,  $\ell_0$  is some primitive element in  $\mathbb{Q}\ell \cap L'$ , and  $L_{\ell_0}(f,s)$  denotes the symmetric square type *L*-series from Definition 6.79.

As a consequence, we conclude that a family of special L-values vanishes, if the associated cusp form is annihilated by the Kudla–Millson lift.

**Corollary 7.15.** In the notation of Theorem 7.14 we find that for any  $f \in \text{ker}(\Lambda_{\text{KM}})$  and all primitive  $\ell_0 \in L'$  of positive norm the following L-value vanishes

$$L_{\ell_0}(f, m^+) \stackrel{\text{def}}{=} \sum_{n \in \mathbb{N}} \frac{a(n\overline{\ell_0}, n^2 q(\ell_0))}{n^{m^+}} = 0.$$
(7.21)

The corollary above will be employed to derive an injectivity result.

### 7.3.2 Injectivity for a hyperbolic split

The aim of this subsection is to prove the following theorem.

**Theorem 7.16.** Let (L, q) be an even lattice of signature  $(m^+, 2)$  with  $m^+ > 3$ . Assume that L splits a hyperbolic plane. Then the Kudla–Millson lift  $\Lambda_{\text{KM}} : S_{L,1+m^+/2} \to \mathcal{H}^{(1,1)}(Y_L)$  is injective.

We begin by recalling that the Kudla–Millson lift associates to an elliptic cusp form  $f \in S_{L,k}$  a form

$$\Lambda_{\mathrm{KM}}(f) = \int_{\mathrm{Mp}_2(\mathbb{Z}) \setminus \mathbb{H}} \langle f(\tau), \Theta(\tau, z; \varphi_{\mathrm{KM}}) \rangle \frac{\mathrm{d} u \, \mathrm{d} v}{v^2} \in \mathcal{H}^{(1,1)}(Y_L).$$

Before continuing, we will sketch the idea of the proof of Theorem 7.16. By Corollary 7.15, the vanishing of  $\Lambda_{\text{KM}}(f)$  for some  $f \in \mathcal{S}_{L,k}$  implies the vanishing of special *L*-values associated to primitive positive  $\ell_0 \in L'$ . Briefly recall that primitivity derived from having to sum over all coefficients whose indices are generated by a rational line in *V*. Under the hypothesis of a hyperbolic split of *L*, we may eliminate the primitivity assumption on  $\ell_0$ . With that advantage, the series  $L_{\ell_0}(f, m^+)$  may be dissected into vanishing subseries, guaranteeing the vanishing of all of its coefficients.

There are two challenges to be addressed in order to prove the vanishing of the initial form f based on the vanishing of the special *L*-values in (7.21) which contain Fourier coefficients of f. The first is, that the only Fourier coefficients  $a(\lambda, n)$  of f ever appearing with the chosen method are of the form  $a(\overline{\ell}, q(\ell))$  for some  $\ell \in L'$  of positive norm; the second is that the relations involve infinitely many of these coefficients – the system of equations does not appear to be determined. While the second issue may be addressed by Hecke theory, the first is solely a property of the lattice L.

By assuming that L splits off a hyperbolic plane, we may eliminate the first problem. In fact, in this case, Lemma 3.68 states that any possible index  $(\lambda, n)$  of a Fourier coefficient of  $f \in S_{L,k}$  may be represented by some primitive (!) vector  $\ell_0 \in L'$  in the sense that  $(\lambda, n) = (\overline{\ell_0}, q(\ell_0))$ .<sup>4</sup> However, the *L*-series  $L_{\ell}(f, s)$  depends solely on the pair  $(\overline{\ell}, q(\ell))$ , so that in conjunction with Corollary 7.15 we immediately conclude the following assertion.

**Corollary 7.17.** Assume (L,q) is an even  $\mathbb{Z}$  lattice of signature  $(m^+,2)$  that splits a hyperbolic plane and set  $k \coloneqq 1 + m^+/2$ . If  $m^+ > 3$  and  $f \in S_{L,k}$  is annihilated by  $\Lambda_{\text{KM}}$ , then for any  $\ell \in L'$  of positive norm the following special L-value vanishes:

$$L_{\ell}(f, m^{+}) \stackrel{\text{def}}{=} \sum_{n \in \mathbb{N}} \frac{a(n\overline{\ell}, \mathbf{q}(n\ell))}{n^{m^{+}}} = 0.$$

As we will see, this also eliminates the second problem in proving the main theorem without resorting to Hecke theory by utilising an inclusion–exclusion trick.

**Lemma 7.18.** Assume L splits a hyperbolic plane and  $f \in S_{L,k}$  for  $m^+ > 3$  is annihilated by  $\Lambda_{\text{KM}}$ . Let  $N \in \mathbb{N}$  be a natural number and  $\ell \in L'$  be of positive norm. Then

$$L_{\ell}^{N}(f,m^{+}) \stackrel{\text{def}}{=} \sum_{\substack{n \in \mathbb{N} \\ \gcd(n,N)=1}} \frac{a(n\overline{\ell},\mathbf{q}(n\ell))}{n^{m^{+}}} = 0.$$

*Proof*: Note that it suffices to consider square free N. We will prove the assertion by induction on the number of distinct prime divisors of N. The case N = 1 is identical to Corollary 7.17. Assume that the statement is true for some  $N \in \mathbb{N}$ . Select a prime number  $p \nmid N$  and note that by the assumption  $m^+ > 3$  absolute convergence guarantees

$$\sum_{\substack{n \in \mathbb{N} \\ \gcd(n,pN)=1}} \frac{a(n\overline{\ell}, \mathbf{q}(n\ell))}{n^{m^+}} = \sum_{\substack{n \in \mathbb{N} \\ \gcd(n,N)=1}} \frac{a(n\overline{\ell}, \mathbf{q}(n\ell))}{n^{m^+}} - \frac{1}{p^{m^+}} \sum_{\substack{n \in \mathbb{N} \\ \gcd(n,N)=1}} \frac{a(np\overline{\ell}, \mathbf{q}(np\ell))}{n^{m^+}}$$
$$= L_{\ell}^N(f, m^+) - \frac{1}{p^{m^+}} L_{p\ell}^N(f, m^+).$$

<sup>&</sup>lt;sup>4</sup>Note that the proof of the Lemma in question may be understood without learning about the notation of the respective subsection.

Now the right hand side vanishes by our induction hypothesis, completing the proof.  $\Box$ 

The lemma above is the last ingredient for the proof of the main theorem.

Proof of Theorem 7.16: Assume there was a non-trivial form  $f \in \ker(\Lambda_{\text{KM}})$  with Fourier expansion

$$f(\tau) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathbf{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n\tau) \mathfrak{e}_{\lambda}.$$

Fix a pair  $(\lambda, n)$  for which  $a(\lambda, n) \neq 0$ . By Lemma 3.68, the hyperbolic split guarantees the existence of some  $\ell \in L'$  of positive norm with  $(\bar{\ell}, q(\ell)) = (\lambda, n)$ . Further, the assumption of  $m^+ > 3$  implies that  $L_{\ell}(f, m^+) = \sum_{d \in \mathbb{N}} a(d\bar{\ell}, d^2 q(\ell))/d^{m^+}$  converges absolutely. Hence, there must be some number  $M \in \mathbb{N}$  such that

$$\sum_{\substack{d\in\mathbb{N}\\d>M}}^{\infty} \frac{|a(d\overline{\ell}, d^2 \mathbf{q}(\ell))|}{d^{m^+}} < |a(\overline{\ell}, \mathbf{q}(\ell))|.$$

Consequently, selecting  $N \coloneqq M!$  results in

$$L_{\ell}^{N}(f, m^{+}) = a(\bar{\ell}, q(\ell)) + \sum_{\substack{1 < d \in \mathbb{N} \\ \gcd(d, N) = 1}} \frac{a(d\bar{\ell}, d^{2}q(\ell))}{d^{m^{+}}} \neq 0.$$
(7.22)

However, f was assumed to be annihilated by  $\Lambda_{\rm KM}$ , implying by Lemma 7.18 that the left hand side of (7.22) vanishes – a contradiction! As a consequence, the assumption of a non-trivial element in the kernel of  $\Lambda_{\rm KM}$  must be incorrect i.e. the Kudla–Millson lift is injective.

An application of Theorem 7.13 yields a converse theorem for Borcherds products (also compare Theorem 7.25).

**Theorem 7.19.** Assume that  $L \simeq K \oplus H$  for some lattice K of signature  $(m^+ - 1, 1)$ with  $m^+ > 3$ . Then every meromorphic modular form F for  $\Gamma(L)$  whose divisor is a linear combination of special divisors  $Z(\mu, n)$  as in (4.31) is (up to a nonzero constant factor) the Borcherds lift  $\Psi(z, f)$  of some weakly holomorphic modular form  $f \in \mathcal{M}^!_{L^-, 1-m^+/2}$ .

**Remark 7.20.** We note that the essential assumption of a hyperbolic split in Theorem 7.16 may not be omitted. This has been shown by [Bru14, Sec. 6.1 p. 333]. Also, the theorem cannot be proven for  $m^+ = 1$  – cf. Theorem 7.11 iii). The case of  $m^+ = 3$  is discussed in

the following, while the case of  $m^+ = 2$  remains out of reach for a single hyperbolic split with the current method.

### The case of $m^+ = 3$

The case of  $m^+ = 3$ , which was previously excluded for the sake of convenience, may still be treated by the same procedure. A careful analysis of the proof reveals that the majority of arguments remains intact if vanishing of the *L*-value  $L_{\ell}(f, m^+)$  is replaced by convergence  $L_{\ell}(f, s) \to 0$  for  $s \to m^+$ . This approach is universally applicable for Witt rank 1 of the lattice *L* which is precisely the case that remains unproven. The crucial additional element for such an advancement is Theorem 3.82, guaranteeing that the subseries  $L_{\ell}^N(f, m^+)$  in fact converges absolutely, provided lev $(L) \mid N$ . Indeed, the sole issue with the case  $m^+ = 3$  in the proof of Theorem 7.16 is convergence.

More explicitly, in order to apply the Siegel–Weil formula to insert the Eisenstein series  $\mathcal{E}$ in (7.15) and obtain holomorphicity at  $s_0 = k - 1/2$ , Weil's convergence criterion is applied. This criterion fails, unless  $L \cap \ell^{\perp}$  has Witt rank at most 1 which is guaranteed if the initial lattice L is assumed to have Witt rank 1. Continuing the proof under this assumption, yields that (7.18) equals

$$2\frac{\Gamma(\overline{s}+k-1)}{(q(\ell_0)4\pi)^{\overline{s}+k-1}} \cdot L_{\ell_0}(f, 2(\overline{s}+k-1))$$

for all  $s \in \mathbb{C}$  in a right half plane, where the *L*-series converges absolutely and is holomorphic in  $\overline{s}$ . This is the case for  $\operatorname{Re}(s) > 0$ . However, the left hand side of (7.18) is holomorphic in  $\overline{s}$  at  $s = m^+$  by Weil's convergence criterion (cf. Theorem 5.21 and Corollary 3.54). Hence, the right hand side of (7.18) can be holomorphically continued to  $s = m^+$ . As a consequence, we obtain the following version of Corollary 7.15.

**Corollary 7.21.** Let (L,q) be an even lattice of signature (3,2), set  $k := \frac{5}{2}$ , let  $f \in S_{L,k}$ and select some  $\ell \in L$  of positive norm such that  $L \cap \ell^{\perp}$  has Witt rank at most 1. Then

$$\int_{\mathcal{Z}(\ell)} \Lambda_{\rm KM}(f) \wedge \Omega^1 = 2C_L(\ell) \frac{\Gamma(k-1)}{(4\pi \,\mathrm{q}(\ell_0))^{k-1}} \cdot \lim_{\substack{s \to m^+ \\ \operatorname{Re}(s) > m^+}} L_{\ell_0}(f,s), \tag{7.23}$$

for some primitive element  $\ell_0 \in \mathbb{Q}\ell \cap L'$ . For the rest of the notation, compare Theorem 7.14.

**Corollary 7.22.** In the notation of Corollary 7.21, assume  $f \in \text{ker}(\Lambda_{\text{KM}})$ . Then

$$\lim_{\substack{s \to m^+ \\ \operatorname{Re}(s) > m^+}} L_{\ell_0}(f, s) \stackrel{\text{def}}{=} \lim_{\substack{s \to m^+ \\ \operatorname{Re}(s) > m^+ }} \sum_{n \in \mathbb{N}} \frac{a(n\ell_0, n^2 \operatorname{q}(\ell_0))}{n^s} = 0,$$
(7.24)

where  $a(\lambda, n)$  denote the Fourier coefficients of f as in (7.13).

The same line of reasoning that was used to prove Corollary 7.17 and Lemma 7.18 can be applied to prove their respective analogues which are subsumed in the following Corollary.

**Corollary 7.23.** Assume the lattice L of signature (3, 2) has Witt rank 1, splits a hyperbolic plane and that some  $f \in S_{L,k}$  is annihilated by  $\Lambda_{KM}$ . Let  $N \in \mathbb{N}$  be a natural number and  $\ell \in L'$  be of positive norm. Then

$$\lim_{\substack{s \to m^+ \\ \operatorname{Re}(s) > m^+}} L_{\ell}(f, s) = \lim_{\substack{s \to m^+ \\ \operatorname{Re}(s) > m^+}} L_{\ell}^N(f, s) = 0.$$

**Proposition 7.24.** Let (L,q) be an even lattice of Witt rank 1 and signature (3,2) such that L splits a hyperbolic plane. Then the lift  $\Lambda_{\text{KM}} : \mathcal{S}_{L,5/2} \to \mathcal{H}^{(1,1)}(Y_L)$  is injective.

*Proof*: The proof is almost identical to the proof of Theorem 7.16, only differing by convergence issues at the critical point s = 3 of the *L*-series involved. Assume there was a non-trivial form  $f \in \ker(\Lambda_{\text{KM}})$  with Fourier expansion

$$f( au) = \sum_{\lambda \in \mathcal{L}} \sum_{n \in \overline{\mathfrak{q}}(\lambda) + \mathbb{Z}} a(\lambda, n) \cdot e(n au) \mathfrak{e}_{\lambda}.$$

Fix a pair  $(\lambda, n)$  for which  $a(\lambda, n) \neq 0$ . By Lemma 3.68, the hyperbolic split guarantees the existence of some  $\ell \in L'$  of positive norm with  $(\bar{\ell}, q(\ell)) = (\lambda, n)$ . Further,  $m^+ = 3$ is chosen such that  $L_{\ell}(f, s) = \sum_{d \in \mathbb{N}} a(d\bar{\ell}, d^2 q(\ell))/d^s$  converges absolutely to the right of  $s = m^+$ . However, for  $N \in \mathbb{N}$  with  $\operatorname{lev}(L) \mid N$ , the series

$$L^N_{\ell}(f,s) = \sum_{\substack{d \in \mathbb{N} \\ \gcd(d,N) = 1}} \frac{a(d\overline{\ell}, d^2 \operatorname{q}(\ell))}{d^s}$$

converges absolutely at  $s = m^+$  by Corollary 3.87. Hence, there must be some number  $M \in \mathbb{N}$  such that

$$\sum_{\substack{M < d \in \mathbb{N} \\ \gcd(d,N) = 1}}^{\infty} \frac{|a(d\overline{\ell}, d^2 \operatorname{q}(\ell))|}{d^{m^+}} < |a(\overline{\ell}, \operatorname{q}(\ell))|.$$

Consequently, selecting  $N \coloneqq (M \cdot \operatorname{lev}(L))!$  results in

$$L_{\ell}^{N}(f, m^{+}) = a(\bar{\ell}, \mathbf{q}(\ell)) + \sum_{\substack{1 < d \in \mathbb{N} \\ \gcd(d, N) = 1}} \frac{a(d\bar{\ell}, d^{2} \mathbf{q}(\ell))}{d^{m^{+}}} \neq 0.$$
(7.25)

However, the form f was assumed to be annihilated by  $\Lambda_{\rm KM}$ , implying by Corollary 7.23 that

$$L_{\ell}^{N}(f,m^{+}) = \lim_{\substack{s \to m^{+} \\ \operatorname{Re}(s) > m^{+}}} L_{\ell}(f,s) = 0,$$

contradicting (7.25). As a consequence, the assumption of a non-trivial element in the kernel of  $\Lambda_{\rm KM}$  must be incorrect, i.e. the Kudla–Millson lift is injective.

This result may be employed to augment the converse theorem for Borcherds products. Recall that Theorem 7.19 already settled the case of a hyperbolic split with  $m^+ > 3$ . For the case of  $m^+ = 3$  and a hyperbolic split, we may find that the lattice L has Witt rank 1 or 2. The case of Witt rank 2 is treated in [Bru14, Thm. 1.2 p. 317] with a minor caveat, while the case of Witt rank 1 is a consequence of Proposition 7.24 in conjunction with Theorem 7.13.

**Theorem 7.25.** Assume  $m^+ > 2$  and that  $L \simeq K \oplus H$  splits a hyperbolic plane. Then there is a sublattice  $K_0 \leq K$  such that every meromorphic modular form F with respect to  $\Gamma(L)$  whose divisor is a linear combination of special divisors  $Z(\mu, n)$  as in (4.31) is (up to a nonzero constant factor) the Borcherds lift  $\Psi(z, f)$  of a weakly holomorphic modular form  $f \in \mathcal{M}^!_{K_0^-, 1-m^+/2}$ .

**Remark 7.26.** Note that the only case, where passing to a proper sublattice  $K_0 < K$  in Theorem 7.25 might be necessary is the case of  $m^+ = 3$  and L having Witt rank 2. We expect the method of proof employed in the current thesis to also work in this case by bypassing Weil's convergence criterion, rendering the transition to a sublattice  $K_0$  superfluous in each instance.

Part IV

Appendix

# A Supplementing proofs

## A.1 Lattices

In this section, we provide proofs of statements from Section 1.1 and also provide a non-trivial example of diagonalising a Gram matrix shedding some light on the O(2,2) case. We assume the reader to be familiar with the notation of Section 1.1.

**Remark A.1.** Let (L, q) be an R lattice of rank m in the quadratic space  $L \otimes_R F$ . Then choosing a Gram matrix S corresponds to defining an isometry to the lattice  $R^m \subseteq F^m$ with bilinear form

$$\mathbf{b}: R^m \times R^m \ni (x, y) \mapsto x^t S y \in F.$$

In the *p*-adic case  $R = \mathbb{Z}_p$ , we may for  $p \neq 2, \infty$  also bring the quadratic form into a standard form according to Proposition 1.11 which then results in choosing a different basis. In case of  $F = \mathbb{Q}_p$ , the additive group (F, +) carries a locally compact topology and hence possesses a Haar measure, which induces a measure on  $F^m$  and will be normalised on the unit cube  $(\mathbb{Z}_p^m \text{ in case } p < \infty)$ . Hence, it is meaningful to assign to L a volume  $\operatorname{vol}(L)$  by taking the measure of the  $\mathbb{Z}_p$  cube generated by a basis of L. Note that for a  $\mathbb{Z}$  lattice L we find the following product formula for the corresponding local lattices  $\prod_{p \leq \infty} \operatorname{vol}(L_p) = 1$ .

**Remark A.2.** (1.30) Let L be a non-degenerate  $\mathbb{Z}$  lattice.

- a) We find  $det(L)L' \leq L$  as well as  $NL' \leq L$ .
- b) We have  $|L'/L| = |\det(L)| = \operatorname{vol}(L)^2$ , in particular, L'/L is finite.
- c) Further, the following division relations are true  $N \mid 2 \det(L) \mid 2N^n$  and  $N \mid \det(L)^2$ . In case of odd level N or even rank m, we find  $N \mid \det(L)$ .
- d) The level N and the determinant det(L) have the same prime divisors.

*Proof*: a) Let S denote a Gram matrix of the lattice L. Then both inclusions follow from 1.23 once  $NS^{-1}$  is identified as an integral matrix. For a basis  $(l_i)_i$  of L and respective

[back]

dual basis  $(l'_i)_i$  we have  $S^{-1} = (b(l'_i, l'_j))_{i,j}$  and verify

$$N b(l'_i, l'_j) = N \left[ q(l'_i + l'_j) - q(l'_i) - q(l'_j) \right] \in \mathbb{Z}.$$

c) The latter implies  $\det(S) \mid \det(S) \det(NS^{-1}) = N^m$ . Since  $2 \det(L)S^{-1} \in M_m(\mathbb{Z})$  and by a) we have  $\det(L)^2 q(L') \in M_m(\mathbb{Z})$ , N must divide both by minimality.

Finally, in case of even lattice and rank m,  $\det(S)S^{-1}$  represents an even lattice. To verify this, the Leibniz formula for a minor of S reading  $\sum_{\sigma \in S_{m-1}} \operatorname{sgn}(\sigma) \cdot \prod_i s_{i,\sigma(i)}$  yields identical terms for permutations  $\sigma$  and  $\sigma^{-1}$ . In case  $\sigma^2 = \operatorname{id}$ , there is a fixed index n, as m-1 is odd, so that  $s_{n\sigma(n)} = s_{nn} \in 2\mathbb{Z}$  is part of the product. As a consequence, all linear combinations of  $NS^{-1}$  and  $\det(S)S^{-1}$  are even, yielding by definition of the level  $N \mid \det(S)$ .

b) Further, the elementary divisor Theorem for  $L \leq L'$  and a straightforward calculation yield  $\det(S) = |L'/L|^2 \det(S)^{-1}$ . The statement about the volume is completely reduced to the case  $\mathbb{Q}_p^n$ . Then Remark 1.23 and the transformation formula for integrals yield the desired result.

**Remark A.3.** Let (L, q) be an even  $\mathbb{Z}$ -lattice of signature  $(m^+, m^-)$ . If  $|\operatorname{sig}(L)| \geq 3$ , then the Witt rank  $r_{\mathbb{Q}}$  of L is maximal, i.e.  $r = \min\{m^+, m^-\}$ .

*Proof*: The definite case is clear. Assume  $V_{\mathbb{Q}}$  is indefinite, i.e.  $m^+, m^- > 0$ . Note that  $|\operatorname{sig}(L)| = |m^+ - m^-| = \max\{m^+, m^-\} - \min\{m^+, m^-\}$ , so that

$$|\operatorname{sig}(L)| \ge 3 \iff \max\{m^+, m^-\} + \min\{m^+, m^-\} \ge 3 + 2\min\{m^+, m^-\}.$$

This translates to  $\dim(V) \ge 3 + 2 \cdot r_{\mathbb{R}}$ , with  $r_{\mathbb{R}} = \min\{m^+, m^-\}$  being the Witt rank over  $\mathbb{R}$ . By applying the Theorem of Meyer (cf. Remark 1.25) inductively to an indefinite space fulfilling the last inequality, we see that it splits at least  $r_{\mathbb{R}}$  hyperbolic planes.

In the following, we present the diagonalisation of two important cases of lattices, one of which may play a key role for the injectivity of the Kudla–Millson lift in signature (2, 2).

Example A.4. Let L be a non-degenerate quadratic  $\mathbb{Z}$  lattice of rank 2 with Gram matrix S. Clearly, given  $\ell \in V = L \otimes_{\mathbb{Z}} \mathbb{Q}$  with  $q(\ell) = 0$ , it cannot be completed to an orthogonal basis, since the lattice was assumed to be non-degenerate. So assume  $\ell$  fulfilled  $q(\ell) \neq 0$ . Then  $\ell^T S \neq 0$  and by non-degeneracy, this vector has 1 dimensional kernel generated by

 $0 \neq v \coloneqq \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} S\ell$ . Now  $\ell, v$  represent an orthogonal basis of V. We find

$$v^{T}S = \ell^{T}S\begin{pmatrix} 0 & -1\\ 1 & 0 \end{pmatrix} S = \ell^{T}\det(S)\begin{pmatrix} 0 & -1\\ 1 & 0 \end{pmatrix} = \det(S)\left[\begin{pmatrix} 0 & 1\\ -1 & 0 \end{pmatrix}\ell\right]^{T}$$

This results in

$$v^T S v = \det(S) \ell^T \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} S \ell = \det(S) 2 \operatorname{q}(\ell).$$

yielding the following Gram matrix with respect to the basis  $\mathcal{B} = (\ell, v)$ 

$$S_{\mathcal{B}} = 2 \operatorname{q}(\ell) \begin{pmatrix} 1 & 0 \\ 0 & \operatorname{det}(S) \end{pmatrix}.$$

Example A.5. Assume L is a non-degenerate  $\mathbb{Z}$  lattice of rank 4, such that  $L = M \oplus N$ splits with  $\operatorname{rk}(M) = 2$ . Note that this renders M, N non-degenerate  $\mathbb{Z}$  lattices of rank 2. We would like to complete  $\ell \in V = L \otimes_{\mathbb{Z}} \mathbb{Q}$  to an orthogonal basis. The vector  $\ell$  must clearly be non-isotropic and we write  $\ell = (\ell_1, \ell_2, \ell_3, \ell_4)^T \in V$  as for the Gram matrix

$$S = S_M \oplus S_N = \begin{pmatrix} s_{11} & s_{12} & & \\ s_{12} & s_{22} & & \\ & & s_{33} & s_{34} \\ & & & s_{34} & s_{44} \end{pmatrix}$$

Without loss of generality,  $\ell|_M$  is not isotropic, meaning  $q(\ell|_M) \neq 0$ . If  $\ell = \ell|_M \oplus 0$ , meaning the restriction  $\ell|_N = 0$  vanishes identically, the situation has been solved in Example A.4, by giving an OB for M and N individually.

So assume  $\ell|_M, \ell|_N \neq 0$ . Then we may, without loss of generality assume that  $\ell_1, \ell_3 \neq 0$ . Further,  $v_1 = (\ell_1 s_{12} + \ell_2 s_{22} - \ell_1 s_{11} - \ell_2 s_{12} \ 0 \ 0)^T \in M'$  is orthogonal to  $\ell$ . In order to complete this to a basis, we have two cases to consider, namely whether  $\ell|_N$  is isotropic or not.

1. Assume  $\ell|_N$  is not isotropic. Then we may repeat the argument above to find  $v_2 = \begin{pmatrix} 0 & 0 & \ell_3 s_{34} + \ell_4 s_{44} & -\ell_3 s_{33} - \ell_4 s_{34} \end{pmatrix}^T \in N'$  to be orthogonal to  $\ell|_N$  and hence also to  $\ell$ . Now in order to find the last vector of the orthogonal basis, we need to solve the following system of equations (we use the computation of Example A.4,

again)

$$\begin{pmatrix} \ell_2 & -\ell_1 & 0 & 0\\ 0 & 0 & \ell_4 & -\ell_3\\ \ell_1 s_{11} + \ell_2 s_{12} & \ell_1 s_{12} + \ell_2 s_{22} & \ell_3 s_{33} + \ell_4 s_{34} & \ell_3 s_{34} + \ell_4 s_{44} \end{pmatrix} w = \begin{pmatrix} 0\\ 0\\ 0 \end{pmatrix}.$$

For the last line, we want to eliminate the second and fourth entry. In order to complete the step, note that  $\ell_1 s_{11} + \ell_2 s_{12} + (\ell_1 s_{12} + \ell_2 s_{22})/\ell_1 = 2 \operatorname{q}(\ell|_M)/\ell_1$ , so that we need to solve the following system:

$$\begin{pmatrix} \ell_2 & -\ell_1 & 0 & 0 \\ 0 & 0 & \ell_4 & -\ell_3 \\ q(\ell|_M)/\ell_1 & 0 & q(\ell|_N)/\ell_3 & 0 \end{pmatrix} w = \begin{pmatrix} 0 \\ 0 \\ 0 \\ 0 \end{pmatrix}.$$

We choose  $w_1 = q(\ell|_N)\ell_1$ , yielding by the first line  $w_2 = q(\ell|_N)\ell_2$ . Now the third line yields  $q(\ell|_M) q(\ell|_N) = -q(\ell|_N)w_3/\ell_3$ , which returns  $w_3 = -q(\ell|_M)\ell_3$ . Combining this fact with the second line results in  $w_4 = q(\ell|_M)\ell_4$ . This completes the construction of the orthogonal basis. We end the computation by determining the representing matrix. To this end, it suffices to reduce the computation to  $w|_M =$  $q(\ell|_N)\ell|_M$ . This yields  $w|_M^T S_M w|_M = q(\ell|_N)^2 q(\ell|_M)$ , which may be combined to  $w^T Sw = (q(\ell|_N)^2 q(\ell|_M) + q(\ell|_M)^2 q(\ell|_N)) = q(\ell|_M) q(\ell|_N) q(\ell)$ . All in all, we obtain the following orthogonal basis with representing Gram matrix

$$\mathcal{B} = \begin{pmatrix} \ell_1 & \ell_1 s_{12} + \ell_2 s_{22} & 0 & q(\ell|_N)\ell_1 \\ \ell_2 & -\ell_1 s_{11} - \ell_2 s_{12} & 0 & q(\ell|_N)\ell_2 \\ \ell_3 & 0 & \ell_3 s_{34} + \ell_4 s_{44} & -q(\ell|_M)\ell_3 \\ \ell_4 & 0 & -\ell_3 s_{33} - \ell_4 s_{34} & -q(\ell|_M)\ell_4 \end{pmatrix}$$
$$= \left(\ell, \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} S_M \ell|_M, \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} S_N \ell|_N, q(\ell|_N)\ell|_M - q(\ell|_M)\ell|_N \right),$$
$$S_{\mathcal{B}} = 2 \begin{pmatrix} q(\ell) & \\ \det(S_M) q(\ell|_M) & \\ & \det(S_N) q(\ell|_N) \\ & & q(\ell|_M) q(\ell|_N) q(\ell) \end{pmatrix}.$$

2. Assume  $\ell|_N$  is isotropic. Then the space  $\ell^{\perp}$  will clearly be isotropic as well, as it contains  $\ell|_N$ . However, we still have  $\ell$  and  $v_1$  as non-isotropic, orthogonal elements.

Also, N is isotropic, non-degenerate and 2 dimensional and as such isometric to a hyperbolic plane, meaning we may assume N = H. We may still assume  $\ell_3 \neq 0$ , forcing  $\ell_4 = 0$ . This leaves us with finding elements from the kernel of the following system that are different from  $(0, 0, \ell_3, 0)$ .

$$egin{pmatrix} \ell_2 & -\ell_1 & 0 & 0 \ 2\,\mathrm{q}(\ell|_M)/\ell_1 & 0 & 0 & \ell_3 \end{pmatrix} u = egin{pmatrix} 0 \ 0 \end{pmatrix}$$
 .

We assume  $u_4 \neq 0$ , say  $-2 q(\ell|_M)/\ell_3$ . Then the last equation yields  $2 q(\ell|_M)/\ell_1 \cdot u_1 = 2 q(\ell|_M)$ , meaning  $u_1 = \ell_1$ . In addition, the first equation yields  $\ell_1 \ell_2 = \ell_1 u_2$ , meaning  $u_2 = \ell_2$ . All in all, we find

$$v_2 \coloneqq u = \begin{pmatrix} \ell_1 & \ell_2 & u_3 & -2\operatorname{q}(\ell|_M)/\ell_3 \end{pmatrix}^T,$$

with  $u_3$  undetermined, yet. We choose  $u_3 = 0$ . Clearly, we have  $\ell|_M = u|_M$ , so that for the orthogonality condition on M, we may reproduce the same computation as above and only have to compute  $u|_N^T S_N = u|_N^T H = \left(-2\operatorname{q}(\ell|_M)/\ell_3 \quad 0\right)^T$ . This yields as a condition for the last vector w

$$\begin{pmatrix} \ell_2 & -\ell_1 & 0 & 0\\ 2\operatorname{q}(\ell|_M)/\ell_1 & 0 & 0 & \ell_3\\ 2\operatorname{q}(\ell|_M)/\ell_1 & 0 & -2\operatorname{q}(\ell|_M)/\ell_3 & 0 \end{pmatrix} w = \begin{pmatrix} 0\\ 0 \end{pmatrix}.$$

We recall that assuming  $w_4 \neq 0$  (as above for u will determine  $w_1$  and vice versa). Further,  $w_1$  determines  $w_2$  by the first line. Hence, w is of the same form as u, but with  $w_3 \neq u_3 = 0$ . In fact, we find from the last line that  $2 q(\ell|_M)/\ell_1 \cdot \ell_1 \ell_3 = 2 q(\ell|_M)w_3$ , meaning  $w_3 = \ell_3$ . We recollect the basis elements

$$\ell = \begin{pmatrix} \ell_1 \\ \ell_2 \\ \ell_3 \\ \ell_4 \end{pmatrix}, v_1 = \begin{pmatrix} \ell_1 s_{12} + \ell_2 s_{22} \\ -\ell_1 s_{11} - \ell_2 s_{12} \\ 0 \\ 0 \end{pmatrix}, v_2 = \begin{pmatrix} \ell_1 \\ \ell_2 \\ 0 \\ -2 \operatorname{q}(\ell|_M)/\ell_3 \end{pmatrix}, v_3 = \begin{pmatrix} \ell_1 \\ \ell_2 \\ \ell_3 \\ -2 \operatorname{q}(\ell|_M)/\ell_3 \end{pmatrix}.$$

Next, we need to compute the representing matrix for  $v_2, v_3$ , finding

$$v_2^T S_H v_2 = 2 \operatorname{q}(\ell|_M), \qquad v_3^T S v_3 = 2 \operatorname{q}(\ell|_M) - 4 \operatorname{q}(\ell|_M) = -2 \operatorname{q}(\ell|_M).$$

Now recall that we have assumed in this case that  $q(\ell|_N) = 0$ , rendering  $q(\ell|_M) = q(\ell)$ . Further, we had assumed  $L = M \oplus H$ , meaning  $\det(S) = \det(S_M) \cdot \det(S_H) = -\det(S_K)$ . This ultimately leads to the following representing matrix for the orthogonal basis  $\mathcal{B} := (\ell, v_1, v_2, v_3)$ :

$$S_{\mathcal{B}} = 2 q(\ell) \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & -\det(S) & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & -1 \end{pmatrix}$$

# A.2 Miscellaneous

#### Norms

We briefly recall the following definition.

**Definition A.6.** Let  $\mathbb{F} \in \{\mathbb{R}, \mathbb{C}\}$  and V be an  $\mathbb{F}$  vector space. A map  $\|\cdot\|: V \to \mathbb{R}_{\geq 0}$  is called a norm, if for  $x, y \in V$  and  $\lambda \in \mathbb{F}$  the following conditions are satisfied.

- a) It is positive definite, i.e.  $||x|| = 0 \iff x = 0$ .
- b) It fulfils the triangle inequality, i.e.  $||x + y|| \le ||x|| + ||y||$ .
- c) It is positive homogeneous of degree one, i.e.  $\|\lambda x\| = |\lambda| \cdot \|x\|$ .

We call the pair  $(V, \|\cdot\|)$  a normed space.

We also recall the following fact about normed spaces that is applied to prove convergence of certain L-series in Lemma 6.76 and Proposition 6.95.

**Remark A.7.** Let  $(V_i, \|\cdot\|_i)$  for  $i \in I$  an index set be normed spaces over a field  $\mathbb{F} \in \{\mathbb{R}, \mathbb{C}\}.$ 

- a) For  $c \in \mathbb{R}_{>0}$ , the space  $(V_i, c \cdot \| \cdot \|_i)$  is a normed space, again.
- b) The direct sum

$$\bigoplus_{i} (V_i, \|\cdot\|_i) \coloneqq \left( \oplus_i V_i, \sum_{i} \|\cdot\|_i \right)$$

is a normed space, again.

c) Let  $V_i$  be finite dimensional. Then any norm  $\|\cdot\|$  is equivalent to  $\|\cdot\|_i$ , i.e. there are constants  $C \ge c > 0$  such that

$$c \parallel \cdot \parallel \leq \parallel \cdot \parallel_i \leq C \parallel \cdot \parallel$$

#### Groups and index transfers

We will briefly recall a property of groups that is required for a key reduction to domains of integration in the proof of the main theorem of this thesis.

**Lemma A.8.** Let G be a group,  $H, K \leq G$  be subgroups. Then we find

$$[K:H\cap K] \le [G:H].$$

*Proof*: First, note that  $H \cap K \leq K$  is a subgroup. Assume, we establish a bijection between the left cosets of  $H \cap K$  in K and the left cosets of H in HK. We will call the latter [HK : H]. Then since  $HK \subseteq G$ , we find

$$[K: H \cap K] = [HK: H] \le [G: H].$$

The bijection is identical to the construction in the first isomorphism theorem of group theory [Bos18, Prop. 8 p. 18]. To be explicit, send

$$(H \cap K)k \mapsto Hk$$

Assume there are two different  $k_1, k_2 \in K$  such that  $Hk_1 = Hk_2$ . Then there is  $h \in H$ such that  $k_1k_2^{-1} = h \in H \cap K$ . In other words,  $(H \cap K)k_2 = (H \cap K)k_1$ , proving that the association is injective. However, surjectivity of the association is apparent, finishing the proof.

**Corollary A.9.** Let G be a group and,  $H, K \leq G$  be subgroups. Then we have

$$[G:H\cap K]\leq [G:H]\cdot [G:K]$$

#### Invariance of Fourier coefficients under imaginary shifts

It has been stated in (2.13) that periodic holomorphic functions on the upper half plane possess a Fourier expansion, i.e. let  $f : \mathbb{H} \to \mathbb{C}$  be holomorphic with  $f(\tau + 1) = f(\tau)$ , then there is a Fourier expansion of f:

$$f(z) = \sum_{n \in \mathbb{Z}} a(n) \cdot e^{2\pi i n \tau}$$

with coefficients

$$a(n) = \int_{[0,1]} f(\tau) \cdot e^{-2\pi i n\tau} \, \mathrm{d}\lambda(\operatorname{Re}(\tau)).$$

From the real perspective on Fourier theory, however, it is unclear why the Fourier coefficients a(n) should be independent of the chosen height  $\text{Im}(\tau)$ . The following computation is based on [FB06, III.5, p. 150].

We investigate the derivative of a(n) with respect to  $\text{Im}(\tau)$  and use the product integration rule to perform the following calculation:

$$\begin{split} \partial_{\mathrm{Im}(\tau)} a(n) &= \int_{[0,1]} \partial_{\mathrm{Im}(\tau)}(f(\tau)) \cdot e^{-2\pi i n \tau} \, \mathrm{d}\lambda(\mathrm{Re}(\tau)) + 2\pi n \int_{[0,1]} f(\tau) \cdot e^{-2\pi i n \tau} \, \mathrm{d}\lambda(\mathrm{Re}(\tau)) \\ &= f(\tau) e^{-2\pi i n \tau} \Big|_{\mathrm{Re}(\tau)=0}^{\mathrm{Re}(\tau)=1} = 0, \end{split}$$

where we have used that f is holomorphic, i.e.  $\frac{\partial f}{\operatorname{Im}(\tau)} = i \frac{\partial f}{\partial \operatorname{Re}(\tau)}$ . In consequence, a(n) is constant with respect to  $\operatorname{Im}(\tau)$ .

# A.3 Relations of Hecke operators

In this section, we present some proofs of lemmas in Subsection 6.2 and 6.3 that have been outsourced in order to improve readability.

**Lemma A.10** (Lemma 6.32). For  $r \in \mathbb{N}$  and  $k, p \nmid N$  as above we find

$$T(p^{r+1}) = T(p)T(p^r) - p \cdot T_{p \cdot \mathcal{I}} \cdot T(p^{r-1})$$

The identity remains valid if the symbol T is replaced by  ${}^{k}T$ .

Proof of Lemma 6.32: Lemma 6.25 yields the following sets of representatives for T(p)

$$R_p = \left\{ \begin{pmatrix} p & 0 \\ 0 & 1 \end{pmatrix} \right\} \bigcup \left\{ \begin{pmatrix} 1 & b \\ 0 & p \end{pmatrix} \middle| b \mod p \right\}$$

and for  $T(p^r)$ 

$$R_{p^{r}} = \left\{ \begin{pmatrix} p^{r_{1}} & b \\ 0 & p^{r_{2}} \end{pmatrix} \middle| \begin{array}{c} r_{1} + r_{2} = r, \\ b \mod p^{r_{2}} \end{array} \right\}.$$

Their product  $R_p R_{p^r}$  will be a system of representatives for  $T(p)T(p^r)$  and reads

$$\left\{ \begin{pmatrix} p^{r_1+1} & pb \\ 0 & p^{r_2} \end{pmatrix} \middle| \begin{array}{c} r_1+r_2=r, \\ b \mod p^{r_2} \end{array} \right\} \bigcup \left\{ \begin{pmatrix} p^{r_1} & b_r+b_1p^{r_2} \\ 0 & p^{r_2+1} \end{pmatrix} \middle| \begin{array}{c} r_1+r_2=r, \\ b_1 \mod p, \\ b_r \mod p^{r_2} \end{array} \right\}.$$

The second set falls only short of a set of representatives of  $T(p^{r+1})$  by the matrix  $\operatorname{diag}(p^{r+1}, 1)$ . Eliminating this element from the first set leaves

$$\left\{ p \cdot \begin{pmatrix} p^{r_1} & b \\ 0 & p^{r_2-1} \end{pmatrix} \middle| \begin{array}{c} r_1 + r_2 = r, \\ r_2 > 0, \\ b \mod p^{r_2} \end{array} \right\} = \left\{ p \cdot \begin{pmatrix} p^{r_1} & b \\ 0 & p^{r_2} \end{pmatrix} \middle| \begin{array}{c} r_1 + r_2 = r - 1, \\ b \mod p^{r_2+1} \end{array} \right\}$$

The associated Hecke element is given by

$$p \cdot T_{p \cdot \mathcal{I}} \cdot T(p^{r-1}) = \frac{p}{(p^{r+1})^{k/2-1}} {}^k T_{p \cdot \mathcal{I}} \cdot {}^k T(p^{r-1}).$$

Combining the information above, we obtain in light of Definition 6.13 that

$$T(p)T(p^r) = T(p^{r+1}) + p \cdot T_{p \cdot \mathcal{I}} \cdot T(p^{r-1}).$$

r			
н			
н			
н			

**Lemma A.11** (Lemma 6.50). We find for  $r \in \mathbb{N}$  and a prime p not dividing N that

$$\mathcal{T}(p^{r+1}) = \mathcal{T}(p)\mathcal{T}(p^r) - p \cdot T_{p\mathcal{I}} \cdot \mathcal{T}(p^{r-1}) - \delta_{r=1} \cdot T_{p\mathcal{I}}.$$

Proof of Lemma 6.50: The classes  $\Gamma(1) \setminus \Gamma(1) \begin{pmatrix} p & 0 \\ 0 & 1 \end{pmatrix} \Gamma(1)$  are represented by

$$R_p^* = \left\{ \begin{pmatrix} p & 0 \\ 0 & 1 \end{pmatrix} \right\} \cup \left\{ \begin{pmatrix} 1 & b \\ 0 & p \end{pmatrix} \middle| b \mod p \right\}.$$

•

For  $p^r$  the set of representatives reads as follows:

$$R_{p^{r}}^{*} = \left\{ \begin{pmatrix} p^{r_{1}} & b_{r} \\ 0 & p^{r_{2}} \end{pmatrix} \middle| \begin{array}{c} r_{1} + r_{2} = r \\ b_{r} \mod p^{r_{2}} \\ 0 < r_{2} < r \implies p \nmid b_{r} \end{array} \right\}$$

Their product  $R_p^*R_{p^r}^*$  has the following decomposition:

$$\left\{ \begin{pmatrix} p^{r_{1}+1} & b_{r}p \\ 0 & p^{r_{2}} \end{pmatrix} \middle| \begin{array}{c} r_{1}+r_{2}=r \\ b_{r} \mod p^{r_{2}} \\ 0 < r_{2} < r \implies p \nmid b_{r} \end{array} \right\}$$
(I)
$$\bigcup \left\{ \begin{pmatrix} p^{r_{1}} & b_{r}+b_{1}p^{r_{2}} \\ 0 & p^{r_{2}+1} \end{pmatrix} \middle| \begin{array}{c} r_{1}+r_{2}=r \\ b_{r} \mod p^{r_{2}} \\ b_{1} \mod p \\ 0 < r_{2} < r \implies p \nmid b_{r} \end{array} \right\}.$$
(II)

The second set's conditions may be reformulated as

$$\left\{ \begin{pmatrix} p^{r_{1}} & b_{r+1} \\ 0 & p^{r_{2}} \end{pmatrix} \middle| \begin{array}{l} 0 < r_{2} \\ r_{1} + r_{2} = r + 1 \\ b_{r+1} \mod p^{r_{2}} \\ 1 < r_{2} < r + 1 \Longrightarrow p \nmid b_{r+1} \\ 0 < r_{2} \\ r_{1} + r_{2} = r + 1 \\ b_{r+1} \mod p^{r_{2}} \\ 0 < r_{2} < r + 1 \Longrightarrow p \nmid b_{r+1} \end{array} \right\}$$
(II.1)  

$$\bigcup \left\{ p \begin{pmatrix} p^{r-1} & 0 \\ 0 & 1 \end{pmatrix} \right\}.$$
(II.2)

Clearly, (II.1) falls short of a set of representatives for  $R_{p^{r+1}}^*$  by diag $(p^{r+1}, 1)$  only which is taken from (I) (the case of  $r_2 = 0$ ), leaving in place of (I)

$$\left\{ p \begin{pmatrix} p^{r_1} & b_{r-1} \\ 0 & p^{r_2} \end{pmatrix} \middle| \begin{array}{c} r_1 + r_2 = r - 1 \\ b_{r-1} \mod p^{r_2 + 1} \\ r_2 < r - 1 \implies p \nmid b_{r-1} \end{array} \right\}$$

$$= \left\{ p \begin{pmatrix} p^{r-1} & b_{r-1} \\ 0 & 1 \end{pmatrix} \middle| \begin{array}{c} b_{r-1} \mod p \\ 1 < r \implies p \nmid b_{r-1} \end{array} \right\}$$
$$\bigcup \left\{ p \begin{pmatrix} p^{r_1} & b_{r-1} \\ 0 & p^{r_2} \end{pmatrix} \middle| \begin{array}{c} 0 < r_2 \\ r_1 + r_2 = r - 1 \\ b_{r-1} \mod p^{r_2 + 1} \\ 0 < r_2 < r - 1 \implies p \nmid b_{r-1} \end{array} \right\}.$$

We modify the elements via multiplication with suitable elements from  $\Gamma(1)$  to obtain

$$\Gamma(1) \simeq \bigsqcup_{i=1}^{p-1+\delta_{r=1}} \left\{ p \begin{pmatrix} p^{r-1} & 0 \\ 0 & 1 \end{pmatrix} \right\}$$
$$\bigcup \bigsqcup_{i=1}^{p} \left\{ p \begin{pmatrix} p^{r_1} & b_{r-1} \\ 0 & p^{r_2} \end{pmatrix} \middle| \begin{array}{c} 0 < r_2 \\ r_1 + r_2 = r - 1 \\ b_{r-1} \mod p^{r_2} \\ 0 < r_2 < r - 1 \implies p \nmid b_{r-1} \end{array} \right\}.$$

Adding the representative from (II.2) yields

$$= \bigcup_{i=1}^{p+\delta_{r=1}} \left\{ p \begin{pmatrix} p^{r-1} & 0 \\ 0 & 1 \end{pmatrix} \right\} \bigcup \bigcup_{i=1}^{p} \left\{ p \begin{pmatrix} p^{r_1} & b_{r-1} \\ 0 & p^{r_2} \end{pmatrix} \middle| \begin{array}{c} 0 < r_2 \\ r_1 + r_2 = r - 1 \\ b_{r-1} \mod p^{r_2} \\ 0 < r_2 < r - 1 \implies p \nmid b_{r-1} \end{array} \right\}$$
$$= \delta_{r=1} \left\{ p \begin{pmatrix} p^{r-1} & 0 \\ 0 & 1 \end{pmatrix} \right\} \bigcup \bigcup_{i=1}^{p} \left\{ p \begin{pmatrix} p^{r_1} & b_{r-1} \\ 0 & p^{r_2} \end{pmatrix} \middle| \begin{array}{c} r_1 + r_2 = r - 1 \\ b_{r-1} \mod p^{r_2} \\ 0 < r_2 < r - 1 \implies p \nmid b_{r-1} \end{array} \right\}.$$

This represents the operator

$$\delta_{r=1} \cdot T_{p\mathcal{I}} + p \cdot T_{p\mathcal{I}} \cdot \mathcal{T}(p^{r-1}).$$

In total we find

$$\mathcal{T}(p)\mathcal{T}(p^r) = \mathcal{T}(p^{r+1}) + p \cdot T_{p\mathcal{I}} \cdot \mathcal{T}(p^{r-1}) + \delta_{r=1} \cdot T_{p\mathcal{I}}$$

# A.4 Gauss sums

This is a complementary section containing elementary computations for different Gauss sums appearing throughout the thesis. The presented sums and a significant portion of the computations are already found in [Ste15] but had to be recomputed in order to detect a computational error in the expression for Fourier coefficients in Theorem 5.4 of this source.

**Definition A.12.** Let (L, q) be a quadratic  $\mathbb{Z}$  lattice of rank  $m \in \mathbb{N}$ , p a prime,  $h \in \mathbb{Z}$  with gcd(h, p) = 1 and  $n, l \in \mathbb{N}$  with  $l \ge n$ , plus  $\chi$  be a Dirichlet Character modulo n. Define

$$G_{L,p}(n,h) := \sum_{v \in L/p^n L} e\left(\frac{h}{p^n} q(v)\right), \qquad (A.1)$$

$$g_p(n,l,h) := \sum_{k \in \mathbb{Z}/p^l \mathbb{Z}} e\left(\frac{h}{p^n} k^2\right),\tag{A.2}$$

$$g_p[n,\chi,h] := \sum_{k \in \mathbb{Z}/p^n \mathbb{Z}} \chi(k) e\left(\frac{hk}{p^n}\right).$$
(A.3)

**Remark A.13.** For odd primes *p*, we have the following identity:

$$g_p(n,l,h) = \left(\frac{h}{p^n}\right)\epsilon(p^n)p^{l-n/2},$$

where

$$\epsilon(k) = \begin{cases} 1, & \text{if } k \equiv 1 \mod 4, \\ i, & \text{if } k \equiv 3 \mod 4. \end{cases}$$

For p = 2, we find

$$g_p(n,l,h) = \left(\frac{p^n}{h}\right)(1+i^h)p^{l-n/2}$$

*Proof*: We find

$$g_p(n,l,h) = \sum_{k_1 \in \mathbb{Z}/p^{l-n}\mathbb{Z}} \sum_{k_2 \in \mathbb{Z}/p^n\mathbb{Z}} e\left(\frac{h}{p^n} (k_1 p^n + k_2)^2\right)$$
$$= p^{l-n} \sum_{k_2 \in \mathbb{Z}/p^n\mathbb{Z}} e\left(\frac{h}{p^n} k_2^2\right).$$

The sum on the right is equal to

$$\sqrt{p^n} \cdot \begin{cases} \left(\frac{h}{p^n}\right) \epsilon(p^n), & \text{if } p > 2, \\ \left(\frac{p^n}{h}\right) (1+i^h), & \text{if } p = 2, \end{cases}$$

by [BEW98, 1.5 Thm. 1.5.2 p. 26] and [BEW98, 1.5 Thm. 1.5.4 p. 27].

**Remark A.14.** For p > 2 we find that the following reduction is true

$$G_{L,p}(n,h) = \left(\frac{h}{p}\right)^{R^n} \underbrace{p^{nR_0} \cdot \prod_{k=0}^{n-1} \left(\epsilon(p^{n-k})\sqrt{p^{n+k}}\right)^{n_k} \cdot \prod_{i=1}^{n_k} \left(\frac{q(v_i) \| q(v_i) \|_p}{p^n \| q(v_i) \|_p}\right)}_{=G_{L,p}(n,1)}$$

for suitable but explicit numbers  $R^n, R_0 \in \mathbb{N}_0$ .

*Proof*: By [Ste15, Lemma 4.3] there is an orthogonal decomposition of  $L/p^n L$  into  $\mathbb{Z}/p^n \mathbb{Z}$  submodules

$$\left(\bigoplus_i L_i\right) \oplus \left(\bigoplus_j M_j\right) \oplus N,$$

where  $L_i = (\mathbb{Z}/p^n\mathbb{Z})v_i$  is one dimensional with  $b(v_i, v_i) \in (\mathbb{Z}/p^n\mathbb{Z})^{\times}$ ,  $M_j = (\mathbb{Z}/p^n\mathbb{Z})v_j$ where  $b(v_j, v_j) \in p^k(\mathbb{Z}/p^n\mathbb{Z})^{\times}$  with  $1 \leq k \leq n-1$  and  $b(N, N) \subseteq p^n\mathbb{Z}$ . We may assume the  $M_j$  being sorted with respect to increasing valuation k of  $q(v_j)$ . Recall that we set

$$n_k = \#\{1 \le i \le m : \nu_p(\mathbf{b}(v_i, v_i)) = k\}, \quad R := \sum_{k=0}^{n-1} (n-k)n_k$$

in Definition 1.38. Here  $\{v_i\}$  denotes, as suggested before, a basis of L as a  $\mathbb{Z}/p^n\mathbb{Z}$  module. With these definitions, we find

$$G_{L,p}(n,h) = \sum_{(x_j)_j \in (\mathbb{Z}/p^n\mathbb{Z})^m} e\left(\sum_i \frac{h \operatorname{q}(v_j)}{p^n} x_i^2\right)$$
  
$$= \prod_{i,\operatorname{q}(v_i) \neq 0} \sum_{x_i \in \mathbb{Z}/p^n\mathbb{Z}} e\left(\frac{h \operatorname{q}(v_i) \| \operatorname{q}(v_i) \|_p}{p^n \| \operatorname{q}(v_i) \|_p} x_i^2\right) \cdot \prod_{i,\operatorname{q}(v_i)=0} p^n$$
  
$$\stackrel{*}{=} \prod_{k=0}^{n-1} \left(\frac{h}{p^{n-k}}\right)^{n_k} \cdot \prod_{0 \leq \nu_p(\operatorname{q}(v_i)) < n} \sum_{x_i \in \mathbb{Z}/p^n\mathbb{Z}} e\left(\frac{\operatorname{q}(v_i) \| \operatorname{q}(v_i) \|_p}{p^n \| \operatorname{q}(v_i) \|_p} x_i^2\right) \cdot \prod_{q(v_i) \in p^n\mathbb{Z}} p^n$$

$$= \left(\frac{h}{p}\right)^R \cdot G_{L,p}(n,1).$$

For  $\star$ , Remark A.13 has been used to obtain for  $0 \le \nu_p(\mathbf{q}(v_i)) < n$  and fixed i

$$g_{p}(n - \nu_{p}(\mathbf{q}(v_{i})), n, h \mathbf{q}(v_{i}) \| \mathbf{q}(v_{i}) \|_{p})$$

$$= \sum_{x_{i} \in \mathbb{Z}/p^{n}\mathbb{Z}} e\left(\frac{h \mathbf{q}(v_{i}) \| \mathbf{q}(v_{i}) \|_{p}}{p^{n} \| \mathbf{q}(v_{i}) \|_{p}} x_{i}^{2}\right)$$

$$= \left(\frac{h}{p^{n} \| \mathbf{q}(v_{i}) \|_{p}}\right) \left(\frac{\mathbf{q}(v_{i}) \| \mathbf{q}(v_{i}) \|_{p}}{p^{n} \| \mathbf{q}(v_{i}) \|_{p}}\right) \epsilon(p^{n - \nu_{p}(\mathbf{q}(v_{i}))}) p^{(n + \| \mathbf{q}(v_{i}) \|_{p})/2}$$

$$= \left(\frac{h}{p}\right)^{n - \nu_{p}(\mathbf{q}(v_{i}))} \cdot g_{p}(n - \nu_{p}(\mathbf{q}(v_{i})), n, \mathbf{q}(v_{i}) \| \mathbf{q}(v_{i}) \|_{p}).$$

This computation, however, yields directly

$$G_{L,p}(n,1) = p^{nR_0} \cdot \prod_{\substack{i \\ 0 \le \nu_p(\mathbf{q}(v_i)) < n}} \left( \frac{\mathbf{q}(v_i) \| \mathbf{q}(v_i) \|_p}{p^n \| \mathbf{q}(v_i) \|_p} \right) \epsilon(p^{n-\nu_p(\mathbf{q}(v_i))}) p^{(n+\| \mathbf{q}(v_i) \|_p)/2}$$
$$= p^{nR_0} \cdot \prod_{k=0}^{n-1} \left( \epsilon(p^{n-k}) \sqrt{p^{n+k}} \right)^{n_k} \cdot \prod_{i=1}^{n_k} \left( \frac{\mathbf{q}(v_i) \| \mathbf{q}(v_i) \|_p}{p^n \| \mathbf{q}(v_i) \|_p} \right)$$

for  $R_0 = \{i \mid q(v_i) \in p^n \mathbb{Z}\} = m - \sum_{k=0}^{n-1} n_k$  Recall that the elements  $v_i$  have been assumed to be sorted with respect to increasing valuation of  $q(v_i)$ 

**Remark A.15.** For  $\chi = \chi_1$  the trivial character on  $(\mathbb{Z}/p^n\mathbb{Z})^{\times}$  or  $\chi = \chi_p = \left(\frac{\cdot}{p}\right)$  the Legendre symbol and  $p \neq 2$ , we find

$$g_p[n,\chi_1,h] = \begin{cases} p^{n-1}(p-1), & \text{if } \nu_p(h) \ge n, \\ -p^{n-1}, & \text{if } \nu_p(h) = n-1, \\ 0, & \text{if } \nu_p(h) < n-1. \end{cases}$$
$$g_p[n,\chi_p,h] = \begin{cases} p^{n-1/2} \left(\frac{h/p^{n-1}}{p}\right) \epsilon_p, & \text{if } \nu_p(h) = n-1, \\ 0, & \text{if } \nu_p(h) \ne n-1. \end{cases}$$

Note that  $\chi_1(0) = \chi_1(p) = 0$  and  $\epsilon_p = 1, i$  depending on  $p \equiv 1, 3 \mod 4$ .
*Proof*: If  $\chi$  is induced by a character mod p, we find

$$g_p[n,\chi,h] = \sum_{k \in \mathbb{Z}/p^n \mathbb{Z}} \chi(k) e\left(\frac{kh}{p^n}\right)$$
$$= \sum_{k_1 \in \mathbb{Z}/p\mathbb{Z}} \chi(k_1) \sum_{k_2 \in \mathbb{Z}/p^{n-1}\mathbb{Z}} e\left(\frac{(k_1 + pk_2)h}{p^n}\right)$$
$$= \sum_{k_1 \in \mathbb{Z}/p\mathbb{Z}} \chi(k_1) e\left(\frac{k_1h}{p^n}\right) \sum_{k_2 \in \mathbb{Z}/p^{n-1}\mathbb{Z}} e\left(\frac{k_2h}{p^{n-1}}\right)$$
$$= \delta_{p^{n-1}|h} \cdot p^{n-1} \cdot \sum_{k_1 \in \mathbb{Z}/p\mathbb{Z}} \chi(k_1) e\left(\frac{k_1h}{p^n}\right).$$

Now, if  $\chi = \chi_1$  is the trivial Dirichlet character (mod p),

$$\sum_{k_1 \in \mathbb{Z}/p\mathbb{Z}} \chi_1(k_1) e\left(\frac{k_1 h}{p^n}\right) = \begin{cases} p-1, & \text{if } \nu_p(h) \ge n, \\ -1, & \text{if } \nu_p(h) = n-1 \end{cases}$$

In addition, if  $\chi_p = \left(\frac{\cdot}{p}\right)$ , we have

$$\sum_{k_1 \in \mathbb{Z}/p\mathbb{Z}} \chi_p(k_1) e\left(\frac{k_1 h}{p^n}\right) = \begin{cases} 0, & \text{if } \nu_p(h) \ge n, \\ \left(\frac{h/p^{n-1}}{p}\right) \epsilon(p)\sqrt{p}, & \text{if } \nu_p(h) = n-1. \end{cases}$$

The last assertion follows from a well known Theorem by Gauss [IK04, 3.5 Thm. 3.3 p. 49], who required years finding an adequate solution to the problem but finally came out with multiple different successful approaches.  $\Box$ 

The following Gauss sum is required in Proposition 6.54 and in the subsequent results building upon it.

**Definition A.16.** Let  $\mathcal{L}$  be a discriminant form. For  $d \in \mathbb{N}$  define

$$G_{\mathcal{L}}(d) = \sum_{\lambda \in \mathcal{L}} e\left(d\,\overline{\mathbf{q}}(\lambda)\right). \tag{A.4}$$

In [BS08], the authors compare [McG03, Lem. 4.6 p. 115] to [Bor00, Thm. 5.4 p. 329] in order to conclude the following statement.

**Remark A.17.** For  $gcd(d, lev(\mathcal{L})) = 1$  the association

$$(\mathbb{Z}/\operatorname{lev}(\mathcal{L})\mathbb{Z})^{\times} \ni d \mapsto \frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(d)}$$

defines a character.

For the following remark that is required in the proof of Corollary 6.72, recall that for a natural number  $d \in \mathbb{N}$  and a discriminant form  $\mathcal{L}$ , the symbol  $_d\mathcal{L}$  denoted the *d*-torsion of  $\mathcal{L}$ .

**Remark A.18.** Let  $\mathcal{L}$  be the discriminant form of a lattice L of rank m and assume  $gcd(d, lev(\mathcal{L})) = 1$ . Then, by [Bar03, Thm. 5.2.2] we find

$$G_{\mathcal{L}}(d) = \sqrt{|\mathcal{L}|} \sqrt{i}^{\operatorname{sgn}(\mathcal{L})} \frac{1}{d^{m/2}} \sum_{\nu \in L/dL} e\left(-\frac{1}{d} \operatorname{q}(\nu)\right).$$

In particular, the case  $d = p^n$  yields

$$G_{\mathcal{L}}(p^n) = \sqrt{|\mathcal{L}|} \sqrt{i}^{\operatorname{sgn}(\mathcal{L})} \frac{1}{p^{nm/2}} \cdot G_{L,p}(n,1).$$

As a consequence,

$$\frac{G_{\mathcal{L}}(1)}{G_{\mathcal{L}}(p)} = p^{m/2} \frac{1}{\overline{G_{L,p}(1,1)}} = \frac{p^{-m/2}}{|_p \mathcal{L}|} G_{L,p}(1,1).$$
(A.5)

*Proof*: Only the last equation has to be proven. By [Sch09, Prop. 3.8] we find the identity  $|G_{\mathcal{L}}(d)| = \sqrt{|d\mathcal{L}||\mathcal{L}|}$ . As a consequence,

$$|G_{L,p}(n,1)| = \sqrt{|p^{mn}||_p \mathcal{L}|}$$

This immediately implies the result.

Do note that explicit formulae for a more general version of  $G_{\mathcal{L}}$  are also contained in [Sch09, Thm. 3.9 p. 11].

### A.5 Whittaker functions and asymptotics

In these complementary notes, we explore convergence and boundedness properties of special functions which are required for Rankin–Selberg integrals in Subsection 6.4.3.

**Lemma A.19.** Let  $\sum_{n \in \mathbb{N}} a_n z^n$  with  $a_n \in \mathbb{C}$  be a power series that is convergent at a point  $0 \neq z \in \mathbb{C}$ . Then it is absolutely and locally uniformly convergent on the interior of the disc  $B_{|z|}(0)$  of radius |z|.

*Proof*: Recall that by the convergence of the series, it must form a Cauchy sequence and as such, there is some  $N \in \mathbb{N}$  such that  $|a_n z^n| < 1$  for all  $n \geq N$ . Now let  $z_2 \in \mathbb{C}$  fulfil  $|z_2| < |z|$ . Then

$$\sum_{n=N}^{\infty} |a_n z_2^n| = \sum_{n=N}^{\infty} |a_n z^n| |\frac{z_2}{z}|^n \le \sum_{n=N}^{\infty} |\frac{z_2}{z}|^n < \infty.$$

In fact, the last term is strictly decreasing, when decreasing  $|z_2|$ . As a consequence, we obtain uniform absolute convergence on the interior of the disc  $B_{|z|}(0)$ .

The following special functions are required for the Fourier expansion of the nonholomorphic Eisenstein series in Proposition 3.40. We follow [DAR84, Sec. 13 p. 189] and [Olv+10, Chap. 13 p. 321] as a reference for the following special functions.

**Definition A.20.** 1. For  $a, b \in \mathbb{C}$  with  $\operatorname{Re}(a) > 0$  and  $z \in \mathbb{R}$  define the integral representation of the *confluent hypergeometric function* 

$$U(a,b,z) \coloneqq \Gamma(a)^{-1} \int_0^\infty e^{-zt} t^{a-1} (1+t)^{b-a-1} \, \mathrm{d}t.$$

2. Define the Whittaker function

$$\mathcal{W}_{\kappa,\mu}(z) \coloneqq e^{-z/2} z^{1/2+\mu} U(1/2 + \mu - \kappa, 1 + 2\mu, z).$$

3. The following special version plays a role for the Fourier expansion of Eisenstein series in Proposition 3.40. For  $v \in \mathbb{R}^{\times}$  and  $k \in \mathbb{Z}/2$  representing the weight in the respective section, set

$$\mathcal{W}_s(v) \coloneqq |v|^{-k/2} \mathcal{W}_{\operatorname{sgn}(v)k/2,(1-k)/2-s}(|v|).$$

**Lemma A.21.** Given  $\delta > 0$ , the modified Whittaker function  $W_s(v)$  is bounded for any  $v \in \mathbb{R} \setminus B_{\delta}(0)$  and any  $s \in \mathbb{C}$  with  $\operatorname{Re}(s) > 1 - k$  by

$$C_s \cdot e^{-\frac{|v|}{2}(1-\varepsilon)},$$

with some arbitrarily small  $\varepsilon > 0$  and a constant  $C_s > 0$  that may be chosen locally

#### uniformly in s.

*Proof*: We write out the modified Whittaker function  $\mathcal{W}_s$  explicitly.

$$\mathcal{W}_{s}(v) = |v|^{-\frac{k}{2}} \cdot e^{-\frac{|v|}{2}} |v|^{\frac{1}{2} + \frac{1-k}{2} - s} U\left(\frac{1}{2} + \frac{1-k}{2} - s - \operatorname{sgn}(v)\frac{k}{2}, 1 + 2\frac{1-k}{2} - 2s, |v|\right) \\ = e^{-\frac{|v|}{2}} |v|^{1-\frac{k}{2} - s} \cdot U\left(1 - \frac{k}{2} - s - \operatorname{sgn}(v)\frac{k}{2}, 2 - k - 2s, |v|\right) \\ = e^{-\frac{|v|}{2}} |v|^{1-s-\frac{k}{2}} \cdot U\left(1 - s - k\delta_{v>0}, 2(1-s) - k, |v|\right).$$

Now, by the differential equation of the confluent hypergeometric functions, we may replace U(a, b, |v|) by a finite sum of such functions with  $a, b \gg 0$  with prefactors that are polynomial in a, b, and |v| (cf. [Olv+10, 13.3.14 p. 325]). Hence, it suffices to bound the above term for  $s \in \mathbb{C}$  with  $\operatorname{Re}(s) \ll 0$ . In that case, we find

$$\begin{aligned} \mathcal{W}_{s}(v) \\ &= \frac{e^{-\frac{|v|}{2}}|v|^{1-\frac{k}{2}-s}}{\Gamma(1-s-k\delta_{v>0})} \cdot \int_{0}^{\infty} e^{-|v|t} t^{-s-k\delta_{v>0}} (1+t)^{2(1-s)-k-(1-s-k\delta_{v>0})-1} dt \\ &= \frac{e^{-\frac{|v|}{2}}|v|^{1-\frac{k}{2}-s}}{\Gamma(1-s-k\delta_{v>0})} \cdot \int_{0}^{\infty} e^{-|v|t} t^{-s-k\delta_{v>0}} (1+t)^{-s-k\delta_{v<0}} dt. \end{aligned}$$

It is clear, that the integral is bounded for |v| above a fixed threshold and that the bound is uniform on vertical strips in s. Recall that the Beta functions is entire, in particular continuous. Hence, the whole expression is bounded by

$$C_s e^{-\frac{|v|}{2}(1-\varepsilon)}$$

for  $\varepsilon > 0$  arbitrarily small and some  $C_s$  that may be chosen as a constant on a sufficiently small neighbourhood.

# **B** Further background material

In this chapter, we present some more general background material that is less subject to direct referencing in the main body of the thesis. It is presented in a less formal manner and thought of as additional notes and not part of the main content.

## **B.1** Some categorical notions

#### **Restricted products**

The construction of  $\mathbb{A}$  as a locally compact Hausdorff ring is the paradigm of a natural extension of locally compact structures in this category. However, it may be generalised and is based on the observation that a product of locally compact topological spaces is again locally compact with respect to the product topology if and only if almost all of the factors are compact. We refer to [Dei10, 5.1 p. 121] for more details on the subject than we have presented below.

**Definition B.1.** Let *I* be an index set and  $X_i$  be locally compact spaces as well as  $K_i \subset X_i$  compact and open. Then

$$\prod_{i \in I} {}^{K_i} X_i = \bigcup_{\substack{E \subset I \\ \text{finite}}} \prod_{i \in E} X_i \times \prod_{i \in I \setminus E} K_i$$

together with the topology generated by the sets

$$\prod_{i \in E} U_i \text{ where } U_i \subset X_i \text{ is open and almost all } U_i \text{ are equal to} K_i$$
(B.1)

is called *restricted product* of the family  $(X_i)_{i \in I}$  with respect to  $(K_i)_{i \in I}$ .

From the above commentary it is immediate that the individual products which are being united are locally compact. However, the restricted product is a locally compact space itself. Note that since every  $K_i$  is open, the sets presented in B.1 form a basis of a topology, meaning they are closed under intersection. The following remark states some elementary properties of  $\prod_{i \in I} K_i X_i$ .

**Remark B.2.** a) For any partition of  $I = A \cup B$ , we have

$$\prod_{i\in I} {}^{K_i}X_i \simeq \left(\prod_{i\in A} {}^{K_i}X_i\right) \times \left(\prod_{i\in B} {}^{K_i}X_i\right).$$

b) The canonical injection

$$\iota: \prod_{i\in I} {}^{K_i}X_i \hookrightarrow \prod_{i\in I} X_i$$

is continuous, but it is only homeomorphic onto its image, if  $K_i = X_i$  for almost all  $i \in I$ .

c) If every space  $X_i$  is locally compact,  $\prod_{i \in I} K_i X_i$  is as well.

Both products in a) are equal as sets. The topological equivalence follows immediately by definition of the restricted product topology and the product topology itself.

In case b) the continuity of  $\iota$  is a consequence of a), by splitting the product according to every open set  $\prod_{i \in E} U_i \times \prod_{i \in I \setminus E} X_i$  in  $\prod_{i \in I} X_i$  and considering both factors separately. The homeomorphicity follows from the fact, that  $\prod_{i \in I} K_i$  is open in  $\prod_{i \in I} X_i$  if and only if  $X_i = K_i$  for almost all I.

For c) let  $(x)_{i\in I} \in \prod_{i\in I} K_i X_i$ , then there is a finite subset  $E \subset I$ , such that  $x_i \in K_i$  for  $i \in I \setminus E$ . Choose compact neighbourhoods  $U_i$  of  $x_i$  for  $i \in E$ , then  $U := \prod_{i\in E} U_i \times \prod_{i\in I\setminus E} K_i$  is, by use of a) and b), a compact neighbourhood of x.

#### **Projective limits**

Both,  $\mathbb{Z}_p$  and  $\mathbb{Z}_p^{\times}$  can be viewed as projective limits of families of finite algebraic structures, inheriting many of their properties. Hence, it might be sufficient to prove certain properties for the families of finite groups they arise from to lift these properties to the adelic structure. We will define projective limits and give  $\mathbb{Z}_p$  and  $\mathbb{Z}_p^{\times}$  as concrete realisations. Compare again [Dei10, 4.5 p. 113] for additional details.

**Definition B.3.** A partially ordered set  $(I, \leq)$  is called *directed*, if to  $a, b \in I$  there exists a common upper bound c, e.g.  $a \leq c$  and  $b \leq c$ .

Projective limits will be defined for rings explicitly, implying the same structure for groups.

**Definition B.4.** A projective system of rings is a family  $(R_i)_{i \in I}$  of rings  $R_i$  where  $(I, \leq)$  is a partially ordered directed set together with homomorphisms

$$\pi_i^j : R_j \to R_i$$

for every pair  $i, j \in I$  with  $i \leq j$ , such that

$$\pi_i^i = \mathrm{id}_{R_i}$$
 and  $\pi_i^j = \pi_i^k \circ \pi_k^j$  for  $i \le k \le j \in I$ .

The last property is called *compatibility* of the homomorphisms. The *projective limit* of the projective system  $((R_i)_{i \in I}, (\pi_i^j)_{i \leq j \in I})$  is the subring of all  $(a_i)_{i \in I} \in \prod_{i \in I} A_i$  with  $a_i = \pi_i^j(a_j)$  for all  $i \leq j \in I$  and is denoted by  $\lim_{i \in I} R_i$ . It comes with mappings  $\pi_i : \lim_{i \in I} R_i \to R_i$ such that for any Y with mappings  $\psi_i : Y \to R_i$  fulfilling the same properties, there is exactly one  $\pi : Y \to \lim_{i \in I} R_i$  such that the following diagram commutes.



The following example illustrates the concept and provides an idea of why the structure is called *projective* limit.

Example B.5. • Let  $(I, \leq) = (\mathbb{N}_0, \leq)$ , p be a prime number and  $R_i := \mathbb{Z}/p^{i+1}\mathbb{Z}$  as well as

$$\pi_i^j : R_j \to R_i, \quad x + p^{j+1}\mathbb{Z} \mapsto x + p^{i+1}\mathbb{Z}$$

be the canonical projections. Then  $((R_i)_{i \in I}, (\pi_i^j)_{i \leq j \in I})$  is a projective system and its limit is isomorphic to  $\mathbb{Z}_p = \{\sum_{i=0}^{\infty} b_i p^i \mid 0 \leq b_i < p-1 \text{ and } b_i \in \mathbb{N}\}$ . In fact the mapping

$$\phi: \mathbb{Z}_p \to \lim_{\leftarrow} R_i, \quad \sum_{i=0}^{\infty} b_k p^k \mapsto \left(\sum_{k=0}^i b_k p^k \mod p^{i+1}\right)_{i \in I}$$

is a topological ring isomorphism (note that both spaces are compact Hausdorff spaces).

• This example is used in [Ser12a, 3.1 p 15] to derive the results presented in 1.4 nd we stick to the notation of the source, denoting  $U := \mathbb{Z}_p^{\times}$  and  $U_n := 1 + p^n \mathbb{Z}_p$  for  $n \geq 1$  (here a projective system of groups is considered). Then  $U \simeq \lim_{\leftarrow} U/U_n$  with respect to the canonical projections:

$$\pi_i^j: U/U_j \to U/U_i, \quad x \cdot (1+p^j \mathbb{Z}_p) \mapsto x \cdot (1+p^i \mathbb{Z}_p).$$

Note that  $U/U_n \simeq (\mathbb{Z}/p^n\mathbb{Z})^{\times}$ , since

$$\mathbb{Z}_p^{\times} = \left\{ \sum_{i=0}^{\infty} b_i p^i \mid 0 \le b_i < p-1 \text{ and } b_i \in \mathbb{N} \text{ and } b_0 \ne 0 \right\}.$$

Then

$$\psi: \mathbb{Z}_p^{\times} \to \lim_{\leftarrow} U/U_i, \quad \sum_{i=0}^{\infty} b_k p^k \mapsto \left(\sum_{k=0}^i b_k p^k \mod p^{i+1}\right)_{i \in \mathcal{I}}$$

is the desired isomorphism, which was directly derivable from example one with  $\psi := \phi|_{\mathbb{Z}_n^{\times}}.$ 

### B.2 Representation theory in characteristic 0

Groups, as such, appear throughout all sciences. In nearly every subfield ob mathematics the author is aware of but also in information technologies, physics (symmetry groups, atomic spectra, ...), biology, chemistry, and more . This is, in fact, due to the general fashion of their notion, alternatively expressed: their poor structure. In order to investigate particular representatives, a different perspective is advantageous: a more geometric approach that results in an apparent enrichment of structure (in fact it is more of a revelation of structure). The author speaks of realising groups as symmetry groups of geometric objects, by embedding them into  $GL_n$ . This is the fundamental idea of finite representation theory and it yields, in the case of complex finite dimensional representations, a self-contained, accessible edifice of ideas. There are, however, significant obstacles when representations in infinite dimensional spaces are considered, which prevent purely algebraic approaches from finding success. In this section we discuss, in a brief fashion, representation theoretic aspects of relevance for our subsequent investigations. They are based on [Ser12b], [Ber07], [Bur98], and [Con94].

**Definition B.6.** A representation  $\rho$  of a (topological) group G [or ring] is a homomorphism  $\rho : G \to \operatorname{Aut}(V)$  for some complex vector space. Two representations  $\rho_1, \rho_2$  of G are isomorphic, if there is a G-equivariant automorphism F of V, meaning for all  $g \in G$ :

 $F\rho_1(g) = \rho_2(g)F$ . Write dim  $\rho := \dim V$ . In case a representation of a Banach (\*) algebra or a  $C^*$  algebra is considered, it will map to  $\mathscr{B}(\mathcal{H})$ , the bounded linear operators on a Hilbert space  $\mathcal{H}$ . A representation is called *unitary* if  $\rho(G)$  lies in the unitary group of  $\mathcal{H}$ .

**Remark B.7.** The notion of a G-equivariant mapping is particularly restrictive in this setting. In fact, there is no non-trivial G-equivariant mapping between two non-isomorphic representations and in the case of identical representations these are given by homotheties. This fact is known as *Schur*'s lemma.

- *Example* B.8. a) If G is a topological group, then the representations of G into  $GL_1(\mathbb{C})$  are quasi characters. Those which are unitary comprise the class of characters (cf. Definition 1.47). This indicates how restrictive the requirement of unitaricity is.
  - b) Let G be compact implying the existence of a right Haar measure  $\mu_R$ . Further, let  $\rho : G \to \mathscr{B}(\mathcal{H})$  be a representation of G. The symmetrised scalar product  $\int_G \langle \rho(g) \cdot, \rho(g) \cdot \rangle \, d\mu_R(g)$  renders  $\rho$  unitary. In particular, all representations of finite groups may be assumed to be unitary.
  - c) Let G = SO(2), then  $\mathbb{T} \simeq SO(2)$  via  $\theta \mapsto \begin{pmatrix} \cos(\theta) \sin(\theta) \\ \sin(\theta) & \cos(\theta) \end{pmatrix} = k_{\theta}$ . In this sense, the unitary representations of SO(2) are exactly the characters of  $\mathbb{T}$ , which are given by mapping  $\theta \mapsto e(k\theta)$  for  $k \in \mathbb{Z}$ .
  - d) The Weil representation (3.16) is an example of a unitary representation of  $Mp_2(\mathbb{Q}_p)$ .
  - e) Let F be a field. Then  $\operatorname{GL}_n(F)$  has a representation on the vector space of  $n \times n$ matrices  $M_n(F)$  via

$$r(M)(X) \coloneqq M \cdot X \cdot M^T.$$

In case n = 2, the matrix  $S = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix}$  is an eigenvector of r(M) for all  $M \in \operatorname{GL}_n$  with Eigenvalue det(M). This matrix plays a major role in the theory of modular forms and also this thesis (cf. (3.13), for instance).

f) Let G be a toplogical (Lie) group [possessing a Haar measure], then the right regular representation  $\rho_R$  maps into the automorphism group of C(G) ( $C^{\infty}(G)$ ) [ $L^2(G)$ ] and is given by

$$\rho_R: g \mapsto \left[ f \mapsto (h \mapsto f(hg)) \right],$$

in short  $[\rho_R(g)f](h) = f(hg)$ . Analogously, the *left regular representation* is given by  $[\rho_L(g)f](h) = f(g^{-1}h)$ . [These are unitary, provided a right, respectively left Haar measure is chosen.] This construction generalises readily to an action  $G \bigcirc X$  for proper X instead of  $G \odot G$  by multiplication, yielding a representation on  $L^2(X)$  for instance.

In fact, the left regular representation is the ultimate example for finite groups as it carries their entire representation theoretic information. In this sense it is the most important example of a representation of finite groups.

**Remark B.9.** Let  $H \leq G$  be groups and  $\sigma : H \to \operatorname{Aut}(V)$  be a representation. The *induced representation* of  $\sigma$  to G is given by

$$\operatorname{Ind}_{H}^{G}(\sigma) = K[G] \otimes_{K[H]} V.$$

As in most theories objects may be used to construct new ones in the category at hand and in this spirit there is the obvious notion of finite direct sums and tensor products of representations. However, there is another naive construction becoming relevant later.

Example B.10. Let  $\rho_G, \rho_H$  be representations of (topological) groups [rings] G, H into  $\operatorname{Aut}(V)$ ,  $\operatorname{Aut}(W)$ , and  $\varphi: V \to W$  be an isomorphism. Then  $\rho_H$  is pulled back along  $\varphi$  via pointwise conjugation to a representation  $\rho'_H$  into  $\operatorname{Aut}(V)$ . If  $\rho_G, \rho'_H$  commute pointwise, there is an obvious representation

$$\rho_{G \times H} : G \times H \to \operatorname{Aut}(V), \quad (g,h) \mapsto [v \mapsto (\rho_G(g) \circ \rho'_H(h))(v)]$$

called the product representation of  $\rho_G, \rho'_H$ .

As usual the basic building blocks of the theory are to be investigated; in this case the constituents of direct sums of representations which are exactly constructed as expected.

**Definition B.11.** A representation  $\rho : G \to \operatorname{Aut}(\mathcal{H})$  is called *irreducible* if there is no proper subspace  $0 \neq \mathcal{K} < \mathcal{H}$  that is invariant under  $\rho$ , meaning  $\rho(G)\mathcal{K} \subseteq \mathcal{K}$ . Otherwise, it is called *reducible*.

The reason for the cautious fashion of the definition is that the theory does not behave particularly well for non-unitary infinite dimensional representations. A closed subspace  $\mathcal{K}$  is obviously complementable such that  $\mathcal{K} \oplus \mathcal{K}^{\perp} = \mathcal{H}$ , but the invariance of  $\mathcal{K}$  under  $\rho$  carries only in general over if G is unitary (cf. Example B.8 b) to recall that any representation of a finite group is essentially unitary).

*Example* B.12. Let G = SO(2) and consider the trivial representation given by the action of G on  $\mathbb{C}^2$ . Then let  $U \coloneqq \begin{pmatrix} 1 & i \\ i & 1 \end{pmatrix}$  and note that  $Uk(\theta)U^{-1} = \begin{pmatrix} e^{i\theta} & 0 \\ 0 & e^{-i\theta} \end{pmatrix}$ . Meaning that conjugation with U yields an isomorphism to an obvious direct sum of characters of SO(2). The next instance depicts the classical theory of finite groups.

**Remark B.13.** Let G be finite. Then any finite dimensional representation  $\rho$  of G decomposes into a direct sum of irreducible representations. Further, let  $(\rho_i)_{i \in I}$  be a complete set of isomorphically distinct irreducible representations of G. For  $\rho$  as above, define the *character*  $\chi_{\rho}(g) \coloneqq \operatorname{tr}(\rho(g))$  of  $\rho$ . It is invariant on conjugacy classes of G. Such functions are called *class functions* and they form a vector space Cl(G) whose dimension equals the number of conjugacy classes of G. It is equipped with the scalar product

$$\langle \psi, \phi \rangle = |G|^{-1} \sum_{g \in G} \psi(g) \overline{\phi}(g).$$

Crucially,  $\langle \chi_{\rho_i}, \chi_{\rho} \rangle$  equals the number of irreducible constituents of  $\rho$  isomorphic to  $\rho_i$ , consequently  $\chi_{\rho}$  characterises  $\rho$  completely and we find  $\langle \chi_{\rho}, \chi_{\rho} \rangle = 1$  if, and only if,  $\rho$  is irreducible. Computing the character of  $\rho_L$  results in

$$\rho_L \simeq \bigoplus_{i \in I} \bigoplus_{k=1}^{\dim \rho_i} \rho_i$$

causing  $\sum_{i \in I} \dim \rho_i \rho_i(g) = \delta_{g,1}$  which in turn implies  $\sum_{i \in I} (\dim \rho_i)^2 = |G|$ . Further, applying this information to the regular representation yields that  $(\rho_i)_i$  is an ortho normal basis of Cl(G) meaning |I| equals the number of conjugacy classes of G. As a corollary one obtains that G is abelian, if, and only if, all  $\rho_i$  are one dimensional. So far everything was also applicable in characteristic different from 0.

If, in addition, characteristic 0 is imposed, another harsh constraint is forced upon the numbers dim  $\rho_i$ : these divide |G|.

## **B.3** Symplectic and metaplectic groups

The term *metaplectic group* describes a covering group of the symplectic group  $SL_2$ , which, in general, will be a central extension<sup>1</sup> of  $SL_2$  by  $\mathbb{Z}/p\mathbb{Z}$  or  $\mathbb{T}$ . In our case, however, an extension by  $\mathbb{T}$  is considered, as it splits over subgroups for which the common  $\mathbb{Z}/2\mathbb{Z}$ extension does not (cf. [Kud94]). It will be referred to as *the metaplectic group* as if there were only one and denoted  $\overline{SL}_2$ . A concrete description is provided by [KRY06, 8.5 pp. 320-336], which accords with the choice in [KY10]. The extension by  $\mathbb{Z}/2\mathbb{Z}$ , usually

<sup>&</sup>lt;sup>1</sup>Cf. [Tao10] for a brief overview of group extensions.

referred to as the metaplectic group, is derivable from  $\overline{\operatorname{SL}}_2$  as the latter determines a cocycle  $\overline{c} \in H^2(\operatorname{SL}_2, \mathbb{T})$ . Since this homology group contains only one non-trivial element of order two, the cover is essentially unique and coordinates are chosen by picking a concrete representative for  $\overline{c}$ , which shall be denoted by c and equal the one from [KRY06, 8.5 p. 320]. If chosen,  $\overline{\operatorname{SL}}_2$  is realised as the set of pairs  $(\gamma, \zeta) \in \operatorname{SL}_2 \times \mathbb{T}$  with multiplication  $(\gamma, \zeta) \cdot (\gamma', \zeta') = (\gamma \gamma', c(\gamma, \gamma')\zeta\zeta')$ . Formally, the information above, supplemented by a few remarks about subgroups and splits mentioned in Section 3.2 in the adelic case, suffices in order to deal with the metaplectic group as far as the major aim of this thesis is concerned. However, the author sought to supplement the thesis by additional details concerning the construction of metaplectic groups.

Throughout this section, the mapping  $x \mapsto 2x$  is usually assumed to be an isomorphism.

#### Abstract symplectic groups

There is a conceptional approach to the metaplectic group motivated by quantum mechanics which shall be discussed in the following paragraphs. It was introduced by André Weil [Wei64] in general in order to describe the theory of theta functions in group theoretic terms (cf. [Gel74]). The concrete case, sufficient for our investigation, is sketched concisely in [Gel76, 2.3 p. 28] or, more crisply but less conceptually in [Kud96], and the basis of its construction shall be presented in greater detail, based on [Wei64]. Starting with locally compact abelian groups G, H isomorphic to their Pontryagin duals  $G^*, H^*$ , respectively, a completion of  $\mathbb{Q}$  with respect to an absolute value function for instance, every morphism  $\alpha : G \to H$  induces a dual morphism  $\alpha^*$  via pullback:

$$\alpha^*: H^* \to G^*, \qquad h^* \mapsto h^* \circ \alpha.$$

The association  $\alpha \mapsto \alpha^*$  defines a contravariant, isomorphism preserving functor, which also conserves the modulus of the morphisms, meaning the scaling factor for pushforward measures. Furthermore, we have  $(\alpha^*)^* = \alpha$  when the canonical isomorphy  $G \simeq (G^*)^*$  is taken into account. Note that product group  $G \times G^*$  is isomorphic to its dual  $G^* \times G$  via

$$\begin{pmatrix} x \\ x^* \end{pmatrix} \mapsto \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} x \\ x^* \end{pmatrix}.$$

Similarly, endomorphisms  $\sigma$  of  $G \times G^*$  may be written in matrixform

$$\begin{pmatrix} x \\ x^* \end{pmatrix} \mapsto \begin{pmatrix} \alpha & \beta \\ \gamma & \delta \end{pmatrix} \begin{pmatrix} x \\ x^* \end{pmatrix}, \quad \text{with dual morphism } \begin{pmatrix} \alpha^* & \gamma^* \\ \beta^* & \delta^* \end{pmatrix} = \sigma^*, \tag{B.2}$$

where  $\alpha, \beta, \gamma$  and  $\delta$  denote morphisms mapping appropriately between G or  $G^*$ , respectively. The pairing

$$F: (G \times G^*) \times (G \times G^*) \to \mathbb{T}, \quad [(x, x^*), (y, y^*)] \mapsto (x, x^*) \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix} \begin{pmatrix} y \\ y^* \end{pmatrix} = y^*(x)$$
(B.3)

will become relevant later, whereas for now we are interested in the pairing

$$\kappa : (G \times G^*) \times (G \times G^*) \ni [\omega_1, \omega_1] \mapsto F(\omega_1, \omega_2) F(\omega_2, \omega_1)^{-1} = y^* x / x^* y \in \mathbb{T},$$
(B.4)

which has the *symplectic group* as its invariance group.

**Definition B.14.** The symplectic group Sp(G) of a locally compact abelian group G is defined to be the group of automorphisms of  $G \times G^*$  which fix the pairing  $\kappa$ . Hence, with the notation as above an automorphism  $\sigma$  is in Sp(G) if  $\kappa(\sigma\omega_1, \sigma\omega_2) = \kappa(\omega_1, \omega_2)$  holds for all  $\omega_1, \omega_2 \in G \times G^*$ .

By direct computation this condition for  $\sigma$  may be expressed equivalently as

$$\sigma^* \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \sigma = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}.$$

Example B.15. In the concrete case of  $G = \mathbb{Q}_p$  the group  $\mathbb{Q}_p$  may be identified with its Pontryagin dual via the exponential mapping and the entries of the endomorphism  $\sigma$  of  $G \times G^* \simeq \mathbb{Q}_p^2$  are interpreted as elements of  $\mathbb{Q}_p$ . Consequently, the invariance condition for a symplectic automorphism  $\sigma$  reduces to

$$1 = (\alpha \delta - \beta \gamma) = \det(\sigma).$$

In particular, this testifies that the symplectic group is exactly the special linear group  $SL_2(\mathbb{Q}_p)$  if G is chosen to be the additive group of a p-adic field. This closes the paragraph about realising  $SL_2$  via a symplectic pairing.

#### The Heisenberg group

The subsequent sections cover the construction of the metaplectic group, following [Wei64]. He viewed the symplectic group as a group of unitary operators on  $L^2(G)$  and utilised tools from the theory of  $C^*$  Algebras to prove the existence of a group extension, which he called *metaplectic group*. As a first step, the Heisenberg group of G is constructed on which Sp(G) acts.

For  $\omega = (u, u^*) \in G \times G^*$  define the unitary operator

$$U(\omega): L^2(G) \to L^2(G), \quad f(x) \mapsto f(x+u) \cdot u^*(x).$$

Moreover, we find

$$U(\omega_1)U(\omega_2) = F(\omega_1, \omega_2) \cdot U(\omega_1 + \omega_2).$$
(B.5)

Consequently, the mapping

$$G \times G^* \times \mathbb{T} \ni (\omega, t) \mapsto U(\omega)t$$
 (B.6)

defines a bijection onto an operator group which can be used to pullback the group law, obtaining

$$(\omega_1, t_1) \cdot (\omega_2, t_2) = (\omega_1 \omega_2, F(\omega_1, \omega_2) t_1 t_2).$$
(B.7)

This law makes  $A(G) := G \times G^* \times \mathbb{T}$  into a locally compact group with the ordinary product topology, which, exported to the corresponding operator group, equals the strong operator topology. The group A(G) equipped with multiplication given by (B.7) is called *Heisenberg group* of G, for a reason which will become clear after the following historical example.

Example B.16 (history, quantum mechanics). In 1925 Werner Heisenberg suggested solving the harmonic oscillator, a physical system, with operators P, Q on which he imposed a commutator relation  $[Q, P] = i \cdot id$ . Nearly simultaneously, Erwin Schrödinger suggested solving the very same system with a partial differential equation. Both resulted in the same physical solution, leaving the community puzzled before these seemingly different approaches.

John Von Neumann showed not only that both descriptions were mathematically equivalent,

but that Heisenberg's abstract condition determines the mathematical system up to unitary equivalence [Neu31] if irreducibility is imposed and that all representations are direct sums of this essentially unique irreducible one. His proof relied upon a one to one relation between self adjoint operators on Hilbert spaces and unitary strongly continuous 1-parameter operator groups suggested by Herrmann Weyl and published by Marshall Harvey Stone [Sto30]. Transferring the operators P, Q to the corresponding unitary operators translates the commutator relation essentially to (B.5), called the *Weyl relation*. Roughly 20 years later, 1949, George Whitelaw Mackey published a theorem analogous to that of von Neumann, but for locally compact abelian (seperable) groups [Mac49], so that the unitary irreducible representations of the Heisenberg group in the above abstract setting were understood.<sup>2</sup>

The centre of A(G) is evidently  $\mathbb{T}$ , which is closed as it is the kernel of the continuous homomorphism  $\pi_1 : A(G) \to G \times G^*$ ,  $(\omega, t) \mapsto \omega$  and therefore the diagram induces a topological group isomorphism



Recall that the concern of this section is to construct a covering group for the symplectic group Sp(G) which is a group of automorphisms of  $G \times G^*$ .

#### The symplectic group as a group of automorphisms of A(G)

The next step is to consider the group of automorphisms B(G) of A(G) and to prove that it contains a copy of  $\operatorname{Sp}(G)$ . This realises the symplectic group as a group of automorphisms of operators and ultimately leads to the existence of a projective representation of  $\operatorname{Sp}(G)$ , called *Weil representation*. Elements of B(G) which operate trivially on  $\mathbb{T}$  form a subgroup, denoted  $B_0(G)$ . An automorphism  $s \in B_0(G)$  evidently induces an automorphism  $\sigma$  on  $G \times G^*$  and may be represented as

$$s(\omega, t) = (\sigma\omega, f(\omega)t),$$

<sup>&</sup>lt;sup>2</sup>Elaborate historical notes discussing the development of unitary representation theory from this context may be found in [Mac76, Appendix p. 209 a.s.].

where  $f: G \times G^* \to \mathbb{T}$  is a continuous mapping. The condition that an automorphism of this form defines an element of B(G) is equivalent to

$$\forall \,\omega_1, \omega_2 \in G \times G^* : \quad f(\omega_1 + \omega_2) \, f(\omega_1)^{-1} \, f(\omega_2)^{-1} = F(\sigma \omega_1, \sigma \omega_2) \, F(\omega_1, \omega_2)^{-1}, \quad (B.8)$$

implying that  $B_0(G)$  may be identified with the group of pairs  $(\sigma, f)$  satisfying (B.8), with multiplication given by

$$(\sigma, f) (\sigma', f') = (\sigma \sigma', f''),$$

where  $f''(\omega) = f(\omega)f'(\sigma\omega)$ . Since the left hand side of equation (B.8) is symmetric in  $\omega_1, \omega_2$  the automorphism  $\sigma$  has to be a symplectic element, leaving  $\kappa$ , defined in (B.4), invariant. This suggests seeking a decomposition of  $B_0(G)$  by Sp(G) and in fact there is a splitting exact sequence:

$$1 \longrightarrow (G \times G^*)^* \longrightarrow B_0(G) \longrightarrow \operatorname{Sp}(G) \longrightarrow 1.$$
 (B.9)

Note that the kernel of the natural projection  $B_0(G) \to \operatorname{Sp}(G)$  is exactly  $(G \times G^*)^*$ , according to condition (B.8). As a result, it suffices to construct the splitting morphism for (B.9), which is done by computation. For the sake of lucidity the application of characters is denoted with duality brackets. Henceforth,  $f \in (G \times G^*)^*$  may be written as  $f(g, g^*) = \langle g, a^* \rangle \cdot \langle a, g^* \rangle$  with suitable  $a \in G, a^* \in G^*$ . Before constructing the splitting morphism for (B.9) explicitly, a general construction concerning generators of characters of second degree has to be carried out. For that sake let f denote a character of second degree<sup>3</sup> of G, that is a function  $f: G \to \mathbb{T}$ , such that

$$G \times G \ni (g, g') \mapsto f(g + g')f(g)^{-1}f(g')^{-1} \in \mathbb{T}$$
(B.10)

is a bilinear pairing. These functions f form a group with pointwise multiplication denoted  $X^2(G)$ . Furthermore, the associated pairing (B.10) may be represented by a morphism  $\rho: G \to G^*$ :

$$f(g+g')f(g)^{-1}f(g')^{-1} = \langle g, \rho g' \rangle.$$

Since the lefthand side is obviously symmetric,  $\rho$  has to be as well, namely  $\rho = \rho^*$  and the mapping  $\pi : f \mapsto \rho$  defines a homomorphism onto the additive group of symmetric morphisms  $\text{Hom}_s(G, G^*)$  and has kernel  $G^*$ . If doubling defines an automorphism of G,

<sup>&</sup>lt;sup>3</sup>The meaning of characters of second degree for bilinear pairings is similar to the one of quadratic forms for bilinear forms in linear Algebra.

namely  $G \ni g \mapsto 2g \in G$  is bijective<sup>4</sup> with inverse  $2^{-1}$ , then the association

$$\iota: \operatorname{Hom}_{s}(G, G^{*}) \ni \rho \mapsto (g \mapsto \langle g, \rho 2^{-1}g \rangle) \in X^{2}(G)$$
(B.11)

defines a section of  $\pi$ , making the following exact sequence split:

$$1 \longrightarrow G^* \longrightarrow X^2(G) \xrightarrow[\iota]{\pi} \operatorname{Hom}_s(G, G^*) \longrightarrow 1.$$
 (B.12)

Ultimately, the split of (B.9) may be dealt with. For this purpose define the function

$$f'(g,g^*) = f(g,g^*) \langle \gamma g^*, -\beta g \rangle,$$

where  $\sigma \in \text{Sp}(G)$  was written in matrixform as in (B.2). Then, equation (B.8) reads

$$f'(g_1 + g_2, g_1^* + g_2^*) = f'(g_1, g_1^*) f'(g_2, g_2^*) \cdot \langle g_1, \beta^* \alpha g_2 \rangle \cdot \langle \delta^* \gamma g_1^*, g_2^* \rangle$$

for f' and the definitions  $h(g)=f'(g,0), h^\ast(g^\ast)=f'(0,g^\ast)$  lead to

$$h(g_1 + g_2) = h(g_1)h(g_2) \cdot \langle g_1, \beta^* \alpha g_2 \rangle,$$
  
$$h^*(g_1^* + g_2^*) = h(g_1^*)h(g_2^*) \cdot \langle \delta^* \gamma g_1^*, g_2^* \rangle.$$

Since  $\sigma$  is assumend to be symplectic, the morphisms  $\beta^* \alpha, \delta^* \gamma$  are symmetric, hence the above relation describes the transformation property of a character of *second degree*. Thereupon, f may be written as a product of two characters of second degree and a bilinear form:

$$f(g,g^*) = h(g)h^*(g^*) \cdot \langle \gamma g^*, \beta g \rangle.$$

In view of (B.12), especially (B.11), the function f is effortlessly expressed in terms of  $\sigma$  as

$$f_{\sigma}(g,g^*) = \langle g, \beta^* \alpha 2^{-1}g \rangle \cdot \langle \delta^* \gamma 2^{-1}g^*, g^* \rangle \cdot \langle \gamma g^*, \beta g \rangle.$$

The association  $\sigma \mapsto f_{\sigma}$  defines a right split of (B.9), which proves that  $B_0(G)$  contains a copy of the symplectic group Sp(G), as claimed. Next, these are related to unitary operators on  $L^2(G)$ .

<sup>&</sup>lt;sup>4</sup>This requirement is in fact equivalent to  $G \times G^*$  having only one equivalence class of irreducible  $\sigma$ -representations [Mac64].

#### A theorem of Segal

The following theorem, initially from Segal [Seg63, Thm. 2 p. 39], and reformulated by Weil [Wei64, 10. Thm. 1 p. 157] is not proven here, but its implications relevant for our discourse are discussed.

Weil shows that every automorphism of A(G) which fixes its centre is given by an inner automorphism of the unitary group of  $L^2(G)$ . This ensures an abstract argument for the existence of the Weil representation (cf. section B.3).

**Theorem B.17.** Let  $\overline{B}_0(G)$  denote the normaliser of A(G) in the group of automorphisms of  $L^2(G)$ . Then every element of  $B_0(G)$  is a restriction of an inner automorphism of the unitary group where the latter may be chosen from  $\overline{B}_0(G)$ . Furthermore, this natural projection  $\pi_0: \overline{B}_0(G) \to B_0(G)$  has kernel  $\mathbb{T}$ , producing the exact sequence:

$$1 \longrightarrow \mathbb{T} \longrightarrow \overline{B}_0(G) \xrightarrow{\pi_0} B_0(G) \longrightarrow 1.$$
 (B.13)

The theorem above provides an operator theoretic characterisation of the automorphism group  $B_0(G)$  of A(G) which contains a copy of Sp(G). In fact, it plays a central role in constructing Weil's abstract metaplectic group Mp which is discussed briefly in the subsequent section.

#### The metaplectic extension

What has been done above may be partly transferred to the situation where G = V is a finite dimensional vector space over a local field F. In this case the theory may be linearised, replacing the Pontryagin dual  $G^*$  with the algebraic dual vector space V'. In particular, equation (B.4) reads

$$\beta: (V \times V') \times (V \times V') \ni [\omega_1, \omega_1] \mapsto F(\omega_1, \omega_2) - F(\omega_2, \omega_1) = (x, x') \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \begin{pmatrix} y \\ y' \end{pmatrix} \in F,$$
(B.14)

with F analogous to (B.3). Similarly, the product  $V \times V' \times F$  becomes a group with the law

$$(\omega_1, t_1) \cdot (\omega_2, t_2) = (\omega_1 + \omega_2, F(\omega_1, \omega_2) + t_1 + t_2), \tag{B.15}$$

denoted H(V). In case of V = F, the associated symplectic group is exactly  $SL_2(F)$ , analogous to example B.15. Inspired by the abstract treatment above, automorphisms of H(V) which are composed of an automorphism  $\sigma$  of  $V \times V'$  and a quadratic form f on  $V \times V'$  and operate via

$$(\omega, t) \mapsto (\sigma \omega, f(\omega) + t)$$

are considered. In order to define an automorphism of H(V) via this association, it is necessary and sufficient to satisfy

$$\forall \,\omega_1, \omega_2 \in V \times V': \quad f(\omega_1 + \omega_2) - f(\omega_1) - f(\omega_2) = F(\sigma\omega_1, \sigma\omega_2) - F(\omega_1, \omega_2). \quad (B.16)$$

Nearly identically to the multiplicative case, these automorphisms comprise a group with group law

$$(\sigma, f)(\sigma', f') = (\sigma\sigma', f'')$$

where  $f''(\omega) = f(\omega) + f'(\sigma\omega)$ . This group is denoted by Ps(V) and named *pseudo* symplectic group of V. Again, by symmetry of the left hand side of (B.16)  $\sigma$  has to be symplectic, where on the other hand, every symplectic element  $\sigma$  determines a quadratic form via (B.16). As a consequence,  $(\sigma, f) \to \sigma$  defines an isomorphism of Ps(V) and Sp(V). In addition, if  $F = \mathbb{Q}_p$  and  $\chi$  denotes a non-trivial character of  $\mathbb{Q}_p$ , then the relations above may be transformed to the multiplicative case by applying  $\chi$  to both sides of the equations. This may be utilised to pull fragments of the multiplicative theory back to the additive case. Explicitly, the mapping

$$H(V) \to A(G), \quad (\omega, t) \mapsto (\omega, \chi(t))$$

defines a homomorphism as well as

$$\mu : \operatorname{Ps}(V) \to B_0(G), \quad (\sigma, f) \mapsto (\sigma, \chi \circ f).$$

The latter is injective, if  $\operatorname{char}(F) \neq 2.5^{5}$  Note that the group was denoted by G, instead of V, since it is considered to be a locally compact topological group with Pontryagin dual, rather than a linear vector space. This difference in notation might appear captious but it is meant to remind the reader which point of view led to the associated results.

<sup>&</sup>lt;sup>5</sup>In fact, the mapping  $\mu$  is surjective if, and only if,  $F = \mathbb{Q}_p$  for some  $p \leq \infty$  (cf. [Rao93, p. 350]). Rao's Ps(H, F) is essentially our Ps(V) (where the second component is the induced multiplicative function) and Ps(H) in his notation is identical with our  $B_0(G)$ .

**Definition B.18.** Define the *abstract metapletic group* Mp(V) to be the subgroup of  $Ps(V) \times \overline{B}_0(V)$ , consisting of those elements  $(s, \overline{s})$ , such that  $\mu(s) = \pi_0(\overline{s})$  (cf. Figure B.3.1).

The natural projection onto Ps(V) will be denoted by  $\pi$ . Its surjectivity is a consequence of  $\pi_0$ 's (cf. Theorem B.17). The kernel of  $\pi$  consists exactly of  $\mathbb{T} < \overline{B}_0(V)$ , so that the upper horizontal sequence in the commutative Diagram B.3.1 is exact. Consequently, Mp(V) is a central extension of Sp(V).

Furthermore, the injectivity of  $\mu$  implies that the projection of Mp(V) into  $\overline{B}_0(G)$  must be injective as well. Hence, Mp(V) admits a faithful unitary continuous representation on  $L^2(G)$ , where the automorphism group is equipped with the strong operator topology. In fact, it even restricts to a representation on the subspace of Schwartz Bruhat functions S(V), since  $B_0(G)$  operates on S(V).

After the introduction of the metaplectic group, Weil constructs the adelic pendant to Mp and proves [Wei64, IV p. 194] that the cover is never trivial and always reduces to a topological central extension

$$1 \longrightarrow \mathbb{Z}_2 \longrightarrow \overline{\mathrm{Sp}}(V) \longrightarrow \mathrm{Sp}(V) \longrightarrow 1,$$

since the cohomology class it determines in  $H^2(G, \mathbb{T})$  is of order 2. If V = F, which is the case we are interested in, this double cover is unique (cf. [Gel76, 2 Prop. 2.3 p. 14]).<sup>6</sup> However, working with the extension Mp(V) is more convenient for the adelic setting due to its splitting over certain subgroups for which the  $\mathbb{Z}_2$  extension does not (cf. [Kud94]). In the process of Weils proceedings, a class of representations named after him is constructed which will be discussed in the following section.

#### The Weil representation

The most concise conceptional approaches to the relevant setting of the Weil representation the author is aware of is found in [Kud96] and the slightly less dense [Pra93]. An extended description is found in [LV80] and a slightly more explicit approach in [Rao93] (note that Rao's notation is slightly different, for instance, he writes Ps(G) for  $B_0(G)$ .). The abstract reason for the existence of a representation of Sp(V) is a theorem by von Neumann [Neu31] which was later generalised by [Mac49] (cf. Remark B.16). This essentially means that any irreducible unitary representation of the Heisenberg group A(G) on  $L^2(G)$  is unique

<sup>&</sup>lt;sup>6</sup>Further, [Wed16, 2.6 p. 34] or [Hat02, 1.3 p. 54] may be considered for a conceptual approach to coverings.



Figure B.3.1: Cf. (B.13) from Theorem B.17 for the horizontal sequence in the middle. The bottom sequence is in accord with the right split of (B.9) discussed in Section B.3. By injectivity of  $\mu$ , the projection of Mp(V) into  $\overline{B}_0(G)$  must be injective as well.

up to unitary equivalence<sup>7</sup>. In fact, if (B.6) defines a unitary representation, then for any element  $\sigma \in \text{Sp}(G) = \text{Sp}(V)$ 

$$A(G) = G \times G^* \times \mathbb{T} \ni (\omega, t) \mapsto U(\sigma\omega)t \tag{B.17}$$

defines another one. Consequently,

$$U(\sigma\omega) = \Gamma(\sigma)U(\omega)\Gamma(\sigma)^{-1}$$

for some unitary operator  $\Gamma(\sigma)$ , which is determined up to a scalar. As a result, the mapping  $\sigma \mapsto \Gamma(\sigma)$  gives rise to a projective representation<sup>8</sup> of the symplectic group  $\operatorname{Sp}(V)$ , meaning a continuous homomorphism into  $\mathcal{U}(L^2(G))/\mathbb{T}$ . This projective representation may be lifted to a central extension of  $\operatorname{Sp}(V)$ , which may be taken as a double cover. As mentioned above there is only one double cover of  $\operatorname{Sp}(V)$ , resulting in a representation of  $\overline{\operatorname{Sp}}(V)$ .

However, the above discussion is not very instructive, especially if computations are required. For this thesis only Weil representations of  $SL_2$ , more accurately its metaplectic cover  $\overline{SL}_2$ , are required. Weil representations for this setting are given more explicitly

<sup>&</sup>lt;sup>7</sup>The proof features elementary fragments of the theory of  $C^*$ -Algebras [Seg63, p. 39, 41], which happen to be of no further use in supporting a deeper understanding of the ultimate goal of this thesis. As a consequence, it will be disregarded.

<sup>&</sup>lt;sup>8</sup>Details about projective representations are found in [Mac58]. If the reader is interested in the historic approach a far more elementary and less sophisticated source is [Sch04]. Note that the multipliers of projective representations are also called *Schur multiplier*.

in terms of integral and multiplication operators in Subsection B.3. Note that the term local Weil representation denotes a representation of the metaplectic extension  $\overline{\operatorname{SL}}_2(F_p)$ of  $\operatorname{SL}_2(F_p)$  into the space of automorphisms of a Schwartz Bruhat space  $S(V_p)$  for a finite vector space  $V_p$  over  $F_p$ , while the global (adelic) representation emerges from their combination. The representation factors through  $\operatorname{SL}_2$  if, and only if, the space  $V_p$  is even dimensional, rendering it an ordinary representation of  $\operatorname{SL}_2$  in that case.

## **B.4** Figures



Figure B.4.1: Illustration of the relation between finite Hecke characters and primitive Dirichlet characters in form of a commutative diagram. Double tipped arrows mark surjective projections, where the ones from  $\mathbb{A}^{\times} \to \hat{\mathbb{Z}}^{\times} \to (\mathbb{Z}/N\mathbb{Z})^{\times}$ are induced by taking the corresponding long paths. However, both of them are canonical and using the Chinese remainder theorem  $\hat{\mathbb{Z}}^{\times} \simeq \lim_{\leftarrow} (\mathbb{Z}/N\mathbb{Z})^{\times}$ may be verified. An in depth description is found in [Opi18, Sec. 1.10 p. 20].



Figure B.4.2: Visualisation of charts for a manifold M of dimension m over a field  $\mathbb{K} \in \{\mathbb{R}, \mathbb{C}\}$  and their respective transition map (represented by a blue arrow). Here,  $U_i, U_j$  are neighbourhoods on M,  $X_i, X_j$  neighbourhoods in  $\mathbb{K}^m$  and  $x_i, x_j$  denote the local coordinates. The illustration is based on [Tho15].

# Index

abs, 22 absolute norm, 34 absolute value function, 22 equivalence, 23 non-Archimedean, 22 *p*-adic, 23 action monoid, 113 Adele Eisenstein series, 171 Adele ring, 24 finite, 24 admissible indices, 116 affine Weierstraß equation, 44 algebra Clifford, 136 exterior, 135 grading, 134 Lie, 271 symmetric, 135 algebraic integers, 32 algebraic number fied fundamental identity, 35 alternating matrix, 75 Atkin–Lehner involution, 210 automorphic form, 53 automorphism

canonical, 135 Clifford, 137 automorphy factor, 53, 64, 152 cocycle relation, 64 bad place, 20prime number, 20 Baily-Borel compactification, 50Bernoulli number, 57, 66 Bessel function, 59 bilinear form, 13 pairing, 321 Birch Swinnerton-Dyer, 44 Borcherds denominator formula, 268 lift, 269 product, 269 boundary component one dimensional, 150 rational, 151 zero dimensional, 150 boundary point generic, 150

special, 150 Cartan decomposition, 271 Cayley transform, 48character, 26 second degree, 324 standard, 29 class number, 34 Clifford alegbra canonical involution, 137 algebra, 136 canonical automorphism, 137 even, 137 group, 138 representation, 138 module odd, 137 norm, 137, 180 representation, 138 cocycle, 152, 320 relation, 53, 64, 152 complementary series, 167 completed Riemman zeta function, 186 complexification, 133 confluent hypergeometric function, 309 congruence subgroup, 54, 67 orthogonal, 148 level, 67 principal, 67 convergence absolute infinite product, 241 cover metaplectic, 80curve

elliptic, 43 modular, 48cusp, 67cusp form, 58decomposition Cartan, 271 Iwasawa, 162 Dedekind  $\eta$ -function, 267 domain, 36 definite positive, 298 degree, 76 tensor, 134denominator formula, 268 determinant, 135 different ideal, 35differential forms, 271 Diophantine Geometry, 43discontinuous action, 54 discrete series, 167 discrete valuation ring, 36 discriminant, 33 function, 44 kernel, 146 division of lattice vectors, 18 divisor rational quadratic, 157 special, 157 divisor function, 57, 66 Eichler transform, 140 Eisenstein series, 46, 66 augmented, 58

classical, 56 adelic, 171 complete, 58 non-holomorphic, 103 orthogonal, 156 elementary tensors, 133 elliptic curve, 43, 44 rank, 44function, 46Euclid algorithm, 43 Euler product, 185 extension of scalars, 133 of  $\mathbb{Q}$ , 33 unramified, 36 exterior algebra, 135 power, 135factor of automorphy, 53, 64, 152 factorisable section, 169 faithful monoid action, 114 Faltings, 43 Fermat last Theorem, 43 field global, 23 local, 23, 37 of fractions, 32Finite Hecke character, 30 forms differential, 271 Fourier

expansion classic case, 57 transform, 26 free monoid action, 114 Fuchsian groups, 53 function confluent hypergeometric, 309 elliptic, 46matching, 177 rapidly decreasing, 109 Whittaker, 309 fundamental domain, 48 general Spin group, 139 generalised upper half plane, 145 generic boundary point, 150genus quadratic lattice, 16 symplectic group, 75 global field, 23 good place, 20 prime number, 20 graded algebra, 134 Gram matrix, 13 Grassmannian, 142 Grothendieck group, 17 group Heisenberg, 322 character, 26 Clifford, 138 covering, 319 discriminant, 20 dual, 26

genus, 75 Grothendieck, 17 Idele, 24 locally compact abelian, 25 metaplectic, 319, 326, 328 modular, 65 special linear, 321 Spin, 139 symmetric, 136 symplectic, 75, 321 half plane lower, 64upper, 64 generalised, 145 Hall divisor, 209 Hasse principle, 25 Hasse-Minkowsky, 43 Hecke theory, 54 algebra, 193 bound, 61 character, 30 eigenform, 201 operator, 192 primitive, 212 subgroup, 190 Heegner divisor, 2, 157, 269 Heisenberg, 322 group, 322Hilbert symbol, 30 holomorphic at  $\infty$ , 65 modular form, 154

homogeneous degree, 298 positive, 298 hyperbolic Laplacian, 42 measure, 66plane, 15 hypergeometric function, 309 hyperplane reflection, 138 ideal class groups, 34 fractional, 34 absolute norm, 34 inverse, 34 Idele group, 24 imaginary part, 142 indices admissible, 116 representable, 117 primitively, 117 inequality triangle, 298 inertia degree, 34 infinite product absolute convergence, 241 integral element, 32 inversion matrix, 65 involution canonical Clifford, 137 irreducible representation, 318 isotropic element, 14 Iwasawa decomposition, 82, 83, 162 j-invariant, 44 Jordan components, 16 K Bessel function, 59 kernel discriminant, 146Klein *j*-invariant, 268 Kloostermann sum, 103 Kudla-Millson-Funke lift, 273 Kudla-Millson lift, 272 Schwartz form, 271 theta function, 272Kähler form, 178 Laplacian, 95 lattice, 15basis, 15discriminant group, 20 division, 18 dual, 18 even, 18maximal, 19 quadratic signature, 18 representation number, 61 scaled, 14 unimodular, 18 vector primitive, 19 lca, 25 Leray cocycle, 80

level of an element, 20quadratic lattice, 19 L-function completion, 209 vector valued symmetric square, 236 Lie algebra, 271 lift Borcherds, 269 Kudla-Millson, 272 Kudla–Millson–Funke, 273 local field, 23 splitting index, 81local field, 23 lowering operator, 275 Maass differential operator, 59 form, 275 weak, 275 lowering operator, 275 raising operator, 275 majorant standard, 95 matrix alternating, 75 Gram, 13 measure hyperbolic, 66 self dual, 29 Mellin transformation, 61 meromorphic at  $\infty$ , 65 modular form, 155

metaplectic extension, 80group abstract, 328 Milgram formula, 20 Model Grassmannian, 142 modular curve, 48 function orthogonal, 151, 154 group, 65 inversion, 65 translation, 65 quasi character, 26, 168 variety, 149 modular form holomorphic, 154 meromorphic, 155 orthogonal, 155 module exterior algebra, 135 symmetric algebra, 135 tensor product, 132 monoid action, 113 diagonal, 115 faithful, 114 free, 114 product, 115 preorder, 113 Monster denominator formula, 268 Mordell, 43 morphism modulus, 320 multiplier system, 153

Möbius function, 187 transformation, 48, 64, 162 newforms, 208 norm, 33 Clifford, 137 Spinor, 140 normalised eigenfunction, 170 normed space, 298 N-primitive, 118 N-square free, 118 n-torsion, 21 order partial, 116 orthogonal congruence subgroup, 148 orthogonal Eisenstein series, 156 orthogonal group, 129 parabolic Poincaré series, 108 Parseval formula, 29 partial order, 116 Petersson product, 54 operator, 42, 53, 65 orthogonal, 156 product, 69 place, 23, 37 bad, 20 good, 20 Plancharel Theorem, 29 Pochhammer symbol, 111 Poincaré series, 108 Polynomial

degree, 95 Pontryagin, 26 dual group, 26, 320 positive definite, 298 positive homogeneous, 298 preorder monoidal, 113 prime ideal indecomposed, 35 nonsplit, 35 ramified, 35 totally, 35 split, 35 umramified, 35 unramified, 35 prime number bad, 20 good, 20 primitive Hecke operator, 212 primitively representable indices, 117 principal congruence subgroup, 67, 148 principal series representation, 168 even, 169 odd, 169 spherical section, 168product Borcherds, 269 quadratic lattice discriminant, 15 determinant, 13 extension discriminant, 33 extension of  $\mathbb{Q}$ , 33 form, 13 bilinear form, 13

non-degenerate, 14 regular, 13 standard majorant, 95 lattice, 15 bad prime, 20discriminant form, 19 discriminant group, 20 dual, 18even, 18 genus, 16 hyperbolic, 15 integral, 18 level, 19 maximal, 19 oddity, 16 rank, 15 representation number, 61split, 17 type, 16 unimodular, 18 module, 13 anisotropic, 14 direct sum, 14 isometry, 14 isomorphism, 14 isotropic, 14 morphism, 14 non-degenerate, 14 orthogonal complement, 14 reflection, 14 regular, 13 tensor product, 17 space, 13 character, 31 reflection, 129 signature, 14 quantum mechanics, 320

raising operator, 275 ramification index, 34 rank elliptic curve, 44 rapidly decreasing function, 109 rational boundary component, 151 real part, 142 real quadratic space, 14 reflection at hyperplane, 138 relation cocycle, 64 representable indices, 117 representation right regular, 317 character, 319 induced, 318 irreducible, 318 number, 61, 62 restricted tensor product, 82, 168 Riemann zeta function completed, 186 Euler product, 185 Riemann Zeta, 61 functional equation, 61Riemmann-Roch theorem, 44right regular representation, 317 ring of integers, 32 Schur multiplier, 329 Schwartz form Kudla-Millson, 271 Schwartz-Bruhat function, 28 section factorisable, 169

spherical, 168, 169 standard, 169 Segal, 326 Siegel domain, 66 theta function, 95 upper half space, 76 signature, 18 slash operator orthogonal, 156 special boundary point, 150linear group, 321 orthogonal cover, 87spherical section, 169Spin group, 139 general, 139 Spinor norm, 140 split, 35 hyperbolic, 17 lattice, 17 splitting homomorphism, 83 standard majorant, 171 section, 169 strong approximation, 84 strong operator topology, 322 strong triangle inequality, 22 sum direct quadratic module, 14 symmetric algebra, 135 group, 136 power, 135Symplectic group, 75
symplectic group, 321 tangent space, 271 tensor algebra, 134 canonical automorphism, 135 degree, 134product, 132 elementary tensors, 133 restricted, 168 theta distribution, 164 function, 97Siegel, 95 kernel, 164 trace, 33 translation matrix, 65 triangle inequality, 298 strong, 22 tube domain, 145 type real quadratic space, 14 uniformiser, 36 uniformly rapidly decreasing, 109 unitary operator, 322 upper half plane, 64 generalised, 145 upper half space, 76 valuation, 36vanishing at  $\infty$ , 65 variety modular, 149 vector representation, 138 vector space complexification, 133

```
Weierstraß
    equation
      affine, 44
    p-function, 46
Weil
    convergence criterion, 176
    representation, 328
      coefficient, 89
      integral, 86
Whittaker
    function, 309
Witt
    decomposition theorem, 15
    index, 15
    rank, 15
    ring, 17
zero quadric, 142
zeta
    integral, 24
```

## Glossary

Notation	Description	Page
		$\mathbf{List}$
R	A commutative, unitary, and associative ring.	13
q	A quadratic form.	13
b	The bilinear form associated to a quadratic form.	13
F	A field. If the characteristic is not mentioned, it will never equal 2.	13
V	A vector space. If the associated field is not	13
	mentioned, it will be a $\mathbb{Q}$ vector space.	
S	The Gram matrix of q.	13
$\det(M)$	The determinant of a free quadratic Module $(M, \mathbf{q})$ ,	13
	also $\det(\mathbf{q}) = \det(S) \in R/(R^{\times})^2$ for the Gram matrix $S$ .	
$N^{\perp}$	The orthogonal complement of a subset $N$ of a	14
	quadratic module $(M, q)$ .	
Н	The hyperbolic plane, meaning an unimodular $\mathbb{Z}$	15
	lattice of rank 2 and signature $(1, 1)$ .	
L	A lattice (usually accompanied by a quadratic form	15
	q).	
$V_p$	The tensor product $V \otimes_{\mathbb{Q}} \mathbb{Q}_p$ usually accompanied by	15
	a quadratic form q.	
$L_p$	The tensor product $L \otimes_{\mathbb{Z}} \mathbb{Z}_p$ usually accompanied by	15
	a quadratic form q.	
$q_1 \otimes q_2$	The tensor product of two quadratic forms.	17
$\operatorname{Groth}(M)$	The Grothendieck group of a cancellation monoid $M$ .	17
W(F)	The Witt ring of a field $F$ .	17
L'	The dual lattice of $L$ .	18
$\mathbb{Q}_p$	The field of <i>p</i> -adic numbers.	23
$\mathbb{Z}_p$	The ring of <i>p</i> -adic integers.	23

Notation	Description	Page
		$\mathbf{List}$
$\mathbb{A}_{f}$	The finite Adele ring - containing all $\mathbb{Q}_p$ for finite	24
	primes $p < \infty$ .	
$\hat{\mathbb{Z}}$	The product of all $\mathbb{Z}_p$ for $p < \infty$ . It is an open and	24
	compact subring of $\mathbb{A}_f$ .	
A	The Adele ring - containing all $\mathbb{Q}_p$ .	24
$\zeta(s)$	The Riemann Zeta function.	24,61
•	The absolute value on $\mathbb{A}^{\times}$ .	25
$\mathbb{A}_1$	The kernel of the absolute value on $\mathbb{A}^{\times}$ .	25
G	A locally compact abelian group with elements $g$ .	25
$\mu$	The Haar measure of $G$ .	25
$\delta_G$	The modular quasi character of $G$ , where	26,168
	$\delta_G(g)\mu(A) = \mu(Ag).$	
$\chi$	A character of $G$ .	26, 327
Т	The group of unitary complex numbers.	26,
		319, 323
$G^*$	The dual group of a locally compact abelian group $G$ .	26
$e_{\infty}$	Bicharacter for $\mathbb{Q}_{\infty}$ .	26
$e_p$	Bicharacter for $\mathbb{Q}_p$ .	26
$\hat{f}$	The Fourier transform of a function $f \in L^1(G)$ .	26
$\psi^{\mathrm{b}}(a,x)$	The concatenation of a character $\psi$ and a	27
	bilinearform b.	
$\mathcal{S}(S)$	The space of Schwartz Bruhat functions on an	28, 79
	appropriate topological space $S$ .	
$\chi_p$	The character associated to the place $p$ .	30
$\left(\frac{v}{n}\right)$	The Legendre symbol of $v \in \mathbb{Z}$ .	31
$\left( \begin{array}{c} p \end{array} \right)$ $\mathbb{F}_{n}$	The finite field with $p$ elements.	31
$\frac{-p}{\chi^b}$	The finite Hecke character associated with $b \in \mathbb{O}$ .	31
$\chi$	The character associated to a quadratic lattice $(L, q)$ .	32
Q(R)	The quotient field of an integral domain $R$ .	32
$\mathcal{O}_{F}$	The ring of integers of a field $F/Q(R)$ .	32
$tr_{E/K}$	The trace of an element $x \in F$ over the extension	33
$\Gamma / \Lambda$	F/Q(R).	
$N_{F/K}$	The norm of an element $x \in F$ over the extension	33
- ` <i>Г</i> / Л	F/Q(R).	
	= / v (-v)	

Notation	Description	Page
		$\mathbf{List}$
$M_x$	The linear Mapping $M_x: F \to F, y \mapsto xy$ associated	33
	to $x \in F$ .	
a	A fractional ideal of $F/Q(R)$ .	34
$J_F$	The group of fractional ideals $F/Q(R)$ .	34
$Cl_F$	The ideal class group of $F/Q(R)$ .	34
Γ	A discrete subgroup of a topological group $G$ –	41
	usually with additional properties to ensure that the	
	quotient is equipped with an appropriate structure	
	(cf. Fuchsian subgroup of $SL_2$ ).	
K	A compact subgroup of a topological group $G$ . In an	41
	adelic setting $K$ will usually assumed to be open.	
$SL_2$	The special linear group of $2 \times 2$ matrices.	42, 319
$SO_2$	The special orthogonal group of $2 \times 2$ matrices.	42, 82
E/F	An elliptic curve over the field $F$ .	44
$\wp$	The Weierstraß $\wp$ -function.	46
$\mathbb{H}$	The upper half plane consisting of all complex	47
	numbers $c$ with $\text{Im}(c) > 0$ .	
$\Gamma(1)$	The modular group $\Gamma(1) = \operatorname{SL}_2(\mathbb{Z})$ .	48
$Y(\Gamma)$	The modular curve $Y(\Gamma) = \Gamma \setminus \mathbb{H}$ for an	48
	automorphism group $\Gamma$ of $\mathbb{H}$ .	
$\mathcal{F}_{\Gamma}$	A fundamental domain for the automorphism group $\Gamma$	48
	of $\mathbb{H}$ .	
$\infty$	The element added to the modular curve $Y(\Gamma(1))$ for	50
	compactification. It is also referred to as the $cusp$ $\infty$	
	or just the cusp.	
$X(\Gamma)$	The compactified modular curve for $\Gamma$ .	50
$j(\gamma,  au)$	The factor of automorphy, relying on $\gamma \in SL_2(\mathbb{Z})$ and	53
	$ au \in \mathbb{H}.$	
$ _k$	Petersson operator of weight $k \in \mathbb{Z}$ , where	53
	$f _k \gamma = j(\gamma, \tau)^{-k} f(\gamma \tau)$ . It defines a right operation of	
	$\operatorname{SL}_2(\mathbb{Z})$ on meromorphic (holomorphic) functions on	
	HI.	
$\mathcal{M}_k(\Gamma)$	The Weierstraß $\wp$ -function.	53
$\Gamma(N)$	The kernel of the natural projection	54
	$\operatorname{SL}_2(\mathbb{Z}) \to \operatorname{SL}_2(\mathbb{Z}/N\mathbb{Z})$ for a natural number N.	

Notation	Description	Page
		$\mathbf{List}$
$\Gamma_0(N)$	A special congruence subgroup.	54
$\Gamma_1(N)$	A special congruence subgroup.	54
$\pi_1(M)$	The fundamental group of a Riemann surface $M$ .	55
$E_k$	The classical normalised Eisenstein series of weight $k \in 2\mathbb{N}$ .	56
$\sigma_s(n)$	The divisor function of the number $n$ with parameter $s \in \mathbb{C}$ .	57
$E_k(\tau,s)$	The normalised (non-holomorphic) augmented Eisenstein series of weight $k \in 2\mathbb{N}$ .	58
$G_k^*(\tau,s)$	The complete augmented Eisenstein series of weight $k \in 2\mathbb{N}$ .	58
$K_s$	The $K$ Bessel function.	59
$R_k$	A Maass differential operator.	59
$ heta_L$	The theta function associated to the quadratic lattice $L$ .	60
$r_L(n)$	Representation number of the natural number $n$ by the quadratic lattice $L$ .	61
$\zeta^*$	The completed Riemann Zeta function.	61
#(S,T)	The representation number of the Gram matrix $T$ by $S$ .	62, 175
$\operatorname{Pos}_n(\mathbb{Z})$	The set of symmetric positive definite $n \times n$ -matrices with coefficients in $\mathbb{Z}$ .	62
${}^{\mathrm{pr}}r_{p}$	The projective representation on SL <sub>2</sub> , inducing the Weil representation.	80
$c_L$	The Leray cocycle.	80
$\gamma(\eta)$	The Weil index of the character $\eta$ of second degree.	80
$q(g_1,g_2)$	The Leray invariant associated to the Leray triple of isotropic subspaces determined by $q_1, q_2$ .	80
$[g,z]_L$	The Leray bracket, parametrising $\overline{SL}_2$ .	80
$\omega_V$	The Weil representation of $\overline{\mathrm{SL}}_2$ into the space $\mathcal{S}(V_{\mathbb{A}})$ .	80
$\psi_{p.a}$	The mapping $x \mapsto e_p(a, x)$ .	80
$M(F_p)$	The matrix group of elements of the form $m(a)$ for $a \in F^{\times}$ .	82
$N(F_p)$	The matrix group of elements of the form $n(b)$ for $b \in F_p$ .	82

Notation	Description	Page
l.	Elements of SO $(\mathbb{D})$ perpendicised by an angle of $\subset \mathbb{T}$	L1St
$\kappa_{artheta}$	Elements of $SO_2(\mathbb{R})$ parametrised by an angle $v \in \mathbb{I}$ . The matrix $n(x)m(\sqrt{n})$ representing the number	00, 100 92, 166
$g_{ au}$	The matrix $n(x)m(\sqrt{y})$ representing the number $\pi = x + iy \in \mathbb{N}$	05, 100
D	The real matrix group of $2 \times 2$ upper triangular	83
$I_{\infty}$	The real matrix group of $2 \times 2$ upper triangular matrices	00
K	The group $\operatorname{SL}_2(\mathbb{Z})$ for $n < \infty$	83
	The group $\operatorname{SL}_2(\mathbb{Z}_p)$ for $p < \infty$ .	83
$K_{f}$	The adelic group $K \times K_c$	83
$P_{\mathbb{A}}$	The adelic Borel group of upper triangular matrices	83
IA	in $G_{\star}$	00
$P'_{\cdot}$	The preimage under the splitting morphism of $P_{\bullet}$ in	83
• A	The preimage under the sphtting morphism of $T_{\mathbb{A}}$ in $G'_{\mathbb{A}}$	00
K'	$\mathcal{O}_{\mathbb{A}}$ . The preimage under the splitting morphism of $K$ in	83
$m_{\infty}$	$G'_{*}$ .	00
$K_0(4)_2$	The completion of $\Gamma_0(4)$ in $K_2$ .	84
$K_0(4)$	The group $K_0(4)_2 \times \prod_{2 \leq n \leq \infty} K_n$ .	84
$K_0$	A subgroup of $K_0(4)$ , which is open and compact in	84
Ŭ	$K_f$ – it is identified with its preimage in $G'_{\mathbb{A}}$ .	
$Mp_2$	A subgroup of $\overline{SL}$ , representing a central extension of	84
1 2	$SL_2$ by $\{\pm 1\}$ .	
${\cal S}_{L'/L}$	The subspace of $\mathcal{S}(V_f)$ generated by indicator	85
	functions of cosets of $L'$ by $L$ .	
$ ho_L$	The discrete Weil representation on $Mp_2(\mathbb{Z})$ to	86
	$\mathbb{C}[L'/L].$	
$\overline{\mathrm{GL}_2^+}$	The metaplectic cover of $GL_2^+$ .	87
$\overline{T}$	The element $\left(\begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, 1\right) \in Mp_2(\mathbb{Z}).$	87
$\overline{S}$	The element $\left( \begin{pmatrix} 0 & -1 \\ 0 & 1 \end{pmatrix}, \sqrt{\tau} \right) \in \mathrm{Mp}_2(\mathbb{Z}).$	87
$\overline{Z}$	The element $\left( \begin{pmatrix} -1 & 0 \\ 0 & -1 \end{pmatrix}, i \right) \in \mathrm{Mp}_2(\mathbb{Z}).$	87
$\overline{U}$	The element $\overline{ST}^{-1}\overline{S}^{-1} \in \mathrm{Mp}_2(\mathbb{Z}).$	88
$R_d$	The element $\overline{ST}^d \overline{S}^{-1} \overline{T}^a \overline{ST}^d \in \mathrm{Mp}_2(\mathbb{Z})$ for suitable	88
	elements $a, b \in \mathbb{Z}$ .	
$\chi_{\mathcal{L}}$	A special character describing the action of	94
	$\Gamma_0(\operatorname{lev}(\mathcal{L}))$ under the discrete Weil representation.	
$\mathcal{O}(M)$	The orthogonal group of a quadratic module $(M, q)$ .	129
$\mathbb{T}$	The unit circle in $\mathbb{C}$ .	130

Notation	Description	Page
		$\mathbf{List}$
$\mathcal{T}(M)$	The tensor algebra of the module $M$ .	134
$\mathcal{C}_M$	The Clifford algebra of the quadratic module $M$ .	136
Ν	The Clifford norm.	137
$\mathfrak{G}_M$	The Clifford group of the quadratic module $M$ .	138
$\operatorname{GSpin}_M$	The general Spin group of the quadratic module $M$ .	139
$\operatorname{Spin}_M$	The Spin group of the quadratic module $M$ .	139
$\mathbb{D}(V)$	The Grassmannian of the quadratic vector space $V$ .	142
$\mathcal{K}$	The projective model of the Grassmannian.	143
$\Gamma(L)$	The discriminant kernel of the lattice $L$ .	146
$\mathcal{Z}_v$	The special divisor of $\mathbb{D}$ induced by some $v \in \mathbb{D}$ .	157
$\theta$	The theta distribution associated to $h \in O_{\mathbb{A}}(V)$ .	164
$ heta_L$	The theta series associated with the quadratic lattice	166
	L.	
H	A closed subgroup of the topological group $G$ .	167
$\pi_H$	A smooth representation of the closed subgroup $H$ .	168
$\operatorname{Ind}_{H}^{G}(\pi_{H})$	The induced representation of $G$ by $\pi_H$ of $G$ .	168
$I_p(s,\chi)$	The local principal series representation associated to	168
	$s, \chi.$	
$\Phi_p(g,s)$	Holomorphic sections through the bundles of local	168
	principal series representations $I_p(s, \chi)$ .	
$\Phi_p^0(g,s)$	The spherical section, characterised by being right $K_p$	168
1	invariant and normalised.	
$I(s,\chi)$	The global principal series representation associated	168
	to $s, \chi$ .	
$\Phi(g,s)$	Holomorphic sections through the bundles of	169
	principal series representations $I(s, \chi)$ .	
$I(g; \varphi)$	The theta integral of the theta distribution associated	176
	with a Schwartz Bruhat form $\varphi$ .	
$ _k$	The Petersson slash operator of weight $k \in \mathbb{Z}$ .	189
T(n)	The standard Hecke operator of level $n \in \mathbb{N}$ .	198
g(n)	The matrix $\begin{pmatrix} n & 0 \\ 0 & 1 \end{pmatrix} \in \operatorname{GL}_2^+(\mathbb{Q})$ for some $n \in \mathbb{N}$ .	212
$\mathcal{T}(n)$	The primitive Hecke operator of level $n \in \mathbb{N}$ , directly	212
	associated to $g(n)$ .	
$\sigma_n$	The eigenvalue of a modular form $f$ under the	223
	primitive Hecke operator $\mathcal{T}(n)$ .	

Notation	Description	Page
		$\mathbf{List}$
$L_{\ell}(f,s)$	The L-series $L_{\ell}(f,s) = \sum_{n \in \mathbb{N}} a(n\overline{\ell}, n^2 \operatorname{q}(\ell))/n^s$	236
	associated to a modular form $f \in \mathcal{M}_{L,k}$ with Fourier	
	coefficients $a(\lambda, m)$ and an isotropic element $\ell \in L'$ .	
$\Phi(z,f)$	Borcherds' regularised additive theta lift.	268
$\Psi(z,f)$	Borcherds' regularised multiplicative lift.	269
$\varphi_{ m KM}$	The Kudla–Millson Schwartz form.	271
$\Theta_{\rm KM}$	The Kudla–Millson theta form.	272
$\Lambda_{ m KM}$	The Kudla–Millson theta lift.	272
$\overline{\mathrm{SL}}_2$	The metaplectic extension of $SL_2$ .	319
$\alpha^*$	The dual morphism to $\alpha: G \to H$ , given as a	320
	pullback.	
F	A bilinear pairing inducing – the multiplier of the	321
	Heisenberg group.	
$\kappa$	The symplectic pairing of $G \times G^*$ .	321
$\operatorname{Sp}(G)$	The symplectic group of $G \times G^*$ .	321
$L^2(G)$	The complex Hilbert space of square integrable	322
	functions on $(G, \mu)$ .	
$U(\omega)$	The unitary operator induced by $\omega \in G \times G^*$ .	322
A(G)	The Heisenberg group of $G$ .	322
[Q, P]	The commutator $QP(PQ)^{-1}$ of $Q, P$ .	322
B(G)	The group of automorphisms of $A(G)$ .	323
$B_0(G)$	The group of automorphisms of $A(G)$ fixing its	323
	centre.	
$\langle\cdot,\cdot angle$	Duality brackets, taking an element of $G \times G^*$ and	324
	merging the components.	
$X^2(G)$	The group of characters of second degree for $G$ .	324
ρ	The morphism associated to a bilinear paring.	324
$\operatorname{Hom}_{s}(G, G^{*})$	The subgroup of symmetric morphisms of	324
	$\operatorname{Hom}(G, G^*).$	
$f_{\sigma}$	The image of $\sigma$ under the section $\operatorname{Sp}(G) \to B_0(G)$ ,	325
	splitting the Sequence (B.9).	
$\overline{B}_0(G)$	The normaliser of $A(G)$ in $Aut(L^2(G))$ .	326
$\pi_0$	The projection of $\overline{B}_0(G)$ onto $B_0(G)$ .	326
H(V)	The Heisenberg group of the Vector space $V$ .	327
$\operatorname{Ps}(V)$	The pseudo symplectic group of a vector space $V$ .	327

Notation	Description	Page
		$\mathbf{List}$
Mp	The abstract metaplectic group.	328
$\mathbb{N}$	The natural numbers, excluding the zero.	
$\mathbb{N}_0$	The natural numbers, including the zero.	
$\mathbb{Z}$	The integers.	
$\mathbb{Q}$	The rational numbers.	
$\mathbb{R}$	The real numbers.	
$\mathbb{C}$	The complex numbers.	
gcd(a, b)	The greatest common divisor of $a, b \in \mathbb{Z}$ .	
$\operatorname{Im}( au)$	The imaginary part of $\tau \in \mathbb{H}$ .	
$\operatorname{Re}(s)$	The real part of $s \in \mathbb{C}$ .	
$\langle \cdot,\cdot angle$	A scalar product.	

## Bibliography

- [AL70] A. O. L. Atkin and J. Lehner. 'Hecke operators on  $\Gamma_0(m)$ '. In: *Mathematische Annalen* 185.2 (June 1970), pp. 134–160. ISSN: 1432-1807. DOI: 10.1007/bf01359701.
- [Art57] Emil Artin. Geometric Algebra. Ed. by L. Bers, R. Courant and J. J. Stoker. Interscience tracts in pure and applied mathematics 3. Catalog number 57-6109. New York, London: Interscience publishers, 1957. URL: https:// archive.org/details/geometricalgebra033556mbp/page/n3/mode/2up? view=theater.
- [Art98] Michael Artin. *Algebra* -. 1993. Aufl. Birkhaeuser Basel, 1998. ISBN: 978-3-764-35938-6.
- [Att15] Dylan Attwell-Duval. 'Algebraic and Geometric Properties of the Boundary of Orthogonal Shimura Varieties'. PhD thesis. McGill University - Montréal, Québec, 2015.
- [Bar03] Alex Barnard. 'The Singular Theta Correspondence, Lorentzian Lattices and Borcherds-Kac-Moody Algebras'. PhD thesis. U.C. Berkeley, 2003.
- [BC79] Armand Borel and William Allen Casselman. Automorphic forms, representations, and L-functions. Ed. by Armand Borel and Bill Casselman. Online-Ausg. Mathematics, Symposium in Pure v. 33.1. Held at Oregon State University, Corvallis, July 11-August 5, 1977. - Includes bibliographical references and index. - Electronic reproduction; Providence, Rhode Island; American Mathematical Society; 2012. - Description based on print version record. Heidelberg: American Mathematical Soc., 1979. 12 pp. ISBN: 978-0-821-81435-2.
- [BCJ18] Vincent Bouchard, Thomas Creutzig and Aniket Joshi. 'Hecke Operators on Vector-Valued Modular Forms'. In: SIGMA 15 (2019), 041, 31 pages 15 (20th July 2018). ISSN: 1815-0659. DOI: 10.3842/SIGMA.2019.041. arXiv: 1807.07703 [math.NT].

[Ber+16]	Nicolas Bergeron et al. 'The Noether-Lefschetz conjecture and generalizations'.
	In: Inventiones mathematicae 208.2 (Sept. 2016), pp. 501–552. ISSN: 1432-1297.
	DOI: 10.1007/s00222-016-0695-z.

- [Ber01] Rolf Berndt. An Introduction to Symplectic Geometry -. Heidelberg: American Mathematical Soc., 2001. ISBN: 978-0-821-82056-8.
- [Ber07] Rolf Berndt. Representations of Linear Groups An Introduction Based on Examples from Physics and Number Theory. Berlin Heidelberg: Springer Science & Business Media, 2007. ISBN: 978-3-834-89401-4.
- [BEW98] Bruce C. Berndt, Ronald J. Evans and Kenneth S. Williams. Gauss and Jacobi sums. Ed. by Ronald J. Evans and Kenneth S. Williams. A Wiley-Interscience publication. Hier auch später erschienene, unveränderte Nachdrucke. New York, NY: John Wiley & Sons, 1998, pp. 978–. 583 pp. ISBN: 0-471-12807-4.
- [BF01] Jan Hendrik Bruinier and Eberhard Freitag. 'Local Borcherds products'. en. In: Annales de l'Institut Fourier 51.1 (2001), pp. 1–26. DOI: 10.5802/aif.1812.
   URL: https://aif.centre-mersenne.org/articles/10.5802/aif.1812/.
- [BF04] Jan H. Bruinier and Jens Funke. 'On two geometric theta lifts'. In: Duke Mathematical Journal 125.1 (Oct. 2004), pp. 45–90. DOI: 10.1215/S0012– 7094-04-12513-8.
- [BF10] Jan H. Bruinier and Jens Funke. 'On the injectivity of the Kudla-Millson lift and surjectivity of the Borcherds lift.' In: Moonshine : the first quarter century and beyond : proceedings of a workshop on the moonshine conjectures and vertex algebras. Ed. by J. Lepowsky, J. McKay and M. Tuite. London mathematical society lecture note series. 372. Cambridge: Cambridge University Press, Sept. 2010, pp. 12–39. URL: http://dro.dur.ac.uk/12403/.
- [BH08] Valentin Blomer and Gergely Harcos. 'Hybrid bounds for twisted L-functions'.
   In: Journal für die reine undangewandte Mathematik 621 (2008), pp. 53–79.
   DOI: 10.1515/CRELLE.2008.058.
- [BK01] Jan Hendrik Bruinier and Michael Kuss. 'Eisenstein series attached to lattices and modular forms on orthogonal groups'. In: manuscripta mathematica 106.4 (Dec. 2001), pp. 443–459. ISSN: 1432-1785. DOI: 10.1007/s229-001-8027-1.
- [BK03] Jan Hendrik Bruinier and Ulf Kühn. 'Integrals of automorphic Green's functions associated to Heegner divisors'. In: International Mathematics Research Notices 2003.31 (Jan. 2003), pp. 1687–1729. ISSN: 1073-7928. DOI: 10.1155/S1073792803204165. eprint: https://academic.oup.com/imrn/article-

pdf/2003/31/1687/2235886/2003-31-1687.pdf. URL: https://doi.org/ 10.1155/S1073792803204165.

- [BM19] Jan Bruinier and Martin Möller. 'Cones of Heegner divisors'. In: Journal of Algebraic Geometry 28.3 (Apr. 2019), pp. 497–517. ISSN: 1534-7486. DOI: 10.1090/jag/734.
- [Boc06] Siegfried Böcherer et al., eds. Automorphic Forms and Zeta Functions. Proceedings of the Conference in Memory of Tsuneo Arakawa. Proceedings of the Conference in Memory of Tsuneo Arakawa : Rikkyo University, Japan, 4-7 September 2004. Description based on publisher supplied metadata and other sources. Singapore: World Scientific Publishing Co Pte Ltd, 2006. 1400 pp. ISBN: 978-9-812-56632-4.
- [Bor00] Richard E. Borcherds. 'Reflection groups of Lorentzian lattices'. In: Duke Mathematical Journal 104.2 (Aug. 2000), pp. 319–366. ISSN: 0012-7094. DOI: 10.1215/s0012-7094-00-10424-3.
- [Bor65] Armand Borel. 'Introduction to Automorphic Forms'. In: Algebraic groups and discontinuous subgroups. Vol. 9. Proceedings of Symposia in pure mathematics
   symposium on algebraic groups. Boulder, Colorado: American mathematical society, 1965, pp. 199–210. ISBN: 0-8218-1409-5.
- [Bor91] Armand Borel. Linear Algebraic Groups -. 130th ed. Cambridge tracts in mathematics. Berlin Heidelberg: Springer Science & Business Media, 1991.
   ISBN: 9781461209416. DOI: 10.1007/978-1-4612-0941-6.
- [Bor95] Richard E. Borcherds. 'Automorphic forms on O(s + 2,2)(R) and infinite products'. In: *Inventiones mathematicae* 120.1 (1995), pp. 161–213. ISSN: 1432-1297. DOI: 10.1007/BF01241126. URL: https://doi.org/10.1007/BF01241126.
- [Bor97] Armand Borel. Automorphic Forms on SL2 (R) -. transferred to digital print.
   2004. Cambridge tracts in mathematics 130. Cambridge [u.a.]: Cambridge University Press, 1997. ISBN: 978-0-521-58049-6.
- [Bor98] Richard E. Borcherds. 'Automorphic forms with singularities on Grassmannians'. In: *Inventiones mathematicae* 132.3 (1998), pp. 491–562. ISSN: 1432-1297. DOI: 10.1007/s002220050232.
- [Bos08] Siegfried Bosch. Lineare Algebra -. Berlin Heidelberg New York: Springer Berlin Heidelberg, 2008. ISBN: 978-3-540-76438-0. DOI: 10.1007/978-3-540-76438-0.

[Bos09]	Siegfried Bosch. <i>Algebra</i> 7. überarb. Aufl. 2009. Berlin, Heidelberg: Springer, 2009. ISBN: 978-3-540-92811-9.
[Bos18]	Siegfried Bosch. Algebra: From the Viewpoint of Galois Theory. Springer International Publishing, 2018. ISBN: 9783319951775. DOI: 10.1007/978-3- 319-95177-5.
[Boy14]	Hatice Boylan. Jacobi Forms, Finite Quadratic Modules and Weil Representa- tions over Number Fields. 1st ed. Berlin, Heidelberg: Springer International Publishing, 2014. ISBN: 978-3-319-12916-7. DOI: 10.1007/978-3-319-12916- 7.
[Bru+08]	Jan Hendrik Bruinier et al. <i>The 1-2-3 of Modular Forms - Lectures at a Summer School in Nordfjordeid, Norway.</i> Ed. by Kristian Ranestad. Berlin Heidelberg: Springer Science & Business Media - Berlin Heidelberg, 2008. ISBN: 978-3-540-74119-0. DOI: 10.1007/978-3-540-74119-0.
[Bru02]	Jan H. Bruinier. Borcherds products on $O(2, l)$ and Chern classes of Heegner divisors. Berlin, Heidelberg: Springer Berlin Heidelberg, 2002. ISBN: 978-3-540-45872-2. DOI: 10.1007/b83278.
[Bru14]	Jan Hendrik Bruinier. 'On the converse theorem for Borcherds products'. In: <i>Journal of Algebra</i> 397 (Jan. 2014), pp. 315–342. ISSN: 0021-8693. DOI: 10.1016/j.jalgebra.2013.08.034.
[Bru99]	J. H. Bruinier. 'Borcherds products and Chern classes of Hirzebruch-Zagier divisors'. In: <i>Invent. math.</i> 138 (1999), pp. 51–83. DOI: https://doi.org/10.1007/s002220050341.
[BS]	Z. I. Borevich and I. R. Shafarevich. <i>Number Theory</i> Amsterdam, Boston: Academic Press. ISBN: 978-0-080-87332-9.
[BS08]	Jan H. Bruinier and Oliver Stein. 'The Weil representation and Hecke operators for vector valued modular forms'. In: <i>Mathematische Zeitschrift</i> 264.2 (2008), pp. 249–270. DOI: 10.1007/s00209-008-0460-0. URL: https://doi.org/10.1007/s00209-008-0460-0.
[Bum98]	Daniel Bump. <i>Automorphic Forms and Representations</i> Revised ed. Cambridge: Cambridge University Press, 1998. ISBN: 978-0-521-65818-8.
[BY09]	Jan Hendrik Bruinier and Tonghai Yang. 'Faltings heights of CM cycles and derivatives of L-functions'. In: <i>Inventiones mathematicae</i> 177.3 (Sept. 2009), pp. 631–681. ISSN: 1432-1297. DOI: 10.1007/s00222-009-0192-8. URL: https://doi.org/10.1007/s00222-009-0192-8.

- [BY21] Jan Hendrik Bruinier and Tonghai Yang. 'Arithmetic degrees of special cycles and derivatives of Siegel Eisenstein series'. In: Journal of the European Mathematical Society 23.5 (Jan. 2021), pp. 1613–1674. ISSN: 1435-9863. DOI: 10.4171/jems/1040.
- [Cas73] William Casselman. 'On some results of Atkin and Lehner'. In: *Mathematische Annalen* 201.4 (Dec. 1973), pp. 301–314. ISSN: 1432-1807. DOI: 10.1007/bf01428197.
- [Cas78] John W. S. Cassels. Rational Quadratic Forms. @London Mathematical Society monographs 13. Mineola, New York: Courier Dover Publications, 1978. 413 pp. ISBN: 0-12-163260-1.
- [Cas82] J.W.S. Cassels. 'Rational Quadratic Forms'. In: Proceedings of the International Mathematical Conference. Elsevier, 1982, pp. 9–26. ISBN: 9780444865106. DOI: 10.1016/s0304-0208(08)70410-9.
- [Cas86] J. W. S. Cassels. Local Fields. Cambridge, UK: Cambridge University Press, Aug. 1986. ISBN: 978-0-521-31525-8. DOI: 10.1017/cbo9781139171885.
- [CF10] J. W. S. Cassels and A. Fröhlich. Algebraic Number Theory Proceedings of an Instructional Conference Organized by the London Mathematical Society (a NATO Advanced Study Institute) with the Support of the International Mathematical Union. London Mathematical Society, 2010. ISBN: 978-0-950-27342-6.
- [Con94] John B. Conway. A Course in Functional Analysis. Berlin Heidelberg: Springer Science & Business Media, 1994. ISBN: 978-0-387-97245-9. DOI: 10.1007/978-1-4757-4383-8.
- [DAR84] Michael Danos, Milton Abramowitz and Johann Rafelski. Pocketbook of Mathematical Functions -. Frankfurt/Main: H. Deutsch, 1984. ISBN: 978-3-871-44818-8.
- [DE14] Anton Deitmar and Siegfried Echterhoff. Principles of Harmonic Analysis.
   Berlin, Heidelberg: Springer International Publishing, 2014. ISBN: 978-3-319-05792-7. DOI: 10.1007/978-3-319-05792-7.
- [Dei10] Anton Deitmar. Automorphe Formen. 1. Aufl. Berlin Heidelberg New York: Springer Berlin Heidelberg, 2010. ISBN: 978-3-642-12390-0. DOI: 10.1007/ 978-3-642-12390-0.

[DS05]	Fred Diamond and Jerry Shurman. A First Course in Modular Forms. 1st. Berlin Heidelberg: Springer Science & Business Media New York, 2005. ISBN: 978-0-387-23229-4. DOI: 10.1007/978-0-387-27226-9.
[Duk88]	William Duke. 'Hyperbolic distribution problems and half-integral weight Maass forms'. In: <i>Inventiones Mathematicae</i> 92.1 (Feb. 1988), pp. 73–90. ISSN: 1432-1297. DOI: 10.1007/bf01393993.
[Ebe12]	Wolfgang Ebeling. Lattices and Codes: A Course Partially Based on Lectures by Friedrich Hirzebruch. 3. Aufl. Berlin Heidelberg: Springer Science & Business Media, 2012. ISBN: 978-3-658-00359-3. DOI: 10.1007/978-3-658-00360-9.
[EGS23]	Philip Engel, Jan Goedgebeur and Peter Smillie. <i>Exact enumeration of fullerenes</i> . 2023. DOI: 10.48550/ARXIV.2304.01655.
[Ehl03]	Stephan Jakob Ehlen. 'CM values of regularized theta lifts'. PhD thesis. TU Darmstadt, 2003.
[Els11]	Jürgen Elstrodt. <i>Maß- und Integrationstheorie</i> . Wiesbaden: Springer Berlin Heidelberg, 2011. ISBN: 978-3-642-17905-1. DOI: 10.1007/978-3-662-57939-8.
[Erd81]	Arthur Erdélyi. <i>Higher Transcendental Functions - Based in Part, on Notes Left by Harry Bateman.</i> Vol. 1. Malabar, Florida: Robert E Krieger Publishing company INC, 1981.
[FB06]	Eberhard Freitag and Rolf Busam. <i>Funktionentheorie 1</i> . Berlin Heidelberg New York: Springer Berlin Heidelberg, 2006. ISBN: 978-3-662-07352-0. DOI: 10.1007/3-540-32058-x.
[Fis10]	Gerd Fischer. <i>Lineare Algebra - Eine Einführung für Studienanfänger</i> . Berlin Heidelberg New York: Vieweg+Teubner, 2010. ISBN: 978-3-834-89365-9. DOI: 10.1007/978-3-8348-9365-9.
[FK17]	Jens Funke and Stephen S Kudla. 'Mock modular forms and geometric theta functions for indefinite quadratic forms'. In: <i>Journal of Physics A: Mathematical and Theoretical</i> 50.40 (Sept. 2017), p. 404001. ISSN: 1751-8121. DOI: 10.1088/1751-8121/aa848b.
[FM12]	Jens Funke and John Millson. 'Boundary behaviour of special cohomology classes arising from the Weil representation'. In: <i>Journal of the Institute of Mathematics of Jussieu</i> 12.3 (July 2012), pp. 571–634. ISSN: 1475-3030. DOI: 10.1017/s1474748012000795.

- [Fra20] J. Franke. 'Rational functions and modular forms'. In: Proceedings of the American Mathematical Society 148.10 (June 2020), pp. 4151–4164. ISSN: 1088-6826. DOI: 10.1090/proc/15034.
- [Fre13] Eberhard Freitag. Siegelsche Modulfunktionen -. Berlin Heidelberg New York: Springer-Verlag, 2013. ISBN: 978-3-642-68649-8.
- [Fre83] Eberhard Freitag. Siegelsche Modulfunktionen. Die Grundlehren der mathematischen Wissenschaften in Einzeldarstellungen 254. Hier auch später erschienene, unveränderte Nachdrucke. Berlin: Springer Berlin Heidelberg, 1983. 341 pp. ISBN: 0387116613. DOI: 10.1007/978-3-642-68649-8.
- [Fre90] Eberhard Freitag. Hilbert Modular Forms. Berlin Heidelberg: Springer Science & Business Media, 1990. ISBN: 978-3-662-02638-0. DOI: 10.1007/978-3-662-02638-0.
- [Fre91] Eberhard Freitag. Singular Modular Forms and Theta Relations. Ed. by
   F. Takens A. Dold B. Eckmann. Springer Berlin Heidelberg, 1991. ISBN:
   0-387-54704-5. DOI: 10.1007/bfb0093829.
- [Fri13] Solomon Friedberg. Automorphic Forms: A Brief Introduction. https:// icerm.brown.edu/materials/Slides/sp-s13-off\_weeks/Automorphic\_ Forms-\_A\_Brief\_Introduction\_]\_Solomon\_Friedberg,\_Boston\_College. pdf. [Online; accessed 23-September-2016]. 2013.
- [Fun02] Jens Funke. 'Heegner Divisors and NonholomorphicModular Forms'. In: Compositio Mathematica 133.3 (2002), pp. 289–321. ISSN: 0010-437X. DOI: 10. 1023/a:1020002121978.
- [Fun94] Jens Funke. 'Spuroperator und Thetareihen quadratischer Formen'. MA thesis. Universität zu Köln, 1994.
- [Gan14] Terry Gannon. 'The Theory of Vector-Valued Modular Forms for the Modular Group'. In: Conformal Field Theory, Automorphic Forms and Related Topics. Ed. by Winfried Kohnen and Rainer Weissauer. Berlin, Heidelberg: Springer Berlin Heidelberg, 2014, pp. 247–286. ISBN: 978-3-662-43831-2. DOI: 10.1007/978-3-662-43831-2\_9.
- [Gar17] Paul Garret. Modern analysis of automorphic forms by examples. http: //www-users.math.umn.edu/~garrett/m/v/Book\_28Aug2017.pdf. [Online; accessed 29-January-2018]. 2017.

[GBL10]	Timothy Gowers, June Barrow-Green and Imre Leader. <i>The Princeton Com-</i> <i>panion to Mathematics</i> Kassel: Princeton University Press, 2010. ISBN: 140-0-830-397
[Gel+90]	Izrail Moiseevich Gelfand et al. <i>Representation Theory and Automorphic Func-</i> <i>tions: Representation Theory and Automorphic Forms.</i> Vol. Vol 6. Generalized Functions, Vol 6. Academic Pr, 1990. ISBN: 0122795067,9780122795060.
[Gel74]	Stephen S. Gelbart. Automorphic forms on the metaplectic group. 1st ed. mimeographed notes. Cornell University, 1974.
[Gel75]	Stephen S. Gelbart. Automorphic Forms on Adele Groups. (AM-83). Annals of Mathematics Studies v.83. Description based upon print version of record. Kassel: Princeton University Press, 1975. 1280 pp. ISBN: 069-1-081-565 URL: https://www.jstor.org/stable/j.ctt1b7x82z.
[Gel76]	Stephen Samuel Gelbart. Weil's Representation and the Spectrum of the Meta- plectic Group. 1st ed. Lecture Notes in Mathematics. Springer Berlin Heidel- berg, 1976. ISBN: 3-540-07799-5,0-387-07799-5. DOI: 10.1007/bfb0077830.
[GF52]	Israel Gelfand and Sergey Fomin. 'Geodesic flows on manifolds of constant negative curvature'. In: Uspekhi Mat. Nauk 7 (1952), pp. 118–137.
[Gro02]	Larry C. Grove. <i>Classical Groups and Geometric Algebra</i> . Graduate studies in mathematics volume 39. Includes bibliographical references (pages 161-163) and index. Description based on print version record. Providence, Rhode Island: American Mathematical Society, 2002. 1181 pp. ISBN: 978-0-821- 82019-3. URL: https://bookstore.ams.org/view?ProductCode=GSM/39.
[GW20]	Ulrich Görtz and Torsten Wedhorn. <i>Algebraic Geometry I: Schemes: With Examples and Exercises.</i> Singapore: Springer Fachmedien Wiesbaden, 2020. ISBN: 978-3-658-30733-2. DOI: 10.1007/978-3-658-30733-2.
[Has23]	Helmut Hasse. 'Über die Darstellbarkeit von Zahlen durch quadratische Formen im Körper der rationalen Zahlen'. In: <i>Journal für die reine und angewandte Mathematik</i> 152 (1923), pp. 129–148.
[Has24]	Helmut Hasse. 'Über die Darstellbarkeit von Zahlen durch quadratische Formen in einem beliebigen algebraischen Zahlkörper'. In: <i>Journal für die reine und angewandte Mathematik</i> 153 (1924), pp. 113–130.
[Hat02]	Allen Hatcher. <i>Algebraic Topology</i> Cambridge: Cambridge University Press, 2002. ISBN: 978-0-521-79540-1.

- [Hec26] Erich Hecke. 'Zur Theorie der elliptischen Modulfunktinonen'. In: *Mathematische Annalen.* 97 (1926), pp. 210–242.
- [Hec83] Erich Hecke. Mathematische Werke. Göttingen, 1983. URL: http://scans. hebis.de/HEBCGI/show.pl?08128597\_toc.pdf.
- [Hej76] Dennis A. Hejhal. The Selberg Trace Formula for PSL(2, ). Vol. 1. Springer Berlin Heidelberg, 1976. ISBN: 9783540379799. DOI: 10.1007/bfb0079608.
- [Hej83] Dennis A. Hejhal. The Selberg Trace Formula for PSL(2, ). Vol. 2. Springer
   Berlin Heidelberg, 1983. ISBN: 9783540409144. DOI: 10.1007/bfb0061302.
- [Hel90] H.G. Zimmer Helmut Hasse. Number theory. 1st ed. Grundlehren der mathematischen Wissenschaften. Springer-Verlag Berlin, 1990. ISBN: 3540082751, 9783540082750.
- [HZ76] F. Hirzebruch and D. Zagier. 'Intersection numbers of curves on Hilbert modular surfaces and modular forms of Nebentypus'. In: *Inventiones Mathematicae* 36.1 (Dec. 1976), pp. 57–113. ISSN: 1432-1297. DOI: 10.1007/bf01390005.
- [IK04] Henryk Iwaniec and Emmanuel Kowalski. Analytic Number Theory. Providence, Rhode Island: American Mathematical Society, 2004. ISBN: 978-0-821-83633-0.
- [Iwa87] Henryk Iwaniec. 'Fourier coefficients of modular forms of half-integral weight'. In: *Inventiones Mathematicae* 87.2 (June 1987), pp. 385–401. ISSN: 1432-1297. DOI: 10.1007/bf01389423.
- [Iwa95] Henryk Iwaniec. Introduction to the spectral theory of automorphic forms. Revista Matematica Iberoamericana, 1995.
- [Jab21] JabRef Development Team. JabRef An open-source, cross-platform citation and reference management software. Version 5.1. 2021. URL: https://www. jabref.org/.
- $[Jac06] Hervé Jacquet. `Local theory for GL(2) \times GL(2)'. In: Automorphic Forms on$ GL(2) Part II. Berlin, Heidelberg: Springer Berlin Heidelberg, 2006, pp. 1–116.ISBN: 978-3-540-37612-5. DOI: 10.1007/bfb0058504.
- [Jac70] R. P. Jacquet H.; Langlands. Automorphic Forms on GL (2) Part 1. 1st ed.
   114. Berlin, Heidelberg: Springer Berlin Heidelberg, 1970. ISBN: 978-3-540-04903-6. DOI: 10.1007/bfb0058988.

Seokho Jin, Jongryul Lim and Subong Lim. 'Hecke Bound of Vector-valued Modular Forms and its Relationship with Cuspidality'. In: <i>Taiwanese Journal</i> <i>of Mathematics</i> 22.2 (2018), pp. 301–311. ISSN: 10275487, 22246851. URL: https://www.jstor.org/stable/90020209 (visited on 14/08/2024).
<ul> <li>Neil J. A. Sloane John Horton Conway. Sphere Packings, Lattices and Groups.</li> <li>3ed. Grundlehren der mathematischen Wissenschaften v. 290. Springer Science</li> <li>&amp; Business Media, 1998. ISBN: 0387985859, 9780387985855.</li> </ul>
H. Jacquet and J. A. Shalika. 'On Euler Products and the Classification of Automorphic Representations I'. In: <i>American Journal of Mathematics</i> 103.3 (June 1981), pp. 499–558. ISSN: 00029327, 10806377. DOI: 10.2307/2374103. URL: http://www.jstor.org/stable/2374103.
Paul Kiefer. 'Orthogonal Eisenstein Series of Singular Weight'. PhD thesis. TU Darmstadt, Oct. 2021.
Paul Kiefer. 'Orthogonal Eisenstein series at harmonic points and modular forms of singular weight'. In: <i>Mathematische Zeitschrift</i> 301.3 (Mar. 2022), pp. 2841–2882. ISSN: 1432-1823. DOI: 10.1007/s00209-022-03011-w.
Yoshiyuki Kitaoka. <i>Arithmetic of Quadratic Forms</i> . Cambridge Tracts in Mathematics. Cambridge University Press, Aug. 1999. ISBN: 052164996X,9780521649964. DOI: 10.1017/cbo9780511666155.
Max Koecher and Aloys Krieg. <i>Elliptische Funktionen und Modulformen</i> Berlin Heidelberg New York: Springer Berlin Heidelberg, 1998. ISBN: 978-3- 662-07012-3. DOI: 10.1007/978-3-662-07012-3.
Stephen S. Kudla and John J. Millson. 'The theta correspondence and har- monic forms. I'. In: <i>Mathematische Annalen</i> 274.3 (Sept. 1986), pp. 353–378. ISSN: 1432-1807. DOI: 10.1007/bf01457221.
Stephen S. Kudla and John J. Millson. 'The theta correspondence and har- monic forms. II'. In: <i>Mathematische Annalen</i> 277.2 (June 1987), pp. 267–314. ISSN: 1432-1807. DOI: 10.1007/bf01457364.
Stephen S. Kudla and John J. Millson. 'Intersection numbers of cycles on locally symmetric spaces and fourier coefficients of holomorphic modular forms

- [Kna92] Anthony W. Knapp. Elliptic Curves. (MN-40), Volume 40. Kassel: Princeton University Press, June 1992. ISBN: 0-691-08559-5. DOI: 10.2307/j. ctv346st5.
- [Kne02] Rudolf Kneser Martin; Scharlau. Quadratische Formen. Berlin Heidelberg New York: Springer Berlin Heidelberg, 2002. ISBN: 978-3-540-64650-1. DOI: 10.1007/978-3-642-56380-5.
- [Kne65] Martin Kneser. 'Strong approximation'. In: Algebraic groups and discontinuous subgroups. Vol. 9. Proceedings of Symposia in pure mathematics - symposium on algebraic groups. Boulder, Colorado: American mathematical society, 1965, pp. 187–196. ISBN: 0-8218-1409-5.
- [Kob12] Neal Koblitz. p-adic Numbers, p-adic Analysis, and Zeta-Functions -. Berlin Heidelberg: Springer Science & Business Media, 2012. ISBN: 978-1-461-21112-9.
- [Koh87] W. Kohnen. 'A simple remark on eigenvalues of Hecke operators on Siegel modular forms'. In: Abhandlungen aus dem Mathematischen Seminar der Universität Hamburg 57.1 (Dec. 1987), pp. 33–35. ISSN: 1865-8784. DOI: 10.1007/bf02941597. URL: https://doi.org/10.1007/BF02941597.
- [KR88] Stephen S. Kudla and Stephen Rallis. 'On the Weil-Siegel formula.' In: Journal für die reine und angewandte Mathematik 387 (1988), pp. 1–68. URL: http: //eudml.org/doc/153025.
- [KR92] Stephen S. Kudla and Stephen Rallis. 'Ramified degenerate principal series representation for Sp(n)'. In: Israel Journal of Mathematics 78.2–3 (Oct. 1992), pp. 209–256. ISSN: 1565-8511. DOI: 10.1007/bf02808058.
- [KR94] Stephen S. Kudla and Stephen Rallis. 'A Regularized Siegel-Weil Formula: The First Term Identity'. In: *The Annals of Mathematics* 140.1 (July 1994), pp. 1–80. ISSN: 0003-486X. DOI: 10.2307/2118540. URL: http://www.jstor.org/stable/2118540.
- [KRY04] Stephen S. Kudla, Michael Rapoport and Tonghai Yang. 'Derivatives of Eisenstein series and Faltings heights'. In: Compositio Mathematica 140 (04 July 2004). DOI: 10.1112/s0010437x03000459.
- [KRY06] Stephen S. Kudla, Michael Rapoport and Tonghai Yang. Modular Forms and Special Cycles on Shimura Curves. (AM-161). Kassel: Princeton University Press, Dec. 2006. ISBN: 9781400837168. DOI: 10.1515/9781400837168.

[KRY99]	Stephen S. Kudla, Michael Rapoport and Tonghai Yang. 'On the deriva-
	tive of an Eisenstein Series of Weight One'. In: International Mathemat-
	ics Research Notices 7.7 (1999), p. 347. ISSN: 1073-7928. DOI: 10.1155/
	s1073792899000185.

- [Kud02] Stephen S. Kudla. 'Derivatives of Eisenstein series and generating functions for arithmetic cycles'. en. In: Séminaire Bourbaki : volume 1999/2000, exposés 865-879. Astérisque 276. talk:876. Société mathématique de France, 2002, pp. 341-368. URL: http://www.numdam.org/item/SB\_1999-2000\_42\_341\_0.
- [Kud03] Stephen S. Kudla. 'Integrals of Borcherds Forms'. In: Compositio Mathematica 137.3 (July 2003), pp. 293–349. ISSN: 0010-437X. DOI: 10.1023/a: 1024127100993. URL: https://doi.org/10.1023/A:1024127100993.
- [Kud04a] Stephen S. Kudla. 'Special Cycles and Derivatives of Eisenstein Series'. In: *Heegner Points and Rankin L-Series*. Cambridge University Press, June 2004, pp. 243–270. DOI: 10.1017/cbo9780511756375.009.
- [Kud04b] Stephen S. Kudla. 'Tate's Thesis'. In: An Introduction to the Langlands Program. Ed. by Joseph Bernstein and Stephen Gelbart. Boston, MA: Birkhäuser Boston, 2004. Chap. Tate's Thesis, pp. 109–131. ISBN: 978-0-8176-8226-2. DOI: 10.1007/978-0-8176-8226-2\_6. URL: https://doi.org/10.1007/978-0-8176-8226-2\_6.
- [Kud94] Stephen Kudla. 'Splitting metaplectic covers of dual reductive pairs'. In: Israel Journal of Mathematics 87.1–3 (Feb. 1994), pp. 361–401. ISSN: 1565-8511. DOI: 10.1007/bf02773003.
- [Kud96] Stephen S. Kudla. *Notes on the local theta correspondence*. Lecture Notes of the European School on Group Theory. Schloss Hirschberg, 1996.
- [Kud97a] Stephen S. Kudla. 'Algebraic cycles on Shimura varieties of orthogonal type'. In: Duke Mathematical Journal 86.1 (Jan. 1997), pp. 39–78. ISSN: 0012-7094. DOI: 10.1215/s0012-7094-97-08602-6. URL: https://doi.org/10.1215/ S0012-7094-97-08602-6.
- [Kud97b] Stephen S. Kudla. 'Central derivatives of Eisenstein series and height pairings'. In: The Annals of Mathematics 146.3 (Nov. 1997), pp. 545-646. ISSN: 0003-486X. DOI: 10.2307/2952456. URL: https://www.jstor.org/stable/2952456.

- [Küm16] Burkhard Kümmerer. Wie man mathematisch schreibt Sprache Stil -Formeln. Berlin Heidelberg New York: Springer Fachmedien Wiesbaden, 2016.
   ISBN: 978-3-658-01576-3. DOI: 10.1007/978-3-658-01576-3.
- [KY10] Stephen S. Kudla and TongHai Yang. 'Eisenstein series for SL(2)'. In: Science China Mathematics 53.9 (9 Sept. 2010), pp. 2275–2316. ISSN: 1869-1862. DOI: 10.1007/s11425-010-4097-1.
- [Lam05] Tsit-Yuen Lam. Introduction to quadratic forms over fields. Vol. 67. Graduate Studies in Mathematics. American Mathematical Society, 2005. ISBN: 978-0-8218-1095-8. URL: https://bookstore.ams.org/view?ProductCode=GSM/ 67.
- [Lam73] Tsit-Yuen Lam. The algebraic theory of quadratic forms. Advanced book program Mathematics lecture note series. New York, USA: W. A. Benjamin, 1973. ISBN: 0-805-35664-3, 0-805-35665-1. URL: http://digitale-objekte. hbz-nrw.de/webclient/DeliveryManager?pid=1943434&custom\_att\_2= simple\_viewer.
- [Lan02] Serge Lang. Algebra. 3rd rev. ed. 2002. Corr. 4th printing 2005. Berlin Heidelberg: Springer Science & Business Media New York, 2002. ISBN: 978-0-387-95385-4. DOI: 10.1007/978-1-4613-0041-0.
- [Lan06] Robert P. Langlands. On the Functional Equations Satisfied by Eisenstein Series -. Berlin, Heidelberg: Springer Berlin Heidelberg, 2006. ISBN: 978-3-540-38070-2. DOI: 10.1007/bfb0079929.
- [Lan13a] Serge Lang. Algebraic Number Theory. Berlin Heidelberg: Springer Science & Business Media New York, 2013. ISBN: 978-1-461-20853-2. DOI: 10.1007/978-1-4612-0853-2.
- [Lan13b] Serge Lang. *Linear Algebra*. 3. Aufl. Berlin Heidelberg: Springer Science & Business Media New York, 2013. ISBN: 978-1-475-71949-9. DOI: 10.1007/978-1-4757-1949-9.
- [Lan98] Serge Lang. SL2(R). 1st ed. 1975. Corr. 2nd printing 1998. Berlin Heidelberg: Springer Science & Business Media, 1998. ISBN: 0-387-96198-4. DOI: https: //doi.org/10.1007/978-1-4612-5142-2.
- [Li74] Wen-Ch'ing Winnie Li. 'Newforms and Functional Equations.' In: Mathematische Annalen 212 (1974), pp. 285–316. URL: http://eudml.org/doc/182749.
- [LMFDB] The LMFDB Collaboration. The L-functions and modular forms database. http://www.lmfdb.org. [Online; accessed 19 April 2022]. 2022.

[Lor06]	Falko Lorenz. <i>Quadratische Formen über Körpern</i> Berlin Heidelberg New York: Springer Berlin Heidelberg, 2006. ISBN: 978-3-540-36270-8. DOI: 10. 1007/bfb0069629.
[Lor08]	Falko Lorenz. <i>Lineare Algebra I</i> 4th ed. Heidelberg: Spektrum Akademischer Verlag, 2008. ISBN: 978-3-8274-1406-9. URL: https://link.springer.com/book/9783827414069.
[Lor92]	Falko Lorenz. <i>Lineare Algebra II</i> 3rd ed. Heidelberg: Spektrum Akademischer Verlag, 1992. ISBN: 978-3-86025-457-8. URL: https://link.springer.com/book/9783860254578.
[LV80]	Gérard Lion and Michèle Vergne. <i>The Weil representation, Maslov index and Theta series.</i> Progress in Mathematics. Bosten, Basel Stuttgart: Birkhaeuser, 1980. ISBN: 3-7643-3007-4. DOI: 10.1007/978-1-4684-9154-8.
[Mac49]	George Whitelaw Mackey. 'A Theorem of Stone and von Neumann'. In: <i>Duke Mathematical Journal</i> 16.2 (June 1949), pp. 313–326. ISSN: 0012-7094. DOI: 10.1215/S0012-7094-49-01631-2.
[Mac58]	George Whitelaw Mackey. 'unitary representations of group extensions'. In: Acta mathematica 99 (June 1958), pp. 266–311.
[Mac64]	George Whitelaw Mackey. 'some remarks about symplectic automorphisms'. In: $AMS$ 32 (1964), pp. 393–397.
[Mac76]	George Whitelaw Mackey. <i>The Theory of Unitary Group Representations</i> . Boston, Basel, Stuttgart: University of Chicago, 1976. ISBN: 0-226-50051-7.
[McG03]	William J. McGraw. 'The rationality of vector valued modular forms associated with the Weil representation'. In: <i>Mathematische Annalen</i> 326.1 (Apr. 2003), pp. 105–122. ISSN: 1432-1807. DOI: 10.1007/s00208-003-0413-1.
[Met19]	Ingmar Metzler. 'Coefficients of Eisenstein series on SL2 and related Whittak- erfunctions'. MA thesis. Technische Universität Darmstadt, 2019.
[MH73]	John Willard Milnor and Dale Husemöller. <i>Symmetric Bilinear Forms</i> . Vol. Band 73. Ergebnisse der Mathematik und ihrer Grenzgebiete. Berlin: Springer Berlin Heidelberg, 1973. ISBN: 354006009X. DOI: 10.1007/978-3-642-88330-9. URL: http://scans.hebis.de/HEBCGI/show.pl?07476044_toc.pdf.
[Mil17]	James S. Milne. <i>Algebraic Number Theory (v3.07).</i> 2017. URL: www.jmilne. org/math/.

- [Miy06] Toshitsune Miyake. Modular Forms. Berlin Heidelberg: Springer Science & Business Media, 2006. ISBN: 978-3-540-29593-8. DOI: 10.1007/3-540-29593-3.
- [Miy71] Toshitsune Miyake. 'On Automorphic Forms on GL 2 and Hecke Operators'. In: The Annals of Mathematics 94.1 (July 1971), pp. 174–189. ISSN: 0003-486X. DOI: 10.2307/1970741. URL: http://www.jstor.org/stable/1970741.
- [Mül24] Manuel K. -H. Müller. The basis problem for modular forms for the Weil representation. 2024. DOI: 10.48550/ARXIV.2407.01205.
- [Mur93] M. Ram Murty, ed. Theta functions From the Classical to the Modern.
   Vol. 1. CRM Proceedings and Lecture Notes. 1993. ISBN: 0-821-87011-4. URL: https://api.semanticscholar.org/CorpusID:59975241.
- [MZ23] Ingmar Metzler and Riccardo Zuffetti. Injectivity of the genus 1 Kudla-Millson lift on locally symmetric spaces. Dec. 2023. DOI: 10.48550/ARXIV.2312. 00572. arXiv: 2312.00572 [math.NT].
- [Neu06] Jürgen Neukirch. Algebraische Zahlentheorie -. Berlin Heidelberg New York: Springer-Verlag, 2006. ISBN: 978-3-540-37663-7. DOI: 10.1007/978-3-540-37663-7.
- [Neu29] J. Neumann. 'Zur allgemeinen Theorie des Masses'. ger. In: Fundamenta Mathematicae 13.1 (1929), pp. 73–116. URL: http://eudml.org/doc/211921.
- [Neu31] John von Neumann. 'Die Eindeutigkeit der Schrödingerschen Operatoren'.
   In: Mathematische Annalen 104.1 (Dec. 1931), pp. 570–578. ISSN: 1432-1807.
   DOI: https://doi.org/10.1007/BF01457956.
- [Nik80] Carl Ludwig Nikulin. 'Integral symmetric bilinear forms and some of their applications'. In: *Mathematics of the USSR-Izvestiya* 14.1 (Feb. 1980), pp. 103–167. ISSN: 0025-5726. DOI: 10.1070/im1980v014n01abeh001060.
- [Olv+10] Frank W. Olver et al. *NIST Handbook of Mathematical Functions*. 1st. USA: Cambridge University Press, 2010. ISBN: 978-0-521-14063-8.
- [OMe73] Onorato Timothy OMeara. Introduction to Quadratic Forms -. 3rd ed. Berlin, Heidelberg: Springer, 1973. ISBN: 3-540-02984-2, 0-387-02984-2. DOI: 10. 1007/978-3-642-62031-7.
- [Opi18] Sebastian Opitz. 'Computation of Eisenstein series associated with discriminant forms'. PhD thesis. TU Darmstadt, 2018.

[OR]	J J O'Connor and E F Robertson. <i>Mathhistory bibliographies</i> . URL: https://mathshistory.st-andrews.ac.uk/Biographies/Borcherds/.
[Osb75]	Scott M. Osborne. 'On the Schwartz-Bruhat Space and the Paley-Wiener Theorem for Locally Compact Abelian Groups'. In: <i>Journal of Functional Analysis</i> 19.1 (May 1975), pp. 40–49. ISSN: 0022-1236. DOI: 10.1016/0022-1236(75)90005–1.
[Pra93]	Dipendra Prasad. 'Weil Representation, Howe Duality, and the Thetacor- respondence'. In: <i>CRM Proceedings and Lecture Notes.</i> Vol. 1. Centre de Recherches Mathématiques. American Mathematical Society, 1993, pp. 105– 127.
[PRR93]	Vladimir Platonov, Andrei Rapinchuk and Rachel Rowen. <i>Algebraic Groups and Number Theory</i> Amsterdam, Boston: Cambridge University Press, Aug. 1993. ISBN: 978-0-080-87459-3. DOI: 10.1017/9781139017756.
[Ran39a]	R. A. Rankin. 'Contributions to the theory of Ramanujan's function (n) and similar arithmetical functions I'. In: <i>Mathematical Proceedings of the Cambridge Philosophical Society</i> 35.3 (July 1939), pp. 351–356. ISSN: 1469-8064. DOI: 10.1017/s0305004100021095.
[Ran39b]	R. A. Rankin. 'Contributions to the theory of Ramanujan's function (n) and similar arithmetical functions II'. In: <i>Mathematical Proceedings of the Cambridge Philosophical Society</i> 35.3 (July 1939), pp. 357–372. ISSN: 1469-8064. DOI: 10.1017/s0305004100021101.
[Ran40]	R. A. Rankin. 'Contributions to the theory of Ramanujan's function (n) and similar arithmetical functions: iii. a note on the sum function of the fourier coefficients of integral modular forms'. In: <i>Mathematical Proceedings of the Cambridge Philosophical Society</i> 36.2 (Apr. 1940), pp. 150–151. ISSN: 1469-8064. DOI: 10.1017/s0305004100017114.
[Rao93]	<ul><li>R. Ranga Rao. 'On some explicit formulas in the theory of Weil representation.'</li><li>In: <i>Pacific Journal of Mathematics</i> 157.2 (1993), pp. 335–371.</li></ul>
[RK88]	Stephen Rallis and Stephen S. Kudla. 'On the Weil-Siegel formula II.' In: Journal für die reine und angewandte Mathematik 391 (1988), pp. 65–84. URL: http://eudml.org/doc/153070.
[Rud90]	Walter Rudin. <i>Fourier Analysis on Groups</i> . New York: Wiley-Interscience publication, Jan. 1990. ISBN: 9780470744819. DOI: 10.1002/9781118165621.

370

- [Sch04] J. Schur. 'Über die Darstellung der endlichen Gruppen durch gebrochen lineare Substitutionen.' In: Journal für die reine und angewandte Mathematik 127 (1904), pp. 20–50. URL: http://eudml.org/doc/149177.
- [Sch09] Nils Scheithauer. 'The Weil Representation of SL2(Z) and Some Applications'. In: International Mathematics Research Notices - INT MATH RES NOTICES 2009.8 (Feb. 2009), pp. 1488–1545. ISSN: 1073-7928. DOI: 10.1093/imrn/ rnn166.
- [Sch15] Nils R. Scheithauer. 'Some constructions of modular forms for the Weil representation of SL<sub>2</sub>(Z)'. In: Nagoya Mathematical Journal 220.none (Dec. 2015), pp. 1–43. ISSN: 0027-7630. DOI: 10.1215/00277630-3335405. URL: https://doi.org/10.1215/00277630-3335405.
- [Sch39] B. Schoeneberg. 'Das Verhalten von mehrfachen Thetareihen bei Modulsubstitutionen'. In: Mathematische Annalen 116 (1939), pp. 511–523. URL: http://eudml.org/doc/160014.
- [SE] Deepl SE. *Deepl*. used for occasional optimisation of text style. URL: https: //www.deepl.com.
- [Seg63] I. E. Segal. 'Transforms for Operators and Symplectic Automorphisms over a Locally Compact Abelian Group'. In: Math. scand. 13 (Sept. 1963), pp. 31–43.
   ISSN: 0025-5521. DOI: 10.7146/math.scand.a-10686.
- [Sel65] Atle Selberg. 'On the estimation of Fourier coefficients of modular forms'. In: Proc. Sympos. Pure Math., Vol. VIII. American Mathematical Society, Providence, R.I., 1965, pp. 1–15.
- [Ser12a] Jean-Pierre Serre. A Course in Arithmetic. Berlin Heidelberg: Springer Science & Business Media, 2012. ISBN: 978-1-468-49884-4. DOI: 10.1007/978-1-4684-9884-4.
- [Ser12b] Jean-Pierre Serre. Linear Representations of Finite Groups. Berlin, Heidelberg: Springer Science & Business Media, 2012. ISBN: 978-1-468-49458-7. DOI: 10.1007/978-1-4684-9458-7.
- [Sha74] J. A. Shalika. 'The Multiplicity One Theorem for GLn'. In: The Annals of Mathematics 100.2 (Sept. 1974), pp. 171–193. ISSN: 0003-486X. DOI: 10.2307/1971071. URL: http://www.jstor.org/stable/1971071.
- [Shi71] Goro Shimura. Introduction to the Arithmetic Theory of Automorphic Functions -. Kassel: Princeton University Press, 1971. ISBN: 9780691080925.

[Shi73]	<ul> <li>Goro Shimura. 'On Modular Forms of Half Integral Weight'. In: <i>The Annals of Mathematics</i> 97.3 (May 1973), pp. 440-481. ISSN: 0003-486X. DOI: 10. 2307/1970831. URL: http://www.jstor.org/stable/1970831.</li> </ul>
[Shi75]	Goro Shimura. 'On the Holomorphy of Certain Dirichlet Series'. In: <i>Proceedings</i> of the London Mathematical Society s3-31.1 (July 1975), pp. 79–98. ISSN: 0024-6115. DOI: 10.1112/plms/s3-31.1.79.
[Sie35]	Carl Ludwig Siegel. 'Über die analytische Theorie der quadratischen Formen'. In: <i>The Annals of Mathematics</i> 36.3 (July 1935), pp. 527–606. ISSN: 0003-486X. DOI: 10.2307/1968644.
[Sie36]	Carl Ludwig Siegel. 'Über die Analytische Theorie der Quadratischen Formen II'. In: <i>The Annals of Mathematics</i> 37.1 (Jan. 1936), p. 230. ISSN: 0003-486X. DOI: 10.2307/1968694.
[Sie37]	Carl Ludwig Siegel. 'Über die Analytische Theorie der Quadratischen Formen III'. In: <i>The Annals of Mathematics</i> 38.1 (Jan. 1937), p. 212. ISSN: 0003-486X. DOI: 10.2307/1968520.
[Sie51a]	Carl Ludwig Siegel. 'Indefinite quadratische Formen und Funktionentheorie I'. In: <i>Mathematische Annalen</i> 124.1 (Dec. 1951), pp. 17–54. ISSN: 1432-1807. DOI: 10.1007/bf01343549.
[Sie51b]	Carl Ludwig Siegel. 'Indefinite quadratische Formen und Funktionentheorie II'. In: <i>Mathematische Annalen</i> 124.1 (Dec. 1951), pp. 364–387. ISSN: 1432-1807. DOI: 10.1007/bf01343576.
[Sil13]	Joseph H. Silverman. <i>The Arithmetic of Elliptic Curves</i> . Berlin Heidelberg: Springer Science & Business Media, 2013. ISBN: 978-1-475-71920-8. DOI: 10.1007/978-1-4757-1920-8.
[Ste15]	Oliver Stein. 'The Fourier expansion of Hecke operators for vector-valued modular forms'. In: <i>Functiones et Approximatio Commentarii Mathematici</i> 52.2 (2015), pp. 229–252. DOI: 10.7169/facm/2015.52.2.4. URL: https://doi.org/10.7169/facm/2015.52.2.4.
[Ste21]	Oliver Stein. 'Correction and addendum to: The Fourier expansion of Hecke operators for vector-valued modular forms'. In: <i>Functiones et Approximatio</i> <i>Commentarii Mathematici</i> 1.1 (Jan. 2021). ISSN: 0208-6573. DOI: 10.7169/ facm/1969. URL: https://arxiv.org/abs/2108.06549.
[Ste23]	Oliver Stein. A converse theorem for Borcherds products and the injectivity of the Kudla-Millson theta lift. 2023. DOI: 10.48550/ARXIV.2306.17660.

- [Sto30] Marshall Harvey Stone. 'Linear transformations in Hilbert space III operational methods and group theory'. In: Proceedings of the National Academy of Sciences of the United States of America. Vol. 16. 2. Feb. 1930, pp. 172–175. URL: https://www.jstor.org/stable/85485.
- [Str13] Fredrik Strömberg. 'Weil representations associated with finite quadratic modules'. In: Mathematische Zeitschrift 275.1 (Mar. 2013), pp. 509–527. ISSN: 1432-1823. DOI: 10.1007/s00209-013-1145-x.
- [Tal13] Nicola L. C. Talbot. Using LATEX to wirte a PhD thesis. https://www. dickimaw-books.com/latex/thesis/thesis-report.pdf. 2013.
- [Tao10] Terence Tao. Some notes on group extensions. https://terrytao.wordpress. com/2010/01/23/some-notes-on-group-extensions/. [Online; accessed 18-May-2018]. 2010.
- [Tat16] John Tate. Collected Works of John Tate. Part I: 1951-1975. Ed. by Jean-Pierre Serre Barry Mazur. AMS, 2016. ISBN: 9780821890929.
- [Tho15] Martin Thoma. Kartenwechsel Abbildung. https://github.com/MartinThoma/ LaTeX-examples/blob/master/tikz/topology-kartenwechsel/topologykartenwechsel.tex. [Online; accessed 09-July-2017]. 2015.
- [Völ18] Fabian Völz. 'Realizing Hyperbolic and Elliptic Eisenstein Series as Regularized Theta Lifts'. en. PhD thesis. Darmstadt: Technische Universität Darmstadt, Sept. 2018. URL: http://tuprints.ulb.tu-darmstadt.de/8144/.
- [Wai18] Fabian Waibel. 'Fourier coefficients of half-integral weight cusp forms and Waring's problem'. In: *The Ramanujan Journal* 47.1 (Aug. 2018), pp. 185–200. ISSN: 1572-9303. DOI: 10.1007/s11139-017-9934-1.
- [Wal63] C.T.C. Wall. 'Quadratic forms on finite groups, and related topics'. In: Topology 2.4 (1963), pp. 281-298. ISSN: 0040-9383. DOI: 10.1016/0040-9383(63) 90012-0. URL: https://www.sciencedirect.com/science/article/pii/ 0040938363900120.
- [Wed16] Torsten Wedhorn. Manifolds, Sheaves, and Cohomology. Berlin, Heidelberg: Springer Fachmedien Wiesbaden, 2016. ISBN: 978-3-658-10633-1. DOI: 10. 1007/978-3-658-10633-1.
- [Wei64] André Weil. 'Sur certains groupes d'opérateurs unitaires'. In: Acta Mathematica 111.0 (1964), pp. 143–211. ISSN: 0001-5962. DOI: 10.1007/bf02391012.

[Wei65]	André Weil. 'Sur la formule de Siegel dans la théorie des groupes classiques'. In: Acta Mathematica 113.0 (1965), pp. 1–87. ISSN: 0001-5962. DOI: 10.1007/ bf02391774.
[Wei82]	André Weil. <i>Adeles and algebraic groups</i> . Boston, Basel, Stuttgart: Birkhaeuser, 1982. ISBN: 3-7643-2092-9.
[Wei84]	Rainer Weissauer. 'Eisensteinreihen vom Gewicht $n + 1$ zur Siegelschen Modulgruppen-ten Grades'. In: <i>Mathematische Annalen</i> 268.3 (Sept. 1984), pp. 357–377. ISSN: 1432-1807. DOI: 10.1007/bf01457064.
[Wol]	Inc. Wolfram Research. <i>Mathematica</i> . Wolfram Research, Inc. URL: https://www.wolfram.com/mathematica.
[YOS79]	Hiroyuki YOSHIDA. 'Weil's representation of the symplectic groups over finite fields'. In: <i>Journal of the Mathematical Society of Japan</i> 31.2 (2 Apr. 1979), pp. 399–426. ISSN: 0025-5645. DOI: 10.2969/jmsj/03120399.
[Yos80]	Hiroyuki Yoshida. 'Siegel's Modular Forms and the Arithmetic of Quadratic Forms'. In: <i>Inventiones Mathematicae</i> 60.3 (04 Oct. 1980), pp. 193–248. ISSN: 1432-1297. DOI: 10.1007/bf01390016.
[Yos84]	H Yoshida. 'On Siegel modular forms obtained from theta series'. In: <i>Journal für die reine und angewandte Mathematik (Crelles Journal)</i> 352.352 (04 July 1984), pp. 184–219. ISSN: 1435-5345. DOI: 10.1515/crll.1984.352.184.
[Yos85]	H Yoshida. 'The action of Hecke operators on theta series'. In: Algebraic and topological theories - to the memory of Dr. T. Miyata (1985), pp. 197–238.
[Zag13]	Don Bernard Zagier. Zetafunktionen und quadratische Körper - Eine Ein- führung in die höhere Zahlentheorie. Berlin Heidelberg New York: Springer- Verlag, 2013. ISBN: 978-3-642-61829-1. DOI: 10.1007/978-3-642-61829-1.
[Zha09]	Wei Zhang. 'Modularity of Generating Functions of Special Cycles on Shimura Varieties'. PhD thesis. COLUMBIA UNIVERSITY, 2009.
[Zhu96]	Yongchang Zhu. 'Modular Invariance of Characters of Vertex Operator Algebras'. In: <i>Journal of the American Mathematical Society</i> 9.1 (1996), pp. 237–302. ISSN: 08940347, 10886834. URL: http://www.jstor.org/stable/2152847 (visited on 26/05/2024).

- [Zuf22] Riccardo Zuffetti. 'Cones of special cycles of codimension 2 on orthogonal Shimura varieties'. In: Transactions of the American Mathematical Society 375.10 (Aug. 2022), pp. 7385–7441. ISSN: 1088-6850. DOI: 10.1090/tran/ 8757.
- [Zuf24] Riccardo Zuffetti. 'Unfolding and injectivity of the Kudla–Millson lift of genus 1'. In: Mathematische Zeitschrift 307.1 (Apr. 2024). ISSN: 1432-1823. DOI: 10.1007/s00209-024-03479-8.

## Erklärungen

Ich versichere, dass die elektronische Version dieser Dissertation mit der vorliegenden schriftlichen Version übereinstimmt.

Ich versichere, noch nicht zu einem vorherigen Zeitpunkt eine Promotion versucht zu haben.

Die Dissertation ist von mir mit einem Verzeichnis aller benutzten Quellen versehen. Ich erkläre, dass ich die Arbeit – abgesehen von den in ihr ausdrücklich genannten Hilfen – selbstständig verfasst habe.

Des Weiteren versichere ich, die "Grundsätze zur Sicherung guter wissenschaftlicher Praxis an der Technischen Universität Darmstadt" und die "Leitlinien zum Umgang mit digitalen Forschungsdaten an der TU Darmstadt" in den jeweils aktuellen Versionen bei der Verfassung der vorliegenden Dissertation beachtet zu haben.

Ich versichere, dass die Arbeit bisher noch nicht zu Prüfungszwecken gedient hat.

29.08.2024, Darmstadt

Unterschrift